City of Sandy

Agenda Planning Commission Meeting Meeting Location: City Hall- Council Chambers, 39250 Pioneer Blvd., Sandy, Oregon 97055 Meeting Date: Monday, April 8, 2019 Meeting Time: 7:00 PM



4.1. Public Hearing

2.1.

4.2. 18-025 continuance

Staff recommends the Planning Commission hold a public hearing to take public testimony regarding the proposal. Staff recommends the Planning Commission **approve** the subdivision application with modifications as recommended in this report. The staff recommendation includes approval of the following three variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a two-tiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff is also recommending that the FSH analysis area be a separate tract owned and maintained by an HOA. This would result in Lots 19-21 being less than the minimum 7,500 square feet; thus, staff is recommending approval of a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum and allowing the rear setback to be reduced to 15 feet.

9 - 572

Page 1 of 572



Since the Planning Commission decided to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation decision.

18-025 SUB/VAR/FSH/TREE - Pdf

5. ITEMS FROM COMMISSION AND STAFF

6. ADJOURN

Sandy Planning Commission Regular Meeting Monday, February 25 2019

Chairman Jerry Crosby called the meeting to order at 7:00 p.m.

1. ROLL CALL

Commissioner Carlton – Present Commissioner Lesowski – Excused Commissioner MacLean Wenzel – Present Commissioner Logan – Present Commissioner Mobley – Present Chairman Crosby – Present

<u>Others present</u>: Planning & Building Director Kelly O'Neill Jr., Associate Planner Emily Meharg, City Attorney David Doughman (via telephone), Planning Assistant Rebecca Casey

2. REQUESTS FROM THE FLOOR – CITIZEN COMMUNICATION ON NON-AGENDA ITEMS None

NEW BUSINESS

3. PUBLIC HEARING – Type III Subdivision for Jacoby Heights Subdivision (18-025 SUB/VAR/TREE/FSH/INT) Chairman Crosby opened the public hearing on File No. 18-025 SUB/VAR/TREE/FSH/INT (Jacoby Heights Subdivision) at 7:01 p.m. Crosby noted that this is a quasi-judicial public hearing. He called for any abstentions, conflicts of interest, ex-parte contact, challenges to the jurisdiction of the Planning Commission, or any challenges to any individual member of the Planning Commission.

Commissioner Mobley recused himself as he is the owner of Lancaster Engineering who conducted the traffic impact study for the applicant.

Commissioner MacLean-Wenzel stated she came close to having ex-parte contact when she attended the tree committee meeting last week. She explained that "luckily" she was stopped before the conversation went further. Chairman Crosby asked the remaining Commissioners if they had any issues over MacLean-Wenzel's explanation and none did.

With declarations noted, Crosby went over the public hearing procedures for a quasi-judicial public hearing and called for the staff report.

Staff Report:

Associate Planner Emily Meharg summarized the staff report and addressed the background, factual information, public comments staff received, applicable criteria, and went over a brief slide show. Meharg finished her report with the summary and conclusion and staff's recommendation.

City Attorney David Doughman explained to the Commission the Code Interpretation is part of the application.

Commissioner Carlton asked Doughman if the definition of "all zones" includes zones such as R1, R2, SFR, etc. Doughman said yes, the definition of "all zones" references every City zone.

<u>Applicant Presentation:</u> Tracy Brown, 17075 Fir Dr., Sandy, OR 97055

Mr. Brown gave his presentation on behalf of the applicant by starting with background information and their response to the Code Interpretation regarding Chapter 17.92.10 (c) (Landscaping & Screening). Brown requested and was approved to discuss the Code Interpretation first as its outcome will have implications on staff's other recommended conditions in the staff report.

Brown handed out a document to both staff and the Commission that included the applicant's response to staff's recommended conditions.

Brown said his main contention is that Chapter 17.92.10 (c) is intended for landscaping which is also the title of the chapter. He said that staff is using this chapter to try to increase the number of trees to retain and that staff is not applying this chapter correctly. Brown said that the Urban Forestry Ordinance (Chapter 17.102) was the intended Ordinance to regulate tree harvesting in the City. Brown continued to dissect Chapter 17.92. and Chapter 17.102 and what he believed the intent of these chapters are.

Brown said that staff will "run-a-muck" if allowed in preserving trees by trying to "switch" the language of "should be preserved" to "shall be preserved".

Brown also stated, "I would wager that there's nobody in the City that's planted more trees than I have or instituted more tree planting programs in this City". He then finished by stating that in regard to City staff "the left hand doesn't know what the right is doing".

Ray Moore, All County Surveyors and Planners, PO Box 955, Sandy, OR 97055 Mr. Moore also mentioned Brad Picking's letter that was given to staff just prior to the meeting and was handed out to the Commission. Mr. Moore said that Mr. Picking was part of the committee that assisted in writing Chapter 17.102 (Urban Forestry) and said from what Mr. Picking understood Chapter 17.92 was not applicable to residential subdivisions.

Mr. Moore requested that the Committee not "muddy" the waters at this time and wait until the process of reviewing the codes is complete.

Chairman Crosby asked City Attorney David Doughman if the Commission could have a discussion and make a decision based only on the Code Interpretation. Doughman said that would be fine and encouraged the public to make comments during this time regarding just the Code Interpretation.

Chairman Crosby followed up and clarified that at this time the Commission will move forward with the Code Interpretation issue first which will allow for Proponent and Opponent Testimony. The rest of the application will be discussed after a motion is made on the Code Interpretation.

Proponent Testimony: (In favor of staff's recommendation) None

Opponent Testimony:

Mac Even, 5360 SE Chase Rd, Sandy, OR 97055

Mr. Even said he wanted to be on the record and is currently working with the applicant on the Jacoby Heights Subdivision. Mr. Even explained that the decision will also affect other property he is considering developing that is an overgrown Christmas tree farm. He said if this code is applied to subdivisions then he will be stumped as to what he will be able to do with the property.

Mr. Even said he's on the City's Tree Code Committee to find a solution and because he sees the

two different sides (staff and applicant). Until there are code changes, he would like to see staff stay with what was established in the past.

Buzz Ortiz, 41525 SE Vista Loop, Sandy, OR 97055

Mr. Ortiz said he would like code clarification as he will be looking at possible development in the near future.

Bonnie Drake, 19270 SE Jacoby Rd, Sandy, OR 97055

Mrs. Drake said she would like code clarification as her property was recently annexed. She also stated that she is against the use of Chapter 17.92 for subdivisions.

Testimony:

Ryan Newman, Portland Tree Company, 4933 SE Henry St., Portland, OR 97206

Mr. Newman said he is the project Arborist hired by the applicant, but he does not want to be for or against this application. In his opinion it should be a "hard rule" that states the limit of what size trees need to be inventoried.

Tim Anders, 19098 SE Jacoby, Sandy, OR 97055

Mr. Anders said his property borders the applicant's property on the north side. He said that over 10 years ago he had to inventory over 600 trees and removed 35 trees to build his house.

Planning & Building Director Kelly O'Neill Jr. said it sounds like the interpretation back then required Mr. Anders to inventory a lot more trees than just 11-inch trees. Mr. Anders agreed and said he did have to inventory hundreds of trees, but he didn't know what interpretation was used during that time.

Staff Recap:

Associate Planner Emily Meharg said that not all of the 60 + trees the applicant has proposed to retain are in good condition and said the applicant is already going to remove over 450 trees.

Meharg explained that good standard practice is to have more than the absolute minimum of three trees per acre as there are many times the retention trees get damaged or knocked down by the wind. Meharg said that staff is not being unpractical or unrealistic by how they are interpreting the code. She explained that staff is working with what was written in the code under Chapter 17.92 and it never mentions "not applying to subdivisions" just as chapter 17.102 doesn't specifically say it does apply to subdivisions.

Planning and Building Director Kelly O'Neill Jr. followed up and said the bottom line is that the tree standards in Chapter 17.102 do not work at all for subdivisions and that is why Council created the Tree Code Committee. He explained that applying Chapter 17.102 as the only guiding code chapter would be a bad thing and said by also applying Chapter 17.92 it gives the applicant more flexibility with trees that are smaller than 11 inches in diameter.

O'Neill also said that typically when you adopt code regulations and want to exempt a certain chapter then somewhere in the Ordinance would state the exemption. There was nothing in the adopted ordinance stating that Chapter 17.92 is not applicable to subdivisions.

O'Neill stated that after forming the new Tree Code Committee he hopes to make these chapters much clearer.

Discussion:

Commissioner MacLean-Wenzel asked staff if they added the 8-inch trees in diameter to allow for diversity if there are not enough 11-inch trees in diameter to choose from. Meharg said the Code does not state that specifically, but it's how staff is applying the code to provide flexibility. Meharg talked about the Marshall Ridge subdivision as an example. She said they had to apply for a tree

variance, but they also had many 8-inch trees in good condition that staff allowed the applicant to count to meet the retention requirement.

Applicant Recap:

Tracy Brown, 17075 Fir Dr., Sandy, OR 97055

Mr. Brown said that he did a very exhaustive look at Chapter 17.92 and did not find any language that would be applicable to subdivisions since Chapter 17.100 addresses street trees. Brown stated that since subdivisions don't have landscape plans, Chapter 17.92 should not be applicable.

Staff Response:

O'Neill responded to Brown and said that almost all subdivisions have landscape plans as they have detention ponds, pedestrian tracts, and/or parking courts all within the subdivision. He explained that there are several cases such as Pioneer Meadows who prior to final plat had to have a landscape plan for their parking court and this was based on Chapter 17.92.

Doughman said it's good to recognize that Chapter 17.92 and Chapter 17.102 are not clear regarding tree preservation.

O'Neill told the Commission that if they agree with the applicant's Code Interpretation request then it should be very clear. He explained that if the Commission takes away Chapter 17.92 for staff to use in Subdivisions, then staff would have no way of enforcing or monitoring landscape plans in tracts.

Discussion:

Commissioner Carlton said he is frustrated and would like to see these conflicts addressed in the upcoming code changes and in a timely manner. He would like to see the code changes complete and in front of Council within three months.

Commissioner Logan said Mr. Brown makes a compelling argument.

Commissioner MacLean-Wenzel stated that over the last 17 years the knowledge base has changed as well as what people know about trees and tree retention.

Chairman Crosby said it would be difficult for him to say Chapter 17.92 does not apply to all zones when it states "All Zones" in the Chapter title. Although, he said there is enough testimony and evidence to show there is some confusion. While throwing out ideas, Crosby said in the Motion they could state that Chapter 17.92.10 (c) does not apply to subdivisions but retain everything else in the Chapter. Crosby said staff could then reference Chapter 17.102 for subdivisions realizing that it wouldn't solve the problem forever.

Motion: Code Interpretation - Although Chapter 17.92 clearly states in its title, "All Zones," the chart in the chapter does not list all zones. Chapter 17.102 is to be applied to subdivisions, in place of 17.92.10 (C), in regard to tree retention." Moved by: Commissioner Carlton Seconded by: Commissioner Logan Yes votes: Commissioner Carlton, Maclean-Wenzel, Logan and Chairman Crosby No votes: None Abstentions: Commissioner Mobley The motion Passed.

After a short break Chairman Crosby asked the applicant to continue with their presentation.

Mr. Brown stated that the applicant agrees with staff's recommendations on the variances, but has concerns with some of the conditions. Before continuing his presentation, Mr. Brown handed

out their response to the list of staff's recommended conditions.

Mr. Brown addressed the Flood Slope Hazard (FSH) overlay along with the trail conditions and tree conditions. He then introduced Arborist Ryan Newman to again address the tree protection fencing in the FSH overlay.

Mr. Newman explained how they will protect and preserve the trees in the FSH. He agreed with staff's request to have protection on these trees 5 feet beyond the drip line and said in some municipalities they require one foot per inch of trunk diameter. Mr. Newman said the trees should be preserved in groups instead of individually as they would be more viable. Mr. Newman believes that even though there is some decay in the trees in the FSH, saving these trees preserves the characteristics of the site in the FSH Overlay.

Commissioner Maclean-Wenzel asked Mr. Newman about the type of tree protection fencing and he explained to the Commission they will be using a 6-foot-high metal fence.

Mr. Brown addressed staff's recommendation to have the applicant dedicate all the FSH overlay to the City. He considers that recommendation "a taking of the applicant's property". Alternatively, he said the applicant would consider a conservation easement, or tree protection easement.

Mr. Brown moved on to Lot 19 and again explained to the Commission that staff is requesting that Lot 19 be saved for tree preservation. He believes that Lot 19 is worth roughly \$130,000 and would like to know if the applicant will be compensated for the taking of that lot. After stating this to the Commission he asked to have this condition removed and said, "It constitutes a taking of the applicant's property without any compensation".

Ray Moore, All County Surveyors and Planners, PO BOX 955, **Sandy**, **OR 97055** Mr. Moore discussed future street plans, access to the FSH on the applicant's property as well as a possible future trail.

Mr. Brown continued his presentation and started addressing the 79 conditions in the staff report. At condition #13, Commissioner Carlton suggested the applicant identify and address just the "main issues" that are a "big deal" and work with staff on the remaining concerns. Staff and the Commission both agreed that reading every staff condition along with the applicant's response will take at least another 2-3 hours. Mr. Brown said, "I don't know how else to do it unless you send staff back to the drawing board". Following this statement Mr. Brown continued and jumped to staff's condition #27 in the staff report addressing sidewalks and then said he was almost done with his presentation. Mr. Brown then covered condition #32 regarding the right-of-way, condition #35 and condition #78 regarding the fee-in-lieu.

Ray Moore, All County Surveyors and Planners, PO Box 955, Sandy, OR 97055

Mr. Moore covered condition #28 (shifting the street), #30 and #31 (extending the street to the property line), #34 (cul-de-sac radius), #38 (ADA ramp), #40 (vehicle non-access reserve VNAR), #48, #52 and #59 (Moore said its redundant as it's the same condition as #45 regarding sanitary sewer lines and should be removed), #49 (sewer laterals), and finished with condition #67 (cut and fill evaluation).

Planning and Building Director Kelly O'Neill Jr told the Commission that some of the conditions addressed by the applicant will need to be discussed in further detail with the City Engineer and City Attorney. O'Neill explained that staff just received the applicant's response to the staff report during the meeting when it was handed out, which allowed no time to review.

Commissioner Carlton suggested a continuance to the application and Chairman Crosby agreed. O'Neill asked City Attorney David Doughman if staff could request the continuance as they were just presented with multiple documents from the applicant. Doughman explained that staff could request the continuance, but the 120-day clock will continue to run where as if the applicant asks for a continuance the clock will stop.

The Commission, the applicant and staff decided on a date certain of April 8, 2019 to continue the hearing. The Applicant officially asked the Commission for a continuance in order to "toll the clock" to the April 8, 2019 meeting date. The Commission asked the applicant to please provide their presentation handouts before the meeting, as they have no time to review it at the meeting.

O'Neill asked the Commission how they would like to proceed in the next meeting for this application if Commissioner Lesowski is present along with potentially a new planning commissioner. Should or could they participate since they were not currently present. Doughman explained that if the other commissioners review the material and this current meeting, they could legally participate but asked if the applicant had any concerns over this. Mr. Brown said they are fine with this,but would like to reserve the right to contest those individual members at the meeting.

Motion: Move to continue the Public Hearing to Monday April 8, 2019 by request of the applicant. Moved By: Commissioner Carlton Yes votes: All Ayes No votes: None Abstentions: None The motion passed

4. ITEMS FROM COMMISSION AND STAFF None

5. ADJOURNMENT

Motion: To adjourn Moved By: Commissioner Logan Seconded By: Commissioner Mobley Yes votes: All Ayes No votes: None Abstentions: None The motion passed.

Chairman Crosby adjourned the meeting at 9:57 p.m.

Chairman Jerry Crosby

Attest:

Kelly O'Neill Jr., Planning & Building Director Date signed:_____



Staff Report

Meeting Date:	April 8, 2019
From	Emily Meharg, Associate Planner
SUBJECT:	18-025 SUB/VAR/FSH/TREE

Background:

All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

Recommendation:

Staff recommends the Planning Commission hold a public hearing to take public testimony regarding the proposal. Staff recommends the Planning Commission **approve** the subdivision application with modifications as recommended in this report. The staff recommendation includes approval of the following three variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a two-tiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff is also recommending that the FSH analysis area be a separate tract owned and maintained by an HOA. This would result in Lots 19-21 being less than the minimum 7,500 square feet; thus, staff is recommending approval of a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum and allowing the rear setback to be reduced to 15 feet.

Since the Planning Commission decided to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests

that the Planning Commission provide a revised motion with a more clear basis for the code interpretation decision.

Code Analysis: See attached Exhibits A-Z

Budgetary Impact: None



SUBJECT: File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision

AGENDA DATE: April 8, 2019

DEPARTMENT: Planning Division

Application Submitted: June 25, 2018 Additional Submittal Items: December 5, 2018 Application Complete: December 27, 2018 Planning Commission Hearing: February 25, 2019 PC Continued Hearing: April 8, 2019 120-Day Deadline: June 7, 2019

STAFF CONTACT: Emily Meharg, Associate Planner

EXHIBITS:

Applicant's Submittals

- A. Land Use Application Form
- B. Narrative
- C. Project Plan Set
 - Sheet 1: Cover Sheet and Preliminary Plat Map
 - Sheet 2: Existing Conditions and Tree Retention Inventory
 - Sheet 3: Existing Trees with Building Setbacks Map
 - Sheet 4: FSH and Slopes Map
 - Sheet 5: Preliminary Street and Utility Plan
 - Sheet 6: Preliminary Street Tree Plan
 - Sheet 7: Preliminary Parking Plan
 - Sheet 8: Future Street and Pedestrian Trail Plan
 - Sheet 9: Grading and Erosion Control Plan
 - Sheet 10: Cut and Fill Evaluation
- D. Code Interpretation Request
- E. Storm Drainage Design and Calculations for the Jacoby Heights Subdivision
- F. Traffic Impact Analysis
- G. Geotechnical Report
- H. Wetland Determination
- I. Easement between Cory Knight and Ian Bird and Kelly Bird
- J. Letter from Public Works Director Re: Pump Station Capacity
- K. Arborist Report

Agency Comments

- L. City Engineer (January 15, 2019)
- M. Transportation Engineer (January 18, 2019)
- N. PGE (January 23, 2019)
- O. Public Works Director (January 26, 2019)
- P. Transit Director (January 28, 2019)

Public Comments

- Q. Annette Giltner, 35000 SE Colorado Road (February 6, 2019)
- R. Shawn Lorenzen, 39343 Stratford Street (February 11, 2019)
- S. Tim and Cherri Anders, 19098 Jacoby Road (February 12, 2019)

Additional Documents Submitted by Staff

- T. Chapter 17.102 of the City of Sandy Development Code
- U. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision Staff Report from February 25, 2019 Planning Commission meeting

Documents Submitted at the February 25, 2019 Planning Commission Hearing

- V. Arborist Notes for Jacoby Heights Subdivision
- W. Review of Staff Recommended Conditions File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision memo by Tracy Brown

Additional Agency Comments after the February 25, 2019 PC Hearing

X. Public Works Director and City Engineer

Additional Documents Submitted by Staff after the February 25, 2019 PC Hearing

Y. Park Facilities Plan

Additional Comments Submitted the day of the February 25, 2019 PC Hearing

Z. Brad Picking

I. BACKGROUND

A. PROCEEDING

Type III Subdivision, Type III Code Interpretation, Type III FSH review, Type III Variance for not orienting towards a transit street, Type III Variance for not providing two tiered lot configuration, Type III Variance to Block Length, and Type II Tree Removal Permit. Note: the proposal includes a newly mapped FSH overlay area. The zone change associated with the new FSH overlay will be processed separately by the City.

B. FACTUAL INFORMATION

- 1. APPLICANT/OWNER: Cory Knight
- 2. PROJECT NAME: Jacoby Heights Subdivision
- 3. LEGAL DESCRIPTION: T2S R4E Section 24A Tax Lot 2300
- 4. SITUS ADDRESS: 19124 Jacoby Road
- 5. PROPERTY LOCATION: East of Jacoby Road, west of Langensand Road, and south of Cascadia Village Drive
- 6. PROPERTY SIZE: 9.68 acres
- PROPOSAL: All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

- 8. COMPREHENSIVE PLAN DESIGNATION: Low Density Residential
- 9. ZONING DISTRICT DESIGNATION: SFR, Single Family Residential
- 10. SERVICE CONSIDERATIONS: The applicant proposes to extend the existing sanitary sewer in Jacoby Road to service the entire subdivision. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. The stormwater for the subdivision would be collected and conveyed to a proposed detention pond located in Tract A at the northwest corner of the site.
- 11. RESPONSE FROM GOVERNMENTAL AGENCIES, UTILITY PROVIDERS, AND CITY DEPARTMENTS:
 - a. City Manager No comments received
 - b. City Engineer- Exhibit L
 - c. Transportation Engineer Exhibit M
 - d. Public Works Director Exhibit O
 - e. Transit Director Exhibit P
 - f. PGE Exhibit N
 - g. ODOT No comments received
 - h. SandyNet Manager No comments received
 - i. Police Department No comments received
 - j. Fire District No. 72 No comments received
 - k. US Postal Service No comments received

C. PUBLIC COMMENTS

Three written comments were received as follows:

- Annette Giltner at 35000 SE Colorado Road (Exhibit Q) does not want to see more development in Sandy.
- Shawn Lorenzen at 39343 Stratford Street (Exhibit R) does not want to see more
 residential units built in Sandy without first strengthening the City's infrastructure to
 support existing residents.
- Tim and Cherri Anders at 19098 Jacoby Road (Exhibit S) expressed concerns about the
 effect of the proposed subdivision on their property, which is directly adjacent to the
 north.

D. APPLICABLE CRITERIA:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Sandy Municipal Code: 17.12 Procedures for Decision Making; 17.14 Request for Interpretation; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 Single Family Residential (SFR); 17.56 Hillside Development; 17.60 Flood and Slope Hazard Overlay District; 17.66 Adjustments and Variances; 17.80 Additional Setbacks on Collector and Arterial Streets; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.86 Parkland and Open Space; 17.92 Landscaping and Screening; 17.98 Parking, Loading, and Access; 17.100 Land Division; 17.102 Urban Forestry; 15.30 Dark Sky Ordinance; and 15.44 Erosion Control.

E. BACKGROUND INFORMATION/HISTORY

There is one existing house at 19124 Jacoby Road. The applicant shall obtain a demolition permit from the City of Sandy Building Department prior to demolition of the existing structure(s) on-site. Per the City Engineer (Exhibit L), the preliminary plans don't identify any existing domestic or irrigation wells on site. Any existing domestic or irrigation wells on site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City Planning Division. Per the City Engineer (Exhibit L), the preliminary plans don't identify any septic tanks on site. Any on-site sewage disposal system shall be abandoned in conformance with Clackamas County Water Environmental Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City Planning Division.

F. PROCEDURAL CONSIDERATIONS

This request is being processed as a Type III Subdivision, Type III FSH Review, three Type III Variances, Type III Code Interpretation, and Type II Tree Removal Permit. Notification of the proposal was mailed to property owners within 300 feet of the subject property and to affected agencies on January 29, 2019. A legal notice was published in the Sandy Post on February 6, 2019.

The Planning Commission reviewed the subject subdivision request at a public hearing on February 25, 2019. The meeting focused primarily on the applicant's request for a code interpretation. Due to inclement weather, staff suggested that the meeting be continued. The continued meeting date was set for April 8, 2019. This staff report is an update to the staff report that was presented at the February 25, 2019 Planning Commission meeting.

At the February 25, 2019 Planning Commission hearing, the applicant contested some of staff's proposed conditions. Due to inclement weather and the general consensus to continue the hearing, staff did not have an opportunity to respond to various issues raised by the applicant. Below is a partial list of some of the objections brought up by the applicant's representatives, followed by staff's response.

- Applicant's Representative: Subdivisions don't include landscaping. Staff's Response: Many subdivisions include parking tracts, stormwater detention areas, and pedestrian tracts that do have landscaping requirements. All subdivisions are required to install street trees.
- Applicant's Representative: The pre-application notes were not comprehensive. Staff's Response: Pre-application notes are never comprehensive. Every proposal is unique and often issues are only discovered after a thorough review of the proposal. All pre-app

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

notes include the following language: "This analysis includes a review of those code sections that may conflict with the proposed design as submitted. This review is not intended to be a comprehensive analysis of all applicable code sections."

- Applicant's Arborist: Excavating within critical root zones could lead to tree failure. Staff's Response: We completely agree.
- Applicant's Arborist: "I agree that the 10 feet from the trunk is not [couldn't hear] enough tree protection for a lot of trees. It needs to be at least the drippage [sic]. The 17.92 says the drippage [sic] plus 5 feet, which I think would be ample." Locating tree protective fencing 5 feet beyond the dripline is good. Staff's Response: We completely agree.
- Applicant's Arborist: Recommends using a 6 foot metal fence. Staff's Response: We completely agree and that is what Section 17.102.50(B.2) of our code says.
- Applicant's Representative: Lot 19 is worth \$130,000. Staff's Response: A majority of Lot 19 contains slopes in excess of 25 percent. Staff is curious who appraised this lot to be \$130,000 as staff does not have evidence of that appraisal amount.

Prior to submitting construction plans or building permits, including grading and erosion control permits, the applicant shall update the plan set and associated documents based on the conditions of approval determined by the Planning Commission and shall submit a full set of the updated plans to Planning Division staff for review and approval.

II. ANALYSIS OF CODE COMPLIANCE

CHAPTER 17.14 – REQUEST FOR INTERPRETATION

Request: The applicant is requesting an interpretation regarding whether Chapter 17.92, Landscaping and Screening, applies to subdivisions.

RESPONSE: Staff reviewed the Development Code and relevant ordinances and consulted with the City Attorney and found nothing indicating that Chapter 17.92 does not apply to subdivisions. In fact, the full title of Chapter 17.92 is Landscaping and Screening General Standards – All Zones. Staff interprets this as meaning that Chapter 17.92 applies to all zones, including Single Family Residential. Chapter 17.92 includes standards related to required tree plantings (including street trees and parking court trees), minimum sizes for trees and plants, and revegetation standards, all of which are highly applicable to subdivision development. The specific general provision that the applicant is questioning (Section 17.92.10(C)) states that significant trees should be integrated into the design of a development and that trees to be retained shall be protected from damage during construction. Staff finds that a subdivision qualifies as a development and as construction. Furthermore, there is nothing contained in Chapter 17.92 or Chapter 17.102 that restricts using Chapter 17.92 during subdivision review. There is also nothing in the ordinance (Ordinance No. 2002-10) that adopted Chapter 17.102 that speaks on this matter. In addition, there are several Comprehensive Plan Goal 5 policies that support retaining significant trees, most notably, the following:

 Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.
- Policy 31: The City of Sandy shall undertake a study to designate "significant/heritage" trees or areas of second growth that exist within the City's urban growth boundary. Designated trees shall be protected and preserved where possible.

At the February 25, 2019 Planning Commission hearing, the City Attorney stated: "There's really nothing in 17.92 or 17.102 that indicates 17.92 would not be applicable during subdivision review." The City Attorney further stated: "When you have [a tension between two code sections] normally what you would do, unless there's an interpretation to the contrary, would be to try to harmonize both 17.102 and 17.92 when it comes to tree preservation." Members of the Planning Commission reiterated that the code clearly states that Chapter 17.92 applies to all zones. Commissioner Carlton got clarification from the City Attorney that all zones means all zones, including SFR, R-1, R-2, etc. Commissioner Crosby stated: "It would be hard to interpret something against what it says in the title 'All Zones." After further discussion, Commissioner Carlton made a motion to interpret Chapter 12.92 (sic) to apply to all zones including subdivisions, but that Section 17.92.10(C) not apply to residential subdivisions. After consultation with the City Attorney on the wording of the motion, Commissioner Carlton restated the motion as "the tree retention standards in 17.102 are the standards to apply to subdivisions versus the tree retention standards in 17.92.10(C)" and Commissioner Logan seconded the motion. The Planning Commissioners voted 4-0 to carry the motion. Staff was confused by the motion and unsure what the basis was for rendering the interpretation that 17.92.10(C) does not apply to residential subdivisions, but the remainder of Chapter 17.92 does apply to subdivisions since the motion contradicted the City Attorney's statement and did not seem to include an analysis or reason for the code interpretation. Staff reached out to the Planning Commission chair for clarification and was provided the following rationale:

"Here's what I recall was our thinking. There is a conflict between the tree retention standards in 17.92.10 (C) and 17.102 in regards to their applicability to subdivisions, and it has produced confusion for developers. (Although 17.92 clearly states in its title, "All Zones," oddly enough the chart in it does not list all zones. That caused both John and I to cock our heads.) To resolve the conflict and avoid confusion (to a point...there will always be confusion!), we simply pointed to 102 and said, 'This is the code to be applied to subdivisions, in place of 17.92.10 (C), in regards to tree retention.'"

Staff is still confused as the tree retention standards in 17.92.10(C) and 17.102 don't conflict. Furthermore, the landscaping chart in Section 17.92.20 that the Planning Commission referenced simply lists zones that have minimum landscaping standards; the SFR, R-1, and R-2 zones do not have minimum standards so do not appear in the chart. Staff is still not clear as to what section of the code the Planning Commission is citing as the basis for their interpretation.

Recommendation: Since the Planning Commission decides to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation

 $w: \city Hall \city Hall \city Hall \city Hall \city Hall \city Hall \city \city Hall \city \c$

decision.

CHAPTER 17.30 - ZONING DISTRICT

The Single Family Residential (SFR) zoning on the site allows a minimum of 3 units per acre and a maximum of 5.8 units per acre. The number of dwelling units permitted on a parcel of land is calculated after determining the amount of restricted development area in accordance with Section 17.60, the Flood and Slope Hazard Overlay District. Density transfer is permitted from restricted development areas consistent with the provisions of Section 17.60.90, although no density transfer is proposed.

RESPONSE: The total gross acreage for the entire property is 9.68 acres. The proposal contains 2.08 acres of area dedicated for public right-of-way and 0.47 acres dedicated for public tracts (Tract A). After removal of the right-of-way and public tracts the net site area for the subject property is reduced to 7.13 acres of net site area (NSA). The subject property also contains 1.20 acres of restricted development area; therefore, the unrestricted site area (USA) is 5.93 acres. Based on required density, the SFR land requires a minimum of 18 dwelling units (5.93 USA x 3). The maximum allowed dwelling units is 41 (calculated as the lesser of (a) 7.13 NSA x 5.8 = 41 and (b) 5.93 USA x 5.8 x 1.5 = 52). The proposed 32 dwelling units are within the allowable density range and therefore meet the density requirement.

CHAPTER 17.34 - SINGLE FAMILY RESIDENTIAL (SFR)

The applicant proposes 32 single family detached dwellings in conformance with minimum and maximum density requirements, as detailed above in Chapter 17.30.

17.34.10 PERMITTED USES

RESPONSE: Section 17.34.10 lists single family detached dwellings as a permitted use. The proposed subdivision includes 32 lots for single family detached dwellings. All homes shall provide building design features in compliance with the standards in Section 17.90.150 and Chapter 17.54.

17.34.30 DEVELOPMENT STANDARDS

Туре	Standard	
A. Minimum Lot Area - Single detached dwelling	7,500 square ft.	
- Other permitted uses	No minimum	
A. Minimum Average Lot Width		
- Single detached dwelling	60 ft.	
C. Minimum Lot Frontage	20 ft. except as allowed by Section	
	17.100.160	
D. Minimum Average Lot Depth	No minimum	
E. Setbacks (Main Building)		
Front yard	10 ft. minimum	
Rear yard	20 ft. minimum	

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Side yard (interior)	7.5 ft. minimum	
Corner Lot	10 ft. minimum on side abutting the street 1	
F. Setbacks (Garage/Carport)	22 ft. minimum for front vehicle access	
	15 ft. minimum if entrance is perpendicular	
	to street (subject to Section 17.90.220)	
	5 ft. minimum for alley or rear access	
G. Projections into Required Setbacks	See Chapter 17.74	
H. Accessory Structures in Required Setbacks	See Chapter 17.74	
I. Structure Height	35 ft. maximum	
J. Building Site Coverage	No minimum	
K. Off-Street Parking	See Chapter 17.98	

RESPONSE: The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes shall meet the development standards of Section 17.34.30.

17.34.40 MINIMUM REQUIREMENTS

- A. Must connect to municipal water.
- B. Must connect to municipal sewer if service is currently within 200 hundred feet of the site. Sites more than 200 feet from municipal sewer, may be approved to connect to an alternative disposal system provided all of the following are satisfied:
 - 1. A county septic permit is to be secured and a copy is provided to the city.
 - 2. The property owner executes a waiver of remonstrance to a local improvement district and/or signs a deed restriction agreeing to complete improvements.
 - 3. The minimum size of the property is one acre or is a pre-existing buildable lot, as determined by the city.
 - 4. Site consists of a buildable parcel(s) created through dividing property in the city, which is less than five acres in size.
- C. The location of any real improvements to the property must provide for a future street network to be developed.
- D. Must have frontage or approved access to public streets.

RESPONSE: All lots will be required to connect to City services. The applicant is also required to extend utilities to the furthest extent of the subject property.

CHAPTER 17.56 – HILLSIDE DEVELOPMENT

17.56.10 APPLICABILITY

These regulations shall apply to any parcel with slopes greater than twenty-five percent (25%) as-shown on the Hillside Development Overlay District Map or with slope hazards mapped by the Department of Geology and Mineral Industries (DOGAMI). This chapter shall apply only to activities and uses that require a building, grading, tree removal and/or land use permit.

¹ Must comply with clear vision requirements of Chapter 17.74.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. <u>General</u>. No person shall develop property in areas designated by SDC 17.56.10, without first demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance or land use approval, the applicant shall agree to implement the recommendations of approved studies and to allow all inspections to be conducted.
 - 2. Where a bond, letter of credit or other guarantee is required, the permit shall not be issued until the bond or guarantee has been obtained and approved.

B. Exemptions:

- 1. An activity or use that avoids slopes of 25% or greater, DOGAMI slope hazard areas, natural drainageways and potentially hazardous analysis areas as defined in Section 17.56.30.A.
- 2. The following activities, regardless of location:
 - a. An excavation that is less than three feet in depth, or which involves less than fifty cubic yards of volume;
 - b. A fill that does not exceed three feet in depth or 50 cubic yards of volume;
 - c. New construction or expansion of a structure resulting in a net increase in ground floor area of less than 1,000 square feet that does not involve grading;
 - d. Emergency actions required to prevent an imminent threat to public health or safety, or prevent imminent danger to public or private property, as determined by the Director; or
 - e. Any land use or activity that does not require a building, grading permit, or land use approval.

RESPONSE: The applicant submitted an FSH and Slopes Map (Exhibit C, Sheet 4) that shows areas of land that exceed 25 percent slope. A majority of the 25 percent or greater slope areas are located in the northeast corner of the property, adjacent to the Tickle Creek basin. The applicant submitted a Geotechnical Report (Exhibit G) by Rapid Soil Solutions LLC that includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations, which was reviewed by the City Engineer. Per the City Engineer (Exhibit L), the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC. The submitted Cut and Fill Evaluation (Exhibit C, Sheet 10) shows proposed cut and fill within the tree protection area south of Trees #419, #523, and #526. The applicant shall update the cut and fill evaluation to include the location of the tree protection fencing as depicted on the existing conditions and tree retention inventory (Exhibit C, Sheet 2). The applicant shall not cut or fill within the proposed tree protection area.

17.56.30 REQUIRED MAP AND STUDIES

A. Topographic Map Required. To determine the location of potentially hazardous areas, the applicant shall submit a scaled topographic map at two-foot contour intervals for the subject property (site) and for land within 25 feet of the site perimeter. In addition to DOGAMI slide hazard areas and slopes of 25% or greater, potentially hazardous "analysis areas" include land within 25 feet of the top or toe of slopes of 25% or greater and the area 25 feet on either side of drainageways that drain 20 acres or more. This map shall be prepared by a registered engineer or land surveyor and shall show:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- 1. Slopes of 25-34%;
- 2. Slopes of 35% and greater;
- 3. The analysis that is within 25' of slopes that are 25% or greater parallel to and within 25 feet of the top of the 25% slope break;
- 4. Mapped DOGAMI slide hazard areas;
- 5. The analysis area within 25 feet of the centerline of drainageways that drain at least 20 acres; and
- 6. The area (in square feet) for each category listed above for the subject property.
- B. <u>Types of Required Studies</u>. There are three types of geological and engineering studies that may be required by this chapter. See Table 1 under Section 17.56.40, below.
 - <u>Geological Assessments</u> are prepared and stamped by a Certified Engineering Geologist and describe the surface and subsurface conditions of a site, delineate areas of a property that may be subject to specific geologic hazards, and assess the suitability of the site for development. Geotechnical Reports shall be conducted according to the requirements of Appendix A (*Geological Assessments*), shall make recommendations as to whether further studies are required, and may be incorporated into or included as an appendix to the geotechnical report.
 - 2. Engineering Geology Reports are prepared and stamped by a Certified Engineering Geologist and provide detailed descriptions of the geology of the site, professional conclusions and recommendations regarding the effect of geological conditions on the proposed development, and opinions and recommendations covering the adequacy of the site to be developed. Engineering Geology Reports shall be prepared in accordance with the requirements of Appendix B (*Guidelines for Preparing Engineering Geology Reports in Oregon* adopted by the Oregon State Board of Geologist Examiners) and may be incorporated into or included as an appendix to the geotechnical report.
 - 3. <u>Geotechnical Reports</u> are prepared and stamped by a Geotechnical Engineer, evaluate site conditions, and recommend design measures necessary to reduce the development risks and facilitate safe and stable development. Geotechnical Reports shall be conducted according to the requirements of Appendix C (Geotechnical Reports), and may be incorporated into or included as an appendix to the Engineering Geology Report.

RESPONSE: Rapid Soil Solutions LLC completed a Geotechnical Report (Exhibit G) dated February 23, 2018 for the subject site. **The applicant shall follow the recommendations outlined in the Geotechnical Report.**

17.56.40 WHERE STUDIES REQUIRED

Additional geological or engineering studies shall be required, or not required, under the following circumstances:

	Type I Development	NON-EXEMPT Grading;
Situation	Applications; Single Family	Type II or III Develop-
	Homes, Duplexes and	ment Applications
	Accessory Uses	
A. Proposed development	1. No further requirements	2. No further requirements
avoids slopes of 25% or		
greater, drainageways,		

TABLE 1: WHERE STUDIES ARE REQUIRED OR NOT REQUIRED

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

DOGAMI slope hazard areas and all analysis areas:		
B. Development proposed on slopes of 25-35% or analysis areas, but avoids drainageways, DOGAMI hazard areas and slopes of 35% or greater:	1. Geological Assessment required; Engineering Geology or Geotechnical Reports may be required*	2. Engineering Geology Report required; Geotechnical Report may be required*
C. Development proposed on DOGAMI hazard areas, slopes of 35% or greater, or drainageway areas:	1. Engineering Geology Report required; Geotechnical Report may be required*	2. Engineering Geology Report and Geotechnical Report required

* Whether additional studies are necessary depends on recommendations of base required study.

RESPONSE: The applicant shall submit a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay. This includes Lots 19, 20, 21, 27, 28, and 29. Additional reports may be required depending on the recommendations of the geological assessment. Once the subdivision is platted and building permits are filed for individual home construction the property owner/contractor shall submit Type I Hillside applications with studies as detailed in Table 1 in Section 17.56.40.

17.56.50 COMPLIANCE WITH STUDY CONCLUSIONS AND RECOMMENDATIONS REQUIRED

- A. <u>Professional Standards</u>. The director shall determine whether Geological Assessments, Engineering Geology Reports, or Geotechnical Reports have been prepared in accordance with Section 17.56.30. The director may require additional information or analysis necessary to meet study requirements.
- B. <u>Peer Review</u>. The director may require peer review of any required report, in which case regulated activities and uses shall be reviewed and accepted through the peer review process before any regulated activity will be allowed.
 - 1. A professional or professional firm of the city's choice that meets the qualifications listed in this chapter shall perform the review.
 - 2. The review shall be at the applicant's expense.
 - 3. Review of report submittals shall determine whether required elements are completed, geologic report procedures and assumptions are accepted, and all conclusions and recommendations are supported and reasonable.
- C. <u>Review Criteria</u>. The approval authority shall rely on the conclusions and recommendations of required reports, as modified by peer review, to determine compliance with this chapter.
- D. <u>Conditions of Approval</u>. Conclusions and recommendations stated in approved reports shall be directly incorporated as permit conditions or provide the basis for conditions of approval for the regulated activity or use.
- E. <u>Expiration</u>. Where an approved assessment or report as defined and required by this chapter has been prepared within the last five years for a specific site, and where the proposed land use activity and surrounding site conditions are unchanged, that report may be utilized and a new report is not

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

required. Should environmental conditions associated with the site or surrounding the site change, or if the proposed land use activity or development has materially changed, the applicant shall submit an amendment to the required assessment or report, which may be reviewed and approved through the peer review process.

RESPONSE: The Geotechnical Report includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.

CHAPTER 17.60 - FLOOD AND SLOPE HAZARD OVERLAY DISTRICT

17.60.00 INTENT

This chapter is intended to promote the public health, safety and general welfare by minimizing public and private adverse impacts from flooding, erosion, landslides or degradation of water quality consistent with Statewide Planning Goals 6 (Air, Land and Water Resources Quality) and 7 (Areas Subject to Natural Disasters and Hazards) and the Sandy Comprehensive Plan (SCP).

RESPONSE: The applicant is not proposing any development within the Flood and Slope Hazard (FSH) overlay district. However, the proposal includes 6 lots that that are proposed to be platted within the FSH overlay. Staff brought the subject subdivision proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board recommends that a tract is created and dedicated to the City and cited three primary reasons that they would like to see the land dedicated: 1) increased protection of the trees located within the FSH buffer area; 2) a larger area through which a connector trail to Tickle Creek Trail could meander; and, 3) to provide a wider riparian buffer that would not only better protect Tickle Creek, but would also enhance the natural beauty along the future extension of the Tickle Creek Trail. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the FSH tract without knowing the existing condition of that area and recommended that, prior to dedicating the FSH tract to the City, the applicant would need to: 1) submit a Phase I environmental assessment and a native vegetation report for the FSH area to be dedicated in the tract, 2) remove all invasive plants and re-plant any areas of exposed soil with native plants, and 3) submit a maintenance bond for two full growing seasons (May 1 - October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Staff supported these recommendations in the February 25, 2019 staff report; however, the applicant's representative indicated that the applicant did not want to incur the expense of having to restore the FSH tract and conduct a Phase I environmental assessment. If the FSH tract is not dedicated to the City, it will not be necessary to perform a Phase I environmental assessment. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants. Due to the site layout, Lots 19-21

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

will end up less than 7,500 square feet once the FSH area is put into a separate tract. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot area and allowing the rear setback to be reduced to 15 feet. This will help avoid confusion over property ownership, fence lines, and unnecessary or illegal removal of riparian vegetation or vegetation on steep slopes.

The proposal also contains two trees that are proposed for removal and appear to be on the FSH boundary line: Trees #510 and #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.

17.60.30 REQUIRED SETBACK AREAS

Setback areas shall be required to protect water quality and maintain slope stability near stream corridors and locally significant wetlands. Setbacks are measured horizontally from, parallel to and upland from the protected feature.

- A. <u>Required Setbacks</u>. The required special setback(s) shall be:
 - 1. 70 feet from the top of bank of Tickle Creek;
 - 2. 50 feet from top of bank along other perennial streams, except for "No Name Creek" east of Towle Drive, as provided in Section 17.60.30.C.2 below.
 - 3. 25 feet around the edge of any mapped locally significant wetland; and
 - 4. 25 feet from the top of any 25% slope break where the slope break occurs within the FSH overlay district as mapped by the city.

RESPONSE: The applicant submitted a Wetland Determination (Exhibit H) by Environmental Science & Assessment, LLC dated May 30, 2018. The wetland determination concluded there is no evidence of intermittent or perennial flow on the site. Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). Initially, the City Engineer required that the wetland report/delineation shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife. In a subsequent email to the applicant's representative, the City Engineer decided his original comment had been met and there was no need to include concurrence as a condition of approval. The City Engineer recommends that the City of Sandy Planning Division review the FSH Overlay District line at the northeast corner of the site.

17.60.50 SPECIAL REPORTS

Where development is proposed on restricted development areas within the FSH overlay district as defined in Section 17.60.20.A, the Director shall require submission of the following special reports. These reports shall be in addition to other information required for specific types of development, and shall be prepared by professionals in their respective fields.

The Director may require one of more of these reports where necessary to address potential adverse impacts from development on buildable land within the FSH overlay district. The Director may exempt Type II permit applications from one or more of these reports where impacts are minimal and the exemption is consistent with the purpose of the FSH overlay zone as stated in Section 17.60.00.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. <u>Hydrology and Soils Report</u>. This report shall include information on the hydrological conditions on the site, the effect of hydrologic conditions on the proposed development, the proposed development's impact on surface and groundwater flows to wetlands and streams, and any hydrological or erosion hazards. This report shall also include soils characteristics of the site, their suitability for development, carrying capacity, and erosion or slumping characteristics that might present a hazard to life and property, or adversely affect the use or stability of a public facility or utility. Finally, this report shall include information on the nature, distribution and strength of existing soils; the adequacy of the site for development purposes; and an assessment of grading procedures required to impose the minimum disturbance to the natural state. A licensed professional engineer registered in Oregon shall prepare the hydrology and soils report.
- B. <u>Grading Plan.</u> The grading plan shall be specific to a proposed physical structure or use and shall include information on terrain (two-foot intervals of property), drainage, direction of drainage flow, location of proposed structures and existing structures which may be affected by the proposed grading operations, water quality facilities, finished contours or elevations, including all cut and fill slopes and proposed drainage channels. Project designs including but not limited to locations of surface and subsurface devices, walls, dams, sediment basins, storage reservoirs, and other protective devices shall form part of the submission. The grading plan shall also include: 1) construction phase erosion control plan consistent with the provisions of Chapter 15.44; and 2) schedule of operations. A licensed professional engineer registered in Oregon shall prepare the grading and erosion control plan.
- C. <u>Native Vegetation Report</u>. This report shall consist of a survey of existing vegetative cover, whether it is native or introduced, and how it will be altered by the proposed development. Measures for revegetation with native plant species will be clearly stated, as well as methods for immediate and long-term stabilization of slopes and control of soil erosion. A landscape architect, landscape designer, botanist or arborist with specific knowledge of native plant species, planting and maintenance methods, survival rates, and their ability to control erosion and sedimentation shall prepare the vegetation report. The applicant shall be responsible for replacing any native plant species that do not survive the first two years after planting, and for ensuring the survival of any replacement plants for an additional two years after their replacement.

RESPONSE: Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). The applicant submitted a wetland determination (Exhibit H) that the City Engineer determined satisfies the requirement.

17.60.80 WATER QUALITY TREATMENT FACILITIES

Tickle Creek, the Sandy River and associated natural drainage ways are vital to Sandy's recreationally based economy and to the quality of life of Sandy residents. Placement of water quality facilities shall be limited as follows:

A. The water quality facility shall not be constructed in restricted development areas, except where necessary to serve approved development within restricted development areas (e.g., a road) and where no reasonable alternative exists in buildable areas of the site.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

B. Where the approval authority determines that a more efficient and effective regional site exists within the sub-basin, the water quality facility may be constructed off-site.

RESPONSE: The proposed detention pond (Tract A) is located in the northwest corner of the site and is not within the mapped FSH overlay area.

17.60.90 DENSITY TRANSFER PROVISIONS

Residential density transfer may be approved subject to the following:

- A. Required Setback Areas. Density may be transferred from restricted development areas (i.e., steep slopes, protected water features and required setbacks) to buildable portions of the site.
- B. Density Maximum. The maximum gross density for the buildable area of the site shall not exceed 150% of the maximum density allowed by the underlying zoning district for that buildable area.
- C. Housing Types Not Permitted in Underlying Zoning District. Housing types not permitted in the underlying zoning district may only be approved through the PD (planned development) or SAP (specific area plan) process.
- D. Transfer Area. Transfer of density may only occur within the same property and/or to properties contiguous to the primary property. The terms "primary property" identify the legal lot from which density is to be transferred to "secondary property(s)". Further development or land use action on the primary or secondary properties shall be reviewed together in the same application.

RESPONSE: Since no development is proposed within the FSH area, a density transfer per Section 17.60.90 is allowed for the buildable area of the site. There are 1.20 acres of restricted development area. See Chapter 17.30 for density analysis.

CHAPTER 17.80 - ADDITIONAL SETBACKS ON COLLECTOR AND ARTERIAL STREETS

17.80.10 APPLICABILITY

Collector: Jacoby Road

RESPONSE: Four lots (Lots 9, 10, 30 and 32) are proposed to contain frontage on Jacoby Road, which is classified as a collector street. All structures on lots along Jacoby Road shall be constructed to comply with the standards of Section 17.80.20.

17.80.20 SPECIFIC SETBACKS

Any structure located on streets listed above or identified in the Transportation System Plan as arterials or collectors shall have a minimum setback of 20 feet measured from the property line. This applies to applicable front, rear and side yards.

RESPONSE: All structures shall maintain a minimum 20-foot setback from the Jacoby Road public right-of-way. The Preliminary Plat (Exhibit C, Sheet 1) depicts building envelopes at 20 feet from the Jacoby Road right-of-way.

CHAPTER 17.82 – SPECIAL SETBACKS ON TRANSIT STREETS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.82.20 BUILDING ORIENTATION

- A. All buildings shall have their primary entrances oriented toward a transit street rather than a parking area, or if not adjacent to a transit street, toward a public right-of-way or private walkway which leads to a transit street.
- B. Buildings shall have a primary entrance connecting directly between the street and building interior. This entrance shall be open to the public during all business hours and shall comply with the accessibility standards of the Uniform Building Code.
- C. In lieu of a building entrance oriented to a transit street, a building's entrance may be enhanced and identified in the following manner:
 - 1. An entrance plaza of at least 150 square feet, at least 100 square feet of which shall be visible from the transit street. The entrance plaza shall be at least 10 feet wide at the narrowest dimension; and
 - 2. A permanent building feature (e.g. a portico, porch or awning) shall be visible from the transit street, signifying an entrance; and
 - 3. Pedestrian-scale lighting shall be required at the entrance; and
 - 4. A clearly marked, convenient, safe and lighted pedestrian route shall be provided to the entrance, from the transit street.
- D. Primary building entrances shall be architecturally emphasized and visible from the street. Building entrances shall incorporate arcades, roofs, porches, alcoves, porticoes, and awnings that protect pedestrians from the rain and sun. Continuous arcades are strongly encouraged.
- E. All building entrances and exits shall be well lit. Lighting shall be a pedestrian scale (3'-12') and the source light shall be shielded to reduce glare.
- F. For commercial buildings with facades over 300 feet in length on a transit street, two or more building entrances on the street must be provided.
- G. If the site has frontage on more than one transit street, the building shall provide one main entrance oriented to a transit street or to a corner where two transit streets intersect.

RESPONSE: Jacoby Road is a designated transit street. Lots 9, 10, 30, and 32 have frontage on Jacoby Road. As part of the Jacoby Heights subdivision proposal, the applicant has requested a variance to the requirements of Section 17.82.20 to allow the front door for the houses on Lots 9, 10, 30, and 32 to face the internal street network instead of Jacoby Road, which is a designated transit street. The variance request is discussed in detail in the analysis of Chapter 17.66 in this staff report.

CHAPTER 17.84 - IMPROVEMENTS REQUIRED WITH DEVELOPMENT

This chapter provides general information regarding improvements required in association with development, and it clarifies the timing, extent, and standards for public and private improvements.

17.84.20 TIMING OF IMPROVEMENTS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. All improvements required by the standards in this chapter shall be installed concurrently with development, as follows:
 - 1. Where a land division is proposed, each proposed lot shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to approval of the final plat.
 - 2. Where a land division is not proposed, the site shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to temporary or final occupancy of structures.

RESPONSE: Submission of preliminary street and utility plans during the land use review process is solely for compliance with the data requirements of Section 17.100.60 (D). **Public improvement** plans are subject to a separate review and approval process. Preliminary plat approval does not connote approval of public improvement construction plans.

B. Where specific approval for a phasing plan has been granted for a planned development and/or subdivision, improvements may similarly be phased in accordance with that plan.

RESPONSE: The applicant did not propose a phasing plan with this application.

17.84.30 PEDESTRIAN AND BICYCLIST IMPROVEMENTS

- A. Sidewalks shall be required along both sides of all arterial, collector, and local streets, as follows:
 - 1. Sidewalks shall be a minimum of 5 ft. wide on local streets. The sidewalks shall be separated from curbs by a tree planting area that provides separation between sidewalk and curb, unless modified in accordance with Subsection 3 below.
 - 2. Sidewalks along arterial and collector streets shall be separated from curbs with a planting area, except as necessary to continue an existing curb-tight sidewalk. The planting area shall be landscaped with trees and plant materials approved by the City. The sidewalks shall be a minimum of 6 ft. wide.
 - 3. Sidewalk improvements shall be made according to city standards, unless the city determines that the public benefit in the particular case does not warrant imposing a severe adverse impact to a natural or other significant feature such as requiring removal of a mature tree, requiring undue grading, or requiring modification to an existing building. Any exceptions to the standards shall generally be in the following order.
 - a) Narrow landscape strips
 - b) Narrow sidewalk or portion of sidewalk to no less than 4 feet in width
 - c) Eliminate landscape strips
 - d) Narrow on-street improvements by eliminating on-street parking
 - e) Eliminate sidewalks
 - 4. The timing of the installation of sidewalks shall be as follows:
 - a) Sidewalks and planted areas along arterial and collector streets shall be installed with street improvements, or with development of the site if street improvements are deferred.
 - b) Sidewalks along local streets shall be installed in conjunction with development of the site, generally with building permits, except as noted in (c) below.
 - c) Where sidewalks on local streets abut common areas, drainageways, or other publicly owned or semi-publicly owned areas, the sidewalks and planted areas shall be installed with street improvements.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) depicts an 8-foot-wide public utility easement along the frontage of all proposed lots to facilitate future installation of utilities. Sidewalks abutting the proposed lots shall be constructed in association with development of the lots. **The applicant shall construct sidewalks along Tract A both on Jacoby Road and Camden Court, prior** to final plat approval. The sidewalks on Camden Court, Woodstock Street, Newton Street, and Middlebury Avenue shall be five feet in width and separated by a five foot planter strip (or 6 foot swale) in areas not transverse by driveways. The sidewalks on Jacoby Road shall be six feet in width and separated by a five foot planter strip.

The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all streets. The section of Newton Street on the east side of the property adjacent to Lots 1 and 2 is proposed to be constructed with half street improvements. This proposed design places the utilities underneath the planter strip; thus, the proposal does not show street trees in front of Lots 1 and 2. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a three-quarter street. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. Street trees shall be installed approximately 30 feet on center in conjunction with issuance of building permits. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted and staked per the City of Sandy standard planting detail; trees shall be tied to the stakes with loosely tied twine. The planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed within one year of installation.

- B. Safe and convenient pedestrian and bicyclist facilities that strive to minimize travel distance to the extent practicable shall be provided in conjunction with new development within and between new subdivisions, planned developments, commercial developments, industrial areas, residential areas, public transit stops, school transit stops, and neighborhood activity centers such as schools and parks, as follows:
 - 1. For the purposes of this section, "safe and convenient" means pedestrian and bicyclist facilities that: are reasonably free from hazards which would interfere with or discourage travel for short trips; provide a direct route of travel between destinations; and meet the travel needs of pedestrians and bicyclists considering destination and length of trip.
 - 2. To meet the intent of "B" above, right-of-ways connecting cul-de-sacs or passing through unusually long or oddly shaped blocks shall be a minimum of 15 ft. wide with 8 feet of pavement.
 - 3. 12 feet wide pathways shall be provided in areas with high bicycle volumes or multiple use by bicyclists, pedestrians, and joggers.
 - 4. Pathways and sidewalks shall be encouraged in new developments by clustering buildings or constructing convenient pedestrian ways. Pedestrian walkways shall be provided in accordance with the following standards:
 - a) The pedestrian circulation system shall be at least five feet in width and shall connect the sidewalk on each abutting street to the main entrance of the primary structure on the site to minimize out of direction pedestrian travel.
 - b) Walkways at least five feet in width shall be provided to connect the pedestrian circulation system with existing or planned pedestrian facilities which abut the site but are not adjacent to the streets abutting the site.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- c) Walkways shall be as direct as possible and avoid unnecessary meandering.
- d) Walkway/driveway crossings shall be minimized. Internal parking lot design shall maintain ease of access for pedestrians from abutting streets, pedestrian facilities, and transit stops.
- e) With the exception of walkway/driveway crossings, walkways shall be separated from vehicle parking or vehicle maneuvering areas by grade, different paving material, painted crosshatching or landscaping. They shall be constructed in accordance with the sidewalk standards adopted by the City. (This provision does not require a separated walkway system to collect drivers and passengers from cars that have parked on site unless an unusual parking lot hazard exists).
- f) Pedestrian amenities such as covered walk-ways, awnings, visual corridors and benches will be encouraged. For every two benches provided, the minimum parking requirements will be reduced by one, up to a maximum of four benches per site. Benches shall have direct access to the circulation system.

RESPONSE: Five foot wide sidewalks are proposed on local streets and six foot wide sidewalks are proposed along Jacoby Road adjacent to the proposed subdivision. These sidewalks provide safe and convenient pedestrian access within the subdivision and connect the site to neighboring properties. To improve pedestrian connectivity, **the applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west. At the February 25, 2019 Planning Commission hearing, the applicant's engineer stated it would be impossible to design an ADA ramp at that location on Jacoby Road due to the slope. The City Engineer and Public Works Director stated that the mid-block ramp should be easily achievable if the street grade is 7.5 percent and recommend the condition remain as written.**

C. Where a development site is traversed by or adjacent to a future trail linkage identified within the Transportation System Plan, improvement of the trail linkage shall occur concurrent with development. Dedication of the trail to the City shall be provided in accordance with 17.84.80.

RESPONSE: The City's current TSP maps were created with the former UGB boundaries (pre-June 2017) and did not include the subject property that was brought into the revised UGB boundaries. Therefore, there are no trail linkages identified in the TSP for this property.

E. To ensure improved access between a development site and an existing developed facility such as a commercial center, school, park, or trail system, the Planning Commission or Director may require off-site pedestrian facility improvements concurrent with development.

RESPONSE: Tickle Creek crosses the northeast corner of the subject property and the future extension of the Tickle Creek Trail is planned to extend along the length of Tickle Creek. At the pre-application meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trails Advisory Board meeting. The Parks and Trails Advisory Board also recommends that the connector trail be constructed as part of this development. While the Park Facilities Plan (Exhibit Y) depicts the future Tickle Creek Trail located along Tickle Creek in the northeast corner of the subject property, the Park Facilities Plan doesn't go as far as

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

identifying specific trail connections to the proposed future Tickle Creek Trail alignment. Without specific trail connectors identified explicitly on the plan, staff cannot compel the applicant to include a trail connecting to the future Tickle Creek Trail. However, staff and the Parks & Trails Advisory Board recommend the applicant include a connector trail on the subject property that provides access to the future Tickle Creek Trail.

17.84.40 TRANSIT AND SCHOOL BUS TRANSIT REQUIREMENTS

- A. Development sites located along existing or planned transit routes shall, where appropriate, incorporate bus pull-outs and/or shelters into the site design. These improvements shall be installed in accordance with the guidelines and standards of the transit agency. School bus pull-outs and/or shelters may also be required, where appropriate, as a condition of approval for a residential development of greater than 50 dwelling units where a school bus pick-up point is anticipated to serve a large number of children.
- B. New developments at or near existing or planned transit or school bus transit stops shall design development sites to provide safe, convenient access to the transit system, as follows:
 - 1. Commercial and civic use developments shall provide a prominent entrance oriented towards arterial and collector streets, with front setbacks reduced as much as possible to provide access for pedestrians, bicycles, and transit.
 - 2. All developments shall provide safe, convenient pedestrian walkways between the buildings and the transit stop, in accordance with the provisions of 17.84.30 B.

RESPONSE: The Transit Director (Exhibit P) states that the proposed development will require construction of one transit amenity at the northwest corner of the subject property adjacent to Tract A and as close to Cascadia Village Drive as possible. The amenity required is a 7 foot by 9.5 foot concrete pad to accommodate a 5 foot by 7.5 foot bus shelter, equipped with one 6 foot long green metal bench. The applicant shall install one bench (Fairweather model PL-3, powder-coated RAL6028) with a concrete pad to City specifications. The applicant shall contact the Transit Director for specific location, amenity information, and pad engineering specifications.

17.84.50 STREET REQUIREMENTS

- A. Traffic evaluations may be required of all development proposals in accordance with the following:
 - 1. A proposal establishing the scope of the traffic evaluation shall be submitted for review to the City Engineer. The evaluation requirements shall reflect the magnitude of the project in accordance with accepted traffic engineering practices. Large projects should assess all nearby key intersections. Once the scope of the traffic evaluation has been approved, the applicant shall present the results with and an overall site development proposal. If required by the City Engineer, such evaluations shall be signed by a Licensed Professional Civil Engineer or Licensed Professional Traffic Engineer licensed in the State of Oregon.
 - 2. If the traffic evaluation identifies level-of-service conditions less than the minimum standard established in the Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered concurrent with a development proposal.

RESPONSE: The submitted Traffic Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10^{th} Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably per City of Sandy and ODOT standards and are projected to continue to operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development.

The City Engineer (Exhibit L) reviewed the Traffic Impact Analysis and noted the following: "There are no reported crashes at the intersection of Jacoby [Road] and Dubarko Road, or at any of the intersections adjacent to the development. However, the crash rate at the intersection of Dubarko Road and Highway 211 exceeds the 90th percentile criteria and meets the warrants for the addition of left turn lanes as identified in the TSP. This intersection should be addressed by ODOT as soon as possible, but not as a component of this proposed development."

The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and does not recommend any modifications to the TIA as the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards. The City Transportation Engineer states the following: "The intersections that will provide access from the subdivision to Jacoby Road are appropriately located and provide adequate sight distance. The existence of a higher than average crash rate at the intersection of OR 211 and Dubarko Road and traffic volumes that suggest left-turn lanes are warranted on OR 211 at this location are indicators that the city should continue to work with ODOT for implementation of enhancements at this intersection. This analysis indicates that a traffic signal is not yet warranted but left-turn lanes on both the northbound and southbound approaches on OR 211 are warranted. The need for these improvements is justified based on regional growth of which this development is only a minor component."

- B. Location of new arterial streets shall conform to the Transportation System Plan in accordance with the following:
 - 1. Arterial streets should generally be spaced in one-mile intervals.
 - 2. Traffic signals should generally not be spaced closer than 1500 ft. for reasonable traffic progression.

RESPONSE: The proposal does not include any new arterial or collector streets.

- C. Local streets shall be designed to discourage through traffic. NOTE: for the purposes of this section, "through traffic" means the traffic traveling through an area that does not have a local origination or destination. To discourage through traffic and excessive vehicle speeds the following street design characteristics shall be considered, as well as other designs intended to discourage traffic:
 - 1. Straight segments of local streets should be kept to less than a quarter mile in length. As practical, local streets should include traffic calming features, and design features such as curves and "T" intersections while maintaining pedestrian connectivity.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- 2. Local streets should typically intersect in "T" configurations rather than 4-way intersections to minimize conflicts and discourage through traffic. Adjacent "T" intersections shall maintain a minimum of 150 ft. between the nearest edges of the 2 rights-of-way.
- 3. Cul-de-sacs should generally not exceed 400 ft. in length nor serve more than 20 dwelling units, except in cases where existing topography, wetlands, or drainage systems or other existing features necessitate a longer cul-de-sac in order to provide adequate access to an area. Cul-de-sacs longer than 400 feet or developments with only one access point may be required to provide an alternative access for emergency vehicle use only, install fire prevention sprinklers, or provide other mitigating measures, determined by the City.

RESPONSE: The proposal includes a cul-de-sac (Camden Court) that does not exceed 400 feet in length. Five lots and Tract A are proposed to gain access from the cul-de-sac.

- D. Development sites shall be provided with access from a public street improved to City standards in accordance with the following:
 - 1. Where a development site abuts an existing public street not improved to City standards, the abutting street shall be improved to City standards along the full frontage of the property concurrent with development.
 - 2. Half-street improvements are considered the minimum required improvement. Three-quarterstreet or full-street improvements shall be required where traffic volumes generated by the development are such that a half-street improvement would cause safety and/or capacity problems. Such a determination shall be made by the City Engineer.
 - 3. To ensure improved access to a development site consistent with policies on orderly urbanization and extension of public facilities the Planning Commission or Director may require off-site improvements concurrent with development. Off-site improvement requirements upon the site developer shall be reasonably related to the anticipated impacts of the development.
 - 4. Reimbursement agreements for ³/₄ street improvements (i.e., curb face to curb face) may be requested by the developer per Chapter 12 of the SMC.
 - 5. A ¹/₂ street improvement includes curb and pavement 2 feet beyond the center line of the right-ofway. A ³/₄ street improvement includes curbs on both sides of the side and full pavement between curb faces.

RESPONSE: The proposed street network and improvements generally comply with City standards, with the exception of one block variance. There are five streets inside or adjacent to the proposed subdivision requiring the following improvements:

<u>Jacoby Road</u>: Half-street construction of Jacoby Road is required within the subject property boundary to collector street standards, including curb, 6-foot wide concrete sidewalks, street lighting, landscaping strip, street trees, and public utilities. The site has approximately 600 linear feet of frontage on Jacoby Road, a designated collector. The right-of-way width adjacent to the site is 50 feet; the standard for collector streets is 60 feet. Per the City Engineer (Exhibit L) and Public Works Director (Exhibit O), the applicant shall dedicate 10 feet of right-of-way along the Jacoby Road frontage of the site.

<u>Woodstock Street</u>: Woodstock Street requires full-street construction to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities. The applicant has applied for one variance to exceed the maximum block length for the north side of Woodstock Street from Camden Court to the east

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

property boundary. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. At the February 25, 2019 Planning Commission hearing, the applicant's representative indicated the applicant doesn't have an easement from the neighbor and obtaining one could be burdensome. In response, the Public Works Director and City Engineer note: "The condition gives the applicant the option of a retaining wall or an easement to allow extension of the street. The condition doesn't require neighbor participation nor is it contingent on same. Based on the grading plan the wall would be a fill wall, less about 2 feet in height. This is neither problematic or [sic] expensive. The bottom of the wall would be above any required utilities (which are shown extending all the way to the property line. The typical method when extending a street with a fill wall is to remove the top course or two and leave the rest of the wall in place and bury it when the street is extended, no removal required. Even if removal would be required it would be no more expensive than removing the soil (whether fill or cut) at the end of the street when extended." The Public Works Director and City Engineer recommend the condition remain as written.

<u>Middlebury Avenue</u>: Middlebury Avenue requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

<u>Newton Street</u>: Newton Street is proposed to align with the existing alignment of Newton Street west of Jacoby Road and to then shift to the south such that it aligns with the southern property boundary. The extension of Newton Street requires full-street improvements to local standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, street trees, ADA ramps, and public utilities adjacent to Lot 9, tapering to three-quarter street improvements adjacent to the remaining lots along Newton Street. The applicant is proposing a three-quarter street improvement for Newton Street through the site transitioning to a half-street section at the west line of proposed Lot 2. Per the Public Works Director (Exhibit O), the proposed Newton Street alignment coincides with an existing access easement serving at least two and possibly as many as four existing dwellings to the east. No more than two dwellings may be served by a half-street improvement. As proposed this narrowed section of Newton Street would serve at least four dwellings and perhaps as many as six. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. Per the City Engineer (Exhibit L), the three-quarter street improvements on Newton Street shall include a 28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot sidewalks on the north side of the roadway. At the February 25, 2019 Planning Commission hearing, the applicant's representative's memo (Exhibit W) indicated the applicant's "proposal to narrow this street was intended to allow the home on the adjacent property to the East to the saved." The Public Works Director and City Engineer (Exhibit X) provided the following response: "All the City has is the application before it. The City shouldn't speculate on what may or may not happen on the adjacent property. The adjacent property owner has been noticed and if they have a better idea of what the future holds for their property or the existing dwelling than [sic] they should submit testimony or comments reflecting this. A 40 ft. right-of-way dedication and a ³/₄-street improvement should be required to serve more than two single-family dwellings. If the applicant wants to accommodate future development of the parcel to the east then the applicant needs to figure out how to do this and still provide the minimum

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

³/4-street section on the development site or acquire land to accommodate it on adjacent properties." The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a three-quarter street.

The preliminary street and utility plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW dedication for staff review and approval.

<u>Camden Court</u>: Camden Court requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

- E. As necessary to provide for orderly development of adjacent properties, public streets installed concurrent with development of a site shall be extended through the site to the edge of the adjacent property(ies) in accordance with the following:
 - 1. Temporary dead-ends created by this requirement to extend street improvements to the edge of adjacent properties may be installed without turn-arounds, subject to the approval of the Fire Marshal.
 - 2. In order to assure the eventual continuation or completion of the street, reserve strips may be required.

RESPONSE: The applicant is not proposing any permanent dead-end streets. The applicant proposes that Newton Street, Woodstock Street, and Middlebury Avenue be temporary dead-end streets with construction of this subdivision until such a time as these streets are extended onto the adjoining properties to the east and south. The applicant shall plat a vehicle non-access reserve (VNAR) strip at the east end of Woodstock Street and along the south side of the right-of-way of Newton Street, with the exception of the easternmost end of the south side of Newton Street to allow access to the newly constructed street from the existing gravel drive; the length of the VNAR for the south side of Newton Street shall be determined during Construction Plan review. The dead-ends on Woodstock Street and Newton Street would be approximately 190 feet. The applicant shall work with the Fire Marshall to determine if a temporary turn-around is needed on Woodstock Street and/or Newton Street, or if houses on particular lots need to be sprinkled. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

F. Where required by the Planning Commission or Director, public street improvements may be required through a development site to provide for the logical extension of an existing street network or to connect a site with a nearby neighborhood activity center, such as a school or park. Where this creates a land division incidental to the development, a land partition shall be completed concurrent with the development.

RESPONSE: The proposal includes the extension of Newton Street from the Cascadia Village subdivision.

G. Except for extensions of existing streets, no street names shall be used that will duplicate or be confused with names of existing streets. Street names and numbers shall conform to the established pattern in the surrounding area and be subject to approval of the Director.

RESPONSE: The applicant is proposing the following new street names: Camden Court, Woodstock Street, and Middlebury Avenue. **The City of Sandy reserves the right to name streets.**

- H. Location, grades, alignment, and widths for all public streets shall be considered in relation to existing and planned streets, topographical conditions, public convenience and safety, and proposed land use. Where topographical conditions present special circumstances, exceptions to these standards may be granted by the City Engineer provided the safety and capacity of the street network is not adversely affected. The following standards shall apply:
 - 1. Location of streets in a development shall not preclude development of adjacent properties. Streets shall conform to planned street extensions identified in the Transportation Plan and/or provide for continuation of the existing street network in the surrounding area.
 - 2. Grades shall not exceed 6 percent on arterial streets, 10 percent on collector streets, and 15 percent on local streets.
 - 3. As far as practical, arterial streets and collector streets shall be extended in alignment with existing streets by continuation of the street centerline. When staggered street alignments resulting in "T" intersections are unavoidable, they shall leave a minimum of 150 ft. between the nearest edges of the two rights-of-way.
 - 4. Centerline radii of curves shall not be less than 500 ft. on arterial streets, 300 ft. on collector streets, and 100 ft. on local streets.
 - 5. Streets shall be designed to intersect at angles as near as practicable to right angles and shall comply with the following:
 - a) The intersection of an arterial or collector street with another arterial or collector street shall have a minimum of 100 ft. of straight (tangent) alignment perpendicular to the intersection.
 - b) The intersection of a local street with another street shall have a minimum of 50 ft. of straight (tangent) alignment perpendicular to the intersection.
 - c) Where right angle intersections are not possible, exceptions can be granted by the City Engineer provided that intersections not at right angles have a minimum corner radius of 20 ft. along the right-of-way lines of the acute angle.
 - d) Intersections with arterial streets shall have a minimum curb corner radius of 20 ft. All other intersections shall have a minimum curb corner radius of 10 ft.
 - 6. Right-of-way and improvement widths shall be as specified by the Transportation System Plan. Exceptions to those specifications may be approved by the City Engineer to deal with specific unique physical constraints of the site.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) does not show the correct measurement of 50 feet of tangent alignment at the intersection of Jacoby Road and Newton Street as required by 17.84.50(H.5.b). The applicant measured the 50 feet of tangent alignment from the centerline of Jacoby Road rather than from the curb line extended. **The applicant shall revise the alignment of Newton Street to conform to the 50 feet of tangent alignment standard.**

17.84.60 PUBLIC UTILITY EXTENSIONS

- A. All development sites shall be provided with public water, sanitary sewer, and storm drainage.
- B. Where necessary to serve property as specified in "A" above, required public utility installations shall be constructed concurrent with development.
- C. Off-site public utility extensions necessary to fully serve a development site and adjacent properties shall be constructed concurrent with development.
- D. As necessary to provide for orderly development of adjacent properties, public utilities installed concurrent with development of a site shall be extended through the site to the edge of adjacent property(ies).
- E. All public utility installations required with development shall conform to the City's facilities master plans.
- F. Private on-site sanitary sewer and storm drainage facilities may be considered provided all the following conditions exist:
 - 1. Extension of a public facility through the site is not necessary for the future orderly development of adjacent properties;
 - 2. The development site remains in one ownership and land division does not occur (with the exception of land divisions that may occur under the provisions of 17.84.50 F above);
 - 3. The facilities are designed and constructed in accordance with the Uniform Plumbing Code and other applicable codes, and permits and/or authorization to proceed with construction is issued prior to commencement of work.

RESPONSE: The applicant's Preliminary Street and Utility Plan (Exhibit C, Sheet 5) depicts the location and type of proposed public utilities including water, sanitary sewer and stormwater. All public utility installations shall conform to the City's facilities master plans. Staff recommends the applicant revise the utility plan to include broadband fiber locations as detailed by the SandyNet Manager. Per the City Engineer (Exhibit L), all public sanitary sewer and waterline mains shall be a minimum of 8 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. No building permits will be issued until all public utilities including sanitary sewer are available to serve the development. The applicant shall pay plan review, inspection, and permit fees as determined by the Public Works Director. The utility improvements proposal and requirements for the Jacoby Heights subdivision are further detailed in Sections 17.100.230, 17.100.240, and 17.100.250 below.

17.84.80 FRANCHISE UTILITY INSTALLATIONS

These standards are intended to supplement, not replace or supersede, requirements contained within individual franchise agreements the City has with providers of electrical power, telephone, cable television, and natural gas services (hereinafter referred to as "franchise utilities").

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx
RESPONSE: Private utility services will be submitted for review and approval by service providers and City staff in association with construction plans, and all utility lines will be extended to the perimeter of the site. All franchise utilities shall be installed underground and in conformance with City standards. PGE submitted a comment (Exhibit N) stating they did not find any conflicts related to the project. Per PGE's request, the applicant shall call the PGE Service Coordinators at 503-323-6700 when the developer is ready to start the project.

17.84.90 LAND FOR PUBLIC PURPOSES

- A. Easements for public sanitary sewer, water, storm drain, pedestrian and bicycle facilities shall be provided whenever these facilities are located outside a public right-of-way in accordance with the following:
 - 1. When located between adjacent lots, easements shall be provided on one side of a lot line.
 - 2. The minimum easement width for a single utility is 15 ft. The minimum easement width for two adjacent utilities is 20 ft. The easement width shall be centered on the utility to the greatest extent practicable. Wider easements may be required for unusually deep facilities.
- B. Public utility easements with a minimum width of 5 feet shall be provided adjacent to all street rights-of-way for franchise utility installations.
- C. Where a development site is traversed by a drainageway or water course, a drainage way dedication shall be provided to the City.
- D. Where a development is traversed by, or adjacent to, a future trail linkage identified within the Transportation System Plan, dedications of suitable width to accommodate the trail linkage shall be provided. This width shall be determined by the City Engineer, considering the type of trail facility involved.
- E. Where existing rights-of-way and/or easements within or adjacent to development sites are nonexistent or of insufficient width, dedications may be required. The need for and widths of those dedications shall be determined by the City Engineer.
- F. Where easement or dedications are required in conjunction with land divisions, they shall be recorded on the plat. Where a development does not include a land division, easements and/or dedications shall be recorded on standard document forms provided by the City Engineer.
- G. If the City has an interest in acquiring any portion of a proposed subdivision or planned development site for a public purpose, other than for those purposes listed above, or if the City has been advised of such interest by a school district or other public agency, and there is a reasonable assurance that steps will be taken to acquire the land, the Planning Commission may require those portions of the land be reserved for public acquisition for a period not to exceed 1 year.
- H. Environmental assessments for all lands to be dedicated to the public or City may be required to be provided by the developer. An environmental assessment shall include information necessary for the City to evaluate potential liability for environmental hazards, contamination, or required waste cleanups related to the dedicated land. An environmental assessment shall be completed prior to the acceptance of dedicated lands in accordance with the following:
 - The initial environmental assessment shall detail the history of ownership and general use of the land by past owners. Upon review of the information provided by the grantor, as well as any site investigation by the City, the Director will determine if the risks of potential contamination warrant further investigation. When further site investigation is warranted, a Level I Environmental Assessment shall be provided by the grantor.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A). Rather than include the FSH overlay area as part of Lots 19, 20, 21, 26, 27, and 28, the applicant shall designate the FSH area as a separate tract on the plat, which shall be owned and maintained by an HOA.

The Plat shall detail the following easements:

- An eight-foot wide public utility easement (PUE) along the frontage of all proposed lots;
- Crossover easements along the common parcel line recorded between pairs of lots that share driveways (Lots 26 and 27, and Lots 28 and 29);
- 20 foot easement along the west boundary of Lots 9 and 10 for public sanitary sewer and public storm line. Per the Public Works Director (Exhibit O), the City reserves the right to widen this easement depending on the final design depth of the sanitary sewer and storm lines per Section 17.84.90(A.2);
- A 15 foot easement between Lot 30 and Tract A for public sanitary sewer;
- A 10 foot private storm drain easement at the rear of Lots 1-18, 22-25, 28, 29, 31, and 32 and a 10 foot private storm drain easement across Lots 19-21 just south of the FSH boundary;
- A blanket conservation easement and pedestrian easement over the entirety of the FSH tract;
- An 8 foot wide pedestrian maintenance access easement to the FSH tract (could be located on the east edge of Lot 19 or elsewhere); and
- A vehicle non-access reserve (VNAR) strip in the following locations:
 - Jacoby Road frontage of Lots 9, 10, 30 and 32 and Tract A; and,
 - South side of the right-of-way of Newton Street, with the exception of the easternmost section as determined necessary to allow access to the newly constructed Newton Street from the existing gravel driveway during Construction Plan review; and,
 - East end of the right-of-way of Woodstock Street.

17.84.100 MAIL DELIVERY FACILITIES

RESPONSE: The applicant will need to coordinate with the United States Postal Service (USPS) to locate mail facilities and these will be approved by the City and USPS. **Mail delivery facilities shall be** provided by the applicant in conformance with Section 17.84.100 and the standards of the USPS. The applicant shall submit a mail delivery plan, featuring grouped lockable mail facilities, to the City and USPS for review and approval prior to installation of mailboxes.

CHAPTER 17.86 – PARKLAND AND OPEN SPACE

17.86.10 MINIMUM PARKLAND DEDICATION REQUIREMENTS

<u>Parkland Dedication:</u> New residential subdivisions, planned developments, multi-family or manufactured home park developments shall be required to provide parkland to serve existing and future residents of those developments. Multi-family developments which provide some "congregate" services

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

and/or facilities, such as group transportation, dining halls, emergency monitoring systems, etc., but which have individual dwelling units rather than sleeping quarters only, are considered to be multi-family developments for the purpose of parkland dedication. Licensed adult congregate living facilities, nursing homes, and all other similar facilities which provide their clients with individual beds and sleeping quarters, but in which all other care and services are communal and provided by facility employees, are specifically exempt from parkland dedication and system development fee requirements.

- 1. The required parkland shall be dedicated as a condition of approval for the following:
 - a. Tentative plat for a subdivision or partition;
 - b. Planned Development conceptual or detailed development plan;
 - c. Design review for a multi-family development or manufactured home park; and
 - d. Replat or amendment of any site plan for multi-family development or manufactured home park where dedication has not previously been made or where the density of the development involved will be increased.
- 2. Calculation of Required Dedication: The required parkland acreage to be dedicated is based on a calculation of the following formula rounded to the nearest 1/100 (0.00) of an acre:

Required parkland dedication (acres) = (proposed units) x (persons/unit) x 0.0043 (per person park land dedication factor)

a. Population Formula: The following table shall be used to determine the number of persons per unit to be used in calculating required parkland dedication:

Type of Unit	Total Persons Per Unit	
Single family residential	3.0	
Standard multi-family unit	2.0	
Manufactured dwelling park	2.0	
Congregate multi-family unit	1.5	

Persons per unit, age distribution, and local conditions change with time. The specific formula for the dedication of land will, therefore, be subject to periodic review and amendment.

b. Per Person Parkland Dedication Factor: The total parkland dedication requirement shall be 0.0043 of an acre per person based on the adopted standard of 4.3 acres of land per one thousand of ultimate population per the Parks Master Plan¹. This standard represents the citywide land-to-

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

¹ Parks Master Plan, Implementation Plan section, Pages 4 and 5 indicate a required park acreage total of 64.5 acres. This number, divided by population (2015) of 15,000 equates to 4.3 acres per 1000 population or 0.0043 per person.

population ratio for city parks, and may be adjusted periodically through amendments to the Parks Master Plan.

RESPONSE: The applicant proposes 32 single-family detached dwellings with this subdivision request. Based upon the calculations adopted by the City and specified within Section 17.86.10, the required dedication area is 0.41 acres of public parkland (32 proposed units x 3 persons per unit x .0043=0.4128 rounded to the nearest 1/100 of an acre = 0.41 acres to be dedicated).

17.86.40 CASH IN LIEU OF DEDICATION

At the city's discretion only, the city may accept payment of a fee in lieu of land dedication. The city may require payment in lieu of land when the park land to be dedicated is less than 3 acres. A payment in lieu of land dedication is separate from Park Systems Development Charges, and is not eligible for a credit of Park Systems Development Charges. The amount of the fee in lieu of land dedication (in dollars per acre) shall be set by City Council Resolution, and it shall be based on the typical market value of developed property (finished lots) in Sandy net of related development costs.

- 1. The following factors shall be used in the choice of whether to accept land or cash in lieu:
 - a. The topography, geology, access to, parcel size, and location of land in the development available for dedication;
 - b. Potential adverse/beneficial effects on environmentally sensitive areas;
 - c. Compatibility with the Parks Master Plan, Public Facilities element of the Comprehensive Plan, and the City of Sandy Capital Improvements Program in effect at the time of dedication;
 - d. Availability of previously acquired property; and
 - e. The feasibility of dedication.
- 2. A. Cash in lieu of parkland dedication shall be paid prior to approval of the final plat or as specified below:
 - a. 50 percent of the payment shall be paid prior to final plat approval, and
 - b. The remaining 50 percent of the payment pro-rated equally among the lots, plus an administrative surcharge, shall be paid as specified by City Council Resolution.

RESPONSE: Based on the calculations specified in Section 17.86.10, the applicant is responsible for dedicating 0.41 acres of public parkland based on 32 dwelling units. No parkland is identified on the subject property in the Parks Master Plan. For these reasons, the applicant shall pay a fee in lieu for the required parkland dedication per the adopted Fee Resolution. Per Resolution 2013-14, the required fee in lieu amount is \$241,000 per acre if the entire amount is paid prior to final plat approval. Therefore, based on the current Fee Resolution, the applicant would be required to pay a fee in lieu of dedication for a total of \$98,810 (0.41 acres of land to be dedicated x \$241,000). Alternatively, Ordinance 2013-03 allows the applicant to pay a minimum of 50 percent of the fee to receive final plat approval with the remaining balance to be paid as a proportionate amount with each building permit. If

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

a portion of the fee is deferred, Resolution 2013-14 specifies a per acre fee of \$265,000. Currently, the Fee Resolution requires payment of \$108,650 if a portion of the fee is deferred, a minimum of 50 percent (\$54,325) paid prior to final plat approval and the remaining 50 percent (\$54,325) divided between the 32 lots (\$1,697.66/lot). If the number of lots is modified from 32, the parks fee in lieu will need to be recalculated.

17.86.50 MINIMUM STANDARDS FOR OPEN SPACE DEDICATION

The applicant through a subdivision or design review process may propose the designation and protection of open space areas as part of that process. This open space will not, however, be counted toward the parkland dedication requirement of Sections 17.86.10 through 17.86.40.

4. Open space areas shall be maintained so that the use and enjoyment thereof is not diminished or destroyed. Open space areas may be owned, preserved, and maintained by any of the following mechanisms or combinations thereof:

a. Dedication to the City of Sandy or an appropriate public agency approved by the City, if there is a public agency willing to accept the dedication. Prior to acceptance of proposed open space, the City may require the developer to submit a Phase I Environmental Site Assessment completed by a qualified professional according to American Society of Testing and Materials (ASTM) standards (ASTM E 1527). The results of this study shall indicate a clean environmental record.

b. Common ownership by a homeowner's association that assumes full responsibility for its maintenance;

c. Dedication of development rights to an appropriate public agency with ownership remaining with the developer or homeowner's association. Maintenance responsibility will remain with the property owner; and/or

d. Deed-restricted private ownership preventing development and/or subsequent subdivision and providing for maintenance responsibilities.

RESPONSE: The applicant is not proposing any dedication of open space. However, the Parks Board recommends that the FSH analysis tract be dedicated to the City. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the tract without knowing the condition that it is in and recommended that, prior to dedicating the FSH tract to the City, the applicant would need to: 1) submit a Phase I environmental assessment and a native vegetation report for the FSH area to be dedicated in the tract, 2) remove all invasive plants and re-plant any areas of exposed soil with native plants, and 3) submit a maintenance bond for two full growing seasons (May 1 – October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Staff supported these recommendations in the February 25, 2019 staff report; however, the applicant's representative indicated that the applicant did not want to incur the expense of having to restore the FSH tract and conduct a Phase I environmental assessment. If the FSH tract is not dedicated to the City, it will not be necessary to perform a Phase I environmental assessment. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

pedestrian easement over the entirety of the FSH tract. Due to the site layout, Lots 19-21 will end up less than 7,500 square feet once the FSH area is put into a separate tract. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot area and allowing the rear setback to be reduced to 15 feet. This will help avoid confusion over property ownership, fence lines, and unnecessary or illegal removal of riparian vegetation and vegetation on steep slopes.

5. In the event that any private owner of open space fails to maintain it according to the standards of this Code, the City of Sandy, following reasonable notice, may demand that the deficiency of maintenance be corrected, and may enter the open space for maintenance purposes. All costs thereby incurred by the City shall be charged to those persons having the primary responsibility for maintenance of the open space.

RESPONSE: The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants. In the event that the HOA fails to maintain the FSH tract according to the standards of the Sandy Municipal Code, the City of Sandy, following reasonable notice, may demand that the deficiency of maintenance be corrected, and may enter the FSH tract area for maintenance purposes. All costs thereby incurred by the City shall be charged to the HOA.

CHAPTER 17.92 – LANDSCAPING AND SCREENING

17.92.10 GENERAL PROVISIONS

- C. Significant plant and tree specimens should be preserved to the greatest extent practicable and integrated into the design of a development. Trees of 25-inches or greater circumference measured at a height of 4-½ ft. above grade are considered significant. Plants to be saved and methods of protection shall be indicated on the detailed planting plan submitted for approval. Existing trees may be considered preserved if no cutting, filling, or compaction of the soil takes place between the trunk of the tree and the area 5-ft. outside the tree's drip line. Trees to be retained shall be protected from damage during construction by a construction fence located 5 ft. outside the dripline.
- L. All landscaping shall be continually maintained, including necessary watering, weeding, pruning and replacing.

RESPONSE: Per Section 17.92.10 (C), trees over 25-inches circumference measured at a height of $4-\frac{1}{2}$ feet above grade are considered significant and should be preserved to the greatest extent practicable and integrated into the design of a development. A 25-inch circumference tree measured at $4-\frac{1}{2}$ feet above grade has roughly an eight-inch diameter at breast height (DBH). Based on the Planning Commission interpretation from February 25, 2019, Subsection 17.92.10(C) does not apply to subdivisions. Tree protection fencing and tree retention will be discussed in more detail under Chapter 17.102 in this document. Per Section 17.92.10(L), all landscaping shall be continually maintained, including necessary watering, weeding, pruning, and replacing.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Note: the applicant is requesting a code interpretation to determine if Section 17.92.10(C) applies to subdivisions.

17.92.30 REQUIRED TREE PLANTINGS

Planting of trees is required for all parking lots with 4 or more parking spaces, public street frontages, and along private drives more than 150 feet long. Trees shall be planted outside the street right-of-way except where there is a designated planting strip or City adopted street tree plan.

The City maintains a list of appropriate trees for street tree and parking lot planting situations. Selection of species should be made from the city-approved list. Alternate selections may be approved by the Director following written request. The type of tree used shall determine frequency of trees in planting areas. Trees in parking areas shall be dispersed throughout the lot to provide a canopy for shade and visual relief.

Area/Type of Planting	Canopy	Spacing
Street Tree	Medium	30 ft. on center
Street Tree	Large	50 ft. on center

Trees may not be planted:

- Within 5 ft. of permanent hard surface paving or walkways, unless specific species, special planting techniques and specifications approved by the Director are used.
- Unless approved otherwise by the City Engineer:
 - * Within 10 ft. of fire hydrants and utility poles
 - * Within 20 ft. of street light standards
 - * Within 5 ft. from an existing curb face
 - * Within 10 ft. of a public sanitary sewer, storm drainage or water line
- Where the Director determines the trees may be a hazard to the public interest or general welfare.
- Trees shall be pruned to provide a minimum clearance of 8 ft. above sidewalks and 12 ft. above street and roadway surfaces.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires trees spaced 30 feet on center along street frontages. The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all of the proposed streets. The proposed plan details 84 street trees. Two (2) of the street trees are proposed to be located in driveways in front of Lots 31 and 32; however, there appears to be sufficient space to relocate at least one of the trees outside of the driveway area. The applicant shall update the plan set to remove street trees from the driveways in front of Lots 31 and 32; at least one of the street trees shall be relocated outside of the driveway area.

The applicant is proposing to mass grade the buildable portion of the site. This will remove top soil and heavily compact the soil. In order to maximize the success of the required street trees, the applicant shall aerate the planter strips to a depth of 3 feet prior to planting street trees. The applicant shall either aerate the planter strip soil at the subdivision stage and install fencing around the planter strips to protect the soil from compaction or shall aerate the soil at the

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

individual home construction phase. The applicant shall call for an inspection with the City after aerating the soil and before planting the street trees.

The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." With extension of the sidewalk and planter strip to the southern property boundary, the proposal could accommodate a third street tree on the south side of Newton Street. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW dedication for staff review and approval. The proposed layout would likely accommodate an additional two (2) street trees on the east end of Newton Street. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. If the plans change in a way that affects the number of street trees (e.g., driveway locations), the applicant shall submit an updated street tree plan for staff review and approval. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted, staked, and the planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed after one growing season (or a maximum of 1 year).

17.92.40 IRRIGATION

Landscaping shall be irrigated, either with a manual or automatic system, to sustain viable plant life.

RESPONSE: Section 17.92.40 requires that all landscaping shall be irrigated, either with a manual or automatic system. As required by Section 17.92.140, the developer and lot owners shall be required to maintain all vegetation planted in the development for two years from the date of completion, and shall replace any dead or dying plants during that period.

17.92.50 TYPES AND SIZES OF PLANT MATERIALS

- A. At least 75% of the required landscaping area shall be planted with a suitable combination of trees, shrubs, or evergreen ground cover except as otherwise authorized by Chapter 17.92.10 F.
- B. Plant Materials. Use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged where possible.
- C. Trees shall be species having an average mature spread of crown greater than 15 feet and having trunks which can be maintained in a clear condition with over 5 feet of clear wood (without branches). Trees having a mature spread of crown less than 15 feet may be substituted by grouping the same so as to create the equivalent of a 15-foot crown spread.
- D. Deciduous trees shall be balled and burlapped, be a minimum of 7 feet in overall height or 1 ¹/₂ inches in caliper measured 6 inches above the ground, immediately after planting. Bare root trees will be acceptable to plant during their dormant season.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- E. Coniferous trees shall be a minimum five feet in height above ground at time of planting.
- F. Shrubs shall be a minimum of 1 gallon in size or 2 feet in height when measured immediately after planting.
- G. Hedges, where required to screen and buffer off-street parking from adjoining properties shall be planted with an evergreen species maintained so as to form a continuous, solid visual screen within 2 years after planting.
- H. Vines for screening purposes shall be a minimum of 1 gallon in size or 30 inches in height immediate after planting and may be used in conjunction with fences, screens, or walls to meet physical barrier requirements as specified.
- I. Groundcovers shall be fully rooted and shall be well branched or leafed. If used in lieu of turf in whole or in part, ground covers shall be planted in such a manner as to provide complete coverage in one year.
- J. Turf areas shall be planted in species normally grown as permanent lawns in western Oregon. Either sod or seed are acceptable. Acceptable varieties include improved perennial ryes and fescues used within the local landscape industry.
- K. Landscaped areas may include architectural features or artificial ground covers such as sculptures, benches, masonry or stone walls, fences, rock groupings, bark dust, decorative hard paving and gravel areas, interspersed with planted areas. The exposed area developed with such features shall not exceed 25% of the required landscaped area. Artificial plants are prohibited in any required landscape area.

RESPONSE: Street trees are typically required to be a minimum caliper of 1.5-inches measured 6 inches from grade. In the past, staff has recommended a larger caliper street tree (2-inches) for subdivision proposals where a substantial number of existing trees were proposed for removal. However, staff has received feedback that fewer species are available at the 2-inch caliper size. To provide species diversity, staff recommends requiring 1.5-inch caliper street trees. All street trees shall be a minimum of 1.5-inches in caliper measured 6 inches above the ground and shall be planted per the City of Sandy standard planting detail. The applicant proposes six distinct street tree species with one tree species per block face. Staff would like to see more diversity in street tree species in general and within each block. The applicant shall update the plan set to detail a minimum of two (2) different tree species per block face for staff review and approval. The proposed street tree list includes paperbark maple and Patmore ash. Due to concerns with Asian Longhorn Beetle and Emerald Ash Borer, staff do not recommend planting new maples or ashes at this time. The applicant shall update the proposed street tree list with at least two additional species from the City of Sandy Street Tree List that aren't maples or ashes. Staff recommends two (or more) species from the following list: cascara (Rhamnus purshiana), kobus magnolia (Magnolia kobus), Skyrocket oak (Quercus robur 'Skyrocket'), Japanese stewartia (Stewartia pseudocamellia), Japanese snowbell (Styrax japonica), or eastern redbud (Cercis canadensis). Providing species diversity adds visual interest and provides additional resistance against disease in conformance with City of Sandy Comprehensive Goal 5 Policy 28. The applicant also proposed Halka ginkgo. Technically, the only ginkgo on the City of Sandy street tree list is Saratoga ginkgo (Ginkgo biloba 'Saratoga'); however, staff recommends that Halka ginkgo be approved as a street tree with this application.

17.92.60 REVEGETATION IN UNLANDSCAPED OR NATURAL AREAS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. Areas where natural vegetation has been removed or damaged through grading or construction activity in areas not affected by the landscaping requirements and that are not to be occupied by structures or other improvements shall be replanted.
- B. Plant material shall be watered at intervals sufficient to assure survival and growth.
- C. The use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged to reduce irrigation and maintenance demands.

RESPONSE: The applicant did not submit any plans for re-vegetation of areas damaged through grading/construction, although most of the areas affected by grading will be improved. **The** applicant shall maintain all unlandscaped and/or revegetated areas for a period of two years following the date of recording of the final plat associated with those improvements.

17.92.130 PERFORMNCE BOND

If weather conditions or other circumstances beyond the control of the developer or owner make completion of the landscaping impossible prior to desired occupancy, an extension of up to 6 months may be applied for by posting "security" equal to 120% of the cost of the landscaping, assuring installation within 6 months. "Security" may consist of a performance bond payable to the city, cash, certified check, time certificates of deposit, assignment of a saving account, letter of credit, or other such assurance of access to funds necessary for completion as shall meet the approval of the City Attorney. Upon acceptance of the security, the developer or owner may be allowed occupancy for a period of up to 180 days. If the installation of the landscaping improvement is not completed within 180 days, the City shall have access to the security to complete the installation and/or revoke occupancy. Upon completion of the installation by the city, any portion of the remaining security shall be assessed against the property and the City shall thereupon have a valid lien against the property, which will come due, and payable.

RESPONSE: The applicant has the option to defer the installation of street trees and/or landscaping for weather-related reasons. Staff recommends the applicant utilize this option rather than install trees and landscaping during the dry summer months. If the applicant chooses to postpone street tree and/or landscaping installation, the applicant shall post a performance bond equal to 120 percent of the cost of the street trees/landscaping, assuring installation within 6 months. The cost of street trees shall be based on the street tree plan and at least \$500 per tree fee, including a three-year maintenance and warranty period. The cost of landscaping shall be based on the average of three estimates from three landscaping contractors; the estimates shall include as separate items all materials, labor, and other costs of the required action, including a two-year maintenance and warranty period.

CHAPTER 17.98 - PARKING, LOADING, AND ACCESS REQUIREMENTS

17.98.20 OFF-STREET PARKING REQUIREMENTS

RESPONSE: Section 17.98.20 requires two off-street parking spaces per single family detached dwelling unit. The 32 dwelling units proposed in this subdivision requires 64 off-street parking spaces. Each lot will have a driveway and the ability to construct a double car garage.

17.98.50 SETBACKS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. Parking areas, which abut a residential zoning district, shall meet the setback of the most restrictive adjoining residential zoning district.
- B. Required parking shall not be located in a required front or side yard setback area abutting a public street except in industrial districts. For single family and two-family dwellings, required off-street parking may be located in a driveway.
- C. Parking areas shall be setback from a lot line adjoining a street the same distance as the required building setbacks. Regardless of other provisions, a minimum setback of 5 feet shall be provided along the property fronting on a public street. The setback area shall be landscaped as provided in this code.

RESPONSE: Garages are required to be at least 22 feet setback from the front property line to meet setback requirements in the SFR zoning district. The Preliminary Plat (Exhibit C, Sheet 1) details a typical 22 foot garage setback.

17.98.60 DESIGN, SIZE AND ACCESS

All off-street parking facilities, vehicular maneuvering areas, driveways, loading facilities, accessways, and private streets shall conform to the standards set forth in this section.

- A. Parking Lot Design. All areas for required parking and maneuvering of vehicles shall have a durable hard surface such as concrete or asphalt.
- B. Size of Space.
 - 1. A standard parking space shall be 9 feet by 18 feet.
 - 2. A compact parking space shall be 8 feet by 16 feet.
 - 3. Handicapped parking spaces shall be 13 feet by 18 feet. Accessible parking shall be provided for all uses in compliance with the requirements of the State of Oregon (ORS 447.233) and the Americans with Disabilities Act.
 - 4. Parallel parking spaces shall be a length of 22 feet. No more than 35 percent of the parking stalls shall be compact spaces

RESPONSE: The applicant shall comply with the parking standards in Section 17.98.60. The parking areas in front of the proposed garages for all lots need to be at least 10 feet in width by 20 feet in length. Driveways for single family homes are required to be at least 10 feet wide as detailed in Section 17.98.100 below. The garages shall be adequate depth to park a vehicle and the on-street parking spaces shall be at least 22 feet in length.

17.98.80 ACCESS TO ARTERIAL AND COLLECTOR STREETS

A. Location and design of all accesses to and/or from arterials and collectors (as designated in the Transportation System Plan) are subject to review and approval by the City Engineer. Where practical, access from a lower functional order street may be required. Accesses to arterials or collectors shall be located a minimum of 150 ft. from any other access or street intersection. Exceptions may be granted by the City Engineer. Evaluations of exceptions shall consider posted speed of the street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- B. No development site shall be allowed more than one access point to any arterial or collector street (as designated in the Transportation System Plan) except as approved by the City Engineer. Evaluations of exceptions shall consider posted speed of street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.
- C. When developed property is to be expanded or altered in a manner that significantly affects on-site parking or circulation, both existing and proposed accesses shall be reviewed under the standards in A and B above. As a part of an expansion or alteration approval, the City may require relocation and/or reconstruction of existing accesses not meeting those standards.

RESPONSE: Jacoby Heights Subdivision is proposed to create two intersections with Jacoby Road, a designated collector street: Woodstock Street and Newton Street. All lots are required to have driveway access from the internal local street network with a vehicle non-access reserve strip (VNAR) detailed on the face of plat along Jacoby Road.

17.98.100 DRIVEWAYS

- A. A driveway to an off-street parking area shall be improved from the public roadway to the parking area a minimum width of 20 feet for a two-way drive or 12 feet for a one-way drive but in either case not less than the full width of the standard approach for the first 20 feet of the driveway.
- B. A driveway for a single-family dwelling shall have a minimum width of 10 feet.
- C. A driveway for a two-family dwelling shall have a minimum width of 20 feet. A driveway approach must be constructed in accordance with applicable city standards and the entire driveway must be paved with asphalt or concrete.
- D. Driveways, aisles, turnaround areas and ramps shall have a minimum vertical clearance of twelve feet for their entire length and width but such clearance may be reduced in parking structures.
- E. No driveway shall traverse a slope in excess of 15 percent at any point along the driveway length.
- F. The location and design of the driveway shall provide for unobstructed sight per the vision clearance requirements. Requests for exceptions to these requirements will be evaluated by the City Engineer considering the physical limitations of the lot and safety impacts to vehicular, bicycle, and pedestrian traffic.

RESPONSE: The minimum driveway width for a single-family dwelling is 10 feet. The Public Works driveway approach standard detail specifies a maximum of 24 feet wide for a residential driveway approach. The Preliminary Parking Plan (Exhibit C, Sheet 7) details driveway curb cuts for all lots. With the exception of a few lots on the cul-de-sac, all driveways are detailed at approximately 20 feet. The applicant is proposing a shared driveway between Lot 29 and Tract A at approximately 36 feet and a shared driveway between Lot 27 and 28 at approximately 26 feet. In order to maximize street tree planting areas and prevent conflicts with streetlights, water meters and dry utility facilities, shared driveway approaches may be required in cul-de-sac bulbs. Per the Public Works Director (Exhibit O), Lot 28 and 29 shall share a driveway approach maximum 24 feet in width.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Lot 26 and 27 shall share a driveway approach maximum 24 feet in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches. The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of Lot 29 to eliminate remnant of Tract A between the access drive and Lot 29. The driveway approach for Tract A shall be a maximum of 12 feet in width. Per the City Engineer (Exhibit L), the access to the detention pond shall be paved or all-weather surface.

17.98.130 PAVING

- A. Parking areas, driveways, aisles and turnarounds shall be paved with concrete, asphalt or comparable surfacing, constructed to city standards for off-street vehicle areas.
- B. Gravel surfacing shall be permitted only for areas designated for non-motorized trailer or equipment storage, propane or electrically powered vehicles, or storage of tracked vehicles.

RESPONSE: As required by Section 17.98.130, all parking, driveway and maneuvering areas shall be constructed of asphalt, concrete, or other approved material.

17.98.140 DRAINAGE

Parking areas, aisles and turnarounds shall have adequate provisions made for the on-site collection of drainage waters to eliminate sheet flow of such waters onto sidewalks, public rights-of-way and abutting private property.

RESPONSE: Section 17.98.140 contains requirements for drainage. Other sections of this order detail the stormwater requirements.

17.98.200 RESIDENTIAL ON-STREET PARKING REQUIREMENTS

A. Residential On-Street Parking Requirements. Residential on-street parking shall conform to the following standards:

- 1. In addition to required off-street parking, all new residential planned developments, subdivisions and partitions shall provide one (1) on-street parking space within 200 feet of each dwelling except as provided in Section 17.98.200(A)(6) below.
- 2. The location of residential on-street parking shall be reviewed for compliance with this section through submittal of a Residential Parking Analysis Plan as required in Section 17.98.10(M).
- 3. Residential on-street parking shall not obstruct required clear vision areas and shall not violate any local or state laws.
- 4. Parallel residential on-street parking spaces shall be 22 feet minimum in length.
- 5. Residential on-street parking shall be measured along the curb from the outside edge of a driveway wing or curb cut. Parking spaces must be set back a minimum of 15 feet from an intersection and may not be located within 10 feet of a fire hydrant.
- 6. Portions of residential on-street parking required by this section may be provided in parking courts that are interspersed throughout a development when the following standards are met:
 - a. No more than eight (8) parking spaces shall be provided in a parking court;

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- b. Parking spaces within a parking court shall be nine (9) feet wide and 18 feet in depth;
- c. Notwithstanding Section 17.98.70, vehicles parked in a parking court are permitted to back onto the public right-of-way from the parking court;
- d. A parking court shall be located within 200 feet of the dwellings requiring parking in accordance with the requirements of Section 17.98.10(M);
- e. No more than two (2) parking courts shall be provided within a block, with only one (1) parking court provided along a block face;
- f. A parking court shall be paved in compliance with the standards of this chapter and the latest adopted grading and drainage standards;
- g. If a parking court is adjacent to a public right-of-way, it shall be publicly owned and maintained;
- h. If a parking court is adjacent to a private drive, it shall be privately owned and maintained. For each parking court there shall be a legal recorded document which includes:
 - i. A legal description of the parking court;
 - ii. Ownership of the parking court;
 - iii. Use rights; and
 - iv. A maintenance agreement and the allocation and/or method of determining liability for maintenance of the parking court;
- i. A parking court shall be used solely for the parking of operable passenger vehicles.

RESPONSE: The Preliminary Parking Plan (Exhibit C, Sheet 7) identifies a total of 62 on-street parking spaces with at least one (1) on-street parking space within 200 feet of each of the 32 lots. The proposed parking plan shows 21 parking spaces on the south side of Newton Street, including parking spaces for Lots 1-4. These on-street parking spaces should not be counted toward meeting the on-street parking requirement since the south side of the street will eventually develop with driveways and/or streets that will eliminate these spaces or prevent the developer of the site to the south from counting these spaces toward their on-street parking requirement. Per the Public Works Director (Exhibit O), the applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval. No parking courts are proposed. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with Construction Plans. The applicant shall revise the Parking Analysis if required fire hydrants affect on-street parking spaces.

CHAPTER 17.100 - LAND DIVISION

Chapter 17.100 presents review procedures, design standards, and improvement requirements for land division. Section 17.100.60 is the subsection relevant to the division of subdivisions.

17.100.60 SUBDIVISIONS

Approval of a subdivision is required for a land division of 4 or more parcels in a calendar year. A twostep procedure is required for subdivision approval: (1) tentative plat review and approval; and (2) final plat review and approval.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: Submittal of preliminary utility plans is solely to satisfy the requirements of Section 17.100.60. Preliminary plat approval does not connote utility or public improvement plan approval which will be reviewed and approved separately upon submittal of public improvement construction plans.

- E. <u>Approval Criteria.</u> The Director or Planning Commission shall review the tentative plat for the subdivision based on the classification procedure (Type II or III) set forth in Section 17.12 and the following approval criteria:
 - The proposed subdivision is consistent with the density, setback and dimensional standards of the base zoning district, unless modified by a Planned Development approval. *RESPONSE:* The applicant requests subdivision approval for a subdivision that is in compliance with most of the applicable development standards. The applicant requests one variance to increase block length beyond the maximum block length standard, one variance to allow houses to not face a transit street, and one variance to allow single-tiered lots. As a result, the application for the subdivision and variances is being processed concurrently through a Type III procedure. The proposal is consistent with density and other dimensional standards of the base zoning district.
 - 2. The proposed subdivision is consistent with the design standards set forth in this chapter. *RESPONSE:* Consistency with design standards in this chapter are discussed under each subsection below. Conditions of approval can be adopted where necessary to bring the proposal into compliance with applicable standards.
 - The proposed street pattern is connected and consistent with the Comprehensive Plan or official street plan for the City of Sandy.
 RESPONSE: The proposed subdivision includes the extension of Newton Street to the east of Jacoby Road and three new proposed local streets, including one cul-de-sac. The subject property was not included in the City's current Transportation System Plan (TSP) because it was outside of the Urban Growth Boundary (UGB) when the TSP was adopted in 2011.
 - 4. Adequate public facilities are available or can be provided to serve the proposed subdivision. *RESPONSE:* All public facilities including water, sanitary sewer and stormwater are available or will be constructed by the applicant to serve the Jacoby Heights Subdivision.
 - 5. All proposed improvements meet City standards. **RESPONSE:** The detailed review of proposed improvements is contained in this report. Staff has identified a few aspects of the proposed subdivision improvements requiring additional information or modification by the applicant, but conditions of approval can be adopted to bring the proposal into compliance with all standards.
 - The phasing plan, if requested, can be carried out in a manner that meets the objectives of the above criteria and provides necessary public improvements for each phase as it develops.
 RESPONSE: The applicant did not submit a phasing plan. This subdivision request is for a 32 lot subdivision in one development phase.

17.100.70 LAND DIVISION DESIGN STANDARDS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

All land divisions shall be in conformance with the requirements of the applicable base zoning district and this chapter, as well as with other applicable provisions of this Code. Modifications to these requirements may be accomplished through a Planned Development. The design standards in this section shall be used in conjunction with street design standards included in the City of Sandy Transportation System Plan and standards and construction specifications for public improvements as set forth in adopted Public Facilities Plans and the Sandy Municipal Code.

RESPONSE: The applicant requested the following three Type III variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff supports the variance to front door orientation, the variance to allow a single-tiered lot configuration along Jacoby Road, and the variance to allow the block length to exceed 400 feet on the north side of Woodstock Street between Camden Court and the east property boundary. The tentative plat shall otherwise be designed to comply with all standards of the City of Sandy Development Code, Transportation System Plan, Facilities Master Plans and Sandy Municipal Code.

17.100.100 STREETS GENERALLY

No subdivision or partition shall be approved unless the development has frontage or approved access to an existing public street. In addition, all streets shall be graded and improved in conformance with the City's construction standards, approved by the City Engineer, in accordance with the construction plans.

A. <u>Street Connectivity Principle.</u> The pattern of streets established through land divisions should be connected to: (a) provide safe and convenient options for cars, bikes and pedestrians; (b) create a logical, recognizable pattern of circulation; and (c) spread traffic over many streets so that key streets (particularly U.S. 26) are not overburdened.

RESPONSE: The proposed development is moderately conducive to walking and biking while accommodating motor vehicles. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north.

- B. <u>Transportation Impact Studies.</u> Transportation impact studies may be required by the city engineer to assist the city to evaluate the impact of development proposals, determine reasonable and prudent transportation facility improvements and justify modifications to the design standards. Such studies will be prepared in accordance with the following:
 - 1. A proposal established with the scope of the transportation impact study shall be coordinated with, and agreed to, by the city engineer. The study requirements shall reflect the magnitude of the project in accordance with accepted transportation planning and engineering practices. A professional civil or traffic engineer registered in the State of Oregon shall prepare such studies.
 - 2. If the study identifies level-of-service conditions less than the minimum standards established in the Sandy Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered as part of the land use decision for the proposal.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: The submitted Transportation Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10th Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably per City of Sandy and ODOT standards and are projected to continue to operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development. The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and concurred that the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards.

C. <u>Topography and Arrangement.</u> All streets shall be properly related to special traffic generators such as industries, business districts, schools, and shopping centers and to the pattern of existing and proposed land uses.

RESPONSE: Considering the site's topography, the proposed street layout is acceptable given the topography and residential use of this site, and the topography and use of adjacent properties.

D. <u>Street Spacing</u>. Street layout shall generally use a rectangular grid pattern with modifications as appropriate to adapt to topography or natural conditions.

RESPONSE: The applicant proposes a rectangular pattern of streets with an offset grid. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north. However, future development to the south will be required to align with the proposed intersections in order to maintain a rectangular grid pattern and maximize pedestrian, bicycle, and vehicular connectivity. An offset grid with "T" intersections is not warranted due to the lack of through traffic.

E. <u>Future Street Plan.</u> Future street plans are conceptual plans, street extensions and connections on acreage adjacent to land divisions. They assure access for future development and promote a logical, connected pattern of streets. It is in the interest of the city to promote a logical, connected pattern of streets. All applications for land divisions shall provide a future street plan that shows the pattern of existing and proposed future streets within the boundaries of the proposed land divisions, proposed connections to abutting properties, and extension of streets to adjacent parcels within a 400 foot radius of the study area where development may practically occur.

RESPONSE: The proposed local street plan has been designed to facilitate the traffic needs of this development while ensuring there are no intersection conflicts with future development. Per the City Engineer (Exhibit L), the applicant shall provide a profile design for a minimum of 200 feet for all future street extensions beyond the project boundary to ensure future street grades can be met.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- F. <u>Connections.</u> Except as permitted under Exemptions, all streets, alleys and pedestrian walkways shall connect to other streets within the development and to existing and planned streets outside the development and to undeveloped properties which have no future street plan. Streets shall terminate at other streets or at parks, schools or other public land within a neighborhood.
 - 1. Where practicable, local roads shall align and connect with other roads when crossing collectors and arterials.
 - 2. Proposed streets or street extensions shall be located to provide direct access to existing or planned transit stops, and existing or planned neighborhood activity centers, such as schools, shopping areas and parks.
 - 3. Exemptions.
 - a. A future street plan is not required for partitions of residentially zoned land when none of the parcels may be redivided under existing minimum density standards.
 - b. Standards for street connections do not apply to freeways and other highways with full access control.
 - c. When street connection standards are inconsistent with an adopted street spacing standard for arterials or collectors, a right turn in/right turn out only design including median control may be approved. Where compliance with the standards would result in unacceptable sight distances, an accessway may be approved in place of a street connection.

RESPONSE: The proposed design extends Newton Street east across Jacoby Road into the site. All proposed streets will allow connection with future development to the south and east, with the exception of Camden Court, which is a cul-de-sac. Due to topography and the location of Tickle Creek, it is not practicable to connect Camden Court to the north.

17.100.110 STREET STANDARDS AND CLASSIFICATION

Street standards are illustrated in the figures included at the end of this chapter. Functional definitions of each street type are described in the Transportation System Plan as summarized below.

- A. Major arterials are designed to carry high volumes of through traffic, mixed with some unavoidable local traffic, through or around the city. Major arterials should generally be spaced at 1-mile intervals.
- B. Minor arterials are designed to collect and distribute traffic from major and minor arterials to neighborhood collectors and local streets, or directly to traffic destinations. Minor arterials should generally be spaced at 1-mile intervals.
- C. Residential minor arterials are a hybrid between minor arterial and collector type streets that allow for moderate to high traffic volumes on streets where over 90% of the fronting lots are residential.
- D. Collector streets are designed to collect and distribute traffic from higher type arterial streets to local streets or directly to traffic destinations. Collector streets should generally be spaced at 1/2-mile intervals.
- E. Local streets are designed to provide direct access to abutting property and connect to collector streets. A general spacing of 8-10 local streets per mile is recommended.
- F. Cul-de-sacs and dead end streets are discouraged. If deemed necessary, cul-de-sacs shall be as short as possible and shall not exceed 400 feet in length.
- G. Public access lanes are designed to provide primary access to a limited number of dwellings when the construction of a local street is unnecessary.
- H. Alleys are designed to provide access to multiple dwellings in areas where lot frontages are narrow and driveway spacing requirements cannot be met.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: Jacoby Road is an existing collector street. All proposed streets in the subdivision are local streets, including the extension of the existing Newton Street, which is proposed to extend east of Jacoby Road into the site. Camden Court is a proposed cul-de-sac and does not exceed 400 feet. Per the City Engineer (Exhibit L), the plans don't clearly indicate the radius for the proposed culde-sac. In the February 25, 2019 staff report, the City Engineer and Public Works Director recommended that the minimum paved cul-de-sac radius be 48 feet as per the Oregon Fire Code and Metro Code Committee. The applicant's representative (Exhibit W) indicated the applicant wants to provide a smaller cul-de-sac and cited the cul-de-sac requirements for manufactured home parks as a basis for allowing a smaller cul-de-sac radius. They also citied other cul-de-sacs in Sandy that don't meet the minimum radius requirement. The City Engineer and Public Works Director (Exhibit X) stated: "The fact that a cul-de-sac of this size was approved on another development does not allow staff to ignore the Fire Code standard on this site. The cul-de-sac bulb dimensions are not solely for emergency vehicle access. The larger radius allows enough room for on-street parking, garbage pickup, street sweeping and snow plowing. Even if lots 26-30 were constructed with residential fire protection systems I would not recommend reducing the cul-de-sac bulb radius.' After further discussion, the Public Works Director determined that a minimum 42 foot paved culde-sac radius would suffice if the Fire Department is willing to sign off on a 42 foot radius cul-desac. The minimum paved cul-de-sac radius shall be 42 feet; a review by the Fire Department to confirm cul-de-sac size compliance shall be required prior to approval of the construction plans.

17.100.120 BLOCKS AND ACCESSWAYS

- A. <u>Blocks</u>. Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features.
- B. <u>Residential Blocks.</u> Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance.

RESPONSE: The applicant is requesting variances to Sections 17.100.120(A) and (B). The proposal includes one lot (Lot 30) that will be a single-tiered lot and one block (north side of Woodstock Street between Camden Court and the eastern property boundary) that exceeds 400 feet.

D. Pedestrian and Bicycle Access Way Requirements. In any block in a residential or commercial district over 600 feet in length, a pedestrian and bicycle accessway with a minimum improved surface of 10 feet within a 15-foot right-of-way or tract shall be provided through the middle of the block. To enhance public convenience and mobility, such accessways may be required to connect to cul-de-sacs, or between streets and other public or semipublic lands or through greenway systems.

RESPONSE: None of the proposed blocks exceed 600 feet in length. At the pre-application meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trail on the adjacent property to the east (tax lot 800). Staff brought this proposal to

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board also recommends that the connector trail be constructed as part of this development. While the Park Facilities Plan (Exhibit Y) depicts the future Tickle Creek Trail located along Tickle Creek in the northeast corner of the subject property, the Park Facilities Plan doesn't go as far as identifying specific trail connections to the proposed future Tickle Creek Trail alignment. Without specific trail connectors identified explicitly on the plan, staff cannot compel the applicant to include a trail connecting to the future Tickle Creek Trail. However, staff and the Parks & Trails Advisory Board recommend the applicant include a connector trail on the subject property that provides access to the future Tickle Creek Trail.

17.100.130 EASEMENTS

A minimum eight (8) foot public utility easement shall be required along property lines abutting a rightof-way for all lots within a partition or subdivision. Where a partition or subdivision is traversed by a watercourse, drainage way, channel or stream, the land division shall provide a stormwater easement or drainage right-of-way conforming substantially with the lines of such watercourse, and such further width as determined needed for water quality and quantity protection.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) details an 8 foot wide public utility easement along all street frontages. The plat shall detail all proposed easements as detailed in Section 17.84.90 above.

17.100.180 INTERSECTIONS

- A. <u>Intersections.</u> Streets shall be laid out so as to intersect as nearly as possible at right angles. A proposed intersection of two new streets at an angle of less than 75 degrees shall not be acceptable. No more than two streets shall intersect at any one point unless specifically approved by the City Engineer. The city engineer may require left turn lanes, signals, special crosswalks, curb extensions and other intersection design elements justified by a traffic study or necessary to comply with the Development Code.
- B. <u>Curve Radius.</u> All local and neighborhood collector streets shall have a minimum curve radius (at intersections of rights-of-way) of 20 feet, unless otherwise approved by the City Engineer. When a local or neighborhood collector enters on to a collector or arterial street, the curve radius shall be a minimum of 30 feet, unless otherwise approved by the City Engineer.

RESPONSE: The proposed intersections are all right angles and meet the required minimum spacing standard of 150 feet as required in Section 17.84.50(C)(2).

17.100.190 STREET SIGNS

RESPONSE: The applicant will be required to pay for the cost of all street signs and the City will install the street signs.

17.100.210 STREET LIGHTING

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

insufficient. The locations of street light fixtures shall be reviewed in detail with construction plans.

17.100.220 LOT DESIGN

- A. The lot arrangement shall be such that there will be no foreseeable difficulties, for reason of topography or other conditions, in securing building permits to build on all lots in compliance with the Development Code.
- B. The lot dimensions shall comply with the minimum standards of the Development Code. When lots are more than double the minimum lot size required for the zoning district, the subdivider may be required to arrange such lots to allow further subdivision and the opening of future streets to serve such potential lots.
- C. The lot or parcel width at the front building line shall meet the requirements of the Development Code and shall abut a public street other than an alley for a width of at least 20 feet. A street frontage of not less than 15 feet is acceptable in the case of a flag lot division resulting from the division of an unusually deep land parcel which is of a size to warrant division into not more than two parcels.
- D. Double frontage lots shall be avoided except where necessary to provide separation of residential developments from arterial streets or to overcome specific disadvantages of topography or orientation.
- E. Lots shall avoid deriving access from major or minor arterials. When driveway access from major or minor arterials may be necessary for several adjoining lots, the Director or the Planning Commission may require that such lots be served by a common access drive in order to limit possible traffic hazards on such streets. Where possible, driveways should be designed and arranged to avoid requiring vehicles to back into traffic on minor or major arterials.

RESPONSE: The Single Family Residential (SFR) zoning district requires lots at least 7,500 square feet in area. The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes are required to comply with setback standards and maximum building height limitations as required in Chapter 17.34. Lot 30 is a double frontage lots that will take access from Camden Court but also abut Jacoby Road. Lots 19-21 and 27 contain more than double the minimum lot size; however, a large portion of each lot's area is within the FSH and, therefore, cannot be divided into a separate buildable lot. To avoid potential conflict between private property owners and the FSH overlay area in the future, the applicant shall designate the FSH analysis area as a separate tract, which shall be owned and maintained by an HOA. No lots are proposed to be accessed from a major or minor arterial. All lots are required to comply with clear vision requirements at all intersections.

17.100.230 WATER FACILITIES

Water lines and fire hydrants serving the subdivision or partition, and connecting the development to City mains, shall be installed to provide adequate water pressure to serve present and future consumer demand. The materials, sizes, and locations of water mains, valves, service laterals, meter boxes and other required appurtenances shall be in accordance with the standards of the Fire District, the City, and the State.

If the city requires the subdivider to install water lines in excess of eight inches, the city may participate in the oversizing costs. Any oversizing agreements shall be approved by the city manager based upon council policy and dependent on budget constraints. If required water mains will directly serve property

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement for the proportionate share of the cost.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of water facilities will be reviewed with construction plans. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. There is no other proposed connection to the water distribution system. Per the Public Works Director (Exhibit O), the applicant shall demonstrate that adequate fire and domestic flow will be available with a single point of connection to the water distribution system or connect to the existing 8-inch water line at the intersection of Street and Amherst Street. Per the City Engineer (Exhibit L), all new waterline mains shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The applicant's proposed Utility Plan (Exhibit C, Sheet 5) depicts new hydrants. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with construction plans.

17.100.240 SANITARY SEWERS

Sanitary sewers shall be installed to serve the subdivision and to connect the subdivision to existing mains. Design of sanitary sewers shall take into account the capacity and grade to allow for desirable extension beyond the subdivision.

If required sewer facilities will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement by nonparticipating landowners for the proportionate share of the cost of construction.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of sanitary sewer facilities will be reviewed with construction plans. Per the City Engineer (Exhibit L), all new public sanitary sewer lines shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The proposed utility plan shows an 8-inch sanitary sewer line extended south from the existing sewer line in Jacoby Road approximately 490 feet to the site. The site will drain to the existing SE sewer pump station located on Jacoby Road south of Trillium Avenue. Per the Public Works Director (Exhibit O), the applicant shall provide a sanitary sewer main line extension. At the February 25, 2019 Planning Commission hearing, the applicant's representative (Exhibit W) requested that the City reimburse the developer for the cost of the lateral. The Public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer for an Advanced Financed Public Improvement reimbursement from the benefitting property owner(s) per section 12.14 of the Municipal Code."

17.100.250 SURFACE DRAINAGE AND STORM SEWER SYSTEM

A. Drainage facilities shall be provided within the subdivision and to connect with off-site drainage ways or storm sewers. Capacity, grade and materials shall be by a design approved by the city engineer. Design of drainage within the subdivision shall take into account the location, capacity and grade necessary to maintain unrestricted flow from areas draining through the subdivision and to allow extension of the system to serve such areas.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- B. In addition to normal drainage design and construction, provisions shall be taken to handle any drainage from preexisting subsurface drain tile. It shall be the design engineer's duty to investigate the location of drain tile and its relation to public improvements and building construction.
- C. The roof and site drainage from each lot shall be discharged to either curb face outlets (if minor quantity), to a public storm drain or to a natural acceptable drainage way if adjacent to the lot.

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A) to be dedicated to the City of Sandy. Detained and treated discharge from the detention pond is proposed to be discharged to the adjacent property to the north. Per the Public Works Director(Exhibit O), the applicant shall demonstrate that the proposed subdivision does not exceed pre-development site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way. Per the City Engineer (Exhibit L), the detention pond shall meet the requirements of the 2016 City of Portland Stormwater Management Manual (SWMM) for landscaping, Section 2.4.1, and escape route, Section 2.30. All new public storm drains shall be a minimum of 12-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The City Engineer (Exhibit L) states the submitted preliminary stormwater calculations meet the water quality and water quantity criteria as stated in the City of Sandy Municipal Code Chapter 13.18 Standards and the City of Portland current Stormwater Management Manual (SWMM) Standards that were adopted by reference into the Sandy Development Code. Per the City Engineer, the applicant shall submit a detailed final stormwater report stamped by a licensed professional to the City for review and approval with the final construction plans.

17.100.260 UNDERGROUND UTILITIES

All subdivisions or major partitions shall be required to install underground utilities (including, but not limited to, electrical and telephone wiring). The utilities shall be installed pursuant to the requirements of the utility company.

RESPONSE: All utilities shall be placed underground.

17.100.270 SIDEWALKS

Sidewalks shall be installed on both sides of a public street and in any special pedestrian way within the subdivision.

RESPONSE: The applicant proposes constructing sidewalks along all public street frontages, with the exception of the proposed three-quarter street section and one-half street section of Newton Street.

17.100.290 STREET TREES

Where planting strips are provided in the public right-of-way, a master street tree plan shall be submitted and approved by the Director. The street tree plan shall provide street trees approximately every 30' on center for all lots.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires installation of trees along all street frontages. Street trees are discussed in Section 17.92.30 of this staff report.

17.100.300 EROSION CONTROL

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Grass seed planting shall take place prior to September 30th on all lots upon which a dwelling has not been started but the ground cover has been disturbed. The seeds shall be of an annual rye grass variety and shall be sown at not less than four pounds to each 1000 square feet of land area.

RESPONSE: The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a re-vegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code and as detailed below. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

17.100.310 REQUIRED IMPROVEMENTS

The following improvements shall be installed at no expense to the city, consistent with the design standards of Chapter 17.84, except as otherwise provided in relation to oversizing.

- A. Drainage facilities
- B. Lot, street and perimeter monumentation
- C. Mailbox delivery units
- D. Sanitary sewers
- E. Sidewalks
- F. Street lights
- G. Street name signs
- H. Street trees
- I. Streets
- J. Traffic signs
- K. Underground communication lines, including broadband (fiber), telephone, and cable. Franchise agreements will dictate whether telephone and cable lines are required.
- L. Underground power lines
- M. Water distribution lines and fire hydrants

RESPONSE: The applicant shall be responsible for the installation of all improvements detailed in Section 17.100.310, including fiber facilities. SandyNet requires the developer to work with the City to ensure that broadband infrastructure meets the design standards and adopted procedures as described in Section 17.84.70. Entry monument signs shall be located entirely outside the public rightof-way and clear vision areas as required by Section 17.74.30. If entry signs are desired, staff recommends the applicant submit a detailed plan showing the location of such signage.

CHAPTER 17.102 – URBAN FORESTRY

17.102.20 APPLICABILITY

This chapter applies only to properties within the Sandy Urban Growth Boundary that are greater than one acre including contiguous parcels under the same ownership.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- A. General: No person shall cut, harvest, or remove trees 11 inches DBH or greater without first obtaining a permit and demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance, the applicant shall agree to implement required provisions of this chapter and to allow all inspections to be conducted.
 - 2. Tree removal is subject to the provisions of Chapter 15.44, Erosion Control, Chapter 17.56, Hillside Development, and Chapter 17.60 Flood and Slope Hazard.
- B. Exceptions: The following tree removals are exempt from the requirements of this chapter.
 - 1. Tree removal as required by the city or public utility for the installation or maintenance or repair of roads, utilities, or other structures.
 - 2. Tree removal to prevent an imminent threat to public health or safety, or prevent imminent threat to public or private property, or prevent an imminent threat of serious environmental degradation. In these circumstances, a Type I tree removal permit shall be applied for within seven days following the date of tree removal.

RESPONSE: Section 17.00.10 states the overall purpose of the Development Code, which is to ensure that development is consistent with the goals and policies of the Sandy Comprehensive Plan. Goal 5 of the Comprehensive Plan addresses natural resources, with a stated intention of establishing policies for conservation of open space and protection of natural and scenic resources in compliance with Statewide Planning Goal 5. The Comprehensive Plan states, "In keeping with the mandate of Statewide Planning Goal 5, Oregon communities are required to provide programs that will 'conserve open space and protect natural and scenic resources.' Additionally, the City of Sandy has the goal of providing parks, open space, and recreational opportunities for its citizens, as well as promoting healthy and visually attractive environments that are in harmony with the natural landscape character of the area." Goal 5 of the Comprehensive Plan includes policies related to natural resources, open space, stream corridor protection, indigenous growth protection, historic and cultural resource protection, and environmental and cultural resource protection. The most relevant policies related to tree retention and mitigation include the following:

- Policy 4: The City promotes development practices which maintain or enhance the environmental quality enjoyed by the residents of the community.
- Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.
- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 18: Where feasible, preserve natural vegetation resource sites through public acquisition, conservation easements or other available methods to permanently limit development.
- Policy 19: Encourage and, if necessary, require reforestation in open space areas to help mitigate the adverse impacts of development.
- Policy 27: The City of Sandy shall promote the planting of trees throughout the community on both public and private properties.
- Policy 28: The City of Sandy shall consider the establishment of a street tree plan that identifies particular trees within different areas of the City but that provide species diversity

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

to add visual interest throughout the community, create individual neighborhood identity, and provide additional resistance through species diversity against disease.

- Policy 29: The City of Sandy shall review existing development standards, including street standards, to assure that the standards support and encourage the maintenance of a sustainable urban forest.
- Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.

As discussed in further detail below, the 9.68 acre site requires retention of a minimum of 29 trees that are in good condition and are 11-inches DBH or greater. Of the 521 trees inventoried on the site, the applicant is proposing to remove 457. Per Section 17.92.10(C), significant trees (8-inches or greater DBH) should be preserved to the greatest extent practicable. In addition, the City of Sandy Comprehensive Plan encourages development that minimizes disturbance of native tree stands, preserves natural open space, and promotes innovative site and building design to minimize the adverse impacts of development on native tree stands.

17.102.50 TREE RETENTION AND PROTECTION REQUIREMENTS

- A. Tree Retention: The landowner is responsible for retention and protection of trees required to be retained as specified below:
 - 1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous ownership.
 - 2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.
 - 3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.
 - 4. If possible, at least two of the required trees per acre must be of conifer species.
 - 5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements.
- **B.** Tree Protection Area: Except as otherwise determined by the Planning Director, all tree protection measures set forth in this section shall be instituted prior to any development activities and removed only after completion of all construction activity. Tree protection measures are required for land disturbing activities including but not limited to tree removal, clearing, grading, excavation, or demolition work.
 - 1. Trees identified for retention shall be marked with yellow flagging tape and protected by protective barrier fencing placed no less than 10 horizontal feet from the outside edge of the trunk.
 - 2. Required fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade.
 - 3. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles.
- C. **Inspection.** The applicant shall not proceed with any tree removal or construction activity, except erosion control measures, until the City has inspected and approved the installation of tree protection measures. Within 15 days of the date of accepting an application for a Type I permit, the city shall complete an onsite inspection of proposed activities and issue or deny the permit. Within 15 days of

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

issuing a Type II or Type III permit, the city shall complete an onsite inspection of proposed activities.

For ongoing forest operations, the permit holder shall notify the city by phone or in writing 24 hours prior to subsequent tree removal. The city may conduct an onsite re-inspection of permit conditions at this time.

RESPONSE: The subject property is 9.68 acres, which requires a minimum of 29 retention trees that are 11-inches or greater DBH and in good health. Per the submitted Tree Retention Inventory (Exhibit C, Sheet 2), the applicant proposes to retain all trees within the FSH overlay (64 total), including 33 trees that are viable and 11-inches or greater DBH. Of the 33 viable trees identified for retention, six (6) appear to be on or very close to the eastern property boundary (Trees # 527, 528, 530, 549, 550, and 3006). The applicant's representative submitted a memo (Exhibit W) at the February 25, 2019 Planning Commission hearing that confirmed Tree #3006 is on the property line and Trees # 527, 528, 530, 549, and 550 are all within 5 feet of the property line. However, the roots and canopies of these trees clearly cross the property boundary; thus, staff is concerned that these trees could be impacted by actions occurring on the neighboring property to the east, which is not in City limits. The applicant shall either identify an additional six (6) retention trees on the subject property, or shall obtain a recorded tree conservation easement on the neighboring property to the east for Trees # 527, 528, 530, 549, 550, and 3006. The tree conservation easement shall include the area under the dripline of the tree plus an additional 5 feet beyond the dripline. If the applicant chooses to retain additional trees, staff recommends retaining all trees on Lot 19 and including the area as part of the FSH tract to be owned and maintained by the HOA. Alternatively, staff recommends the applicant consider directional boring in the utility easements to retain Trees #109, 110, 120, 122, 336, 340, 366, 380, 383 and 392, and/or reducing the square footage of Lots 9 and 10 and creating an approximately 20 foot wide tree preservation buffer tract along Jacoby Road, which would also be placed in a conservation easement. A tree buffer along Jacoby Road would also support the applicant's request to have the houses on Lots 9 and 10 face the internal street network rather than Jacoby Road.

Of the 33 viable trees, an additional four (4) include notes such as "poor trunk taper," "windthrow potential," "wound on trunk," or "some trunk decay" (Trees # 520, 531, 534, and 560). At the February 25, 2019 Planning Commission hearing, the applicant's arborist submitted Arborist Notes for Jacoby Heights subdivision (Exhibit V). The notes state that "minor defects noted about 'viable' trees that do not necessarily affect survivability or lead to tree mortality are considered if proximity to construction will lead to additional stress that will affect the tree's ability to compartmentalize or outgrow the defect. The mention of these defects does not mean the trees are hazardous or declining. As trees being preserved are largely part of, and adjacent to, a mature stand of native trees they have a strong likelihood of survival and reduced risk of windthrow." Staff recommends obtaining a second opinion on the health of these four (4) trees and their likelihood of growing to maturity. **The applicant shall obtain a third-party arborist opinion to evaluate Trees # 520, 531, 534, and 560, and shall pay the third-party arborist fee.**

It appears that there are two trees that are proposed for removal located on the FSH boundary line: Tree #510 and Tree #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay. Both Tree #510 and Tree #516 are viable per the arborist report.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Section 17.102.50(B.1) requires tree protection fencing be placed no less than 10 horizontal feet from the outside edge of the trunk. Per the Pacific Northwest International Society of Arboriculture (ISA), the ISA defines the critical root zone (CRZ) as "an area equal to a 1-foot radius from the base of the tree's trunk for each 1 inch of the tree's diameter at 4.5 feet above grade (referred to as diameter at breast height)." Often the drip-line is used to estimate a tree's CRZ; however, it should be noted that a tree's roots typically extend well beyond its drip-line. In addition, trees continue to grow, and roots continue to extend. Thus, a proactive approach to tree protection would take into consideration the fact that the tree and its root zone will continue to grow. The arborist notes for Jacoby Heights Subdivision submitted by the applicant's arborist at the February 25, 2019 Planning Commission hearing (Exhibit V) state: "The TPF [tree protection fence] will be installed 5 feet beyond the drip-lines of trees being preserved that border the TPF. Trees being preserved within the core of the FSH will have ample protection." With regards to Trees # 510 and 516, the arborist's notes state: "T510 (Tree 510) and T516 (Tree 516) will be added to the preservation plan. These two trees are at the edge of the FSH. A grading cut will be made to the southwest of these two trees. Grading may encroach into the root protection zones (RPZs) of the two trees but will be no closer than 10 ft. from either tree. Grading will slope away from the trees which preserves more of the root crown than a vertical cut. The project arborist will be onsite to supervise grading that occurs within the RPZs of these trees and to prune roots if needed." As stated by the applicant's arborist, tree protection fencing shall be located at least 5 feet beyond the dripline of all retention trees, with the exception of Trees #510 and #516. For Trees #510 and #516, tree protection fencing shall be placed as close as possible to 5 feet beyond the dripline but may encroach in the critical root zone under supervision of the arborist, who shall be on site to supervise grading that occurs within the critical root zones of Trees # 510 and #516. In no case shall grading or other construction activity occur within 10 feet from the trunk of Tree #510 or Tree #516. Tree protection fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade. Staff recommends 6 foot tall no-jump horse fencing. Staff also recommends the applicant affix a laminated sign (minimum 8.5 inches by 11 inches) to the tree protection fencing indicating that the area behind the fence is a tree retention area and that the fence shall not be removed or relocated. No construction activity shall occur within the tree protection zone, including, but not limited to, dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. The applicant shall request an inspection of tree protection measures prior to any tree removal, grading, or other construction activity on the site.

OTHER CONSIDERATIONS FOR TREES:

To ensure protection of the required retention trees, the applicant shall record a tree protection covenant specifying protection of these trees and limiting removal without submittal of an Arborist's Report and City approval. This document shall include a sketch identifying the required retention trees. All trees marked for retention shall be retained and protected during construction regardless of desired or proposed building plans; plans for future houses on the proposed lots within the subdivision shall be modified to not encroach on retention trees and associated tree protection fencing.

The properties directly north and east of the subject site (tax lots 6100 and 800) contain a number of existing trees, some of which are located close to the shared property line and have canopies that extend onto the subject property. The applicant shall update the plan set to detail tree protection fencing on the subject property located 5 feet beyond the dripline of all trees located on the properties to the north and east.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

CHAPTER 17.66 – ADJUSTMENTS AND VARIANCES

17.66.60 VARIANCES

Variances are a means of requesting a complete waiver or major adjustment to certain development standards. They may be requested for a specific lot or as part of a land division application. The Type II variance process is generally reserved for major adjustments on individual lots, while variances to development standards proposed as part of a land division are processed as a Type III application (requiring a public hearing).

RESPONSE: The applicant proposes the following three Type III variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

17.66.70 TYPE II AND TYPE III VARIANCE CRITERIA

The authority to grant a variance does not include authority to approve a development that is designed, arranged or intended for a use not otherwise approvable in the location. The criteria are as follows:

- A. The circumstances necessitating the variance are not of the applicant's making.
- B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located.
- C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan.
- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land.
- F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control.

BLOCK LENGTH VARIANCE

17.100.120 (B) Residential Blocks. "Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance."

Request: The applicant requests approval to exceed the 400 foot block length standard for the north side of Woodstock Street between Camden Court and the eastern property boundary (approximately 426 feet).

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin. The City Engineer reviewed the future street plan (Exhibit C, Sheet 8) and states "given the wetlands/Tickle Creek is [sic] located to the north, a stubbed street is not feasible. Extending the block length for approximately 100 feet appears to be reasonable and is compatible with the location of the existing house on the adjacent property to the east."

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance. Granting a block length variance will better protect the natural resources on the site.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." **RESPONSE:** Approval of the variance will not have an impact on any of the policies or goals of the Comprehensive Plan or any other applicable policies and standards adopted by the City. The applicant's narrative (Exhibit B) states: "Granting this variance will further the City'[s] goal of protecting natural resources and will not adversely affect implementation of the Comprehensive Plan."

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." **RESPONSE:** The proposed development will not be significantly different than if Woodstock Street was limited to 400 feet in length. Inclusion of a connector trail to the future Tickle Creek Trail on the subject property north of Woodstock Street will break up the block length and provide additional pedestrian connectivity.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Recommendation: Staff recommends the Planning Commission approve a variance to allow the north side of Woodstock Street between Camden Court and the eastern property boundary to exceed 400 feet as proposed in the submitted plan set.

BUILDING ORIENTATION VARIANCES

The requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located.

Request: The applicant requests a Type III variance to Chapter 17.82.20 to allow the front door of the future homes constructed on Lots 9, 10, 30 and 32 to face the internal local street network instead of Jacoby Road, a designated transit street.

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The Development Code does not allow driveway access to higher classification streets such as Jacoby Road, a collector street. The front door could be oriented to Jacoby Road with a rear or side loaded garage oriented to the internal street network; however, staff recognizes that the front doors on Jacoby Road would essentially be false front doors, which is not the internet of the code.

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." **RESPONSE:** The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity. Orientation of the houses on Lots 9, 10, 30, and 32 to the internal street network will have no negative effect on surrounding properties. Per the applicant's narrative (Exhibit B), "Homes directly across Jacoby Road from the subject property have a similar configuration to what is being proposed."

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." **RESPONSE:** Orienting the houses on Lots 9, 10, 30, and 32 to the local street will allow construction of a traditional structure containing a garage and front door on the same street elevation, which will create a consistent look in the neighborhood. However, the requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located. If Planning Commission approves this variance, staff recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30 and 32 per the following:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Builders of individual lots shall construct a decorative fence on the Jacoby Road facing side of Lots 9, 10, 30, and 32 to enhance the visual appeal of these lots from Jacoby Road. The fence shall include the following design details:
 - Constructed of vertical black metal or faux metal fencing material.
 - No greater than 3-inch gap between vertical pickets.
 - o 4-feet to 6-feet in height.
- The rear elevations of Lot 30 and the side elevation of Lots 9, 10, and 32 shall incorporate all of the following design details where applicable:
 - Decorative gables including two or more of the following:
 - A window with grids.
 - A trimmed vent. The trim must match the trim on the windows and the vent must be at least 4 square feet in area.
 - Cross or diagonal bracing, shingles, trim, corbels, exposed rafter ends, or brackets.
 - Decorative 'belly-band' with an alternative paint color to the siding color, between building floors.
 - *Mixture of siding materials, including shake or horizontal lap siding with an alternative paint color to the primary siding color.*
 - *Recessed or covered rear entries (rear elevation only).*
 - The covered area must be at least 48 square feet and a minimum of 8 feet wide.
 - The recessed entry must feature vertical support posts.
 - o Minimum four-inch wide trim or 12-inch wide shutters around all windows.

Builders shall submit proposed decorative fence and elevation designs for staff review and approval.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." **RESPONSE:** It would create an unconventional design to have the rear of the houses on Lots 30 and 32 facing the proposed Camden Court cul-de-sac. While the front doors of the houses on Lots 9 and 10 could face Jacoby Road, the garages would not be able to gain access from Jacoby Road. A more traditional orientation would be to have both the front door and garage face the local street as requested by the applicant.

Recommendation: Staff recommends the Planning Commission approve a variance to allow the front door of the future homes constructed on Lots 9, 10, 30, and 32 to not face Jacoby Road. Staff further recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30, and 32.

TWO-TIERED LOT VARIANCE

17.100.120 (A) Blocks. "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Request: The applicant requests approval to provide a one-tier lot configuration for Lot 30 between Jacoby Road and Camden Court, instead of the required two-tier configuration.

 $W: \label{eq:linear} W: \lab$

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The applicant's narrative (Exhibit B) states that the design of the proposed subdivision, including the single-tiered lot on the cul-de-sac, is due to "topographic and natural resource constraints of the property associated with the location of Tickle Creek and steep slopes." The proposal only includes one single-tiered lot. In addition, the Development Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A), which states: "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." **RESPONSE:** The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." **RESPONSE:** The proposal contains one single-tiered lot (Lot 30) and the proposed development will not be significantly different than if the applicant proposed a two-tiered lot configuration so that Lot 30 was somehow not oriented as a single tiered lot.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The applicant's narrative states "the subject property contains unique circumstances (collector street, steep slopes, and creek) that warrant approval of an alternative design." The Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A) as explained in Criteria A.

Recommendation: Staff recommends the Planning Commission approve the applicant's request to allow one-tiered lot configuration for Lot 30 between Jacoby Road and Camden Court.

CHAPTER 15.30 - DARK SKY ORDINANCE

All exterior lighting is required to conform to the requirements of this chapter.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined insufficient. The locations of these fixtures shall be reviewed in detail with construction plans. Full cut-off lighting shall be required.

CHAPTER 15.44 – EROSION CONTROL

15.44.20 AREA OF APPLICATION

This chapter applies to all ground disturbing activities within the city limits whether or not a permit is required, unless such activities are otherwise exempted by the Sandy Development Code. All non-permitted ground disturbing activities that are permanent or temporary in nature shall comply with this chapter unless otherwise noted.

RESPONSE: All on-site earthwork activities including any retaining wall construction should follow the requirements of the current edition of the Oregon Structural Specialty Code (OSSC). If the proposal includes a retaining wall, the applicant shall submit additional details on the proposed retaining wall for staff review and approval. All on-site earthwork activities should follow the Geotechnical Report (Exhibit G) prepared by Rapid Soil Solutions LLC, dated February 23, 2018. Any deviations from the initial design parameters shall be reviewed by Rapid Soil Solutions for applicability. The City Engineer (Exhibit L) reviewed the Geotechnical Report and recommends that the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the above noted report and OSSC. Site grading should not in any way impede, impound or inundate the adjoining properties. All the work within the public right-of-way and within the paved area should comply with American Public Works Association (APWA) and City requirements as amended. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a revegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code. The proposed subdivision is greater than one acre which typically requires approval of a DEO 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

15.44.50 APPLICANT'S RESPONSIBILITY

The developer, subsequent developers or property owners shall bear responsibility for the installation, construction, protection of all stockpiles on the site, materials transported from the site, maintenance including re-vegetation of all graded areas, inspection and disposition of erosion control measures. In order to meet the purpose of this chapter, the developer shall:

C. Apply permanent or temporary soil stabilization to denuded development site areas in conformance with the following schedule:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- 1. Between October 1 and April 30, all denuded sites shall immediately be provided with either temporary or permanent soil stabilization.
- 2. Between May 1 and September 30, temporary erosion and sediment control measures to reduce dust and sediment transport shall be applied as soon as practicable, but in no case more than seven days after ground disturbing activity occurs.
- 3. Ground cover shall be installed on any portion of a site that is denuded for more than six months. Sports fields or playgrounds surrounded by vegetative cover or permanently installed curbing are exempt from this requirement.
- 4. Temporary measures shall be maintained until permanent measures are established.
- 5. Plant replacement vegetative cover that does not include plants listed as nuisance or prohibited plants on the City of Portland Plant List (available in the Planning & Development office) or the City's Native Plant List. Agriculture, timber production or residential crop growing activities are exempted from this requirement.
- 6. Secure or protect soil stockpiles throughout the project with temporary or permanent soil stabilization measures. Depositions of soil may be subject to additional regulations requiring permit, review or erosion and sediment control.
- 7. Select BMPs from the Erosion Control Manual.

RESPONSE: Section 15.44.50 contains requirements for maintenance of a site including revegetation of all graded areas. The applicant's Erosion Control Plan shall be designed in accordance with the standards of Section 15.44.50.

III. SUMMARY AND CONCLUSION

All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

As detailed in this report, the proposed subdivision generally complies with most Development Code requirements.

Biggest Issues/Staff Recommendations:

- Create a separate FSH tract, which will be owned and maintained by an HOA. Approve variance to allow Lots 19-21 to be less than 7,500 square feet and allow rear setback to be reduced to 15 feet.
- Add additional decorative fence and building features on Jacoby Road for Lots 9, 10, 30, and 32.
- Provide a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Design Newton Street with a three-quarter street section to the eastern property boundary. Provide the minimum three-quarter street improvements on Newton Street on the subject property or acquire land on adjacent properties to accommodate a three-quarter street.
- Identify an additional six (6) retention trees on the subject property, or obtain a recorded tree conservation easement on the neighboring property to the east for Trees # 527, 528, 530, 549, 550, and 3006. The tree conservation easement shall include the area under the dripline of the tree plus an additional 5 feet beyond the dripline.
- Obtain a second opinion from the third-party arborist to evaluate Trees # 520, 531, 534, and 560, and pay the third-party arborist fee.

IV. RECOMMENDATION

Staff recommends the Planning Commission hold a public hearing to take public testimony regarding the proposal. Staff recommends the Planning Commission **approve** the subdivision application with modifications as recommended in this report. The staff recommendation includes approval of the following three variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff is also recommending that the FSH analysis area be a separate tract owned and maintained by an HOA. This would result in Lots 19-21 being less than the minimum 7,500 square feet; thus, staff is recommending approval of a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum and allowing the rear setback to be reduced to 15 feet.

Since the Planning Commission decided to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation decision.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx
EXHIBIT A LAND	USE APPLICATION FORM (Please print or type the information b-low) Planning Department 39250 Pioncer Blvd. Sandy OR 97055 503-668-4886	
Name of Project		
Location or Address 19124 SE Jacoby Road, Sa	andy, OR	
Map & Tax Lot Number T 2S , R 4E , So	ection 24A ; Tax Lot(s) 2300	
Plan Designation SFR Zoning D	esignation LDR Acres 9.64	
Request: Type III, 31 lot residential subdivision including two Type III Variances and a Type II Tree Removal Permit. I am the (check one) 🖾 owner 🗆 lessee of the property listed above and the statements and information contained herein are in all respects true, complete and correct to the best of my		
knowledge and belief.		
Applicant Cory Knight	Owner Cory Knight	
Address 16513 SE Orchard View Lane	Address	

 Address
 16513 SE Orchard View Lane
 Address

 City/State/Zip
 Damascus, OR 97089
 City/State/Zip

 Phone
 503-481-7789
 Phone

 Email
 cory@taylorgrouprealty.com
 Email

 Signature
 Signature
 Signature

If signed by Agent, owner's written authorization must be attached.

RUAUINE		2
File No. 8-025 Date 6/918	Rec. No.	Fee \$ 967100
Type of Review (circle one): Type I	Type II Type III Type	e IV
)

G\Forms All Departments\Planning\Form Updates 2014\Applications\General Land Use Application .doe

Page 1 of 1

24E24402300



Revised December 2018

Project Details

Project Location:	19124 SE Jacoby Road, East side of Jacoby Road; south
	and east of Cascadia Village Drive

Legal Description: Map 24E 24A, Tax Lot 2300

Zoning District SFR, Single Family Residential

Site Size: 9.68 ac. (421,661 sf)

Owner / Applicant Cory Knight 16513 SE Orchard View Lane Damascus, Or 97089 cory@taylorgrouprealty.com 503-481-7789

Representative: <u>Civil Engineer / Surveyor</u> All County Surveyors & Planners, Inc. Ray Moore, P.E., P.L.S. P.O. Box 955 Sandy, OR 97055 Phone: 503-668-3151 Fax: 503-668-4730 Email: ray@allcountysurveyors.com

Jacoby Heights Subdivision Revised Narrative

Page 1 of 49

Consultant Team:

Planning Tracy Brown Tracy Brown Planning Consultants, LLC 17075 Fir Drive Sandy, OR 97055 Phone: 503-781-0453 tbrownplan@gmail.com

> <u>Geotechnical Engineer</u> Mia Mahedy-Sexton, PE, GE Rapid Soil Solutions 3915 SW Plum Street Portland, Or 97219 Mia@rapidsoilsolutions.com 503-816-3689

Traffic Engineer William Farley Lancaster Engineering 321 SW 4th Ave., Suite 400 Portland, OR 97204 Phone: 503-248-0313 todd@lancasterengineering.com

Wetland Consultant Jack Dalton Environmental Science & Assessment LLC 107 SE Washington Street Suite 249 Portland, Or 97214 Jack@esapdx.com Office: 503-478-0424

> <u>Arborist</u> Peter Torres Multnomah Tree Experts, Ltd 8325 SW 42nd Ave Portland, Or 97219 Peter@multnomahtree.com 503-452-8160

Cell: 971-413-6738

Jacoby Heights Subdivision Revised Narrative

Page 2 of 49

I. General Project Description

The project site consists of one parcel located at Township 2 South, Range 4 East, Section 24A, tax lot 2300 of the Willamette Meridian. The property contains approximately 9.64 acres and an existing single family dwelling. A tributary of Tickle Creek flows through the northeast corner of the property.

The site is zoned SFR, Single Family Residential. The applicant proposes constructing a 32 lot subdivision in a single phase and intends to remove the existing residence with construction of the subdivision. All of the lots are proposed to be accessed by a new street system constructed off Jacoby Road that can be extended to adjacent parcels in the future. The subject property gradually slopes to the west and north with the northeast corner of the property containing steeper slopes sloping to the northeast. This area of the property contains a small stretch of Tickle Creek.

The applicant proposes platting lots within the FSH Overlay identified on the property but no development is proposed within this area. A 20,283 square foot tract (Tract A) is also proposed to be dedicated to the city of Sandy to be used for stormwater detention and water quality purposes.

A pre-application conference was held with the City to review the project on January 3, 2018. Based on input received at this meeting modifications were made to the subdivision layout. Application materials were initially submitted to the City on June 25, 2018. On July 12, 2018, the City sent the applicant a letter deeming the application incomplete. The revised narrative addresses the items in this letter.

With this application the applicant is requesting a Code Interpretation per the requirements of Chapter 17.14. The applicant requests the Planning Commission make a determination regarding pertinent regulations applicable to retaining trees within the proposed subdivision. Staff has recently been applying the language in Section 17.92.10(C) in addition to Chapter 17.102 to regulating trees. The applicant believes only the requirements of Chapter 17.102 are applicable for this purpose.

II. Application Approval Requests

The applicant requests the following approvals with this application:

- Type III Code Interpretation regarding tree protection regulations applicable to the proposed subdivision;
- Type III Subdivision including a Type II Tree Removal Permit;
- Type III, Variance to Section 17.82.20 to allow all homes adjacent to Jacoby Road to be constructed with their front doors facing the internal local street rather than Jacoby Road;
- Type III Variance to Section 17.100.120(A) to allow creation of a single tier of lots (Lot 30) adjacent to Jacoby Road, a collector street; and

Jacoby Heights Subdivision Revised Narrative

Page 3 of 49

• Type III Variance to Section 17.100.120(B) to exceed the 400 foot block length maximum for the north side of Woodstock St., from Camden Ct. to the eastern property line.

III. Items Submitted With This Application

- Land Use Application
- Notification List and Mailing Labels
- Exhibit A Project Narrative (revised)
- Exhibit B Code Interpretation Request
- Exhibit C Civil Plans
 - Sheet 1 Cover Sheet and Preliminary Plat Map
 - Sheet 2 Existing Conditions and Tree Inventory
 - Sheet 3 Existing Trees with Building Setbacks
 - Sheet 4 FSH and Slopes Map
 - Sheet 5 Preliminary Street and Utility Plan
 - Sheet 6 Preliminary Street Tree Plan
 - Sheet 7 Preliminary Parking Plan
 - Sheet 8 Future Street and Pedestrian Trail Plan
 - Sheet 9 Grading and Erosion Control Plan
 - Sheet 10 Cut and Fill Evaluation
- Exhibit D Storm Drainage Report
- Exhibit E Traffic Impact Analysis
- Exhibit F Geotechnical Report
- Exhibit G Wetland Determination Memo
- Exhibit H Existing Easement (Document 2008-012774)
- Exhibit I Public Works Correspondence RE: Sewer Pump Station
- Exhibit J Tree Inventory
- Exhibit K Incompleteness Letter dated July 12, 2018

IV. Review of Applicable Approval Criteria

Development applications are required to meet development standards set forth in the City of Sandy Development Code. This section addresses all applicable review criteria. Pertinent code provisions are cited below in regular text followed by a response describing how the proposal complies with this standard in *italics*. The following code chapters have been reviewed in this narrative:

<u>Chapter</u>	Title
17.30	Zoning District
17.34	Single Family Residential (SFR)
17.60	Flood and Slope Hazard Overlay
17.80	Additional Setbacks on Collector and Arterial Streets
17.82	Special Setbacks on Transit Streets
17.84	Improvements Required with Development
17.86	Parkland and Open Space

Jacoby Heights Subdivision Revised Narrative

Page 4 of 49

- 17.90 Design Standards
- 17.92 Landscaping and Screening
- 17.98 Parking, Loading, and Access Requirements
- 17.100 Land Division
- 17.102 Urban Forestry
- 15.30 Dark Sky Ordinance

CHAPTER 17.30 - ZONING DISTRICTS

17.30.20 RESIDENTIAL DENSITY CALCULATION PROCEDURE

The number of dwelling units permitted on a parcel of land is calculated after the determination of the net site area and the acreage of any restricted development areas (as defined by Chapter 17.60). Limited density transfers are permitted from restricted development areas to unrestricted areas consistent with the provisions of the Flood and Slope Hazard Area Overlay District, Chapter 17.60.

Response: The applicant proposes a 32 lot subdivision with a single tract to be dedicated to the city (Tract A) for stormwater drainage purposes. The subject property contains a gross site area of 9.68 acres. After deducting public rights-of-way and the proposed stormwater tract, the net site area (NSA) is 7.09 acres. Because the subject property contains restricted development areas (RDA) as defined by Chapter 17.60 these areas are also deducted from the net site area to determine the unrestricted site area (USA). The formula used in this calculation is: NSA - RDA = USA.

The subject property contains 52,444 square feet (1.20 acres) of restricted development area (RDA) as shown on Sheet 4. Subtracting this area from the 7.09 net site area (NSA) results in an unrestricted site area (USA) containing 5.89 acres.

The SFR zone allows a minimum of 3 and a maximum of 5.8 units per net acre. The minimum density is calculated by multiplying the USA x the required minimum density (5.89 acres x = 17.67 units round up to 18 units)

The maximum density is determined by these two formulas and using the lesser number of units.

a. NSA (in acres) x Maximum Density of Zoning District (units/acre).

(7.09 acres x 5.8 units/acre = 41.122 (rounded to 41 units)).

b. USA (in acres) x Maximum Density of Zoning District (units/acre) x 1.5 (maximum allowable density transfer based on Chapter 17.60)

(5.89 x acres x 5.8 units/acre x 1.5 density transfer = 51.243 (rounded to 51 units)

As a result of these calculations the density range for the subject property is a minimum of 18 units and a maximum of 41 units.

Jacoby Heights Subdivision Revised Narrative

Page 5 of 49

CHAPTER 17.34 - SINGLE-FAMILY RESIDENTIAL (SFR)

17.34.00 - INTENT

The district is intended to implement the Low Density Residential Comprehensive Plan designation by providing for low-density residential development in specific areas of the city. The purpose of this district is to allow limited development of property while not precluding more dense future development, as urban services become available. Density shall not be less than 3 or more than 5.8 units per net acre.

Response: As detailed above, the proposed 32 lots complies with the density range in the SFR zoning district of 18 to 41 units for the subject property.

17.34.10 - PERMITTED USES

A. Primary Uses Permitted Outright:

1. Single detached dwelling subject to design standards in Chapter 17.90; *Response:* The applicant proposes constructing single family detached dwellings as permitted in this zoning district.

	11.24.20 -	DEVELOPMENT	STANDARDS	
1				T
	T			1

47.24.20 DEVELOBMENT STANDADDS

Туре	Standard
A. Minimum Lot Area - Single detached dwelling - Other permitted uses	7,500 square ft. No minimum
B. Minimum Average Lot Width - Single detached dwelling	60 ft.
C. Minimum Lot Frontage	20 ft. except as allowed by Section 17.100.160
D. Minimum Average Lot Depth	No minimum
E. Setbacks (Main Building) Front yard Rear yard Side yard (interior) Corner Lot	10 ft. minimum 20 ft. minimum 7.5 ft. minimum 10 ft. minimum on side abutting the street
F. Setbacks (Garage/Carport)	22 ft. minimum for front vehicle access 15 ft. minimum if entrance is perpendicular to street (subject to Section 17.90.220) 5 ft. minimum for alley or rear access
G. Projections into Required Setbacks	See Chapter 17.74
H. Accessory Structures in Required Setbacks	See Chapter 17.74
I. Structure Height	35 ft. maximum
J. Building Site Coverage	No minimum
K. Off-Street Parking	See Chapter 17.98

Jacoby Heights Subdivision Revised Narrative

Page 6 of 49

Response: As shown on Sheet 1 of the plan set, all lots in the proposed subdivision contain at least 7,500 square feet, have at least 20 feet of street frontage, contain an average lot width of at least 60 feet as required. The proposed building footprints shown on this sheet indicates that all lots are capable of complying with applicable setbacks in the zone. The details of these development standards will be reviewed with submittal of building permits. Compliance with required off-street parking as shown on Sheet 7 is reviewed in Chapter 17.98 below.

17.34.40 - MINIMUM REQUIREMENTS

- A. Must connect to municipal water. *Response:* The applicant proposes extending water service to serve all dwellings in the new subdivision.
- B. Must connect to municipal sewer if service is currently within 200 feet of the site. Sites more than 200 feet from municipal sewer, may be approved to connect to an alternative disposal system provided all of the following are satisfied:
 - 1. A county septic permit is secured and a copy is provided to the city;
 - The property owner executes a waiver of remonstrance to a local improvement district and/or signs a deed restriction agreeing to complete improvements, including but not limited, to curbs, sidewalks, sanitary sewer, water, storm sewer or other improvements which directly benefit the property;
 - 3. The minimum size of the property is one acre or is a pre-existing buildable lot, as determined by the city;
 - 4. Site consists of a buildable parcel(s) created through dividing property in the city, which is less than five acres in size. *Response: The existing dwelling is currently served by a septic system. This system will be decommissioned in accordance with applicable regulations and the applicant will provide proof of the decommissioned system with construction documents.*
- C. The location of any real improvements to the property must provide for a future street network to be developed. *Response:* A new street network will be constructed to serve each dwelling as required.
- D. Must have frontage or approved access to public streets. **Response:** Each new residence constructed in the subdivision will gain access from a public street.

17.34.50 - ADDITIONAL REQUIREMENTS

A. Design review as specified in Chapter 17.90 is required for all uses.

Jacoby Heights Subdivision Revised Narrative

Page 7 of 49

Response: Only Section 17.90.150, Residential Design Standards of Chapter 17.90 is applicable to residential developments. This section is reviewed below.

B. Lots with 40 feet or less of street frontage shall be accessed by a rear alley or a shared private driveway.

Response: Lots 27/28 contain less than 40 feet of street frontage but comply with the 20 foot minimum street frontage standard. Due to slopes it is not feasible to access these lots by a rear alley, however a shared curb cut/ driveway is a possibility if required.

CHAPTER 17.60 - FLOOD AND SLOPE HAZARD (FSH) OVERLAY

17.60.10 - INTERPRETATION AND MAPPING

The Director has the ultimate responsibility for maintaining the FSH Overlay District on the City of Sandy Zoning Map, determining on-site measuring methods, and otherwise interpreting the provisions of this chapter. Technical terms used in this chapter are defined in Chapter 17.10, Definitions. This chapter does not regulate development on lots or parcels entirely outside the FSH Overlay District.

A. <u>FSH Overlay District</u>. The only areas subject to the restrictions and prohibitions of the FSH overlay district are those indicated on the City of Sandy Zoning Map on file in the Planning Department. This chapter does not regulate lots or parcels entirely outside the FSH Overlay District. *Response: The city's Zoning Map dated May 18, 2018 shows that only the northeast corner of the subject property is affected by the FSH Overlay and the city's FSH Overlay Analysis map dated December 11, 2012 does not identify any restricted areas on the subject property. The applicant hired a wetland*

expert to evaluate the property to determine if any wetlands or waters of the state exist on the subject property. As detailed in Exhibit G submitted with this application, this memo indicates "there is no evidence of intermittent or perennial flow on the site" and no wetlands are present.

As part of the completeness items, the City requested the applicant map the FSH Overlay on the site (including 25% slopes and setbacks, 200 feet from the top of bank of Tickle Creek, 100 ft. from the top of bank of other streams, wetlands with setback, etc.) and pay a Type III FSH review fee. The applicant is confused by this request since this code section clearly states the "only areas subject to the restrictions and prohibitions of the FSH overlay district are this indicated on the City of Sandy Zoning Map on file in the Planning Department".

Regardless, as requested the applicant's Engineer completed the FSH overlay mapping as shown on Sheet 4.

Jacoby Heights Subdivision Revised Narrative

Page 8 of 49

B. <u>Development Approval Required</u>. No development shall occur within the FSH overlay district without first obtaining City approval under the provisions of this chapter. The Director shall notify the Oregon Division of State Lands whenever any inventoried wetland is proposed for development, in accordance with ORS 227.350. In riverine situations, the Director shall notify adjacent communities and the State Coordinating Office prior to any alteration or relocation of a watercourse, and submit copies of such notification to the administrator.

Response: As shown on the Preliminary Plat Map, lots are proposed to be platted within the FSH overlay district, but no development will occur in this area.

- C. <u>Applicant Responsibilities</u>. The applicant for alteration or development within the FSH overlay district shall be responsible for preparing a survey of the entire site, based on site specific field surveys or Corps of Engineers data that precisely maps and delineates the following areas:
 - The name, location and dimensions of affected streams or rivers, and the tops of their respective banks. *Response: As noted in the section above, no development is proposed* within the FSH overlay district on the subject property.
 - 2. 100-year floodplain and floodway boundaries and elevations as determined by the June 17, 2008 FIS for Clackamas County and Incorporated Areas. *Response: The 100-year floodplain or floodway boundaries have not been identified for this potion of Tickle Creek and does not affect the subject property.*
 - 3. The City of Sandy FSH overlay district boundary as depicted on the City of Sandy FSH Map.
 - 4. The water quality and slope setback area(s) as defined in Section 17.60.30.
 - 5. The size and location of locally significant wetlands shall be determined based on the City of Sandy Locally Significant Wetland Inventory (2002) unless modified by a wetland delineation approved by the Oregon Division of State Lands and submitted to the City. Wetland delineations that have formal concurrence from the Division of State Lands shall be valid for the period specified in that agency's administrative rules.
 - 6. Steep slope areas where the slope of the land is 25% or greater within the FSH overlay district boundary.
 - 7. The area enclosed by a continuous line, measured 25 feet horizontally, parallel to and upland from the top of a steep slope area, where the top of the steep slope is within the FSH overlay district boundary.
 - 8. Existing public rights-of-way, structures, roads and utilities.
 - 9. Natural vegetation, including trees or tree clusters and understory within the FSH Overlay District boundary.
 - 10. Existing and proposed contours at 2-foot intervals.

Jacoby Heights Subdivision Revised Narrative

Page 9 of 49

Response: The applicant has shown the FSH Overlay on Sheet 4. This boundary is included on all plan sheets.

17.60.20 - PERMITTED USES AND ACTIVITIES

- A. <u>Restricted Development Areas</u>. Restricted development areas within the FSH overlay district as shown on the City of Sandy Zoning Map include:
 - Slopes of 25% or greater that (a) encompass at least 1,000 square feet and (b) have an elevation differential of at least 10 feet.
 - 2. Protected water features, including locally significant wetlands, wetland mitigation areas approved by the Division of State Lands, and perennial streams.
 - 3. Required setback areas as defined in section 17.60.30. *Response:* No development is proposed within the FSH overlay as identified on either the city's official Zoning Map or the applicant's mapped boundary.
- B. <u>Permitted Uses</u>. Permitted uses within restricted development areas are limited to the following: *Response:* None of the uses identified in this section are proposed within the FSH overlay on the subject property.
- C. <u>Platting of New Lots</u>. No new lot shall be platted or approved for development that is exclusively in restricted development areas as defined in subsection 17.60.20.A.

Response: As shown on the Preliminary Plat several of the lots are proposed to be platted within the FSH overlay. All of these lots have sufficient area to ensure development can occur outside the FSH overlay.

17.60.30 - REQUIRED SETBACK AREAS

- A. <u>Required Setbacks</u>. The required special setback(s) shall be:
 - 1. 70 feet from the top of bank of Tickle Creek;
 - 3. 25 feet around the edge of any mapped locally significant wetland; and
 - 25 feet from the top of any 25% slope break where the slope break occurs within the FSH overlay district as mapped by the city. *Response: These boundaries are identified as applicable on the applicant's FSH mapping included on Sheet 4.*
- B. <u>Minimize Impacts</u>. Natural vegetation shall be preserved and enhanced and excavation minimized within required water quality setback areas. *Response: No disturbance or development is proposed within water quality setback areas on the subject property.*

17.60.40 - REVIEW PROCEDURES

Review of development requests within the FSH Overlay District shall occur subject to the following procedures. Unless otherwise indicated below, the

Jacoby Heights Subdivision Revised Narrative

Page 10 of 49

Director may approve Type I permits over the counter or following a field check. Type II and III development applications shall be reviewed to ensure consistency with Section 17.60.60-70. Section 17.60.50 special reports shall also be required, unless specifically exempted by the Director.

Response: As noted above, no development is proposed within the FSH Overlay District.

17.60.80 - WATER QUALITY TREATMENT FACILITIES

Tickle Creek, the Sandy River and associated natural drainage ways are vital to Sandy's recreationally based economy and to the quality of life of Sandy residents. Placement of water quality facilities shall be limited as follows:

- A. The water quality facility shall not be constructed in restricted development areas, except where necessary to serve approved development within restricted development areas (e.g., a road) and where no reasonable alternative exists in buildable areas of the site.
- B. Where the approval authority determines that a more efficient and effective regional site exists within the sub-basin, the water quality facility may be constructed off-site.

Response: The proposed water quality (Tract A) is located outside the FSH overlay.

CHAPTER 17.66 - ADJUSTMENTS AND VARIANCES

Adjustments and variances are procedures to vary development standards normally applied to a particular district.

Response: The applicant is requesting three variances with this application as follows:

- Variance No. 1 Type III Special Variance to Section 17.82.20 to allow homes on lots adjacent to Jacoby Road to face the internal local street rather than Jacoby Road;
- Variance No. 2 Type III Variance to Section 17.100.120(A) regarding two tiers of lots.
- Variance No. 3 Type III Variance to Section 17.100.120(B) regarding block length standard.

<u>VARIANCE NO. 1</u> - The applicant is requesting a Type III variance to Section 17.82.20 to allow homes constructed on lots adjacent to Jacoby Road to have their front doors face the internal local street rather than Jacoby Road. The City's Transit Master Plan identifies the portion of Jacoby Road from Dubarko Road to Cascadia Village Drive as a future transit route. Section 17.82.10 indicates that the requirements of Chapter 17.82 are only applicable to properties within 400 feet of an existing or future transit route. Only Lot 30 is considered a double frontage lot with access proposed from Camden Court.

Jacoby Heights Subdivision Revised Narrative

Page 11 of 49

A. The circumstances necessitating the variance are not of the applicant's making.

Response: The applicant is requesting a variance to orient the homes adjacent to Jacoby Road towards the internal local street. The location and roadway classification of Jacoby Road is not of the applicant's making and these lots are restricted from gaining access Jacoby Road by Section 17.100.220(E). In addition, a grade differential exists between the elevation of the sidewalk along Jacoby Road and the future homes located on some of these lots making construction of a pedestrian connection more challenging. The applicant also wants to ensure that these lots are provided with a large private backyard which is desirable selling feature for a large lot, and is concerned construction of a pedestrian connection through this yard will limit the desirability of this feature. In addition, Chapter 17.84 requires Camden Court to be spaced at least 150 feet from the intersection of Woodstock Street with Jacoby Road. As a result, Lot 30 is considerably deeper than it would otherwise be and the backyard on this lot will be roughly 40 deep. This added depth will separate this home from Jacoby Road and ensure a spacious backyard. To require construction of a pedestrian connection through this yard would defeat the benefit of this configuration. All other lots (Lots 9, 10, and 32) that may be subject to these section are corner lots and are best suited to be oriented to the internal local street as the lots next to them will be.

B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located. *Response:* Approval of this variance would allow the property to be developed with large lots featuring a relatively large private backyard. This configuration is similar to all other lots in the subdivision and the majority of similarly sized lots in the City of Sandy. Approval of a variance to this standard will allow homes to be constructed on these lots similar to homes constructed directly across Jacoby Road from the subject property. Approval of this variance will not allow otherwise prohibited uses.

 C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan.
 Response: The Sandy Comprehensive Plan contains limited language specifically related to the proposed code requirement. Goal 12, Transportation however does include the following policies:

Pedestrian Friendly Street and Streetscape Design

- 8. Encourage the planting of street trees in tree-deficient area of the city.
- 10. Encourage the development of sidewalks on both sides of all streets, especially in high pedestrian activity areas such as near schools and in the downtown area.
- 11.Develop street, bicycle, and pedestrian facilities that encourage pedestrian- friendly streetscapes.

Jacoby Heights Subdivision Revised Narrative

Page 12 of 49

If the requested variance is approved, all homes constructed in the subdivision will be provided with a street and sidewalk in front of the home that will connect to the sidewalk along Jacoby Road. This streetscape will include the planting of street trees as required by code. It would hard to imagine how approval of this variance would adversely affect implementation of policies of the Comprehensive Plan.

- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity. *Response:* Approval of this variance to allow homes on lots adjacent to Jacoby Road to face the internal street rather than Jacoby Road will have no effect on the public welfare or other property in the vicinity. Homes directly across Jacoby Road from the subject property have a similar configuration to what is being proposed.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land.

Response: Approval of this variance will allow homes to be constructed in a manner that is similar to the majority of homes in the city. Some examples of homes constructed under similar circumstances include:

- Cascadia Village between Wallingford Ave. and Jacoby Road (across Jacoby from the proposed subdivision;
- Timberline Trails Subdivision between Dubarko and Wall St;
- Sandy Bluff Subdivision south side of Coralburst St;
- Sandy Bluff Annex west side of Penny Ave;
- Hamilton Ridge southern section of Hamilton Ridge Drive;
- Sleepy Hollow Subdivision between Van Tassel and Sandy Heights;
- Deer Pointe west side of Antler Ave.; and
- Deer Pointe between Highway 26 and Meadow Ave.
- F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control. *Response:* The applicant is requesting approval of this variance because of topographic considerations and the physical characteristics of the subject lots. Lot 30 proposed to be accessed from and face Camden Court is approximately 130 feet. This added depth will allow a large home to be constructed on this lots with a substantial backyard. These characteristics are somewhat unique and very desirable in the City of Sandy. The other lots that may be subject to this standard are corner lots.

<u>VARIANCE NO. 2</u> - The applicant is requesting a variance to Section 17.100.120(A) for the block containing Lot 30. Section 17.100.120(A) <u>Blocks</u>, contains the following language: "Blocks shall have sufficient width to provide for two tiers of

Jacoby Heights Subdivision Revised Narrative

Page 13 of 49

lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Response: Based on input received at the pre-application conference, City staff indicated the applicant will need to apply for a variance to Section 17.100.120(A) for Lot 30. As shown on submitted plans the separation between Jacoby Road and Camden Court contains only a single tier of lots. This design is due to topographic and natural resource constraints of the property associated with the location of Tickle Creek and steep slopes. The language in this section specifies that, "exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features". The language in this section suggests that granting an exception without requiring a variance is the preferred method to resolve conflicts with this section. There are numerous examples in the city of subdivisions that have been approved without requiring a variance to this standard. These include the following:

- Cascadia Village between Wallingford Ave. and Jacoby Road (across Jacoby from the proposed subdivision;
- Timberline Trails Subdivision between Dubarko and Wall St;
- Sandy Bluff Subdivision south side of Coralburst St;
- Sandy Bluff Annex west side of Penny Ave;
- Hamilton Ridge southern section of Hamilton Ridge Drive;
- Salmon Creek Estates north side of Chinook St;
- Sleepy Hollow Subdivision between Van Tassel and Sandy Heights;
- Deer Pointe west side of Antler Ave.; and
- Deer Pointe between Highway 26 and Meadow Ave.

The variance criteria regarding the requested variance are addressed below as requested.

A. The circumstances necessitating the variance are not of the applicant's making.

Response: Tickle Creek flows through the northeast corner of the subject property and moderately steep slopes descend from the upper parts of the property down towards the creek. The location of these features limit the creation of a typical two tier lot development pattern. In addition, Jacoby Road is classified as a collector street and direct access to lots from a local street (proposed cul-de-sac) is preferred. A cul-de-sac is proposed to provide access to the five lots in this area. Due to unique site constraints, the circumstances necessitating the variance are not of the applicant's making.

B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located.

Jacoby Heights Subdivision Revised Narrative

Page 14 of 49

Response: The requested variance is necessary to allow an efficient development pattern on the subject property. The hardship created by requiring compliance with the two tier standard in this section does not arise from a violation of the Code and approval of this variance will not allow otherwise prohibited uses in the SFR zoning district in which the property is located. Without approval of this variance development of this portion of the property would be severely limited.

 C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan.
 Response: As noted above, a similar design has been approved and constructed in a number of other places through-out the community. The proposed request is not unique and to allow a single tier of lots in this portion of the development will not set a precedence or adversely affect implementation of the Comprehensive Plan. A review of the Comprehensive

Plan does not reveal any policies specifically addressing this standard.

- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity. *Response:* The requested variance will provide the developer with increased flexibility in the development of the lots affected by this request. The location and conditions of the variance will have no effect on adjacent or surrounding properties. Approval of the variance will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land.

Response: The proposed design is due to the unique topographic and natural resource constraints of the property associated with the location of Tickle Creek and steep slopes. The language in this section specifies that certain conditions warrant an exception to this standard: "blocks adjacent to arterial streets or natural features". The language in this section suggests that granting an exception without requiring a variance is the preferred method to resolve conflicts with this section. As noted above, numerous examples exist in the city of subdivisions that have been approved without requiring a variance to this standard. These include the following:

- Cascadia Village between Wallingford Ave. and Jacoby Road (across Jacoby from the proposed subdivision;
- Timberline Trails Subdivision between Dubarko and Wall St;
- Sandy Bluff Subdivision south side of Coralburst St;
- Sandy Bluff Annex west side of Penny Ave;
- Hamilton Ridge southern section of Hamilton Ridge Drive;
- Salmon Creek Estates north side of Chinook St;

Jacoby Heights Subdivision Revised Narrative

Page 15 of 49

- Sleepy Hollow Subdivision between Van Tassel and Sandy Heights;
- Deer Pointe west side of Antler Ave.; and
- Deer Pointe between Highway 26 and Meadow Ave.

Because of physical limitations associated with the classification of Jacoby Road, steep slopes and the location of Tickle Creek, development of this portion of the property is challenging. Approval of an exception to this standard or variance will allow the property to be developed to the greatest extent that is reasonably possible.

F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control. *Response:* As discussed above, the subject property contains unique circumstances (collector street, steep slopes, and creek) that warrant approval of an alternative design. The applicant does not have any control over these circumstances. Approval of this variance will result in a similar lot configuration as a number of previously approved subdivisions located in the city.

<u>VARIANCE NO. 3</u> - The applicant requests a variance to Section 17.100.120(B) block length standard for the north side of Woodstock Street from Camden Court (cul-de-sac) to the end of this street on the subject property.

This standard states: <u>Residential Blocks</u>. Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance. *Response:* The submitted plan features a block on the north side of Woodstock St. from Camden Court to the eastern property line of 420 feet, approximately 20 feet longer than allowed by this section. This design is due the unique topographic conditions of the property and the location of Tickle Creek prohibiting another street to be constructed north of Woodstock Street. As shown on Sheet 8, Future Street Plan, the extension of Woodstock Street to the east of the subject property provides opportunities to provide public access to the creek corridor if it desired in the future. This section clearly identifies an adjustment or variance

A. The circumstances necessitating the variance are not of the applicant's making.

Response: The location of steep slopes and Tickle Creek limit construction of a north-south street north of Woodstock Street. The location of these natural features are not of the applicant's making.

Jacoby Heights Subdivision Revised Narrative

Page 16 of 49

B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located.

Response: The proposed block exceeds the 400 foot maximum block length standard by approximately 20 feet. Because of the physical constraints discussed above, the construction of a north-south street in this location is also not desirable as this street is extended to the east. Granting a variance to allow construction of the street as proposed would not allow otherwise prohibited uses in the SFR zoning district.

- C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan. *Response:* Due to the unique physical constraints and the location of Tickle Creek, it is not practicable to construct a street north of Woodstock Street. No street is shown in this location on the city's Transportation System Plan or any other long range planning document. Granting this variance will further the City' goal of protecting natural resources and will not adversely affect implementation of the Comprehensive Plan.
- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity. *Response:* The extension of Woodstock Street to the east will provide an opportunity for the adjacent property to gain access to this street and extend utilities for future development. Property to the north of Tickle Creek in the Timberline Trails Subdivision has already been provided access. Construction of a new street north of Woodstock Street would be significantly more injurious to properties north of the subject property than not constructing this street. Granting this variance will not adversely affect the public welfare or be materially injurious to other property in the vicinity.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land. *Response:* Approving this requested variance will allow the property to be developed with a subdivision of large lots for future residential home construction. The applicant proposes establishing a pedestrian access to the Tickle Creek corridor from the north end of Camden Court. Approval of the variance will be similar to development permitted in compliance with this standard.
- F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control. *Response:* As noted above, the location of Tickle Creek and topographic constraints make construction of a street north of Woodstock Street, east of

Jacoby Heights Subdivision Revised Narrative

Page 17 of 49

Camden Court impracticable and undesirable. These conditions are generally unique to the subject property and result from physical limitations of the property.

CHAPTER 17.80 - ADDITIONAL SETBACKS ON COLLECTOR AND ARTERIAL STREETS

17.80.10 - APPLICABLITY

These regulations apply to all property abutting the following streets:

- A. Minor Arterials.
 - SE 362nd Avenue (Duncan Road)
 - Bluff Road
 - Kelso Road
 - Ten Eyck Road
 - Langensand Road
 - Bornstedt Road
 - Bell Street
- B. CollectorStreets.
 - Industrial Way
 - Sandy Heights (Wewer Road) Street
 - Tupper Road
 - Meinig Road (south of Proctor)
 - Meinig Road (First Avenue)
 - McCormick
 - Van Fleet Street
 - Gary Street
 - Pleasant Street
 - Sunset Street
- C. Residential Minor Arterial
 - Dubarko Road

Response: The subject property is not located adjacent to any of the streets listed in this section.

17.80.20 - SPECIFIC SETBACKS

Any structure located on streets listed above or identified in the Transportation System Plan as arterials or collectors shall have a minimum setback of 20 feet measured from the property line. This applies to applicable front, rear and side yards.

Response: The City's Transportation System Plan identifies Jacoby Road adjacent to the subject property as a "Collector" street. The Preliminary Plat shows the proposed building envelopes for all lots adjacent to this road will maintain a minimum 20 foot setback to this road.

CHAPTER 17.82 - SPECIAL SETBACKS ON TRANSIT STREETS 17.82.00 INTENT

Jacoby Heights Subdivision Revised Narrative

Page 18 of 49

The intent is to provide for convenient, direct, and accessible pedestrian access to and from public sidewalks and transit facilities, provide a safe, pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk and/or transit street; and promote the use of pedestrian, bicycle, and transit modes of transportation to retail and commercial activities. **Response:** The applicant has submitted a variance request to the requirements of this section as detailed in Chapter 17.66 above to allow homes along Jacoby Road to face the adjacent internal local street.

CHAPTER 17.84 - IMPROVEMENTS REQUIRED WITH DEVELOPMENT 17.84.20 - TIMING OF IMPROVEMENTS

- A. All improvements required by the standards in this chapter shall be installed concurrently with development, as follows:
 - 1. Where a land division is proposed, each proposed lot shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to approval of the final plat.

Response: All lots in the proposed subdivision will be required to install public and franchise utility improvements or financially guaranteed these improvements prior to final plat approval.

2. Where a land division is not proposed, the site shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to temporary or final occupancy of structures.

Response: This section is not applicable because a land division is proposed.

B. Where specific approval for a phasing plan has been granted for a planned development and/or subdivision, improvements may similarly be phased in accordance with that plan.

Response: The applicant does not intend constructing the proposed subdivision in phases.

17.84.30 - PEDESTRIAN AND BICYCLIST REQUIREMENTS

A. Sidewalks shall be required along both sides of all arterial, collector, and local streets, as follows:

1. Sidewalks shall be a minimum of 5 ft. wide on local streets. The sidewalks shall be separated from curbs by a tree planting area that provides separation between sidewalk and curb, unless modified in accordance with Subsection 3 below.

Response: All proposed sidewalks on the internal (local) streets will be five feet wide as required and separated from curbs by a tree planting area.

Jacoby Heights Subdivision Revised Narrative

Page 19 of 49

2. Sidewalks along arterial and collector streets shall be separated from curbs with a planting area, except as necessary to continue an existing curb-tight sidewalk. The planting area shall be landscaped with trees and plant materials approved by the City. The sidewalks shall be a minimum of 6 ft. wide.

Response: A six foot sidewalk is proposed to be constructed along Jacoby Road and will include a planter strip as required.

- 3. Sidewalk improvements shall be made according to city standards, unless the city determines that the public benefit in the particular case does not warrant imposing a severe adverse impact to a natural or other significant feature such as requiring removal of a mature tree, requiring undue grading, or requiring modification to an existing building. Any exceptions to the standards shall generally be in the following order.
 - a) Narrow landscape strips
 - b) Narrow sidewalk or portion of sidewalk to no less than 4 feet in width
 - c) Eliminate landscape strips
 - d) Narrow on-street improvements by eliminating on-street parking
 - e) Eliminate sidewalks

Response: No exceptions or modifications to the sidewalk standards of this section are requested with this application.

- 4. The timing of the installation of sidewalks shall be as follows:
 - a) Sidewalks and planted areas along arterial and collector streets shall be installed with street improvements, or with development of the site if street improvements are deferred.
 - b) Sidewalks along local streets shall be installed in conjunction with development of the site, generally with building permits, except as noted in (c) below.
 - c) Where sidewalks on local streets abut common areas, drainageways, or other publicly owned or semi-publicly owned areas, the sidewalks and planted areas shall be installed with street improvements. *Response:* The applicant intends to construct all sidewalk improvements as required by this section. The sidewalk along Jacoby Road will be constructed prior to final plat approval, or at the time of home construction whichever the city prefers. Sidewalks along the local streets will be constructed at the time of home construction.
- B. Safe and convenient pedestrian and bicyclist facilities that strive to minimize travel distance to the extent practicable shall be provided in conjunction with new development within and between new subdivisions, planned developments, commercial developments, industrial areas, residential areas, public transit stops, school transit stops, and neighborhood activity centers such as schools and parks, as follows:

Jacoby Heights Subdivision Revised Narrative

Page 20 of 49

1. For the purposes of this section, "safe and convenient" means pedestrian and bicyclist facilities that: are reasonably free from hazards which would interfere with or discourage travel for short trips; provide a direct route of travel between destinations; and meet the travel needs of pedestrians and bicyclists considering destination and length of trip.

Response: No pedestrian or bicycle facilities other than sidewalks are proposed. The applicant had originally identified a pedestrian access from the end of Camden Court to connect with a future trail along Tickle Creek. After reviewing the location and details of facility further, the applicant's Engineer has identified the preferred location for this facility to be on the adjacent property to the east of the subject property. The reasons this facility should not be located on the subject property is due to steep slopes. As shown on Sheet 8, a future Tickle Creek Trail is likely to be located on the north side of the creek. The location of a connector/access trail will need to accommodate construction of bridge to cross the creek from the vicinity of the subject property in order to reach the trail. The applicant does not favor constructing this facility on the subject property due to steep grades and resulting construction challenges (stairs with handrail) associated with constructing the facility on steeps slopes.

- To meet the intent of "B" above, right-of-ways connecting cul-de-sacs or passing through unusually long or oddly shaped blocks shall be a minimum of 15 ft. wide with 8 feet of pavement. *Response: As noted above, no facilities are proposed.*
- 3. 12 feet wide pathways shall be provided in areas with high bicycle volumes or multiple use by bicyclists, pedestrians, and joggers. *Response:* The details of the Tickle Creek detail have not been determined in the vicinity of the subject property. No facility is proposed with the subdivision.
- 4. Pathways and sidewalks shall be encouraged in new developments by clustering buildings or constructing convenient pedestrian ways. Pedestrian walkways shall be provided in accordance with the following standards:
 - a) The pedestrian circulation system shall be at least five feet in width and shall connect the sidewalk on each abutting street to the main entrance of the primary structure on the site to minimize out of direction pedestrian travel.
 - b) Walkways at least five feet in width shall be provided to connect the pedestrian circulation system with existing or planned pedestrian facilities which abut the site but are not adjacent to the streets abutting the site.
 - c) Walkways shall be as direct as possible and avoid unnecessary meandering.
 Response: No pedestrian pathways are proposed, only sidewall

Response: No pedestrian pathways are proposed, only sidewalks adjacent to public streets.

Jacoby Heights Subdivision Revised Narrative

Page 21 of 49

- d) Walkway/driveway crossings shall be minimized. Internal parking lot design shall maintain ease of access for pedestrians from abutting streets, pedestrian facilities, and transit stops.
- e) With the exception of walkway/driveway crossings, walkways shall be separated from vehicle parking or vehicle maneuvering areas by grade, different paving material, painted crosshatching or landscaping. They shall be constructed in accordance with the sidewalk standards adopted by the City. (This provision does not require a separated walkway system to collect drivers and passengers from cars that have parked on site unless an unusual parking lot hazard exists).
- f) Pedestrians amenities such as covered walk-ways, awnings, visual corridors and benches will be encouraged. For every two benches provided, the minimum parking requirements will be reduced by one, up to a maximum of four benches per site. Benches shall have direct access to the circulation system.
 Response: The requirements of these sections are not applicable to the proposed subdivision.
- C. Where a development site is traversed by or adjacent to a future trail linkage identified within the Transportation System Plan, improvement of the trail linkage shall occur concurrent with development. Dedication of the trail to the City shall be provided in accordance with 17.84.80. *Response:* No trails are identified in the City's Transportation System Plan or Parks Master Plan on the subject property.
- D. To provide for orderly development of an effective pedestrian network, pedestrian facilities installed concurrent with development of a site shall be extended through the site to the edge of adjacent property(ies). *Response:* No pedestrian facilities, except sidewalks are proposed.
- E. To ensure improved access between a development site and an existing developed facility such as a commercial center, school, park, or trail system, the Planning Commission or Director may require off-site pedestrian facility improvements concurrent with development.
- *Response:* No off-site pedestrian improvements have been identified. 17.84.40 - TRANSIT AND SCHOOL BUS TRANSIT REQUIREMENTS
 - A. Development sites located along existing or planned transit routes shall, where appropriate, incorporate bus pull-outs and/or shelters into the site design. These improvements shall be installed in accordance with the guidelines and standards of the transit agency. School bus pull-outs and/or shelters may also be required, where appropriate, as a condition of approval for a residential development of greater than 50 dwelling units where a school bus pick-up point is anticipated to serve a large number of children.

Jacoby Heights Subdivision Revised Narrative

Page 22 of 49

Response: The subject property is not located along an existing or planned transit route as shown on the city's Transit Master Plan or does it contain greater than 50 dwellings (32 proposed). During the pre-application conference the city Transit Manager identified a required transit amenity at the northwest corner of the property adjacent to Tract A. The location of this facility is shown on the plan set.

- B. New developments at or near existing or planned transit or school bus transit stops shall design development sites to provide safe, convenient access to the transit system, as follows:
 - 1. Commercial and civic use developments shall provide a prominent entrance oriented towards arterial and collector streets, with front setbacks reduced as much as possible to provide access for pedestrians, bicycles, and transit.
 - 2. All developments shall provide safe, convenient pedestrian walkways between the buildings and the transit stop, in accordance with the provisions of 17.84.30 B.

Response: The proposed residential subdivision complies with the requirements of this section.

17.84.50 - STREET REQUIREMENTS

- A. Traffic evaluations may be required of all development proposals in accordance with the following:
 - 1. A proposal establishing the scope of the traffic evaluation shall be submitted for review to the City Engineer. The evaluation requirements shall reflect the magnitude of the project in accordance with accepted traffic engineering practices. Large projects should assess all nearby key intersections. Once the scope of the traffic evaluation has been approved, the applicant shall present the results with and an overall site development proposal. If required by the City Engineer, such evaluations shall be signed by a Licensed Professional Civil Engineer or Licensed Professional Traffic Engineer licensed in the State of Oregon.
 - If the traffic evaluation identifies level-of-service conditions less than the minimum standard established in the Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered concurrent with a development proposal. *Response: A Traffic Impact Study has been included with this application as requested by the City. This study does not identify any required mitigation.*
- B. Location of new arterial streets shall conform to the Transportation System Plan in accordance with the following:
 - 1. Arterial streets should generally be spaced in one-mile intervals.
 - 2. Traffic signals should generally not be spaced closer than 1500 ft. for reasonable traffic progression.

Jacoby Heights Subdivision Revised Narrative

Page 23 of 49

Response: No new arterial streets are required as part of this project.

- C. Local streets shall be designed to discourage through traffic. NOTE: for the purposes of this section, "through traffic" means the traffic traveling through an area that does not have a local origination or destination. To discourage through traffic and excessive vehicle speeds the following street design characteristics shall be considered, as well as other designs intended to discourage traffic:
 - 1. Straight segments of local streets should be kept to less than a quarter mile in length. As practical, local streets should include traffic calming features, and design features such as curves and "T" intersections while maintaining pedestrian connectivity.
 - 2. Local streets should typically intersect in "T" configurations rather than 4way intersections to minimize conflicts and discourage through traffic. Adjacent "T" intersections shall maintain a minimum of 150 ft. between the nearest edges of the 2 rights-of-way.

Response: The proposed subdivision does not include any long straight street segments. The distance between the proposed cul-de-sac and Jacoby Road is 150 feet in accordance with this section. All streets have been designed in accordance with the requirements of these sections.

3. Cul-de-sacs should generally not exceed 400 ft. in length nor serve more than 20 dwelling units, except in cases where existing topography, wetlands, or drainage systems or other existing features necessitate a longer cul-de-sac in order to provide adequate access to an area. Cul-de-sacs longer than 400 feet or developments with only one access point may be required to provide an alternative access for emergency vehicle use only, install fire prevention sprinklers, or provide other mitigating measures, determined by the City.

Response: Camden Court, a proposed cul-de-sac, is approximately 219 feet long and will serve five lots in compliance with this section. Servicing this area of the subject property is challenging due to topographic and natural resource considerations and this is the reason a cul-de-sac has been proposed.

- D. Development sites shall be provided with access from a public street improved to City standards in accordance with the following:
 - 1. Where a development site abuts an existing public street not improved to City standards, the abutting street shall be improved to City standards along the full frontage of the property concurrent with development. *Response:* All homes will gain access from a public street improved to city standards in compliance with this section.
 - 2. Half-street improvements are considered the minimum required improvement. Three quarter-street or full-street improvements shall be

Jacoby Heights Subdivision Revised Narrative

Page 24 of 49

required where traffic volumes generated by the development are such that a half-street improvement would cause safety and/or capacity problems. Such a determination shall be made by the City Engineer.

Response: All new streets are proposed as full street improvements with the exception of the extension of Newton Street on the subject property. This street is proposed as a full street adjacent to Lot 9, a three-quarter street adjacent to Lots 2-8, tapering from a three-quarter street to a half street on Lot 2, and is proposed be constructed as a half street adjacent to Lot 1. The applicant is proposing tapering this street to a one-half street improvement at the east end of the subdivision in order to allow this street to be extended to the east without impacting the existing home located at 19260 SE Jacoby Road. As proposed this road when extended as a one-half street would be located about 15 feet from the front porch of this home rather than one-foot if this road were extended as a threequarter street.

- 3. To ensure improved access to a development site consistent with policies on orderly urbanization and extension of public facilities the Planning Commission or Director may require off-site improvements concurrent with development. Off-site improvement requirements upon the site developer shall be reasonably related to the anticipated impacts of the development. *Response:* No off-site improvements have been identified or are warranted with construction of this subdivision.
- 4. Reimbursement agreements for 3/4 street improvements (i.e., curb face to curb face) may be requested by the developer per Chapter 12 of the SMC. Response: The applicant may request reimbursement for this street per the requirements of Chapter 12 following construction of this improvement.
- 5. A ½ street improvement includes curb and pavement 2 feet beyond the center line of the right-of-way. A ¾ street improvement includes curbs on both sides of the side and full pavement between curb faces. Response: Both the short section of the one-half street section and the three-quarter street section are proposed to be constructed in accordance with these standards.
- E. As necessary to provide for orderly development of adjacent properties, public streets installed concurrent with development of a site shall be extended through the site to the edge of the adjacent property(ies) in accordance with the following:
 - 1. Temporary dead-ends created by this requirement to extend street improvements to the edge of adjacent properties may be installed without turn-arounds, subject to the approval of the Fire Marshal.
 - 2. In order to assure the eventual continuation or completion of the street, reserve strips may be required.

Jacoby Heights Subdivision Revised Narrative

Page 25 of 49

Response: The proposed street layout results in one temporary dead-end street (Woodstock Street) and one street that will be used to provide access to the two rural dwellings located at 19260 and 19270 Jacoby Road just east of the subject property (Newton Street). The applicant is aware the Fire Marshal will need to review the proposal. In addition, the applicant is aware that reserve strips will likely be required at the end of these streets.

- F. Where required by the Planning Commission or Director, public street improvements may be required through a development site to provide for the logical extension of an existing street network or to connect a site with a nearby neighborhood activity center, such as a school or park. Where this creates a land division incidental to the development, a land partition shall be completed concurrent with the development. *Response:* The applicant does not anticipate that any public street improvements will be required to be extended beyond the site boundaries. No such improvements were identified at the pre-application conference.
- G. Except for extensions of existing streets, no street names shall be used that will duplicate or be confused with names of existing streets. Street names and numbers shall conform to the established pattern in the surrounding area and be subject to approval of the Director.

Response: The southern most proposed street is an extension of Newton Street and will carry the same name. The proposed names for the other streets in the subdivision have been selected to conform to names of New England towns, similar to those used in the Cascade Village Subdivision to the west. These names include: Camden Court, Woodstock Street, and Middlebury Avenue.

- H. Location, grades, alignment, and widths for all public streets shall be considered in relation to existing and planned streets, topographical conditions, public convenience and safety, and proposed land use. Where topographical conditions present special circumstances, exceptions to these standards may be granted by the City Engineer provided the safety and capacity of the street network is not adversely affected. The following standards shall apply:
 - 1. Location of streets in a development shall not preclude development of adjacent properties. Streets shall conform to planned street extensions identified in the Transportation Plan and/or provide for continuation of the existing street network in the surrounding area.

Response: A future street plan is submitted with this application as Sheet 8. This plan shows that the proposal will facilitate and not preclude development on adjacent properties. No roads identified on the TSP are shown on the subject property.

Jacoby Heights Subdivision Revised Narrative

Page 26 of 49

 Grades shall not exceed 6 percent on arterial streets, 10 percent on collector streets, and 15 percent on local streets.
 Response: All streets in the proposed development are local streets. The maximum grade on these streets is less than the maximum allowed by this section.

 As far as practical, arterial streets and collector streets shall be extended in alignment with existing streets by continuation of the street centerline. When staggered street alignments resulting in "T" intersections are unavoidable, they shall leave a minimum of 150 ft. between the nearest edges of the two rights-of-way. *Response:* The alignment of Jacoby Road, a collector street, has already been established. The proposed plan features all street intersections a minimum of 150 feet between each other. The distance between Jacoby

minimum of 150 feet between each other. The distance between Jacoby Road and the proposed cul-de-sac (Camden Court) is 150 feet and the distance between Camden Court and Middlebury Avenue to the east is approximately 193 feet in compliance with this standard.

- 4. Centerline radii of curves shall not be less than 500 ft. on arterial streets, 300 ft. on collector streets, and 100 ft. on local streets. *Response:* All proposed streets in the subdivision with the exception of short segments of the western end of Newton Street are straight. The short segment of this street has a 100 foot centerline radii in compliance with this standard.
- 5. Streets shall be designed to intersect at angles as near as practicable to right angles and shall comply with the following:
 - a) The intersection of an arterial or collector street with another arterial or collector street shall have a minimum of 100 ft. of straight (tangent) alignment perpendicular to the intersection.
 - b) The intersection of a local street with another street shall have a minimum of 50 ft. of straight (tangent) alignment perpendicular to the intersection.
 - c) Where right angle intersections are not possible, exceptions can be granted by the City Engineer provided that intersections not at right angles have a minimum corner radius of 20 ft. along the right-of-way lines of the acute angle.
 - d) Intersections with arterial streets shall have a minimum curb corner radius of 20 ft. All other intersections shall have a minimum curb corner radius of 10 ft.

Response: All proposed streets are designed to insect at right angles with the intersecting street and comply with the requirements of this section. Newton Street constructed on the subject property intersecting with Jacoby Road maintains a 50 foot straight tangent as required as shown on Sheets 1 and 5.

Jacoby Heights Subdivision Revised Narrative

Page 27 of 49

6. Right-of-way and improvement widths shall be as specified by the Transportation System Plan. Exceptions to those specifications may be approved by the City Engineer to deal with specific unique physical constraints of the site.

Response: The proposed right-of-way width of all full-width local streets is 50 feet in compliance this standard. The right-of-way width of the extension of Newton Street to be constructed along the southern property line is designed in compliance with the standards for a three-quarter street and a one-half street as applicable.

J. Private streets may be considered within a development site provided all the following conditions are met: *Response:* No private streets are proposed.

17.84.60 - PUBLIC FACILITY EXTENSIONS

- A. All development sites shall be provided with public water, sanitary sewer, broadband (fiber), and storm drainage.
 Response: The submitted Utility Plan shows the location of proposed public water, sanitary sewer, and stormwater drainage facilities. Broadband fiber service will be detailed in construction plans.
- B. Where necessary to serve property as specified in "A" above, required public facility installations shall be constructed concurrent with development. *Response:* All of the utilities identified above will be constructed concurrent with the development.
- C. Off-site public facility extensions necessary to fully serve a development site and adjacent properties shall be constructed concurrent with development. *Response:* The applicant will extend all utilities as necessary to serve the development as required by this section.
- D. As necessary to provide for orderly development of adjacent properties, public facilities installed concurrent with development of a site shall be extended through the site to the edge of adjacent property(ies).
 Response: As shown on the submitted Utility Plan, all public facilities are proposed to be extended through the site to edge of adjacent properties.
- E. Private on-site sanitary sewer and storm drainage facilities may be considered provided all the following conditions exist: *Response:* No private utilities are proposed.
- 17.84.70 PUBLIC IMPROVEMENT PROCEDURES Response: The applicant is aware of and intends to comply with the requirements of this section.

Jacoby Heights Subdivision Revised Narrative

Page 28 of 49

17.84.80 - FRANCHISE UTILITY INSTALLATIONS

These standards are intended to supplement, not replace or supersede, requirements contained within individual franchise agreements the City has with providers of electrical power, telephone, cable television, and natural gas services (hereinafter referred to as "franchise utilities").

A. Where a land division is proposed, the developer shall provide franchise utilities to the development site. Each lot created within a subdivision shall have an individual service available or financially guaranteed prior to approval of the final plat.

Response: Franchise utilities will be provided to all lots within the proposed subdivision as required. The location of these utilities will be identified on construction plans and installed or guaranteed prior to final plat approval.

B. Where necessary, in the judgment of the Director, to provide for orderly development of adjacent properties, franchise utilities shall be extended through the site to the edge of adjacent property(ies), whether or not the development involves a land division.

Response: The applicant does not anticipate extending franchise utilities beyond the site.

- C. The developer shall have the option of choosing whether or not to provide natural gas or cable television service to the development site, providing all of the following conditions exist:
 - 1. Extension of franchise utilities through the site is not necessary for the future orderly development of adjacent property(ies);
 - 2. The development site remains in one ownership and land division does not occur (with the exception of land divisions that may occur under the provisions of 17.84.50 F above); and
 - 3. The development is non-residential.

Response: The applicant anticipates installing natural gas and cable television service will be installed as required.

- D. Where a land division is not proposed, the site shall have franchise utilities required by this section provided in accordance with the provisions of 17.84.70 prior to occupancy of structures. *Response: A land division is proposed, as such this section is not applicable.*
- E. All franchise utility distribution facilities installed to serve new development shall be placed underground except as provided below. The following facilities may be installed aboveground:
 - 1. Poles for street lights and traffic signals, pedestals for police and fire system communications and alarms, pad mounted transformers, pedestals, pedestal mounted terminal boxes and meter cabinets, concealed ducts, substations, or facilities used to carry voltage higher than 35,000 volts;

Jacoby Heights Subdivision Revised Narrative

Page 29 of 49

- Overhead utility distribution lines may be permitted upon approval of the City Engineer when unusual terrain, soil, or other conditions make underground installation impracticable. Location of such overhead utilities shall follow rear or side lot lines wherever feasible. *Response:* All franchise utilities will be installed underground with the exception of street lights as allowed by this section.
- F. The developer shall be responsible for making necessary arrangements with franchise utility providers for provision of plans, timing of installation, and payment for services installed. Plans for franchise utility installations shall be submitted concurrent with plan submittal for public improvements to facilitate review by the City Engineer.

Response: The developer will make all necessary arrangements with franchise utility providers as required by this section.

- G. The developer shall be responsible for installation of underground conduit for street lighting along all public streets improved in conjunction with the development in accordance with the following:
 - 1. The developer shall coordinate with the City Engineer to determine the location of future street light poles. The street light plan shall be designed to provide illumination meeting standards set by the City Engineer.
 - 2. The developer shall make arrangements with the serving electric utility for trenching prior to installation of underground conduit for street lighting. *Response:* The developer will install underground conduit for street lighting in accordance with the requirements of this section.

17.84.90 - LAND FOR PUBLIC PURPOSES

- A. Easements for public sanitary sewer, water, storm drain, pedestrian and bicycle facilities shall be provided whenever these facilities are located outside a public right-of-way in accordance with the following:
 - 1. When located between adjacent lots, easements shall be provided on one side of a lot line.
 - 2. The minimum easement width for a single utility is 15 ft. The minimum easement width for two adjacent utilities is 20 ft. The easement width shall be centered on the utility to the greatest extent practicable. Wider easements may be required for unusually deep facilities. Response: A 15 foot wide easement is proposed to be centered on Lot 30 and Tract A for sanitary sewer service and a 20 foot wide easement adjacent to Jacoby Road on Lots 9, 10, 30, and 32 is proposed for sanitary sewer and stormwater purpose in compliance with this section. Ten foot private storm easements are located at the back of Lots 1 18, 19 25, 28, 29, 31, and 32. No other easements are proposed.
- B. Public utility easements with a minimum width of 5 feet shall be provided adjacent to all street rights-of-way for franchise utility installations.

Jacoby Heights Subdivision Revised Narrative

Page 30 of 49

Response: Despite the language in this section, eight foot wide public utility easements are provided along all lots adjacent to street rights-of-way for future franchise utility installations.

- C. Where a development site is traversed by a drainageway or water course, a drainage way dedication shall be provided to the City. *Response:* No dedication along Tickle Creek is proposed at this time.
- D. Where a development is traversed by, or adjacent to, a future trail linkage identified within the Transportation System Plan, dedications of suitable width to accommodate the trail linkage shall be provided. This width shall be determined by the City Engineer, considering the type of trail facility involved. *Response: No future trail is identified in the TSP on subject property.*
- E. Where existing rights-of-way and/or easements within or adjacent to development sites are nonexistent or of insufficient width, dedications may be required. The need for and widths of those dedications shall be determined by the City Engineer.

Response: The only existing right-of-way adjacent to the development is Jacoby Road. A ten foot dedication to accommodate improvements to road is proposed along the entire property adjacent to this road.

F. Where easement or dedications are required in conjunction with land divisions, they shall be recorded on the plat. Where a development does not include a land division, easements and/or dedications shall be recorded on standard document forms provided by the City Engineer.

Response: All easements and dedications will be identified on the plat as required.

17.84.100 - MAIL DELIVERY FACILITIES

Response: The location and type of mail delivery facilities will be coordinated with the City Engineer and the Post Office as part of the construction plan process. The conceptual location of these facilities are shown on Sheet 7, Preliminary Parking Plan.

CHAPTER 17.86 - PARKLAND and OPEN SPACE 17.86.00 - INTENT

The availability of parkland and open space is a critical element in maintaining and improving the quality of life in Sandy. Land that features trees, grass and vegetation provides not only an aesthetically pleasing landscape but also buffers incompatible uses, and preserves sensitive environmental features and important resources. Parks and open space, together with support facilities, also help to meet the active and passive recreational needs of the population of Sandy. This chapter implements policies of Goal 8 of the Comprehensive Plan and the Parks Master Plan by outlining provisions for parks and open space in the City of Sandy.

Jacoby Heights Subdivision Revised Narrative

Page 31 of 49

Response: The city's master plans do not show any parks or trails on the subject property. The subject property contains a small section of Tickle Creek at the northeast corner of the property. The subject property also contains relatively steep slopes behind Lots 19-21 and 26 - 28.

17.86.10 - MINIMUM PARKLAND DEDICATION REQUIREMENTS

Parkland Dedication: New residential subdivisions, planned developments, multifamily or manufactured home park developments shall be required to provide parkland to serve existing and future residents of those developments. **Response:** The proposed residential subdivision is subject to the provisions of this chapter.

- 1. The required parkland shall be dedicated as a condition of approval for the following:
 - a. Tentative plat for a subdivision or partition;
 - b. Planned Development conceptual or detailed development plan;
 - Design review for a multi-family development or manufactured home park; and
 - d. Replat or amendment of any site plan for multi-family development or manufactured home park where dedication has not previously been made or where the density of the development involved will be increased. *Response:* No parkland has been identified on the tentative plat.
- 2. Calculation of Required Dedication: The required parkland acreage to be dedicated is based on a calculation of the following formula rounded to the nearest 1/100 (0.00) of an acre:

Required parkland dedication (acres) = (proposed units) x (persons/unit) x 0.0043 (per person park land dedication factor)

Response: The proposed 32 lot subdivision results in the following formal: 32 (proposed units) x 3 (persons/unit) x 0.0043 (per person park land dedication factor) = 0.4128 rounded to 0.41.

17.86.40 - CASH IN LIEU OF DEDICATION

At the city's discretion only, the city may accept payment of a fee in lieu of land dedication. The city may require payment in lieu of land when the park land to be dedicated is less than 3 acres. A payment in lieu of land dedication is separate from Park Systems Development Charges, and is not eligible for a credit of Park Systems Development Charges. The amount of the fee in lieu of land dedication (in dollars per acre) shall be set by City Council Resolution, and it shall be based on the typical market value of developed property (finished lots) in Sandy net of related development costs.

Response: City staff has indicated they would consider recommending parks feein-lieu credit if the applicant dedicates parkland but a formal proposal has not been brought forward. The applicant requests the city consider providing park fee-in-lieu credit for any land dedication. The amount of this fee will be

Jacoby Heights Subdivision Revised Narrative

Page 32 of 49

determined based on the City Council Fee Resolution with approval of the subdivision.

CHAPTER 17.92 - LANDSCAPING AND SCREENING GENERAL STANDARDS -ALL ZONES

Response: This chapter has limited applicability to subdivisions so only those applicable sections are reviewed in this submittal.

17.92.10 - GENERAL PROVISIONS

- A. Where landscaping is required by this Code, detailed planting plans shall be submitted for review with development applications. No development may commence until the Director or Planning Commission has determined the plans comply with the purposes clause and specific standards in this chapter. All required landscaping and related improvements shall be completed or financially guaranteed prior to the issuance of a Certificate of Occupancy.
- B. Appropriate care and maintenance of landscaping onsite and landscaping in the adjacent public right-of-way is the right and responsibility of the property owner, unless City ordinances specify otherwise for general public and safety reasons. If street trees or other plant materials do not survive or are removed, materials shall be replaced in kind within 6 months.
- C. Significant plant and tree specimens should be preserved to the greatest extent practicable and integrated into the design of a development. Trees of 25-inches or greater circumference measured at a height of 4-1/2 ft. above grade are considered significant. Plants to be saved and methods of protection shall be indicated on the detailed planting plan submitted for approval. Existing trees may be considered preserved if no cutting, filling, or compaction of the soil takes place between the trunk of the tree and the area 5-ft. outside the tree's drip line. Trees to be retained shall be protected from damage during construction by a construction fence located 5 ft. outside the dripline. *Response: City staff has recently started to use this section to specify tree protection within residential subdivisions. The applicant has submitted an interpretation request per the requirements of Chapter 17.14 with this application. This request asks that the Planning Commission make a formal determination regarding applicable tree protection regulations for residential subdivisions.*

17.92.20 - MINIMUM IMPROVEMENTS - LANDSCAPING AND SCREENING *Response:* The Single Family Residential zone is not listed in this section requiring minimum landscaping.

CHAPTER 17.98 - PARKING, LOADING, AND ACCESS REQUIREMENTS 17.98.10 GENERAL PROVISIONS

Jacoby Heights Subdivision Revised Narrative

Page 33 of 49

- M. <u>Residential Parking Analysis Plan</u>. A Residential Parking Analysis Plan shall be required for all new residential planned developments, subdivisions, and partitions to include a site plan depicting all of the following:
 - a. Location and dimension of required parking spaces as specified in Section 17.98.200.
 - b. Location of areas where parking is not permitted as specified in Sections 17.98.200(A)(3) and (5).
 - c. Location and design of parking courts (if applicable).

Response: A Residential Parking Analysis Plan as required by this section is included as Sheet 7 of the plan set.

17.98.80 - ACCESS TO ARTERIAL AND COLLECTOR STREETS

Response: No lots are proposed to gain access from an arterial or collector street.

17.98.90 - ACCESS TO UNIMPROVED STREETS

Response: All streets proposed in the subdivision will be improved to city standards.

17.98.100 - DRIVEWAYS

- A. A driveway to an off-street parking area shall be improved from the public roadway to the parking area a minimum width of 20 feet for a two-way drive or 12 feet for a one-way drive but in either case not less than the full width of the standard approach for the first 20 feet of the driveway. *Response:* All lots will have a standard 24 foot wide curb cut and driveway approach.
- B. A driveway for a single-family dwelling shall have a minimum width of 10 feet. *Response:* All lots will have a standard 24 foot wide curb cut and driveway approach. Due to the narrow street frontage of Lots 27/28, these lots will likely be accessed by a joint curb cut serving both lots.
- C. A driveway for a two-family dwelling shall have a minimum width of 20 feet. A driveway approach must be constructed in accordance with applicable city standards and the entire driveway must be paved with asphalt or concrete. *Response:* All of the proposed lots will be constructed with a single family dwelling so this section is not applicable.
- D. Driveways, aisles, turnaround areas and ramps shall have a minimum vertical clearance of twelve feet for their entire length and width but such clearance may be reduced in parking structures. *Response:* All driveways will be designed in compliance with this standard.
- E. No driveway shall traverse a slope in excess of 15 percent at any point along the driveway length.

Jacoby Heights Subdivision Revised Narrative

Page 34 of 49
Response: All driveways will be designed in compliance with this standard.

F. The location and design of the driveway shall provide for unobstructed sight per the vision clearance requirements. Requests for exceptions to these requirements will be evaluated by the City Engineer considering the physical limitations of the lot and safety impacts to vehicular, bicycle, and pedestrian traffic.

Response: All driveways will be designed in compliance with this standard.

17.98.110 - VISION CLEARANCE

A. Except within the Central Business District, vision clearance areas shall be provided at intersections of all streets and at intersections of driveways and alleys with streets to promote pedestrian, bicycle, and vehicular safety. The extent of vision clearance to be provided shall be determined from standards in Chapter 17.74 and taking into account functional classification of the streets involved, type of traffic control present at the intersection, and designated speed for the streets.

Response: The subject property is located in the SFR zone requiring compliance with this section. The requirements of this section will be considered in placing landscaping in these areas with construction of homes.

B. Traffic control devices, streetlights, and utility installations meeting approval by the City Engineer are permitted within vision clearance areas. *Response:* The exceptions contained in this section will be considered in the design and placement of these structures.

17.98.200 - RESIDENTIAL ON-STREET PARKING REQUIREMENTS

- A. Residential On-Street Parking Requirements. Residential on-street parking shall conform to the following standards:
 - 1. In addition to required off-street parking, all new residential planned developments, subdivisions and partitions shall provide one (1) on-street parking space within 200 feet of each dwelling except as provided in Section 17.98.200(A)(6) below.
 - 2. The location of residential on-street parking shall be reviewed for compliance with this section through submittal of a Residential Parking Analysis Plan as required in Section 17.98.10(M).
 - 3. Residential on-street parking shall not obstruct required clear vision areas and shall not violate any local or state laws.
 - 4. Parallel residential on-street parking spaces shall be 22 feet minimum in length.

Jacoby Heights Subdivision Revised Narrative

Page 35 of 49

5. Residential on-street parking shall be measured along the curb from the outside edge of a driveway wing or curb cut. Parking spaces must be set back a minimum of 15 feet from an intersection and may not be located within 10 feet of a fire hydrant.

Response: A Residential On-Street Parking Analysis designed in compliance with the requirements of this section is included as Sheet 7 of the application package. One on-street parking space at least 22 feet in length has been identified within 200 feet of each of the 32 lots. An additional 30 on-street parking spaces have also been identified as shown on the Parking Plan.

6. Portions of residential on-street parking required by this section may be provided in parking courts that are interspersed throughout a development when the following standards are met: *Response:* No parking courts are proposed.

CHAPTER 17.100 - LAND DIVISION

- 17.100.20 LAND DIVISION CLASSIFICATION TYPE I, II OR III PROCEDURES
- C. Type II Land Division (Major Partition or Subdivision). A major partition or subdivision shall be a Type II procedure when a street is extended, satisfactory street conditions exist and the resulting parcels/lots comply with the standards of the zoning district and this chapter. Satisfactory street conditions exist when the Director determines one of the following:
 - 1. Existing streets are stubbed to the property boundaries and are linked by the land division.
 - 2. An existing street or a new proposed street need not continue beyond the land division in order to complete an appropriate street system or to provide access to adjacent property.
 - 3. The proposed street layout is consistent with a street pattern adopted as part of the Comprehensive Plan or an officially adopted City street plan. *Response:* The proposed subdivision complies with all applicable code requirements to be process as a Type II application with the exception of three items noted above. The applicant is requesting three variances to applicable code standards and as such, the entire application will be processed as a Type III application.

17.100.60 - SUBDIVISIONS

Approval of a subdivision is required for a land division of 4 or more parcels in a calendar year. A two-step procedure is required for subdivision approval: (1) tentative plat review and approval; and (2) final plat review and approval. *Response:* As defined by this section the 32 lot land division is a subdivision.

A. Preapplication Conference. The applicant for a subdivision shall participate in a preapplication conference with city staff to discuss procedures for approval,

Jacoby Heights Subdivision Revised Narrative

Page 36 of 49

applicable state and local requirements, objectives and policies of the Sandy Comprehensive Plan, and the availability of services.

Response: A pre-application conference was held with the city on January 3, 2018.

B. Application Requirements for a Tentative Plat. Subdivision applications shall be made on forms provided by the planning department and shall be accompanied by:

Response: All of the items required by this section were included with the submittal.

E. Approval Criteria. The Director or Planning Commission shall review the tentative plat for the subdivision based on the classification procedure (Type II or III) set forth in Section 17.12 and the following approval criteria:

1. The proposed subdivision is consistent with the density, setback and dimensional standards of the base zoning district, unless modified by a Planned Development approval.

Response: As reviewed in the narrative above, the proposed subdivision is designed to be consistent with density, setback, and dimensional standards in the SFR zoning district.

2. The proposed subdivision is consistent with the design standards set forth in this chapter.

Response: With the exception of three variances requested with this application, the proposal complies with the design standards in this chapter.

- 3. The proposed street pattern is connected and consistent with the Comprehensive Plan or official street plan for the City of Sandy. *Response:* As illustrated on the submitted Future Street Plan (Sheet 8), the proposed street system is consistent with the City's Transportation System Plan and Comprehensive Plan.
- 4. Adequate public facilities are available or can be provided to serve the proposed subdivision.

Response: The City of Sandy has confirmed (Exhibit I) the sanitary sewer pump station constructed with the Timberline Trails Subdivision located along Jacoby Road north of the subject property has sufficient capacity to accommodate the proposed development. The city has indicated that all other public facilities have capacity to serve the proposed subdivision.

5. All proposed improvements meet City standards. *Response:* With the exception of the three variances requested above, all improvements in the proposed development are designed in compliance with City standards.

Jacoby Heights Subdivision Revised Narrative

Page 37 of 49

6. The phasing plan, if requested, can be carried out in a manner that meets the objectives of the above criteria and provides necessary public improvements for each phase as it develops.

Response: The applicant proposes developing the subdivision in a single phase.

17.100.80 - CHARACTER OF THE LAND

Land which the Director or the Planning Commission finds to be unsuitable for development due to flooding, improper drainage, steep slopes, rock formations, adverse earth formations or topography, utility easements, or other features which will reasonably be harmful to the safety, health, and general welfare of the present or future inhabitants of the partition or subdivision and the surrounding areas, shall not be developed unless adequate methods are formulated by the subdivider and approved by the Director or the Planning Commission to solve the problems created by the unsuitable land conditions.

Response: A small area of the subject property is affected by the FSH overlay identified on the City of Sandy Zoning Map. The area of this overlay has been refined by the applicant as shown on Sheet 4 of the plan set. The applicant does not propose any development within this area. A Geotechnical Evaluation for the property is included as part of the application package (Exhibit F).

17.100.90 - ACCESS CONTROL GUIDELINES AND COORDINATION

A. Notice and coordination with ODOT required. The city will coordinate and notify ODOT regarding all proposals for new or modified public and private accesses on to Highways 26 and 211.

Response: The subject property does not abut Highways 26 and 211.

17.100.100 - STREETS GENERALLY

- A. <u>Transportation Impact Studies</u>. Transportation impact studies may be required by the city engineer to assist the city to evaluate the impact of development proposals, determine reasonable and prudent transportation facility improvements and justify modifications to the design standards. Such studies will be prepared in accordance with the following:
 - 1. A proposal established with the scope of the transportation impact study shall be coordinated with, and agreed to, by the city engineer. The study requirements shall reflect the magnitude of the project in accordance with accepted transportation planning and engineering practices. A professional civil or traffic engineer registered in the State of Oregon shall prepare such studies.
 - If the study identifies level-of-service conditions less than the minimum standards established in the Sandy Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered as part of the land use decision for the proposal. *Response:* A traffic impact analysis prepared in compliance with city standards has been submitted with the application (Exhibit E). This study does not identify any issues requiring mitigation by the applicant.

Jacoby Heights Subdivision Revised Narrative

Page 38 of 49

- B. <u>Topography and Arrangement.</u> All streets shall be properly related to special traffic generators such as industries, business districts, schools, and shopping centers and to the pattern of existing and proposed land uses. *Response:* None of the special traffic generators listed in this section are located near the subject property. All existing and proposed residential uses have been considered in development of the proposed street pattern. A future street plan is submitted with this application (Sheet 8) showing how streets can be extended beyond the subject property in the future.
- C. <u>Street Spacing</u>. Street layout shall generally use a rectangular grid pattern with modifications as appropriate to adapt to topography or natural conditions. *Response:* The proposed street layout represents a rectangular grid pattern.
- D. Future Street Plan. Future street plans are conceptual plans, street extensions and connections on acreage adjacent to land divisions. They assure access for future development and promote a logical, connected pattern of streets. It is in the interest of the city to promote a logical, connected pattern of streets. All applications for land divisions shall provide a future street plan that shows the pattern of existing and proposed future streets within the boundaries of the proposed land divisions, proposed connections to abutting properties, and extension of streets to adjacent parcels within a 400 foot radius of the study area where development may practically occur. Response: A future street plan in compliance with the requirements of this

section was submitted as part of the application package (Sheet 8). This plan assures that access for future development will promote a logical and connected pattern of streets.

E. <u>Connections.</u> Except as permitted under Exemptions, all streets, alleys and pedestrian walkways shall connect to other streets within the development and to existing and planned streets outside the development and to undeveloped properties which have no future street plan. Streets shall terminate at other streets or at parks, schools or other public land within a neighborhood.

Where practicable, local roads shall align and connect with other roads when crossing collectors and arterials.

Proposed streets or street extensions shall be located to provide direct access to existing or planned transit stops, and existing or planned neighborhood activity centers, such as schools, shopping areas and parks. **Response:** All streets proposed in the subdivision will allow connection to other streets with the exception of the proposed cul-de-sac, Camden Court. Due to the location of Tickle Creek and steep slopes on the subject property, it is not practical or desirable to serve this portion of the property with a connected street network.

17.100.120 - BLOCKS AND ACCESSWAYS

Jacoby Heights Subdivision Revised Narrative

Page 39 of 49

- A. <u>Blocks.</u> Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features. *Response:* All blocks within the proposed subdivision have sufficient width to provide for two tiers of lots within the exception of the block containing Lot 30. Because direct access is not permitted to Jacoby Road a street is required on the east side of this lot to provide access. The applicant is requesting a variance to this section as reviewed in Chapter 17.66 above.
- B. <u>Residential Blocks.</u> Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance. *Response:* As reviewed in Chapter 17.66 above, the applicant is requesting a Type III Variance to this section for the block on the north side of Woodstock Street between Camden Court and the eastern property line. The length of this block is 420 feet, 20 feet greater than the maximum block length.
- D. <u>Pedestrian and Bicycle Access Way Requirements.</u> In any block in a residential or commercial district over 600 feet in length, a pedestrian and bicycle accessway with a minimum improved surface of 10 feet within a 15-foot right-of-way or tract shall be provided through the middle of the block. To enhance public convenience and mobility, such accessways may be required to connect to cul-de-sacs, or between streets and other public or semipublic lands or through greenway systems.

Response: None of the blocks within the proposed subdivision exceed 600 feet in length. The applicant is not proposing access to the Tickle Creek corridor in this subdivision but rather as shown on Sheet 8, a pedestrian access trail is identified in a more suitable location on the property directly east of the subject property.

17.100.130 - EASEMENTS

A minimum eight (8) foot public utility easement shall be required along property lines abutting a right-of-way for all lots within a partition or subdivision. Where a partition or subdivision is traversed by a watercourse, drainage way, channel or stream, the land division shall provide a stormwater easement or drainage right-of-way conforming substantially with the lines of such watercourse, and such further width as determined needed for water quality and quantity protection. *Response:* Eight foot wide public utility easements are identified along all property lines abutting a public right-of-way. In addition, a 20 foot wide easement to accommodate both sanitary sewer and stormwater facilities is proposed along the western line of Lots 9, 10, 30 and 32 and a 15 foot wide sanitary sewer easement between Lot 30 and Tract A is also proposed. Private stormwater easements are located along the back of Lots 1 - 18, 19 - 25, 28, 29, 31, and 32.

Jacoby Heights Subdivision Revised Narrative

Page 40 of 49

17.100.140 - PUBLIC ALLEYS

Response: No alleys are proposed in this development.

17.100.150 RESIDENTIAL SHARED PRIVATE DRIVES

Response: No residential shared private drives are proposed in this development

17.100.160 PUBLIC ACCESS LANES

Response: No public access lanes are proposed in this development

17.100.170 - FLAG LOTS

Flag lots can be created where it can be shown that no other street access is possible to achieve the requested land division. The flag lot shall have a minimum street frontage of 15 feet for its accessway. The following dimensional requirements shall apply to flag lots:

A. Setbacks applicable to the underlying zoning district shall apply to the flag lot.

B. The access strip (pole) may not be counted toward the lot size requirements. *Response:* No flag lots are proposed.

17.100.180 - INTERSECTIONS

A. <u>Intersections.</u> Streets shall be laid out so as to intersect as nearly as possible at right angles. A proposed intersection of two new streets at an angle of less than 75 degrees shall not be acceptable. No more than two streets shall intersect at any one point unless specifically approved by the City Engineer. The city engineer may require left turn lanes, signals, special crosswalks, curb extensions and other intersection elements justified by a traffic study or necessary to comply with the Development Code. *Response:* All streets in the proposed subdivision have been designed to

Response: All streets in the proposed subdivision have been designed to intersect at right angles to the opposing street as required.

B. <u>Curve Radius.</u> All local and neighborhood collector streets shall have a minimum curve radius (at intersections of rights-of-way) of 20 feet, unless otherwise approved by the City Engineer. When a local or neighborhood collector enters on to a collector or arterial street, the curve radius shall be a minimum of 30 feet, unless otherwise approved by the City Engineer. *Response:* All streets in the proposed subdivision have a minimum curve radius as required by this section.

17.100.190 - STREET SIGNS

The subdivider shall pay the cost of street signs prior to the issuance of a Certificate of Substantial Completion. The City shall install all street signs and upon completion will bill the developer for costs associated with installation. In addition, the subdivider may be required to pay for any traffic safety devices related to the development. The City Engineer shall specify the type and location of the street signs and/or traffic safety devices.

Jacoby Heights Subdivision Revised Narrative

Page 41 of 49

Response: The applicant understands it will be his responsibility to pay the cost of street signs and the city will install these signs.

17.100.200 - STREET SURFACING

Public streets, including alleys, within the development shall be improved in accordance with the requirements of the City or the standards of the Oregon State Highway Department. An overlay of asphalt concrete, or material approved by the City Engineer, shall be placed on all streets within the development. Where required, speed humps shall be constructed in conformance with the City's standards and specifications.

Response: All streets in the proposed subdivision will be improved in accordance with City standards.

17.100.210 - STREET LIGHTING

A complete lighting system (including, but not limited to: conduits, wiring, bases, poles, arms, and fixtures) shall be the financial responsibility of the subdivider on all cul-de-sacs, local streets, and neighborhood collector streets. The subdivider will be responsible for providing the arterial street lighting system in those cases where the subdivider is required to improve an arterial street. Standards and specifications for street lighting shall be coordinated with the utility and any lighting district, as appropriate.

Response: The applicant is aware of the requirements of this section. A lighting plan will be coordinated with PGE and the city prior to installation of any fixtures.

17.100.220 - LOT DESIGN

- A. The lot arrangement shall be such that there will be no foreseeable difficulties, for reason of topography or other conditions, in securing building permits to build on all lots in compliance with the Development Code. *Response:* All of the lots in the proposed subdivision have been designed so that no foreseeable difficulties due to topography or other conditions will exist in securing building permits on these lots. A Geotechnical Evaluation report has been submitted with this application (Exhibit F).
- B. The lot dimensions shall comply with the minimum standards of the Development Code. When lots are more than double the minimum lot size required for the zoning district, the subdivider may be required to arrange such lots to allow further subdivision and the opening of future streets to serve such potential lots.

Response: All lots in the proposed subdivision comply with the minimum standards in the SFR zoning district. As shown on the Preliminary Plat, Lots 19 - 21 and 27 contain more than double the minimum lot size of 7,500 square feet required in this zone. Due to the configuration of these lots and steep slopes, further division of these lots is not possible.

Jacoby Heights Subdivision Revised Narrative

Page 42 of 49

C. The lot or parcel width at the front building line shall meet the requirements of the Development Code and shall abut a public street other than an alley for a width of at least 20 feet. A street frontage of not less than 15 feet is acceptable in the case of a flag lot division resulting from the division of an unusually deep land parcel which is of a size to warrant division into not more than two parcels.

Response: All lots in the proposed subdivision contain at least 20 feet of frontage along a public street. As noted above, no flag lots are proposed.

- D. Double frontage lots shall be avoided except where necessary to provide separation of residential developments from arterial streets or to overcome specific disadvantages of topography or orientation. *Response:* Only Lot 30 is considered a double frontage lots because it has frontage on both Jacoby Road and Camden Court. Because direct access to this lot from Jacoby Road (a collector street) is discouraged, access is proposed from Camden Court, a local street. The arrangement of this lot represents a logical pattern and due to the specific conditions of the property this arrangement is warranted.
- E. Lots shall avoid deriving access from major or minor arterials. When driveway access from major or minor arterials may be necessary for several adjoining lots, the Director or the Planning Commission may require that such lots be served by a common access drive in order to limit possible traffic hazards on such streets. Where possible, driveways should be designed and arranged to avoid requiring vehicles to back into traffic on minor or major arterials. *Response: Jacoby Road is designated as a collector street in the City's Transportation System Plan. The applicant proposes accessing all lots in the subdivision from local streets.*

17.100.230 - WATER FACILITIES

Water lines and fire hydrants serving the subdivision or partition, and connecting the development to City mains, shall be installed to provide adequate water pressure to serve present and future consumer demand. The materials, sizes, and locations of water mains, valves, service laterals, meter boxes and other required appurtenances shall be in accordance with the standards of the Fire District, the City, and the State.

If the city requires the subdivider to install water lines in excess of eight inches, the city may participate in the oversizing costs. Any oversizing agreements shall be approved by the city manager based upon council policy and dependent on budget constraints. If required water mains will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement for the proportionate share of the cost. **Response:** The applicant intends to install all water lines and fire hydrants in compliance with applicable standards.

Jacoby Heights Subdivision Revised Narrative

Page 43 of 49

17.100.240 - SANITARY SEWERS

Sanitary sewers shall be installed to serve the subdivision and to connect the subdivision to existing mains. Design of sanitary sewers shall take into account the capacity and grade to allow for desirable extension beyond the subdivision.

If required sewer facilities will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement by nonparticipating landowners for the proportionate share of the cost of construction.

Response: The applicant intends to install sanitary sewer lines in compliance with applicable standards. The Public Works Director has indicated the pump station located in the Timberline Trails Subdivision that will serve the subdivision has adequate capacity (Exhibit I). As indicated in this Exhibit an analysis was completed for the contributing basin to the pump station. The pump station, wet well, and force main were all designed to accommodate 525 dwelling units. The basin contains 284 existing dwelling units and as identified by the Public Works Director has sufficient capacity to accommodate the proposed 32 lot subdivision without any required expansion.

17.100.250 - SURFACE DRAINAGE AND STORM SEWER SYSTEM

A. Drainage facilities shall be provided within the subdivision and to connect with off-site drainage ways or storm sewers. Capacity, grade and materials shall be by a design approved by the city engineer. Design of drainage within the subdivision shall take into account the location, capacity and grade necessary to maintain unrestricted flow from areas draining through the subdivision and to allow extension of the system to serve such areas.

Response: A stormwater water quality and detention facility is proposed to be located at the northwest corner of the site (Tract A). This facility has been sized and located to accommodate all public stormwater generated by the subdivision. A stormwater report is included with this application as Exhibit D. In addition, the proposed stormwater system has been designed to capture water entering the site from the south and convey this water through the site as required.

- B. In addition to normal drainage design and construction, provisions shall be taken to handle any drainage from preexisting subsurface drain tile. It shall be the design engineer's duty to investigate the location of drain tile and its relation to public improvements and building construction. *Response: No subsurface drain tiles are known to exist on the site.*
- C. The roof and site drainage from each lot shall be discharged to either curb face outlets (if minor quantity), to a public storm drain or to a natural acceptable drainage way if adjacent to the lot. *Response:* All roof and site drainage will be discharged to curb face outlets or another approved system as required.

Jacoby Heights Subdivision Revised Narrative

Page 44 of 49

17.100.260 - UNDERGROUND UTILITIES

All subdivisions or major partitions shall be required to install underground utilities (including, but not limited to, electrical and telephone wiring). The utilities shall be installed pursuant to the requirements of the utility company. *Response: The applicant intends to install all utilities underground as required.*

17.100.270 - SIDEWALKS

Sidewalks shall be installed on both sides of a public street and in any special pedestrian way within the subdivision.

Response: Sidewalks will be installed of both sides of all streets with the exception of the three-quarter and one-half street sections of Newton Street extended on the subject property.

17.100.280 - BICYCLE ROUTES

If appropriate to the extension of a system of bicycle routes, existing or planned, the Director or the Planning Commission may require the installation of bicycle lanes within streets. Separate bicycle access ways may be required to reduce walking or cycling distance when no feasible street connection is available. **Response:** No existing, planned, or proposed bicycle routes exist on the subject property. The applicant is aware that street improvements along Jacoby Road may require completion of a bicycle lane as part of these improvements.

17.100.290 - STREET TREES

Where planting strips are provided in the public right-of-way, a master street tree plan shall be submitted and approved by the Director. The street tree plan shall provide street trees approximately every 30' on center for all lots. **Response:** Planter strips will be provided along all frontages as required. Street trees in accordance with City standards will be provided in these areas. A Street Tree Plan is included with the submittal package as Sheet 6.

17.100.300 - EROSION CONTROL

Grass seed planting shall take place prior to September 30th on all lots upon which a dwelling has not been started but the ground cover has been disturbed. The seeds shall be of an annual rye grass variety and shall be sown at not less than four pounds to each 1000 square feet of land area.

Response: Grass seeding will be completed as required by this section. The submitted erosion control plan (Sheet 9) provides additional details to address erosion control concerns. A separate Grading and Erosion Control Permit will be required prior to any grading on the site.

17.100.310 - REQUIRED IMPROVEMENTS

The following improvements shall be installed at no expense to the city, consistent with the design standards of Chapter 17.84, except as otherwise provided in relation to oversizing.

A. Drainage facilities

Jacoby Heights Subdivision Revised Narrative

Page 45 of 49

- B. Lot, street and perimeter monumentation
- C. Mailbox delivery units
- D. Sanitary sewers
- E. Sidewalks
- F. Street lights
- G. Street name signs
- H. Street trees
- I. Streets
- J. Traffic signs
- K. Underground communication lines, including broadband (fiber), telephone, and cable. Franchise agreements will dictate whether telephone and cable lines are required.
- L. Underground power lines
- M. Water distribution lines and fire hydrants

Response: All of the improvements specified in this section will be installed by the developer at no expense to the City of Sandy consistent with the design standards of Chapter 17.84 and applicable standards.

CHAPTER 17.102 - URBAN FORESTRY

17.102.20 - APPLICABILITY

This chapter applies only to properties within the Sandy Urban Growth Boundary that are greater than one acre including contiguous parcels under the same ownership.

A. General: No person shall cut, harvest, or remove trees 11 inches DBH or greater without first obtaining a permit and demonstrating compliance with this chapter.

- 1. As a condition of permit issuance, the applicant shall agree to implement required provisions of this chapter and to allow all inspections to be conducted.
- 2. Tree removal is subject to the provisions of Chapter 15.44, Erosion Control, Chapter 17.56, Hillside Development, and Chapter 17.60 Flood and Slope Hazard.

Response: The subject property contains 9.68 and the standards of this chapter are applicable to the proposed subdivision. The applicant intends removing the majority of the trees on the property to accommodate development of a residential subdivision. The proposed tree removal and protection plan have been designed in accordance with the standards of this chapter and the provisions of Chapters 15.44, 17.56, and 17.60 as applicable. As noted in a review of Chapter 17.92, Landscaping above, the applicant has submitted a request for interpretation regarding the applicability of this Chapter to the proposal.

17.102.50 - TREE RETENTION AND PROTECTION REQUIREMENTS

A. Tree Retention: The landowner is responsible for retention and protection of trees required to be retained as specified below:

Jacoby Heights Subdivision Revised Narrative

Page 46 of 49

- 1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous ownership.
- 2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.
- 3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.
- 4. If possible, at least two of the required trees per acre must be of conifer species.
- 5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements. Response: A Tree Inventory was conducted on the site by a professional Arborist who inventoried all trees six-inches and greater DBH within the FSH Overlay and all trees eight-inches and greater on the rest of the site as requested by City staff. This inventory included as Exhibit J lists all trees by number, species, condition, and recommended root protection zone. Sheets 2 and 3 of the plan set shows the location of all trees with trees proposed to be retained colored green. Sheet 2 also includes a portion of the tree inventory listing trees proposed to be retained. Sheet 3 shows these trees with the proposed building envelopes as the City requested.

The subject property contains 9.68 acres requiring retention of 29 trees, 11 inches and greater DBH (9.68 x 3 = 28.92 rounded up to 29.04 trees). The submitted plan indicates that 64 trees are proposed to be retained. Of these trees, 33 trees are 11-inches or greater and in "viable" condition as identified by the Arborist in compliance with the tree retention standards of this Chapter. Of the 33 trees 11-inches or greater, 66 percent (22) are conifer species (Western Red Cedar, Western Hemlock and Douglas fir) with the other 11 trees hardwood species (Red Alder and Bigleaf Maple) as is preferred by subsection (4) above. All of the trees proposed to be retained are located within the FSH Overlay at the back of Lots 19-21, and 27.

In addition to the 33 retained trees, 10 mostly conifer trees are proposed to be retained identified a "viable" but are less than 11 inches.

As indicated on the tree inventory, the site contains over 500 trees of varying sizes, species, and condition. The primary reason more trees cannot be retained can be understood best by viewing the Cut and Fill Evaluation included as Sheet 10 with this submittal. As shown on this plan, due to topographic variability of the site, the majority of the site, with the exception of the FSH Overlay area, will need to be cut or filled to make it suitable for road, infrastructure, and home construction.

Jacoby Heights Subdivision Revised Narrative

Page 47 of 49

- B. Tree Protection Area: Except as otherwise determined by the Planning Director, all tree protection measures set forth in this section shall be instituted prior to any development activities and removed only after completion of all construction activity. Tree protection measures are required for land disturbing activities including but not limited to tree removal, clearing, grading, excavation, or demolition work.
 - 1. Trees identified for retention shall be marked with yellow flagging tape and protected by protective barrier fencing placed no less than 10 horizontal feet from the outside edge of the trunk.
 - 2. Required fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade.
 - 3. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. *Response:* The requirements of this section will be complied with prior to any grading or tree removal on the site.

17.102.60 - TREE REPLANTING REQUIREMENTS

- 1. All areas with exposed soils resulting from tree removal shall be replanted with a ground cover of native species within 30 days of harvest during the active growing season, or by June 1st of the following spring.
- 2. All areas with exposed soils resulting from tree removal occurring between October 1 and March 31 shall also be covered with straw to minimize erosion.
- 3. Removal of hazard trees as defined shall be replanted with two native trees of quality nursery stock for every tree removed.
- 4. Tree Removal allowed within the FSH Overlay District shall be replanted with two native trees of quality nursery stock for every tree removed.
- 5. Tree Removal not associated with a development plan must be replanted following the provisions of OAR Chapter 629, Division 610, Section 020-060 *Response: No trees are proposed to be replanted at this time.*

17.102.70 - VARIANCES

Response: The submitted plan is designed in compliance with the standards in this chapter and a variance to these standards is not requested or required.

CHAPTER 15.30 - DARK SKY ORDINANCE

15.30.000 - PURPOSE

The purpose of the Sandy Dark Sky Ordinance is to regulate outdoor lighting in order to reduce or prevent light pollution. This means to the extent reasonably possible the reduction or prevention of glare and light trespass, the conservation of energy, and promotion of safety and security. (Ord. 2002-11)

Jacoby Heights Subdivision Revised Narrative

Page 48 of 49

15.30.030 - EXEMPTIONS AND EXCEPTIONS

D. Full cutoff street lighting, which is part of a federal, state, or municipal installation.

15.30.060 - GENERAL STANDARDS

D. All outdoor lighting systems shall be designed and operated so that the area 10 feet beyond the property line of the premises receives no more than .25 (one quarter) of a foot-candle of light from the premises lighting system. **Response:** The applicant understands the requirements of this chapter. A detailed lighting plan will be submitted with construction plans following land use approval.

CONCLUSION

The proposed 32 lot subdivision as reviewed in this document complies with all relevant code standards with the exception of three sections. The applicant has submitted a narrative and the required fee for variances to the following sections:

- Section 17.82.20 to allow all homes adjacent to Jacoby Road to be constructed with their front doors facing the internal local street rather than Jacoby Road;
- Section 17.100.120(A) to allow creation of a single tier of lots (Lot 30) adjacent to Jacoby Road, a collector street; and
- Section 17.100.120(B) to exceed the 400 foot block length maximum for the north side of Woodstock St., from Camden Ct. to the eastern property line.

In addition, the applicant has requested the Planning Commission consider a Type III Code Interpretation regarding tree protection regulations applicable to the proposed subdivision. As reviewed in this document, the applicant requests the Planning Commission approve the application as submitted.

Jacoby Heights Subdivision Revised Narrative

Page 49 of 49





	7 4 111 1	<u> </u>	ODV	
PRESERV			ORY	
	DBH	CONDITION	ACTION	
BIG LEAF MAPLE	10	VIABLE	PRESERVE	
	12			
WESTERN HEMLOCK	14	VIABLE, WOUND ON TRUNK	PRESERVE	
	18			
WESTERN RED CEDAR	18	VIABLE	PRESERVE	
WESTERN RED CEDAR	42	BASAL DECAY, TRUNK DECAY FOR 30 FT	PRESERVE	
WESTERN RED CEDAR	15		PRESERVE	
WESTERN RED CEDAR	8	525		
WESTERN RED CEDAR	12	VIABLE	PRESERVE	
WEGTERN RED CEDAR	9		PRESERVE	
WESTERN RED CEDAR	3	VIABLE VIABLE, POOR TRUNK	PREJERVE	
WESTERN RED CEDAR	12	POTENTIAL	PRESERVE	
DOUGLAS FIR	9	TAPER, WINDTHROW POTENTIAL	PRESERVE	
DOUGLAS FIR	6	VIABLE, SUPPRESSED	PRESERVE	
DOUGLAS FIR	15	VIABLE, POOR TRUNK TAPER	PRE9ER∨E	
WESTERN HEMLOCK	13	20-FT TRUNK SCAR, POOR RESPONSE GROWTH,	PRESERVE	
WESTERN HEMLOCK	8	35-FT TRUNK SCAR,	PRESERVE	
WESTERN RED CEDAR	п	VIABLE	PRESERVE	
WESTERN HEMLOCK	10	VIABLE, MINOR MECHANICAL DAMAGE TO	PRESERVE	
	16	TO TRUNK VIABLE, GROWS FROM	PRESERVE	
	8	NURSE STUMP LISTED, HUNG UP IN TREE		
RED ALDER	10	541 VIABLE, TRUNK SWOOP	PRESERVE	
WESTERN HEMLOCK	12	VIABLE	PRESERVE	
DOUGLAS FIR	8			
DOUGLAS FIR	25	VIABLE	PRESERVE	
DOUGLAS FIR	18		PRESERVE	
RED ALDER	ГІ	CORRECTED, EXCESSIVE LEAN	PRESERVE	
RED ALDER	14	VIABLE, ON #		
WESTERN RED CEDAR	21	TRUNK DECAY	PRESERVE	
	22			
WESTERN RED CEDAR	13		PRESERVE	
DEAD	5	DEAD	PRESERVE	
RED ALDER	6 15		PRESERVE PRESERVE	
DOUGLAS FIR	5	SUPPRESSED, UNDERSIZE	PRESERVE	
	11	VIABLE, AT CREEK VIABLE, SOME TRUNK	PRESERVE	
WESTERN RED CEDAR	50			
RED ALDER	9	EXCESSIVE LEAN, TRUNK	PRESERVE	
RED ALDER	8	TERMINAL DECLINE	PRESERVE	
BIGLEAF MAPLE	24	VIABLE		
RED ALDER	12		PRESERVE PRESERVE	
RED ALDER	12	VIABLE, LEANS TOWARDS SITE	PRESERVE	
BIGLEAF MAPLE	П	VIABLE	PRESERVE	
BIGLEAF MAPLE WESTERN HEMLOCK	5	UNDERSIZE, STUMP SUCKER VIABLE, UNDERSIZE	PRESERVE PRESERVE	
WESTERN RED CEDAR	27	WOUND ON TRUNK, TRUNK	PRESERVE	
WESTERN RED CEDAR	15	WOUND ON BASE, WOOD	PRESERVE	
WEGTERN RED CEDAR	43	LARGE TRUNK SWOOP FROM BASE, BROKEN STEM, LOW VIGOR	PRESERVE	
WESTERN RED CEDAR	9	VIABLE	PRESERVE	
WESTERN RED CEDAR	9	VIABLE BROKEN STEM. TRUNK	PRESERVE	
BIGLEAF MAPLE	32	DECAY, RE-TRENCHED	PRESERVE	
	22	ARBORIST	PRESERVE	
RED ALDER BIGLEAF MAPI F	- 15	VIABLE, LISTED AND SELF-CORRECTED ADDED	PRESERVE	
RED ALDER BIGLEAF MAPLE	15	VIABLE, LISTED AND SELF-CORRECTED, ADDED BY ARBORIST VIABLE, ON 12, GROWS ON NURSE STUMP, BASE IS 4.5	PRE3ERVE	
RED ALDER BIGLEAF MAPLE WESTERN RED CEDAR	15	VIABLE, LISTED AND SELF-CORRECTED, ADDED BY ARBORIST VIABLE, ON ¢, GROWS ON NURSE STUMP, BASE IS 4.5 FT HIGH, ADDED BY ARBORIST	PRE3ERVE PRE3ERVE	

NOTE: INDICATES TREES II" DBH AND GREATER DEEMED TO BE VIABLE BY ARBORIST INSPECTION.

TOTAL NUMBER OF VIABLE, 11"+ DBH TREES TO BE PRESERVED: 33

PROJECT: Decoder PROJECT: Decoder JACOBY HEIGHTS SUBDIVISION Scale JACOBY HEIGHTS SUBDIVISION Date: JACOBY HEIGHTS SUBDIVISION Date: DATE: De/OI/2018 EXISTING CONDITIONS & TREE RETENTION INVENTORY ICATION MALE TREE RETENTION INVENTORY DESCION MALE TREE RETENTION INVENTORY DESCION MALE DESCION	I B	H	S	ынее 2	T) •	
PROJECT. DECLET. DECLET. DATE: DATE: DATE: DATE: DATE: DATE: DATE: DATE DATE <t< th=""><th>REVISION</th><td></td><td>TH T</td><td>OF H1</td><td>1</td><td>Ο</td></t<>	REVISION		TH T	OF H1	1	Ο
PROJECT DECALE DECALE </th <th>DATE NO. – –</th> <td></td> <td>DESIGNED: C</td> <td>DRAWN: C</td> <td>CHECKED: R</td> <td>APPROVED: R</td>	DATE NO. – –		DESIGNED: C	DRAWN: C	CHECKED: R	APPROVED: R
PROJECT DECL SCALE VERT: NA JACOBY HEIGHTS SUBDIVISION DATE: 06/01/2018 LEXISTING CONDITIONS & DATE: 06/01/2018 TREE RETENTION INVENTORY ELE: 11-199-FLANNING.dwg LOCATION 19124 SECTION NO 16/01 LOCATION 19124 SECTION NO 16/01 LOCATION 19124 SECTION 100 100	EST NOINE	Tar and hear in the	V (ORECON	AT NOOR	DENEWAL DATE: 12/31 /2018	NENEWAL DATE. 12/01/2010
PROJECT. DECALE SCALE HORI JACOBY HEIGHTS SUBDIVISION DATE: 06 DATE: EXISTING CONDITIONS & TREE RETENTION INVENTORY LOCATION 19124 SE JACOBY RD,SANDY, OR 97055 24	Z: 1"=40"	ANNING.dwg	٦L	P. RANGE	Ť	с 4 П
> PROJECT > JACOBY HEIGHTS SUBDIVISION LUCATION LOCATION BI24 SE JACOBY RD, SANDY, OR 97055	SCALE VERT	иліс. 00 ILE: 11-199-РL	LEG/	SECTION TW		27 72
		EXISTING CONDITIONS &	TREE RETENTION INVENTO			19124 SE JACOBI HU'SANDI, OH 9/055
		CORY KNIGHT	0153 5E ORCHARU VIEW LANE DAM45CUS, OR 97089			





-SURVEYED TOP OF BANK



	Slop	es Table	
Number	Minimum Slope	Maximum Slope	Color
1	0.00%	10.00%	
2	10.01%	24.99%	
3	25.00%	Vertical	



AREAS SHOWN ABOVE ARE LIMITED TO FSH WITHIN THE PROJECT BOUNDARY









		_ _		<u></u>				
, ₩			6					
REVISION			Т	OF	1	0		
<u>ب</u> ۲				CTH	RLM	RLM		
DATE NO.				AWN:	ECKED:	PROVED:		
	LINA SONA	Joon /						
CRED PRO	SCH GINE		DX (precon	1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1				
N/A "=40'	'2018	AING.dwg		RANGE	ļ	44		
VERT: HORIZ:	06/01	39-PLAN	LEGAL	TWP.	;	<u>۲</u>		
SCALE	DATE:	FILE: 17-19		SECTION		24		
						19124 SE JACOBY HD, SANDY, OH 9/05		
	County	CNIGHT Surveyors & Planners, Inc.		P.O. Box 955 Sandy, OR 97055	Fax: (503) 668-4730	DATE OF PLOT 12/4/2018		
		ЯY	HOH U U U U U U U					

GREEN VASE ZELKOVA - SE JACOBY RD GINKO HALKA - NEWTON ST

STREET TREE LEGEND

- VILLAGE GREEN ZELKOVA MIDDLEBURY AVE
- PATMORE ASH WOODSTOCK ST (SOUTH)
- AMERICAN HOPHORNBEAM WOODSTOCK ST (NORTH)
- PAPERBARK MAPLE CAMDEN CT

















				- B/	SHE	ET
RESERV	ATIC	ON INVENT	ORY	SION	OF	10
SPECIES	DBH	CONDITION	ACTION	REVI		
LEAF MAPLE	12	VIABLE	PRESERVE PRESERVE			ΣΣ
OUGLAS FIR	12 9	VIABLE	PRESERVE PRESERVE			
	14	VIABLE, WOUND ON TRUNK	PRESERVE	21		ä
ERN HEMLOCK	11	VIABLE VIABLE	PRESERVE	DATE	ESIGN RAWN:	FECKE
RN RED CEDAR	18	BASAL DECAY, TRUNK	PRESERVE			
RN RED CEDAR	15	DECAY FOR 30 FT VIABLE	PRESERVE	(SIONA)	12 M	/31/20
RN RED CEDAR	8	LISTED, HUNG UP IN TREE 525	PRESERVE	PROF	LOON BEEN	E: 12,
RN RED CEDAR	12	VIABLE	PRESERVE PRESERVE		A Star	AL DAT
N RED CEDAR	9		PRESERVE	1305		RENEW
IN RED CEDAR	12	VIABLE, POOR TRUNK TAPER, WINDTHROW	PRESERVE		Бл т	
		POTENTIAL VIABLE, POOR TRUNK		018 018	NG.d	#
	9	TAPER, WINDTHROW POTENTIAL	PRESERVE	01/2(ANNI	
UGLAS FIR	15	VIABLE, SUPPRESSED	PRESERVE	<u>/ERT:</u> -loriz	-EGA	25
RN HEMLOCK	13	20-FT TRUNK SCAR, POOR RESPONSE GROUTU	PREAEDVE		NO NO	
	8	WINDTHROW POTENTIAL 35-FT TRUNK SCAR,		CALE	LE: I	24
N RED CEDAR	17	TERMINAL DECLINE	PRESERVE	S S	<u> </u>	
RN HEMLOCK	10	VIABLE, MINOR MECHANICAL DAMAGE TO	PRESERVE			
RN HEMLOCK	16	VIABLE, GROWS FROM	PRESERVE			
UGLAS FIR	8	LISTED, HUNG UP IN TREE 541	PRESERVE		\succ	
D ALDER	10	VIABLE, TRUNK SWOOP	PRESERVE		₩ Å	
IGLAS FIR	8	VIABLE	PRESERVE PRESERVE		50	52
GLAS FIR	9		PRESERVE			970
GLAS FIR	18	VIABLE	PRESERVE		Ĕ≩	Ë
D ALDER	п	LISTED, AND SELF- CORRECTED, EXCESSIVE LEAN	PRESERVE	တ		Υ, Ο
ALDER	14	VIABLE, ON R	PRESERVE	ျပ	66	AND
RED CEDAR	21	TRUNK DECAY	PRESERVE	1 1 5	ΩĔ	D,S,C
RED CEDAR	22 25			0	心町	ΥR
RED CEDAR	13	VIABLE	PRESERVE		「二」	Ö
D ALDER	6	VIABLE, UNDERSIZE	PRESERVE		5 CK	NAC
JGLAS FIR	15 5	VIABLE SUPPRESSED, UNDERSIZE			山田	щ
D ALDER	JI.	VIABLE, AT CREEK	PRESERVE		Ω	24
N RED CEDAR	33	DECAY, AT CREEK	PRESERVE	U I Ŭ		þ
D ALDER	9	EXCESSIVE LEAN, TRUNK BOILS 60 ET TO SOUTH	PRESERVE	Ă		
	8	TERMINAL DECLINE	PRESERVE			žo
D ALDER	24 10	VIABLE	PRESERVE PRESERVE	PRO		OCA
D ALDER	12		PRESERVE	N		
EAF MAPLE	12	SITE	PRESERVE		*	
EAF MAPLE	5	UNDERSIZE, STUMP SUCKER	PRESERVE		ind 97055	
N RED CEDAR	7	WABLE, UNDERSIZE	PRESERVE		ing a OR 9	-4/30
N RED CEDAR	15	WOUND ON BASE, WOOD	PRESERVE		lanni jineer andy, () 668	9 668-
N RED CEDAR	43	LARGE TRUNK SWOOP FROM BASE, BROKEN STEM. LOW	PRESERVE	× {	1 Eng 55 Sc (503	4/2018
N RED CEDAR	9	VIGOR	PRESERVE		Civi Civi tox 9	-121
N RED CEDAR	9		PRESERVE		Su Su	PLOT
	32	DECAY, RE-TRENCHED	PRESERVE	No. Contraction		TE OF
ALDER	22	ARBORIST VIABLE, LISTED AND	PRESERVE		~	DA
EAF MAPLE	15	SELF-CORRECTED, ADDED BY ARBORIST	PRESERVE		¥	
RED CEDAR	13	VIABLE, ON 12, GROUS ON NURSE STUMP, BASE 19 4.5 FT HIGH, ADDED BY	PRESERVE	-	99 89	
RED CEDAR	8	ARBORIST VIABLE	PRESERVE		DE COLOR	
3 II" DBH AND GRE ORIST INSPECTION. DBH TREES TO BE	ATER DEEM	ED TO BE D: 33			CORT KNIC 3 SE ORCHARD DAMASCUS, OR	















EXHIBIT D

REQUEST FOR CODE INTERPRETATION Jacoby Heights Subdivision File No. 18-025

Request: The applicant requests a Planning Commission code interpretation to determine the applicable code section/chapter regarding tree retention in residential subdivisions. With the current application, Planning Department staff has required the applicant to address the requirements of both Chapter 17.92, Landscaping & Screening, General Standards All Zones specifically Section 17.92.10, General Provisions, subsection (C) and the requirements of Chapter 17.102, Urban Forestry Ordinance. It is the applicant's belief that only the requirements of Chapter 17.102, Urban Forestry Ordinance are applicable to regulating trees on the subject property. The requested interpretation is relevant not only to the proposed Jacoby Heights Subdivision application but may also to all similarly zoned properties in the City of Sandy.

Introduction: The purpose of this document is to show that the requirements of Section 17.92.10(C) are not intended to be used to regulate tree retention on the subject property. The document starts by reviewing the legislative history leading up to the adoption of these code chapters and it then analyzes the code language of these chapters in detail.

Legislative History: Chapter 17.92 was adopted by the City Council as part of an overhaul of the Development Code on October 20, 1997 by Ordinance 9-97. The regulations of Chapter 17.102 were developed after an extensive public involvement process that involved work by an advisory committee. This code was adopted by Ordinance 2002-10 on November 18, 2002, about four years after work on this code had begun. Because Chapter 17.92 was adopted and in place prior to the start of any work on Chapter 17.102, it is clear that these regulations should not be used to regulate trees in residential subdivisions. If Section 17.92.10(C) was applicable there would have been no reason to develop new regulations.

A review of the legislative history provides valuable insight into understanding the intent and applicability of these adopted regulations. The legislative history of Chapter 17.102 is especially revealing in that this chapter was developed over several years before adoption in 2002. The following documents relevant to this discussion were gleaned from the City's online archive system.

- Attachment 1 Council Goals 1997-98 "Adopt a Tree Ordinance/Urban Forestry Program".
- Attachment 2 March 15, 1999 City Council appoints Urban Forestry Committee to develop tree regulations.
- Attachment 3 October 30, 2000 Council Minutes CC reviews draft Urban Forestry Ordinance.
- Attachment 4 Council Goals 2001 "Adopt an Urban Forestry Ordinance".
- Attachment 5 June 17, 2002 Lazenby (City Manager) memo to PC regarding "Tree Ordinance".
- Attachment 6 PC Staff Report, August 26, 2002
- Attachment 7 Other Code Amendments

Jacoby Heights Code Interpretation

Page 1 of 6

- Attachment 8 November 4, 2002 Council Minutes public hearing on revised draft Urban Forestry Ordinance.
- Attachment 9 Ordinance 2002-10 Findings of Fact
- Attachment 10 Pre-application Notes and Email from James Cramer Regarding Tree Inventory Required.

Conclusion of Legislative History Review: It is clear after reviewing the various documents produced after Chapter 17.92 had already been adopted, that the reason the City Council directed Planning staff to develop the tree protection regulations in Chapter 17.102, was because adequate regulations did not already exist. Since the regulations in Chapter 17.92, were adopted prior to this date (1997), it is reasonable to conclude that everyone involved (staff, public, Planning Commission, and City Council) believed the language in Chapter 17.92 did not apply to residential subdivisions. This conclusion is supported by the Findings adopted with Ordinance 2002-10, Council goals, meeting minutes, and various other documents listed above. The "Additional Code Amendments" document (Attachment 7), although never adopted, provides additional evidence supporting this conclusion. This amendment added clarification that retained trees required by Chapter 17.102 could be counted towards required landscaping. This language would not have been included if Chapter 17.92 was intended to regulate tree retention.

Comparison of Chapters 17.92 and 17.102: A review of the pure language in these regulations also supports the conclusion that the language in Chapter 17.92 should not be used to regulate tree retention in residential subdivisions. The following section starts by comparing the intent sections of these chapters and then evaluates specific regulatory language.

Chapter 17.92: The title of this chapter, "Landscaping & Screening, General Standards - All Zones" speaks to the intent of this chapter. Section 17.92.00, Intent, includes the following language:

"The City of Sandy recognizes the <u>aesthetic and economic value of landscaping</u> and <u>encourages its use</u> to establish a pleasant community character, unify developments, and <u>buffer or screen unsightly features</u>; to <u>soften and buffer large scale structures</u> <u>and parking lots</u>; and to aid in energy conservation by providing shade from the sun and shelter from the wind. The community desires and intends all properties to be landscaped and maintained."

"<u>This chapter prescribes standards for landscaping, buffering, and screening</u>. While this chapter provides standards for frequently encountered development situations, detailed planting plans and irrigation system designs, when required, shall be reviewed by the City with this purposes clause as the guiding principle."

The following terms are relevant when evaluating the intent of this chapter:

Jacoby Heights Code Interpretation

Page 2 of 6
Landscaping: The arrangement of trees, grass, bushes, shrubs, flowers, gardens, fountains, patios, decks, outdoor furniture, and paving materials in a yard space. It does not include the placing or installation of artificial plant materials. (from SDC) Buffer - A combination of physical space and vertical elements, such as plants, berms, fences or walls, designed to provide space or distance, obstruct undesirable views, serve as an acoustic barrier, generally reduce impacts of adjacent development, or separate and screen incompatible land uses from each other. (from SDC)

Buffer: A combination of physical space and vertical elements, such as plants, berms, fences or walls, designed to provide space or distance, obstruct undesirable views, serve as an acoustic barrier, generally reduce impacts of adjacent development, or separate and screen incompatible land uses from each other.

Screen: No definition found in SDC. Websters Dictionary defines a screen as, "something that shelters, protects, or hides: such as: a growth or stand of trees, shrubs, or plants."

Even though the title of this chapter includes the words "All Zones" the language in the intent statement speaks to landscaping in the broad use of the term and including items typically considered landscaping: trees, bushes, grass, etc., arranged in a development or yard. The examples used in the intent statement include: "<u>buffering and screening</u> <u>unsightly features</u>" and "<u>softening and screening large scale structures and parking</u> <u>lots</u>". These examples are relevant to multi-family, commercial, and industrial projects but do not apply to tree retention requirements in residential subdivisions.

Chapter 17.102, Urban Forestry: In contrast, the intent of Chapter 17.102 speaks to regulating tree retention and the benefits trees provide.

- A. This chapter is intended to <u>conserve and replenish the ecological</u>, <u>aesthetic and</u> <u>economic benefits of urban forests</u>, by regulating tree removal on properties <u>greater than one acre within the Sandy Urban Growth Boundary</u>.
- B. This chapter is intended to facilitate planned urban development as prescribed by the Sandy Comprehensive Plan, through the appropriate location of harvest areas, landing and yarding areas, roads and drainage facilities.
- C. <u>This chapter shall be construed in a manner consistent with Chapter 17.60 Flood</u> <u>and Slope Hazard Overlay District. In cases of conflict, Chapter 17.60 shall</u> <u>prevail</u>.

Section 17.92.10(C): The language Planning staff has recently been using in addition to Chapter 17.102 to regulate tree retention, is found in this section.

C. "Significant plant and tree specimens <u>should</u> be preserved to the greatest extent practicable and integrated into the design of a development. Trees of 25-inches or greater circumference measured at a height of 4-1/2 ft. above grade are considered significant. Plants to be saved and methods of protection shall be

Jacoby Heights Code Interpretation

Page 3 of 6

indicated on the detailed planting plan submitted for approval. Existing trees may be considered preserved if no cutting, filling, or compaction of the soil takes place between the trunk of the tree and the area 5-ft. outside the tree's drip line. Trees to be retained shall be protected from damage during construction by a construction fence located 5 ft. outside the dripline."

The first sentence of this section states: "Significant plant and tree specimens <u>should</u> be preserved to the greatest extent practicable ...". There are two problems with using the language in this sentence. The first is use of the word "should". This word when used in regulatory code is permissive and aspirational and generally denotes a guideline or recommendation. This is opposed to the word "shall" which conveys a mandatory requirement. The second problem with this section is the use of the phrase, "preserved to the greatest extent practicable". This phase is unclear and is subject to interpretation as to what is required. Who is responsible for defining "greatest extent practicable"?

A comparison of the language in Chapter 17.102 with Section 17.92.10(C) reveals a number of differences between these two sections. Where the regulations in Chapter 17.102 only apply to properties greater than one acre and defines trees as 11-inches diameter or greater, Section 17.92.10(C) doesn't contain a property size limitation and defines trees as 25-inches or greater circumference (7.9 inches). Chapter 17.102 specifically allows retained trees to be located anywhere on the site at the landowner's discretion while Section 17.92.10(C) requires trees to be integrated into the design of the development. In addition to these differences, the protection requirements for retained trees are considerable more restrictive in Chapter 17.92 than the regulations in Chapter 17.102. A table comparing the differences between these regulations is included below.

Chapter 17.102 - Urban Forestry	Chapter 17.92 - Landscaping & Screening
Applicable to properties greater than one acre.	No property size specified.
Shall retain three trees/acre.	Trees <u>should</u> be preserved to the greatest extent practicable.
Retained trees can be located <u>anywhere</u> on the site at the landowner's discretion before the harvest begins.	integrated into the design of a development.
Trees are defined as 11-inches or greater.	Trees are defined as 25-inches or greater circumference.
Trees proposed for retention shall be healthy.	No tree condition requirement.
At least two of the three retained trees/acre must be conifer species, if possible.	No species preference.
Retained trees shall be protected by fencing 10 horizontal feet from the outside edge of the trunk.	Retained trees shall be protected by a constructed fence five feet outside the drip line.

Comparing Code Requirements

Jacoby Heights Code Interpretation

Page 4 of 6

Conclusion of Code Language Review: As reviewed in this section, the language in Section 17.92.10(C) encourages tree protection ("should be preserved") rather than requiring protection ("shall retain three trees/acre") as specified in Chapter 17.102. Where Chapter 17.92 speaks to landscaping and screening, Chapter 17.102 speaks to the benefits of the urban forest. Where the standards in Chapter 17.102 are prescriptive (three trees per acre on properties greater than one acre) the language in Section 17.92.10(C) is permissive, aspirational, and subjective ("tree specimens should be preserved to the greatest extent practicable"). Even if the language in Chapter 17.92 was found to apply to residential subdivisions, this language encourages but does not require trees retention.

Summary: The applicant requests the Planning Commission consider this request for a Code Interpretation to determine the applicable Code language regulating tree retention for the proposed residential subdivision. As discussed in this document, the applicant believes City staff is incorrectly applying the language in Section 17.92.10(C) to residential subdivisions. The basis for this conclusion includes the following:

- 1. Chapter 17.102 was adopted four years after Chapter 17.92 was adopted.
- 2. The record shows that everyone involved in the development and adoption of the tree regulations in Chapter 17.102 believed regulations did not already exist in the Code to protect trees in residential subdivisions.
- 3. The intent of Chapter 17.92 speaks to landscaping, buffering, and screening, aspects that are generally not relevant at the subdivision level. These requirements are generally only applicable to commercial, industrial, and multi-family projects.
- 4. The intent of Chapter 17.102 speaks to regulating tree retention and the benefits trees provide.
- 5. The language in Section 17.92.10(C) conflicts with the tree retention standards in Chapter 17.102, Urban Forestry.
- 6. The tree retention language in 17.92.10(C) is subjective and does not specify a mandatory requirement. The use of the word "should" denotes a recommendation or guideline and the phrase, "to the greatest extent practicable" is subject to interpretation.
- 7. The language in Chapter 17.102 is clear and objective and mandates a tree retention requirement.

Why is this Important?

 If Section 17.92.10(C) is applied to subdivisions, the phrase "Significant plant and tree specimens <u>should</u> be preserved to the greatest extent practicable..." will make it virtually impossible to layout out a subdivision on a heavily wooded site. It will be

Jacoby Heights Code Interpretation

Page 5 of 6

impossible to meet required street spacing, lot densities, street grades and numerous other design requirements that control subdivision layouts, all while saving trees to the "greatest extent practicable".

- The language in Chapter 17.102 is clear and objective and contains mandatory tree retention standards while the language in Section 17.92.10(C) is subjective and leaves significant room for interpretation. The use of the word "should" in Section 17.92.10(C) also denotes that tree retention is a guideline or recommendation while the language in Chapter 17.102 is mandatory.
- The difference between the tree retention requirements of Chapter 17.102 and those in Section 17.92.10(C) can be significant. Where Chapter 17.102 clearly requires three trees 11-inches or greater/acre to be retained anywhere on the site at the owner's discretion, Section 17.92.10(C) could be interpreted to require all trees 8-inches or larger to be retained ("to the greatest extent practicable").
- Using two different chapters with conflicting requirements is confusing to developers and to staff. As included in Attachment 10, the pre-application conference notes prepared by staff (January 3, 2018), specify that trees 11-inches and greater (six inches within the FSH Overlay) shall be inventoried. This standard is stated again in a email to the applicant by city staff on February 12, 2018. It was only after the applicant paid an Arborist to complete this inventory and it was included with the original application package that staff then told the applicant in the incompleteness letter (dated July 12, 2018) that all trees 8-inches and greater needed to be inventoried. The additional cost to the applicant to have an Arborist return to the site and inventory these additional trees was about \$3,100.

Jacoby Heights Code Interpretation

Page 6 of 6

CITY COUNCIL 19976-987 GOAL STATEMENT

1. Improve the city's infrastructure to meet current standards and provide for managed future growth.

This Year

Secure a permit for future sewer plant improvements and operations, including an outfall on the Sandy River.

Develop standards and funding for a Neighborhood Traffic Management Program, including "traffic calming" measures.

Complete Tickle Creek-local-improvement district. Develop a Stormwater Management Plan.

Adopt a Wastewater Pretreatment Ordinance

<u>Complete the Acquire the state forestry site for Vista Loop reservoir expansion.</u>

Begin planning for a Salmon River water plant and Sandy River wastewater outfall.

Next Two Years

Find a stable funding source for street maintenance.

Pursue funding to implement alternative transportation methods (e.g., bike/ped paths, routes, maps; improved bus service).

Explore a Sidewalk Repair Program.

Long Range

Pursue options for alleviating Highway 26 traffic congestion, including a bypass, and safe pedestrian crossings in downtown Sandy.

2. Manage growth pressures through careful planning.

This Year

Revise the Comprehensive Plan to reflect the Sandy 2040 plan.

D:\MSOFFICE\WINWORD\POLICIES\GOALS98A.DOC

Develop an Economic Development Strategy

Adopt the Downtown Plan.

Adopt development regulations for dedication of parks and other neighborhood amenities.

Adopt a Tree Ordinance/Urban Forestry Program.

Next Two Years

Begin implementing the downtown plan.

3. Build the city's tax base by supporting and expanding the business community.

This Year

Build communication channels between the city and businesses.

Work with the Chamber of Commerce in marketing the city as a place to live and work.

Support business efforts to improve the appearance and effectiveness of the downtown core.

Encourage the formation of a Downtown Development Organization.

Attract a movie theater.

In conjunction with businesses and land owners, develop a downtown parking plan

4. Build on the city's superb quality of life.

This Year

Encourage growth of neighborhood associations and neighborhood block watch. <u>Provide</u> more outreach by the City Council to neighborhoods.

Finish Tupper Park.

Coordinate, improve, and publicize recreational programs and facilities.

Adopt a parks and recreation master plan.

D:\MSOFFICE\WINWORD\POLICIES\GOALS98A.DOC

ATTACHMENT 2 COUNCIL MINUTES Regular Meeting March 15, 1999

Mayor Malone called the regular meeting of the City Council to order at 7:03 p.m., on March 15, 1999, in the Council Chambers at City Hall.

Roll Call Mayor Malone - present Councilor Smith - present Councilor Topliff - present Council President Bardon - present

Councilor Hammons - present Councilor Allen - present Councilor Kennedy - excused

<u>Shadow Council</u> Sunil Loprinzi - present Jon Liepold - absent Tessa Brookhart - absent Tyson Bell - present

Casey Butler - present Morgan Tigli - present Laura Pollard - absent

Also attending were Scott Lazenby, City Manager, Department Directors, Sue Barker, Recording Secretary and interested citizens.

Changes to Agenda

City Manager Lazenby included two items under new business: 1) Discussion on issues regarding the sewer plant close out 2) Review of a request to serve alcohol at the Senior Center.

Public Comment

<u>Consent Calendar</u> The items on the consent calendar included:

- A. Approval of Council Minutes for March 1, 1999.
- B. Approval of Liquor License Renewal for El Gallo Restaurant
- C. Approval of Resolution No. 99-04, A Resolution Setting Forth Corrective Measures for Deficiencies Disclosed in the FY97-98 Audit Report

Motion Councilor Allen moved, seconded by Councilor Smith and approved by unanimous voice vote:

To Approve the Consent Calendar as Submitted.

Action Items

the control and the second block of a local second matter where

Appointment to Urban Forestry Committee Mayor Malone thanked citizens for their response to serve on the committee. She reported there were 20 applicants.

Council President Bardon supported appointment of Margaret Holman and any members who served on the prior committee.

Page 2

Councilor Topliff also supported appointment of prior members along with Steel and John Belton.

Mayor Malone also added Bill O'Brion to the list.

Councilor Hammons recommended appointment of all applicants.

Mayor Malone felt that if all applicants were appointed, the committee would not be balanced. She also preferred a smaller committee.

Councilor Allen suggested appointing the seven other appointments as alternates to the committee.

Mayor Malone mentioned she would appoint alternates in case some of the other committee members decide to drop out.

Mayor Malone appointed the following to the Urban Forestry Committee: Brad Picking, Susan Drew, Kevin Frostad, Margaret Holman, Bob Skipper, Bill Trimble, Dan Loftis, Bob Rockwell, Jerry Carlson, John Belton, Elinor Majeski, Steel and William O'Brion. Bob Skipper will be Chairman the committee and Susan Drew will serve as Vice Chair.

Alternates: Dorothy Douglas, Jerry Lawson, Nita Stamm, Dick Sleeper, Mort Spence, and Sara Hicks.

Appointment to the Transit Committee

Mayor Malone reported that Bud Fullen asked to be on the committee. Mayor Malone appointed the following: Sherry Landauer, Mt. Hood National Forest, Gary Boley, John Charles, Janet Pitts, Bud Pullen, Caren Topliff, Don Allen and Roberta Kennedy. She also noted the committee is still looking for a business representative.

Council Policy Discussion - None

<u>Public Hearing</u> - None <u>Ordinance No. 99-02</u> - Assigning Zoning Designation

Mayor Malone called for the second reading of Ordinance No. 99-02 by title only.

COUNCIL MINUTES Special Meeting October 30, 2000

Following the pledge of allegiance, Mayor Malone called the regular meeting of the City Council to order at 7:02 p.m., on October 16, 2000, in the Council Chambers at City Hall.

Roll CallMayor Malone - presentCouncilor Hammons - excusedCouncilor Smith - presentCouncilor Allen - presentCouncilor Topliff - presentCouncilor Kennedy - presentCouncil President Bardon - presentFresent

Also attending were Heather Brinton, City Attorney, City Manager, Department Directors, Scott Lazenby, Recording Secretary and interested citizens.

Changes to Agenda

Public Comment

Action Items

Council Policy Discussion - None

New Business

Draft Tree Ordinance and Urban Forestry

Kevin Liburdy, Assistant City Planner, explained that the group consisted of four subcommittees: Undeveloped Lands Developed Lands Street Trees Heritage Committee

City Manager Lazenby reported that the proposal includes a new Urban Forestry Chapter 17.102 has been added to the development code. He noted that Urban Forestry Chapter would deal with both developed and undeveloped lands. He mentioned that Heritage Trees fit under the existing Goal 5 type issues under Chapter 17.62 and street trees are under Section 12.11.

Chapter 17.102 - Urban Forestry

City Manager Lazenby advised that this ordinance would come into effect when you have five or more trees six inches or greater in diameter being cut.

Bob Kallen, 39517 Scenic, asked for clarification on developed and undeveloped properties. City Manager Lazenby mentioned that the Urban Forestry ordinance really doesn't make a distinction. He noted that if you have a home with a lot of trees over six inches, you would still fall under this ordinance if you cut five or more trees.

Bill O'Brion, 18410 Meinig, noted that if you are on a singlefamily lot that may be considered underdeveloped because it could be sold to someone who could develop it to a greater density. He suggested having an exemption for single family residential that is not able to redevelop.

Kevin Frostad, 37270 Dubarko Drive, mentioned that he is pleased to see this in code language. He felt this is a very modest urban forestry ordinance. He noted this is a big step forward. He felt the two most important aspects are: (1) it provides for a retention requirement (25% of the basal area of the trees on a parcel) and (2) transfers the jurisdiction of urban forestry matters from the Oregon Dept. of Forestry to the City where we feel they belong. Mr. Frostad would support either version of the ordinance.

Margaret Holman, 39385 Idleman, mentioned that the Sandy Forest Practices Act will be a little more stringent than the State Forest Practices Act and noted there were four changes that were inadvertently omitted from the second draft.

1. - Page 44 - Road Maintenance (5) Amended to read "When applying water and/or environmental safe road oil or other surface stabilizing materials

City Manager Lazenby noted this changed would be placed under Section 17.102.40 (D) - Road Construction and Maintenance.

Councilor Smith questioned what is environmental safe road oil.

Bill O'Brion suggested wording it as read binders to minimize dus: and that would be environmental safe.

The Council agreed with that change.

- Page 45 -629-630-000 Purpose (3)

The purpose of the harvesting rules is to establish standards for forest practices that will maintain the productivity of forestland, minimize soil and debris entering waters of the state, and protect "water quality and quantity and " and wildlife and fish habitat. (inserting water quality and quantity"

- 3. Page 46 629-630-600(1) Felling, Removal of Slash
- (1) Operators shall not fell, buck and limb trees in ways that minimize disturbance to disturb channels, soils and retain vegetation in riparian management areas, stream, lakes and all wetlands greater than one quarter acre, and minimize shall avoid slash accumulations in channels, significant wetlands and lakes.

- 629-630-400(4) - Treatment of Waste Materials Operators shall dispose of remove from the site all other debris such as machine parts, old wire rope, and used tractor tracks so that such materials do not enter water of the state.

Councilor Smith noted that you want to say "remove and legally dispose of " ... and the Council agreed with this change.

Ms Holman questioned if state regulations override city regulations.

City Attorney Brinton advised that generally there is a state preemption of forestry regulations in Oregon which would preempt. However, there is an exemption when cities have adequately adopted their own regulations to deal with certain forest operations.

Bill O'Brion noted that Salem Oregon Forestry officials have reviewed this proposal and they were pleased with it. He also advised that Oregon Forestry officials would be available for expert assistance.

City Manager Lazenby advised that jurisdiction would shift to the city and there is still a role for DSL and other agencies.

City Manager Lazenby reported that a policy issue for the Council is whether or not to exempt anything with a single family home. He also noted that there is a commercial tree cutting exception in the draft if they are harvesting.

City Manager Lazenby presented scenarios and how this ordinance would apply to those situations.

Councilor Smith expressed concerns that the ordinance would prevent people from cutting trees that were planted for the purpose of harvesting.

Page 4

Bill O'Brion mentioned that the trade off is that if you decide not to develop but still want to cut trees, you need to sign off that you will not develop the property for several years. He noted the consultant kicked out this particular section.

He also noted that the committee was trying to leave placement of the trees being retained up to the landowner as long as they leave 25% basal area.

Bob Kallen has an acre of property with a grove of Douglas Fir averaging more than 20 inches. He noted his neighbors on Scenic would have the same situation.

Bob Rockwell, 40305 McCormick Court, mentioned that there were a number of people on the committee that did not favor taking property away from and people and regulating how many trees they can cut. He noted that the full committee did not review this proposal.

Barbara Sah, 16605 Bluff Road, mentioned that they have planted a lot of trees and opposed the regulations since it would be impossible to measure the number of trees.

Bill O'Brion noted that this ordinance would protect trees and avoid clear cutting trees when new development occurs.

Mayor Malone mentioned that 17.102.25 Exceptions would exempt commercial timber harvest outside the FSHR Overlay District.

City Manager Lazenby suggested taking out "properties enrolled in a forest property tax assessment program".

Mayor Malone suggested taking out (8) city-owned lands from exceptions.

Councilor Topliff suggested taking out (9) those associated with the establishment or alteration of any public park from exceptions.

Bill Knapp, P O Box 880, questioned 17.102.30 Urban Forest Conservation Requirements. He felt this ordinance would force him to cut his trees before the ordinance is adopted. He also questioned if a piece of property outside of FSH setback area, meets the criteria for forest conservation area?

Page 5

He also noted they have EFU (Exclusive Farm Use) ground and this is not handled the same as a Forest Practice Use. He would like to see an exception for farm use. He mentioned this is in addition to the park dedication requirement and could represent a taking issue.

City Manager Lazenby noted this would provide an incentive for retaining the trees in the riparian area since they would count towards the 25%. He mentioned that the city is hopeful this will apply in the Urban Growth Boundary.

Bill Knapp felt that the FSHR ordinance will take half his farmland and this ordinance will prevent him from using natural tree covers as a rotation crop.

City Manager Lazenby pointed out that the concern is with the farmer who sells his property to a developer and the trees are removed. He pointed out that the property can be cleared if it is not developed within ten years. He felt it should be easy to retain 25% of the trees.

Mayor Malone favored the farm exception similar to the Forest Practices Act.

Bob Skipper, 78800 SE Short Road, Gresham, expressed concerns that everyone should be treated the same with no exceptions. He felt if property owners have to set aside a percentage of their property there should be some benefit to the property owner. He noted that property owners with trees are being penalized. He also mentioned that ten years is too long of a time frame.

Bill O'Brion felt that if you use trees for a cover crop, it is a forest and would fall under number 10 of exceptions.

Bill O'Brion proposed some changes:

<u>17.102.20 - Procedures and Application Requirements</u> <u>C - Forest Thinning - Removal of Trees of less than 6 inches.</u> He felt this is not forest thinning but pre-commercial thinning since the trees are too small.

 On lots of greater than an acre, forest thinning shall be reviewed by the Planning Director.

He felt it should apply to all. He did not recommend flagging trees prior to removal, rather paint should be used to mark trees.

Page 6

He also mentioned that noticing neighboring property owners should be consistent.

F. Certified Arborist Review. He would like to see foresters included in this section.

<u>G. Agency Review</u> He noted that the City can request a third-party review.

I Enforcement

He would like to see a cap for the amount of damages. City Manager Lazenby advised the amount would be up to the judge.

17.102.30 Urban Forest Conservation Requirements

 <u>C. Criteria for Determining Forest Conservation Areas.</u>
 <u>1. The conservation area shall be part of a contiguous area</u> with at least 80% tree canopy coverage.

He felt this would be the Council choice whether or not to leave contiguous in but as a subcommittee they felt that if the trees were scattered, it would be the property owner's prerogative.

 If there is insufficient basal area within the riparian setback area to meeting the 25% retention standard, the forest conversation area shall include trees adjacent to the riparian setback area.

He did not like the word shall.

 The conservation area includes all land within five horizontal feet of the driplines of protected trees.

He would like to see it say "driplines of protected tree(s)" to pover significant trees.

F. Harvesting Period.

He completely disagreed with limiting to a time period during the year.

17.102.40 E. Division 630 Harvesting

He noted there will be situations when harvesting trees will make it necessary to violate some of things in this section. It would still have to go to the Planning Commission. By disallowing, you take away the ability for things are more beneficial to the environment.

City Manager Lazenby did not recommend regulating removal of four or fewer trees on developed property.

Councilor Topliff suggested removing "with surrounding soil heaving" from Section 17.102.25 Exceptions A(7).

Mayor Malone noted that a workshop will be set for the November 20¹⁰ meeting starting at 5:30 p.m.

Margaret Holman questioned how we can promote safety in the neighborhood when trees are being removed. She suggested that if a permit is required, the city could recommend a professional.

Mayor Malone suggested striking the section-requiring street trees where there are not currently street trees.

Bob Skipper mentioned that when permits are issued for existing commercial development, they should comply with the street tree requirements.

Glenda Woodcock, Evans Street, questioned who would maintain the trees. She suggested that there needs to be a program for long term care. She also noted that current codes are not currently being enforced.

Councilor Smith noted that Section 12.11.30 covers maintenance and it is the responsibility of the property owner.

Councilor Topliff spoke of a program with the Friends of Trees which would give group prices to purchase trees and they would help you plant trees.

A citizen also suggested a program for senior or handicap citizens to help maintain the trees.

Margaret Holman noted that the street sweeper will take down the address and report to the city any areas where the trees need to be trimmed.

Mayor Malone continued this matter to November 20, 2000 at 5:30 p.m.

Mayor Malone adjourned the meeting tat 9:30 p.m.

Linda K. Malone, Mayor

ATTEST:

Refine downtown design standards in the city's development code. Draft a downtown plan map as a development/planning tool.

Adopt an Urban Forestry ordinance.

Consider new standards for city street standards ("skinny streets").

3. Build the city's tax base by supporting and expanding the business community.

This Year

Encourage the development of a movie theater.

In conjunction with businesses and land owners, develop a downtown parking plan

Explore options to encourage telecommuting. Encourage home-based businesses.

Next 4. Build on the city's superb quality of life.

This Year

Encourage growth of neighborhood associations and neighborhood block watch. Provide more outreach by the City Council to neighborhoods.

Encourage the use of the Neighborhood Traffic Management program. Develop the library book and materials collection. Next Two Years Explore a connection between the Sandy bike/pedestrian trail system and the, Springwater and Estacada trails. ortinue Contplete Complete the development of balifields as a cooperative venture between the Oregon Trail School District. Ne Pursue the Sandy River part/trail loop plan Adopt a "Dark Skies" Ordinance. Long Range Explore the creation of a parks and recreation district. C:\My Documents\POLICIES\GOALS01.DOC

13	ATTACHMENT 5
City o	of Sandy Staff Report
DATE:	June 17, 2002
TO:	Planning Commission
FROM:	Scott Lazenby, City Manager
RE:	TREE ORDINANCE
ATTACI Developm	EMENTS: Chapter 17.102; Required Changes to other sections of the Sandy nent Code.
In March tree ordin preferenc but would that could	, the Planning Commission and City Council held a joint workshop on the draft nance. At that meeting, the majority of the members in attendance expressed e for an ordinance that would not prohibit property owners from removing trees, d provide regulations on how the trees are removed, and establish permit fees a be used as a source of revenue for an Urban Forestry Fund.
Subseque and now allowing proposal trees per attached.	nt to this workshop, Bill O'Brion, a member of the Urban Forestry Committee a City Council member, explored ways of preserving existing trees while still property owners a fair measure of flexibility and discretion. His compromise is based on the existing state forestry regulations, which require retention of two acre, for properties that are 25 acres or more in size. This draft ordinance is
<u>Key Feat</u>	ures of the Ordinance
• Tree ordination to ens	removal in the areas protected by the steep slope and buffer areas of the FSH ance are covered by the FSH ordinance (they are referred to in the tree ordinance are that it encompasses all land in the UGB).
 For an DBH 	reas of one acre or larger that can be further developed, three trees of at least 11 must be retained. At least half of these trees must be a conifer species.
• The accommutation fully for the retention of the second	creage threshold and retention requirement are based on contiguous areas under on ownership. For example, suppose a person owns a one-acre parcel that is orested, and an adjacent ten-acre parcel that is pastureland with no trees. The ion requirement would be 33 trees, based on the two contiguous parcels under me ownership.
the sa	







City of Sandy Staff Report

SUBJECT: 02-016 Urban Forestry

AGENDA DATE: August 26, 2002

DEPARTMENT: Planning & Development

STAFF CONTACT: Tracy Brown

Exhibits:

- A. Revised Chapter 17.102 (updated since July PC meeting)
- B. Other Proposed Amendments Related To Trees
- C. Written Testimony Friends of Tickle Creek (submitted 7/22/02)
- D. Written Testimony John Gardiner (submitted 7/22/02)

I. BACKGROUND

In March, the Planning Commission and City Council held a joint workshop on the draft tree ordinance. At that meeting, the majority of the members in attendance expressed preference for an ordinance that would not prohibit property owners from removing trees, but would provide regulations on how the trees are removed, and establish permit fees that could be used as a source of revenue for an Urban Forestry Fund.

The Planning Commission opened the public hearing on 7/22/02 and took testimony on the proposed code language. The Planning Commission then closed the public hearing and continued deliberations to the 8/26/02 meeting. Since that meeting, staff reviewed the ordinance language and made modifications to include separate sections for land inside the city limits and land outside the city limits but inside the Urban Growth boundary. The link to the revised language is listed above.

Ballot Measure 56 (ORS 227.160 et seq.) was adopted by the voters on November 3, 1998. This law requires the city to notify a property owner if proposed regulations will affect the permissible uses of property and may reduce the value of the property. The original notice was mailed to property owners within the city limits and an additional notice was mailed to property owners outside the city limits but inside the urban growth boundary.

To afford the property owners outside the city limits an opportunity comment on the proposed urban forestry regulation, the Planning Commission should reopen the public hearing and take additional testimony. After close of the public hearing, the Planning Commission could then deliberate and prepare a recommendation to the City Council.

\\Tracy\c\Planning 2000\Forestry\2002\02_016 staff report 08262002.doc

At the hearing on 7/22/02, the Planning Commission had a lengthy discussion and decided that a transcript of the discussion should be prepared to assist them in collating issues and preparing a recommendation. The minutes include a detailed transcript of the discussion and this language is also included in this staff report.

Key Features of the Ordinance

- Tree removal in the areas protected by the steep slope and buffer areas of the FSH ordinance are covered by the FSH ordinance (they are referred to in the tree ordinance to ensure that it encompasses all land in the UGB).
- For areas of one acre or larger that can be further developed, three trees of at least 11 DBH must be retained. At least half of these trees must be a conifer species.
- The acreage threshold and retention requirement are based on contiguous areas under common ownership. For example, suppose a person owns a one-acre parcel that is fully forested, and an adjacent ten-acre parcel that is pastureland with no trees. The retention requirement would be 33 trees, based on the two contiguous parcels under the same ownership.
- No permit is required for tree removal on residential lots that are smaller than one acre.

Advantages of the Draft Ordinance for Tree Preservation

- 1. The draft ordinance does prohibit removal of all large trees.
- 2. The draft ordinance is more stringent than the state forest practices act:
 - a. It retains 50% more trees per acre.
 - b. It has a much lower threshold acreage (i.e., one acre vs. twenty-five acres).
 - c. It applies to the total area of contiguous ownership, not just the area proposed for the forest operation. Within an urban area, this can significantly increase the retention requirement.
- 3. The ordinance does not "penalize" owners of property with many existing trees. In fact, it gives them more flexibility in future use of the property, since they have more choices over which trees to retain. Owners of developable property with few trees have less flexibility.

Advantages to Property Owners

- 1. The property owner has full control over which trees are retained (as long as the standards of the ordinance are met).
- 2. Trees protected by the FSH ordinance can be counted toward the retention requirement.
- 3. There is no lengthy permit or review period. The property owner must only show that the required number of trees will be retained during and after the forest operation.
- 4. There is a variance procedure that could serve as a relief valve for property owners. For example, if the existing trees fall short of the retention requirement, then none of the trees could be removed. But it could be possible that a tree sits in the middle of the only logical location for a street extension. The planning commission could issue a variance (but the burden of proof on a hardship condition would be on the property owner).

 Property owners can increase their future options by planting trees. The more large trees on their property, the more flexibility they will have over which trees must be retained. Ease of Administration

- 1. The ordinance is much simpler to administer than the original draft, therefore increasing the likelihood that it will be uniformly enforced.
- 2. Compliance with the ordinance will be simple to verify. There is no need for complicated basal area calculations or urban development master plans.

\\tracy\c\planning 2000\forestry\2002\02_016 staff report 08262002.doc

3. There is no need to track the location and identity of retained trees. All that is required is a survey of trees at the time of each forest operation. The trees marked for retention in one forest operation could be replaced by others ten years later, if additional trees had been retained or if other trees grew to 11" DBH.

Recommendation:

The revised draft ordinance will be a first step in regulating tree removal on undeveloped land. Staff recommends that the Planning Commission forward this to the City Council for consideration.

Revisions will also be necessary to other sections of the Development Code dealing with property divisions (partitions, subdivisions), to ensure that retained trees remain protected during the subdivision process. The additional revisions are also included in this packet and a recommendation should include a reference to these additional amendments.

\\tracy\c\planning 2000\forestry\2002\02_016 staff report 08262002.doc

	ATTACHMENT 7
NE	URBAN FORESTRY CESSARY CHANGES TO OTHER CHAPTER OF THE SANDY DEVELOPMENT CODE
CHA	PTER 17.62 - CULTURAL AND HISTORIC RESOURCES
17.62	.30 CRITERIA FOR DESIGNATION
The evalu desig Socia landn	Planning Commission and City Council shall use the following criteria and standards to ate whether a particular object, site, structure or place merits a CHR designation. In order to nate a CHR, the Planning Commission and City Council must find, through an Economic l, Environmental and Energy (ESEE) analysis, that the benefits of designating the proposed hark outweigh the benefits of continuing the conflicting use or uses without the designation.
A. A	association with historic or famous events; or
 B. U 1. 2. 3. 4. 5. 	 inique architectural design or mode of construction because of: Representative character of a period or a particular architectural style, building type or method of construction. Extraordinary or unusual architectural merit by reason of design, detail, use of materials or craftsmanship or Identification as the work of an architect, designer or master builder whose individua work has influenced development in the nation, state or community; or Significance as the only remaining, or one of the few remaining, resources of a particular style, building type, design, material or method of construction.
C .	Inclusion in an official Register of Historic Places; or
D.	Relationship to the broad cultural history of the nation, state or community; or
E.	Identification with a person or persons who have significantly contributed to the history of the city; or
F.	Identification as a unique object representing an aesthetic or educational feature of the community.
G.	Archaeological site designation.
H.	 Environmental Significance. Significance as a visual landmark. Integrity of surrounding land-use of the historic period represented. Significance because the resource contributes to the continuity of historical character of the street, neighborhood and/or community.
C:\Pla	nning 2000\Forestry\2002\02-020 Other Required Amendments.doc



3. Zoning, size and dimensions of the tract to be partitioned 4. Size, dimensions and identification of proposed parcels (Parcel 1, Parcel 2, Parcel 3) 5. Approximate location of any structures on the tract to be partitioned, including setbacks to proposed parcel boundaries 6. Location, names and widths of streets, sidewalks and bikeways within the tract to be partitioned and extending 400 feet beyond the tract boundaries 7. Location, width and purpose of existing and proposed easements on the tract to be partitioned 8. Location and size of sewer, water and drainage facilities proposed to serve the tract to be partitioned 9. Natural features such as waterways, drainage area, significant vegetation or rock outcroppings, AND TREES REQUIRED TO BE RETAINED UNDER THE PROVISIONS OF CHAPTER 17.102. 17.100.60 SUBDIVISIONS D. Data Requirements for Tentative Plat. 1. Scale of drawing, north arrow, and date. 2. Location of the subdivision by section, township and range, and a legal description sufficient to define the location and boundaries of the proposed tract. 3. A vicinity map, showing adjacent property boundaries and how proposed streets may be extended to connect to existing streets. 4. Names, addresses, and telephone numbers of the owner(s) of the property, the engineer or surveyor, and the date of the survey. 5. Streets: location, names, paved widths, alleys, and right-of-way (existing and proposed) on and within 400 feet of the boundaries of the subdivision tract. 6. Easements: location, widths, purpose of all easements (existing and proposed) on or serving the tract. 7. Utilities: location of storm drainage, sanitary sewers and water lines (existing and proposed) on and abutting the tract. If utilities are not on or abutting the tract, indicate the direction and distance to the nearest locations. 8. Ground elevations shown by contour lines at two-foot vertical intervals for ground slopes of less than 10 percent and at ten-foot vertical intervals for ground slopes exceeding 10 percent. Ground elevation shall be related to an established benchmark or other datum approved by the Director. 9. Natural features such as marshes, rock outcroppings, watercourses on and abutting the property, location of wooded areas, AND TREES REQUIRED TO BE RETAINED UNDER THE PROVISIONS OF CHAPTER 17.102. 10. Approximate location of areas subject to periodic inundation or storm sewer overflow, location of any floodplain or flood hazard district. 11. Location, width, and direction of flow of all water courses. 12. Identification of the top of bank and boundary of mandatory setback for any stream or water course. 13. Identification of any associated wetland and boundary of mandatory setback. 14. Identification of any wetland and boundary of mandatory setback. 15. Location of at least one temporary bench mark within the tract boundaries. C:\Planning 2000\Forestry\2002\02-020 Other Required Amendments.doc Page 3



Sandy City Council Regular Meeting November 4, 2002

Mayor Linda Malone called the regular meeting of the Sandy City Council to order at 7:00 p.m. in the Council Chambers at City Hall. The meeting opened with the Pledge of Allegiance to the flag.

Roll Call

Mayor Linda Malone - Present Councilor Tina Frostad – Present Councilor Gene Smith – Present Council President Don Allen – Present Councilor Dick Steiner – Present Councilor Roberta Kennedy – Present Councilor Bill O'Brion – Present

Shadow Councilors

Sarah Cancellieri - Present Laura Penland - Absent Nathan Visan - Absent John Reed - Present Ryan Montgomery- Present Matt Frick - Present

Also Present: City Manager Scott Lazenby, Department Directors, Recording Secretary, Nancy Enabnit, and interested citizens.

Changes to Agenda: None

Public Comment: None

Reports and Comments from Neighborhood Associations: None

Youth Council Report: None

Appointments:

Library Board Appointment - Staff recommends Linda Gettmann be appointed.

Motion: Councilor Kennedy made a motion to appoint Linda Gettmann to fill vacancy on Library Board, seconded by Councilor Allen. There were no abstentions or nay votes.

Yea: 7- Frostad, Smith, Allen, Steiner, Kennedy, O'Brion, Malone Nay: 0 Abstentions: 0 Motion carried.

Consent Agenda:

Sandy City Council Minutes of October 21, 2002

Bill O'Brion asked that a list of citizens present at be included with the minutes. He noted that Drew Anderson as his Eagle Project installed the stairs at Meinig Park on November 2nd. Over 20 Scouts assisted him.

Motion: Councilor Allen made a motion to accept the consent agenda with spelling correction of Nancy Jaksich's name, seconded by Councilor O'Brion. There were no abstentions or nay votes.

Yea: 7- Frostad, Smith, Allen, Steiner, Kennedy, O'Brion, Malone Nay: 0 Abstentions: 0 Motion carried.

Old Business: None

Public Hearing: File 02-016, Chapter 17.102 Urban Forestry Ordinance

Public Hearing opened at 7:08.

Mayor Malone called for any abstentions, any declarations of conflict of interest, any exparte contact, any challenges to the jurisdiction of the Planning Commission in this matter, or any challenges to individual members. Answers were negative and unanimous in all cases.

Mayor Malone then reviewed the public hearing procedures.

Staff Report:

Kevin Liburdy stated that the major points of this proposal is regulation of tree harvesting and "trees" are defined only as those 11 inches or greater in diameter breast height. Tree harvest on sites greater than 1 acre inside the UGB requires a permit issued by the City. This process also requires a dedication of three, 11 inch or greater trees per acre, under contiguous ownership, prior to harvest. It requires a 70-ft, no-harvest buffer adjacent to Tickle Creek and a 50-ft buffer along other perennial streams inside the UGB, outside the city limits, but inside the UGB. It eliminates ODF regulation inside the UGB. Kevin then gave an overview of the history of the Ordinance.

Scott Lazenby added that there had been feedback from the Oregon Department of Forestry, and that some of their observations had been added to the Ordinance. He also stated that a few of the citizens had contacted the city with concerns over procedural issues. Scott then read a letter received from the Oregon Department of Forestry in support of an Urban Forestry Ordinance.

Public Testimony:

Bob Kallen - 39517 Scenic St. - Sandy, OR 97055

Mr. Kallen reiterated his concerns that he would not be able to maintain the 'scenic view.' He stated that his property is taxed at a higher rate due to the view.

Councilor O'Brion gave an explanation of the Urban Forestry Ordinance versus the FSH ordinance to clarify some of the questions Mr. Kallen raised.

Mr. Kallen still feels that this ordinance is to complicated for the average property and he recommends the creation of a separate section in the ordinance to address 'scenic properties.'

Wally Hewitt - P O Box 462 - Sandy, OR 97055

Mr. Hewitt stated that he has a farm-woodlot operation outside the city limits, but within the UGB. He had concerns about having to consult an arborist or forester to cut hazard trees. He also expressed concerns about the repeated need for permits every time he takes a tree down. He suggested an ongoing permit for woodlot operations. Mr. Hewitt stated that he supports the ordinance in general.

Bob Rockwell - P O Box 724 - Sandy, OR 97055

Mr. Rockwell stated that he was on the original tree committee. He added that the ordinance should be as simple as possible and this version is the best he has seen.

Mayor Malone closed the public hearing at 7:51 p.m.

Staff Recap:

Scott Lazenby suggested that by increasing the validity of permits from 180 days to 1 year it would address Mr. Hewitt's concerns.

Councilor O'Brion thought that any activity during that year should require notification to the City.

Mr. Hewitt stated that prior notification is not always possible with an ongoing operation.

Councilor O'Brion stated that if every operation was like Mr. Hewitt's there would be no problem, but he feels that 24-hour notification is not too much to ask.

Councilor Frostad stated that she agrees with Councilor O'Brion's comments. Councilor Smith asked for clarification on 17.02.50 b. (page 12). He stated that it looked contradictory and felt it needed clarification. Councilor Allen stated that he would like more specifics regarding blow-down, to encourage groves and clarify options. Councilor O'Brion stated that there are professional foresters on call and available if staff is not comfortable with placement of retained trees. Councilor Steiner asked how the number 3 (trees) came about. Councilor O'Brion stated that it came about after review and much discussion trying to reach a compromise for this ordinance. Councilor Kennedy asked approximately how many forested acres are located within the city limits?, within UGB? She stated that she felt that the ordinance is a simple as it could be and still meet it's goals. Scott Lazenby stated that we don't have an actual total. Councilor O'Brion stated that it really is not about acres but about preservation. He stated that this ordinance has been longer than 3 years in the making. He would like to thank staff and committee members for their diligence. Some of the goals accomplished were: 1. 1. City has jurisdiction, which eliminates disputes. 2. 2. Tree retention 3. 3. Dedicated tree fund He added that 17.102.20 b 2 - need to have a permit after the fact (after emergency removal). He would like to see the addition of a clause that states once you have applied for your permit you will know within 15 days whether you are approved or denied. Page 11, letter D, he stated that the threshold of a Type 1 permit could be even larger to prevent unnecessary appeals. Motion: Councilor Smith made a motion to that removal of 50 trees would require a Type 2 procedure, seconded by Councilor Steiner. Yea: 5- Smith, Allen, Steiner, Kennedy, O'Brion Nav: 2 - Frostad, Malone Abstentions: 0 Motion carried.

Councilor O'Brion continued with line item E letter A. He felt that this was an unnecessary difficulty to impose. It was pointed out that the Title companies keep a more up-to-date list of property owners and provide the service free of charge. 17.102.40 - page 12 - letter b. - clause that requires 24 hour advance notification. 17.102.50 - letter a #1 that there be line stating that it does not have to be on every specific acre, a total of. 17.102.60 - tree re-planting - under #3 & 4 - need to exclude the 11/2-inch caliper. It is not so much the size of the tree, but the quality of the tree. Councilor O'Brion pointed out that a board or committee needs to be appointed. Mayor Malone stated that although this ordinance does not satisfy everyone, she feels that it is a good ordinance. Motion: Councilor O'Brion made a motion to direct staff to prepare an ordinance and findings for approval at next meeting, seconded by Councilor Allen. There were no abstentions or nay votes. Yea: 7- Frostad, Smith, Allen, Steiner, Kennedy, O'Brion, Malone Nay: 0 Abstentions: 0 Motion carried. New Business: Community Development Block Grant Applications: Scott Lazenby stated that staff needs direction from Council as to whether these are good projects to submit and what priority Council would like to place on each project. Project descriptions are in Council Packets. CDBG Malone Allen Smith Steiner Frostad Kennedy O'Brion 3 4 2 Ł 3 2 1 2 3 1 3 3 4 1 4 3 4 1 4 2 1 3 5 4 2 2 5 5 2 4 2 5 5 5 1 4 5 5 3

Shadow Councilor Frick stated that the ADA items are good and that # 5 part C seems like a good thing.

Shadow Councilor Montgomery rated 3 & 4 as highest priority. Shadow Councilor Cancellieri rated 3 & 4 as highest priority.

Sandy River Park Land Acquisition:

Scott Lazenby stated that there is a deadline as far as the State Parks Grant is concerned. They did award \$250,000 towards the development of the Park. We have tried to obtain



Exhibit B Ordinance No. 2002-10 Page 2

considered a range of alternatives before adopting the final version of the regulations.

- iii) Through the public hearing process, significant effort has been expended to provide greater clarity regarding key provisions and definitions used in the regulations.
- iv) Throughout the process, the City has engaged in and maintained a dialogue with the Department of Forestry regarding the regulations.
 Based upon its conversations with the Department, the City has incorporated into the regulations many of the Department's suggestions.
- v) Upon successful acknowledgement of the regulations, the City will seek to amend its Growth Management Agreement with Clackamas County in order to regulate forest practices within Sandy's UGB but outside of its city limits.
- Goal 3: Agricultural Lands. This goal is not applicable to the regulations.
- 4. <u>Goal 4: Forest Lands</u>. The regulations mostly affect property within Sandy's city limits property not zoned as forest land. However, the City Council acknowledges that it intends to eventually regulate forest practices on property outside Sandy's city limits and within Sandy's UGB including property zoned forest land by Clackamas County. With regard to those lands, compliance with Goal 4 is demonstrated via the following:
 - Via ORS 527.722, the Oregon Legislature has specifically authorized Sandy to regulate forest practices on such lands.
 - ii) Land use designations on such lands shall remain the same. With the exception of a modest 3 tree per acre retention requirement and stream buffers, forest operations shall continue to be allowed on forest lands.
 - iii) The regulations shall not alter existing land divisions standards on forest lands, nor shall the regulations alter existing dwelling citing standards on such lands.
 - iv) Through the retention requirement, a modest amount of forest land is guaranteed to be conserved, consistent with the Goal's plain language.
 - v) In certain areas where water quality is threatened and/or natural hazards exist, the regulations will interface with Sandy's Flood Slope Hazard Overlay zone, which will regulate forest practices consistent with sound management of soil, air, water, and fish and wildlife resources.
 - vi) Overall, the regulations will maintain the forest land base and will continue to allow economically efficient forest practices that assure a continuous supply of harvest stock as the leading use on forest land.
 - 5. <u>Goal 5: Natural Resources</u>. Based on the following discussion, the City Council concludes that Goal 5 does not apply to the regulations. As defined at OAR 660-023-0010(5), the regulations are a "post-acknowledgement plan amendment" (PAPA) because, in pertinent part, they constitute an "adoption of [a] new ... land use regulation." In adopting a PAPA, local governments

Exhibit B Ordinance No. 2002-10 Page 3

are required to apply Goal 5 only if the PAPA "affects a Goal 5 resource." OAR 660-023-0250(3). As defined further in 0250(3), a PAPA affects a Goal 5 resource "only" if the PAPA (1) "creates or amends" a "resource list;" or "a portion of a land use regulation adopted in order to protect a significant Goal 5 resource or to address specific requirements of Goal 5;" or (2) "allows new uses that could be conflicting uses with a particular significant Goal 5 resource site on an acknowledged resource list."

- i) Based on the exclusive categories above, the PAPA does not affect any Goal 5 resource.
 - Trees are not a part of any resource list maintained by the City of Sandy – therefore the PAPA is not amending such a list. Neither is a resource list being created via the regulations.
 - (2) The regulations are not creating or amending a land use regulation adopted in order to protect a significant Goal 5 resource, because trees are not a significant Goal 5 resource under OAR 660, Division 23. Nor will the regulations create or amend a land use regulation adopted in order to address specific requirements of Goal 5, because the City Council does not intend to address those requirements through these regulations. LUBA precedent supports this conclusion. See Home Builders Ass'n of Lane County v. City of Eugene, 41 Or LUBA 370 (2002) (no authority for the proposition that local jurisdiction is required to apply Goal 5 before it regulates resources that are not a part of the local jurisdiction's Goal 5 inventory); Ramsey v. City of Portland, 30 Or LUBA 212 (1995) (an ordinance regulating the cutting of trees does not affect any Goal 5 site nor implicate Goal 5).
 - (3) Finally, the regulations are not providing for any new "uses;" therefore there will be no conflicts with any particular Goal 5 resource sites.

 <u>Goal 6: Air, Land and Water Resources Quality</u>. The City Council finds that Goal 6 is met in the following ways:

 By requiring trees to be retained on certain properties within Sandy's UGB, the regulations will have a modest effect on improving water quality when compared to the current lack of a tree retention requirement, because runoff and sedimentation to sensitive surface and groundwater areas will likely decrease.

ii) The retention requirement, alone and in combination with other applicable provisions of Sandy's development code, shall further ensure that waste and process discharges from future development will comply with applicable state and federal environmental statutes and rules

- iii) The retention requirement will also contribute to ensuring that such discharges will not degrade, overload, or threaten the availability of air, water or land with Sandy's UGB.
- <u>Goal 7: Natural Disasters and Hazards</u>. Chapters 17.56 and 17.60 of the Sandy Municipal Code primarily address Goal 7. However, these regulations

Exhibit B Ordinance No. 2002-10 Page 4

will also address Goal 7 by adding additional safeguards to development in potentially hazardous areas. As applied to potentially hazardous areas, the tree retention requirement will reduce the possibility of landslides and encourage a less dense development pattern in such areas. Goal 7 is met.

8. <u>Goal 8: Recreation</u>. By requiring tree retention, the regulations assist in preserving trees for Sandy's future. Thus, the regulations will only increase general recreational opportunities today and in the future for Sandy's residents and visitors. However, the regulations do not authorize nor address the citing of recreational and destination resorts. The City Council finds that Goal 8 is met.

- Goal 9: Economy of the State. The regulations will have a negligible effect on the profitability of forest practices. The value of certain properties may increase as a result of this ordinance. Goal 9 is met.
- Goal 10: Housing. Because a developer may choose where on a given property to retain trees, and because the vast majority of developments voluntarily retain trees above the regulations' retention requirement, the regulations will have little to no effect on Sandy's ability to provide needing housing. Goal 10 is met.
- <u>Goal 11: Public Facilities and Services</u>. The regulations will have no effect on the ability of the City to provide public facilities and services. Goal 11 is met.
- <u>Goal 12: Transportation</u>. The regulations will have no effect the City's Transportation System Plan. Goal 12 is met.
- <u>Goal 13: Energy Conservation</u>. The regulations arguably encourage a more energy-efficient and compact urban growth form. The retention requirement could also provide shade in the summer months. Otherwise, the regulations have a negligible affect on energy conservation. Goal 13 is thus met.
- <u>Goal 14: Urbanization</u>. As above, the regulations arguably encourage a more compact urban growth form, thus reducing pressure on agricultural and farmland. Otherwise, the regulations have a negligible affect on urban growth management objectives. Goal 14 is thus met.

C. Conclusion.

For all of the above reasons, the Sandy City Council finds Ordinance No 2002-10 to be in conformance with the State of Oregon's Statewide Planning Goals.

PRE-APPLICATION CONFERENCE NOTES

Project Name: Jacoby Estates Subdivision

Pre-Application Conference Date: January 3, 2018 Applicant Name: Cory Knight and Mac Even Engineer Name: All County Surveyors and Planners Staff: Thomas Fisher, Emily Mcharg, Kelly O'Neill Jr., James A. Cramer & Mike Walker

PLANNING DEPARTMENT REVIEW

Sandy Development Code (SDC): Sandy Development Code Sections 17.12 Procedures for Decision Making; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 SFR Single Family Residential; 17.60 FSH Overlay; 17.80 Additional Setbacks on Collectors; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.86 Parkland and Open Space; 17.90 Landscaping and Design Standards; 17.92 Landscaping and Screening; 17.98 Parking, Loading and Access Requirements; 17.100 Land Division; 17.102 Urban Forestry; and Chapter 15.30 Dark Sky.

Caveat: This analysis includes a review of those code sections that may conflict with the proposed design as submitted. This review is not intended to be a comprehensive analysis of all applicable code sections.

Parking Analysis

- Locations of the driveways should be identified for review (SDC 17.90.90.B.5).
- 2 off-street parking spaces per dwelling required (SDC 17.98.20), demonstrate compliance.
- On-street parking plan shall be submitted for review. One space required for every dwelling unit within 200 feet of each lot.

Access and Utilities

- Frontage improvements along east side of SE Jacoby Road are required with development of property.
- Need analysis to determine if sewer pump station in Timberline Trails has sufficient capacity for this development proposal (based on 32 lots).
- Need analysis on depth of sanitary sewer and storm in Lots 9 and 10 to determine if 20 foot wide casement is sufficient.
- Submit a traffic impact analysis (TIA). Need \$1,500 for traffic consultant.
- Streets shall be named in common with Cascadia Village (i.e. New England towns).
- Distance between Jacoby Road and any internal streets shall be spaced at least 150 feet. Plan only details the intersection spacing on Street B at 125 feet so lots by Street A need to be deeper.
- If wider area is dedicated at Newton Street and Jacoby Road to create proper alignment then street improvements need to be designed accordingly.
- Vision clearance areas must remain unobstructed (SDC 17.74.30).
- Transit requires a transit amenity on Jacoby Road in NW corner of Tract B. The amenity required is a 6 foot long green metal bench (Fairweather model PL-3, powder-coated RAL6028) mounted on a 7 foot by 9.5 foot concrete pad which could accommodate a 5 foot by 7.5 foot bus shelter.
- Existing access easement on Newton Street shall be submitted for review.
- Future "conservation easement" document within Lots 19-21 required be submitted for review.
- Dedicate both Tracts A and B.
- Any well and septic system associated with the existing single-family home will need to be abandoned and the house be removed (with demolition permit) prior to plating subdivision.

Other Planning Items

- Parks fee in-lieu required at approximately \$99,485.
- Landscaping is required on Tract B (Landscape standards within SDC 17.92).
- Density needs to be determined based on net acreage (9.64 acres per County GIS records, density is 3 to 5.8 units per net acre). Right-of-way, Tract A, Tract B, and pedestrian connection to Tract A need to be removed from the gross acreage.
- A variance would be required to allow a single stack housing development (Lots 29-32), two tier required per SDC 17.100.120 (Jacoby is a Collector St.)
- Houses would have to face "Jacoby" or you would have to apply for a variance (SDC 17.82 Special Setbacks on Transit Streets).
- Proposed building footprints should be identified for review (SDC 17.90.90.B.2). Setback standards can be found in SDC 17.34.30. Additionally, Section 17.80.20 states any structure located on streets identified in the Transportation System Plan as collectors shall have a minimum setback of 20 feet measured from the property line. This applies to applicable front, rear and side yards. (Jacoby is identified as a collector).
- Blocks can't be greater than 400 feet unless justified by topographic, natural area, or other physical conditions. Blocks greater than 400 feet require a variance. Blocks greater than 600 feet require a pedestrian and bicycle access way (17.100.120.B).
- A dedicated tract from the terminus of Street C to connect to Tract A for future development of the Tickle Creek Trail. The trail will need to be improved for pedestrian access.
- Are there any proposed retaining walls? If so, what are their locations, heights, and materials? Section 17.74.40 details code standards which could require a variance.
- A geotechnical study will need to be done for any area at 25 percent slope or greater that is proposed to contain development.
- A wetland mitigation study will define restricted development areas on the site, which in turn will define tree retention requirements in those areas. Will need to define the top of bank for Tickle Creek.
- Submit an arborist report and tree plan for trees 11-inches DBH and greater (and trees 6-inches DBH and greater in the restricted development area). Tree retention at 3 trees per acre. Trees must be 11"
 DBH or greater and in good health. Identify on the plans which trees are to be removed as well as retained.

Application Process: Type III SUB review with requested variances (most likely three or more variances).

Projected Processing Steps:

- Submittal Requirements: See requirements lists on City of Sandy website. https://www.ci.sandy.or.us/Planning-Requirements/
- Fees: \$3,000 for Type III subdivision review plus \$80 per lot; \$1,000 per variance; \$700 for FSH Overlay review; \$150 for Tree Removal review; \$1,500 for Third Party traffic consultant.
- Staff review for completeness (30 days max.), if determined incomplete then the applicant submits
 additional information as required, staff then reviews for completeness again, if the application is
 deemed complete then the application is processed.

From: James Cramer <jcramer@ci.sandy.or.us> Date: February 12, 2018 at 3:04:57 PM PST To: cory@taylorgrouprealty.com Subject: Jacoby Estates

Cory,

Kelly passed along the document you left for him to look over and respond too. Below is the CIty's response to your inquiries as well as a couple items for consideration we came across during our research and observation of the property.

- A wetlands report will not be required for the development. Section 17.60 details the requirements for development within the FSH overlay and sense no development is proposed within the overlay, or within the required setbacks of this code section, Staff is comfortable not requiring a report at this time.
- As a reminder, a Geological Assessment is required for any area at 25 percent slope or greater that is proposed to contain development (17.56.40).
- We discussed the potential for a credit to the Park In-Lieu fee based on the appraised value of Track B. Should you propose to be given a credit for that land area we would need to see an appraised value to determine how we would like to proceed with that element of the project.
- Should the owner decide to dedicate any portion of land the appropriate documents per Section 17.86.30 would be required with the initial submission for consideration.
- As mentioned in the preapplication notes, an arborist report and tree plan for trees 11-inches DBH and greater (and trees 6-inches DBH and greater in the restricted development area). Tree retention at 3 trees per acre. It has been observed that the site may have been removing trees without approval. If that is the case those actions need to stop immediately until approval for their removal has been given.
- Additionally 17.102.60 details planting that needs to occur when removing trees which does not appear to be occurring.

Please let me know if you have any additional question and fill me in on any improvements being made to the property so we can ensure they are meeting code.

James A. Cramer Associate Planner City of Sandy <u>39250 Pioneer Blvd</u> Sandy. OR 97055 phone (503) 783-2587 jcramer@ci.sandy.or.us



Storm Drainage Design and Calculations For the Jacoby Heights Subdivision

June 5, 2018

Prepared By: All County Surveyors and Planners, Inc. Tyler Henderson, E.I. Ray L. Moore, P.E., P.L.S. P.O. Box 955 Sandy, Oregon 97055 Phone: (503) 668-3151 Fax: (503) 668-4730 Job #16-045

> Prepared For: Mr. Cory Knight 16513 SE Orchard View Lane Damascus, OR 97089



Table of Contents:

Page(s)

Table of Contents and Appendices	1
Purpose	ż
Project Location and Description	2
Proposed Improvements	2
Hydrograph Parameters	2-3
Detention Sizing Results	4
Water Quality Design	5
Conclusions	5

Appendices:

- Appendix A Vicinity Map Site Layout
- -Time of Concentration Plan -Pre-Developed Areas
- -Developed Areas

- Appendix B Standard Formulas
- Coefficients
- Coencients
 SCS Runoff Curve Numbers (CN)
 Pre-Developed Time of Concentration Calculations
 Developed Time of Concentration Calculations
 Hydrograph Analysis Summary
 Detention System Summary

- -Stage Storage Summary -Rectangular, Sharp Crested Weir Calculations -Contech CDS2015-4-C Detail

17-199-Storm Report-Prelim Body.doc

PURPOSE:

The purpose of this analysis is to:

- Describe existing and proposed site conditions.
- Provide detention calculations for the 2-yr, 5-yr, 10-yr, and 25-yr storm events.
- Provide water quality calculations.

PROJECT LOCATION AND DESCRIPTION:

The project site is located on the east side of Jacoby Road in Sandy, Oregon. The site includes only taxlot 2300. The +/-9.7-acre site consists of grassy fields, and plentiful tree cover. The land is generally sloped to the north and west with an average slope of about 10%. A Vicinity Map and Site Layout (with proposed storm sewer layout) can be found in Appendix A.

PROPOSED IMPROVEMENTS

The proposed Jacoby Heights Subdivision project will consist of 31-single family residential lots ranging from 7,500 SF to 16,500 SF. The site improvements will include streets, curbs, sidewalks, utilities, etc.

New storm sewer pipes, manholes, and catch basins will be installed to convey storm water to a detention pond located in Tract B. A new water quality manhole will be installed downstream of the detention pond (See Site Layout – Appendix A).

HYDROGRAPH PARAMETERS:

Rainfall

The rainfall distribution numbers below were taken from the City of Sandy Stormwater Website: http://www.ci.sandy.or.us/Stormwater/

2 year, 24 hr. rainfall = 3.5" 5 year, 24 hr. rainfall = 4.5" 10 year, 24 hr. rainfall = 4.8" 25 year, 24 hr. rainfall = 5.5"

Soils

The soil data for this site is from *Soil Survey of Clackamas County, Oregon* published by the United Stated Department of Agriculture (USDA). The post-development soil is assumed to be the same as pre-development.

Soil Type: 15B, Cazadero silty clay loam. Hydrologic Group "C" 15D, Cazadero silty clay loam. Hydrologic Group "C" 24B, Cottrell silty clay loam. Hydrologic Group "C"

(See Appendix B for Runoff Curve Numbers)

17-199-Storm Report-Prelim Body.doc

2

Areas and Curve Numbers

Drainage basin areas were determined using a topographic map drafted in AutoCAD. See the Pre-Developed Area and Developed Area in Appendix A.

The impervious area for these post-developed basins includes the proposed roofs, streets, sidewalks, planter strips, driveways, and curbs. See the following tables for a specific breakdown of these areas.

	Pre-Devel	opment
Areas	CN	Land Use Description
Pervious (9.42 acres)*	82	Lawn & Young Second Growth Forest Land
Impervious (0.26 acres)	98	Existing Building, AC, Concrete
	Post-Devel	opment
Areas	CN	Land Use Description
Pervious (7.02 acres)	86	Lawns "Good Condition"
Impervious (2.66 acres)	98	Buildings, AC, Sidewalks, etc.

Pre-Developed Pervious CN: Weighted CN

*Lawn "Good Condition" 1.08 AC: CN=86

Wood or Forest Land "Young Second Growth" 8.34 AC: CN=81 [(1.08AC x 86) + (8.34AC x 81)] / (1.08+8.34) = 81.57= 82.0

Pre-Developed Impervious CN: See Runoff Curve Numbers Appendix B

Post-Developed Pervious CN: See Runoff Curve Numbers Appendix B Post-Developed Impervious CN: See Runoff Curve Numbers Appendix B

Time of Concentration

The times of concentrations (Tc), were calculated using the equations and spreadsheets shown in Appendix B.

Pre-development Tc=	48.0 minutes
Post-development Tc=	5.8 minutes

Hydrograph Modeling Results

Hydrographs for the site were determined using a spreadsheet based on the King County, Washington Hydrograph Program, version 4.21B, which uses the Santa Barbara Urban Hydrograph (SBUH) method.

17-199-Storm Report-Prelim Body.doc

DETENTION SIZING RESULTS:

(-)

The Post-Development flows were routed through a proposed 4-foot deep detention pond. The 4-foot deep detention pond has been designed so that the Post-Developed release rates for the entire site do not exceed the Pre-Developed rates for the 2-year, 5-year, 10year, and 25-year storm events per the City of Sandy public Works Design Standards. See the Detention System Summary in Appendix B.

The required storage volume is 21,793-cubic feet. This can be contained in a 4-foot deep pond with a bottom area of 3,781 square feet.

Flow Control:

The flow control orifices were designed to release the Post-development Peak-Q's at or below the Pre-developed Peak-Q's.

(See the Detention System Summary - Appendix B)

	Orifice Table	
Orifice	Dia. (inches)	Height (feet)
Bottom	7.65	0.0
Тор	8.85	3.17

17-199-Storm Report-Prelim Body.doc

WATER QUALITY DESIGN:

CDS Storm Water Treatment Device

A CDS manhole by Contech Stormwater Solutions was designed for water quality for the site - see detail in Appendix B. The impervious area for the site includes AC pavement, sidewalks, planter strips, and roofs. The impervious area is 2.66-acres.

Proposed asphalt, walks, pla	inters etc.:	0.88 acres
Roof, Patio, Driveway*:		1.78 acres
Total Impervious Area:		2.66 acres
*40'x50' Building footprint:	2000SF	
20'x20' Driveway:	400SF	
10'x10' Patio:	100SF	
Total:	2,500SF X 31	lots = 77,500SF

The flow (Q) from this runoff was calculated using the rational method (Q = CIA)

Where Q = flow (cfs)

C = runoff coefficient = 0.90 pavement and Roofs I = Intensity = 0.2 inches per hour (Water Quality Design Storm) A = Impervious Area = 2.66 Acres

Q = 0.90 X 0.2 X 2.66 Q = 0.48 cfs

The Contech Storm Water Treatment Device Model: CDS2015-4-C has a treatment capacity of 0.7 cfs which exceeds the required 0.48 cfs.

A Storm Water Treatment Device CDS Model CDS2015-4-C can be used to adequately treat the water for the site

CONCLUSIONS:

- The conveyance system for the proposed Jacoby Heights Subdivision site has been sized to handle the peak 25-year, 24-hour storm.
- On-site detention has been designed to maintain existing downstream storm water runoff characteristics in accordance with the City of Sandy requirements.
- A CDS Storm Water Treatment Device will be used for water quality.

17-199-Storm Report-Prelim Body.doc

5

Appendix A

(

-Vicinity Map -Site Layout -Time of Concentration Plan -Pre-Developed Areas -Developed Areas

Θ











Appendix B

-Standard Formulas -Coefficients

Ô

- -Coefficients -Curve Runoff Numbers (CN) -Pre-Developed Time of Concentration Calculations -Developed Time of Concentration Calculations -Hydrograph Analysis Summary -Detention System Summary -Stage Storage Summary -Rectangular, Sharp Crested Weir Calculations -Contech CDS2015-4-C Detail

()

Standard formulas used for the Time of Concentration Calculations

Overland Flow (max 300' total)



Тс = time of concentration for less than 300' of travel (minutes) Ns = sheet flow Manning's effective roughness coefficient = flow length (ft) L P2 = 2-year, 24 hour rainfall (in) = slope of hydraulic grade line (land slope, ft/ft) So

Shallow Concentrated Flow (after initial 300')



Т L So k

= travel time for sheet flow (min) = flow length (ft) = slope of hydraulic grade line (land slope, ft/ft) = time of concentration velocity factor (ft/s)

Flow in Swales

Q = (1.486/n) x A x R^2/3 x S^1/2 (Manning's Equation)



Flow in gutters



Tc = time of concentration for gutter flow (minutes) V = average velocity of flow (ft/sec) Q = quantity of flow (ft^3/sec) S = street longitudinal slope (ft/ft) Sx = street cross slope (ft/ft) Т = total width of flow in the gutter (ft) = sheet flow Manning's (pavement = 0.018) n L = Length of flow (ft)

Flow in pipes Mannings Equation



COEFFICIENTS

Ns = = Manning's coefficient (<u>sheet flow</u>) n values are for sheet flow only

Design Value

0.011 Concrete or asphalt

0.010 Bare soil

0.020 Graveled surface

0.020 Bare clay - loam (eroded)

0.150 Grass (short prairie)

0.240 Grass (dense lawn)

0.410 Grass (bermuda)

0.400 Woods (light underbrush)

0.800 Woods (dense underbrush)

k = _ = time of concentration velocity factor (ft/s) Design Value

3 Forest with heavy ground cover and meadows (n=0.10)

5 Brushy ground with some trees (n=0.060)

8 Fallow or cultivation (n=0.040)

9 High grass (n=0.035)

11 Short grass, pasture or lawns (n=0.030)

13 Nearly bare ground (n=0.025)

27 Paved and gravel areas (n=0.012)

n = = Manning's coefficient (<u>channel</u>) Design Value

CONSTRUCTED CHANNELS

A. Earth, straight and uniform

0.018 Earth (straight and uniform)

0.025 Gravel (straight and uniform)

0.027 Grass (with weeds)

B. Earth, winding and sluggish

0.025 Earth (no vegetation)

0.030 Grass (some weeds)

0.035 Dense weeds (deep channel)

0.030 Earth (rubble bottom and sides)

0.035 Stony bottom and weedy banks 0.040 Cobble bottom with clean sides

C. Rock lined

0.035 Smooth and uniform

0.040 Jagged and irregular

D. Channels not maintained (weeds and brush uncut)

0.050 Dense weeds (high as flow depth)

0.050 Clean bottom (brush on sides)

0.100 Dense brush (high stage)

0.200 Water quality swales (mowed regulary) NATURAL STREAMS

0.029 Clean (straight no pools)

0.035 Clean (straight no pools with weeds and stones)

0.039 Clean (winding pools)

0.042 Clean (winding pools weeds and stones)

0.052 Clean (winding pools weeds and large stones)

0.065 Weedy (sluggish with deep pools)

0.112 Very weedy (sluggish with deep pools)

SCS RUNOFF CURVE NUMBERS (CN) For Selected Land Uses

0

()

			RUNOF	F CURVE	NUMBE	RS (CN)
			BY HY	DROLOG	IC SOIL C	GROUP
LAND US	E DESCRIPTIC)N	Α	в	С	D
Cultivated land (1):	winter condition	n	86	91	94	95
Mountain open areas:	low growing b	rush and grasslands	74	82	89	92
Meadow or pasture:			65	78	85	89
Wood or forest land:	undisturbed o	r older second growth	42	64	76	81
Wood or forest land:	young second	growth or brush	55	72	81	86
Orchard	with cover cro	p	81	88	92	94
Open spaces, lawns, parks, go	f courses, cem	etaries, landscaping				
Good condition:	grass cover o	n 75%	68	80	86	90
	or more of th	e area				
Fair condition:	grass cover o	n 50%	77	85	90	92
	to 75% of the	area				
Gravel roads and parking lots			76	85	89	91
Dirt Roads and parking lots			72	82	87	89
Impervious surfaces:	pavement, roo	of, etc.	98	98	98	98
Open water bodies:	lakes, wetland	ls, ponds, etc.	100	100	100	100
Single Family Residential (2)						
Dwelling Unit/Gross Acre (DU/	GA)	% Impervious (3)				
1.0		15				
2.0		25				
3.0		34	Select	separate	curve nui	nbers
4.0		42	for the	pervious	and impe	rvious
5.0		48	portion	of the sit	e or basir	3.
6.0		52				
7.0		56				
Planned unit developments, co	ndominiums,	Use actual				
apartments, commercial busine	sses	impervious area.				
and industrial areas.						
(1) Detailed information relating	g to specific ag	ricultural land uses is a	vailable in th	e Nationa	al Engine	ering
Engineering Handbook, Se	ction 4, Hydrol	ogy, chapter 9, August 1	1972.		-	-

(2) Assume site drains to storm system.
 (3) For this land use, the remaining pervious areas are assumed to be lawn in good condition.

		(\bigcirc
		(
Project N	lame:	Jacoby	Heights	Subdivision	
PRE-DE	VELOPE	D - TIM	E OF COM	ICENTRATION C	ALCULATIONS
ob #	17-199				
Date:	2/5/2018				
				48.0	= Total Tc (min)
Overland Flo	ow (max 300)' total)		total	1
TC =	17.0	17.6		34.6	= travel time for less than 300' (min)
	110	190		300	1= Manning's coefficient (<u>sneet flow</u>)
P2 =	3.5	3.5			= 2-vear. 24 hour rainfall (in)
So =	3.90%	10.60%			= slope of the land (%)
Shallow Cor	ncentrated F	low (after	initial 300')	total	
T =	1.7	7.2	4.4	13.3	= travel time for sheet flow (min)
L = So =	14 00%	4 50%	8 10%	620	= now length (π) = slope of the land (%)
k=	3	3	3		= time of concentration velocity factor (ft/s)
	Ť,			I	
Flow in Swa	lles			total	
Tc =	0.00			0.0	= travel time in swale (min)
A =	6.00				= area of flow (sf)
<u> </u>	0.59				j= hydraulic radius (it)
0 =	4.12			<u> </u>	= side slope wet (ii) = ouantity of flow (ft/3/sec)
V =	0.52				= velocity
L =	0			0	= flow length (ft)
Ve =	a 1				= vertical distance of side
Ho =	4				= horizontal distance of side
Bw =	24				= base width of swale (in)
<u> </u>	1.00%	- 1			= depth of flow ESTIMATE (In)
n=	0.2			· · · · · ·	= Manning's coefficient (channel)
				I	
Flow in Gut	ters			total	
Tc =	0.0			0.0	= travel time in gutter (min)
tps =	0.02]= average velocity of flow (ft/sec)
0c =	0.0				- calculated width of now in the gutter (it) = quantity of flow (as caluclated O=CIA) (ftA3/sec)
C =	0.90				= runoff coefficient for rational method (paved=0.9)
=	2.75				= rainfall intensity (assume 5 min tc)
W =	18.00				= width of pavement draining to CB
S =	8.00%				= street longitudinal slope (%)
Sx =	2.50%			<u></u>	= street cross slope (%)
n = =	0.010				<pre> = wammy s coemcent (<u>pavement</u> = 0.0 (6) = length of flow and drainage basin (ff)</pre>
	0.0				- iengin ei new and drainage basin (it)
Flow in Pipe	es			total	
Tc =	0.1			0.1	= travel time in pipe (min)
V =	7.03				= calculated velocity pipe full (ft/sec)
Q =	12.41				= quantity of flow (ft^3/sec)
n =	0.011			<u> </u>	_= Maπning's coefficient (<u>pipe</u>)
<u> </u>	18			<u> </u>	- pipe diameter (in)
S =	1 11 1 17/4 1				

		1				S-2
Project N	Name:	lacoby	Heights	Subdiv	sion	
DEVELO	PED - TI	MEOFO	ONCENT	RATION	CALCULA	TIONS
Job #	17-199				ONLOULN	
Date:	2/5/2018					
					5.8 =	Total Tc (min)
Overland Fl	low (max 300	' total)			total	
IC =	0.011				0.4 = 1	ravel time for less than 300' (min)
L =	20				20 = f	low length (ft)
P2 =	3.5			1	= 2	2-year, 24 hour rainfall (in)
So =	1.00%				= 9	slope of the land (%)
Shallow C-	nontrated F		initial 2009		tetel	
T =	ncentrated F	iow (aner	initial 300')			ravel time for sheet flow (min)
L=	0				0 = f	low length (ft)
So =	2.00%				= 9	slope of the land (%)
k =	1				= t	ime of concentration velocity factor (ft/s)
Flow in Sw					intel.	
Tc =	0.00					ravel time in swale (min)
A =	0.01				= 8	area of flow (sf)
R =	0.04				=	nydraulic radius (ft)
Ls =	0.12				= 5	side slope wet (ft)
Q =	0.00				= (quantity of flow (ft^3/sec)
L =	0.02		—			low length (ft)
Ve =	1				= \	vertical distance of side
Ho =	1				1=	norizontal distance of side
Bw =	1				=t	pase width of swale (in)
 	1 00%					depth of flow ESTIMATE (in)
n=	1.0078					Manning's coefficient (channel)
Flow in Gut	ters				total	
fne =	3.4				<u>3.4</u> = t	ravel time in gutter (min)
T =	2.9				= ;	average velocity of flow in the cutter (6)
Qc =	0.09				=	quantity of flow (as caluclated Q=CIA) (ft^3/sec)
C =	0.90				= r	unoff coefficient for rational method (paved=0.9)
10/-	1.73				= 1	ainfall intensity (assume 5 min tc)
vv = S =	0.50%				= \	wroth or pavement draining to CB
Sx =	2.50%		 +			street cross slope (%)
n =	0.016				=	Manning's coefficient (pavement = 0.016)
L =	175.0		1		175 = 1	ength of flow and drainage basin (ft)
Elou la Dia					Antal	
To =	0.8	0.4	0.3	0.4		ravel time in pipe (min)
V =	9.07	10.64	10.15	12.83	1.9 - 1	calculated velocity pipe full (ff/sec)
Q =	7.12	8.35	7.96	10.07	= 0	quantity of flow (ft^3/sec)
n =	0.013	0.013	0.013	0.013	= 1	Manning's coefficient (<u>pipe</u>)
D=	12	12	12	12	= r	pipe diameter (in)
3=	4.00%	273.0	162.0	201 0	203 - 1	slope of pipe (%)
	4000	213.0	102.0	431.0	093 =	endra or bibe (it)

Project Name: Jacoby Heights Subdivision Hydrograph Analysis Summary Job # 17-199 Date: 2/5/2018

Rainfall	Rainfall		Pre-Develop	bed	Developed			Note: The hydrogenetic
year)	(inches	}	Pervious		Pervious		-	Note The hydrographs
2		1.50	Area =	9.42 acres	Area =	7.02 a	cres	Shown are based on the
5	i 4	1.50	CN =	82 na	CN =	86 na	B	storm using the CRUH
10		08.4	Impervious		Impervious			method based on the King
25	8 1	5.50	Area =	0.26 acres	Area =	2.66 a	cres	County Model
100) (00.0	CN =	98 na	CN =	96 na	a	County model
		a.a. 1	Tc =	48 min	Tc =	5.8 m	in	
			Total A =	9.68 acres	Total A =	9.68 a	cres	
					and the second se		A	

Ô

()

	P	re-Develop	ed Hydro	oraphs			Develope	d Hydroa	raphs		
Year		2	5	10	25	100	2	5	10	25	10
apeak	cfs =>	2.38	3.71	4.13	5.12	0.00	6.21	8.75	9.52	11.33	0.
Volume	cf =>	63,452	93,363	102,635	124,657	-	84,978	117,658	127,619	151,060	
Tpeak	min =>	490	490	490	490	10	470	470	470	470	1
peak	hr =>	8.17	8.17	8.17	8.17	0.17	7.83	7.83	7.83	7.83	0,
lydrogra	aph Name=>	2	5	10	25	100	2	5	10	25	1
ime	Time	Hyd	Hyd	Hyd	Hyd	Hyd	Hyd	Hyd	Hyd	Hyd	Н
min)	(hr)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	(0
	0 0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00	0.00	0.00	
	10 0.17	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
1	20 0.33	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	0.00	
	30 0.50	0.00	0.00	0.00	0.00	0.00	0.00	0.01	0 01	0.02	
	40 0.67	0.00	0.00	0.00	0.00	0.00	0.01	0.03	0.04	0.07	
	50 0.83	0.00	0.00	0.00	0.00	0.00	0.03	0.07	0.08	0.11	
1	60 1.00	0.00	0.00	0.00	0.01	0.00	0.05	0.10	0.11	0.15	
	70 1.17	0.00	0.01	0.01	0.01	0.00	0 07	0.12	0,14	0 18	
- 9.	80 1.33	0.00	0.01	0.01	0.01	0.00	0.09	0.14	0,16	0.20	
	90 1.50	0.00	0.01	0.01	0.01	0.00	0.10	0.16	0.18	0.22	
1	00 1.67	0.01	0.01	0.01	0.01	0.00	0,11	0.17	0.19	0.24	
1	10 1.83	0.01	0.01	0.01	0.02	0.00	0.14	0.21	0.23	0.28	
1	20 2.00	0.01	0.01	0.02	0.02	0.00	0.17	0.25	0 27	0.33	
1	30 2.17	0.01	0.02	0.02	0.02	0.00	0.18	0.26	0.29	0.34	
1	40 2.33	0.01	0.02	0.02	0 02	0.00	0.19	0 27	0 30	0 36	
1	50 2.50	0.01	0.02	0.02	0.03	0.00	0.20	0.28	0.31	0.38	
1	60 2 67	0.01	0.02	0.02	0.03	0.00	0.20	0.29	0.32	0.42	
1	70 2.83	0.02	0.02	0.03	0.03	0.00	0.23	0.33	0.37	0.51	
1	80 3.00	0 02	0.03	0.03	0.03	0.00	0.26	0.38	0.44	0.60	
1	90 3.17	0.02	0.03	0.03	0.04	0.00	0.27	0.43	0.49	0.66	
2	00 3.33	0 02	0.03	0.03	0.06	0.00	0 27	0.47	0.54	0.71	
2	10 3.50	0.02	0.03	0.04	0.08	0.00	0.29	0.50	0.58	0.76	
2	20 3.67	0.02	0.04	0.05	0.11	0.00	0.31	0.54	0.62	0.80	
2	30 3.83	0.02	0.05	0.07	0.14	0.00	0.36	0.62	0.71	0.91	
2	40 4 00	0.03	0.07	0.10	0 19	0.00	0.42	0 71	0.80	1.03	
2	50 4.17	0.03	0.09	0.13	0.23	0.00	0.45	0.75	0.85	1.09	
2	60 4.33	0.03	0.12	0.16	0.28	0 00	0.48	0.79	0.89	1.13	
2	70 4.50	0.04	0 15	0.20	0.33	0.00	0.51	0.83	0.93	1.18	
2	80 4.67	0.05	0.19	0.24	0.38	0.00	0.54	0.86	0.97	1 22	
2	90 4 83	0.07	0.23	0.29	0.45	0.00	0.61	0.97	1.08	1 36	
3	00 5.00	0.09	0.27	0.34	0.52	0.00	0.69	1.09	1.21	1.51	
3	10 5.17	0.12	0 32	0.40	0.59	0.00	0.73	1 13	1 26	1.57	
3	20 533	0.14	0.37	0.45	0.65	0.00	0.76	1 17	1.30	1.61	
3	30 5.50	0.17	0.42	0.50	0.72	0.00	0.70	1.20	1.34	1.65	
	40 5.57	0.20	0.47	0.50	0.78	0.00	0.91	1 24	1 37	1 60	
3	50 5.83	0.20	0.52	0.62	0.86	0.00	0.01	1.37	1.51	1.86	
3	60 6 00	0.24	0.52	0.02	0.00	0.00	1.00	1.50	1.66	2.04	
2	70 617	0.20	0.05	0.05	1 0.4	0.00	1.04	1 55	1 71	2.04	
2	80 6.22	0.32	0.00	0.00	1.10	0.00	1.04	1.00	1.75	2 43	
2	00 6.53	0.40	0.71	0.80	1.14	0.00	1.00	1,00	1.79	2.13	
-	00 6.50	0.40	0.77	0.09	1,13	0.00	1,09	1.02	1.70	2.17	
	10 6.07	0.50	0.03	1.00	1 20	0.00	1.14	2.00	1.01	2.20	
5	20 7.00	0.50	1.00	1.00	1.39	0.00	1,30	2.00	2,19	2.00	
4	20 7.00	0.59	1.00	4 3 4	1.3/	0.00	1.03	2.30	2 01	3.13	
1	40 7.17	0.07	1.10	1.34	1.74	0.00	1,69	2.40	2.09	3.24	
4	7.33	0.79	1.35	1.03	1,97	0.00	2.03	2.92	3.20	3.65	
9	50 7.50	0.93	1.5/	1.//	2.27	0.00	2.40	3.44	3.75	4.50	
4	7.67	1.21	2.00	2.25	2.85	0.00	3 54	5 04	5,50	6.57	
4	7.83	1.83	2.94	3.29	4.12	0.00	6.21	8./5	9.52	11.33	
4	80 8.00	2.32	3 66	4.08	5 07	0.00	6 11	8.55	9,29	11.02	
4	90 8.17	2.38	3.71	4.13	5.12	0.00	3 69	5,13	5.57	6.59	
5	8.33	2.29	3.54	3.93	4 86	0.00	2.57	3.56	3.85	4.55	
5	10 8.50	2.17	3.33	3,70	4.56	0.00	2.17	3.00	3 25	3.83	
5	20 8.67	2.07	3.17	3.51	4.32	0.00	2.16	2.97	3 22	3 7 9	

Hyd-base-2-ORIFICE.xls

		Pro Douolon	od bludes	arabe		
Year ==		2	ea Hyaro 6	10	25	100
Opeak	cfs =>	2.38	3.71	4.13	5.12	0.00
Volume	cf =>	63,452	93,363	102,635	124,657	
peak	min =>	490	490	490	490	10
peak	hr =>	8.17	8.17	8.17	8.17	0.17
iyorograpi	Time	Z	5 Liuni	10 Dual	25	100
min)	(hr)	(cfa)	(cfe)	(ofe)	(cfn)	Hyd (cfa)
530	883	195	2.96	3 28	(03)	(CIS)
540	9.00	1 79	2 30	3.00	3.68	0.00
550	9.17	1.67	2.52	2 78	3.40	0.00
560	9.33	1.57	2 36	2 61	3 18	0.00
570	9.50	1.50	2.24	2.46	3.00	0.00
580	9.67	1.44	2 14	2 35	2.86	0.00
590	9.63	1.39	2 06	2 26	2.75	0.00
600	10.00	1.35	2.00	2.19	2.66	0.00
610	10.17	1.32	1.95	2,14	2.59	0.00
620	10.33	1 30	1.91	2.10	2.53	0.00
630	10.50	1.29	1.88	2,06	2.49	0.00
640	10.67	1.28	1.86	2.04	2.46	0.00
650	10.83	1.25	1.82	1.99	2.39	0.00
660	11.00	1,21	1.75	1,91	2.30	0.00
670	11.17	1.17	1.70	1.86	2.23	0.00
680	11.33	1,15	1 65	1 81	2.17	0.00
690	11.50	1.13	1.62	1.77	2.12	0.00
700	11.57	1.11	1 60	1.74	2.09	0.00
710	11.83	1,10	1.58	1.72	2.06	0.00
720	12.00	1.09	1.56	1.70	2.04	0.00
730	12.17	1.09	1.55	1.69	2.02	0.00
740	12.33	1,08	1.54	1.68	2.01	0.00
750	12 50	1.08	1.54	1.68	2.00	0.00
760	12.57	1.08	1.53	1.67	1.99	0.00
770	12.83	1.00	1.50	1.64	1,95	0.00
700	13.00	0.00	1.40	1,50	1.00	0.00
800	13 33	0.95	1 37	1.33	1.02	0.00
810	13 50	0.95	1.34	1.46	1.73	0.00
820	13.67	0.93	1.32	1 43	1 70	0.00
830	13.83	0.92	1.30	1 4 1	1.68	0.00
840	14.00	0.91	1.29	1.40	1.66	0.00
850	14.17	0.91	1.28	1.39	1.65	0.00
860	14.33	0.90	1.27	1,38	1.63	0.00
870	14.50	0.90	1.26	1.37	1.63	0.00
880	14.67	0.90	1.26	1.37	1.62	0.00
890	14.83	0.88	1.24	1.35	1,60	0.00
900	15.00	0.87	1.21	1.32	1.56	0.00
910	15.17	0.85	1.19	1.29	1.53	0.00
920	15.33	0.84	1.17	1.27	1.51	0.00
930	15.50	0.83	1.16	1.26	1.49	0.00
940	15.67	0.82	1.15	1.24	1.47	0.00
950	15.83	0.82	1.14	1.23	1,46	0.00
960	16.00	0.81	1.13	1.23	1.45	0.00
970	16.17	0.81	1.13	1.22	1.44	0.00
980	16.33	0.81	1.12	1.22	1,44	0.00
990	18.50	0.80	1.12	1.21	1.43	0.00
1010	10.0/	0.50	1.12	1.21	1.43	0,00
1020	10.03	0,79	1.09	4.4.4	1.40	0.00
1030	17.17	0.70	1.00	1 11	1 33	0.00
1040	17.33	0.72	1.00	1.08	1.28	0.00
1050	17.50	0.71	0.98	1.05	1.25	0.00
1060	17.67	0.70	0.96	1.04	1.23	0.00
1070	17.83	0.69	0.95	1.03	1.21	0.00
1080	18.00	0.68	0.94	1.02	1.20	0.00
1090	18.17	0.67	0.93	1.01	1.19	0.00
1100	18.33	0.67	0.93	1.00	1.18	0.00
1110	18.50	0.67	0.92	1.00	1.17	0.00
1120	18.67	0.66	0.92	0.99	1.17	0.00
1130	18.83	0.66	0.91	0.99	1.17	0.00
1140	19.00	0.66	0.91	0 99	1.16	0.00
1150	19,17	0.66	0,91	0,98	1.16	0.00
1160	19.33	0.66	0.91	0.98	1.16	0.00
1170	19.50	0.66	0.91	0.98	1.16	0.00
1180	19.67	0.66	0,91	0.98	1.16	0.00
1190	19.83	0.66	0,91	0.98	1.16	0.00
1200	20.00	0.66	0.91	0.98	1,16	0.00
1210	20,17	0.66	0.91	0.98	1.16	0.00
1220	20.33	0.66	0.91	0.98	1,16	0.00
1230	20.50	0.66	0.91	0.98	1.16	0.00

0

Hyd-base-2-ORIFICE_xis

Pre-Developed Hydrographs											
Year		=>	2	6	10	25	100				
Opeak		cfs =>	2.38	3.71	4.13	5.12	0.00				
Volume		cf =>	63,452	93,363	102,635	124,657					
Tpeak	F	nin =>	490	490	490	490	10				
Tpeak		hr =>	8.17	8.17	8.17	8.17	0.17				
Hydrog	raph Nan	16=>	2	5	10	25	100				
Time	Time		Hyd	Hyd	Hyd	Hyd	Hyd				
(min)	(hr)		(cfs)	(cfs)	(cfs)	(cfs)	(cfs)				
12	240	20.67	0.66	0 91	0.98	1.16	0.00				
12	250	20.83	0.66	0.91	0.98	1,16	0.00				
12	260	21.00	0.66	0.91	0.98	1.16	0.00				
12	270	21.17	0 66	0.91	0.98	1.16	0.00				
12	280	21.33	0.66	0.91	0.99	1.16	0.00				
12	290	21.50	0.66	0.91	0.99	1.16	0.00				
13	300	21.67	0.67	0.91	0.99	1,16	0 00				
10	10	21.83	0.67	0.91	0.99	1.16	0.00				
10	320	22.00	0.67	0.91	0.99	1.16	0.00				
10	330	22.17	0.67	0.92	0.99	1.16	0.00				
13	340	22.33	0.67	0.92	0.99	1.16	0.00				
13	350	22.50	0.67	0.92	0 99	1.16	0.00				
13	360	22.67	0.67	0.92	0.99	1.16	0 00				
1:	370	22 83	0.67	0.92	0.99	1.16	0.00				
13	380	23 00	0.67	0.92	0 99	1.17	0.00				
13	390	23.17	0.67	0.92	0.99	1.17	0.00				
14	400	23.33	0.67	0.92	0.99	1.17	0.00				
14	410	23.50	0.67	0.92	1.00	1.17	0.00				
1.	420	23.67	0.67	0.92	1.00	1.17	0.00				
14	430	23.83	0.68	0 92	1.00	1.17	0.00				
14	440	24.00	0.68	0.92	1.00	1.17	0.00				
14	450	24.17	0.61	0 84	0.90	1 06	0.00				
14	460	24.33	0.50	0.68	0.73	0.86	0.00				
14	470	24.50	0.40	0.55	0.59	0.70	0.00				
1	480	24.67	0.33	0.45	0.48	0.57	0.00				
1.	490	24.67	0.27	0.36	0.39	0.46	0.00				
1	500	24.67	0.22	0.29	0.32	0.37	0,00				

()

Developed Hydrographs 2 5 10 5.21 8.75 9.52 25 100 9.52 11.33 0.00 11.33 151.060 470 7.83 25 Hyd (cfs) 1.23 1.23 1.23 117,658 470 7.83 5 127.619 470 7.83 10 84.978 470 7.83 2 10 0.17 Hyd (cfs) 0 00 0 00 0 00 0 00 0 00 Hyd (cfs) 0.74 0.74 Hyd (cfs) 0.98 0.99 Hyd (cfs) 1.06 1.06 0 99 0 99 0 99 1.06 1.06 1.06 0.99 0.99 0.99 0.99 0.99 0.99 0.99 0.99 0.53 0.04 0.00 1.06 0.00 1.06 1.07 1.07 0.00 0.00 0.00 0.57 0.04 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00 0.00

0.00

0.00

0.00

0.00

0

Hyd-base-2-ORIFICE.xls

\bigcirc

	Street.	
1		
1		
. \		
	Sec. 1	

Pre-Developed Hydrographs								
Year		2	5	10	25	100		
Qpeak	cfs =>	2.38	3.71	4.13	5.12	0.00		
Volume	d =>	63,452	93.363	102.635	124 657			
Tpeak	min =>	490	490	490	490	10		
Tpeak	hr =>	8.17	8.17	8 17	8.17	0.17		
Hydrogr	aph Name=>	2	5	10	25	100		
Time	Time	Hyd	Hyd	Hyd	Hyd	Hyd		
(min)	(hr)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)		

Develope	d Hydrog	raphs		
2	5	10	25	100
6,21	8.75	9.62	11.33	0.00
84,978	117.658	127.619	151.060	
470	470	470	470	10
7.83	7.83	7.83	7.83	0.17
2	5	10	25	100
Hyd	Hyd	Hyd	Hyd	Hvd
(cfs)	(cfs)	(cfs)	(cfs)	(cfs)



Hyd-base-2-ORIFICE xis

\bigcirc

0

	F	Pre-Develop	ed Hydro	graphs		
Year		2	5	10	25	100
Opeak	cfs =>	2.38	3.71	4,13	5.12	0.00
Volume	cf =>	63,452	93,363	102.635	124,657	
Tpeak	min =>	490	490	490	490	10
Tpeak	hr =>	8.17	8.17	8.17	8.17	0.17
Hydrogr	aph Name=>	2	5	10	25	100
Time	Time	Hyd	Hyd	Hyd	Hyd	Hyd
(min)	(hr)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)



Hyd-base-2-OR FICE xls

Ó



Pre-Developed Hydrographs							
Year		2	5	10	25	100	
Opeak	cfs =>	2.38	3.71	4.13	5.12	0.00	
Volume	cf =>	63,452	93,363	102,635	124,657		
Tpeak	min =>	490	490	490	490	10	
Tpeak	hr =>	8.17	8.17	8.17	8.17	0.17	
Hydrogr	aph Name=>	2	5	10	25	100	
Time	Time	Hyd	Hyd	Hyd	Hyd	Hyd	
(min)	(hr)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)	

2	5	10	25	100
6.21	8.75	9.52	11.33	0.00
84 978	117.658	127.619	151,060	
470	470	470	470	10
7.83	7.83	7.83	7.83	0.17
2	5	10	25	100
Hyd	Hyd	Hyd	Hyd	Hyd
(cfs)	(cfs)	(cfs)	(cfs)	(cfs)



Hyd-base-2-OR/FICE xis





	P	re-Develop	ed Hydro	graphs		
Year	======>	2	5	10	25	100
Opeak	cfs =>	2.38	3.71	4.13	5.12	0.00
Volume	cf =>	63,452	93,363	102.635	124,657	
Tpeak	min =>	490	490	490	490	10
Tpeak	hr =>	8.17	8 17	8.17	8.17	0.17
Hydrogra	aph Name=>	2	5	10	25	100
Time	Time	Hyd	Hyd	Hyd	Hyd	Hyd
(min)	(hr)	(cfs)	(cfs)	(cfs)	(cfs)	(cfs)

evelope	d Hydrog	raphs		
2	5	10	25	100
6.21	8.75	9.52	11.33	0.00
84,978	117,658	127.619	151,060	-
470	470	470	470	10
7.83	7.83	7.83	7.83	0.17
2	5	10	25	100
Hyd	Hyd	Hyd	Hyd	Hyd
(cfs)	(cfs)	(cfs)	(cfs)	(cfs)



Hyd-base-2-ORIFICE xis

	Jacoby Hei	ghts Subdivi	sion		1.0
Detention System Summary					
lob #	17-199				
Jate:	2/5/2018	The detention sys	tem design i	e based on t	ha Kina
I) Detention Facility Design Input:	Count	ty Model "Facility I	Design Routi	ne".	ne King
2) Type of facility:		DETENTION PO	DND		
b) Pond side slopes () Pond storage denth:		3 to 1	of sound to	august and	
5) Vertical permeability		0 min/in	or pond to	overnow)	
6) Number of orifices:		2			
7) Riser dia. =>	1	2 in 2 (hubically 0.62)			
3) IE - bottom orifice:	0.8	0 ft (distance belo	w bottom of	oond - Negal	ive #)
0) Max Q Bottom Orif. #1	3.1	8 cfs			
11) Top Orif #2 Height =	3.1	7 ft	0-15		
13) Mid Orif #3 Height =	0.0	0 ft	Onlice not a	being used	
Detention Facility Design Results			Crinica nor i	senig used	
Performance	Developed	Pre-Developed	Actual	Peak	Storage
year	Inflow	Outflow	Outflow	Stage	
100	CIS	Cts	Cfs	<u>ft</u>	Cf
25	11.33	5.12	5.12	4.00	21,793
10	9.52	4.13	4.12	3.47	18,058
2	8.75 6.21	3.71	3.51	3.26	16,669
		2.00	Required S	torage ====	21,793
	Bottom Orif	Middle Orif.	Top Orif.	Optional W	eir Desian
Fotal Q =	3.18	0.00	1.94	(for top orifi	ce)
tead (π) = Dist_from bottom of pond (ft) =	4.00	0.00	0.83	1.09	La (ft)
Orif. Dia. (in) =	7.65	0.00	8.85	Weir is an o	option
FLOW CONTROL STRUCTURE SCH	EMATIC		1 1 1 1 1	8	
		+	12	(in) Riser di	a.
davimum water surface elevation	100000				
					£ #7
4.0		9.0	8.85	(in) Dia. Ori (cfs) Max O	ton Orif #2
4.0 Storage depth or tank dia.	(fi)	-0-	8.85 1.94	(in) Dia. Ori (cfs) Max C	top Orif #2
4.0 Storage depth or tank dia.	(ft)		8.85 1.94 NA	(in) Dia. Ori (cfs) Max C (in) Dia. Ori	f #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft)	(fi) 3.17		8.85 1.94 NA NA	(in) Dia. Ori (cfs) Max Q (in) Dia. Ori (cfs) Max Q	f #3 Mid Orif #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft) Middle Orif #3 Height (ft)	(fi) 3.17	 	8.85 1.94 NA NA	(in) Dia. Ori (cfs) Max Q (in) Dia. Ori (cfs) Max Q	f #3 Mid Orif #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft) Middle Orif #3 Height (ft)	(fi) 3.17		8.85 1.94 NA NA	(in) Dia. Ori (cfs) Max Q (in) Dia. Ori (cfs) Max Q	f #3 Mid Orif #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft) Middle Orif #3 Height (ft) Bottom of pond / tank	(fl) 3.17		8.85 1.94 NA NA	(in) Dia. Ori (cfs) Max Q (in) Dia. Ori (cfs) Max Q	f #3 Mid Orif #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft) Middle Orif #3 Height (ft) Bottom of pond / tank Bottom Orif depth below pond / tank (ft)	(fi) 3.17 NA		8.85 1.94 NA NA	(in) Dia. Ori (cfs) Max C (in) Dia. Ori (cfs) Max C	f #2 f #3 Mid Orif #3
4.0 Storage depth or tank dia. Top Orif #2 Height (ft) Middle Orif #3 Height (ft) Bottom of pond / tank Bottom Orif depth below pond / tank (ft)	(ft) 3.17 NA		8.85 1.94 NA NA 7.65	(in) Dia. Ori (cfs) Max C (in) Dia. Ori (cfs) Max C	f#1

Detention Summary Page 1

Ò

Hyd-base-2-ORIFICE xls



All Storm Hydrographs Routed Through The Detention Facility

Hyd-base-2-ORIFICE xls

Detention Summary Page 2







Project Name: Jacoby Heights Subdivision Stage Storage Summary Job # 17-199 Dete: 2/5/2018

()



()

Hyd-base-2-ORIFICE.xls

Stage Storage Summary Page 1

Stage ft Discharge Storage cf cfs 3.30 16,924,44 3.65 3 35 17,256.03 3.81 3.40 3.45 17,590.07 3 95 4 08 3.50 18,265.52 4.20 3.55 18,606.96 4 31 3.60 3.65 3.70 18,950.88 4.41 19,646.20 4.61 3.75 3.80 3.85 19,997.62 4 70 4 79 4 87 4 96 20,351.57 20,708.04 3,90 21,067.05 3.95 4.00 4.05 21,428.60 21,792.71 22,159.39 5.04 5 12 5 19 4.10 22,528.64 5.27 4.15 4.20 4.25 22,900.47 5.34 5.42 5.49 23,274 89 23,651 92 4.30 24,031.55 5 56 4.35 4.40 4.45 4.50 24,413 80 24,798 68 25,186 20 5 62 5 69 5 76 25,576 36 5 82 4.50 4.50 25,576.36 5 82 25,576 36 25,576 36 5 82 5 82 4.50 4.50 25,576.36 5.82 4.50 4.50 25,576,36 25,576,36 25,576,36 5.82 5 82 5 82 4.50 25,576.36 5 82 25,576.36 25,576.36 25,576.36 4.50 4.50 5.82 5 82 5 82 4,50 4.50 25 576 36 5.82 4.50 4.50 25 576 36 25 576 36 5,82 5.82 5.82 4.50 25 576 36 4.50 25 576 36 5.82 4.50 4.50 4.50 25 576 36 25 576 36 5.82 5.82 5.82 25 576 36 4.50 25 576 36 5.82 5 82 5 82 5 82 4.50 25 576 36 25 576 36 4.50 25 576 36 4.50 4.50 4.50 25 576 36 5,82 25 576.36 25 576.36 5.82 5.82 4.50 25 576.36 5.82 4 50 4 50 4 50 4 50 25 576.36 5.82 25,576.36 25,576.36 5.82 5.82 4.50 25 576.36 5.82 4.50 25,576.36 25,576.36 25,576.36 5.82 5.82 4.50 5.82 25,576.36 5 82 4.50 4.50 25,576.36 5.82 25 576.36 25 576.36 5.82 5.82 4.50 4.50 25,576.36 5.82 4 50 4 50 25,576.36 25,576.36 5.82 5.82 5.82 4.50 25,576.36 4 50 25,576.36 5.82 4 50 4 50 25,576.36 5 82 25,576,36 5.82 4.50 25 576 36

Ó

Ô

Hyd-base-2-ORIFICE_xis

4.50

25,576,36

5.82

Stage Storage Summary Page 2

Stage	Storage	Discharge
ft	cí	cfs
4.50	25 576 36 25 576 36	5.82 5.82
4.50	25 576 36	5 62
4.50	25,576.36	5.82
4.50	25 576 36	5.82
4.50	25 576 36	5.82
4.50	25,576.36	5 82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576,36	5.82
4 50	25,576.36	5.82
4.50	25,576.36	5 82
4.50	25,576.36	5.82
4.50	25 576 36	5 82
4.50	25,576.36	5,82
4.50	25,576 36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.62
4 50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576,36	5.82
4.50	25,576 36	5.82
4.50	25 576 36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576 36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5.82
4 50	25,576.36	5.62
4 50	25,576.36	5.82
4 50	25,576.36	5,82
4 50	25,576,36	5.82
4.50	25,576 36	5.82
4.50	25,576.36	5.82
4.50	25,576,36	5 82
4.50	25,576,36	5.82
4,50	25,576.36	5.82
4 50	25,576,36	5.82
4 50	25,576.36	5.82
4 50	25,576.36	5.82
4.50	25,576.36	5.82
4.50	25,576,36	5,82
4.50	25,576,36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5_82
4.50	25,576 36	5.82
4.50	25,576.36	5.82
4.50	25,576.36	5 82
4 50	25,576,36	5 82
4 50	25,576.36	5.82
4 50	25,576.36	5.82
4 50	25,576.36	5.82
4 50	25,576,36	5 82
4.50	25,576,36	5 82 5 82

Ò

 \bigcirc

Hyd-base-2-ORIFICE.xls

Stage Storage Summary Page 3

 \bigcirc

Hyd-base-2-ORIFICE xis

Stage Storage Summary Page 4

Ó
Project Name: Jacoby Heights Subdivision Rectangular, Sharp Crested Weir Calculations Job # 17-199 Date: 2/5/2018

Weir Equation: $Q = C(L-0.2H)H^{3/2}$

- Q = Flow over weir (cfs)
- C = 3.27 + 0.40 H/P (ft)
- L = Adjusted length of weir (La 0.1H x 2) this is to account for side constraints
- La = Actual length of weir along pipes interior circumference (ft)
- H = Distance from bottom of weir to maximum head (ft)
- P = Distance from bottom of weir to outfall invert elevation (ft)
- D = Inside riser pipe diameter (in)
- < = Angle of opening for weir (maximum 180 degrees)

Given:

Q	1.94	cfs
Н	0.83	ft
Р	3.17	ft
D	12	in

Find:

C	3.37	ft
L	0.92	ft
La	1.09	ft
<	125	degrees



H

P





Table of Contents

0

0

le

secutive Summary	l
troduction	2
Project and Location Description	2
Vicinity Streets	2
Study Intersections	3
Traffic Counts	3
e Trips	6
Trip Generation	6
Trip Distribution	6
perational Analysis	9
Background Volumes	9
Background Volumes plus Site Trips	5
Intersection Capacity Analysis	2
fety Analysis	4
Crash Data Analysis	4
Sight Distance Analysis	5
Warrant Analysis	5
nclusions	5
pendix	7

Table of Figures

Figure 1: Vicinity Map	Ļ
Figure 2: Existing Conditions	;
Figure 3: Site Trip Distribution and Assignment	3
Figure 4: Year 2020 Background Conditions)
Figure 5: Year 2020 Background Plus Site Trips Conditions	

Table of Tables

Table 1: Trip Generation Summary	(
Table 2: Intersection Capacity Analysis Summary	13

Jacoby Road Subdivision - Traffic Impact Analysis

ü



Executive Summary

- 1. A subdivision is proposed that will include the construction of 31 single-family detached houses located at 19124 SE Jacoby Road in Sandy, Oregon.
- 2. The trip generation calculations show that the proposed development is projected to generate an additional 23 morning peak hour and 31 evening peak hour site trips. Trip generation was based on a previously submitted site plan that showed 32 houses, and should be considered conservative.
- 3. All study intersections are currently operating acceptably per City of Sandy and ODOT standards and are projected to continue operating acceptably upon build-out of the proposed development through year 2020. No operational mitigation is necessary or recommended.
- 4. A crash pattern was identified at the intersection of OR-211 at Dubarko Road. The planned turn lanes and signal at the intersection outlined in the Sandy TSP will help to mitigate this crash pattern. No safety mitigation is recommended as part of this development.
- 5. Adequate sight distance is available at both proposed site access locations.
- 6. Consistent with improvements outlined in the City's TSP, left-turn lane warrants are met under existing conditions for the intersection of OR-211 at Dubarko Road.
- 7. Preliminary traffic signal warrants are not projected to be met for the intersection of OR-211 at Dubarko Road, even with the completion of the proposed development.

Jacoby Road Subdivision — Traffic Impact Analysis



Introduction

A subdivision is proposed that will include the construction of 31 single-family detached houses located at 19124 SE Jacoby Road in Sandy, Oregon. This report addresses the traffic impacts of the proposed development on the nearby street system. Based on correspondence with John Replinger, the City's consulting transportation engineer, and City of Sandy staff, the report conducts safety and capacity/level-of-service analyses at the following intersections:

- Eagle Creek-Sandy Highway No. 172 (OR-211) at Dubarko Road;
- SE Jacoby Road at Dubarko Road;
- Newton Street at SE Jacoby Road (south site access); and
- Street B at SE Jacoby Road (proposed north access intersection).

The purpose of this study is to determine whether the transportation system within the vicinity of the site is capable of safely and efficiently supporting the existing and proposed uses and to determine any mitigation that may be necessary to do so. Detailed information on traffic counts, trip generation calculations, safety analyses, and level-of-service calculations is included in the appendix to this report.

Project and Location Description

The 31-lot subdivision is proposed for development at 19124 SE Jacoby Road in Sandy, Oregon. The subject site is located near the southern edge of city limits in a developing area, surrounded by low density houses to the east, west, and south. The Zion Meadows, Marshall Ridge, Mt. View Ridge, and Snowberry subdivisions to the west are currently under construction and were included as in-process trips in the site trips and capacity analysis sections.

Upon development of the site, the subdivision will take access to SE Jacoby Road via an eastern extension of Newton Street and a new site access between Newton Street and Cascadia Village Drive currently designated as Street B.

Vicinity Streets

OR-211 is classified by the City of Sandy as a Major Arterial and by the Oregon Department of Transportation (ODOT) as a District Highway. The roadway generally has a two-lane cross-section and has a posted speed of 45 mph within the site vicinity. Curbs, sidewalks, and bike lanes are not generally provided on either side of the roadway; however, an off-street path is provided along the western side of the roadway for a short distance north of Dubarko Road.

Dubarko Road is classified by the City of Sandy as a Residential Minor Arterial roadway between OR-211 and Jacoby Road and as a Minor Arterial to the east of Jacoby Road. The roadway has a two-lane cross-section



and a posted speed limit of 25 mph. Curbs and sidewalks are generally provided along both sides of the roadway. Bicycle lanes are present along both sides of the road in the site vicinity.

SE Jacoby Road is classified by the City of Sandy as a Collector. The roadway has a two-lane cross-section and a posted speed limit of 25 mph. Curbs and sidewalks are intermittently provided along both sides of the roadway.

Newton Street is classified by the City of Sandy as a Local Street. The roadway has a two-lane cross-section and a statutory residential speed limit of 25 mph. Curbs and sidewalks are provided along both sides of the roadways.

Study Intersections

The intersection of OR-211 at Dubarko Road is a four-legged intersection that is stop-controlled for the minor street eastbound and westbound approaches of Dubarko Road. The northbound approach has one shared lane for all turning movements. The southbound, eastbound, and westbound approaches have one shared through / left-turn lane, one right-turn lane, and a bicycle lane located between the two vehicle lanes. Crosswalks are unmarked at the intersection.

The intersection of Dubarko Road at SE Jacoby Road is a three-legged intersection that is stop-controlled for the northbound minor street approach of SE Jacoby Road. Each approach has one shared lane for all turning movements. The eastbound and westbound approach of Dubarko Road has bicycle lanes located to the right of the travel lane. Crosswalks are unmarked but available across all approaches.

The intersection of SE Jacoby Road at Newton Street is a three-legged intersection that is stop-controlled for the eastbound minor street approach of Newton Street. Each approach has one shared lane for all turning movements. Crosswalks are unmarked at the intersection.

A vicinity map displaying the project site, vicinity streets, and the study intersections with their associated lane configurations is shown in Figure 1 on page 4.

Traffic Counts

Traffic counts were conducted at the existing study intersections on Tuesday, May 1st, 2018, from 4:00 to 6:00 PM and on Wednesday, May 2nd, 2018, from 7:00 to 9:00 AM.

Per requirements established in ODOT's *Analysis Procedures Manual*, a seasonal adjustment factor of 1.05 was calculated for traffic along OR-211 based on the commuter seasonal trend. The adjustment factor was applied to the highway's through movement traffic volumes to reflect the 30th highest hour volumes along the ODOT facility.

Figure 2 on page 5 shows the existing morning and evening peak hour traffic volumes at the study intersections.





1e

Site Trips

Trip Generation

The proposed subdivision will construct 31 single-family houses while removing one existing house for a net increase of 30 houses. A prior site plan for the subdivision called for the construction of 32 single-family homes. Analysis within this report is based on the higher lot count.

To estimate the number of trips that will be generated by the proposed development, trip rates from the *Trip* Generation Manual, were used. Data from land-use code 210, Single-Family Detached Housing, was used to estimate the proposed development's trip generation based on the number of dwelling units.

The trip generation calculations show that the proposed development of 31 additional homes is projected to generate an additional 23 morning peak hour and 31 evening peak hour site trips. The trip generation estimates of the proposed development are summarized in Table 1 below. Detailed trip generation calculations are included in the technical appendix to this report.

	ITE Code	Size	Morni	ing Peak	Hour	Even	ng Peak	Hour	Weekday
	TIE Code	5120	Enter	Exit	Total	Enter	Exit	Total	Total
Existing Home	210	1 unit	0	1	1	1	0	1	10
Proposed Development	210	32 units	6	18	24	20	12	32	302
Net Increase		31 units	6	17	23	19	12	31	292

Table 1: Trip Generation Summary

Trip Distribution

The directional distribution of site trips to and from the project site was estimated based on locations of likely trip destinations, locations of major transportation facilities in the site vicinity, and existing travel patterns at study intersections.

The following trip distribution was estimated and used for analysis:

¹ Institute of Transportation Engineers (ITE), *Trip Generation Manual*, 10th Edition, 2017. Jacoby Road Subdivision — Traffic Impact Analysis



- Approximately 30 percent of site trips will travel to/from the south along OR-211;
- Approximately 25 percent of site trips will travel to/from the west along OR-211;
- Approximately 25 percent of site trips will travel to/from the west along Dubarko Road; and
- Approximately 20 percent of site trips will travel to/from the north along along OR-211.

For a conservative analysis of the study intersection of Jacoby Road at Dubarko Road, it was assumed that all site trips will travel north on Jacoby Road to the intersection rather than take Cascadia Village Drive to Bornstedt Road. Under existing and potentially future conditions, the site may be served by multiple routes/roadways to gain access to the greater transportation system, including Cascadia Village Drive, Newton Street, Village Boulevard, and Arletha Court.

The trip assignment for the site trips generated by the proposed development during the morning and evening peak hours are shown in Figure 3 on page 8.

Jacoby Road Subdivision - Traffic Impact Analysis



Operational Analysis

Background Volumes

To provide analysis of the impact of the proposed development on the nearby transportation facilities, an estimate of future traffic volumes is required.

In order to calculate the future traffic volumes for non-ODOT facilities, a compounded growth rate of two percent per year for an assumed build-out condition of two years was applied to the measured existing traffic volumes to approximate year 2020 background conditions.

To estimate the future traffic volumes for ODOT facilities, a linear growth rate of 3.2 percent per year was calculated for the traffic volumes along OR-211 using ODOT's 2035 Future Volume Tables. This growth rate was applied to the measured existing traffic volumes over a two-year period to determine year 2020 background traffic volumes for the through traffic traveling along OR-211. A compounded growth rate of two percent per year for an assumed build-out condition of two years was applied to all other turning movement traffic volumes.

In addition to the traffic volume growth described above, there are four in-process developments that are currently or will be approved for construction near the site vicinity and are expected to impact nearby study intersections. The in-process developments include the Zion Meadows, Marshall Ridge, Mt. View Ridge, and Snowberry subdivisions.

The four in-process developments are currently not fully contributing trips to the transportation system, but were assumed to be completed by the 2020 build-out year of the proposed development. Additional trips corresponding to each in-process development were added to the existing year traffic volumes in addition to the two years of traffic growth at each of the applicable study intersections.

Figure 4 on page 10 shows the projected year 2020 background traffic volumes at the study intersections during the morning and evening peak hours.

Background Volumes plus Site Trips

Peak hour trips calculated to be generated by the proposed development, as described earlier within the *Site Trips* section, were added to the projected year 2020 background traffic volumes to obtain the expected 2020 background volumes plus site trips.

Figure 5 on page 11 shows the projected year 2020 peak hour background traffic volumes plus proposed development site trips at the study intersections during the morning and evening peak hours.







Intersection Capacity Analysis

A capacity and delay analysis was conducted for each of the study intersections per the unsignalized intersection analysis methodologies in the *HIGHWAY CAPACITY MANUAL* (HCM)². According to the City of Sandy's Transportation System Plan (TSP), both signalized and unsignalized intersections are required to operate at level of service (LOS) D or better. The LOS of an intersection can range from LOS A, which indicates very little or no delay experienced by vehicles, to LOS F, which indicates a high degree of congestion and delay.

OR-211 is an ODOT facility located outside the Portland Metropolitan Region. According to Table 6 – *Volume to Capacity Ratio Targets Outside Metro* of the *Oregon Highway Plan* (OHP), intersections on District Highways with speed limits of 45 mph or more within an Urban Growth Boundary and not inside a Metropolitan Planning Organization are required to operate with a v/c ratio of 0.90 or less for the major-street approach. The volume to capacity (v/c) ratio is a measure that compares the traffic volume (demand) against the available capacity of an intersection. For the purposes of this analysis the highest calculated v/c ratio is reported, regardless of the intersection approach.

The intersection of OR-211 at Dubarko Road currently operates at LOS C with v/c ratios of 0.36 or less during both peak hours, and under all future analysis scenarios is projected to operate at LOS D with v/c ratios of 0.50 or less.

The intersection of Dubarko Road at SE Jacoby Road currently operates at LOS A during the AM peak hour and B during the PM peak hour. Under full build-out conditions it is projected to operate at LOS B during both peak hours. All analysis scenarios operate with a v/c ratio of 0.11 or less.

The site access intersections of Street B at SE Jacoby Road and Newton Street at SE Jacoby Road operate with low traffic volumes and little vehicle delay. All analysis scenarios are projected to operate at LOS A with v/c ratios of 0.02 or less.

The v/c, delay, and LOS results of the capacity analysis are shown in Table 2 for the morning and evening peak hours. Detailed calculations as well as tables showing the relationship between delay and LOS are included in the appendix to this report.

² Transportation Research Board, *HIGHIFAY CAPACITY MANUAL*, 2000. Jacoby Road Subdivision — Traffic Impact Analysis



Table 2: Intersection Capacity Analysis Summary

Ó

	Mo	orning Peak H	lour	Ev	ening Peak H	our
	LOS	Delay (s)	v/c	LOS	Delay (s)	v/c
OR-211 at Dubarko Road						
2018 Existing Conditions	С	23	0.37	С	24	0,25
2020 Background Conditions	D	30	0.48	D	32	0.35
2020 Background Plus Site Conditions	D	33	0.52	D	33	0.38
Dubarko Road at SE Jacoby Road						
2018 Existing Conditions	А	10	0.08	В	10	0.07
2020 Background Conditions	Λ	10	0.08	В	11	0.07
2020 Background Plus Site Conditions	В	10	0.11	В	11	0.10
Street B at SE Jacoby Road						
2020 Background Plus Site Conditions	Α	9	0.01	Δ	9	0.01
SE Jacoby Road at Newton Street						
2018 Existing Conditions	Λ	9	0.01	Δ	9	0.01
2020 Background Conditions	A	9	0.01	А	9	0.01
2020 Background Plus Site Conditions	А	9	0.01	А	9	0.01

Based on the results of the operational analysis, all study intersections are currently operating acceptably per City of Sandy and ODOT standards and are projected to continue operating acceptably upon build-out of the proposed development through year 2020. No operational mitigation is necessary or recommended.

Jacoby Road Subdivision - Traffic Impact Analysis

Safety Analysis

Crash Data Analysis

Using data obtained from the ODOT's Crash Analysis and Reporting Unit, a review of the most recent available five years of crash history (January 2012 to December 2016) at the study area intersections was performed. The crash data was evaluated based on the number of crashes, the type of collisions, the severity of the collisions, and the resulting crash rate for the intersection. Crash rates provide the ability to compare safety risks at different intersections by accounting for both the number of crashes that have occurred during the study period and the number of vehicles that typically travel through the intersection. Crash rates were calculated using the common assumption that traffic counted during the evening peak period represents 10 percent of average daily traffic (ADT) at the intersection. Crash rates in excess of one to two crashes per million entering vehicles (CMEV) may be indicative of design deficiencies and therefore require a need for further investigation and possible mitigation.

No crashes were found to be reported at the intersection of SE Jacoby Road at Dubarko Road during the analysis period.

One crash was reported at the intersection of SE Jacoby Road at SE Newton Street during the analysis period. The crash was a turning movement collision where a cyclist ran through a stop sign, failing to yield the right-of-way to a car, and collided with the vehicle, resulting in a non-incapacitating injury sustained by the cyclist. The crash rate was calculated to be 0.91 CMEV.

There were 28 crashes reported at the intersection of Dubarko Road at OR-211 during the analysis period. Of these crashes, 16 occurred before and 12 occurred after the intersection was improved to allow eastbound and westbound vehicles to approach the intersection at a right angle.

Thirteen of the crashes were angle-type collisions and occurred as a result of westbound or eastbound vehicles failing to yield the right-of-way to northbound or southbound vehicles. Six of the crashes were rearend collisions, of which three occurred during snowy or icy conditions and two are attributed to be caused by slipping or swerving due to the snow or ice. Six of the crashes were turning-movement collisions and were the result of turning vehicles failing to yield the right-of-way to through-moving vehicles. One of the crashes was a sideswipe collision. One of the crashes was an improper backing movement collision. One of the crashes was a fixed-object collision.

There were 11 crashes that resulted in property damage only while the other 17 crashes resulted in reports of injuries. There were 26 reports of possible injuries or complaints of pain and 7 reports of non-incapacitating injuries. The crash rate was calculated to be 1.83 CMEV. As this intersection is an ODOT facility, the crash rate was compared to the 90th percentile rate outlined in the ODOT *Analysis Procedures Manual* ³ (APM). As seen in Exhibit 4-1 of the APM, the 90th percentile crash rate for a 4-leg minor stop-control rural roadway is 1.08 CMEV. Since the calculated crash rate is higher than the 90th percentile rate, improvements to the

³ https://www.oregon.gov/ODOT/Planning/Documents/APMv2_Ch4.pdf Jacoby Road Subdivision — Traffic Impact Analysis

intersection may be required. Installation of the planned turn lanes and traffic signal outlined in the Sandy TSP will likely reduce the crash rate at this intersection.

Based on the most recent five years of crash data, no other significant trends or crash patterns were identified at the study intersections. Accordingly, no specific safety mitigation is necessary or recommended.

Sight Distance Analysis

Sight distances were measured and evaluated in accordance with standards established in *A Policy of Geometric Design of Highways and Streets*⁴. According to AASHTO standards, the driver's eye is assumed to be 14.5 feet from the near edge of the nearest travel lane of the intersecting street and at a height of 3.5 feet above the minor-street approach pavement. The vehicle driver's eye-height along the major-street approach is assumed to be 3.5 feet above the cross-street pavement.

Based on the posted speeds of 25 mph, the minimum recommended intersection sight distance to ensure safe and efficient operation of intersections along both roadways are 280 feet viewing in both directions along the major-street from the minor-street approaches.

For the proposed intersection of Street B at Jacoby Road, ISD was measured to be 350 feet to the north and 410 feet to the south. For the proposed southern site access at the intersection of Newton Street at Jacoby Road, ISD was measured to be 545 feet to the north and 595 feet to the south.

Based on the detailed analysis, adequate sight distance is available in both directions at both locations. No mitigations are necessary or recommended.

Warrant Analysis

Left-turn and traffic signal warrants were examined for the study intersections where such treatments would be applicable.

The intersection of OR-211 at Dubarko Road is planned for construction of a northbound right-turn lane, a southbound left-turn lane, a northbound left-turn lane, and a traffic signal according to the City of Sandy TSP Preferred Plan. These plans are identified as a Long-Term Project with ODOT specified as the lead agency.

Preliminary traffic signal warrants are not projected to be met for the intersection of OR-211 at Dubarko Road under any analysis scenario, including year 2020 with completion of the proposed subdivision. As such, no new traffic signals are recommended as part of this development.

A left-turn refuge lane is primarily a safety consideration for the major-street, removing left-turning vehicles from the through traffic stream. The left-turn lane warrants used were developed from methodology outlined in the NCHRP Report 457, published by the Transportation Research Board in 2001. Turn lane warrants

⁴ American Association of State Highway and Transportation Officials (AASHTO), A Policy on Geometric Design of Highways and Streets, 6th Edition, 2011.



were evaluated based on the number of advancing and opposing vehicles as well as the number of turning vehicles, the travel speed, and the number of through lanes. Left-turn lane warrants are met under existing conditions for all three turn lanes outlined in the Sandy TSP.

Left-turn lane warrants were examined for the intersections of SE Jacoby Road at Dubarko Road, SE Jacoby Road at Street B, and SE Jacoby Road at Newton Street. Left turn lanes are not warranted under any analysis scenarios. Detailed warrant analysis calculation sheets are provided in the technical appendix of this report.

Conclusions

All study intersections are currently operating acceptably per City of Sandy and ODOT standards and are projected to continue operating acceptably upon build-out of the proposed development through year 2020. No operational mitigation is necessary or recommended.

A crash pattern was identified at the intersection of OR-211 at Dubarko Road. The planned turn lanes and signal at the intersection outlined in the Sandy TSP will help to mitigate this crash pattern. No safety mitigation is recommended as part of this development.

Adequate sight distance is available in both directions at both proposed site access intersections. No sight distance mitigations are necessary or recommended.

Turn lane warrants are met under existing conditions for three turn lanes outlined in the Sandy TSP for the intersection of OR-211 at Dubarko Road. A traffic signal is also planned at that intersection, but signal warrants are not yet met under any analysis scenario. Turn lane warrants are not met under any analysis scenario for the other three study area intersections. No turn lane or signal mitigations are recommended as part of this development.



Jacoby Road Subdivision - Traffic Impact Analysis



 \bigcirc



Hwy 211 & Dubarko Dr

Wednesday, May 02, 2018 7:00 AM to 9:00 AM



5-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		Northi Hwy	bound 211			South Hwy	bound 211			East	nound riko Dr			Westi Duba	oound rko Dr		Interval		Peder	strians swalk	
Time	L	T	R	Bikes	L	T	R	Bikes	L	т	R	Bikes	L	Т	R	Bikes	Total	North	South	East	West
7:00 AM	3	14	2	0	0	12	0	0	2	1	1	0	- 4	4	3	0	46	1	0	1	0
7:05 AM	3	27	0	0	0	11	0	0	1	0	4	0	4	3	3	0	56	0	Ó	0	0
7 10 AM	5	21	3	0	0	7	0	0	1	1	1	0	7	7	5	0	58	0	0	0	0
7 15 AM	14	21	0	0	0	11	1	0	2	1	1	0	3	4	7	0	65	0	0	0	0
7:20 AM	5	19	2	0	1	17	0	0	2	1	5	0	2	3	2	0	59	0	0	0	0
7:25 AM	6	29	3	0	0	15	1	0	2	0	7	0	11	4	5	0	63	0	0	0	0
7 30 AM	5	23	1	0	0	15	1	0	3	0	3	0	7	4	2	0	64	1 1	0	0	0
7:35 AM	5	19	0	0	0	15	0	0	0	1	5	0	1	0	0	0	46	0	0	0	0
7:40 AM	1	23	0	0	0	20	0	0	0	0	5	0	5	1	1	0	56	0	0	0	0
7:45 AM	4	9	4	0	0	11	0	0	0	1	3	0	2	4	2	0	40	0	0	0	0
7:50 AM	2	15	1	0	0	6	1	0	2	0	4	0 1	2	3	1	0	37	0	0	0	0
7:55 AM	2	19	2	0	0	9	1	0	1	1	3	0	4	5	0	0	47	1 0	0	0	0
8:00 AM	1	14	0	0	0	9	0	0	2	1	1	0	2	1	3	0	34	1 0	0	0	0
8:05 AM	3	17	0	0	1	11	0	0	0	1	3	0	1	0	1	0	38	1 0	0	0	0
8:10 AM	2	20	4	0	1	9	1	0	1	0	3	0	1	2	3	0	47	1 0	0	0	0
8.15 AM	2	19	0	0	0	9	0	0	0	0	1	0	4	4	1	0	40	0	0	0	0
8:20 AM	1	13	2	0	0	8	0	0	0	0	3	0	1	6	5	0	39	0	0	0	0
8.25 AM	4	17	1	0	0	5	1	0	0	1	1	0	1	1	1	0	33	0	0	1	0
8 30 AM	3	15	0	0	0	9	1	0	1	0	1	0	0	3	1	0	34	0	Ö	0	0
8:35 AM	5	19	1	0	0	9	0	0	0	2	1	0	1	2	2	0	42	0	0	0	0
840 AM	2	12	0	0	0	10	3	0	1	0	1	0	4	0	1	0	34	0	0	0	0
8 45 AM	2	10	2	0	0	5	1	0	0	3	0	0	1	1	0	0	25	0	0	0	0
8 50 AM	1	23	2	0	2	9	0	0	2	2	4	0	3	3	4	0	55	0	0	0	0
8.55 AM	1	24	2	0	1	12	1	0	2	2	5	0	1	5	2	0	58	0	0	0	0
Total Survey	82	442	32	0	6	254	13	0	25	19	66	0	72	70	55	0	1,136	2	0	2	٥

15-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		Northi	211			South	bound 211			Easti Duba	rko Dr			West: Duba	rko Dr		Interval		Peder	trians swalk	
Time	L	T	R	Bikes	L	T	R	Bikes	L	T	R	Bikes	L	т	R	Bikes	Total	North	South	East	West
7:00 AM	11	62	5	0	0	30	0	0	4	2	6	0	15	14	11	0	160		0	1	0
7.15 AM	25	69	5	0	1	43	2	0	6	2	13	0	16	11	14	0	207	0	0	0	0
7.30 AM	11	65	1	0	0	50	1	0	3	1	13	0	13	5	3	0	166	1 1	0	0	0
7.45 AM	8	43	7	0	0	26	2	0	3	2	10	0	8	12	3	0	124	0	0	0	0
8:00 AM	6	51	- 4	0	2	29	1	0	3	2	7	0	4	3	7	0	119	0	0	0	0
B 15 AM	7	49	3	0	0	22	1	0	0	1	5	0	6	11	7	0	112	0	0	1	0
6 30 AM	10	46	1	0	0	28	4	0	2	2	3	0	5	5	- 4	0	110	0	0	0	0
6 45 AM	4	57	6	0	3	26	2	0	4	7	9	0	5	9	6	0	138	0	0	0	0
Total Survey	82	442	32	0	6	254	13	0	25	19	66	0	72	70	55	O	1,136	2	0	2	0

Peak Hour Summary 7:00 AM to 8:00 AM

By		North	bound 211			South	bound 211			Eastb Duba	rko Dr			Westl Duba	ound rko Dr		Total		Pader	swalk	
Abbioacti	In	Out	Total	Bikes	ln I	Out	Total	Bikes	In	Out	Total	Bikes	In	Out	Total	Bikes		North	South	East	West
Volume	312	243	555	0	155	286	441	0	65	102	167	0	125	26	151	0	657	1 2	0	1	0
%HV		- 4	5%			97	7%			13	8%			41	3%		6.7%				
PHF	I	0	79			0.1	76		Ī	0	71	-		0	73		0 79	1			
By		North	bound / 211	1		South	bound 211			East	ound rko Dr			Westi	ound rko Dr		Total]			
By Movement	L	North Hwy T	bound / 211 R	Total	Ļ	South Hwy T	bound 211 R	Total	L	East Duba	iound iko Dr R	Total	L	Westi Duba T	rko Dr R	Total	Total]			
By Movement Volume	L 55	North Hwy T 239	bound / 211 R 18	Total 312	L 1	South Hwy T 149	bound 211 R 5	Total	L 16	East Duba T 7	iound iko Dr R 42	Total 65	L 52	Westi Duba T 42	rko Dr R 31	Total 125	Total 657				
By Movement Votume %HV	L 55 5.5%	North Hw T 239 3.3%	211 211 R 18 16 7%	Total 312 4.5%	L 1 0 0%	South Hwy T 149 10 1%	bound 211 R 5 0 0%	Total 155 9.7%	L 16 12 5%	East Duba T 7 25 6%	iound rko Dr R 42 11 9%	Total 65 13 6%	L 52 1 9%	Westi Duba T 42 7 1%	rko Dr R 31 6 5%	Total 125 4 8%	Total 657 6.7%				

Rolling Hour Summary 7:00 AM to 9:00 AM

Interval Start		Northi Hwy	211			South Hwy	bound 211			East Duba	iound rko Dr			West	rko Dr		Interval		Pades	strians swalk	
Time	L	T	R	Bikes	L	T	R	Bikes	L	Т	R	Bikes	L	T	R	Bikes	Total	North	South	East	West
7.00 AM	55	239	18	0	1	149	5	0	16	7	42	0	52	42	31	0	657	2	0	1	0
7 15 AM	50	228	17	0	3	148	6	0	15	7	43	0	41	31	27	0	616	1 1	0	0	0
7.30 AM	32	208	15	0	2	127	5	0	9	6	35	0	31	31	20	0	521	1	0	1	0
7.45 AM	31	189	15	0	2	105	8	0	6	7	25	0	23	31	21	0	465	0	0	1	0
8 00 AM	27	203	14	0	5	105	8	0	9	12	24	0	20	28	24	0	470	0	0	1	0





Hwy 211 & Dubarko Dr

Wednesday, May 02, 2018 7:00 AM to 9:00 AM



Heavy Vehicle 5-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		North Hwy	211			South Hwy	bound 211			Easti Duba	rko Dr			Westi Duba	rko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	T	R	Total	L	T	R	Total	Total
7:00 AM	0	0	0	0	0	2	0	2	0	1	0	1 1	0	0	0	0	3
7:05 AM	0	1	0	1	Ö	1	Ó	1	0	0	0	0	1	0	0	1	3
7:10 AM	0	0	0	0	0	0	0	0	0	0	0	0	0	1	1	2	2
7:15 AM	2	2	0	4	0	0	0	0	1	1	1	3	0	0	1	1	В
7 20 AM	0	0	0	0	0	1	0	1	0	0	0	D	0	0	0	0	1
7.25 AM	0	1	0	1 1	0	2	0	2	0	Û.	1	1.1	0	0	0	0	4
7:30 AM	1	1	1	3	Ö	3	0	3	1	0	0	1 1	0	1	0	1	6
7.35 AM	0	2	0	2	Ó	2	0	2	0	0	0	0	0	0	0	0	4
7:40 AM	0	0	0	0	0	2	0	2	0	0	2	2	0	1	Ó	1	5
7:45 AM	0	0	2	2	0	0	0	0	0	0	1	1 1	0	0	0	0	3
7:50 AM	0	1	0	(1	0	1	0	1 1	0	0	0	0	0	0	0	0	2
7:55 AM	0	0	0	0	0	1	0	1 1	0	0	0	0	0	0	0	0	1
8:00 AM	0	1	0	1	0	2	0	2	0	0	0	0	0	0	0	0	3
8:05 AM	1	3	0	4	0	3	0	3	0	0	0	0	0	0	0	0	7
8.10 AM	0	1	0	1	0	1	0	1 1	0	0	1	1	0	0	0	0	3
8.15 AM	0	1	0	1	0	0	0	0	0	0	0	0	0	0	0	0	1
8 20 AM	0	2	0	2	0	1	0	1.1	0	0	0	0	1	1	1	3	6
8:25 AM	2	0	0	2	0	2	0	2	0	0	0	0	0	0	0	0	4
8:30 AM	2	1	0	3	0	1	0	10.10	0	0	0	0	0	0	0	0	4
8 35 AM	0	1	0	1	0	2	0	2	0	D	0	0	Ó	0	0	0	3
8-40 AM	0	0	0	0	0	2	1	3	0	0	0	0	0	0	0	0	3
8:45 AM	0	0	0	0	0	3	0	3	0	0	0	0	0	0	0	0	3
8:50 AM	0	1	0	1	0	1	0	1 1	0	0	1	1	0	0	0	0	3
8.55 AM	0	1	0	1	0	0	0	0	0	0	1	1	0	0	0	0	2
Total Survey	8	20	3	31	0	33	1	34	2	Z	8	12	z	4	3	9	86

O

Heavy Vehicle 15-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		North	bound 211			South	bound 211			Duba	bound irka Dr			Duba	hound rko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	Т	R	Total	L	Т	R	Total	Total
7:00 AM	0	1	0	42.1.	0	3	0	3	0	1	0	1	1	1	1	3	8
7 15 AM	2	3	0	5	0	3	0	3	1	1	2	4	0	0	1	1	13
7.30 AM	1	3	1	5	0	7	0	7	1	0	2	3	0	2	0	2	17
7 45 AM	0	1	2	3	0	2	0	2	0	0	1	1	0	0	0	0	6
MA 00 B	1	5	0	6	0	6	0	6	0	0	1	1	0	0	0	0	13
8.15 AM	2	3	0	5	0	3	0	3	0	0	0	0	1	1	1	3	11
8 30 AM	2	2	0	4	0	5	1	6	0	0	0	0	0	0	0	0	10
8 45 AM	0	2	0	2	0	4	0	4	0	0	2	2	0	0	0	0	8
Totai Survey	8	20	3	31	٥	33	1	34	2	2	8	12	2	4	3	9	86

Heavy Vehicle Peak Hour Summary 7:00 AM to 8:00 AM

1100 -1111			***										
Ву		North Hwy	bound 211		South Hwy	bound 211		East	rko Dr		West	bound rite Dr	Total
Abbioetu	In	Out	Total	In	Out	Total	in	Out	Total	In	Out	Total	
Volume	14	21	35	15	12	27	9	6	15	6	5	11	44
PHF	0.58			0.54			0.56			0.38			0.65

By		North Hwy	bound 211		-	South Hwy	bound 211			Eastb Duba	iound rko Dr		-	West	nko Dr		Total
MOARTINUT	L	T	R	Total	L	T	R	Total	L	Т	R	Total	L	T	R	Total	
Volume	3	8	3	14	0	15	0	15	2	2	5	9	1	3	2	6	44
PHF	0.38	0.50	0.38	0.5B	0.00	0.54	0.00	0.54	0.50	0.50	0.42	0.56	0.25	0.38	0.25	0.38	0.65

Heavy Vehicle Rolling Hour Summary 7:00 AM to 9:00 AM

Interval Start		North Hwy	211			South Hwy	bound 211			East	rko Dr			Wes Dub	tbound arko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	T	R	Total	L	T	R	Total	Total
7:00 AM	3	- ô	3	14	0	15	0	15	2	2	5	9	1	3	2	6	44
7 15 AM	4	12	3	19	0	18	0	18	2	1	6	0	0	2	1	3	49
7.30 AM	4	12	3	19	0	18	0	18	1	0	4	5	1	3	1	5	47
7:45 AM	5	11	2	18	0	16	1	17	0	0	2	2	1	1	1	3	40
8 00 AM	5	12	0	17	0	18	1	19	0	0	3	3	1	1	1	3	42





4:00 PM to 6:00 PM

PHF PH Out 366 in 330 Peak Hour Summary 4:25 PM to 5:25 PM

5-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		North Hwy	bound 211			South! Hwy	211			Easti Duba	nound ake Dr			West	rko Dr		Interval		Pedes	trians Iwalk	
Time	L	T	R	Bikes	E.	T	R	Bikes	L	Ŧ	R	Bikes	L	Т	R	Bikes	Total	North	South	East	West
4:00 PM	1	15	3	0	0	20	1	0	0	5	4	0	0	3	2	0	54		0	0	0
4:05 PM	4	23	3	0	0	19	1	0	1.	4	7	0	2	3	3	0	70	0	0	0	1
4 10 PM	1	25	2	0	2	24	0	0	1	4	2	0	2	3	0	0	66	0	0	0	0
4 15 PM	5	12	3	0	1	23	1	0	1	4	6	0	3	4	5	0	70	0	0	0	0
4.20 PM	2	21	3	0	1	22	0	0	1	1	1	0	0	3	2	0	57	1	0	0	0
4.25 PM	5	30	6	0	2	25	0	D	0	1	3	0	0	3	1	0	76	0	0	0	0
4 30 PM	5	16	3	0	1	18	2	0	1	4	6	0	0	4	0	0	60	1	1	0	1.1
4 35 PM	3	12	3	0	0	27	2	0	1	5	1	0	1	5	2	0	62	0	0	0	0
4 40 PM	7	17	5	0	1	27	1	0	1	1	3	0	4	6	3	0	76	0	0	0	0
4 45 PM	2	21	1	0	1	26	1	0	1	6	8	0	2	2	1	0	72	0	0	0	0
4:50 PM	2	21	2	0	1	16	1	0	0	- 4	3	0	3	5	3	0	61	0	1	0	0
4:55 PM	3	16	- 4	0	4	23	1	0	1	9	4	0	0	0	0	0	65	0	0	0	0
5.00 PM	7	- 14	- 4	0	0	19	1	1 1	2	5	9	0	1	2	5	0	69	0	0	0	0
5:05 PM	5	15	9	0	0	24	1	0	0	3	4	0	3	3	3	0	70	0	0	0	0
5:10 PM	7	18	3	0	1	30	1	0	1	0	7	0	2	4	4	0	78	0	0	0	0
5:15 PM	3	21	6	0	0	24	2	0	0	5	7	0	6	1	1	0	76	0	0	0	0
5.20 PM	3	30	1	0	1	22	1	0	0	4	5	0	3	2	3	0	75	0	0	0	0
5:25 PM	4	17	4	0	- 4	21	1	0	0	8	5	0	0	3	4	0	71	0	0	0	0
5:30 PM	7	12	4	1	2	17	2	0	1	2	3	0	3	1	4	0	58	0	0	0	0
5:35 PM	8	21	- 5	0	0	24	0	0	1	2	3	0	1	2	D	0	67	11_0_	0	0	0
540 PM	4	15	6	0	0	18	1	0	0	4	6	0	2	2	1	0	59	10	0	0	0
5.45 PM	6	20	7	0	1	24	1	0	1	3	5	0	3	5	3	0	79	1 0	0	0	0
5:50 PM	7	14	4	0	1	21	0	0	0	7	7	0	2	3	2	0	65	0	0	0	0
5.55 PM	3	22	7	0	1	14	1	0	0	8	5	0	0	4	3	0	65	0	0	0	1
Total Survey	104	448	95	1	25	528	23	1	15	99	116	0	43	73	55	0	1,627	2	z	D	3

15-Minute Interval Summary

4:00 PM to 6:00 PM Pedestrians Crosswalk South East Interval Start Eastbound Dubarko Dr Northbo Hwy 211 South Westbound Dubarko Dr Hwy 211 T R Interval Start Time 4 00 PM 4 15 PM 4 30 PM 4 30 PM 4 45 PM 5 00 PM 5 15 PM 5 30 PM 5 45 PM Total Bikes Bikes Total 190 203 198 198 217 222 184 215 B Bikes Ť R Bikes West 13 13 9 63 63 45 58 47 68 48 56 8 12 11 7 16 11 15 63 70 72 65 73 67 59 0000100 6 12 15 7 19 10 19 16 355696 10 15 7 9 0 4 2 6 2 3 2 3 0 2 6 10 19 8 17 12 10 15 20 17 12 17 8 ò 0 000000 12 000 0 18 59 16 104 448 98 1 25 528 23 1 15 99 116 0 43 73 55 0 1,627 2 2 0 3 Survey

Peak Hour Summary 4:25 PM to 5:25 PM

By		North	bound 211			South	bound 211			Eastb Duba:	ound ke Dr			Westb	iound ko Dr		Total		Pedes	trians swalk	
Approach	In	Out	Total	Bikes	In	Out	Total	Bikes	In	Out	Total	Bikes	fn l	Out	Total	Bikes		North	South	East	West
Volume	330	366	696	0	307	265	572	1.1	115	103	218	0	68	106	194	0	840		2	0	1
%HV		1	8%			1.3	3%			0 9	P%			4 5	56		1.6%	1			
0		0				0.	00			0.1	70			0.1	70		0.02	-			
PHF		0	90			0.	<u>da</u>			Ų	/ a		·		¢.		0.92				
By		North	bound / 211			South Hwy	bound 211			Eastb	ound ko Dr			West	ound to Dr		Total	Ì			
By Movement	L	North Hwy T	bound / 211 R	Total	L	South Hwy T	bound 211 R	Total	L	Eastb Dubar T	iound tko Dr R	Total	L	Westi Duba	ko Dr R	Total	Total]			
By Movement Volume	L 52	North Hwy T 231	bound 211 R 47	Total 330	L 12	South Hwy T 261	bound 211 R 14	Total 307	L	Eastb Duba T 47	ra iound rko Dr R 60	Total 115	L25	Westi Duba T 37	icound tko Dr R 26	Total 86	Total 840				
By Movement Volume %HV	L 52 3.8%	North Hwy T 231 1 3%	bound 211 R 47 2 1%	Total 330 1.6%	L 12 0 0%	South Hwy T 261 1 4%	bound 211 R 14 0.0%	Total 307	L 8 00%	Eastb Duba T 47 0 0%	iound rko Dr R 60 1 7%	Total 115 0.9%	L25 8 0%	Westi Duba T 37 5 4%	ko Dr R 26 0 0%	Total 86 4 5%	Total 840 1 8%				

Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		North	211			South Hwy	bound 211	1		East	bound Irko Dr			West Duba	bound rko Dr	i	Interval		Peder	striaris swalk	
Time	L	T	R	Bikes	L	Ť	R	Bikes	L	Т	R	Bikes	L	T	R	Bikes	Total	North	South	East	West
4 00 PM	40	229	38	0	14	270	11	: 0	9	48	50	0	17	41	22	0	789	2	2	0	2
4.15 PM	53	213	46	0	13	280	12	1	10	43	57	0	19	41	29	0	816	2	2	0	1
4:30 PM	51	218	45	0	14	277	15	1	B	54	62	0	25	37	29	0	835	1	2	0	1
4 45 PM	55	221	49	1	14	264	13	1	7	52	64	0	26	27	29	0	821	0	1	0	0
5 00 PM	64	219	60	1	11	258	12	1	6	51	66	0	26	32	33	0	838	0	0	0	1





Hwy 211 & Dubarko Dr

Tuesday, May 01, 2018 4:00 PM to 6:00 PM



Heavy Vehicle 5-Minute Interval Summary 4:00 PM to 6:00 PM

interval Start		North Hwy	bound 211			South Hwy	bound 211			Easti Duba	rko Dr			West	bound rko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	Т	R	Total	Ł	Т	R	Total	Total
4:00 PM	0	1	0	1 1	0	2	0	2	0	0	0	: 0 (0	0	0	0	3
4 05 PM	0	1	0	1	0	1	0	1	0	0	0	0	0	0	1	1	3
4:10 PM	0	1	0	1 1	0	0	0	0	1	0	0	1	0	0	0	0	2
4 15 PM	0	0	0	0	0	1	0	1	0	0	1	1 1	1	0	0	1 1	3
4.20 PM	0	1	0	1	0	1	0	1	0	0	0	0	0	0	0	0	2
4.25 PM	1	0	0	1	0	0	0	0	0	0	0	0	0	0	0	0	1
4 30 PM	1	0	0	1 1	0	0	0	0	0	0	0	0	0	0	0	0	1
4:35 PM	0	0	0	0	0	1	0	1	0	0	0	0	0	0	0	0	1
4 40 PM	0	0	1	1	0	1	0	1	0	0	0	0	0	0	0	0	2
4 45 PM	0	0	0	0	0	2	0	2	0	0	0	0	0	0	0	0	2
4:50 PM	0	0	0	0	0	0	0	0	0	0	0	0	1	2	0	3	3
4 55 PM	0	1	0	1	0	0	0	0	0	0	0	0	0	0	0	0	1
5 00 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
5.05 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
5:10 PM	0	2	D	2	0	0	0	0	0	0	1	1 1	1	0	0	1	4
5:15 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
5:20 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	i _0	0	0	0
5.25 PM	0	0	0	0	1	Q	0	1	0	0	0	0	0	0	0	0	1
5:30 PM	0	0	0	0	0	2	0	2	0	0	0	0	0	0	0	0	2
5:35 PM	0	0	0	0	0	1	0	1	0	0	0	0	0	0	0	0	1
5.40 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
5 45 PM	0	0	1	1	0	3	0	3	0	0	0	0	0	0	D	0	4
5:50 PM	0	0	0	0	0	0	0	0	0	0	0	D	0	0	0	0	0
5:55 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
Total Survey	2	7	2	11	1	15	0	16	1	0	2	3	3	2	1	6	36

Heavy Vehicle 15-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		North Hwy	bound / 211			South Hwy	bound 211			East Duba	bound Irko Dr			West Duba	bound rko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	Т	R	Total	L	T	R	Total	Total
4 00 PM	0	3	0	3	0	3	0	1 3	1	0	. 0	1 1	0	0	1	1 1	8
4 15 PM	1	1	0	2	0	2	0	2	0	0	1	1	1	0	0	1	6
4:30 PM	1	0	1	2	0	2	0	2	0	0	0	0	0	0	0	0	4
4 45 PM	0	1	0	1	0	2	0	2	0	0	0	0	1	2	0	3	6
5 00 PM	0	2	D	2	0	0	0	0	0	0	1	1	1	0	0	1	4
5.15 PM	0	0	0	0	1	0	0	1	0	0	0	0	0	0	0	0	1
5 30 PM	0	0	0	0	0	3	0	3	0	0	0	0	0	0	0	0	3
545 PM	0	0	1	1	0	3	0	3	0	0	0	0	0	0	0	0	4
Total Survey	2	7	2	- 11	1	15	0	16	1	0	2	3	3	2	1	6	36

Heavy Vehicle Peak Hour Summary 4:25 PM to 5:25 PM

By Approach		North	211	1	South	bound 211		East	bound irko Dr		Duba	bound Irko Dr	Total
.,	10	Out	10181	In	Out	Total	10	Out	Total	11	Out	Tota	
Volume	6	7	13	4	3	7	1	- 4	5	4	1	5	15
PHF	075	1.11		0.25		_	0.25		_	0.33			0.54

By Northbound Movement L T R	bound 211	_		South Hwy	bound 211			Eastb Duba	ound ko Dr			West	ound iko Dr		Total		
NIOVOIDEIN	L	T	R	Total	_L	T	R	Total	L	T	R	Total	L	Т	R	Total	
Volume	2	3	1	6	0	4	0	4	Û	0	1	1	2	2	0	4	15
PHF	0.25	0.38	0.25	0.75	0 00	0.25	0 0 0	0.25	0.00	0.00	0.25	0.25	0.50	0 25	0.00	0.33	0.54

Heavy Vehicle Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		Norih Hwj	bound 211		_	South Hwy	bound 211			Easti Duba	rko Dr	_		West	bound rko Dr		Interval
Time	L	T	R	Total	L	T	R	Total	L	T	R	Total	L	T	R	Total	Total
4:00 PM	2	5	1	8	0	9	0	9	1	0	1	2	2	2	1	5	24
4.15 PM	2	4	1	7	0	6	0	6	0	0	2	2	3	2	0	5	20
4:30 PM	1	3	1	5	1	4	0	5 1	0	0	1	1	2	2	0	4	15
4 45 PM	0	3	0	3	1	5	0	6	0	0	1	1	2	2	0	4	14
5 00 PM	0	2	1	3	1	6	0	7	0	0	1	1	1	0	0	1	12





5-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East	rko Dr			Westbou Dubarko	ind Dr	Interval		Pedea	trians swalk	
Time	L	R	Bikes	Bikes	Т	R	Bikes	L	T	Bikes	Total	North	South	East	West
7:00 AM	5	0	0	0	1	2	0	1	8	0	37	0	0	0	0
7:05 AM	4 -	2	0	0	1	1	0	0	8	0	16	0	0	0	0
7 10 AM	6	0	0	0	3	0	0	0	14	0	23	0	0	0	0
7 15 AM	6	1	0	0	3	1	0	1	10	0	22	0	0	0	0
7.20 AM	3	0	0	0	3	4	0	0	8	0	18	11 0	0	0	0
7.25 AM	1	1	0	0	2	0	0	1	14	0	19	1 0	0	0	0
7.30 AM	2	0	0	0	Z	2	0	0	10	0	16	1 0	0	0	0
7.35 AM	0	2	0	0	1	0	0	1	2	0	6	1 0	Ð	0	0
7:40 AM	0	0	0	0	1	1	0	0	8	0	10	1 0	0	0	0
7.45 AM	3	2	0	0	2	1	0	0	3	0	11	1 0	0	0	0
7:50 AM	2	3	0	0	1	2	0	2	6	0	16	10	0	0	0
7:55 AM	5	0	0	0	1	3	0	0	5	0	14	0	0	0	0
8.00 AM	4	0	0	0	1	1	0	1	3	0	10	0	0	0	0
8 05 AM	5	1	0	0	2	1	0	0	2	0	11	0	0	0	0
8 10 AM	2	0	0	0	2	1	0	1	4	0	10	0	0	0	0
8 15 AM	4	0	0	0	1	1	0	1	4	0	11	0	0	0	0
8.20 AM	3	0	0	0	3	1	0	0	9	0	16	0	0	0	0
6.25 AM	2	0	0	0	2	1	0	0	Z	0	7	0	Ó	0	0
8 30 AM	4	3	0	0	1	1	0	1	1	0	11	0	1	0	0
8 35 AM	0	2	0	0	2	0	0	2	6	0	12	0	0	0	0
840 AM	2	. 0	0	0	3	1	0	0	3	Ó	9	0	0	0	0
845 AM	0	2	0	0	1	3	0	1	3	0	10	0	0	0	0
8 50 AM	6	1	0	0	2	1	0	2	5	0	17	0	0	0	0
8.55 AM	1	0	0	0	2	2	0	0	6	0	11	0	1	0	0
Total Survey	70	20	0	0	43	31	0	15	144	0	323	0	2	0	0

15-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East	bound arko Dr			Westbou Dubarko	nd Dr	Interval		Pedes	trians Walk	
Time	L	R	Bikes	Bikes	Т	R	Bikes	L	T	Bikes	Total	North	South	East	West
7:00 AM	15	2	0	0	5	3	0	1	30	0 1	56	0	0	0	0
7 15 AM	10	2	0	0	8	5	0	2	32	0	59	0	0	0	0
7:30 AM	2	2	0	0	4	3	0	1	20	0	32	0	0	0	0
7:45 AM	10	5	0	0	4	6	0	2	14	0	41	0	0	0	0
8 00 AM	11	1	0	0	5	3	0	2	9	0	31	0	0	0	0
8:15 AM	9	0	0	0	6	3	0	1	15	O I	34	0	0	0	Ő.
8:30 AM	6	5	0	0	6	2	0	3	10	0	32	i o	1	0	0
B:45 AM	7	3	0	0	5	6	0	3	14	0	38	0	1	0	0
Total Survey	70	20	0	0	43	31	0	15	144	0	323	0	2	0	0

Peak Hour Summary 7:00 AM to 8:00 AM

By		Norti Jaci	bound by Rd			South Jaco	bound by Rd		12	Eastb	iound rko Dr	- 27		Westi Duba	nko Dr		Total		Pedes	trians swalk	
Abbinact	_ In	Out	Total	Bikes	in	Out	Total	Bikes	In	Out	Total	Bikes	In	Out	Total	Bikes		North	South	East	West
Volume	48	23	71	0	0	0	0	0	38	133	171	0	102	32	134	0	186	0	0	0	0
%HV		10	37%			0	0%			13	2%			6 1	9%	-	10.6%				
PHF			63			0	00			0.0	58			0.	75		0.75	1			
By Movement		Nort	bound by Rd			South Jaco	bound by Rd			East: Duba	iound rko Dr			West Duba	bound rko Dr		Total]			
	L		R	Total				Total		Т	R	Total	L	T		Total					
Volume	37	W	11	48				٥		21	17	38	6	96	1	102	168	1			
%HV	10.6%	NA	36 4%	16.7%	NA	NA	NA	0.0%	NA	19 0%	5.9%	13.2%	16.7%	6 3%	NA	69%	10 6%	1			
PHF	0.58		0.55	0.63				0.00		0.58	0.71	0.68	0.75	0.75		0.75	0.75	1			

Rolling Hour Summary 7:00 AM to 9:00 AM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East Duba	nound rko Dr			Westbou Dubarko	nd Dr	Interval		Pedes	trians Iwelk	
Time	L	R	Bikes	Bikes	т	R	Bikes	L.	T	Bikes	Total	North	South	East	West
7:00 AM	37	11	0	0	21	17	0	6	96	0	188		0	0	0
7 15 AM	33	10	0	0	21	17	0	7	75	0	163	1 0	0	0	0
7.30 AM	32	8	0	0	19	15	0	6	58	0	138	0	D	0	0
7.45 AM	36	11	0	0	21	14	0	В	48	0	136	11 0	1	0	Ū.
MA 00:8	33	9	0	0	22	14	0	9	48	0	135	i õ	2	0	0



Jacoby Rd & Dubarko Dr

Wednesday, May 02, 2018 7:00 AM to 9:00 AM



()) 0 0

Heavy Vehicle 5-Minute Interval Summary 7:00 AM to 9:00 AM

interval Start	No J	acoby Rd		Southbound Jacoby Rd	Eastb Duba	iound rko Dr			Dubarko	Dr	Interval
Time	L	R	Total	Total	Т	R	Total	L	T	Total	Total
7:00 AM	1	0	1	0	0	1	1 1	0	0	0	2
7 05 AM	0	1	1	0	0	0	0	0	1	1 1	2
7.10 AM	2	0	2	0	0	0	0	0	1	1 1	3
7.15 AM	0	1	1	0	1	0	1 1	0	2	2	4
7.20 AM	1	0	1	0	1	0	1	0	0	0	2
7 25 AM	0	0	0	0	0	0	0	0	0	0	0
7 30 AM	0	0	0	0	1	0	1 1	0	0	0	1
7:35 AM	0	1	1	0	0	0	0	1	0	1	2
7 40 AM	0	D	0	0	0	0	0	0	1 1	1	1
7:45 AM	0	1 1	1	0	1	0	1	0	0	0	2
7:50 AM	0	0	0	0	0	0	0	0	1	1	1
7:55 AM	0	0	0	0	0	0	0	0	0	0	0
8 00 AM	0	0	0	0	0	0	0	0	0	0	0
8 05 AM	0	0	0	0	0	0	0	0	0	0	0
8.10 AM	0	0	0	0	0	0	0	0	0	0	0
8 15 AM	0	0	0	0	D	0	0	1	0	1	1
8 20 AM	1	0	1	0	0	0	0	0	2	2	3
8:25 AM	0	0	0	0	0	0	0	0	0	0	0
8 30 AM	0	1	1	0	0	1	1 1	0	0	0	2
8:35 AM	0	1	1	0	0	0	0	0	0	0	1
8 40 AM	0	0	0	0	0	0	0	0	0	0	0
8 45 AM	0	0	0	D	0	0	0	0	0	0	0
8:50 AM	0	0	0	0	0	0	0	0	0	0	0
8:55 AM	0	0	0	0	0	0	0	0	0	0	0
Total Survey	5	6	11	0	4	2	6	2	в	10	27

Heavy Vehicle 15-Minute İnterval Summary 7:00 AM to 9:00 AM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East Duba	rko Dr			Westbou Dubarko	and Dr	latorval
Time	L	R	Total	Total	т	R	Total	L	T	Total	Total
7:00 AM	3	1	1 4	0	0	1	1 1	0	2	2	7
7:15 AM	1	1	2	0	2	0	2	0	2	2	6
7.30 AM	0	1	1 1	0	1	0	1	1	1	2	4
7 45 AM	0	1	1	0	1	0	1	0	1	1	3
8 00 AM	0	0	0	0	0	0	0	0	0	0	0
8:15 AM	1	0	1	0	0	0	0	1	2	3	4
8 30 AM	0	2	2	0	0	1	1	0	0	0	3
8 45 AM	0	0	0	0	0	0	0	0	0	0	0
Total Survey	5	6	11	0	4	2	6	2	а	10	27

Heavy Vehicle Peak Hour Summary 7:00 AM to 8:00 AM

By Approach	Northbound Jacoby Rd In Out Total	Southbound Jacoby Rd In Out Total	Eastbound Dubarko Dr In Out Total	Westbound Dubarko Dr In Out Total	Totat
Valume	8 2 10	0 0 0	5 10 15	7 8 15	20
	0.00	0.00	0.00	0.44	000

By		Northbound Jacoby Rd]	Southbo Jacoby	und Rd	Eastb Duba	ound rko Dr			Westbo	o Dr	Total
MOABUMBLIN	L	R	Total		Total	T	R	Total	L	. T .	Total	
Volume	4	4	÷ 6		0	4	1	5	1	6	7	20
PHF	0.33	0 50	0.50	1	0.00	0.50	0.25	0.63	0.25	0.38	0.44	0.56

Heavy Vehicle Rolling Hour Summary 7:00 AM to 9:00 AM

Interval Start	N	orthbound acoby Rd		Southbound Jacoby Rd	East Duba	bound Irko Dr			Westbor Dubarko	nd Dr	Interval
Time	L	R	Total	Total	Т	R	Total	L	T	Total	Total
7:00 AM	4	4	8	0	4	1	5	1	6	7	20
7.15 AM	1	3	4	0	4	0	4	1	4	5	13
7:30 AM	1	2	3	0	2	0	2	2	4	6	11
7.45 AM	1	3	4	0	1	1	2	1	3	4	10
8.00 AM	1	2	3	0	0	1 1	1 1	1	2	3	7





5-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East: Duba	iound rko Dr			Westbour Dubarko	nd Dr	Interval		Pedes Cros	itrians swalk	
Time	L	R	Bikes	Bikes	Ť	R	Bikes	L	T	Bikes	Total	North	South	East	West
4:00 PM	4	1	0	0	10	2	0	0	6	0	23	10	0	0	0
4 05 PM	1	1	0	0	11	3	0	0	6	0	22	0	0	0	0
4 10 PM	3	0	0	0	7	2	0	2	5	0	19	0	0	0	0
4.15 PM	1	0	0	ö	6	3	0	2	6	0	16	0	0	0	0
4:20 PM	1	2	0	0	7	3	0	1	5	0	19	0	0	0	0
4:25 PM	5	4	0	0	5	1	0	1	4	0	20	0	0	0	0
4.30 PM	2	2	0	0	4	6	0	3	4	0	21	0	0	D	0
4:35 PM	3	2	0	0	5	5	0	0	6	0	21	0	0	D	0
4 40 PM	4	0	0	0	4	3	0	1	7	1	19	0	0	0	0
4 45 PM	2	1	0	0	8	4	0	0	6	0	21	0	0	0	0
4 50 PM	7	0	0	0	7	5	0	0	6	0	25	0	0	0	0
4.55 PM	5	0	0	0	13	7	0	1	1	0	27	11 0	0	0	0
5 00 PM	0	1	0	0	7	7	0	2	4	0	21	11-0-	0	2	0
5:05 PM	6	0	0	D	9	5	0	0	7	0	27	0	0	0	0
5:10 PM	2	1	0	0	6	4	0	1	8	0	22	0	0	0	1
5:15 PM	1	1	0	0	13	7	0	Ö	1 4 1	0	26	0	0	0	0
5:20 PM	3	0	0	0	7	5	0	0	5	0	20	0	0	0	0
5:25 PM	3	0	0	0	6	6	0	1	4	0	24	0	0	0	0
5.30 PM	2	1	0	0	8	7	0	1	4	0	23	1 0	D	0	0
5.35 PM	0	0	0	0	8	8	0	0	2	0	16	0	0	0	0
5 40 PM	1	2	0	0	10	3	0	Ó	3	1	19	0	0	0	0
5 45 PM	3	0	0	0	13	2	0	2	11	0	31	1 0	0	0	0
5 50 PM	4	1	0	0	9	3	0	4	8	0	29	0	0	0	0
5.55 PM	4	2	0	1 0	15	5	0	1	5	0	32	0	0	0	0
Total Survey	67	22	0	0	198	108	0	23	127	2	545	0	0	2	1

15-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	East Duba	rka Dr			Westbou Dubarko	nd Dr	Interval		Pedes	trians walk	
Time	L	R	Bikes	Bikes	T	R	Bikes	L	T	Bikes	Total	North	South	East	West
4:00 PM	8	2	0	0	28	7	0	2	17	0	64	0	0	0	0
4 15 PM	7	6	0	0	15	7	0	4	15	0	57	0	0	0	0
4.30 PM	9	4	0	0	13	14	0	4	17	1	61	0	0	0	0
4:45 PM	14	1	0	0	28	16	0	1	13	0	73	0	0	0	0
5.00 PM	8	2	0	0	22	16	0	3	19	0	70	0	0	2	1
5:15 PM	7	1	0	0	28	20	0	1	13	0	70	0	0	0	0
5 30 PM	3	3	0	0	24	18	0	1	9	1	58	0	0	0	0
5.45 PM	11_	3	0	0	37	10	0	7	24	0	92	0	0	0	0
Total Survey	67	22	0	0	198	108	0	23	127	2	545	0	0	2	1

Peak Hour Summary 5:00 PM to 6:00 PM

By		North	bound by Rd			South	bound by Rd			Eastb Duba	iound rko Dr			Duba	iound ko Dr		Total		Pedes	triacus swalk	
Abbioacti	In	Out	Total	Bikes	In	Out	Total	Bikes	ln.	Qui	Total	Bikes	In	Out	Total	Bikes		North	South	East	West
Volume	38	76	114	0	0	0	0	0	175	94	269	0	77	120	197	1	290	0	0	2	1
%HV		2	6%			01	0%			1.1	1%			28	156		17%			_	
PHE	1	0	68			0	00			0.	0.1			0.	22		0.70	-			
FIN			.00			0	00		<u> </u>	ν.	<i></i>			Q.1	16		9.78				
By		North	abound aby Rd			South	bound by Rd			East	iound rko Dr	1000		West	ound ko Dr		Total	ן ן			
By Movement		North	bound by Rd R	Total		South	bound by Rd	Total		East Duba	iound rko Dr R	Total	L	Westi Duba	ound ko Dr	Total	Total	Ì			
By Movement Volume	L 29	North	bound by Rd R 9	Total 38		South	bound by Rd	Total 0		East Duba T	iko Dr R 64	Total 175	L 12	Westi Duba T 65	bound the Dr	Total 77	Total 290				
By Movement Votume %HV	L 29 0 0%	North	bound by Rd R 9 11 1%	Total 38 2.6%	NA	South Jaco	bound by Rd NA	Total 0 0 0%	NA	East Duba T 111 0.9%	iound rko Dr R 64 1.6%	Total 175 1 1%	L 12 0.0%	Westl Duba T 65 3 1%	oound tko Dr NA	Tota1 77 2.6%	Total 290 1 7%				

Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		Northbound Jacoby Rd		Southbound Jacoby Rd	Easth Duba	ound ko Dr			Westbou Dubarko	Dr	Interval		Pedes	trians Iwalk	
Time	L	R	Bikes	Bikes	T	R	Bikes	L	T	Bikes	Total	North	South	East	West
4:00 PM	38	13	0	0	87	44	0	11	62	1	255	0	0	0	0
4 15 PM	38	13	0	0	61	53	0	12	64	1	261	0	0	2	1
4 30 PM	38	8	0	0	91	66	0	9	62	1	274	0	0	2	1
4 45 PM	32	7	0	0	102	70	0	6	54	1	271	0	0	2	1
5 00 PM	29	9	0	0	111	64	0	12	65	1	290	0	0	2	1



4:00 PM to 6:00 PM

Peak Hour Summary 5:00 PM to 6:00 PM

t

+

_

Heavy Vehicle 5-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start	No. Ji	acoby Rd		Southbound Jacoby Rd	East Duba	oound rko Dr			Westbou Dubarko	and Dr	Interval
Time	L	R	Total	Total	T	R	Total	L	T	Total	Total
4:00 PM	0	0	0	0	0	0	0	0	0	0	0
4 05 PM	0	1	1	0	0	0	0	0	1	1 1	2
4.10 PM	0	0	0	0	0	0	0	0	0	0	0
4:15 PM	0	0	0	0	0	0	0	0	1	1	1
4.20 PM	1	0	1	0	0	0	0	0	0	0	1
4.25 PM	0	0	0	0	0	1	1	0	0	0	1
4 30 PM	0	D	0	0	0	0	0	1	0	1 1	1
4 35 PM	0	0	0	0	0	0	0	0	0	0	0
4 40 PM	0	0	0	. 0	0	0	0	0	0	0	0
4 45 PM	0	0	0	0	0	0	0	0	0	0	0
4.50 PM	0	0	0	0	0	0	0	0	2	2	2
4:55 PM	0	0	0	0	0	0	0	0	0	0	0
5 00 PM	0	0	0	0	0	0	0	0	0	0	0
5 05 PM	0	0	0	0	0	0	0	0	0	0	0
5 10 PM	0	0	0	0	0	0	0	0	1	1 1	1
5 15 PM	0	1	1	0	0	0	0	0	0	0	1
5 20 PM	0	0	0	0	0	0	0	0	0	0	0
5.25 PM	0	0	0	0	0	1	1	0	0	0	1
5 30 PM	0	0	0		0	D	0	0	1	1 1	1
5 35 PM	0	0	0	0	0	0	0	0	0	0	0
5 40 PM	0	0	0	0	0	0	0	0	0	0	0
5 45 PM	0	0	0	0	1	0	1	0	0	0	1
5 50 PM	0	0	0	0	0	0	0	0	0	0	0
5:55 PM	0	0	0	0	0	0	0	0	0	0	0
Total Survey	1	2	3	0	1	2	3	1	6	7	13

Heavy Vehicle 15-Minute Interval Summary 4:00 PM to 6:00 PM

Interval		Northbound		Southbound	East	bound			Westbo	bnuc	
Start		Jacoby Rd		Jacoby Rd	Duba	irko Dr			Dubark	o Dr	Interval
Time	L	R	Total	Total	1	R	Total	L	т	Total	Total
4:00 PM	0	1	1 1	0	0	0	0	0	1	1	2
4:15 PM	1	0	1	Û	0	1	1	0	1	1	3
4:30 PM	0	0	0	0	0	0	0	1	0	1	1
4 45 PM	0	0	0	0	0	0	0	0	2	2	2
5:00 PM	0	0	0	0	0	0	0	0	1	1	1
5:15 PM	0	1	1	0	0	1	1 1	0	0	0	2
5.30 PM	0	0	0	0	0	0	0	0	1	1	1
5.45 PM	0	0	0	0	1	0	1	0	0	0	1
Total Survey	1	2	3	0	1	2	3	1	6	7	13

Heavy Vehicle Peak Hour Summary 5:00 PM to 6:00 PM

By Approach	Northbound Jacoby Rd In Out Total	Southbound Jacoby Rd In Out Total	Eastbound Dubarko Dr In Out Total	Westbound Dubarko Dr In Out Total	Total
Volume	1 1 2	0 0 0	2 2 4	2 2 4	5
PHF	0.25		0.50	0.50	0.63

By		Northbound Jacoby Rd		Southt Jacob	ound y Rd	Eastbo Dubari	ound ko Dr			Westbor Dubarko	und Dr	Total	
NUMBER	L.	R	Total		Total	τ	R	Total	L	T	Total		
Volume	0	1	1	- E	0	1.1	1	2	0	2	2	5	Ĩ
PHF	0.00	0 25	0.25		0.00	0.25	0.25	0.50	0.00	0.50	0.50	0.63	

Heavy Vehicle Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		Northbound Jacoby Rd	Southbound Jacoby Rd	Eastbound Dubarko Dr	Westbound Dubarko Dr	Interval
Time	L	R Total	Total	T R Total	L T Total	Total
4:00 PM	1.1	1 2		0 1 1	1 4 5	8
4:15 PM	1 1	0 1	0	0 1 1	1 4 5	7
4 30 PM	0	1 1 1	1 0	0 1 1	1 3 4	6
4 45 PM	0	1 1 1	0	0 1 1	0 4 4	6
5:00 PM	0	1 1	0	1 1 2	0 2 2	5





5-Minute Interval Summary 7:00 AM to 9:00 AM

7:00 AM to 9:00 AM

Interval Start		Northbou Jacoby F	and	South Jacob	bound by Rd			Eastbound Newton St		Westbound Newton St	Interval		Pedes	trians swalk	
Time	L	T	Bikes	T	R	Bikes	L	R	Bikes	Bikes	Total	North	South	East	West
7:00 AM	0	0	0	0	0	0	1	0	0	0	1	0	0	0	0
7:05 AM	0	3	0	0	0	0	1	0	0	0	4	0	Ó	0	0
7.10 AM	0	2	0	0	0	0	2	0	0	0	4	Ó	0	0	0
7 15 AM	0	1	0	1	0	0	0	0	0	0	2	0	0	0	0
7:20 AM	0	4	0	2	1	0	0	0	0	0	7	0	0	0	0
7.25 AM	0	0	0	0	1	0	0	0	0	0	1	0	0	0	0
7.30 AM	0	0	0	0	1	0	0	0	0	0	1	0	0	0	1
7:35 AM	0	0	0	0	0	0	0	0	0	0	0	0	Ó	0	0
7.40 AM	0	0	0	1	0	0	0	0	0	0	1	0	Ö	0	1
7:45 AM	0	1	0	0	0	0	1	0	0	0	2	0	Ó	0	0
7:50 AM	0	4	0	1	0	0	1	0	0	0	6	0	Ó	0	0
7:55 AM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
8 00 AM	0	2	0	1	0	0	1	0	0	0	4	0	0	0	0
8-05 AM	0	3	0	1	0	0	0	0	0	0	4	0	0	0	0
B 10 AM	0	2	0	0	0	0	0	0	0	0	2	l ő	1	0	0
8 15 AM	0	2	0	0	0	0	1	0	0	0	3	D	D	0	0
8:20 AM	0	2	0	0	0	0	0	0	0	0	2	0	0	D	0
B.25 AM	0	2	0	0	0	0	0	0	0 1	0	2	0	0	0	0
8 30 AM	0	0	0	1	0	0	2	1	0	0	4	l o	0	0	0
8 35 AM	0	2	0	0	0	0	1	1	0	0	4	LI o	Ö	0	0
840 AM	0	0	0	0	0	0	0	0	0	0	0	i o	Ó	0	0
845 AM	0	1	0	2	1	0	0	1	0	0	5	l ő	0	ň	0
8 50 AM	0	1	0	0	1	0	0	0	0	0	2	1 0	0	0	0
8.55 AM	_ 0	1	0	0	0	0	0	0	0	0	i i	1 o	0	0	0
Total Survey	0	33	0	10	5	0	11	3	٥	0	62	0	1	0	2

Peak Hour Summary 7:50 AM to 8:50 AM

15-Minute Interval Summary 7:00 AM to 9:00 AM

tniervat Start		Northbo Jacoby	und Rd	South Jacob	bound by Rd			Esstbound Newton St		Westbound Newton St		Interval		Peder	trians swalk	
Time	L	T	Bikes	T	R	Bikes	L	R	Bikes		Bikes	Total	North	South	East	West
7:00 AM	0	5	0	0	0	0	4	0	0		0	9	0	0	0	0
7.15 AM	0	5	0	3	2	0	0	0	0		0	10	0	0	0	0
7:30 AM	0	0	0	1	1	0	0	0	0		0	2	0	0	0	2
7:45 AM	0	5	0	1	0	0	2	0	0		0	6	0	0	0	0
MA 00.8	0	7	0	2	0	0	1	0	0		0	10	0	1	0	0
8:15 AM	0	6	0	0	0	D	1	0	0		0	7	0	0	0	0
8:30 AM	0	2	0	1	0	0	3	2	0		D	B	l ō	0	0	0
8:45 AM	0	3	0	2	2	0	0	1	0		0	6	i o	0	0	0
Total Survey	0	33	D	10	5	0	11	3	0		0	62	0	1	0	2

Peak Hour Summary 7:50 AM to 8:50 AM

By		Northi Jacob	bound by Rd			South Jacob	bound by Rd			East New	bound ton SI			West! New!	on St		Total		Pades	swalk	
Abbioacti	In	Out	Total	Bikes	In	Qut	Total	Bikes	In	Out	Total	Bikes	In	Out	Total	Bikes		North	South	East	West
Volume	20	9	29	0	7	26	33	0	9	1	10	0	0	0	0	0	36	0	1	0	0
%HV		5 (1%			0.0	3%			11	1%			0 (0%		5 6%				
PHF		0.1	71			0	58			0	45			0	00		0.90				
By		North	oound by Rd			South	bound by Rd			East	bound ton St			West	bound on St		Total]			
By Movement	L	North Jacob T	bound by Rd	Total		South Jacol T	bound by Rd R	Total	L	East	bound ton St R	Total		West	bound on SI	Total	Total]			
By Movement Volume	L	North Jacob T 20	oound by Rd	Total 20		South Jacol T 6	bound by Rd R	Total 7	L	East	bound fon St R 3	Total 9		West	bound on St	Total	Total 36				
By Movement Volume %HV	L 0 00%	North Jacob T 20 5 0%	oound by Rd NA	Total 20 5 0%	NA	South Jacol T 6 0 0%	bound by Rd R 1 0 0%	Total 7 0.0%	L 6 0.0%	East New NA	bound ton St R 3 33 3%	Total 9	NA	West New	oound on SI	Total 0 0 0%	Total 36 5.6%				

Rolling Hour Summary 7:00 AM to 9:00 AM

-Interval Start		Jacoby I	und Rd	South	bound by Rd			Eastbound Newton St		Westbound Newton St	Interval		Pedes	trians Iwalk	
Time	L	T	Bikes	Т	R	Bikes	L	R	Bikes	Bikes	Total	North	South	East	West
7:00 AM	0	15	0	5	3	0	6	0	0	0	29		0	0	2
7.15 AM	0	17	0	7	3	0	3	0	0	0	30	0	1	0	2
7.30 AM	0	18	0	4	1	0	- 4	0	0	0	27	0	1	0	2
7.45 AM	0	20	0	4	0	0	7	2	0	0	33	0	1	Ő	0
8.00 AM	0	18	0	5	2	0	5	3	0	0	33	l ő	1	0	0





Jacoby Rd & Newton St

Wednesday, May 02, 2018 7:00 AM to 9:00 AM



Heavy Vehicle 5-Minute Interval Summary 7:00 AM to 9:00 AM

Start Time		Jacoby I	Rd	South Jacol	bound by Rd		E	estbound lewton SI		Westbound Newton St		Interval
Time	L	Т	Total	Ť	R	Total	L	积	Total		Total	Total
7:00 AM	0	0	0	0	0	0	1	0	1 1		0	1
7:05 AM	0	0	0	0	0	0	0	0	0		0	0
7 10 AM	0	1	1	0	0	0	1	0	1		0	2
7.15 AM	0	0	0	0	0	0	0	0	0		0	0
7:20 AM	0	1	1	0	0	0	0	0	0		0	1
7.25 AM	0	0	0	0	0	0	0	0	0		0	0
7:30 AM	0	0	0	0	0	0	0	0	0	- 1	0	0
7:35 AM	0	0	0	0	0	0	0	0	0		0	0
7:40 AM	0	0	0	0	0	0	0	0	0		0	0
7 45 AM	Ö	Ö	0	0	0	0	0	0	0		0	0
7:50 AM	0	0	0	0	0	0	0	0	0		0	0
7.55 AM	0	0	0	0	0	0	0	0	0		0	0
8 00 AM	0	0	0	0	0	0	0	0	0		0	0
8.05 AM	0	0	0	0	0	0	0	0	0		0	0
8 10 AM	0	0	0	0	0	0	0	0	0		0	0
8 15 AM	0	0	0	0	0	0	0	0	0		0	0
8.20 AM	0	0	0	0	0	0	0	0	0		0	0
8:25 AM	0	0	0	0	0	0	0	0	0		0	0
8 30 AM	0	0	0	0	0	0	0	1	1		0	1
8 35 AM	0	1	1	0	0	0	0	0	0		0	\$
84DAM	0	0	0	0	0	0	0	0	0		D	D
8 45 AM	0	Ó	0	0	0	0	0	0	0		0	0
8 50 AM	0	0	0	0	0	0	0	0	0		0	0
8.55 AM	0	0	0	0	0	0	0	0	0		0	0
Total Survey	0	3	3	0	0	0	2	1	3		0	6

Ó

Heavy Vehicle 15-Minute Interval Summary 7:00 AM to 9:00 AM

Interval Start		Northbo Jacoby	und Rd	South Jaco	bound by Rd		l	Eastbound Newton SI		Westbound Newton St		Interval
Time	L	T	Total	т	R	Total	L	R	Total		Total	Total
7:00 AM	0	1	1	0	0	0	2	0	2		0	3
7 15 AM	0	1	1	0	D	0	0	0	0		0	1
7.30 AM	0	0	0	0	0	0	0	0	0		0	0
7.45 AM	0	0	0	0	0	0	0	0	0		0	0
MA 00.8	0	0	0	0	0	0	0	Ó	0		0	0
8 15 AM	0	0	0	0	0	0	0	0	0		0	0
8 30 AM	0	1	1 1	-0	0	0	0	1	1		0	2
8:45 AM	0	0	0	0	0	0	0	0	0		0	0
Total Survey	0	3	3	D	0	0	2	1	3		0	6

Heavy Vehicle Peak Hour Summary 7:50 AM to 8:50 AM

By Approach	in	North Jaco Out	bound by Rd Total	In	Soutt Jaco Out	bound by Rd Total	In	East New Out	ton SI Total	In	West New Out	bound ton St Total	Total
Volume	1	1	2	0	1	1	1	0	(1	0	0	0	2
PHF	0.25		1.	0.00			0.25			0.00			0.25

By Movement	Northbound Jacoby Rd			Southbound Jacoby Rd			Eastbound Newton St			Westbound Newton St	Total
	L	Т	Total	т	R	Total	L.	R	Tota	Total	L
Volume	0	1	1	0	0	0	0	- 1	1	0	2
PHF	0.00	0.25	0.25	0 00	0.00	0.00	0 00	0.25	0.25	0.00	0.25

Heavy Vehicle Rolling Hour Summary 7:00 AM to 9:00 AM

Interval Start Time	Northbound Jacoby Rd			Southbound Jacoby Rd			1	Eastbound Newton St	Westbound Newton St	Interval
	L	T	Total	Т	R	Totat	L	R Total	Total	Total
7:00 AM	0	2	2	0	0	0	2	0 2	0	4
7.15 AM	0	1	1	0	0	0	0	0 0	0	1
7:30 AM	0	0	0	0	0	0	0	0 0	0	0
7:45 AM	0	1	1	0	0	0	0	1 1	0	2
8:00 AM	0	1	1	0	0	0	0	1 1	0	2




5-Minute Interval Summary 4:00 PM to 6:00 PM

Interval		Northbo	und	Southb	ound		E	astbound		Westbound			Peder	trians	
Start		Jacoby	Rd	Jacob	y Rd			Newton St		Newton St	[interval		Cros	swalk	
Time	_L_	T	Bikes	T	R	Bikes	L	i R	Bikes	Bikes	Total	North	South	East	West
4:00 PM	0	0		1	0	0	0	0	0	I I 0	1	0	0	0	0
4:05 PM	0	1	0	0	0	0	0	0	0	0	1	0	0	0	0
4.10 PM	0	0	0	÷ 0	2	0	1	0	0		3	0	0	0	0
4:15 PM	0	0	0	1	2	0	0	0	0	0	3	0	0	0	0
4.20 PM	0	1	0	0	1	0	1	; 0	0	0	3	0	0	0	0
4:25 PM	0	0	0	1 1	0	0	1	0	0	0	2	0	0	0	0
4 30 PM	0	0	0	2	1	0	1	0	0	0	4	0	0	0	0
4:35 PM	0	0	0	0	3	0	0	Î O	0	0	3	0	0	0	0
4 40 PM	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
4 45 PM	0	1	0	3	D	0	1	0		0	5	0	0	0	0
4.50 PM	0	1	0	0	1	0	1	0	1 0 1	0	3	0	0	0	0
4.55 PM	0	0	0	0	0	0	0	1 0	0	0	0	0	0	0	0
5:00 PM	0	0	0	1 4	3	0	0 1	1 0	0	0 1	7	0_0	0	0	0
5:05 PM	0	0	0	2	1	0	0 1	1 0	0	1 1 0	3	0	0	0	0
5:10 PM	0	2	0	1	1	1 0 I	1	0	0	0	5	0	0	0	0
5:15 PM	0	0	0	1 1	3	0	0 1	0	0	i i o	- 4	0	0	0	0
5:20 PM	0	4	0	D	4	0	0	0	0		8	0	0	0	0
5:25 PM	0	2	0	1	2	0	1	0	0		6	0	0	0	0
5 30 PM	0	0	0	1	1	0	0	0	0	0	2	0	0	0	1 1
5:35 PM	0	0	0	1	0	0	0	0	0	0	1	11- <u>0</u> -	0	0	1
5.40 PM	0	2	0	1	0	0	2	1 0	0		5	0	0	0	0
5 45 PM	0	0	0	1	3	0	0	0	0		4	11-0-	0	0	0
5 50 PM	0	0 1	0	1	5	0	4	1 0	1 0	0	10	1 0	0	0	0
5.55 PM	0	1	0	0	2	0	2	0	10		5	1 0	0	Ō	Ő
Total Survey	0	15	0	22	35	0	16	0	0	0	88	0	0	D	2

15-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		Nort Jac	thbour oby R	ndi d		South Jaco	bound by Rd			E: N	estbour lewton 1	nd SI		Westbound Newton St		Interval		Peder Cros	strians swalk	
Time	L L	5 T.	1	Bikes	÷	T	R	Bikes	L			R	Bikes	1 1	Bikes	Total	North	South	East	West
4:00 PM	0	1 1	ì	: 0	316	1	2	0	1	1	i i	0	0		0	5	0	. 0	0	1 0
4:15 PM	0	1		0		2	3	0	2	1		0	0		0	8	0	0	0	0
4 30 PM	0	0		0	1	2	4	0	1	Ī	1.1	ò i	0	L L	0	7	0	0	0	0
4 45 PM	0	2		0		3	1	0	2	1		0	0		0	8	0	0	0	0
5 00 PM	0	2		0		7	5	0	1	1		0 1	0	1	0	15	0	-0	0	: 0
5 15 PM	0	6		0		2	9	0	1			0	0		0	16	0	0	0	0
5.30 PM	0	2		0	1	3	1	0	2			0	0		0	8	0	0	0	2
5 45 PM	0	1 1	1	0	1	2	10	0	6	1.1		0	0	1	0	19	0	0	0	0
Total Survey	0	15		0		22	35	0	. 16	ļ		D	0		0	88	0	0	0	2

Peak Hour Summary 5:00 PM to 6:00 PM

By		Norti Jaci	nbound oby Rd			South Jaco	bound by Rd			East	bound ton St			West New	bound ton St		Total		Pedes	itrians swalk	
Approach	ាត	Out	Tota	Bikes	п	Out	Total	Bikes	la.	Qut	Total	Bikes	In	Qut	Total	Bikes		North	South	East	West
Volume	11	14	25	0	39	21	60	0	10	25	35	0	0	0	0	0	60	0	0	0	2
%HV		0	0%			D I	0%			0	0%			0	3%		0.0%	1			
PHF		C	.48	- minimi min voin		0	81			0	42			0	00		0 79				
By		Nort Jac	bound by Rd			South Jaco	bound by Rd			East	bound ton SI			West	bound Ion St		Total	1			
WAVEGICIETA	L	T	1	Total		T	R	Total	L		R	Total				Total					
	0	5.4		11		14	25	39	10		0	10		l.	1	0	60	1			
Volume	F 0																a second s				
Volume %HV	0 0%	0.0%	NA	0.0%	NA	0 0%	0.0%	0 0%	0.0%	NA	0.0%	0 0%	NA	NA	NA	0.0%	0.0%				

Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		No J	irihi acol	bound by Rd			South Jaco	bound by Rd	4			Eastbo	und St			Westbound Newton St		Interval		Pedes Cross	trians walk	
Time	L		r P		Bikes		T	R	E	ikes	L		R	Bike	85	i i i	Bikes	Total	North	South	East	West
4:00 PM	0	1.1.4	1		0	-	8	10	1	0	6		0	0			0	28	0	0	0	0
4:15 PM	0	1.1	5		0		14	13	1	0	6		0	0			0	38	0	0	0	0
4:30 PM	0	1	0		0	1	14	19		0	5		0	0	_1	1	0	48	0	0	0	0
4:45 PM	0	1	2		0		15	16		0	6		0	0	- I	1	0	49	0	0	0	2
5:00 PM	0	1	1		0		14	25	1	0	10	1	0	0			0	60	0	0	0	2





Jacoby Rd & Newton St

Tuesday, May 01, 2018 4:00 PM to 6:00 PM



Heavy Vehicle 5-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		Jacoby I	and Rd	South Jaco	bound by Rd		E	astbound lewton St		Westbound Newton St		Interval
Time	_L	T	Total	T	R	Total	L	R	Total		Total	Total
4:00 PM	0	0	0	0	0	0	0	0	0	1151 BC	0	0
4:05 PM	0	0	0	0	0	0	0	0	0		0	0
4:10 PM	0	0	0	0	0	0	0	0	0		0	0
4.15 PM	0	0	0	0	0	0	0	0	0		0	0
4.20 PM	0	1	1	0	0	0	0	0	0		0	1
4.25 PM	0	0	0	1	0	1	0	0	0		0	1
4.30 PM	0	0	0	0	0	0	0	0	0		0	0
4.35 PM	0	0	0	0	0	0	0	0	0		0 L	0
4:40 PM	Ð	0	0	0	0	0	0	0	0		0	0
4:45 PM	0	0	0	0	0	0	0	0	0		0	0
4:50 PM	0	0	0	0	0	0	0	0	0		ů l	0
4:55 PM	0	0	0	0	0	0	0	0	0		o l	0
5:00 PM	0	0	0	0	0	0	0	0	0		0	0
5:05 PM	0	0	0	0	0	0	0	0	0		0	0
5:10 PM	0	0	0	0	0	0	0	0	0		0	0
5:15 PM	0	0	0	0	0	0	0	0	0	1.1	0	0
5.20 PM	0	0	0	0	0	0	0	0	0		0	0
5:25 PM	0	0	0	0	0	0	0	0	0		0	0
5.30 PM	0	0	0	0	0	0	0	0	0		0	0
5.35 PM	0	0	0	0	0	0	0	0	0		0	0
5:40 PM	0	0	0	0	0	0	0	0	0		0	0
5:45 PM	0	0	0	0	0	0	0	0	0		0	0
5-50 PM	0	0	0	0	0	0	0	0	0		0	0
5:55 PM	0	0	0	0	0	0	0	0	0		0	Ô
Total Survey	0	1	1	1	0	1	0	0	0		0	2

Heavy Vehicle 15-Minute Interval Summary 4:00 PM to 6:00 PM

Interval Start		Jacoby	und Rd	South Jaco	bound by Rd		I	Essibound Newton St		Westbound Newton St	Interval
Time	1.	Ť	Total	Т	R	Total	L	R	Total	Tot	al Total
4:00 PM	0	0	0	0	0	0	0	0	0	0	0
4.15 PM	0	1	1	1	0	1 1	Ó	0	0	0	2
4:30 PM	0	0	0	0	0	0	0	0	0	0	0
4 45 PM	0	0	0	0	0	0	0	0	0	0	0
5 00 PM	0	0	0	0	0	0	0	0	0	0	0
5:15 PM	0	0	0	0	0	0	0	0	0	0	0
5:30 PM	0	0	0	0	0	0	0	0	0	0	0
5:45 PM	0	0	0	0	0	0	0	0	0	0	0
Total Survey	0	11	1	1	D	1	0	0	0	0	2

Heavy Vehicle Peak Hour Summary 5:00 PM to 6:00 PM

By Approach	Northbou Jacoby F In Out To	and Rd otal	In	South Jacos Out	bound by Rd Total	In	East New	ton St Total	In	West New Out	bound ton SI Total	Total
Volume	0 0	0	0	0	0	0	0	0	0	0	0	0
PHF	0.00		0.00			0.00			0.00			0.00

By	Northbour Jacoby R	nd d	South Jacol	bound by Rd			Eastbound Newton St		Westbound Newton SI	Total
HIG WEITHEITE	L T	Total	Т	R	Total	L	R	Total	Total	
Volume	0 0	0	0	Û	0	0	0	0	- 0	0
PHF	0.00 0.00	0.00	0.00	0.00	0.00	0 00	0.00	0.00	0.00	0.00

Heavy Vehicle Rolling Hour Summary 4:00 PM to 6:00 PM

Interval Start		Northbo Jacoby I	und Rd	South Jaco	ibound by Rd		E	astbound Vewton St		Westbound Newton SI	Interval
Time	L	Т	Total	T	R	Total	L	R	Total	Total	Total
4 00 PM	0	1 1 1	1	- 1 - 1 -	0	1	0	0	0	0	2
4 15 PM	0	1	1	1	0	1	0	0	0	0	2
4 30 PM	0	0	0	0	0	0	Ó.	0	0	0	
4 45 PM	0	0	0	0	Ū.	0	0	0	0	0	0
5:00 PM	0	0	0	0	0	0	0	0	0	0	n n



180 1/2018 0P SANDY. 5 D	CLACKANAS COU	ALRI			DOBARU	OREGON DE TRANSF TRANSF D RD AL EAGLE	PARTMENT OF ORTATION DI CRX-SAMDY 1 = 4	TRANSPOL ATA SECTI NO UNBAN NO UNBAN NO UNBAN NO	NTTION TR ON CTASH A H-SVSTEM CRA of Sandy, C 28 Cramh	unisportation anavLysis and an Listing Jatkama Cou records show	n DEVELOPMENT D REPORTING mer, 01/01/:	unit unit 2012 to 12,	9102/TE/					Fage: 1
	S M DATE C O DAY H R TIME L K LAT 03/10/2014	CLASS DIST FROM LOHO 6 LOHO	CITY STREET FIRST STREET SECOND STREET LAS DUBARKO RD EACLE CEK-SAMDY HY	RD CHAR DIRECT LOCTN INTER	INT TYPE IMEDIAUI LEGS (#LANES) CHOSS	INT-REL TRAF CONTL N STOP SIGN	OFFRD N RNDBT 5 DRVNY L N C	THR CR URF CO IGHT SVI LR 0-1	ASH LL ISTOP Va ISTOP 01	SPCL USE TRLR OTY OWNER TYPE NONE 0 PRVTH	MOVE FROM TO BACK	PRTC PRTC	INJ SVRTY	A S G B LICNS E X RES	PED LOC ERROR	ACT EVENT	CAUSE 10	
	45 23 23 23 23	76 122 15 48 39	017200100500	1 Q	ō		. 2	A A A	5	FSHCR CAR PSHCR CAR NONE 0 FRVTE FSNCR CAR	a a 21.55 au	01 DRVR 03 DRVR	INIC	22 M OR-Y OR<25 26 F OR-25 DR<25	011	000 000 000 000 000	50 00 00 00	Q.
21 22 X	N N 02/07/201 FR 4P 25 23 26 23 26 76124	4 16 0 122 15 48 19 9599	DURARIO RD EAGLE CRK SANDY HY	1 NTER 06	CHOSS	% 510P 81G8		NOW S. NO RE.	1570P 01 AR 0	NONE 0 UNX34 UNX30MN NONE 0 PRVTE PSNGR CAR	STRGHT B H STOP F - M	ORVR	NONE	00 Unk UrkK UnkK 38 P OR Y OR AZE	047,0	600 600 011 000	10 00 10 00 00	
22 22 23	02/07/201 FR 12 22.76124 22.76124	4 16 0 - 122 15 48, 1879599	DUBARKO RD EAGLE CRK-SAUDY HY	14172R 6 0.6	0 L4025	H STOP SIGN	x x x	AY PDM	1570P 01 AR 0 02:1 1 1	HOHE 0 PRVTE PENGR CAR PENGR CAR PENGR CAR PENGR CAR PENGR CAR PENGR CAR	S170541 5.700 5.100 5.100 5.100 5.100 5.100	01 DRVR 01 DRVR 01 DRVR	3RON	21 F 0R-Y 0R-25 19 M 0R-Y 0R-25 00 Unk UNK	026 000	124 001 124 000 011 011 011 000	10 00 00 00 00 00 00 00	(
N N 94 6	02/07/2014 FR 4P 45 23 22.76124	16 0 -122 15 46, 3879599	BUBARKO RD EAGLE CRK SAIDY HY	INTER 8 06	CHOSS 0	H STOP SICK	R R C	KEM 6	1510P 01	HONE 0 PRVTE PSNGR CAR NONE 0 PRVTE PSNGR CAR	STRGHT E .M STOP E .M	01 DRVR 01 DRVR	3NOM	36 М ОR-Y 0R-25 45 М ОR-25 45 М ОR-25	000	124 000 124 000 011 011	07 00 00 00 00	()
N N N	N N 12/23/2014 SU 5P 45 23 45 23	2 16 16 122 15 48 3879667	DUBARKO RD EAGLE CRK SAYDY HY 017200100500	INTER SW OS	CROSS D	N STOP SIGN	× N N	11 LUI LUI	7 K 08.1	NONE 0 PRVTE PSNGR CAR	STRGHT N = S	01 DRVR	anon	47 M OR Y OR<25	08	124,043 000 124,043 000	10 00 10	

Page: 2		Ċ									\langle	Ć					eport forms is voperify
	CAUSE	00 00	00	00	07,29	00	27,07,32	00	27,07,32	00	00	00	07	00	0.0	00	ause submital of crash r 04, may result in fewer p
	ACT EVENT 000 124,043	000 000 324,043 000	022 000	000	000	012 000		000	010	000	012 000	000		000	000	012 000	Homers However, bec
	RRACARS	000	000		043,026	000			016,043,052	000	000	0,00			000	000	r sporting required
	A S LECKS PED B LECKS PED B X RES LOC	۵۷ ۵۶ ۴ ۲۹ ۴	30 \$ 08 ¥ 08 ¢25		19 M 08 Y 08<25	36 H 0R-Y 0R-25			30 M 08-Y 0R≼25	4 10	18 F OR-Y 201-26	08×25 26 M 08×∑ 08×25			0.0 Unk UNK UNK	00 Unk Urik Urik	To providing the highest qualit tanges to DM/V's vehicle crash
r r to 12/31/2016	PRTC INJ TYPE SVRTY	2 PSNG INJC	I DRVR NOWE		I DRVR INJC	I DRVR INJC			I DRVR NONE	5 PSNG NO<5	L DRVR NONE	I DRVR INCH			I DRVR NONE	I DRVR NONE	e hore Legislative c
DEVELOPMENT DI REPORTING UNIT (ty. 01/01/2012	MOVE FROM TO STRUHT N - F	0 STRICHT N - S 0 CTB CUT	D N-S	51RGHT 5W - NE	0	SM-NE 0	STRGHT	NS SH	0	STRGHT NE-SW 03	STOP HE-SW 01	STRCHT SW-RE 01	STRGHT	S -12	0	5 -%	h Analysis and Repo
TRANSPORTATION SH ANANLYSIS AND CHASH LISTING , Clackamas Coun ish records show	SPCL USE TRLA OTY OMNER VO TYPE 01 NOVE 0	PSNGR CAR 01 NOHE D PRVTE PSNGR CAR	PRVTE U PSNGR CAR	01 NONE 0 PRVTE	PSNGR CAR	02 NONE 0 PRVTE PSNGR CAR	01 NOME 0	PRVTE	PSNGR CAR	01 NONE O PRVTE PSNGR CAR	02 NONE 0 PRVTE PSNGR CAR	03 NGNE 0 PRVTE PSNGR CAR	6 BNON TO	N/A	PSNGR CAR	D2 NONE 9 N/A PSNGR CAR	AS 811 720 The Cres take pertamang to a san
SPORTATION CTION CRAW NUN-SYSTEM Ity of Sandy dty of Sandy	CRASH COLL SVRTY			5.15TOP REAR	CHI		S-1570P	SS-0	L741				s-1570P	REAR	6D0	٠	rasregurad in C made that all de
THENT OF TRAN ATION DATA SE URBAU K-SANDY HT, C	JFFRD MTHR URBT SURF DRUMY LIGHT			r CLR	ł pusk		t CLD	I DRY	AV DAY				RAIN	Taw 1	(DLIT		ni of Transportation can assumences be
RECON. DEPAR TRANSPORT	187-REL RAP-			I SNO	-			ITOP SIGN	-				~	1 BADS	-		i Dregon Departme re represented nor
DUBALICO R	INT-TYPE (MEDJAM) 1 LEGS 7 (MLANES) (CROSS 3	0		CROSS	3	Q				CROSS N	~	0		នេះសាការដែល fo the សេក្ខាកាក្ខ បានសំទន ៨ ស
	RD CHAR DIRECT LOCTH			INTER	06		INTER	10 M	9.0				INTER	SW	06		poice crash repo
	CITY STREET FIRST STREET SECNID STREET LRS			DUBARKO RD EAGLE CRK-SANDY HY	017200100200		DUBARKO RD	EAGLE CRM-SAMDY HY	017200100500				DUBARKO RD	EAGLE CRK-SANDY HY	017200100500		uled from mownaug driver and s and Reporting Unit can not g ewide Crash Data File
A.I.MO.	CLASS DIST FROM			1015 \$6	2.76 -122 15	21 m DØ	016 16		2 76 122 15	h D F			016 16		2.76 -122 15	A. - D P	d an this report is com er. the Crash Analysis or inclusion an the Stat
0 2018 ~ SAIDY, CLACKAMAS C	S D P R S M DATE E A U C O DAY E L G H R TIME D C S L K LAT			и и и 07/27/2 МО	8P 45 23 2		N N N N 03/04/2	FR	4P 45 23 2				N N N N 02/17/2	ME	5P 45 23 2:		 The information contained made of the individual dim rady crashes being eligible R
CDS38 06/06/	SER# INVEST RD DPT UNLOC7			03026 CITY	N N		01095	CITY	И				E9400	CITY	NN		Disclarm the respt damage i

e used			(Ó										Ó						ash report forms is ver property
	20 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	07	0.4	00	00	30,03	00	30,03	8.8	02	00	00	00	03	02	00	00	00 02		course submittal of or 004, may result at fer
	1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 -		000	012	012		000	000	000		000	000	000	015		000	000	015 000		stomers However as tent, effective 01/01/2
	8 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	2012	026	000	000			058, 021	000			000	000	038			000	028		uakry crash data to cu ash reporting required
	E LICHS PED		M OR Y OR ZS	P 0R-Y	0X < 25			M 07H-Y 0R<25	Unk OTH=Y N-RES			F 0R Υ 08<25	×	M 0R-Y 0R-25			₽ \$U\$₽ 0R<25	Unk UNK	UDAK	providing the highest q ges to DMV's volucle ci
/31/2016	INJ C	2 4 4 5	NONE 18	11430 26	110×5 04			NONE 21	LE BRON			₽Z ZNCN	INJC 21	110ME 16			1NJC 52	DO SHCH		is committed to Legrslative chang
tent division the CMIT 1/2012 to 1:	PRIC	1 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	01 DRVR	01 DRVP	03 PSHC			01 DAVR	OI DRVR			01 DRVR	02 PSNG	01 DRVR			OI DRVR	OI DRVR		nd Reporting Univ accurate Note
ION DEVELOPM AND REPORTI G County, 01/0 hown	MOVE	57RGHT	ere oli " da ĉe	SWIRE	SM-NE	STRGHT	3 - 14		STRGHT S -N	STRGHT	N S		STRCHT 5 N	STRGHT W -8	STRGHT	S- 11		TURN-L E .S		Cash Analysis & single crash are
TRANSPORTAT SH ANAVLYSIS CRASH LISTIN , Clackaaas (ash records s	SPCL USE TRLR OTY DMNER	01 NONE 0	PSNGR CAR	02 NONE 0 PRVTE PSNCR CAR	02 NONZ 0 PRVTE PSNGA CAR	01 NDNE 0	PRVTE	PSNGR CAR	02 NOME 0 PRVTE PSMGR CAR	0 3NON 10	PRVTE	PSRGR CAR	0.1 NONE 0 PRVTE PSNGR CAR	02 NONE 0 PRVTE PSNGR CAR	01 NONB 0	PRVTE	PSNGR CAR	0.2 NONE 0 UNIXU UNIXUOWN		RS 811 720 The (tais pertaming to a
(SPORTATION - ECTION - CRA N NON-SYSTEM N NON-SYSTEM	CRASH COLL	S-1STOP	114.2			AJGL-OTH	ARGL	6DO		ANGL-OTH	TURN	LAI			ANGL-OTH	TURM	CH1) as required in 0 made lihal all de
MENT OF TRAU ATION DATA S URBA L'SAMDY HY, (9 = 12	FFRD WTHR MDBT SURF	RAIN	DARK			RAIN	WET	TAT		GP	DRY	NSUG			CLR	DRY	DAY			t of Transportation an assurances by
EGON DEPAR TRANSPORT	IT REL C					N	OP SIGN N	14		H	N NDIS 40	14			X	N NDIS 40	14			report Departmen represented nor o
10 DURANKO RU	INT TYPE (MEDIAN) IN LEGS TH	CROSS N	6			CROSS N	57			CROSS N	ST				ROSS N	51	~			submitted to the (Myarg crashes are
	RD CHAR DIRECT	INTER	04			INTER	CH	04		INTER	CN	4			INTER	CH	03			olice crash reports inarriee (hat all qui
	CITY STREET FIRST STREET SECOND STREET	DUBARKO RD	017200100500			DUBARKO RD	EAGLE CRK SAUDY HY	017200100500		DUBARKO RD	EAGLE CRK-SANDY HY	017200100500			DUBARKO RD	EAGLE CRK-SANDY HY	017200100500			ed from indvndual drivet and p ind Reporting Und can nol gu wde Crash Data File
4	CLASS DIST FROM	16	122 15	48_3879621		16		-122 15 48 3879667		16		-122 15 48 1879667			16		-122 15			is report is compli- e Crash Analysis . usion in the Stater
10 2018 F SARDY, CLACKAMAS COUR	E D F R S M DATE E A U C G DAY E G M R TIME D C C E K TAT	N N N N 01/06/2012	5P 5P 45 23	22,7612385		Y N N N 02/08/2012	WE	12P 45 23 22,7612384		N N N N 03/24/2012	SA	6P 45 23 25 7612384			N N N N 08/03/2012	FR	5P 45 23 22 7612384			sr The information contained in Ib subling of the excinctual dimer. Ib any crashes being eligible for incl
CDS36 06/06/ CITY 0	SER# INVEST RD DPT	00010	N			00516	CITY	н		01124	CITY	N			02844	CITT	н			Desclamme the respo damage

		Ò		
Pad				- Almadoud version
	Cattoli 02 00	00 00 02 02	02 00 02 00 00	92 90 90 92 92 92 92 93 93 93 93 93 90 90 90 90 90 90 90 90 90 90 90 90 90
	ACT EVENT	000 000 000 015	000 000	000 000 015 015 015 000 000 000 000 000
	ROR	000 823	028	000 B25 B00 B00 B00 B00 B00 B00 B00 B00 B00 B0
	LICNS PED	08 4 25 08 4 25 08 4 25 08 4 2	08.Y 08.2% 08.Y 08.Y	OR - Y OR - Y OR - Y OR - Y OR - Z OR - Z OR - Z OR - Z OR - Z OR - Z S OR - Z OR - Z O
2016	LU A CU	ONE 36 M O<5 02 P ONE 21 P	D11E 1.6 F \$4.18 25 M	ONE 25 M ONE 25 M NJC 25 M NJC 78 M SJC 78 M
HUISIUN TIN 12/31	PATC I	OI DRVR N #2 PSNG N 01 DRVR N	01 DRVR N 01 DRVR 1	01 DRVR // // // // // // // // // // // // //
DEVELOPMENT REPORTING U NEY, 01/01/2 4 n.	MOVE FRCM TO SIRGHT S -N	STRGHT S - H STRGHT W - B	STRGHT E -M STRGHT N = S	Sthour structur struc
TRANSPORTATION SH ANAYLYEIS AND CRASH LISTING • Clackass Coun ssh records show	SPCL USE TRLR OTY OMNER Ve TYPE 01 NONE 0 PRVTE	ESNGR CAR 01 NOILE 0 PENGR CAR PENGR CAR 02 HOME 0 PENGR CAR	01 NONE 0 FRVTE PSNJR CAR 02 NONE 0 FRVTE PSNJR CAR	01 HUNE 0 PRVTE PRVTE 0 PRVTE 0 PRVTE 0 PRVTE 0 PRVTE 0 PSVER CAR PSVER CAR PSVER CAR PSVER CAR PSVER CAR
SPORTATION - CTICN - CRAN HUN SYSTEM Lty of Sandy of 28 Cra	CRASH COLL SVRTY ANGL-OTH ANGL	QG	ANGL-OTH JOINA CHI	Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph Audul-orph
TION DE TRAN TION DATA SE URBAN SANDT HT, C 13 - 15	FRD WTHR DBT SURF VMY LIGHT CLR	DAY	CLR DRY DARK	cLR BUSK CLD CLD BRY BAY CTD A R A A A CLR A A CLR A A CLR A A CLR A A CLR A CLR A CLR A C CLD A CLD A C CLD A CLD A C CLD A CLD A C CLD A CLD CLD A CLD CLD CLD CLD CLD CLD CLD A CLD CLD CLD CLD CLD CLD CLD CLD
031 DEFARTY TRANSPORTA e eagine crr.	RZL OF 1 D3 1 D3 1 D3 1 D3 1 D3 1 D3 1 D3 1 D3	. 2	61GN 26	stana k stana k stana k stana k n stana k stana k stattana k stana k s
OREG	T.TYPE EDIANJ INT- LEGS TRAF LANES) CONT SS N STOP		N 012	IS N N STORE STORE
-	IN CHAR (M ECT (M ER CROS	0	CRO.	ER CR05 ER C905 seaf-reports sub
	DIRI DIRI LOCT	5	INT CN 01	1711 1711 10 10 10 10 10 10 10 10 10
	C1TY STREET FIRST STREET SECOND STREET LRS DUBARKO RD DUBARKO RD	01/2001 00500	DUBARKO RD EAGLE CRK SAUDI	DUTBARKO RD EAGLE CRK-SAND) 017200100509 DUTBARKO RD EAGLE CRK-SAND) 017200100500 017200100500 017200100500
Ĭ	CLASS DIST FROM LONG 16	-122 £5 48 387966	16 -122 15 48 387966	16 = 122 15 48 387348 16 = 122 15 48.387348 48.387348
10 2016 P SNDY, CLACKANAS CODA	5 D P R S M DATE 7 E A U C O DAY 8 L G H R TIME 0 C S L K LAT 1 N N 09/05/2012 51	65 55 22.1912384	ы и и № 99/08/2012 БА 9P 25 23 2306	 N. H. H. K. 01/08/2013 F. R. 8 9 S. 22, 5623199 S. 22, 5623199 K. K. 04/21/2013 K. M. 04/21/2013 S. 12 S. 22, 5623199 S. 22, 5623199 S. 24, 562319 S. 25, 5623

Page: 5		Ó		6	u report forms is report
	CAUSE 00	00 02 00 02 02	2 8 8 8 8 1	02 00 03 03 03 03 03	00 00 conce submitted of cras COA, may result of fores
	ACT EVENT 000	000 000 000 000	000 000 000	000 000 000 000 000 000 000	000 000 015(Chrefe Hovevier br ument, effective 01/01(
	ERIFOR	000 000	000 000	028,004 010 028 028	aon aky crean dee to
uriston 417 13 to 12/31/3016	PRTC INJ G E LICKS PED PRTYPE SVRTV E X RES LOC	02 PSNG INJC 85 P 01 DRVR NOWE 18 P OR-25 01 DRVN NOWE 30 M OTH-Y NI-RES	01 DRVR 110348 39 P 08.Y 08.255 02 PSUG 134JC 12 P	01 DRVR INJC 67 M 0R-Y 01 DRVR INJKE 36 P 0R-Y 03 DRVR INJKE 35 P 0R-Y 03-25 03-25	92 PSHG INJE 57 P porting Um is committed to providing the highest qu and Mate Legislative changes to DMP3 vence of
DSVELOPMENT REPORTING UN Ity, 01/01/20	BYCM FROM TO TRGHT N- S	STRGHT K • 2 STRGHT S = N	STRGHT N - S STRGHT N - S TRGHT N - S	5118GHT 8-18GHT 8-18GHT 8-18GHT 18-18 8-18GHT 18-18 8-18GHT 18-18 8-18GHT 18-18 18-	STRGIT
. TRAMSPORTATION SH ANAYLYSIS AND CRASH LISTING CRASH LISTING , Clackamas Coun	SPCL USE TRLR QTY OMNER VM TYPE 0 RONE 0 PRVTE	PSNGR CAR 01 NOVE 0 PRVTE 0 PSNGR CAR PSNGR CAR PSNGR CAR	01 MONE 0 PRVTE PSHCR CAR PSHCR CAR 01 NONE 0 PRVTE PSHCR CAR PSHCR CAR 02 NONE 0	PSNICE CAR PSNICE CAR PSNORE OR PSNORE CAR PSNORE CAR PSNORE CAR PSNORE CAR PSNORE CAR PSNORE CAR PSNORE CAR	P NOTE 6 PRATE CAR PISICA CAR PISICA CAR PISICA CAR PISICA CAR PISICA CAR PISICA CAR
SPORTATION SCTION CHA NON-SYSTEM NON-SYSTEM Standy Standy	CRASH COLL SVRTY	AKCL-OTH AKCL PDO	D-1 L-TURN TURN INJ	ANGL-OTH ANGL-OTH ANGL-OTH TURN TURN	as required en C
MENT OF TRAUN MTION DATA SU URBAN -SANDY MY, C 16 = 19	FFRD WIHR SURF SURF LIGHT	CLR DRY DAY	CLLD DUSK	CLR DRY DAY CLD CLD DAY	o d'Transportation assurances be
CON DEPART TRANSPORT	1.8EL 0 LF- 81 LT- D	N N SIGN	2 2 2 4	12 SIGN 23 12 SIGN 23 13 SIGN 23 14 SIGN 24	repartmen
ORE CLE CLEARED	INT-TYPE IMEDIAN) INT LEGS TRA	N SSO	R055 * 1	ROSS N STT STT STT STT	utumated to the O
	AD CHAR DIRECT LOCTH	INTER C	INTER C	INTER C CM 03 03 03 03 03 0 0	like cash reports andee that all qua
	CITY STREET FIRST STREET SECOND STREET LAS	DURARNO RD EAGLE CRK.SANDY AV 017200106500	UTANKON RU EAJLE CRK. SANDY HY 017200100500	EAGLE CR. END HY EAGLE CR. END HY 017280106588 017280106588 EAGLE CR. EAND HY EAGLE CR. EAND HY	od bras andri dana dana dana dana dana dana dana dan
4.L.N	CLASS DIST FROM LONG	3 16 -122 15 1 48,187348	3 16 	12215 48.187559 48.187559 16	this report a comp the Crange Analysis obsect in the Stat
10380 6/06/2018 ITY OP SANDY, CLACKAMAS COUN	SA D SA P S M DATE NYEST E A U C O DAY D DFT E L G H R TIME VLOC? D C S L K LAT	стос/го/то и и и и 5,52 СМ 71 сс 22 52,55 25	4702 N.N.N.N.N.12/06/2013 117 FR 49 22.7623199	0137 И И И И 01/25/2014 ОИЕ БА 15 45 23 45 23 22.76124 17 50 109/2014 17 50 109/2014 15 15 15 15 15 15 15 15 15 15 15 15 15	restand for intermation contained in restand in the intermation contained in the intermation contained on the intermation contained of the intermation contained

NDV, CLACKAWAS COUNTY			DUBARK	OREGON. DE TRANSP O RD at EAGLE	PARTHENT (ORTATION CRK-SANDT 20 -	PF TRANSPON DATA SECTI URBAN NOU HT, CLTY 24 of 24 of	ithium	RANSPORTATION ANAVLYSIS ANI VASH LISTING Clackaman Cou Liccords show	DEVELOPMENT D REPORTING (Rty, 01/01/2 m.	UNIT UNIT 1012 to 11/1	1/2016						
D H # M DATE CLASS A U C O DAY DIST L G H R TIME FROM	CITY STREET FIRST STREET SECOND STREET	RD CHAR DIRECT	1HT-TYPE (MEDIAN) LEGS	ENT-RIL TRAF	OFFRD RNDBT	MTHR CRI SURF COL	L L	EPCL USE TRLR OTV OWNER	MOVE FRCM	PR1C PA TYOF	1%J SVDTV	A S EPC	13 124 121	BOB		CATER	
CSLKIAT WAG	DRS .	INCIN	ICONVITA I	1	I MAND	1017	0	PRVTE 0 PSNCR CAR	S- M	01 DRVR	awow	61 P OR-Y OR-Z	6	0	000	00	
N N N 11/18/2014 16	DUBARKO RD	INTER	CROSS	N STOP STCN	8 8	CLR ABI	10 H10-11	RVTE 0	STRGHT N -S						000	02	(
45 23 22,76 -122 15	017200100500	01	0	2	. 2	DLIT PIX		PSNGR CAR	2	OI DRVR	NONB	21 M OR-Y OR<2	0	a	000	00	3
5 5 9							1 0.	PRVTE PRVTE PSHGR CAR	E N	O1 DRVR	HORE	38 M OR-Y 0R<21	0		015 000	00	
1 N N % N 11/22/2015 16	DUBARKO RD	INTER	CROSS	28	Н	CLD AB	10 HLO-71	NONE 0	TURN-L							03	
50	EAGLE CRK SANDY HY	CN		STOP SIGN	ы	DRY TUI	214	PRVTE	av- w						000	00	
4P 45 23 22 76 122 15	017200100500	03			и	DAY IN.		PSNGR CAR		01 DRVR	BLUI	53 F 0TH-7	0	24	000	50	
35 m 26 to							02	PRVTE 0 PSNGR CAR	STRCHT ME SW	01 DRVR	2NCH	19 P OR-Y	8		000	00	
							6.0	PRVTE PSNGR CAR	STRGHT EW-NE	01 DRVR	NONE	41 M 0R-7 08<21	0	o	000	00	
I N N H H 12/25/2015 16	DUBARKO RD	INTER	CROSS	И	Z	CLID ANC	1L-07H 01	10118 0	STRGHT							02	
법	EAGLE CRK-SANDY HY	ĩ		STOP SIGN	W	NET AND	JL.	BRUTE	8- N						000	00	
6P 45 23 22,76 122 15	017200100500	10	0		H	DLIT 134,		ANGR CAR		DRVR	NONE	58 M 08-Y 08-29	0	0	000	¢ΰ	
7. 1. 2. 2.							03	PRVTE PRVTE PSNGR CAR	STRCHT B .W	O1 DRVR	11/30	53 M 08-Y 0R<2	0	8	000	00	Ô
N N 06/05/2015 16	DUBARKO RD	INTER	CROSS	н	н	CLR MMC	10 HLL0-7:	NONE 0	STRGHT							02	
fr'R	EAGUE CRK-SAMDY HY	CN		NDIS dOLS	И	DRY TUI	222	PRVTE	M - 3						015	00	
7A 45 23 22.76 -122 15 48.39	017200100800	04	0		PE	DAY PDI		PSNGR CAR		01 DAVR	NONE	24 M DR-Y 0R<2	8	8	000	0.2	
							03	PRVTE PRVTE PSHGR CAR	TURN-L SN-N	01 DRVR	INDNE	29 M OR-Y OR≺25	00	0	000	00	

Fage: 7	\bigcirc		0	property as
	CAUSE 02 00 02 00 00	03 00 00 00 00 00	01,12 00 00 00 00 00 00 00	ause submital ol crasi 04. may result er fewer
	ACT. EVENT 015 000 000	015 000 000 000 000		starters However, bec
	C 678.08	900 000 000	00000000000000000000000000000000000000	puality crash dala to cu rash reporting require
	A S LICHS PEL B LICHS PEL B X RES LOO N N.RES N N.RES	и Р ОТН-У Н. RES F И. RES F ОК-У В Р ОК-У	Unk UNK Unk Unk Unk Unk Unk Unk Unk Unk Unk	pronding the highest q gas to DMV's vehicle o
ISIDK to 13/31/3016	PRTC INJ TYDE SVRTY (DRVR INJC 7: DRVR NONE 40	DRVR IIJJC 61 PSND IIJJC 61 PSND IIJJC 06	DRVR NICKE 00 DRVR NICKE 00 DRVR NICKE 00 DRVR NICKE 00	ng Unit is commuted to Note Legistative chan
DEVELOPMENT DIV REPORTING UNIT N ty, 01/01/2013	HOVE FROM FROM FTOHT E - M 01 6 FROHT 0 01	STRGHT M = E STRGHT M = E STRGHT M = C STRGHT S = N 0 1	STRGHT N. E STRGHT N. S. 01 S. N. 01 I. S. 01 I. S. 01 C. M. 01 01 01	h Analyzis and Report.
TRANSPORTATION 5H AMAYLYSIS AND CRASH LISTING , Clackamae Cour	SPCL USE TRLR OTY ONNER ONNER D1 NONE PSNGE CAR PSNGE CAR PSNGE CAR PSNGE CAR PSNGE CAR	01 HONE 0 FRVTE PSNCJR CAR PSNCJR CAR PSNCJR CAR PSNCJR CAR PSNCJR CAR PSNCJR CAR PSNCJR CAR PSNCJR CAR	01 100/12 9 11/A P PSNGR CAR PSNGR CAR PSNGR CAR PSNGR CAR PSNGR CAR PSNGR CAR	RS 811 720 The Crass Mets pertaming to a sing
ANSPORTATION - SECTION - CRAK ANN NON-SYSTEM CLLY OF SENDY Of 28 Cra	CRASH COLL I SVRTY ANGL-OTH ANGL	ANGL-OTH ANGL INJ	ANGL-OTH ANGL-OTH ANGL-OTH PDO	ion as required in O
RPARTMENT OF TR PORTATION DATA URL I CIX-SANDY HY, 25 28	OFFRD KTHR RUIDET SUDEP BAVWY LICH H CLR H DRY H DAY	и ссля и оду и оду	И СLR И 047К И DAV И 25. К И 25. К И 24. С. В. И 24. С. В. И 24. С. В. И 24. С. В. В. В. В. В. В. В. В. В. В. В. В. В.	fron can assurances fron can assurances
OREGON DI TRANS RED RD at EAGL	P8 NTANF SCONTL SCONTL STOP SIGN	H STOP SIGN	11 10 10 10 10 10 10 10 10 10 10 10 10 1	s to the Chegan Depa
DUAN	INT.77 HAR (MEDIA LEGS I (#LANE R CROSS	a CROSS	C CROSS	sh reports submitter hat all qualifying cra
	T BE C DIRES LLCCTI LLCCTI LLTE LLTE LLTE LLTE CI	INTEL SDP HY CH 04	JATER MOV HY CH 03 MOV HY CH 01	driver and police cra 1.can not guarantee 1 148
	CITY STREET FIRST STREET SECOND STREE LRS DUBARKO RD EAGLE CRK SA 017200100500	DURARKO RD EAGLE CRA-SA 017200100500	DURARIO RD EAGLE CRK-EA 017200100500 DURARIO RD EAGLE CRK-EA 017200100500	npaied from andrudual us end Reporting Uni atennic Crash Dala F atennic Crash Dala F
COUNTY	CLASS DIST FROM FROM (2016 16 16 22 76 -122 15 28 76 -122 15	2016 16 22.76 122 15 40.39	2016 16 22,76 -122 15 48 39 2016 16 2016 16 23,76 -122 15 23,76 -122 15 23,76 -122 15	ed in this report is co mer, the Crash Analy for inclusion in the St
B ANDY, CLACKANAS	R R R R R R R R R R R R R R R R R R R	0E/40 N N N N UT 421 65 28	И И 05/11. 11. 45 23 8 И И И 05/06/ 48 45 23	he information contait why of the indendual of crashes being eligible
CDS380 06/06/201 017Y 0F S	5588 5588 5588 5588 549 5589 111111111111111111111111111111111	03967 H CITY N N	02427 N CITY N N 02031 N N N	Disclamar 7 the responsed damage only

Paje: 1	Ó	\bigcirc	sport forms is
	CAUSE 03,222 03 00 00		ocense sukmera el crash n Docense sukmera el crash el cra
	ACT EVENT 110 015 000 000		ustomer However J
	FED ERROR LOC ERROR 1 11/10 023		gines quak cran daa l
	A 5 LICHS G E LICHS B 20 M E 79 M 0R-Y		miled to proveing the tv are changes to Dikir's ve
AT DIVISION G UNIT : to 12/31/2016	PH TYPE 5.VR 01 BIKE INJ 01 DRVR NON		d Reporting Und is com- acculate Note Legisla
KTION DEVELOPMI AND REPORTING ING 01/01/201	RRICH FRICH 1-1-11-11-11-11-11-11-11-11-11-11-11-11		ne Crash Anelysa a
(Ol - TRANSPORTJ CRASH ANAVLVST GTASH ANAVLVST STEM CRASH LIGT COMMAN COMMIN	SFFCL US PARA GT MARZE VM TTPE PRVTE PSNGR C		ad in ORS 811 720 71 1 ad óin ais partainnag
ASPORTATI SECTION - AN NON-SY of Sundy ,	CRASH BIKE TURN INJ		be made that
ortheart of Tru- ortation DATA Ure Wrich 51, C4ty	OFFRD MTHR RUGHT STURF N CLR N CLR N DAY		o Terreporter Inor can assurances
ORECCH. DE TRANS	PE TRAF- REA- S) CONTL N STOP SLOK		i la iho Oregon Depa shes are represente
	117-71 110		qualifying cra
	RD CHAR DIRECT LLOCTN INTER CN CN		and police creat rep not puerantee that a
	CITY STREET SECOND STREET LES JACOBY RD JACOBY RD		ubed from indrivatual dime a and Reporting Unit can fewede Crash Data Felo
,ã	CLASS DIST FRICH 10010 2 017419 27 7 9 07419		I this report is complete the Crash Analysis The Crash Analysis Inclusion at the Stat
DY, CLACKWAS COU	D R S H DATE O C O DAY O C O DAY O C C DAY C C DAY O C		a branching a program of the secon contained a navel bacon contained a solution of the secon of the secon second seco
CDS380 06/07/2018 CETY OF SAN	SER# 5 111/157 E 8 111/157 E 8 012210 M 8 02210 M 8 M		Declamer The the responsibilit damage only of

Ô

06/07/2018

Ċ

Intersection		-						-		_	-	
Int Delay slueh	5.5								1000	-		
incodidy, areas	0.0			-								
Movement	EBL	EBT	EBR	WBL.	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR
Lane Configurations		्र	1		र्स	1		4			र्भ	1
Traffic Vol, veh/h	16	7	42	52	42	31	55	251	18	1	156	5
Future Vol, veh/h	16	7	42	52	42	31	55	251	18	1	156	5
Conflicting Peds, #/hr	2	0	0	0	0	2	0	0	1	1	0	0
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free
RT Channelized	-	-	None	-	-	None	•		None		-	None
Storage Length	-		100	-	-	120		•	-	•		350
Veh in Median Storage	e,# -	0	-	1 12-	0	•	-	0	-	+	0	
Grade, %	-	0	-		0		-	0			0	
Peak Hour Factor	79	79	79	79	79	79	79	79	79	79	79	79
Heavy Vehicles, %	14	14	14	5	5	5	5	5	5	10	10	10
MVmt Flow	20	9	53	66	53	39	70	318	23	1	197	6
Major/Minor	Minor2		-2°k	Minor1	(S)		Major1			Major2		2010
Conflicting Flow All	717	681	197	704	676	333	203	0	0	342	0	0
Stage 1	199	199		471	471	-	-	-		-		
Stage 2	518	482	-	233	205							
Critical Hdwy	7.24	6.64	6.34	7.15	6.55	6.25	4.15	-		42		
Critical Hdwy Stg 1	6.24	5.64		6.15	5.55				1.1			
Critical Hdwy Stg 2	6.24	5.64	-	6.15	5.55	-		-		-	-	
Follow-up Hdwy	3.626	4.126	3.426	3.545	4.045	3.345	2.245		-	2.29	-	
Pot Cap-1 Maneuver	330	358	815	348	371	702	1351		-	1174	-	
Stage 1	776	714	-	568	554	•			-			
Stage 2	519	534	- 1	763	726		-		-			-
Platoon blocked, %								-				
Mov Cap-1 Maneuver	261	334	815	303	347	700	1351		-	1173	•	
Mov Cap-2 Maneuver	261	334		303	347	-		-		-		-
Stage 1	726	713		531	518	-	1		-	-	-	-
Stage 2	411	499	-	704	725	. ×	-	-	-		-	
							0			- 1. 		
Approach	FR			WR		5 101	NA			92		In case of the local division of the local d
HCM Control Delay	13.1			10.6			1 2	-	1000	00	-	-
HCM LOS	B			13.0	1.000		1.3			Ų		
	0			U						8		
Minor Long Hilling M.	-4	Arm	NIDE	Aller							-	
Minitor Lane/Major MVn	T	NBL	NBL	NBR	EBLINI	EBLN2V	VBLn1	VBLn2	SBL	SBT	SBR	and the
Capacity (veh/h)		1351	-	•	280	815	321	700	1173	-	-	
HUM Lane V/C Ratio		0.052	-	-	0.104	0.065	0.371	0.056	0.001	•	-	
HUM Control Delay (s)		7.8	0	•	19.3	9.7	22.7	10.4	8.1	0	-	
HUM Lane LOS		A	A	-	C	A	С	В	A	Α	-	
HUM 95th %tile Q(veh)	0.2	1	-	0.3	0.2	1.7	0.2	0	-	•	

Jacoby Road Subdivision TIS $\,$ 05/15/2018 Existing Conditions AM Peak Hour RM $\,$

Ó

06/07/2018

0

Jacoby Road Subdivision TIS 05/15/2018 Existing Conditions AM Peak Hour RM

Ô

06/07/2018

Ċ

Intersection			lang ?	-		
Int Delay, s/veh	1.9					
Movement	EBL	EBR	NBL	NBT	SBT	SBR
Lane Configurations	M			đ	1	ODIT
Traffic Vol, veh/h	6	0	0	15	5	3
Future Vol, veh/h	6	0	0	15	5	3
Conflicting Peds, #/hr	0	0	2	0	0	2
Sign Control	Stop	Stop	Free	Free	Free	Free
RT Channelized	-	None	-	None	-	None
Storage Length	0	-	-	-	-	
Veh in Median Storage	e,# 0	-	-	0	0	
Grade, %	0	-	-	0	0	
Peak Hour Factor	56	56	56	56	56	56
Heavy Vehicles, %	33	33	13	13	0	0
Mvmt Flow	11	0	0	27	9	5
Major/Minor	Minor?	- 1	Ininet		laine?	1000
Conflicting Flow All		14	16	N	ndjui Z	0
Stars 1	41	14	10	U	-	U
Stace 2	14	•	-			-
Critical Liduar	673	6 52	4 00	-	•	-
Critical House Start	5.72	0.33	4.23	-	•	-
Critical Howy Sty 1	0.73		-	-		-
Chucai nuwy Sig Z	0./3	2 507	- 2.247	-	•	-
Pot Con 1 Monormore	3./9/	3.09/	4522	10	-	
Store 1	03/	903	1033	•	-	
Stage 1	934	1.5				-
Stage Z	921	-	-	•	-	
matoon blocked, %	000	004	4500	-		
Nov Cap-1 Maneuver	093	981	1530	-		•
Nov Cap-2 Maneuver	893		•	-		
Stage 1	932		•	-	•	•
Stage 2	919	-	-			
Approach	EB		NB		SB	
HCM Control Delay, s	9.1	Service Service	0		0	
HCM LOS	А					
	1					
Minor Lang Mains Ha		MDL	NOT	KDL-4	COT	000
Connective (underline)	IL.	ACOC	ND I		301	SBR
Capacity (ven/n)	1011-1-1	1530	-	893		
HCM Cashel Delaw (-)		-	-	0.012		
HCM Lang LOS		0	-	9.1	-	
HOM LARE LUS	1	A	-	A	-	
LCW ADIU MILle CI(Aep	1	U	-	0	-	

Jacoby Road Subdivision TIS 05/15/2018 Existing Conditions AM Peak Hour RM

0

06/07/2018

 \bigcirc

ntersection					6	Ш.,		an Xin			-			
Int Delay, s/veh	4.8													
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR		
Lane Configurations		đ	1		đ	1		4			đ	1		
Traffic Vol. veh/h	6	51	66	26	32	33	64	230	60	11	271	12	15-16-16 A.C. 181	
Future Vol, veh/h	6	51	66	26	32	33	64	230	60	11	271	12		
Conflicting Peds, #/hr	0	0	0	0	0	0	1	0	0	0	0	1	Contention of the last	START
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	1000000000	
RT Channelized		-	None	-	-	None			None	-	-	None		
Storage Length	-	-	100	•		120	10			-	-	350		
Veh in Median Storage	,# -	0			0		-	0	-		0	-		
Grade, %	-	0	-	-	0	-	10	0	-		0	-		
Peak Hour Factor	91	91	91	91	91	91	91	91	91	91	91	91		
Heavy Vehicles, %	1	1	1	1	1	1	1	1	1	3	- 3	3		
Mvmt Flow	7	56	73	29	35	36	70	253	66	12	298	13	1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	
Major/Minor N	Minor2	22.13		Minor1	1		Major1		1	Major2	-	- 10499		miles a
Conflicting Flow All	785	782	299	819	762	286	312	0	0	319	0	0		
Stage 1	323	323	-	426	426	-			-	•	-	•		
Stage 2	462	459		393	336	-	-			-	-	-		
Critical Hdwy	7.11	6.51	6.21	7.11	6.51	6.21	4.11	-		4.13	-			
Critical Hdwy Stg 1	6.11	5.51	•	6.11	5.51	-	-	8	-	-	-			
Critical Hdwy Stg 2	6.11	5.51	•	6.11	5.51		-	-	-	-	-			
Follow-up Hdwy	3.509	4.009	3.309	3.509	4.009	3.309	2.209		-	2.227	-	-		
Pot Cap-1 Maneuver	311	327	743	295	336	755	1254			1235	-	-		
Stage 1	691	652		608	588	•	- •			- 2		-		
Stage 2	582	568	-	634	644	-					-	-		
Platoon blocked, %	227			- 7-				-	-					_
Mov Cap-1 Maneuver	254	301	742	215	309	755	1253	-		1235	-		- 4 - 0.02 - 3	
Mov Cap-2 Maneuver	254	301	-	215	309	-	-	-	-	-		•		
Stage 1	643	644	•	567	548	-	-	-	-	-	-	S - 3.		
Stage 2	483	529		516	636	•	•		-	-	-		i sira in co	
a T	-			14.00			NE			00				
Approach	EB	-		WB			NB			55	122	-		
HCM Control Delay, s	15.1		-	18.6	1 and		1.5			0.3				di mini di s
HCM LOS	C		-11-2-	C					1					1.55
Minor Lano/Major Mum	at	MQI	MRT	NED	CDI -4	CD1 n 24	MBI n41	MBLoD	CRI	CRT	COD	1 10-5		AV/SI III
Canadia (ush /h)	.u.	40F9		NOR	DUIT	740	OED	755	1000	301	JOOK			
Capacity (Ven/n)	Series.	1253	20,0	-	290	742	200	100	1230	-	-			
HOM Caster Delaw (a)		0.000		-	0.21Z	10.090	0.24/ 22 E	0.040	7.0			-		
HCM Long LOC		0	U		20.0	10.4	23.0	10	1.9	0		и —		
HOW Lane LUS	1	A	A	-	0.0	0.2	0.0	0.0	A	A				
LOW ADD WING MAN	1	0.2			0.8	0.3	0.9	0,2	U	-	-			

Jacoby Road Subdivision TIS $\,$ 05/15/2018 Existing Conditions PM Peak Hour RM $\,$

Ó

06/07/2018

0

and the second se				1000		
Intersection						
Int Delay, s/veh	1.7					
Movement	EBT	EBR	WBI	WRT	NBI	NBR
Lane Configurations	Ť.	all the later of the	a rords	1	M	1.9941.4
Traffic Vol. veh/h	111	64	12	65	20	0
Future Vol. veh/h	111	64	12	65	20	9
Conflicting Dode #/hr	0	-0	12	00	25	3
Sign Control	Erec	Eroc	Erec	Erec	Stor	Cior
DT Channelized	Fiee	None	Fiee	Mana	Stop	Ness
Ni Unannelized		NODE	•	NONE	•	NONE
Storage Length	-	•	-		0	-
ven in Median Storage,	# U	100	-	0	0	•
Grade, %	0	-	-	0	0	•
Peak Hour Factor	79	79	79	79	79	79
Heavy Vehicles, %	1	1	3	3	3	3
Mvmt Flow	141	81	15	82	37	11
Majoriklings	Iniant		Intert		diam-1	_
Major/Mittor N	najori		viajor2		Minor1	-
Conflicting Flow All	0	0	222	0	295	184
Stage 1	-	-		-	182	
Stage 2	-	-	•	-	113	
Critical Hdwy	-	-	4.13	-	6.43	6.23
Critical Howy Stg 1	1.61.00.00	-		-	5.43	
Critical Hdwy Stg 2	-		-	-	5.43	-
Follow-up Hdwy			2.227		3.527	3.327
Pot Cap-1 Maneuver		-	1341	-	694	856
Stage 1					847	
Stage 2			-		909	
Platoon blocked %			-		203	and the second second
Mov Can-1 Manouver			13/1	-	685	854
Mov Cap 2 Maneuver		100	1041		000	004
Stage 4	-	30	•		000	-
Stage 1		-			837	•
Stage 2	•	•	-	-	908	-
			M-AL			18 0
Approach	EB	11-20-00	WB	11-11-1	NB	
HCM Control Delay s	0		12		10.4	
HCM LOS	v		1,4	and the second second	D	-
HOW LOO		1111-11-11		-	D	
				1.2.2		din e
Minor Lane/Major Mvml	1	VBLn1	EBT	EBR	WBL	WBT
Capacity (veh/h)		719	-		1341	-
HCM Lane V/C Ratio		0.067			0.011	-
HCM Control Delay (s)		10.4			77	0
HCM Lane LOS		R			Δ	Δ
HCH 05th % tile O(ush)	0 1 0	0.0	-		A	A
ICM SOUL WILL (Veh)		0.2	•		0	-

Jacoby Road Subdivision TIS $\,$ 05/15/2018 Existing Conditions PM Peak Hour RM $\,$

Ó

06/07/2018

Intersection						
Int Delay, s/veh	1.5					
Movement	EBL	EBR	NBL	NBT	SBT	SBR
Lane Configurations	W			đ,	ţ,	
Traffic Vol. veh/h	10	0	0	11	14	25
Future Vol. veh/h	10	0	0	11	14	25
Conflicting Peds. #/hr	0	0	2	0	0	2
Sign Control	Stop	Stop	Free	Free	Free	Free
RT Channelized		None		None		None
Storage Length	0	-	1	-	-	-
Veh in Median Storage	.# 0	-	-	0	0	-
Grade. %	0	-	-	0	0	-
Peak Hour Factor	79	79	79	79	79	79
Heavy Vehicles %	0	0	0	0	0	0
Mymt Flow	13	Ő	0	14	18	32
	19			••	10	
Major/Minor	Minor2	1	Major1	A	lajpr2	
Conflicting Flow All	50	36	52	0	1.0	0
Stage 1	36	-	-			-
Stage 2	14			-	-	
Critical Hdwy	6.4	6.2	4.1		-	
Critical Hdwy Stg 1	5.4		-			-
Critical Hdwy Sto 2	5.4	-	-	-	-	
Follow-up Hdwy	3.5	3.3	2.2	-	-	-
Pot Cap-1 Maneuver	964	1042	1567	-		
Stage 1	992	-	-		-	-
Stage 2	1014		-			
Platoon blocked %	1919					
May Can-1 Maneuver	080	1040	1564			
Mov Cap-1 Maneuver	020	1040	1004			
Stage 1	000	-	-	-	-1-2	
Stage 7	1012				-	
Stage 2	1012	-	-	-	-	-
Luca une colta		-		90 - A.L.		
Approach	EB		NB		SB	
HCM Control Delay, s	8.8	-	0	(All all a	0	
HCM LOS	A					
	1012					104
Manage Barrow Barrow FT		MOL	A IPS IN	7DI - 4	007	000
Minor Lane/Major Myr	u	NBL	NBI	ZULN1	SBI	SBR
Capacity (veh/h)		1564	-	960	-	•
HCM Lane V/C Ratio				0.013	-	•
HCM Control Delay (s)		0	-	8.8	-	
HCM Lane LOS		Α		A	-	
HCM 95th %tile Q(veh)	0		0	-	-

Jacoby Road Subdivision TIS 05/15/2018 Existing Conditions PM Peak Hour RM

Ó

06/07/2018

Ò

Intersection		100			White	0.008							
Int Delay, s/veh	6.5	-	-01-C3										
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR	
Lane Configurations		र्भ	1		÷f	1		4			et.	1	
Traffic Vol, veh/h	17	7	47	56	45	33	66	289	22	1	173	5	
Future Vol, veh/h	17	7	47	56	45	33	66	289	22	1	173	5	in a second
Conflicting Peds, #/hr	2	0	0	0	0	2	0	0	2	2	0	0	
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	
RT Channelized	-	-	None	-	-	None	-	-	None	-	-	None	
Storage Length			100		-	120			-			350	
Veh in Median Storage	3,# -	0	-	-	0	-	-	0	-	-	0	-	
Grade, %		0			0	-	•	0		-	0	-	
Peak Hour Factor	79	79	79	79	79	79	79	79	79	79	79	79	
Heavy Vehicles, %	14	14	14	5	5	5	5	5	5	10	10	10	
Mvmt Flow	22	9	59	71	57	42	84	366	28	1	219	6	
Major/Minor	Minor			Amont	-		Voient			Inia di	<u>.</u>		_
Coefficient Flow All	WILLIOTZ	705	040		799	00/	major1	-	-	viajor2			
Conflicting Flow All	821	/85	219	808	777	384	225	0	0	396	0	0	
Stage 1	221	221		550	550	-	-	-	-	-	-	-	
Stage 2	500	564	-	258	227	-	1.45	-			-		
Critical Howy	7.24	0.04	6.34	7.15	6.55	6.25	4.15	-	•	4.2	-	-	
Critical Howy Stg 1	0.24	5.04	-	6.15	5.55		-	-		-		•	_
Critical Howy Stg 2	0.24	5.04	-	0.15	5.55	-	-	-	•	-	-		
Pollow-up Howy	3.020	4.120	3.420	3.345	4.045	3.345	2.245	•	•	2.29	•	•	_
Stage 1	200	311	792	290	520	100	1320	•	•	1120	•		
Stage 1	100	400	•	740	714		-	-	-	-	•		
Bistop blocked %	400	490	-	740	- 11	-	-	-		-	-	•	
May Cap 1 Manaurar	210	700	702	250	000	EE A	1200	-	-	4440	-	-	_
Mov Cap-1 Maneuver	210	203	192	200	230	004	1320	-	1	1110	-	-	
Stare 1	60/	609	1200-	471	460		-	-	-	•	-		
Stage 2	352	000	-	675	710								
oluge 2	000			010	710			-					
Approach	EB			WB	Carrier -		NB	- The		SB		- Siles	ĺ
HCM Control Delay, s	14.4	1.42	- 254	25.2		the or	1.4	11-14/100		Ö	and the second second	111	Ĩ
HCM LOS	В			D									
										20.18			
Minor Lane/Major Mvn	nt	NBL	NBT	NBR	EBLn1	EBLn2	VBLn1V	VBLn2	SBL	SBT	SBR		
Capacity (veh/h)		1326	-	-	227	792	269	654	1118	-	-		
HCM Lane V/C Ratio		0.063	-	•	0.134	0.075	0.475	0.064	0.001	-	-		
HCM Control Delay (s)		7.9	0	-	23.3	9.9	29.9	10.9	8.2	0	-		
HCM Lane LOS		A	A	•	C	A	D	В	A	Α	-		
HCM 95th %tile Q(veh)	0.2		-	0.5	0.2	2.4	0.2	0	- 1.+	-		

Jacoby Road Subdivision TIS 05/15/2018 2020 Background Conditions AM Peak Hour RM

06/07/2018

Int Delay, s/veh 2.7 Movement EBT EBR WBL WBT NBL NBR Lane Configurations - 0	Intersection	-	- Contra		line of			
Movement EBT EBR WBL WBT NBL NBR Lane Configurations - 0 - 0 0 - 0 0 - 0 0 - 0 0 - 0 0 - 0 0 - 0 0 - - 0 0 - - 0 - - - 15 - - 5 - - - -	Int Delay, s/veh	2.7					- A.	
Lane Configurations Image: Configurations <	Movement	FBT	EBR	WBI	WBT	NBI	NBR	
Conting analoties P No No No Traffic Vol, veh/h 25 18 6 101 38 11 Conflicting Peds, #/hr 0 0 0 0 0 0 0 0 Sign Control Free Free Free Free Free Stop Stop RT Channelized None None None None None None Storage Length - - 0 0 - 0 0 Grade, % 0 - - 0 0 - 0 0 - Peak Hour Factor 75 75 75 75 75 75 11 11 Major/Minor Major1 Major2 Minor1 - - 45 - Stage 1 - - 45 - Stage 1 - - 5.57 - - 5.57 - - 5.57 - -	Lane Configurations	1	THE REF Y	1100	\$	M	TIMIN	
Haine Vol, veh/h 25 18 6 101 38 11 Conflicting Peds, #hr 0	Traffic Vol. veh/h	25	18	6	101	38	11	
Number 2.5 16 0 101 36 111 Conflicting Peds, #hr 0	Future Vol. veh/h	25	10	6	101	29	11	
Connecting reds, mm Co Stop Co Go Go <t< td=""><td>Conflicting Dade #/hr</td><td>20</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td></t<>	Conflicting Dade #/hr	20	0	0	0	0	0	
Sign Control Pree Pree Pree Pree Stop Stop RT Channelized - - - 0 - - 0 - Grade, % 0 - - 0 0 - - 0 0 - Grade, % 0 - - 0 0 - - 0 0 - Peak Hour Factor 75 75 75 75 75 75 75 75 75 17 17 17 Mvmt Flow 33 24 8 135 51 15 Major/Minor Major1 Major2 Minor1 - 45 - Stage 1 - - 45 - Stage 2 - - 151 - - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557 - 557	Sign Control	Eroo	Eroo	Eroo	Eroo	Stop	Stop	
RT Chainelized - None - None <th -="" non<="" td=""><td>Sign Control</td><td>riee</td><td>None</td><td>riee</td><td>Mono</td><td>Stop</td><td>Mono</td></th>	<td>Sign Control</td> <td>riee</td> <td>None</td> <td>riee</td> <td>Mono</td> <td>Stop</td> <td>Mono</td>	Sign Control	riee	None	riee	Mono	Stop	Mono
Storage Length - - - 0 - Veh in Median Storage, # 0 - - 0 0 - Grade, % 0 - - 0 0 - Peak Hour Factor 75 75 75 75 75 75 Heavy Vehicles, % 13 13 7 7 17 17 Mvmt Flow 33 24 8 135 51 15 Major/Minor Major1 Major2 Minor1 17 17 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - 45 - 557 6.37 Critical Hdwy Stage 2 - - 5.57 - 5.57 Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 755 984 Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 841	Ctorned Longth	-	NUTE	•	NONE	-	NOUG	
Ven in Median Storage, # 0 - - 0 0 - Grade, % 0 - - 0 0 - Grade, % 0 - - 0 0 - Peak Hour Factor 75 75 75 75 75 75 Heavy Vehicles, % 13 13 7 7 17 17 Mvmt Flow 33 24 8 135 51 15 Major/Minor Major1 Major2 Minor1 17 17 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - 4.17 6.57 6.37 Critical Hdwy Stg 1 - - 5.57 - Follow-up Hdwy - 2.263 3.653 3.453 Pot Cap-1 Maneuver - 1516 755 984 Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 934 - Stage 2	Storage Length		-	-	-	0	-	
Grade, % 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - - 0 0 - 7 15 <th1< td=""><td>ven in Median Storage,</td><td># U</td><td>-</td><td>-</td><td>0</td><td>0</td><td>-</td></th1<>	ven in Median Storage,	# U	-	-	0	0	-	
Peak Hour Factor / 5 / 5 / 5 <td>Grade, %</td> <td>0</td> <td>-</td> <td>-</td> <td>0</td> <td>0</td> <td>-</td>	Grade, %	0	-	-	0	0	-	
Heavy Vehicles, % 13 13 7 7 17 17 Mymt Flow 33 24 8 135 51 15 Major/Minor Major1 Major2 Minor1 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - - 45 - Stage 2 - - 151 - Critical Hdwy - - 4.17 - 6.57 6.37 - Critical Hdwy Stg 1 - - - 5.57 - - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 - - - 5.57 - - Follow-up Hdwy - 2.263 - 3.653 3.453 - - - 9.84 -	Peak Hour Factor	75	75	75	75	75	75	
Mvmt Flow 33 24 8 135 51 15 Major/Minor Major1 Major2 Minor1 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - - 45 - Stage 2 - - 151 - Critical Hdwy - 4.17 - 6.57 6.37 - Critical Hdwy Stg 1 - - 5.57 - - 5.57 - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 - 760 984 Stage 1 - - 940 - Stage 2 - - 841 - Platoon blocked, % - - - 55 - Stage 1 - - 755 984 Mov Cap-2 Maneuver - - 1516 - 755 984 - Stage 2	Heavy Vehicles, %	13	13	7	7	17	17	
Major/Minor Major1 Major2 Minor1 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - - 45 - Stage 2 - - - 151 - Critical Hdwy - 4.17 6.57 6.37 Critical Hdwy Stg 1 - - 5.57 - Critical Hdwy Stg 2 - - 5.57 - Follow-up Hdwy - 2.263 3.653 3.453 Pot Cap-1 Maneuver - 1516 760 984 Stage 1 - - - 940 - Stage 2 - - 841 - Platoon blocked, % - - - 755 984 Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 841 - Stage 2 - - -	Mvmt Flow	33	24	8	135	51	15	
Major/Minor Major1 Major2 Minor1 Conflicting Flow All 0 0 57 0 196 45 Stage 1 - - - 45 - Stage 2 - - - 151 - Critical Hdwy - 4.17 - 6.57 6.37 Critical Hdwy Stg 1 - - - 5.57 - Critical Hdwy Stg 2 - - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 - 760 984 Stage 1 - - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 755 984 Mov Cap-1 Maneuver - 1516 - 755 - Stage 1 - - - <								
Integrit	Major/Minor	Iniort	Ā	Coint.		linor1	-	
Connicung Flow All U D 57 U 196 45 Stage 1 - - - 45 - Stage 2 - - 45 - Critical Hdwy - - 4.17 - 6.57 6.37 Critical Hdwy Stg 1 - - - 5.57 - Critical Hdwy Stg 2 - - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 - 760 984 Stage 1 - - - 940 - Stage 2 - - 841 - Platoon blocked, % - - - 755 984 Mov Cap-2 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 841 - Stage 2 - - - 841 - Mo			0			400	AF	
Stage 1 - - - 45 - Stage 2 - - 151 - - 151 - Critical Hdwy - 4.17 6.57 6.37 - - 5.57 - Critical Hdwy Stg 1 - - 5.57 - - 5.57 - Critical Hdwy Stg 2 - - - 5.57 - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 - - 843 - - 984 Stage 1 - - - 5.57 - - - 984 Stage 2 - - - 841 - - - - 841 - - - 5.57 - Stage 1 - - - 7.55 - - Stage 1 - - - 7.55 - Stage 2 - - - 841 - - - - 841 - - - - <td< td=""><td>Contricting Flow All</td><td>0</td><td>U</td><td>5/</td><td>0</td><td>196</td><td>45</td></td<>	Contricting Flow All	0	U	5/	0	196	45	
Stage 2 - - - 151 - Critical Hdwy - - 4.17 - 6.57 6.37 Critical Hdwy Stg 1 - - - 5.57 - Critical Hdwy Stg 2 - - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 - 760 984 Stage 1 - - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - 1516 - 755 984 Mov Cap-2 Maneuver - - - 934 - Stage 1 - - - 934 - Stage 2 - - - 841 - Mov Cap-2 Maneuver - - - 841 - Mot Cap-2 Maneuver - - - </td <td>Stage 1</td> <td>•</td> <td>•</td> <td></td> <td>•</td> <td>45</td> <td></td>	Stage 1	•	•		•	45		
Critical Hdwy - - 4.17 - 6.57 6.37 Critical Hdwy Stg 1 - - 5.57 - Critical Hdwy Stg 2 - - 5.57 - Follow-up Hdwy - 2.263 3.653 3.453 Pot Cap-1 Maneuver - 1516 760 984 Stage 1 - - 940 - Stage 2 - - 841 - Platoon blocked, % - - - 841 Platoon blocked, % - - - 841 Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - - 934 - Stage 1 - - - 934 - - Stage 2 - - - 841 - - Mov Cap-2 Maneuver - - 841 - - Mov Cap-2 Maneuver - - - 841 - CM Control Delay, s 0 <td>Stage 2</td> <td>•</td> <td>•</td> <td>•</td> <td>•</td> <td>151</td> <td>•</td>	Stage 2	•	•	•	•	151	•	
Critical Hdwy Stg 1 - - 5.57 Critical Hdwy Stg 2 - - 5.57 Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 760 984 Stage 1 - - 940 - Stage 2 - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 755 - Stage 1 - - 934 - Stage 2 - - 841 - Mov Cap-2 Maneuver - - 934 - Stage 2 - - - 841 - Mov Cap-2 Maneuver - - 841 - Mov Cap-2 Maneuver - - 841 - Mov Cap-2 Maneuver - - 841 - Mov Cap-1 Maneuver - - <	Critical Hdwy			4.17	-	6.57	6.37	
Critical Hdwy Stg 2 - - - 5.57 - Follow-up Hdwy - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - 1516 - 760 984 Stage 1 - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - 1516 - 755 984 Mov Cap-2 Maneuver - - - 755 984 Mov Cap-2 Maneuver - - - 934 - Stage 2 - - - 841 - Approach EB WB NB - HCM Control Delay, s 0 0.4 9.9 - HCM LOS A - - - 1516 - HCM LOS A - 0.005 - - - 1516 - HCM Lane V/C Ratio 0.082 -	Critical Hdwy Stg 1			•		5.57		
Follow-up Hdwy - - 2.263 - 3.653 3.453 Pot Cap-1 Maneuver - - 1516 - 760 984 Stage 1 - - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - - 1516 755 984 Mov Cap-2 Maneuver - - - 755 - Stage 1 - - - 934 - Stage 2 - - - 841 - Approach EB WB NB - - 841 - HCM Control Delay, s 0 0.4 9.9 - - - 841 - HCM LOS A - - - 841 - - Minor Lane/Major Mvmt NBLn1 EBT EBR WB NB - - - - - - <td>Critical Hdwy Stg 2</td> <td></td> <td>-</td> <td>-</td> <td>-</td> <td>5.57</td> <td>-</td>	Critical Hdwy Stg 2		-	-	-	5.57	-	
Pot Cap-1 Maneuver - - 1516 - 760 984 Stage 1 - - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 755 - Stage 1 - - - 934 - Stage 2 - - - 841 - Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A - - - Minor Lane/Majpr Mvmt NBLn1 EBT EBR WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Lane LOS A - A A </td <td>Follow-up Hdwy</td> <td>-</td> <td>-</td> <td>2.263</td> <td>-</td> <td>3.653</td> <td>3.453</td>	Follow-up Hdwy	-	-	2.263	-	3.653	3.453	
Stage 1 - - - 940 - Stage 2 - - - 841 - Platoon blocked, % - - - 841 - Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 755 - Stage 1 - - - 934 - Stage 2 - - - 841 - Approach EB WB NB - - HCM Control Delay, s 0 0.4 9.9 - HCM LOS A - - - 1516 - Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	Pot Cap-1 Maneuver		-	1516	-	760	984	
Stage 2 - - - 841 - Platoon blocked, % - </td <td>Stage 1</td> <td></td> <td>-</td> <td>-</td> <td>-</td> <td>940</td> <td>-</td>	Stage 1		-	-	-	940	-	
Platoon blocked, % - - Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 755 - Stage 1 - - 934 - Stage 2 - - 934 - Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM LOS A - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Control Delay (s) A - A A	Stage 2		-		-	841	-	
Mov Cap-1 Maneuver - 1516 755 984 Mov Cap-2 Maneuver - - 755 - Stage 1 - - 934 - Stage 2 - - 934 - Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A A	Platoon blocked, %							
Mov Cap-2 Maneuver - - 755 Stage 1 - - 934 - Stage 2 - - - 934 - Stage 2 - - - 841 - Approach EB WB NB - - 841 - Approach EB WB NB - - 841 - Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	Mov Cap-1 Maneuver			1516		755	984	
Stage 1 - - - 934 - Stage 1 - - - 934 - Stage 2 - - - 841 - Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A	Mov Cap-2 Maneuver	_				755	001	
Stage 2 - </td <td>Store 1</td> <td>10.5</td> <td>-</td> <td>-</td> <td></td> <td>034</td> <td>-</td>	Store 1	10.5	-	-		034	-	
Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A HCM Dith % iso (veh) 0.2	Stage 1	-	-	Sec.	•	044		
Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	Stage 2	-				041	-	
Approach EB WB NB HCM Control Delay, s 0 0.4 9.9 HCM LOS A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	A CONTRACTOR OF A CONTRACT OF	14.5 2		1000	10.00			
HCM Control Delay, s 0 0.4 9.9 HCM LOS A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM LOS A - A A	Approach	EB		WB	- never	NB	a me	
HCM LOS A Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM LOS A - A A	HCM Control Delay s	0		0.4		9.9	-	
Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM LOS A - A A	HCM LOS			w. T		Δ		
Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	LION LOO	-	2		133	~		
Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 797 - - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A								
Capacity (veh/h) 797 - 1516 - HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A HCM 06th % bio Q(reh) 0.2 0 0	Minor Lane/Major Mvm	t	NBLn1	EBT	EBR	WBL	WBT	
HCM Lane V/C Ratio 0.082 - 0.005 - HCM Control Delay (s) 9.9 - 7.4 0 HCM Lane LOS A - A A	Capacity (veh/h)		797		-	1516		
HCM Control Delay (s) 9.9 7.4 0 HCM Lane LOS A A A HCM (Stip (Visio Q(uch)) 0.2	HCM Lane V/C Ratio		0.082			0.005		
HCM Lane LOS A A A	HCM Control Delay (s)		9.9			7.4	0	
HCM DEth % file O(uch) 0.2 0	HCM Lane LOS		A			A	A	
	HCM 95th %tile O(veh)	110	0.3	-	-	0	-	

Jacoby Road Subdivision TIS $\,05/15/2018\,2020$ Background Conditions AM Peak Hour RM $\,$

HCM 2010 TWSC 3: SE Jacoby Road & Street B

0

06/07/2018

0

Intersection	-			Plante and		
Int Delay, s/veh	0				1000	
Movement	WBI	WRP	NAT	MBP	SBL	SRT
Lana Configurations	WOL	WDI/		NOK	ODU	100
Traffic Vol. voh/h	T	0	12	٥	0	•
Future Vol. veh/h	0	U	22	0	0	0
Confliction Dede #/hr	0	U	22	0	U	8
Contacting Peas, #/nr	Chan	U	0	0	U	0
Sign Control	Stop	Stop	Free	Free	Free	Free
Ri Unannelized	-	None	- 0-	None		None
Storage Length	0	-	-			
Veh in Median Storage	e,# 0	-	0	-	-	0
Grade, %	0	-	0	. *		0
Peak Hour Factor	92	92	92	92	92	92
Heavy Vehicles, %	2	2	2	2	2	2
Mvmt Flow	0	0	24	0	0	9
Major/Minor	Minort	ī	Inion		(alas)	-
Confliction Flow Att	WITTOFT		najoni	-	viajor2	-
Conflicting Flow All	33	24	0	0	24	0
Stage 1	24	-		-	-	
Stage 2	9	-	•	-		•
Critical Hdwy	6.42	6.22	-	-	4.12	•
Critical Hdwy Stg 1	5.42	+	•	-	-	•
Critical Hdwy Stg 2	5.42	-	-	-	-	-
Follow-up Hdwy	3.518	3.318	-		2.218	4
Pot Cap-1 Maneuver	980	1052	-	-	1591	-
Stage 1	999		-		2	
Stage 2	1014	12	-			-
Platoon blocked, %						
Mov Cap-1 Maneuver	980	1052			1591	
Moy Cap-2 Maneuver	980	-				
Stage 1	000					
Stage 2	1014	-	The second se			-
Oldye z	1014	•				
		-				1103
Approach	WB	Better,	NB	-	SB	
HCM Control Delay, s	0		Û	1.251	0	
HCM LOS	A					
W. Starting		100-0-00	and a set		83 - A	
No. 1		A LINE OF				
Minor Lane/Major Mvn	nt	NBT	NBRV	VBLn1	SBL	SBT
Capacity (veh/h)		•	-	-	1591	-
HCM Lane V/C Ratio		•	-	-	•	-
HCM Control Delay (s)		-	0	0	
HCM Lane LOS		-	-	Α	Α	-
HCM 95th %tile Q(veh)			-	0	

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Background Conditions AM Peak Hour RM $\,$

0

06/07/2018

 \bigcirc

Intersection	10.00	The state				200	The List	64-74	and a	100	-	Seco	and the second
Int Delay, s/veh	1.8				_						_		
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR	
Lane Configurations		4			44			4.			4.		
Traffic Vol, veh/h	6	0	0	0	0	0	0	16	0	0	5	3	
Future Vol, veh/h	6	0	0	0	0	0	0	16	0	0	5	3	
Conflicting Peds, #/hr	0	0	0	0	0	0	2	0	0	0	0	2	
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	
RT Channelized	-	-	None	-	-	None			None		-	None	
Storage Length	-		-	-		-		(a)	-		-	-	
Veh in Median Storage	,# -	0	-	-	0	-	-	0	-	-	0	-	
Grade, %		0		-	0	-	-	0	-		0	-	
Peak Hour Factor	56	56	56	56	56	56	56	56	56	56	56	56	N
Heavy Vehicles, %	33	33	33	0	0	0	13	13	13	0	0	0	
Mvmt Flow	11	0	0	0	0	0	0	29	0	0	9	5	
											1.1.1.1		
Major/Minor	Minor2			Minor1			Major1		1	hajor2	a materia		
Conflicting Flow All	43	43	14	41	45	29	16	0	0	29	0	0	
Stage 1	14	14		29	29	-			-				
Stage 2	29	29		12	16							-	
Critical Hdwy	7.43	6.83	6.53	7.1	6.5	6.2	4.23			4.1	-	Statistics.	
Critical Hdwy Stg 1	6.43	5.83	-	6.1	5.5	-	-		-	-	-	-	
Critical Hdwy Sto 2	6.43	5.83	-	6.1	5.5	-	-	-	-	-	-	-	Same and the second
Follow-up Hdwy	3,797	4.297	3.597	3.5	4	3.3	2.317		-	2.2	-		
Pot Cap-1 Maneuver	887	792	983	968	851	1052	1533	-	-	1597	-	•	
Stage 1	932	826		993	875					. •			
Stage 2	914	813		1014	886			-	-				
Platoon blocked, %				and an an instant				-	-		-	-	
Mov Cap-1 Maneuver	885	790	981	968	849	1052	1530	12	-	1597	-		and the second second
Mov Cap-2 Maneuver	885	790	-	968	849	-	-	-	-	-	-	-	
Stage 1	930	824		993	875	-	-	-	-	-	-	-	
Stage 2	914	813		1014	884			-	-			-	
		and the second	n anna		ana an	en an				(and the second			and a manager of the
Approach	EB	No. Con	di terre	WB	14.50		NB		1160	SB			
HCM Control Delay, s	9.1			0			0			0			
HCMLOS	A			A									
		-						5.11		Ti-a-			
Minor Lane/Major Myn	nt	NBL	NBT	NBR	EBLn1	WBLn1	SEL	SBT	SBR	and a start			anticol in Fallen I
Capacity (veh/h)	-	1530	-	-	885		1597		-	14.1			
HCM Lane V/C Ratio			-		0.012								
HCM Control Delay (s		Ó			9.1	0	0			1	-		
HCM Lane LOS		A	-	-	A	A	A						
HCM 95th %tile O(veh	ñ	0	-	-	0		0	-	- 11		_	- 111-	
Tour oour tours offact	1	U			0	-	v						

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Background Conditions AM Peak Hour RM $\,$

06/07/2018

0

ntersection	neek.		Northern Party	NIV.	New N	2.4	100-		1		74			
Int Delay, s/veh	5.6	19.5					- 19							
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR		
Lane Configurations		र्स	1		र्भ	1		4			स	1		
Traffic Vol, veh/h	6	54	77	32	34	35	71	260	66	12	314	13		
Future Vol, veh/h	6	54	77	32	34	35	71	260	66	12	314	13		_
Conflicting Peds, #/hr	0	0	0	0	0	0	1	0	0	0	0	1		
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	12.11-	
RT Channelized	-	-	None			None	-	-	None		-	None		
Storage Length		•	100	-		120			-	-	•	350		
Veh in Median Storage	s,# -	0	-	-	0	-	-	0	100	-	0			
Grade, %	•	0		-	0			0	-	-	0	-		
Peak Hour Factor	91	91	91	91	91	91	91	91	91	91	91	91		
Heavy Vehicles, %	1	1	1	1	1	1	1	1	1	3	3	3		
Mvmt Flow	7	59	85	35	37	38	78	286	73	13	345	14		
Major/Minor	Minor?	2		Minor1	i i		Valort			Major?			THE REAL PROPERTY AND	
Conflicting Flow All	888	887	346	920	865	322	360	0	0	350	0	0	and the second sec	
Stare 1	372	372	340	470	470	525	500	U	v	333	U	U		-
Stage 2	516	515		450	386		-				-	-		
Critical Hdwy	7 11	6.51	6.21	7 11	6.51	6.21	4 11			4 13				
Critical Hdwy Sto 1	6.11	5.51	-	6.11	5.51	9.61	-		-	4.10		-		111
Critical Hdwy Sto 2	6.11	5.51	-	6.11	5.51	-	-	-			-	-		
Follow-up Hdwy	3.509	4.009	3,309	3,509	4.009	3.309	2.209		-	2.227				
Pot Cap-1 Maneuver	265	284	699	249	293	720	1204			1194	-	-		
Stage 1	651	621		570	557	-	-			-	-	-		
Stage 2	544	536	-	590	612	-	-			-	-	-		100
Platoon blocked, %												-		
Mov Cap-1 Maneuver	208	257	698	167	265	720	1203	-	-	1194		-		14/2
Mov Cap-2 Maneuver	208	257	-	167	265				-			-		
Stage 1	597	612		523	511			-	-	-		-		
Stage 2	438	492	•	462	603			•		-	-	-		
Annmach	ER			WP			NP			92				
HCM Control Delov. e.	16.9		1000	24.2			1 =	-		0.2		1		
HCM LOS	10.0			24.3			1.0	(ii)——(i)		0.3				-
TIOM EOS				U		W - 55 -				- 38				
Minor Lane/Major Mvm	nt	NBL	NBT	NBR	EBLn1	EBLn2	VBLn1V	VBLn2	SBL	SBT	SBR			
Capacity (veh/h)	1	1203	-	-	251	698	206	720	1194	- 11	-			
HCM Lane V/C Ratio		0.065	-	-	0.263	0.121	0.352	0.053	0.011	-	-			-
HCM Control Delay (s)		8.2	0	-	24.4	10.9	31.7	10.3	8	0			TE VILLE T	
HCM Lane LOS	0.20 - 0.00 - 1 	Α	Α	-	С	В	D	8	Α	A	-			
HCM 95th %tile Q(veh)	0.2	-	-	1	0.4	1.5	0.2	0	•	•			

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Background Conditions PM Peak Hour RM $\,$

06/07/2018

0

ntersection	110			1315		
Int Delay, s/veh	1.6					
Movement	Lat	200	14/21	WDT	MRI	NDD
I one Configurations	DI	COR	TYDL		NDL	NDR
Lane Configurations	447	07	40		1 20	0
I ramic Vol, veh/h	11/	67	12	12	30	y
Future Vol, veh/h	11/	6/	12	12	30	9
Conflicting Peds, #/hr	0	0	0	0	1	2
Sign Control F	Free	Free	Free	Free	Stop	Stop
RT Channelized	10.00	None	-	None	-	None
Storage Length	100	•	•	•	0	-
Veh in Median Storage, #	¥ 0	•	•	0	0	-
Grade, %	0	-	•	0	0	-
Peak Hour Factor	79	79	79	79	79	79
Heavy Vehicles, %	1	1	3	3	3	3
Mvmt Flow	148	85	15	91	38	11
					-	
Major Minor	Ind		Inland.		Incat	
Majoriminor Ma		-	vialor2		MINOT'	400
Conflicting Flow All	0	0	233	0	313	193
Stage 1	-		1-	-	191	-
Stage 2	-	-	-	-	122	•
Critical Hdwy	-		4.13		6.43	6.23
Critical Hdwy Stg 1	-	-	-		5.43	-
Critical Hdwy Stg 2			-		5.43	-
Follow-up Hdwy	•	-	2.227	•	3.527	3.327
Pot Cap-1 Maneuver		-	1329		678	846
Stage 1	•			-	839	
Stage 2	-	-	-	-	901	-
Platoon blocked. %	-	-				
Mov Cap-1 Maneuver		-	1329		669	R44
Mov Can-2 Maneuver			IULU		660	011
Stane 1					820	
Stage 1		-			000	
olage z			-		900	
Marine States and						
Approach	EB		WB	10.00	NB	100
HCM Control Delay, s	0		11		10.5	
HCMLOS					B	
		and a local data		i i mana		
Minor Lane/Major Mvmt	1. 1.	NBLn1	EBT	EBR	WBL	WBT
Capacity (veh/h)		703	1	-	1329	-
HCM Lane V/C Ratio		0.07	-	-	0.011	
HCM Control Delay (s)		10.5	-	-	7.7	0
HCM Lane LOS		8	-	1	Δ	Δ
						11

Jacoby Road Subdivision TIS 05/15/2018 2020 Background Conditions PM Peak Hour RM

HCM 2010 TWSC 3: SE Jacoby Road & Street B

0

 \bigcirc

Intersection						
Int Delay, s/veh	0					-
Movement	(A/D)		NOT	NED	CDI	COT
Long Configurations	WDL	NOR	NDI	NOR	SBL	281
Troffic Vol. uch/h	T	0	4	0	0	
Future Vol. ven/n	0	U	21	U	0	41
Puture vol, Ven/n	0	U	21	0	0	41
Connicung Peds, #/hr	0	U	0	0	0	0
Sign Control	Stop	Stop	Free	Free	Free	Free
KI Channelized		None		None	-	None
Storage Length	0		•			1.7
Veh in Median Storage	e,# 0	-	0	-	-	0
Grade, %	0	-	0			0
Peak Hour Factor	92	92	92	92	92	92
Heavy Vehicles, %	2	2	2	2	2	2
Mymt Flow	0	0	23	0	0	45
			~~		0	VF
NA 1 19 19	A. 41 - 2					
Major/Minor	Minor1	N	Aajor1		Major2	152-2
Conflicting Flow All	68	23	0	0	23	0
Stage 1	23	-	-	-		-
Stage 2	45				-	
Critical Hdwy	6.42	6.22	-		4.12	-
Critical Hdwy Sto 1	5.42	-				
Critical Hdwy Sto 2	5.42		-		-	-
Follow-up Hdwy	3.518	3.318			2 218	
Pot Can-1 Maneuver	937	1054	-	-	1502	
Stane 1	1000	1034			1092	•
Store 2	1000	-	-	-	-	•
Stage 2	9//		•		- -	•
Flatoon blocked, %			•			
Mov Cap-1 Maneuver	937	1054	•	-	1592	-
Mov Cap-2 Maneuver	937		-	1.0	•	
Stage 1	1000		-			-
Stage 2	977		-		-	
	1000					
Appmach	WB		MR		SB	-
HCM Control Delaw	0	-	0		00	•
HCMLOS	0		U		0	
ILUS	A		-	-	_	
	· · · ·					
Minor Lane/Major Mym	nt	NBT	NBRV	VBLn1	SBI	SBT
Canacity (yeh/h)					1502	
HCM Lane V/C Patio		-		Acres 10	1082	
HCM Control Delay (a)		-		0	-	
HCML and LOS		-	Sec. 2	U	U	-
HOM DOM DOM NATE OF		-		A	A	
HUM 95th %tile Q(veh)	-	-	-	0	-

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Background Conditions PM Peak Hour RM $\,$

 \bigcirc

06/07/2018

()

Int Delay, Sveh 1.4 Movement EBL EBR WBL WBR NBL NBT NBR SEL SBT SBR Lane Configurations	Intersection													
Novement EBL EBR WBL WBT WBR NBL NBT NBR SBL SBT SBR Lane Configurations 4 4 4 4 4 4 Traffic Vol, veh/h 10 0 0 0 0 11 0 15 26 Future Vol, veh/h 10 0 0 0 0 0 0 0 0 0 0 0 0 0 2 0 0 0 2 0	Int Delay, s/veh	1.4										-		
Markenistical Lot Lot Hoi <	Movement	TDI.	COT	EPD		WPT	MIDD	NPL	MPT	MPD	SPI	CPT	999	
Laine Configurations Laine Con	Long Configurations	COL	EBI	EDR	AADL	101	AADIN	NDL	TICIPI	NDR	JUL	301	ODK	-
Iname Vol, Vehini 10 0 0 0 0 0 11 0 0 13 20 Future Vol, Veh/n 10 0 0 0 0 0 0 0 11 0 0 15 26 Conflicting Peds, #hr 0 0 0 0 0 0 0 0 0 13 20 0 0 15 26 Storage Length -	Traffic Vol. unb/h	10	49	0	٥	4	0	Ô.	449	٥	0	49	20	
Indication Indication <thindication< th=""> Indication Indication<td>Future Vol. veh/h</td><td>10</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>0</td><td>11</td><td>0</td><td>0</td><td>15</td><td>20</td><td></td></thindication<>	Future Vol. veh/h	10	0	0	0	0	0	0	11	0	0	15	20	
Major/Minor Minor Stop	Conflicting Bods #/hr	0	0	0	0	0	0	2	0	0	0	15	20	-
Sign Online Stop Stop <td>Sign Control</td> <td>Stop</td> <td>Stop</td> <td>Stop</td> <td>Ston</td> <td>Stop</td> <td>Stop</td> <td>Free</td> <td>Free</td> <td>Free</td> <td>Free</td> <td>Froo</td> <td>Fran</td> <td></td>	Sign Control	Stop	Stop	Stop	Ston	Stop	Stop	Free	Free	Free	Free	Froo	Fran	
Non-matrice Non- No- Non- Non-	RT Channelized	otop	Jubp	None	otop	otop	None	1100	1166	None	1100	1166	None	
Consigned Langun -	Storage Length			None		-	HUHG			NUNE			NUNE	
Minimum Median Golage, M. C. O. O. C. C. O. C. C. O. C. C. C. O. C. C	Veh in Median Storage	# .	0	-		0			0			0		-
Order, b. O O 79	Grade %	,π -	0			0			0			0		
Non-Houri Non-H	Peak Hour Factor	79	79	79	79	79	79	79	79	79	70	79	79	in the
Merry Follow 13 0 0 0 0 0 0 0 14 0 0 19 33 Major/Minor Minor2 Minor1 Major1 Major2 Conflicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 19 33 Major/Minor Minor2 Minor1 Major1 Major2 Major2 Conflicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 Stage 1 38 38 14 14 -	Heavy Vehicles %	0	2	0	0	0	0	0	0	0	0	0	0	
Major/Minor Minor2 Minor1 Major1 Major2 Conflicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 Stage 1 38 38 - 14 14 - <td< td=""><td>Mymt Flow</td><td>13</td><td>Ď</td><td>n n</td><td>0</td><td>0</td><td>0</td><td>Ő</td><td>14</td><td>0</td><td>0</td><td>19</td><td>33</td><td>5</td></td<>	Mymt Flow	13	Ď	n n	0	0	0	Ő	14	0	0	19	33	5
Major/Minor Minor1 Major1 Major2 Conflicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 Stage 1 38 38 - 14 14 - <td></td> <td>10</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>14</td> <td></td> <td>v</td> <td></td> <td>00</td> <td></td>		10							14		v		00	
Major Minor2 Minor1 Major1 Major2 Conflicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 Stage 1 38 38 - 14 14 - <td>Na 1 (0.4)</td> <td></td> <td>-</td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td></td> <td>17.7</td> <td></td> <td></td> <td></td>	Na 1 (0.4)		-								17.7			
Contlicting Flow All 52 52 38 50 68 14 54 0 0 14 0 0 Stage 1 38 38 - 14 14 -	Major/Minor	vinor2			winor1			vajor1	-	N	najor2	-	-	
Stage 1 38 38 - 14 14 - <th< td=""><td>Conflicting Flow All</td><td>52</td><td>52</td><td>38</td><td>50</td><td>68</td><td>14</td><td>54</td><td>0</td><td>0</td><td>14</td><td>0</td><td>0</td><td>à</td></th<>	Conflicting Flow All	52	52	38	50	68	14	54	0	0	14	0	0	à
Stage 2 14 14 - 36 54 - <th< td=""><td>Stage 1</td><td>38</td><td>38</td><td></td><td>14</td><td>14</td><td>-</td><td></td><td>-</td><td></td><td></td><td>•</td><td>•</td><td></td></th<>	Stage 1	38	38		14	14	-		-			•	•	
Critical Howy 7.1 6.52 6.2 7.1 6.5 6.2 4.1 - - 2.1 - - - - - 2.2 - - 2.2 - - 2.2 - - 2.2 - - 2.2 - - 2.2 <th< td=""><td>Stage 2</td><td>14</td><td>14</td><td>-</td><td>36</td><td>54</td><td>-</td><td></td><td></td><td>-</td><td></td><td></td><td>•</td><td>1</td></th<>	Stage 2	14	14	-	36	54	-			-			•	1
Critical Hdwy Stg 1 6.1 5.52 - 6.1 5.5 - <td< td=""><td>Critical Hdwy</td><td>7.1</td><td>6.52</td><td>6.2</td><td>7.1</td><td>6.5</td><td>6.2</td><td>4.1</td><td></td><td>•</td><td>4.1</td><td></td><td></td><td></td></td<>	Critical Hdwy	7.1	6.52	6.2	7.1	6.5	6.2	4.1		•	4.1			
Critical Howy Stg 2 6.1 5.52 - 6.1 5.5 - <td< td=""><td>Critical Hdwy Stg 1</td><td>6.1</td><td>5.52</td><td></td><td>6.1</td><td>5.5</td><td>-</td><td>-</td><td>•</td><td>•</td><td>•</td><td>•</td><td>•</td><td>5</td></td<>	Critical Hdwy Stg 1	6.1	5.52		6.1	5.5	-	-	•	•	•	•	•	5
Pollow-up Hdwy 3.5 4.018 3.3 3.5 4 3.3 2.2 - 2.2 - - 1617 - - 1617 -	Critical Hdwy Stg 2	6.1	5.52	-	6.1	5.5	-	-	•	11.1				
Pot Cap-1 Maneuver 952 839 1040 955 826 10/2 1564 - - 1617 - - Stage 1 982 863 - 1011 888 -	Follow-up Hdwy	3.5	4.018	3.3	3.5	4	3.3	2.2	-		2.2	-	-	2
Stage 1 982 863 - 1011 888 -	Pot Cap-1 Maneuver	952	839	1040	955	826	1072	1564	-	-	1617	-	-	
Stage 2 1011 884 - 985 854 -	Stage 1	982	863	-	1011	888		-	-		-			à.
Platoon blocked, % -	Stage 2	1011	884	-	985	854	-	-		-	-	-	-	-
Mov Cap-1 Maneuver 950 837 1038 955 824 1072 1561 - - 1617 - - Nov Mov Cap-2 Maneuver 950 837 - 955 824 -	Platoon blocked, %	070	807	4000	0.55	004	4070	4504		•	4047			-
Milor Cap-2 Mathedver 500 637 - 935 624 - <t< td=""><td>Mov Cap-1 Maneuver</td><td>950</td><td>03/</td><td>1038</td><td>900</td><td>024</td><td>1072</td><td>1001</td><td>-</td><td>-</td><td>101/</td><td>-</td><td></td><td></td></t<>	Mov Cap-1 Maneuver	950	03/	1038	900	024	1072	1001	-	-	101/	-		
Stage 1 500 601 - 1011 686 -	Nov Cap-2 Maneuver	950	03/	•	900	024		-				-		1
Stage 2 1011 004 - 900 032 - -	Stage 1	900	001	1	1011	000		•	•		•			
Approach EB WB NB SB HCM Control Delay, s 8.8 0 0 0 HCM LOS A A A Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - - A A - - HCM 1ane LOS A - - A A - -	Stage 2	1011	004	-	900	002			-				-	2
Approach EB WB NB SB HCM Control Delay, s 8.8 0 0 0 HCM LOS A A A 0 0 Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - A A - - - HCM S5th %tile Q(veh) 0 - 0 - 0 - -														
HCM Control Delay, s 8.8 0 0 0 HCM LOS A A A Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - HCM Control Delay (s) 0 - - 8.8 0 - HCM Lane LOS A - - A A - - HCM S5th %tile Q(veh) 0 - 0 - 0 - -	Approach	EB			WB			NB			SB			
HCM LOS A A Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - - A A - - HCM S5th %tile Q(veh) 0 - 0 - 0 - -	HCM Control Delay, s	8.8			0			0			0			
Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - - A A - - HCM 95th %tile Q(veh) 0 - 0 - 0 - -	HCM LOS	Α			Α									
Minor Lane/Major Mvmt NBL NBT NBR EBLn1WBLn1 SBL SBT SBR Capacity (veh/h) 1561 - 950 - 1617 - HCM Lane V/C Ratio - - 0.013 - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - - A A - - HCM 95th %tile Q(veh) 0 - 0 - 0 - -				.8.11							S., 1			
Capacity (veh/h) 1561 - 950 - 1617 - - HCM Lane V/C Ratio - - 0.013 - - - - HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - A A A - - HCM 95th %tile Q(veh) 0 - 0 - 0 - -	Minor Lane/Major Mym	ıt	NBL	NBT	NBR	EBLn1	WBLn1	SBL	SBT	SBR		1 57	2. Mar 21	
HCM Lane V/C Ratio - - 0.013 -	Capacity (veh/h)		1561	-	-	950	-	1617	-	-				
HCM Control Delay (s) 0 - - 8.8 0 - - HCM Lane LOS A - - A A - - HCM 95th %tile Q(veh) 0 - 0 - 0 - -	HCM Lane V/C Ratio		-	-	-	0.013	2	-		-		1.5		
HCM Lane LOS A A A A HCM 95th %tile Q(veh) 0 - 0 - 0	HCM Control Delay (s)		0	-	-	8.8	0	0	-				-	
HCM 95th %tile Q(veh) 0 0 - 0	HCM Lane LOS	and a state of the	A	-	-	A	A	A	-		-	Construction of the local		
	HCM 95th %tile Q(veh)	0		-	0	-	0	-					

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Background Conditions PM Peak Hour RM $\,$

0

06/07/2018

 \bigcirc

Intersection						2							
Int Delay, s/veh	7.2												
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR	
Lane Configurations		đ	1		đ,	1		4			1		-
Traffic Vol. veh/h	17	9	47	61	49	37	66	289	24	2	173	5	-
Future Vol, veh/h	17	9	47	61	49	37	66	289	24	2	173	5	
Conflicting Peds. #/hr	0	0	0	0	0	0	1	0	0	Ū.	0	1	
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	
RT Channelized		-	None			None	-		None	-	-	None	
Storage Length			100			120				-		350	-
Veh in Median Storage	.# -	0	-	-	0	-	÷.	0		-	0	-	-23
Grade, %	-	0		-	0			0			0		
Peak Hour Factor	79	79	79	79	79	79	79	79	79	79	79	79	1
Heavy Vehicles, %	14	14	14	5	5	5	5	5	5	10	10	10	
Mvmt Flow	22	11	59	77	62	47	84	366	30	3	219	6	
Major/Minor	Minor2	-	-	Minor1	12	- 0	Major1	19715		Major2	-1.37		
Conflicting Flow All	830	790	220	812	781	381	226	0	0	396	0	0	
Stage 1	226	226	-	549	549		-	-		-		-	
Stage 2	604	564		263	232					-			
Critical Hdwy	7.24	6.64	6.34	7.15	6.55	6.25	4.15			4.2	-	-	
Critical Hdwy Stg 1	6.24	5.64		6.15	5.55		-		1.1				
Critical Hdwy Stg 2	6.24	5.64	-	6.15	5.55	-	-	-	-	-	-		
Follow-up Hdwy	3.626	4,126	3.426	3.545	4.045	3.345	2.245			2.29			
Pot Cap-1 Maneuver	276	309	791	294	323	660	1325	-	-	1120	-		
Stage 1	750	695		515	512					-		2	
Stage 2	465	490		736	707	-	-		-	-		-	
Platoon blocked, %								-					
Mov Cap-1 Maneuver	201	282	790	246	295	660	1324	-		1120		-	
Mov Cap-2 Maneuver	201	282	-	246	295					-		-	
Stage 1	688	692		473	470			-	-			-	
Stage 2	344	450		667	704	-	-	-	-			-	
	with a	1000					1 Alt				1 all the last	U.,	
Approach	EB		COLUMN T	WB			NB	21175	-	SB			
HCM Control Delay, s	14.9			27.1			1.4			01		-	-
HCM LOS	B	-		D		10-11-11-11-11-11-11-11-11-11-11-11-11-1	1.1			0.1	Sec. 1		-
		20					4.7					-	
Minor Lane/Malor Mym	it	NBL	NBT	NBR	EBLn1	EBLn2	VBL n 1	VBL n2	SBI	SBT	SBR		
Capacity (yeh/h)		1324		-	223	790	266	660	1120	001	ULIN		-
HCM Lane V/C Ratio		0.063		-	0 148	0.075	0.523	0.071	0.002			-	
HCM Control Delay (s)	-	7.9	0		23 0	9.010	32.5	10.071	8.00Z	0	-	S	
HCM Lane LOS	-	A	A	-	C.0	Δ	D	R	Δ	A			
HCM 95th %tile O(veh)		0.2	-		0.5	02	28	0.2	0	-	-	- 291	
the section of the		v.2.			0.0	0.2	2,0	0.2	U	-	1 Carrow	- 11-	

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Site Trips Conditions AM Peak Hour RM $\,$

Ô

ntersection						
Int Delay, s/veh	3.2	-				
Movement	FPT	E80		WPT	NPI	NED
I ano Configurations	ED1	EDK	VADE	VVDT	NDL	NOK
Lane Configurations	4	00	7	404	T	45
rame voi, ven/n	25	23	1	101	51	15
Future Vol, veh/h	25	23	1	101	51	15
Conflicting Peds, #/hr	_ U	- 0	0	0	0	0
Sign Control	Free	Free	Free	Free	Stop	Stop
RI Channelized	-	None	-	None	-	None
Storage Length	•	•	•	•	0	72
Veh in Median Storage,	# 0	•	•	0	0	-
Grade, %	0	-	-	0	0	÷.
Peak Hour Factor	75	75	75	75	75	75
Heavy Vehicles, %	13	13	7	7	17	17
Mymt Flow	33	31	9	135	68	20
			111	_		
And in the second s	-					
Major/Minor N	lajor1	- Contract	Major2		Minor1	1000
Conflicting Flow All	0	0	64	0	202	49
Stage 1	-	-	-		49	
Stage 2		-	-	-	153	-
Critical Hdwy			4 17		6.57	6.37
Critical Hdwy Sto 1					5 57	0.01
Critical Hduay Sta 2	-				5.57	-
Collow up Udwa			0.060		3 663	2 452
Pollow-up Huwy	•	-	2.203	-	3.003	3.433
Pot Cap-1 Maneuver		-	1007	-	/04	9/9
Stage 1	-	-	-	-	936	-
Stage 2		-	-	-	840	-
Platoon blocked, %	-	-		-		
Mov Cap-1 Maneuver	-		1507	•	749	979
Mov Cap-2 Maneuver				-	749	
Stage 1			-		930	-
Stage 2		-	-		840	-
orago z					040	
Approach	EB		WB		NB	
HCM Control Delay, s	Ö		0.5		10.1	111
HCM LOS		-	-		В	
La constantia asses						
Minor Lane/Major Mvm	t	NBLn1	EBT	EBR	WBL	WBT
Capacity (veh/h)		791	-		1507	-
HCM Lane V/C Ratio		0.111	-		0.006	-
HCM Control Delay (s)		10.1			74	D
HCM Lane LOS		R	13-11-13		Δ	Δ
HCM 05th %tile O(uch)	1	0.4			0	~
TOW SOUL WINE CIVEN.	120.10	0.4	-		U	-

()

06/07/2018

Jacoby Road Subdivision TIS 05/15/2018 2020 Site Trips Conditions AM Peak Hour RM

HCM 2010 TWSC 3: SE Jacoby Road & Street B

06/07/2018

 \bigcirc

		-	_			
Intersection		-		-	-	
Int Delay, s/veh	2.3					
Movement	WBL	WBR	NBT	NBR	SBL	SBT
Lane Configurations	M		Ť.			\$
Traffic Vol. veh/h	0	11	28	0	4	10
Future Vol. veh/h	0	11	28	0	4	10
Conflicting Peds. #/hr	0	0	0	0	Ö	0
Sign Control	Stop	Ston	Free	Free	Free	Free
RT Channelized	-	None	-	None	-	None
Storage Length	0		-	-		-
Veh in Median Storage	e # 0		0			Ω
Grade %	0		0	-		0
Peak Hour Factor	92	92	02	02	02	02
Heavy Vehicles %	32	2	32	32	32	32
Mumt Flow	4	10	20	4	2	44
	U	12	30	U	4	- 11
Major/Minor	Minor1	1	Najor1	1	Major2	4
Conflicting Flow All	49	30	0	0	30	0
Stage 1	30				-	
Stage 2	19	-			-	
Critical Hdwv	6.42	6.22			4.12	-
Critical Hdwy Sto 1	5 42	-				
Critical Hdwy Sto 2	5.42	-		-		
Follow-up Hdwy	3 518	3 318			2 218	
Pot Can_1 Maneuver	960	1044		-	1593	
Stane 1	000	1044	-		1000	-
Stage 7	1004		-		•	
Diatoon blocked %	1004		-	-		-
Fidtuun Diücken, %	007	1014	-		4500	-
Mov Cap-1 Maneuver	957	1044	-	•	1583	-
Mov Cap-2 Maneuver	95/	. *	-		-	
Stage 1	990	-	-			-
Stage 2	1004		-	-	-	•
				6.191		
Approach	WB		NB	1015-1	SB	Tre-se
HCM Control Delay	8.5	-	0	-	21	-
HCM LOS	۵.0	-	0	-	4.1	1000
	~					
Minor Lane/Major Mvn	nt	NBT	NBRV	VBLn1	SBL	SBT
Capacity (veh/h)		-		1044	1583	-
HCM Lane V/C Ratio		-	-	0.011	0.003	-
HCM Control Delay (s)	-	-	8.5	7.3	Ó
HCM Lane LOS		-		A	A	A
HCM 95th %tile Q(veh	0		-	0	0	-

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Site Trips Conditions AM Peak Hour RM $\,$

 \bigcirc

06/07/2018

Õ

Intersection				1	-			11-21					and the second second
Int Delay, s/veh	3.2												
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR	
Lane Configurations		4			4.			4.			4.		
Traffic Vol, veh/h	6	0	0	0	0	6	0	16	0	2	5	3	
Future Vol, veh/h	6	0	0	0	0	6	0	16	0	2	5	3	
Conflicting Peds, #/hr	0	0	0	0	0	0	2	0	0	0	0	2	
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free	11.000
RT Channelized			None			None	-	-	None	-	-	None	1000
Storage Length				-		-		-		•	•	-	
Veh in Median Storage	.# -	0	-	-	0	-		0		•	0		
Grade, %	-	0	-	-	0	-		0		•	0	-	
Peak Hour Factor	56	56	56	56	56	56	56	56	56	56	56	56	
Heavy Vehicles, %	33	33	33	0	0	0	13	13	13	0	0	0	
Mvmt Flow	11	0	0	0	0	11	0	29	0	4	9	5	
Major/Minor	Minor2			Minor1			Major1	1		Major2			144 E
Conflicting Flow All	57	51	14	49	53	29	16	0	0	29	0	0	
Stage 1	22	22	-	29	29	•	1111	- 11	-	-	-		
Stage 2	35	29	-	20	24	-	-		-	-	-		
Critical Hdwy	7.43	6.83	6.53	7.1	6.5	6.2	4.23		- 11	4.1			1
Critical Hdwy Stg 1	6.43	5.83		6.1	5.5	-	-		-	-		-	
Critical Hdwy Stg 2	6.43	5.83	-	6.1	5.5	-	-		-		-	100 I 1 - 1	
Follow-up Hdwy	3.797	4.297	3.597	3.5	4	3.3	2.317	•	•	2.2	-		
Pot Cap-1 Maneuver	868	784	983	956	842	1052	1533		-	1597	-	-	
Stage 1	922	819	-	993	875		•	•	•	-		-	
Stage 2	907	813		1004	879	- 11-		•		-	-	() (-	
Platoon blocked, %								-	-			-	
Mov Cap-1 Maneuver	856	780	981	954	838	1052	1530		-	1597	-		
Mov Cap-2 Maneuver	856	780	-	954	838	-	-	-	-	-	-		
Stage 1	920	815		993	875	311 / A#	-	-		-		1	
Stage 2	898	813		1001	875	•	•	•	•			(a)	
Approach	EB			WB	2.10		NB))=0.,	3. 3.6	SB			
HCM Control Delay, s	9.3			8.5			0	- 14	den inge	1.5			
HCM LOS	A	_	1000	A	_		_	-	-		_		
VE I BI-l- H		MIDI	NIDT	MDD	CD1	1001 - 1	001	007	000	1.14	-		
Minor Lane/Major MVIT	u	NBL	NBI	NBR		VBLN1	SBL	SBI	SBR	-			
Capacity (veh/h)		1530	•	-	856	1052	1597		-		Orana I.		A REAL PROVIDE
HCM Lane V/C Ratio		•	•	•	0.013	0.01	0.002	-	-	_	_		
HCM Control Delay (s)		0	-	•	9.3	8.5	7.3	0	-			Neurie	
HCM Lane LOS		A	•		A	A	A	A	-				
HCM 95th %tile Q(veh)	0	-	-	0	0	0	-	-				

Jacoby Road Subdivision TIS 05/15/2018 2020 Site Trips Conditions AM Peak Hour RM

06/07/2018

Intersection												
Int Delay, s/veh	5.9					-			-	1.11	at a set of	
Mauramant	EDI	697	CDD	LA/DI	14/07	14/00	8.1031	NOT	A LTD PD	001	007	080
	EDL	CDI	EBR	AARF	WBI	WBR	URL	NBI	NBR	SBL	SBI	SER
Lane Configurations	-	- स	7		र्भ	7	-	- 44			र्स	7
I rattic Vol, veh/h	6	59	77	36	37	37	71	260	71	16	314	13
Future Vol, veh/h	6	59	77	36	37	37	71	260	71	16	314	13
Conflicting Peds, #/hr	0	0	0	0	0	0	1	0	0	0	0	1
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free
RT Channelized	-	-	None	-	•	None	-		None	-	•	None
Storage Length	-		100		-	120		-	-	•	-	350
Veh in Median Storage	e,# -	0	- 10		0	-	-	0	-	-	0	-
Grade, %	-	0	•		0			0	-		0	-
Peak Hour Factor	93	93	93	93	93	93	93	93	93	93	93	93
Heavy Vehicles, %	1	1	1	1	1	1	1	1	1	3	3	3
Mvmt Flow	6	63	83	39	40	40	76	280	76	17	338	14
Major/Minor	Minor2			Minor1	i gene		Major1	-		Major2	1.00	SALE
Conflicting Flow All	883	881	339	922	857	318	353	0	0	356	0	0
Stage 1	373	373	-	470	470			-		-		-
Stage 2	510	508		452	387		12			-	-	
Critical Hdwy	7.11	6.51	6.21	7.11	6.51	6.21	4.11	-	-	4 13	-	
Critical Hdwy Sto 1	6.11	5.51	-	6.11	5.51	-	-				-	
Critical Hdwy Sto 2	6.11	5.51	-	6.11	5.51		1.1.1	-	-			
Follow-up Hdwv	3,509	4,009	3,309	3,509	4.009	3,309	2,209			2,227		-
Pot Cap-1 Maneuver	267	287	706	252	296	725	1211	-		1197	-	
Stage 1	650	620	-	576	562	-			-			
Stage 2	548	540	-	589	611	-					1	
Platoon blocked, %		0.0	1			2.71		-				
Moy Cap-1 Maneuver	207	259	705	168	268	725	1210		1000	1197		
Moy Cap-2 Maneuver	207	259		168	268			-				
Stage 1	598	608	1	530	518					1.12		
Stage 2	440	497	-	457	599	-	-					-
				101	300		2 Y I					
Approach	FB	-		WR	Ething	-	NR			SP		
HCM Control Delay	17 1			25.1			1.4	-	10 10 10	0.4		
HCMLOS	0			20.1 D		-	1.4			0.4		
	U		200	U			EX		2 - L	-	-	
Minor Lane/Malor Mar	nt	MBI	NRT	NRP	EBI n1	COL n24	MPI n4	MPI -7	201	CDT	COD	
Capacity (uph(h)	in .	1210	NDI	NON		201121	207	TOLIZ	1407	ODI	SDR	
HCML and MC Defin	2.50 33	0.000		100	203	100	207	120	1197	-	•	
HCM Control Datau (a)	1	0.003	-	•	0.270	U.11/	0.3/9	0.000	0.014	-	•	1000
HCM Leng LOS)	0.2	U	•	24.0	10.8	32.6	10.3	8.1	U	-	
HOW Lane LUS	4	A	A	•	U d	B	U	B	A	A	•	
HOW SOLU WING OLAN	1	0.2	-	-	1.1	0.4	1.7	0.2	0	•	-	

Jacoby Road Subdivision TIS 05/15/2018 2020 Site Trips Conditions PM Peak Hour RM

Intersection 2 Int Delay, s/veh Movement EBT EBR WBL WBT NBL NBR Lane Configurations न 72 Ψ Þ 117 39 Traffic Vol, veh/h 17 12 81 Future Vol, veh/h 117 81 17 72 39 12 Conflicting Peds, #/hr 0 0 0 0 2 1 Sign Control Free Free Free Free Stop Stop **RT** Channelized -None -None None Storage Length 0 Veh in Median Storage, # 0 0 0 . . . Grade, % 0 0 0 • . -Peak Hour Factor 79 79 79 79 79 79 Heavy Vehicles, % 1 1 3 3 3 3 148 Mvmt Flow 103 22 91 49 15 Majon/Minor Majort Major2 Minor1 **Conflicting Flow All** 0 0 251 0 336 202 200 Stage 1 -. ---Stage 2 136 ---Critical Hdwy 4.13 6.23 ---6.43 Critical Hdwy Stg 1 ---• 5.43 • Critical Hdwy Stg 2 5.43 --• • -- 2.227 Follow-up Hdwy - 3.527 3.327 . Pot Cap-1 Maneuver - 1309 657 836 • -Stage 1 831 ---Stage 2 -888 . . --Platoon blocked, % _ Mov Cap-1 Maneuver - 1309 645 834 --Mov Cap-2 Maneuver 645 -_ . . -Stage 1 816 -. + • -Stage 2 . . -. 887 -WB NB €B Approach HCM Control Delay, s 0 10.8 1.5 HCM LOS В Minor Lane/Major Mvmt NBLn1 EBT EBR WBL WBT Capacity (veh/h) 681 -1309 -HCM Lane V/C Ratio 0.095 -0.016 HCM Control Delay (s) 0 10.8 - 7.8 -HCM Lane LOS В A А --HCM 95th %tile Q(veh) 0.3 0.1 . . .

Jacoby Road Subdivision TIS 05/15/2018 2020 Site Trips Conditions PM Peak Hour RM

Synchro 10 Report Page 2

06/07/2018

HCM 2010 TWSC 3: SE Jacoby Road & Street B

O

Intersection		- W		-		
Int Delay, s/veh	1.9					-
Movement	WRI	WRR	NRT	NRD	SRI	SPT
Lana Configurations	TTDL	TUN	TUF	MUN	ODL	001
Lane Conligurations	T	ō	6	0	40	4
Tranic Vol, ven/n	0	9	24	U	13	41
Future Vol, Ven/n	0	9	24	0	13	4/
Conflicting Peds, #/nr	0	U	0	0	0	0
Sign Control	Stop	Stop	Free	Free	Free	Free
RI Channelized	-	None	-	None	-	None
Storage Length	0	•	-	-		•
Veh in Median Storage	e,# 0	-	0	-		0
Grade, %	0	1	0	-		0
Peak Hour Factor	92	92	92	92	92	92
Heavy Vehicles, %	2	2	2	2	2	2
Mymt Flow	0	10	26	0	14	51
ethere in the second				-		
Li : 40	-	_				
Major/Minor	Minor1		Majori	44	Major2	and the second
Conflicting Flow All	105	26	0	0	26	0
Stage 1	26	-			÷	-
Stage 2	79	-	-			-
Critical Hdwy	6.42	6.22	-		4.12	-
Critical Hdwy Sto 1	5.42					
Critical Hdwy Stg 2	5.42				The start of	-
Follow-up Hdwa	3 518	3 318		-	2 218	
Pot Cap. 1 Managemer	902	1050			1590	
Ciaco 1	000	1000		-	1000	
Staye 1	997	-	-	-	-	
Stage 2	944		-	-		
Platoon blocked, %			-	-		-
Mov Cap-1 Maneuver	885	1050	-		1588	-
Mov Cap-2 Maneuver	885		-	•	•	-
Stage 1	988	-	-	-		-
Stage 2	944			-		-
			den 1			
Annmach	VA/D		AID	-	00	
HOM Cartral Dal	WD		NB		38	-
HOM CONTrol Delay, S	8.5	2	0		1.6	
HCM LOS	A		_	_		
C. C. Standard Street						
Minor Lane/Major Mvm	nt	NBT	NBRV	VBLn1	SBL	SBT
Capacity (veh/h)				1050	1588	-
HCM Lane V/C Ratio		and the second second		0 000	0.000	
HCM Control Delay (e)				2.000	72	0
HCM Lana LOS	9-0-		-	0.J	7.J	U A
HCM 05th Mala Ofers	1			A	A	A
HOW SOU WING CIAN	J	-	- 12	0	U	-

 \bigcirc

06/07/2018

Jacoby Road Subdivision TIS $\,$ 05/15/2018 2020 Site Trips Conditions PM Peak Hour RM $\,$

(

06/07/2018

()

Intersection								20				
Int Delay, s/veh	2.4											
Movement	EBL	EBT	EBR	WBL	WBT	WBR	NBL	NBT	NBR	SBL	SBT	SBR
Lane Configurations		4			4			4			4	
Traffic Vol, veh/h	10	0	0	0	0	6	0	11	0	6	15	26
Future Vol, veh/h	10	0	0	0	0	6	0	11	0	6	15	26
Conflicting Peds, #/hr	0	0	0	0	0	0	2	0	0	0	0	2
Sign Control	Stop	Stop	Stop	Stop	Stop	Stop	Free	Free	Free	Free	Free	Free
RT Channelized	-	-	None	-	-	None	-		None			None
Storage Length		-			- AND	- CO	•	-			-	
Veh in Median Storage.	# -	0		-	0		-	0	1	-	0	-
Grade, %		0			0		-	0	7	10	0	-
Peak Hour Factor	79	79	79	79	79	92	79	79	79	79	79	79
Heavy Vehicles, %	0	0	0	0	0	0	0	0	0	0	0	0
Mymt Flow	13	Ð	0	0	0	7	0	14	0	8	19	33
Major/Minor N	/linor2	2		Vinor1		1	Major1	10100	1	lajor2		
Conflicting Flow All	72	68	38	66	84	14	54	0	0	14	0	0
Stage 1	54	54	-	14	14	-	-	-		-		
Stage 2	18	14		52	70	-	-	-	÷	-	-	
Critical Hdwy	7.1	6.5	6.2	7.1	6.5	6.2	4.1	-		4.1	-	
Critical Hdwy Stg 1	6.1	5.5		6.1	5.5					-		
Critical Hdwy Stg 2	6.1	5.5	-	6.1	5.5	-						
Follow-up Hdwy	3.5	4	3.3	3.5	4	3.3	2.2	-	-	2.2	-	-
Pot Cap-1 Maneuver	924	826	1040	932	810	1072	1564	-	-	1617	-	-
Stage 1	963	854	-	1011	888	-	-			-	-	
Stage 2	1006	888	-	966	841	-	-	-	-	-	-	(#
Platoon blocked, %												
Mov Cap-1 Maneuver	913	820	1038	928	804	1072	1561	-		1617		
Mov Cap-2 Maneuver	913	820	-	928	804							
Stage 1	961	848		1011	888				-			
Stage 2	1000	888	-	961	835	-	-			-	-	
		The state			1000							1
Approach	EB			WB			NB			SB	TET	
HCM Control Delay, s	9			8.4			0			0.9		
HCM LOS	A			A								
Minor Lane/Major Mym	t	NBL	NBT	NBR	EBLn1	W8Ln1	SBL	SBT	SBR	13. J. T	EN DS	105
Capacity (yeh/h)	The state	1561	-	-	913	1072	1617	-	-		-	
HCM Lane V/C Ratio		1001			0.014	0.006	0.005	1		· · · · ·		
HCM Control Delay (e)		0			0.014	8.4	7 2	0				
HCM Lane LOS	P. 12	4			5	Δ	Δ	Δ	-			
HCM 95th %tile O/veh	1	0			0	0	- -	-	1			
LIDULINUE ALABIT		U			U	U	U		-			

Jacoby Road Subdivision TIS 05/15/2018 2020 Site Trips Conditions PM Peak Hour RM





Ŀ

TRIP GENERATION CALCULATIONS

Land Use: Single-Family Detached Housing Land Use Code: 210 Setting/Location General Urban/Suburban Variable: Dwelling Units Variable Value: 40

AM PEAK HOUR

Trip Rate: 0.74

0

	Enter	Exit	Total
Directional Distribution	25%	75%	
Trip Ends	8	22	30

	Enter	Exit	Total
Directional Distribution	63%	37%	
Trip Ends	25	15	40

PM PEAK HOUR

Trip Rate: 0.99

WEEKDAY

Trip Rate: 9.44

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	189	189	378

SATURDAY

Trip Rate: 9.54

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	191	191	382

Source: TRIP GENERATION, Tenth Edition
1e

TRIP GENERATION CALCULATIONS

Land Use: Single-Family Detached Housing Land Use Code: 210 Setting/Location General Urban/Suburban Variable: Dwelling Units Variable Value: 5

AM PEAK HOUR

Trip Rate: 0.74

	Enter	Exit	Total
Directional Distribution	25%	75%	
Trip Ends	1	3	4

	Enter	Exit	Total
Directional Distribution	63%	37%	
Trip Ends	3	2	5

PM PEAK HOUR

Trip Rate: 0.99

WEEKDAY

Trip Rate: 9.44

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	24	24	48

Source: TRIP GENERATION, Tenth Edition

SATURDAY

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	24	24	48

10

TRIP GENERATION CALCULATIONS

Land Use: Single-Family Detached Housing Land Use Code: 210 Setting/Location General Urban/Suburban Variable: Dwelling Units Variable Value: 11

AM PEAK HOUR

Trip Rate: 0.74

0

	Enter	Exit	Total
Directional Distribution	25%	75%	
Trip Ends	2	6	8

	Enter	Exit	Total
Directional Distribution	63%	37%	
Trip Ends	7	4	11

PM PEAK HOUR

Trip Rate: 0.99

WEEKDAY

Trip Rate: 9.44

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	52	52	104

Source: TRIP GENERATION, Tenth Edition

SATURDAY

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	52	52	104

1e

TRIP GENERATION CALCULATIONS

Land Use: Single-Family Detached Housing Land Use Code: 210 Setting/Location General Urban/Suburban Variable: Dwelling Units Variable Value: 1

AM PEAK HOUR

Trip Rate: 0.74

	Enter	Exit	Total
Directional Distribution	25%	75%	
Trip Ends	0	1	1

	Enter	Exit	Total
Directional Distribution	63%	37%	
Trip Ends	1	0	1

PM PEAK HOUR

Trip Rate: 0.99

WEEKDAY

Trip Rate: 9.44

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	5	5	10

Source: TRIP GENERATION, Tenth Edition

SATURDAY

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	5	5	10

TRIP GENERATION CALCULATIONS

Land Use: Single-Family Detached Housing Land Use Code: 210 Setting/Location General Urban/Suburban Variable: Dwelling Units Variable Value: 32

AM PEAK HOUR

Trip Rate: 0.74

 \bigcirc

	Enter	Exit	Total
Directional Distribution	25%	75%	
Trip Ends	6	18	24

	Enter	Exit	Total
Directional Distribution	63%	37%	
Trip Ends	20	12	32

PM PEAK HOUR

Trip Rate: 0.99

le

WEEKDAY

Trip Rate: 9.44

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	151	151	302

Source: TRIP GENERATION, Tenth Edition

SATURDAY

	Enter	Exit	Total
Directional Distribution	50%	50%	
Trip Ends	153	153	306

-		•			1	
Project:	18067 - SE Jacob	y Road Subdiv	ision		φ	
Date:	6///2018 2049 Evicting Oct					
Scenario:	2018 Existing Col	nditions				
Major Street:	OR 211		Minor Street:	Dubarko Road		
Number of Lanes:	1		Number of Lanes:	1		
PM Peak Hour Volumes:	621		PM Peak Hour Volumes:	214		
Warrant Used:						
X	100 percent of stan	dard warrants us	ed			
	70 percent of stand	ard warrants use	d due to 85th perce	entile speed in exce	ess	
	of 40 mph or isolate	ed community wit	h population less th	nan 10,000.		
Number of	Lanes for Moving	ADT on I	Major St.	ADT on M	linor St.	
Traffic on	Each Approach:	(total of both	approaches)	(higher-volum	e approach)	
WARRANT 1, CO		100%	70%	100%	70%	
Major St.	Minor St.	Warrants	Warrants	Warrants	Warrants	
1	1	8,850	6,200	2,650	1,850	
2 or more	1	10,600	7,400	2,650	1,850	
2 or more	2 or more	10,600	7,400	3,550	2,500	
1	2 or more	8,850	6,200	3,550	2,500	
WARRANT 1, CO	NDITION B					
1	1	13,300	9,300	1,350	950	
2 or more	1	15,900	11,100	1,350	950	
2 or more	2 or more	15,900	11,100	1,750	1,250	
1	2 or more	13,300	9,300	1,750	1,250	
	Note: ADT volumes assume 8th highest hour is 5.6% of the daily volume					
		Approach	Minimum	Is Signal Warrant Met2		
Warrant 1		volumes	Volumes	AACH CALL INCL		
Condition A: Minir	num Vehicular Volum	e				
Major Street		6,210	8,850			
Minor Street*		2,140	2,650	No		
Condition B: Inten	ruption of Continuous	Traffic				
Major Street		6.210	13.300			
Minor Street*		2,140	1,350	No		
Combination Wan	rant					
Major Street		6,210	10,640			
Minor Street*		2 1/0	2 120	No		

Ó

 \bigcirc

Traffic Signa	al Warrant Ar	nalvsis			
Project:	18067 - SE Jacol	by Road Subdi	vision		6
Date:	6/7/2018	-,			て
Scenario:	2020 Background	d Conditions			
Major Street:	OR 211		Minor Street:	Dubarko Road	
- Number of Lanes	^{ie} 1		Number of Lane:	si: 1	
PM Peak Hour Volumes:	719		PM Peak Hour Volumes:	252	
Warrant Used:					
X	100 percent of star	idard warrants u	sed		
	70 percent of stand	lard warrants us	ed due to 85th per	centile speed in exce	ess
<u> </u>	of 40 mph or isolate	ed community w	ith population less	than 10,000.	
Number of	Lanes for Moving	ADT on	Major St.	ADT on M	linor St.
Traffic on	Each Approach:	(total of boti	n approaches)	(higher-volum	e approach)
WARRANT 1, COI	NDITION A	100%	70%	100%	70%
Major St.	Minor St.	Warrants	Warrants	Warrants	Warrants
1	1	8,850	6,200	2,650	1,850
2 or more	1	10,600	7,400	2,650	1,850
2 or more	2 or more	10,600	7,400	3,550	2,500
1 ി	2 or more	8,850	6,200	3,550	2,500
WARRANT 1, COI	NDITION B				
1	1	13,300	9,300	1,350	950
2 or more	1	15,900	11,100	1,350	950
2 or more	2 or more	15,900	11.100	1.750	1.250
1	2 or more	13,300	9,300	1,750	1,250
		Note: ADT v	olumes assume 8th hi	ghest hour is 5.6% of the	daily volume
		Approach Volumes	Minimum Volumes	Is Signal Warrant Met?	
Warrant 1					
Condition A: Minim	num Vehicular Volum	e			
Major Street		7,190	8,850		
Minor Street*		2,520	2,650	No	
Condition B: Intern	uption of Continuous	Traffic			
Major Street		7,190	13,300		
Minor Street*		2,520	1,350	No	
Combination Warn	ant				
oomonighter them		7 400	10 640		
Major Street		(.190	10,040		

in anne eigen		aryono			1			
Project: Date:	18067 - SE Jacob 6/7/2018	18067 - SE Jacoby Road Subdivision						
Scenario:	2020 Background	Conditions						
Major Street:	OR 211		Minor Street:	Dubarko Road				
Number of Lanes:	1		Number of Lanes:	1				
PM Peak Hour Volumes:	710		PM Peak Hour Volumes:	238				
Warrant Used:								
X	_100 percent of stan	dard warrants u	sed					
	70 percent of stand	ard warrants use	ed due to 85th perce	entile speed in exce	ess			
	of 40 mph or isolate	ed community wi	th population less th	nan 10,000.				
Number of	Lanes for Moving	ADT on	Major St.	ADT on N	linor St.			
Traffic on	Each Approach:	(total of both	n approaches)	(higher-volum	e approach)			
WARRANT 1, COI	NDITION A	100%	70%	100%	70%			
<u>Major St.</u>	Minor St.	Warrants	Warrants	Warrants	Warrants			
1	1	8,850	6,200	2,650	1,850			
2 or more	1	10,600	7,400	2,650	1,850			
2 or more	2 or more	10,600	7,400	3,550	2,500			
ୀ	2 or more	8,850	6,200	3,550	2,500			
WARRANT 1, CO	NDITION B							
1	1	13,300	9,300	1,350	950			
2 or more	1	15,900	11,100	1,350	950			
2 or more	2 or more	15,900	11,100	1,750	1.250			
1	2 or more	13,300	9,300	1,750	1,250			
		Note: ADT v	olumes assume 8th high	est hour is 5.6% of the	daily volume			
		Approach	Minimum	Is Signal				
Warrant 1		volumes	volumes	vvarrant iviet?				
Condition A: Minin	num Vehicular Volum	e						
Major Street		7 100	8 850					
Minor Street*		2,380	2.650	No				
Oradillar Di tat		_,						
Condition B: Inten Major Street	uption of Continuous	7 100	13 300					
Minor Street*		2,380	1.350	No				
		2,000	1,000					
Combination Wari	rant	-	10.010					
Major Street		7,100	10,640					

Ó ()

 \bigcirc

Project:	18076 - SE Jacoby Road TIS
Intersection:	Dubarko Road at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - AM

2-lane roadway (English)

ļ	N	Ρ	U	T		

Variable	Value
85 th percentile speed, mph:	30
Left-turns in advancing volume (V _A), veh/hr:	7
Advancing volume (V _A), veh/h:	108
Opposing volume (V ₀), veh/h:	49

Ŀ

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	744	
Suidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9

Ô

Project:	18076 - SE Jacoby Road TIS
Intersection:	Dubarko Road at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - PM

2-lane INPUT	roadway	(English)
1		

Variable	Value
85 th percentile speed, mph:	30
Left-turns in advancing volume (V _A), veh/hr:	17
Advancing volume (V _A), veh/h:	89
Opposing volume (V ₀), veh/h:	198

()

Ŀ

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	391	
Guidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9

 \bigcirc

Project:	18076 - SE Jacoby Road TIS
Intersection:	Newton Road at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - AM

2-lane roadway (English) INPUT

Ν	Pι	Л		
-	_	_	_	_

Variable	Value
85 th percentile speed, mph:	25
Left-turns in advancing volume (V _A), veh/hr:	2
Advancing volume (V _A), veh/h:	10
Opposing volume (V ₀), veh/h:	16

Ŀ

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	500	
Guidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9

Ó

Project:	18076 - SE Jacoby Road TIS
Intersection:	Newton Road at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - PM

2-lane roadway (English)

INPUT	
-------	--

Variable	Value
85 th percentile speed, mph:	25
Left-turns in advancing volume (V _A), veh/hr:	6
Advancing volume (V _A), veh/h:	47
Opposing volume (V ₀), veh/h:	11

()

le

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	603	
Guidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9

 \bigcirc

Project:	18076 - SE Jacoby Road TIS
Intersection:	Street B at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - AM

2-lane roadway (English)

INPUT

Variable	Value
85 th percentile speed, mph:	25
Left-turns in advancing volume (V _A), veh/hr:	4
Advancing volume (V _A), veh/h:	14
Opposing volume (V ₀), veh/h:	28

1e

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	436	
Guidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9

Ô

Project:	18076 - SE Jacoby Road TIS
Intersection:	Street B at SE Jacoby Road
Date:	6/7/2018
Scenario:	2020 Total Traffic - PM

2-lane roadway (English)

INPUT

Variable	Value
85 th percentile speed, mph:	25
Left-turns in advancing volume (V _A), veh/hr:	13
Advancing volume (V _A), veh/h:	60
Opposing volume (V _O), veh/h:	24

 \bigcirc

le

OUTPUT

Variable	Value	
Limiting advancing volume (V _A), veh/h:	481	
Guidance for determining the need for a major-road left-turn bay:		
Left-turn treatment NOT warranted.		



Variable	Value
Average time for making left-turn, s:	3.0
Critical headway, s:	5.0
Average time for left-turn vehicle to clear the advancing lane, s:	1.9



Speed?

 Project:
 18067 - SE Jacoby Road Subdivision

 Date:
 6/7/2018

 Scenario:
 2018 Existing Conditions - 211 at Dubarko

0

45 mph

	PM Peak Hour	
18	Right-Turn Volume	60
155	Approaching DHV	281
No	Lane Needed?	Yes
	18 155 No	PM Peak Hour 18 Right-Turn Volume 155 Approaching DHV No Lane Needed?

72 kmh



Ô

Note:

If there is no right turn lane, a shoulder needs to be provided.

If this intersection is in a rural area and is a connection to a public street, a right turn lane is needed.

Project: 13106 - Silverstone Intersection: Nehalem Highway 47 at David Hill Road Date: 6/7/2018 Scenario: 2020 Background

Speed? 50 mph

AM Peak Hour	90	PM Peak Hour	PM Peak Hour		
Left-Turn Volume		Left-Turn Volume	Left-Turn Volume 69		
Approaching DHV	326	Approaching DHV	349		
# of Advancing Through Lanes	1	# of Advancing Through Lanes	1		
Opposing DHV	304	Opposing DHV	485		
# of Opposing Through Lanes	1	# of Opposing Through Lanes	1		
O+A DHV	630	O+A DHV	834		
Lane Needed?	Yes	Lane Needed?	Yes		

()



*(Advancing Vol/ # of Advancing Through Lanes)+ (Opposing Vol/ # of Opposing Through Lanes)

Note: The criterion is not met from zero to ten left turn vehicles per hour, but careful consideration should be given to installing a left turn lane due to the increased potential for accidents in the through lanes. While the turn volumes are low, the adverse safety and operational impacts may require installation of a left turn. The final determination will be based on a field study.



Project: 18067 - Jacoby Road Subdivision Intersection: OR 211 at Dubarko Road Date: 6/7/2018 Scenario: 2020 Background plus Site Trips

0

Speed? 45 mph

	PM Peak Hour		AM Peak Hour
71	Left-Turn Volume	66	Left-Turn Volume
390	Approaching DHV	367	Approaching DHV
1	# of Advancing Through Lanes	1	# of Advancing Through Lanes
329	Opposing DHV	173	Opposing DHV
1	# of Opposing Through Lanes	1	# of Opposing Through Lanes
719	O+A DHV	540	O+A DHV
Yes	Lane Needed?	Yes	Lane Needed?



*(Advancing Vol/ # of Advancing Through Lanes)+

(Opposing Vol/ # of Opposing Through Lanes)

Note. The criterion is not met from zero to ten left turn vehicles per hour, but careful consideration should be given to installing a left turn lane due to the increased potential for accidents in the through lanes. While the turn volumes are low, the adverse safety and operational impacts may require installation of a left turn. The final determination will be based on a field study

Project:	18067 - Jacoby Road Subdivision
Intersection:	OR 211 at Dubarko Road
Date:	6/7/2018
Scenario:	2020 Background plus Site Trips

45 mph

0

Speed?

AM Peak Hour Left-Turn Volume	2	PM Peak Hour Left-Turn Volume	16
Approaching DHV # of Advancing Through Lanes	173 1	Approaching DHV # of Advancing Through Lanes	329 1
Opposing DHV # of Opposing Through Lanes	367 1	Opposing DHV # of Opposing Through Lanes	390 1
O+A DHV	540	O+A DHV	719
Lane Needed?	No	Lane Needed?	Yes

 \bigcirc



*(Advancing Vol/ # of Advancing Through Lanes)+ (Opposing Vol/ # of Opposing Through Lanes)

Note: The criterion is not met from zero to ten left tum vehicles per hour, but careful consideration should be given to installing a left turn lane due to the increased potential for accidents in the through lanes. While the turn volumes are low, the adverse safety and operational impacts may require installation of a left turn. The final determination will be based on a field study.

















TABLE OF CONTENTS

0

(

1.0 GENERAL INFORMATION
2.0 SITE CONDITIONS
2.1 Surface Conditions
2.2 Regional Geology
2.3 Field Explorations and Surfaces Conditions
2.3.1 Field Explorations
2.3.2 Surface Conditions
2.3.3 Groundwater
3.0 GEOTECHNICAL DESIGN RECOMMENDATIONS
3.1 Foundation
3.2 Floor Slabs
3.3 Seimic Design
3.4 Slopes and Hazards
4.0 CONSTRUCTION RECOMMENDATIONS
4.1 Site Preparation
4.1.1 Proof Rolling
4.1.2 Wet Weather Conditions
4.2 Excavation
4.3 Structural Fills
4.3.1 Native Soils
4.3.2 Imported Granular Fill
4.3.3 Pavement Base Aggregate.
4.4 Drainage Considerations
5.0 CONSTRUCTION OBSERVATIONS
6.0 LIMITATIONS

SUPPORTING DATA

Appendix A - F	igures
Figure 1	Location Plan
Figure 2	Assessor's Map
Figure 3	Preliminary site plan with testing locations

2

Appendix B – Soil Logs, Laboratory data

Jacoby Road Subdivision

1.0 PROJECT AND SITE DESCRIPTIONS

Rapid Soil Solutions (RSS) has prepared this geotechnical report, as requested, for the proposed 32-lot subdivision of the Clackamas County parcel currently assigned the street address of 19124 SE Jacoby Road in the unincorporated outskirts of Sandy, Oregon (97055). The site is situated on the eastern side of SE Jacoby Road roughly 1,300 feet south of its intersection with Dubarko Road. It is tucked between sites assigned the street address of 19098 and 19364 Jacoby Road with a shared driveway for 19260, 19264 and 19270 SE Jacoby Road (tucked behind the subject site and adjacent parcels) running adjacent the southern edge of the subject site. West of the subject site are 12 lots of the Cascade Village and Cascade Village No. 5 plats (Cascade Village East Phase I and II). The properties adjacent to the north and west of the subject site are situated within the incorporated area of Sandy. The site can be found in the northeast quarter of Section 24, Township 2-South, Range 4-East (W.M.) in Clackamas County and can be distinguished by the lot number 2300 (TL 24E24A 02300). The site is roughly 0.43 miles east-southeast of Highway No. 172 (Eagle Creek), 0.75 miles southwest of U.S. Highway 26 (Mt. Hood Highway) and is 0.83 miles south of the Proctor Blvd where is diverges from Pioneer Blvd to form two one-way sections of HWY-26 that pass through downtown Sandy. The site is not part of a subdivision. The latitude and longitude of the site are 45.385063 and -122.256340 (45°23'06.2"N, 122°15'22.8"W). The site can be found within the southeastern corner of the Sandy, OR 7.5minute quadrangle (NE 1/4 of the Boring 15' Quadrangle).

2.0 SITE CONDITIONS

2.1 Surface Conditions

This 9.64-acre subject site is situated along the southeastern edge of the City of Sandy, Oregon at the toe of the Oregon Cascade Range at the foothills of Mount Hood. The site is surrounded on all sides by low-density development. The site and properties north, east and south of the site are situated within the 'low-density' designation of the comprehensive plan for the City of Sandy. The site and those to the east and south are outside the city limits and are zoned by the County as Rural Residential Farm Forest 5-Acres (RRFF5) which is included with a set districts collectively referred to as rural residential and future urban residential zoning districts. The site to the north of the subject site is zoned as 'Single Family Residential' by the City of Sandy, and area intended to implement the Low Density Residential Comprehensive Plan designation by allowing limited development of properties while not precluding more dense future development, as urban services become available.

The properties west of the subject site are zoned low-density residential, providing an urban level of low-density residential development. The sites north, east and south of the subject sites contain single-family residences on lots between 4.00 acres and 10.04 acres. The site to the north of the subject site contains a single-family dwelling constructed in 2008 on a 1.47-acre lot. A section of the land adjacent to the northern edge of the subject site contains a flood slope hazard overlay and is not developed. The properties west of the subject site range in size from 0.13 to 0.11 containing dwellings that were constructed between 2000 and 2002.

The subject site currently contains one, 2700 square foot, single-family residential dwelling

Jacoby Road Subdivision

that was originally constructed in 1966. The site is situated centrally in the northern half of the subject site. The rest of the parcel contains forestland that was recently aggressively thinned. Sections of the parcel have been entirely cleared of trees, while other areas were left with a few individual or clumps of trees. Historical aerial imagery indicates that some of the thinning occurred between 5/2017 and 7/2017 with a majority of the trees harvested after 7/2017 and before 01/2018. The treed areas in the southern half of the parcel appear to have been planted based on their relatively uniform appearance in images taken prior to 2017. The oldest available historical image of the subject site indicates that the southern half of the site was relatively clear of large trees in 1995, and numerous small trees are visible across the southern half of the site in the next available image, taken in 2000. These two early images also depict the northern half of the site as thickly canopied. It appears that between 08/2003 and 08/2005, the trees from the northwestern quarter of the subject site were harvested. Since that image the northeastern quarter of the site appears to have only contained a thin scattering of individual trees. Additionally, some of the trees in the southern half of the site appear to have been removed and not replaced. The northeastern quarter of the site appears to have contained a moderately thick canopy with a mix of evergreen and deciduous trees since the earliest available image (1995).

On site observations indicate that there is flowing water in the north-descending swale-like topographic feature that transects the western half of the subject site. This north-descending depression was observed to contain water that actively flowed across the ground surface, through a culvert under the driveway and to an earthen damn that appears to encourage the surface water to transition to a shallow sub-surface flow. A rock-lined drainage ditch, with at least one small wooden bridge, was observed on the adjacent site. At the time of the site visit this ditch was observed to not contain flowing water, but the existence of the feature speaks to the presence of surface water flowing through the sites. This saturated lowland area includes the entirety of proposed lots 8 and 11 along with large portions of proposed lots 29, 30, 31 and 32.

The site contains a variable amount of slope. The southeastern corner of the subject site is relatively low relief, with slopes modeled by the 5-Ft DEM of Portland Oregon and adjacent areas as falling within the categories of less than 5% (blue) and 5-10% (green). Other areas of the parcel contain slopes of 10-15% (yellow) and 15-20% (orange), The steepest section of the parcel is in the northeastern quarter of the site, where the slopes are descending towards the northwest-flowing Tickle Creek. These slopes, when observed by RSS, appeared irregular. Much of this irregularly appeared to be associated with the scarring produced by the machines that removed the trees from the site. These gouges transecting the hillside appeared to obscure the natural shape of the slopes. The Google Earth DEM indicates that there are slopes of at least 40% on this northeastern quarter of the subject site. The excising conditions map with 1' contours and the proposed lots indicates that lot 20 and 19 have an average slope of 33% below the slope break. This slope regime is also found in the northeastern half of lot 21 and along the eastern edge of lot 26. The existing driveway is presumably constructed on fill where is crosses the swale in the western half of the site, and it contains relatively steep drop-offs on both it's northern and southern sides. The slope ascending from the swale to the existing house and lower relief portion of the parcel (southeastern quarter) is greater than 15%, with

Jacoby Road Subdivision



2.2 Regional Geology

Current geologic literature 1²2³4⁵5⁶7⁸9¹⁰ classifies the slopes underlying the subject site as ancient river rock of the Portland Basin. Various workers have further classified the ancient river rock as part of the Springwater Formation (Madin, 2004; Trimble, 1963) and as

¹ http://www.oregongeology.org/geologicmap/

² Ma, L., Madin, I.P., Duplantis, S., and Williams, K.J., 2012, Lidar-based surficial geologic map and database of the greater Portland, Oregon, area, Clackamas, Columbia, Marion, Multnomah, Washington, and Yamhill Counties, Oregon, and Clark County, Washington: Oregon Department of Geology and Mineral Industries, Open-File Report 0-2012-02, scale 1:8,000.

³ English, J.T., Coe, D.E., and Chappell, R.D., 2013, Channel migration hazard data and maps for the Sandy River, Multnomah and Clackamas Counties, Oregon: Oregon Department of Geology and Mineral Industries, Open-File Report O-2013-10, scale 1:12,000.

⁴ IMSBurns, W.J., Mickelson, K.A., and Duplantis, S., 2012, Landslide inventory maps of the Sandy quadrangle, Clackamas and Multnomah Couties, Oregon: Oregon Department of Geology and Mineral Industries, Interpretive Map Series 38, scale 1:8,000.

⁵ Burns, W.J., Mickelson, K.A., Jones, C.B., Pickner, S.G., Hughes, K.L.B., and Sleeter, Rachel, 2013, Landslide hazard and risk study of northwestern Clackamas County, Oregon: Oregon Department of Geology and Mineral Industries, Open-File Report O-2013-08, scale 1:8,000.

⁶ Schlicker, H.G., and Finlayson, C.T., 1979, Geology and geologic hazards of northwest Clackamas County: Oregon Department of Geology and Mineral Industries, Bulletin 99, scale 1:24,000.

Snyder, D.T., 2008, Estimated depth to ground water and configuration of the water table in the Portland, Oregon area: U.S. Geological Survey, Scientific Investigations Report SIR-2008-5059, scale 1:60,000. 8 Trimble, D.E., 1963, Geology of Portland, Oregon and adjacent areas: U.S. Geological Survey, Bulletin 1119, scale 1:62,500.

⁹ Treasher, R.C., 1942, Geologic map of the Portland area: Oregon Department of Geology and Mineral Industries, Quadrangle Map 9, scale 1:96,000.

¹⁰ Madin, I.P., 2004, Geologic mapping and database for the Portland area fault studies: Final report, Clackamas, Multnomah, and Washington Counties, Oregon: Oregon Department of Geology and Mineral Industries, Open-File Report O-04-02, scale 1:100,000.

Pliocene-Pleistocene Gravels (Schlicker & Finlayson, 1979). The site is tucked along the eastern edge of the Portland basin on slopes ascending towards the Oregon Cascade Range.

Geologic History

The subject site is generally situated within the forearc basin of the Cascadia subduction system between the Cascade Range (volcanic arc) and the Coastal Range (accretionary/subduction complex). The site is tucked along the eastern edge of the Portland Basin, which is one of several topographic and structural depressions that collectively constitute the Puget-Willamette forearc trough. This topographic and structural basin generally has low topographic relief. The basin formed due to tectonic compressional stress that both intimated the basin's formation and produced prolonged the enlargement of the basin. As the Portland Basin continued to subside during the late Miocene and Pliocene, it filled with continental fluvial and lacustrine sediments that were transported through the Cascade Range by the ancestral Columbia River as well as with locally derived detritus carried in by tributaries draining the surrounding highlands. These sediments were deposited primarily in fluvial environments, mostly by the rivers and streams that traversed the Portland basin as it was developing. This resulted in a thick accumulation of material preserving a complex record of deposition and erosion (aggradation and incision). The modern Columbia River has carved a channel through the current-day basin.

Much of the region was inundated by a series of massive floods at the end of the last ice age. . The floodwaters, which reached an elevation of 400 feet above sea level, soured many areas down to bedrock and buried others beneath thick layers of gravel, sand and silt that can be divided into a fine-grained and course-grained units. The subject site appears to be above the inundation elevation of these floodwaters and is not draped in a blanket of the associated sediments.

The ancient river rocks of Ma et al (2012) comprises the youngest bedrock in the Portland area, and it typically comprised of cemented and compacted sediments deposited by the ancestral Tualatin, Willamette, Clackamas, Sandy and Columbia Rivers. They include layers of sandstone, mudstone, siltstone, and conglomerate or cemented gravel. In the Tualatin Valley, there are almost 1,000 feet of siltstone, mudstone and sandstone mostly derived from older sedimentary rocks of the Coast Range. These rocks may have been laid down as long ago as 15 million years after the cessation of the Columbia River Basalt lava flows, as a recently as 2.5 million years when the Boring volcanic field became active.

The deposits on site are further classified as part of the Springwater Formation. This unit is described as a fluvial conglomerate, volcaniclastic sandstone, siltstone, and debris flows derived from the Cascade Range. The conglomerates within this unit typically consist of well-rounded pebbles, cobbles and boulders of basalt, andesite and dacite with rare exotic metamorphic and plutonic rocks. The matric material is typically comprised of sand and silt. It is generally massive and profoundly weathered producing a strongly varicolored material with reds, broths, grey-greens and oranges. When un-weathered, the conglomerate is typically grey to brown. Debris flows contain angular to rounded clasts of basalt, andesite and dacite lava, scoria and pumice in a matrix of clay, ash and sand. Sandstones range from fine to course grained. The siltstones and mudstones generally consists of quartzo-feldspahtic

Jacoby Road Subdivision

silt, ash and clay. The unit overlays the Troutdale Formation and some exposures may be a mixture of the two units and the contact appears to be conformable and gradual. Boring Lava commonly overlies or is interbedded with Springwater Formation rocks.



2.3 Field Exploration and Subsurface Conditions

2.3.1 Field Explorations

Seven (7) hand augur holes were excavated. The locations of the borings are shown on Figure 3 in Appendix A. A GIT (geologist in training) observed and logged the subsurface soils and they were reviewed and written up by a registered professional engineer. Logs detailing materials encountered are in Appendix B. The logs were created using the Unified Soil Classification and Visual Manual Procedure (ASTM-D 2488). Samples were transported to the laboratory ACS Testing of Tigard, Oregon for further classification in seal bags. Please see Appendix B for further laboratory results.

2.3.1 Subsurface Conditions

The soil conditions were stiff to very stiff SILT to a depth of 8 feet. There are some locations were the organics are up to 2.5ft thick. Borings had moisture contents ranging from 32.7% to 44%.

2.3.2 Groundwater

Groundwater was encountered in HA2 at 9inches.

Jacoby Road Subdivision

3.0 GEOTECHNICAL DESIGN RECOMMENDATIONS

3.1 Foundation Design

The building foundations may be installed on either engineered fill or firm native subgrade that is found at a depth of about 2 feet. This depth may be locally variable and should be confirmed by a geotechnical engineer or their representative at the time of construction.

Continuous wall and isolated spread footings should be at least 16 and 24 inches wide, respectively. The bottom of exterior footings should be at least 16 inches below the lowest adjacent exterior grade. The bottom of interior footings should be at least 12 inches below the base of the floor slab.

Footings placed on engineered fill or firm native sub-grade should be designed for an allowable bearing capacity of 2,000 *pounds per square foot* (**psf**). The recommended allowable bearing pressure can be doubled for short-term loads such as those resulting from wind or seismic forces.

Based on our analysis the total post-construction settlement is calculated to be less than 1 inch, with differential settlement of less than 0.5 inch over a 50-foot span for maximum column, perimeter footing loads of less than 100 kips and 6.0 kips per linear foot.

Lateral loads on footings can be resisted by passive earth pressure on the sides of the structures and by friction at the base of the footings. An allowable lateral bearing pressure of 100 *pounds per cubic foot* (**psf/f**) below grade may be used. Adjacent floor slabs, pavements or the upper 12-inch depth of adjacent, unpaved areas should not be considered when calculating passive resistance. An angle of internal friction of 28 degrees can be used.

If construction is undertaken during wet weather, we recommend a thin layer of compacted, crushed rock be placed over the footing sub-grades to help protect them from disturbance due to the elements and foot traffic.

3.2 Floor Slabs

Satisfactory sub-grade support for building floor slabs can be obtained from the native subgrade prepared in accordance with our recommendations presented below. A 6-inch-thick layer of imported granular material should be placed and compacted over the prepared subgrade. Imported granular material should be crushed rock or crushed gravel that is fairly well graded between coarse and fine, contains no deleterious materials, have a maximum particle size of 1 inch, have less than 5 percent by weight passing the U.S. Standard No. 200 Sieve, and meet OSSC 02630.10 – Dense Graded Aggregate 1"-0". The imported granular material should be placed in 6-inch-thick lifts and compacted to at least 95 percent of the maximum dry density as determined by American Society for Testing and Materials (ASTM) D 1557. A sub-grade modulus of 125 pounds per cubic inch (pci) may be used to design the floor slab.

Installation of a vapor barrier is required for all the houses built on this lot. It will reduce the potential for moisture transmission through, and efflorescence growth on, the floor slabs. Additionally, flooring manufacturers often require vapor barriers to protect flooring and flooring adhesives and will warrant their product only if a vapor barrier is installed according to their recommendations. The selection and design of an appropriate vapor barrier, if needed, should be based on discussions among members of the design team.

3.3 Seismic Design Criteria

The seismic design criteria for this project found herein is based on the USGS Earthquake Hazards Program. A summary of IBC 2012/2015 code section seismic design criterion below: using a Lat of 45.385063 and Long of -122.256340

Short Period	1 Second
Ss = 0.764g	S1 = 0.334 g
Sms = 0.913	Sm1 = 0.578
Sds = 0.608	Sd1=0.386
	$\label{eq:short Period} \begin{aligned} &Short Period\\ &Ss = 0.764g\\ &Sms = 0.913\\ &Sds = 0.608 \end{aligned}$

3.4 Slopes and Hazards

The Oregon HazVu: Statewide Geohazard Viewer11 and Metromap12 were reviewed on 22 January 2017 to investigated mapped geological hazards. This review indicates that the northeastern corner of the parcel overlaps with the Effective FEMA 100 year Floodplain as presented by HazVu where Tickle Creek transects the subject site. The expected earthquake-shaking hazard is classified as 'very strong' with no mapped earthquake liquefaction hazard classification. The nearest fault, mapped by and classified as active by DOGAMI, is a NE-SW oriented fault near Boring, Oregon roughly 6.8 miles northwest of the subject site. Additional faults are likely located closer to the subject site, but they are not mapped. There are no landslides mapped on or adjacent to the subject site. The site is classified as having a landslide susceptibility classification of 'low' (landsliding unlikely) in the southeastern corner of the parcel, 'moderate' (landsliding possible) across most of the parcel and 'high' (landsliding likely) in the northeastern quarter of the subject site. William et al (2013) indicates that the site has a low susceptibility to deep-seated landslides across the subject site. The William et al (2013) shallow-landslide susceptibility map is consistent with the Oregon HazVu map susceptibility values.

¹¹ http://www.oregongeology.org/hazvu/

¹² http://gis.oregonmetro.gov/metromap/

4.0 CONSTRUCTION RECOMMENDATIONS

4.1 Site Preparation

Demolition should include removal of existing improvements throughout the project site. Underground utility lines, vaults, basement walls or tanks should be removed or grouted full if left in place. I recommend that soil disturbed during grubbing operations be removed to firm, undisturbed sub-grade. The excavations should then be backfilled with compacted structural fill. On this site only disturb the area in which can be covered with rock during the day. The moisture sensitive SILT soil when exposed to wet weather becomes soft and yielding. See wet weather conditions below.

4.1.1 Proof Rolling

Following stripping and prior to placing aggregate base course, pavement the exposed sub-grade should be evaluated by proof rolling. The sub-grade should be proof rolled to identify soft, loose, or unsuitable areas. Please give 24 hour notice to observe the proof rolling. Soft or loose zones identified during the field evaluation should be compacted to an unyielding condition or be excavated and replaced with structural fill, as discussed in the *Structural Fill* section of this report.

4.1.2 Wet Weather Conditions

The near-surface soils will be difficult during or after extended wet periods when the moisture content of the surface soil is more than a few percentage points above optimum. Soils that have been disturbed during site preparation activities, or soft or loose zones identified during probing or proof rolling, should be removed and replaced with compacted structural fill. Track-mounted excavating equipment will be required during wet weather. The imported granular material should be placed in one lift over the prepared, undisturbed sub-grade and compacted using a smooth drum, non-vibratory roller. If construction is undertaken during the wet weather the builder may choose to cement treat the top 12" of sub-grade soil on the site. This will save time over the duration of the project due to the moisture sensitive SILT becoming soft and yielding which will require repeated over excavation and replacement with structural fill. The geotextile fabric can be eliminated if the sub-grade is cement treated.

4.2 Excavation

Subsurface conditions of accessible cleared areas of the project site show predominately SILT to the depth explored (10 feet). Excavations in the upper soils may be readily accomplished with conventional earthwork equipment with smooth faced bucket.

4.3 Structural Fills

Fills should be placed over sub-grade prepared in compliance with Section 4.1 of this report. Material used, as structural fill should be free of organic matter or other unsuitable materials and should meet specifications provided in OSSC, depending upon the application. A discussion of these materials is in the following sections.

Jacoby Road Subdivision

As there is a large fill on the site the following below figure shall be followed. It is a diagram of placing fill on a slope. The keyway at the bottom shall be at least 4ft deep and 10ft wide and shall be hard and non-yielding. Prior to placement of any fill RSS or there representative shall inspection this keyway. See below benching figure. Compaction testing is required every 1.5ft and a proof roll on all fill areas.



If there are springs into the fill slope they shall be directed into solid pipes and brought to the drainage swale at the south end of job.

4.3.1 Native

Laboratory testing indicates that the moisture content of the near-surface is greater near the optimum moisture content of the soil required for satisfactory compaction. This is depending on the weather conditions at the time of excavation. See section 4.3.2 for imported granular fill. ASTM T-99 and 95% compaction is required when using native soils.

4.3.2 Imported Granular Fill

The imported granular material must be reasonably well graded to between coarse and fine material and have less than 5% by weight passing the US Standard No.200 Sieve. Imported granular material should be placed in lifts 8 to 12 inches and be compacted to at least 92% of the maximum dry density, as determined by ASTM D 1157. Where imported granular material is placed over wet or soft soil sub-grades, we recommend that a geo-textile serve as a barrier between the sub-

Jacoby Road Subdivision

grade and imported granular material. Please supply the engineer with a 5gallon bucket of material 48hours prior to any compaction tests required.

4.3.3 Pavement Base Aggregate

Imported base aggregate for roads and parking lots should be clean, crushed rock or crushed gravel. The base aggregate should meet the gradation defined in OSSC 02630.10 – Dense Graded Aggregate 1 1/2"-0," with the exception that the aggregate should have less than 5% passing a US Standard No. 200 Sieve. The base aggregate should be compacted to at least 92% of the maximum dry density, as determined by ASTM D 1557. Please supply the engineer with a 5gallon bucket of material 48hours prior to any compaction tests required.

4.4 Drainage Considerations

The Contractor shall be made responsible for temporary drainage of surface water and groundwater as necessary to prevent standing water and/or erosion at the working surface. We recommend removing only the foliage necessary for construction to help minimize erosion. Slope the ground surface around the structures to create a minimum gradient of 2% away from the building foundations for a distance of at least 5 feet. Surface water should be directed away from all buildings into drainage swales or into a storm drainage system.

The swales that are placed on the roadway fills shall be lined and then connected into the storm water system.

5.0 CONSTRUCTION OBSERVATIONS

Satisfactory pavement and earthwork performance depends on the quality of construction. Sufficient monitoring of the activities of the contractor is a key part of determining that the work is completed in accordance with the construction drawings and specifications. I recommend that a geotechnical engineer observe general excavation, stripping, fill placement, and sub-grades in addition to base. Subsurface conditions observed during construction should be compared with those encountered during the subsurface explorations. Recognition of changed conditions requires experience. Therefore, qualified personnel should visit the site with sufficient frequency to detect whether subsurface conditions changes significantly from those anticipated.

6.0 LIMITATIONS

This report has been prepared for the exclusive use of the addressee, and their architects and engineers for aiding in the design and construction of the proposed development. It is the addressee's responsibility to provide this report to the appropriate design professionals, building officials, and contractors to ensure correct implementation of the recommendations.

The opinions, comments and conclusions presented in this report were based upon information derived from our literature review, field investigation, and laboratory testing. Conditions between, or beyond, our exploratory borings may vary from those encountered. Unanticipated soil conditions and seasonal soil moisture variations are commonly encountered and cannot be fully determined by merely taking soil samples or soil borings. Such variations may result in

Jacoby Road Subdivision
changes to our recommendations and may require that additional expenditures be made to attain a properly constructed project. Therefore, some contingency fund is recommended to accommodate such potential extra costs.

If there is a substantial lapse of time between the submission of this report and the start of work at the site; if conditions have changed due to natural causes or construction operations at, or adjacent to, the site; or, if the basic project scheme is significantly modified from that assumed, it is recommended this report be reviewed to determine the applicability of the conclusions and recommendations.

The work has been conducted in general conformance with the standard of care in the field of geotechnical engineering currently in practice in the Pacific Northwest for projects of this nature and magnitude. No warranty, express or implied, exists on the information presented in this report. By utilizing the design recommendations within this report, the addressee acknowledges and accepts the risks and limitations of development at the site, as outlined within the report.

Jacoby Road Subdivision

2/23/2018







Figure 2: Subject site location on the Clackamas County Assessor's Map



	0		$\langle \zeta \rangle$	
		APPENDIX B		-
		т. Т.		. 60
Jacoby R	oad Subdivision	15	2/23/201	8



1 of 3 1/15/18

Sample Date

Depth: 4'

Project Name: Jacoby Rd - Sandy Subdicision

Sample Number: HA#I

()

Moisture

Sample number	HA#1	HA#2	HA#3	HA#4	HA#5
1 Date and time in oven	1/16/18 2:30 PM	1/16/18 2 30 PM			
2 Date and time out of oven	1/18/18 10:30 AM	1/18/18 10 30 AM	1/18/18 10:30 AM	1/18/18 10:30 AM	1/18/18 10 30 AM
3 Depth (ft)	4	3	4	8	4
4 Tare No.	3	6	7	8	9
5 Tare Mass	234	233	230	233	231
6 Tare plus sample moist	784	1041	1022	983	1080
7 Tare plus sample dry	636	795	815	798	817
8 Mass of water (g)	148	246	207	185	263
9 Mass of soil (g)	402	562	585	565	586
10 Water Content (%)	36.82	43.77	35.38	32.74	44.88

Atterberg Limit Test

	Liquid Limit			Plastic Limit	
	1	2			2
I Tare No.	D#2.1	D#2.2	D#.23	R#1.1	R#1.2
2 Tare Mass (g)	39.59	39.46	39.3	39.89	38.94
3 Tare Plus Wet Soil (g)	78.34	70.15	71.22	50.81	50.34
4 Tare Plus Dry Soil (g)	65.66	59.72	60.15	47.91	47.33
5 Mass of Water (g)	12.68	10.43	11.07	2.9	3.01
6 Mass of Soil (g)	26.07	20.26	20.85	8.02	8.39
7 Water Content (g)	48.64	51.48	53.09	36.16	35.88
8 No. Blows	27	20	13		

























EXHIBIT H



Environmental Science & Assessment, LLC

MEMORANDUM

DATE:	May	30.	2018
	TALK I A	$\phi\phi_1$	2010

TO: Cory Knight

FROM: Jack Dalton

RE: Jacoby Road Subdivision – Wetland Determination (19124 SE Jacoby Road, Sandy, OR, 97055)

This memo provides findings of a wetlands determination conducted by Environmental Science & Assessment, LLC (ES&A) at 19124 SE Jacoby Road in Clackamas County. The 9.64-acre parcel (TL 24E24A02300) is located directly east of SE Jacoby Road (Attachment A). The parcel boundaries and the aerial photo were provided by All county Surveyors and Planners, Inc.

A 31-lot subdivision site is planned for the site. The project developer contracted ES&A to determine the presence of jurisdictional resources on site and whether the site development may potentially impact any of these resources.

METHODOLOGY

Two levels of investigation were conducted for the determination. The first involved a review of existing information for the parcel. The second involved an on-site determination by ES&A staff.

Reviewed background data included the following information:

- Aerial Photography (Google Earth 2017);
- LIDAR (Oregon Department of Geology and Mineral Industries [DOGAMI], 2018);
- National Wetland Inventory (US Fish and Wildlife Service [USFWS], 2018);
- Topography (Metro Data Resource Center's MetroMap, 2018);
- Web Soil Survey of Clackamas County, Oregon (Natural Resource Conservation Service [NRCS], 2018).
- ORMAP (Taxmap Explorer, 2018)

Site determination data was collected at two locations on May 24, 2018. The entire site was inspected and any potential wetlands or water features on or adjacent to the parcel were noted. The site slopes at most 10 percent towards the north and the west corner. Data plots were collected in the northern end of the site to determine whether hydrology and soils meet currently recognized wetland criteria.

107 SE Washington Street, #249 Portland, OR.: 97214 v 503.478.0424 www.esapdx.com

19124 SE Jacoby Road Wetland Determination Page 2

The wetland areas on the site were identified using methods consistent with the methodology provided in the *Regional Supplement to the Corps of Engineers Wetlands Delineation Manual: Western Mountains, Valleys, and Coast Region (Version 2.0)* (USACE, 2010). This methodology defines criteria for hydrology, soils, and vegetation that determine the jurisdictional status of wetlands.

The data plot locations and any potential wetland boundaries were recorded in the field with field measurements from know points and transferred to base topographical map provided by the client (Attachment A).

The following attachments are included in this memo:

Attachment A:	Figures
Attachment B:	Site photographs
Attachment C:	Wetland Determination Data Forms

EXISTING CONDITIONS

A shallow swale extends north through the western end of the site. A mapped segment of Tickle Creek is located in the northeast corner of the lot and was noted at the time of the site determination. The creek segment was mapped with the base topographic survey and this area will be protected in an open space tract as part of the proposed site development.

The 9.64-acre parcel at 19124 SE Jacoby Road is currently developed with one single family residence. Cascadia Village subdivision resides across SE Jacoby Road to the west and single-family dwellings reside to the north with rural residential area residing directly south and east of the site (Figure 1).

Site vegetation included a sparse canopy of western red cedar (*Thuja plicata*), big-leaf Maple (*Acer macrophyllum*) and western hemlock (*Tsuga heterophylla*), with red elderberry (*Sambucus racemosa*), and thimbleberry (*Rubus parviflorus*), Himalayan blackberry (*Rubus armeniacus*), western swordfern (*Polystichum munitum*), western brackenfern (*Pteridium aquilinium*), little western bittercress (*Cardamine oligosperma*), prickly lettuce (*Lactuca serriola*) and Siberian spring beauty (*Claytonia sibirica*) in the understory.

Topography on the lot slopes from the southern end of the site (1025 feet elevation) to the northern portion of the lot (984 feet elevation). Two wetland determination data plots (DP-1, DP-2) (Attachment C) were established along the shallow swale (Figure 3) (Attachment A).

Soils mapped on-site are Cazadero silty clay loams, 0 to 7 percent and 12 to 20 percent slopes (Map Units 15B and 15D), and Cottrell silty clay loam, 2 to 8 percent slopes (Map Unit 24B).

19124 SE Jacoby Road Wetland Determination Page 3

Tickle Creek flows from the east, entering the site briefly in the northwest corner and continues northwest to a culvert crossing at Jacoby Road approximately 500 feet north of the site (Metro 2018).

Sensitive Areas

The swale extends north to south through the site. A culvert at the existing driveway allows seasonal storm flow to pass into the north end of the site. The shallow swale stops approximately 80' south of the north property line. It was concluded that there was no channel morphology and no subsurface (wetland) hydrology and, therefore, no connection to off-site waterway north of the site.

The north end of site, being the lowest point of site, was investigated to determine the presence of any wetland features. The data collected indicate existing conditions in the swale do not meet wetland criteria. A summary of the data is provided below.

Data collected at north end of shallow swale (DP-1) included a variety of plants, soils, and minimal hydrology. In the surrounding area at the data plot there was nipplewort (*Lapsana communis*), velvet grass (*Holcus lanatus*), stickywilly (*Galium aparine*), annual bluegrass (*Poa annua*), fireweed (*Chamerion angustifolium*), Himalayan blackberry, big-leaf Maple, and western red cedar. There was evidence of seasonal flow lines at DP-1 with ponding in pockets, indicating ephemeral channel hydrology only. Soil profile was silt loam down to 13 inches and gravelly loam below to 20 inches. Soil did not meet hydric soil criteria. See Wetland Determination Data Forms in Attachment C.

Data collected north of the existing driveway (DP-2) included sparse western red cedar canopy with sparse weedy cover of nipplewort, velvet grass, bitter dock (*Rumex obtusifolius*) and Himalayan blackberry. There was evidence of seasonal storm ponding at the base of a berm (fill pile) across the swale which backed up the flow north. No defined channel was observed north of the berm and no wetland hydrology conditions extended north of the berm. Soil profile was silt loam down to 13 inches and gravelly loam below to 20 inches. Soil did not meet hydric soil criteria. See Wetland Determination Data Forms in Attachment C.

CONCLUSION

No evidence of intermittent or perennial flow was found on site. Evidence indicates seasonal ephemeral flow in the swale in west end of site and ponding at a culvert and a berm across channel. No flow or channel is present in the north end of site, thus there is no downstream connection to off-site waters. No wetland soils or hydrology is present, based on the data collected on site.

19124 SE Jacoby Road Wetland Determination Page 4

REFERENCES

Google Earth. Aerial Photography. Available at: <u>http://earth.google.com/</u>. Accessed May 2018.

Metro. 2017. Metro Data Resource Center's MetroMap GIS Application. Available online at: <u>http://gis.oregonmetro.gov/metromap/</u>. Accessed May 2018.

U.S. Army Corps of Engineers (USACE). 2010. Regional Supplement to the Corps of Engineers Wetland Delineation Manual: Western Mountains, Valleys, and Coast Region (Version 2.0), ed. J.S. Wakeley, R. W. Lichvar, and C.V. Noble. ERDC/EL TR-10-3. Vicksburg, MS: U.S. Army Engineer Research and Development Center.

Natural Resources Conservation Service, United States Department of Agriculture. Web Soil Survey for Clackamas County, Oregon. Available online at: <u>http://websoilsurvey.nrcs.usda.gov/</u>. Accessed May 2018.

U. S. Fish and Wildlife Service. 2009. National Wetlands Inventory website. U.S. Department of the Interior, Fish and Wildlife Service, Washington, D.C. Available online at: <u>http://www.fws.gov/wetlands/</u>. Accessed May 2018.

















Photo 1: View north of site - site slopes to the north.

Ó



Photo 2: View west of driveway in middle of site.



Photo 3: View south from driveway - shallow ephemeral flow channel.



0

Photo 4: View south of DP-1 north of driveway at the downstream end.



Photo 5: View north of DP-2 at north end of site.



Photo 6: View NE at the north end of site - No defined channel exiting site.



$\label{eq:wetland} \textbf{WETLAND DETERMINATION DATA FORM-Western Mountains, Valleys, and Coast Region$

Project/Site: Jacoby Subdivision	City/County:	Sandy/Clackam	as Sam	pling Date:	05/24/	18	
Applicant/Owner:		State: OR	Sampling Point:	DP-1			
Investigator(s): Jack Dalton	Section, T	ownship, Range:	24, 2S, 4E	12 6.3	1000		Contraction and
Landform (hillstope, terrace, etc.):	Lo	cal relief (concave	, convex, none):	concave		Slope (%):	7.5
Subregion (LRR): A-NW Forests & Coasts	Lat: 45.385	i Long:	-122.256	Datum:	N/A		
Soil Map Unit Name: Cazadero silty clay loan	, Cottrell silty cl	ay loam	NWI class	sification:	N/A		
Are climatic / hydrologic conditions on the site ty	oical for this time	e of year? Yes	X No (If r	10, explain i	n Remark	(S.)	
Are Vegetation, Soil, or Hydrok	igy <u>sign</u> if	icantly disturbed?	Are "Normal Ci	ircumstance	s" preser	t? Yes X	No
Are Vegetation , Soil , or Hydrold	gy 🔜 natur	ally problematic?	(If neede	ed, explain a	iny answe	ers in Remark	s.)
SUMMARY OF FINDINGS – Attach s	te map show	wing sampling	g point locatio	ons, trans	ects, i	mportant f	eatures, etc.
Hydrophytic Vegetation Present? Yes X Hydric Soil Present? Yes Yes Wetland Hydrology Present? Yes	No X No X	Is the Sampled	Area within a We	atland?	Yes	No	<u> </u>
Remarks: Lacks hydric soil conditions. Hydrolog	y evidence of e	phemeral flow only	1.				

VEGETATION – Use scientific names of plants.

ree Stratum (Plot size: 20'x30')	Absolute <u>% Cover</u>	Dominant Species?	Indicator Status	Number of Dominant Species	
. Thuja plicata	20	Y	FAC	That Are OBL, FACW, or FAC: 3	(A)
Acer macrophyllum	10	Y	FACU	Total Number of Dominant Species Across All Strata: 5	(B)
·				Percent of Dominant Species That Are OBL, FACW, or FAC: 60	(A/B)
Parling/Chash Chathan (District)	30	= Total Cove	er	Prevalence Index worksheet:	
Public ormanizaus	15	V	EAC	Total % Cover of: Multiply by:	
Rubus armeniacus	13		FAG	OBL species v1=	
·				FAC vv species X Z =	
		-		FAC species X 3 =	
				FACU species x 4 =	
	4.5	n Total Cour		UPL species x 5 =	
Jorh Stratum (Plat size: r = 5'	10		51	Column Totals: (A)	(B)
(Plut size. 1=5)	40	v	EACU	Prevalence Index = B/A =	
	40	T	FACO		
Holcus lanatus	20	T	FAC	Hydronhytic Venetation Indicators:	
Gailum apanne	10	N	PAGU	Denid Test for Undershulin Ver	station
Chamenon angustitolium	- 10	N	TAC .	T - Rapid Test for Hydrophylic Veg	letation
5. <u>Poa annua</u>		IN	FAC	2 - Dominance Test is >50%	
				3 - Prevalence Index Is \$3.0	
		Cost in the		4 - Morphological Adaptations' (Pr	ovide supportii sheet)
)		the second s		5 - Wetland Non-Vascular Plants ¹	
				Problematic Hydrophytic Vegetatic	n ¹ (Explain)
		7-1-1-0		Indicators of hydric soil and wetland l be present upless disturbed or proble	nydrology musi matic
Noody Vine Stratum (Plot size: r=15)	90	_ = 10(a) Cov	er		
	0	= Total Cov	er	Hydrophytic Vegetation	
% Bare Ground in Herb Stratum 10	_			Present? Yes X No	
Remarks: Weedy community. Tree cover sparse al	ong swale.			I <u> </u>	
	Ť				

Depth (inches)	cription: (Describe	to the dept	h needed to docum	ent the inc	dicator or co	nfirm the ab	Sampling Point: sence of indicators.)	DP - 1
(interred)	Matrix Color (moist)		Color (moist)	Redox Fea	tures		Tautura	D =
0-4	7.5 YR 2.5/2	100			_туре			<u>Remarks</u>
4-13	7.5 YR 3/2	99	7.5 YR 3/3	1			Silt loom	
13-15	7.5 YR 3/2	95	7.5 YR 4/3			 PI	Gravelly loam	
15-20	7.5 YR 3/2	95	7.5 YR 5/6			 PI	Gravelly toam	
							Graveny Idam	
		<u> </u>						
							·	
¹ Type: C=C	concentration, D=Dep	letion, RM=	Reduced Matrix, CS=	Covered o	or Coated Sa	nd Grains.	² Location: PL=Pore L	ining, M=Matrix.
Hydric Soi	I Indicators: (Applie	able to all	LRRs, unless other	wise note	d.)	Indic	ators for Problematic	Hydric Soils ³ :
Histoso	bl (A1)	_	_ Sandy Redox (S5	i)		2	ст Muck (A10)	,
Histic E Black H	Epipedon (A2) Histic (A3)		Stripped Matrix (S Loamy Mucky Mic	56) Secal (E1) (evcent MI P	A 1) - R	ed Parent Material (TF	2)
Hydrog	en Sulfide (A4)	_	Loamy Gleyed M	atrix (F2)	аксарт мск	~" <u> </u>	ther (Explain in Rema	rks)
Deplete Thick D	ed Below Dark Surfac Jark Surface (A12)	e (A11)	 Depleted Matrix (Bedox Dark Surfa 	F3) 3ce (F6)		3,	ndland	
Sandy	Mucky Mineral (S1)	_	Depleted Dark Su	Inface (F7)		- I W	etland hydrology must	be present.
Sandy	Gleyed Matrix (S4)		Redox Depressio	ns (F8)	,	u	nless disturbed or prot	lematic
estrictive La	ayer (if present):							
Туре:					Hydric Sol	I Present?	Yes	No X
Surface W High Wate Saturation Water Mar	ators (minimum of one ater (A1) r Table (A2) (A3) ks (B1) Deposits (B2)	<u>e required; c</u>	heck all that apply) Water-Stainer MLRA 1, 2, 4, Salt Crust (B1 Aquatic Invert Hydrogen Sul Oxidized Rhiz Roots (C3) Presence of R	I Leaves (I A, and 4B) 1) ebrates (B fide Odor (ospheres a Reduced Iro	39) (except) 13) C1) along Living on (C4)	Second Wa 4A, Dra Dry Sat Gee Sha	ary Indicators (2 or mo ler-Stained Leaves (BS and 48) inage Patterns (B10) -Season Water Table (uration Visible on Aeria pmorphic Position (D2) illow Aquitard (D3)	ore required) (MLRA 1, 2, (C2) al Imagery (C9)
Sediment I Drift Depos Algal Mat o Iron Depos Surface So	or Crust (B4) sits (B5) bil Cracks (B6)		Soils (C6) Stunted or Str (LRR A)	eduction in ressed Plar	n Tilled nts (D1) ks)	FAC	sed Ant Mounds (D6) (LRR A)
Sediment I Drift Depos Algal Mat o Iron Depos Surface So Inundation Sparsely V	or Crust (B4) bits (B5) bit Cracks (B6) Visible on Aerial Ima regetated Concave So attons:	gery (B7) urface (B8)	Soils (C6) Stunted or Str (LRR A) Other (Explain	eduction in essed Plar n in Remar	n Tilled nts (D1) ks)	FA0 Rai Fro	sed Ant Mounds (D6) (st-Heave Hummocks (LRR A) D7)
Sediment I Drift Depos Algal Mat o Iron Depos Surface So Inundation Sparsely V eld Observa urface Wated atter Table P aturation Pre actudes capil	sits (B3) or Crust (B4) oil Cracks (B6) Visible on Aerial Ima /egetated Concave So ations: r Present? Yes esent? Yes esent? Ilary fringe) Yes	gery (B7) urface (B8) No No No	X Depth (inches): X Depth (inches):	eduction in ressed Plar	nts (D1) ks)	FAC	sed Ant Mounds (D6) (st-Heave Hummocks (gy Present? Yes	LRR A) D7) NoX
Sediment I Drift Depos Algal Mat o Iron Depos Surface So Inundation Sparsely V eld Observa riface Water ater Table F ituration Pre- cludes capil cribe Recor	or Crust (B4) or Crust (B4) oil Cracks (B6) Visible on Aerial Ima regetated Concave So attons: r Present? Yes Present? Yes esent? Illary fringe) Yes ded Data (stream gat	gery (B7) urface (B8) No No uge, monitor	X Depth (inches): X Depth (inches): Depth (inches):	eduction ir essed Plar n in Remar	nts (D1) ks) wet	FAC Rai Fro. tland Hydrolo), if available:	sed Ant Mounds (D6) (st-Heave Hummocks (Dgy Present? Yes	LRR A) D7) NoX

$\langle \zeta \rangle$				0
WETLAND DETERMINATIO	ON DATA F	ORM – We	estern Mou	intains, Valleys, and Coast Region
oject/Site: Jacoby Subdivision C	ity/County:	Sandy/Clack	amas	Sampling Date: 05/24/18
plicant/Owner:		State: OR	_ Sampling	Point: DP-2
vestigator(s): Jack Dalton	Section, Tov	wnship, Range	e: <u>24, 2S, 4</u>	
horden (LBB): A NAL Foresta & Capata	LOCa	u relier (conca	ive, convex, n	Dotum: N/A
il Man Lloit Name: Cazadero silty clay loam (at. 40.000	Lung.	-122.200	All classification: N/A
e climatic / hydrologic conditions on the site typic	al for this time	of year? Yes	X No	(If no, explain in Remarks.)
Vegetation . Soil . or Hydrology	signific	antly disturbed	d? Are "No	ormal Circumstances" present? Yes X No
e Vegetation, Soil, or Hydrology	natural	ly problematic	? (If needed, explain any answers in Remarks.)
JMMARY OF FINDINGS – Attach site drophytic Vegetation Present? Yes Yes X rdric Soil Present? Yes etland Hydrology Present? Yes	map show	ing sampli Is the Sampl	ing point li led Area with	in a Wetland? Yes NoX
marks:				
EGETATION – Use scientific names of	of plants.			
	Ahsolute	Dominant	Indicator	Dominance Test worksheet:
ee Stratum (Plot size: 20'x30')	<u>% Cover</u>	Species?	Status	Number of Dominant Species
Thuja plicata	10	Y	FAC	That Are OBL, FACW, or FAC: (A)
				Total Number of Dominant
				Species Across All Strata: 5 (B)
			1	That Are OBL, FACW, or FAC: 80 (A/B)
Charles (Charles Charles)	10	= Total Cov	er	Prevalence Index worksheet:
Bubus armaniacus	15	v	EAC	Total % Cover of: Multiply by:
			17.0	OBL species x 1 =
				FACW species x 2 =
			1	FAC species x 3 =
				FACU species x 4 =
				UPL species x 5 =
(and the second s	15	_ = Total Cov	er	Column Totals: (A) (B)
erb Stratum (Plot size: r = 5')			EAOU	Bravelenes Index a B/A a
Lapsana communis	10	Y	FACU	Frevalence index = B/A =
Rumex obtusifolius	10	Y	FAC	Hydrophytic Vegetation Indicators:
rumen obtailoilas			170	1 - Rapid Test for Hydrophytic Vegetation
				2 - Dominance Test is >50%
				3 - Prevalence Index is ≤3.0 ³
				4 - Morphological Adaptations' (Provide supporting
		8		data in Remarks or on a separate sheet)
				5 - Wetland Non-Vascular Plants
)				Problematic Hydrophytic Vegetation" (Explain)
l			V	Indicators of hydric soil and wetland hydrology must
In adu Mine Stratum (Distaire)	30	= Total Cov	/er	be present, unless disturbed or problematic.
(Plot size: r=15))				
		= Total Cov	/er	Hydrophytic
· · · ·				Vegetation
Bare Ground in Herb Stratum 70		-		Present? Yes X No

Western Mountains, Valleys, and Coast - Version 2.0

Profile Des	cription: (Describe)	to the dept	h needed to docum	ent the inc	licator or co	onfirm the al	Sampling Poir	11: DP - 2
Depth (inches)	Matrix Color (moist)		Color (moist)	Redox Fea %	tures Type		Tevtura	Pemarks
0-6	7.5 YR 3/2	100					Silt loam	<u> </u>
6-10	7.5 YR 3/2	99	7.5 YR 3/4	1	C	M	Silt loam	
10-13	7.5 YR 3/2	98	7.5 YR 3/4	2		M	Silt loam	
13-16	7.5 YR 3/2	95	7.5 YR 4/6	5	с	PL	Gravelly loam	
16-20	7.5 YR 2.5/2	95	7.5 YR 4/4		C	PL	Gravelly loiam	
Туре: С=С	Concentration, D=Dep	letion, RM=	Reduced Matrix, CS	=Covered of	or Coated Sa	nd Grains.	² Location: PL=Pore	Lining, M=Matrix.
Black F Hydrog Deplete Thick D Sandy Sandy	itstic (A3) en Sulfide (A4) ed Below Dark Surfac Dark Surface (A12) Mucky Mineral (S1) Gleyed Matrix (S4)		Loamy Mucky Mi Loamy Gleyed M Depleted Matrix (Redox Dark Surf Depleted Dark S Redox Depression	ineral (F1) latrix (F2) (F3) ace (F6) urface (F7) ons (F8)	except MLR		/ery Shallow Dark Su Dther (Explain in Rem Indicators of hydroph vetland hydrology mu inless disturbed or pr	rface (TF12) iarks) ytic vegetation and st be present, oblematic
Depth (ind	nes):							
Depth (inc narks: Soil i	s well-drained.				I			
Depth (inc narks: Soil i DROLOC atland Hyd imary Indica	s well-drained. SY rology Indicators: ators (minimum of one	e required;	check all that apply)		I	Secon	dary Indicators (2 or r	more required)
Depth (inc narks: Soil i DROLOO atland Hyd imary Indica Surface W High Wate Saturation Water Mar Sediment Drift Depo Algal Mat Iron Depo Surface S Inundatior Sparsely V	s well-drained. s well-drained. s well-drained. status (minimum of one (ater (A1) er Table (A2) (A3) (A) (A) (A) (A) (A) (A) (A) (A	e required; gery (B7) urface (B8)	check all that apply) Water-Staine MLRA 1, 2, 4 Salt Crust (B Aquatic Inver Hydrogen Su Oxidized Rhi Roots (C3) Presence of Recent Iron F Soils (C6) Stunted or St (LRR A) Other (Explai	d Leaves (i JA, and 4B 11) tebrates (B ilfide Odor i zospheres Reduced In Reduced In Reduced In Reduced Pla in in Remai	I B9) (except) 13) c(1) along Living on (C4) n Tilled nts (D1) ks)	<u>Secon</u> 	dary Indicators (2 or rater-Stained Leaves (, and 4B) ainage Patterns (B10) y-Season Water Table turation Visible on Ae comorphic Position (D allow Aquitard (D3) allow Aquitard (D3) C-Neutral Test (D5) ised Ant Mounds (D6 ost-Heave Hummocks	nore required) B9) (MLRA 1, 2,) e (C2) rial Imagery (C9) 2)) (LRR A) 5 (D7)
Depth (inc narks: Soil i DROLOO etland Hyd imary Indica Surface W High Wate Saturation Water Man Sediment Drift Depo Algal Mat Iron Depo: Surface Si Inundation Sparsely V eld Observ Inface Wate ater Table fa uration Pro-	swell-drained. swell-drained. swell-drained. swell-drained. siswell-drained. siswell-drained. sitalors: (A3) rable (A2) (A3) rable (A2) (A3) rable (A2) (A3) rable (A2) (A3) rable (A2) (A3) sits (B1) Deposits (B2) sits (B3) or Crust (B4) sits (B5) oil Cracks (B6) visible on Aerial Ima /egetated Concave S rations: r Present? Yes sent? illary fringe) Yes	e required; i ligery (B7) urface (B8)	check all that apply) Water-Staine MLRA 1, 2, 4 Salt Crust (B Aquatic Inver Hydrogen Su Oxidized Rhi Roots (C3) Presence of I Recent Iron F Soils (C6) Stunted or St (LRR A) Other (Explain X Depth (inches) X Depth (inches)	d Leaves (IA, and 4B 11) tebrates (B Ifide Odor i zospheres Reduced In Reduction is tressed Pla in in Reman	B9) (except) 13) (C1) along Living on (C4) n Tilled nts (D1) ks) We	Secon Wi 44 Dr Dr Sa Sa Sa Sa FA FA FA	dary Indicators (2 or n ater-Stained Leaves (, and 48) ainage Patterns (B10) ->Season Water Tabh turation Visible on Ae comorphic Position (D allow Aquitard (D3) C-Neutral Test (D5) ised Ant Mounds (D6) ost-Heave Hummocks	more required) B9) (MLRA 1, 2,) e (C2) trial Imagery (C9) 2)) (LRR A) s (D7) es No
Depth (inc narks: Soil i DROLOO ettand Hyd imary Indica Surface W High Wate Saturation Water Man Sediment Drift Depo Algal Mat Iron Depo: Surface Sa Inundation Sparsely V etd Observ inface Wate ater Table F aturation Pro- cludes cap	s well-drained. s well	e required; i igery (B7) urface (B8) No No uge, monito	check all that apply) Water-Staine MLRA 1, 2, 4 Salt Crust (B Aquatic Inver Hydrogen Su Oxidized Rhi Roots (C3) Presence of I Recent Iron F Soils (C6) Stunted or SI (LRR A) Other (Explain X Depth (inches) X Depth (inches) x Depth (inches) x Depth (inches)	d Leaves (i IA, and 4B 11) tebrates (B Ifide Odor d zospheres Reduced In Reduction is tressed Pla in in Reman	I B9) (except) 13) (C1) along Living on (C4) n Tilled nts (D1) its) We s inspections	Secon Wi 44 Dr Dr Sa 	dary Indicators (2 or r ater-Stained Leaves (, and 4B) ainage Patterns (B10) y-Season Water Tabh turation Visible on Ae comorphic Position (D allow Aquitard (D3) C-Neutral Test (D5) ised Ant Mounds (D6) ost-Heave Hummocks logy Present? Ye	nore required) B9) (MLRA 1, 2, e (C2) riral Imagery (C9) 2)) (LRR A) 5 (D7) es NoX
3								
---	--	---	--	--	--	--		
С								
EASEMENT								
6 l Between								
Cory Alan Knight								
DIS And	Clackamas County Official Records Sherry Hall, County Clerk 2008-012774							
Ian Bird and Kelly Bird	\$81.00							
After recording, return to:	01190129200800127740070073 02/25/2008 10:44:45 AM							
Ian Bird and Kelly Bird	\$35.00 \$16.00 \$10.00 \$20.00							
0. 1234 Spinnaker Way 2. Sugar Land, TX 77478								
ູ								
X								
THIS AGREEMENT made a Cory Alan Knight, hereinafter called hereinafter called the second party, V WHEREAS: The first party i Clackamas County, State of Oregon,	THIS AGREEMENT made and entered into on February <u>D</u> , 2008 by and between, Cory Alan Knight, hereinafter called the first party, Ian Bird and Kelly Bird, Husband and Wife, hereinafter called the second party, WITNESSETH: WHEREAS: The first party is the record owner of the following described real property in							
Exhibit "A" attached l	hereto and made a part hereof							
and has the unrestricted right to gra estate; and the second party is the re county and state, to-wit:	nt the easement hereinafter described relative to the real cord owner of the following described real property in that							
Exhibit "B" attached l	nereto and made a part hereof							
NOW, THEREFORE, in view party to the first party paid, the rece	v of the premises and in consideration of \$0.00 by the second ipt of which is acknowledged by the first party, it is agreed:							
The first party hereby grants, for ingress, egress and access, over th	, assigns and sets over to the second party a private easement he following described portion of first party's property, to wit:							
SEE ATI	ACHED EXHIBITS "C" and "C-1"							
The second party shall have a (including the right from time to tim	Il rights of ingress and egress to and from the real estate e, except as hereinafter provided, to cut, trim and remove							
		-						

trees, brush, overhanging branches and other obstructions) necessary for the second party's use, enjoyment, operation and maintenance of the easement hereby granted and all rights and privileges incident thereto.

The second party agrees to save and hold the first party harmless from any and all claims of third parties arising from the second party's use of the rights herein granted.

The period of this easement shall be perpetual, always subject, however, to the following specific conditions, restrictions and considerations:

NONE

During the existence of this easement, maintenance of the easement and costs of repair of the easement, if damaged by natural disasters or other events for which all holders of an interest in the easement are blameless, shall be the responsibility of both parties but only to the extent that the easement is used by both parties. If the easement is used exclusively by the second party, then the responsibility will be solely the second parties. The first party shall hold the second party harmless for such.

During the existence of this easement, holders of an interest in the easement who are responsible for damage to the easement because of negligence or abnormal use shall repair the damage at their sole expense.

In the event suit or action be filed by either of the parties hereto, to enforce any of the terms of this agreement, the parties hereto agree that the prevailing party shall be entitled to reasonable attorney's fees, to be set by the court, including any appeal thereon

This agreement shall bind and inure to the benefit of, as the circumstances may require, not only the parties hereto but also their respective heirs, executors, administrators, assigns, and successors in interest.

In construing this agreement, where the context so requires, the singular includes the plural and all grammatical changes shall be made so that this agreement shall apply equally to individuals and to corporations. If the undersigned is a corporation, it has caused its name to be signed and its seal, if any, affixed by an officer or other person duly authorized to do so by its board of directors.

IN WITNESS WHEREOF, the parties have hereunto set their hands in duplicate on the day and year first written above.

Corv Alan Knight

SECOND PA

Ian Ri

Kelly Rird

STATE OF OREGON) County of <u>Clackanas</u>)

This instrument was acknowledged before me on February 20, 2008 by

Cory Alan Knight

bodington in Notary Public for Oregon



STATE OF TEXAS

County of

This instrument was acknowledged before me on February ______2008 by

))

Ian Bird and Kelly Bird

LJSA DOMINGUEZ Notary Public STATE OF TEXAS My Comm, Exp. 08-04-11



3

Exhibit "A"

Part of the Southwest one-quarter of the Northeast one-quarter of Section 24, Township 2 South, Range 4 East of the Willamette Meridian, in the County of Clackamas and State of Oregon, described as follows:

Beginning at an iron pipe at the Northeast corner of a tract of land conveyed to Millard D. Wilder and Virginia Wilder, his wife, by deed recorded September 13, 1944, in Deed Book 331, Page 436; thence East 660.00 feet along the North line of the Southeast one-quarter of the Southwest one-quarter of the Northeast one-quarter of said Section 24; thence North 660.00 feet; thence West 660.00 feet along the South line of that particular tract of land conveyed to George E. Strope, et ux, by deed recorded November 20, 1944, in Deed Book 335,

Page 105; thence South 660.00 feet to the place of baginning. EXCEPTING THEREFROM the West 20 feet used for road purposes.

Exhibit "B"

A tract of land in the Northeast one-quarter of Section 24, Township 2 South, Range 4 East, of the Willamette Meridian, in the County of Clackamas and State of Oregon, described as follows:

Beginning at a point in the East line of the West one-half of the Southeast one-quarter of the Northeast one-quarter that is South 8.25 feet from the Southwest corner of the Northeast one-quarter of the Southeast one-quarter of the Northeast one-quarter of the Southwest one-quarter of the Northeast one-quarter of the Southwest one-quarter of the Northeast one-quarter of the Northeast one-quarter of the Southwest one-quarter of the Northeast one-quarter of said Section 24, a distance of 660.00 feet, more or less, to the Southeast corner thereof; thence West along the South line of said division 640.00 feet, more or less, to the East line of Dubarko Road (County Road No. 1836); thence South along the Easterly line of said road 8.25 feet to a point; thence East parallel with the North line of said division and the Easterly extension thereof 1,300.00 feet, more or less, to the point of beginning.

TOGETHER WITH and subject to a non-exclusive easement to be used in common with others for ingress, egress and all utility purposes over a tract of land 16.50 feet in width, the North line of which is described as follows:

Beginning at the intersection of the East line of Dubarko Road (County Road No. 1836) with the North line of the Southeast one-quarter of the Southwest one-quarter of said Section 24; thence East along the North line of said division and the Easterly extension thereof to a point that is Easterly 25. 00 feet from the East line of said division.

Exhibit "C"

6 0

LEGAL DESCRIPTION FOR EASEMENT

A TRACT OF LAND IN THE NORTHEAST QUARTER OF SECTION 24, TOWNSHIP 2 SOUTH, RANGE 4 EAST OF THE WILLAMETTE MERIDIAN, CLACKAMAS COUNTY OREGON, FURTHER DESCRIBED AS FOLLOWS:

BEGINNING AT THE SOUTHWEST CORNER OF THE LAND DESCRIBED IN DOCUMENT NO. 2004-111676 CLACKAMAS COUNTY DEED RECORDS, SAID POINT BEING ON THE EAST RIGHT OF WAY LINE OF JACOBY ROAD, THENCE NORTH 00°51'44" EAST, ALONG SAID RIGHT OF WAY LINE, A DISTANCE OF 3.50 FEET; THENCE SOUTH 88°37'40" EAST, A DISTANCE OF 239.43 FEET; THENCE NORTH 72°56'14" EAST, A DISTANCE OF 31.62 FEET; THENCE SOUTH 88°37'40" EAST, A DISTANCE OF 30.00 FEET; THENCE SOUTH 70°11'33" EAST, A DISTANCE OF 31.62 FEET; THENCE SOUTH 88°37'40" EAST, A DISTANCE OF 30.757 FEET; THENCE SOUTH 00°53'48" WEST, A DISTANCE OF 3.50 FEET; THENCE NORTH 88°37'40" WEST, A DISTANCE OF 637.00 FEET TO THE POINT OF BEGINNING.



()





A tract of the West of the Dorshowly Charter of Great of Great of Great of the West of Colors, and the second of Great of the Second of Colors, and the Second of Colors, and the Second of the Second of Colors, and the Second of the Second of Colors, and the Second of the Second of the Second of Colors, and the Second of the Second of Colors, and the Second of Colo EXHIOTA .V. See EXHIBIT A Attached IF SACE INSUFFICIENT, CONTINUE DES To Have and to Hold the same such the said grantee and grantee's heirs, successors and assigns forever. And said granter haraby covenants to and with said grantee and grantee's heirs, successors and assigns, that granter is lawfully setted in its simple of the above granted prantises, itse irons all encumbrances \$200\$; 4.1 those of record. ... grantor will warrant and lorever datend the above granted premises and every part and parcel thereol against the law-lal claims and demands of all persons whomsoever, except those claiming under the above described encumbranose. The true and estual consideration paid for this transfer, stated in terms of dollars, is \$19,000,00..., particular, the adual consideration consists of or includes other property or value given or promised which is is what x pavil of Put May 6 19.74 -OTANInigest act and deed 3 1 12 (OFFICIAL STALT 18 OF U thats (), If and L Ex-Officie Micro Cart Dargon, fan certify that certify that certify that WARRANTY DEED DEED PIETI ... 70 turned of bern REVEL 74 12923 ¹ Groups D, Puppez, 4 Revealer of Communics of the Common of the Common of the Common the Vicible Intrometry of Und recorded in the record Und recorded in the record 1970, MAX 16, PM 1 STATE OF ORECON, County of Contained, AFTER RECORDIN Arthur D. Revel Rt. 3, Box 32 Sandy, Ore. 97055 No. 23 UNTEL A CHARGE IS REQUESTED MAIL le FUGURE TAX STATEMENTS TO _=. By Deputy AATI 20 16-21-2 1. 140 Ú. (•)

EXHIBIT J

Tyler Henderson

To: Subject: Ray Moore RE: 17-199 - Fw: Jacoby Subdivision

From: Mike Walker Sent: Friday, January 5, 2018 1:35 PM To: Ray Moore Cc: Thomas Fisher ; Kelly O'Neill Subject: Jacoby Subdivision

Ray,

Attached please find the pre-design report for the pump station and a capacity analysis of the gravity system downstream of the force main discharge.

We did an analysis of the existing dwelling units in the contributing basin. There are 284 existing dwelling units and assuming build-out of vacant or under-developed properties there will be between 360 and 400 dwelling units at build-out of the area identified in the pre-design report.

Since the pump station, wet well and force main were designed for a total of 525 dwelling units it does not appear that the existing pump station will need to be expanded to accommodate your client's development. However you should address this in the utilities section of your land use submittal narrative.

Please let me know if you have any questions or need more information.

••

Mike Walker

Director of Public Works

City of Sandy

39250 Pioneer Blvd.

Sandy, OR 97055

503-489-2162 V

503-668-8714 F

www.ci.sandy.or.us

EXHIBIT K

19124 Jacoby Rd. Sandy 8" and larger

Action RPZ Tag Species Diameter Remarks 8 topped for high voltage wires; hedgerow 8 1 Douglas fir 12 2 western redcedar 19 viable; hedgerow; pruned for high votage wires 8 3 cascara buckthorn 9 topped for high voltage wires; hedgerow 20 30 viable; hedgerow; pruned for high votage wires 4 Douglas fir 11 topped for high voltage wires; hedgerow 8 5 Douglas fir 6 Douglas fir 5 topped for high voltage wires; hedgerow 8 7 Douglas fir 5 topped for high voltage wires; hedgerow 8 14 8 western redcedar 22 topped for high voltage wires; hedgerow 8 9 Douglas fir 8 topped for high voltage wires; hedgerow 10 Douglas fir 11 topped for high voltage wires; hedgerow 8 14 11 western redcedar 22 topped for high voltage wires; hedgerow 12 Douglas fir 10 topped for high voltage wires; hedgerow 8 8 13 Douglas fir 12 topped for high voltage wires; hedgerow 7 topped for high voltage wires; hedgerow 8 14 Douglas fir 15 Douglas fir 8 topped for high voltage wires; hedgerow 8 16 Douglas fir 6 topped for high voltage wires; hedgerow 8 8 17 Douglas fir 5 topped for high voltage wires; hedgerow 18 Douglas fir 9 topped for high voltage wires; hedgerow 8 12 19 Douglas fir 19 topped for high voltage wires; hedgerow 12 20 Douglas fir 19 topped for high voltage wires; hedgerow 8 21 Douglas fir 8 topped for high voltage wires; hedgerow 22 Douglas fir 9 topped for high voltage wires; hedgerow 8 23 Douglas fir 7 topped for high voltage wires; hedgerow 8 8 24 Douglas fir 7 topped for high voltage wires; hedgerow

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

25	western redcedar	26	topped for high voltage wires; broken top	16
26	Douglas fir	11	topped for high voltage wires; hedgerow	8
27	Douglas fir	6	topped for high voltage wires; hedgerow	8
28	Douglas fir	9	topped for high voltage wires; hedgerow	8
29	Douglas fir	6	topped for high voltage wires; hedgerow; dead	0
30	Douglas fir	5	topped for high voltage wires; hedgerow	8
31	Douglas fir	6	topped for high voltage wires; hedgerow	8
32	Douglas fir	12	topped for high voltage wires; hedgerow	8
33	Douglas fir	9	viable; hedgerow; pruned for high votage wires; windthrow potential	8
34	Douglas fir	9	viable; grown in grove	8
35	Douglas fir	6	viable; grown in grove	8
36	Douglas fir	9	viable; grown in grove	 8
37	Douglas fir	8	viable; grown in grove	8
38	Douglas fir	12	viable; grown in grove; so-dominant @ 22'	 8
39	Douglas fir	9	viable; grown in grove	8
40	Douglas fir	11	viable; grown in grove	8
41	Douglas fir	9	viable; grown in grove	 8
42	Douglas fir	12	viable; grown in grove	 10
43	sweet cherry	7	invasive species	 8
44	Douglas fir	11	viable	 8
45	Douglas fir	16	viable	10
46	Douglas fir	11	viable; grown in grove	8
47	Douglas fir	10	viable; grown in grove	8
48	Douglas fir	12	viable; grown in grove	8
49	Douglas fir	8	viable; grown in grove	8

8" and larger

	r i i i i i i i i i i i i i i i i i i i			
50	Douglas fir	9	viable; grown in grove	8
51	Douglas fir	10	viable; grown in grove	8
52	Douglas fir	5	undersize	8
53	Douglas fir	11	viable; grown in grove	8
54	Douglas fir	10	viable; windthrow potential; grown in grove	8
55	Douglas fir	12	viable; grown in grove	 8
56	Douglas fir	9	viable; windthrow potential; grown in grove	8
57	Douglas fir	13	viable; grown in grove	8
58	Douglas fir	9	viable; grown in grove	8
5 9	Douglas fir	11	viable; grown in grove	8
60	Douglas fir	12	viable; grown in grove	8
61	Douglas fir	10	dead	0
62	Douglas fir	8	viable; grown in grove	9
63	Douglas fir	11	viable; grown in grove	9
64	Douglas fir	10	viable; grown in grove	 9
65	Douglas fir	9	viable; grown in grove	8
66	Douglas fir	10	viable; grown in grove	8
67	Douglas fir	10	viable; grown in grove	 8
68	Douglas fir	14	viable	8
69	Douglas fir	11	viable; grown in grove	8
70	western redcedar	15	viable	10
71	western redcedar	12	viable	9
72	Douglas fir	9	viable; grown in grove	 9
73	Douglas fir	14	viable	8
74	red alder	14	co-dominant from base	8

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

75	Douglas fir	14	viable; grown in grove	8
76	Douglas fir	7	viable; grown in grove	8
77	Douglas fir	12	viable; grown in grove	8
78	Douglas fir	8	viable; grown in grove	8
79	Douglas fir	10	viable; grown in grove	8
80	Douglas fir	5	viable; grown in grove	8
81	Douglas fir	. 9	viable; grown in grove	8
82	Douglas fir	9	viable	8
83	sweet cherry	5	invasive species	8
84	bigleaf maple	12	viable	8
85	Douglas fir	23	viable; dead branches	16
86	bigleaf maple	8	excessive lean	8
87	red alder	14	viable	10
88	red alder	13	viable	8
89	red alder	13	viable	8
90	mountain ash	16	invasive species	10
91	Douglas fir	16	viable; co-dominant @ 12 ft.	10
92	Douglas fir	30	viable; co-dominant @ 3 ft.	20
93	Douglas fir	17	viable; hedgerow	12
94	Douglas fir	16	viable	10
95	American elm	7	viable; hedgerow	8
96	American elm	7	viable; hedgerow; excessive lean; trunk bowed	8
97	American elm	33	viable; hedgerow; 6 co-dominant stems from base	20
98	Douglas fir	4	viable; hedgerow	8
99	American elm	22	dead; 7-inch Douglas fir grows from base	0

10/23/2018

100	Douglas fir	15	poor taped; low LCR	10
101	Douglas fir	17	viable	12
102	Douglas fir	20	viable	14
103	Douglas fir	19	viable	12
104	Douglas fir	18	viable	12
105	Douglas fir	21	viable	13
106	Douglas fir	17	viable	12
107	Douglas fir	16	viable	10
108	Douglas fir	20	viable	14
109	Douglas fir	19	viable	12
110	Douglas fir	17	viable	12
111	Douglas fir	17	viable	12
112	Douglas fir	18	viable	12
113	Douglas fir	17	viable	12
114	Douglas fir	17	viable	12
115	western redcedar	63	base is hollow; co-dominant stems from base	25
116	Douglas fir	19	viable	12
117	Douglas fir	16	viable	10
118	Douglas fir	19	viable	12
119	Douglas fir	21	viable	14
120	Douglas fir	22	viable	14
121	Douglas fir	17	viable	12
122	Douglas fir	17	viable	12
123	Douglas fir	20	viable; small co-dominant stem	14
124	Douglas fir	17	viable	12

8" and larger

125	Douglas fir	15	viable		10
126	Douglas fir	10	viable; grown in grove		8
127	Douglas fir	16	viable; grown in grove		10
128	Douglas fir	9	viable; grown in grove		8
129	Douglas fir	8	viable; grown in grove		8
130	Douglas fir	14	viable; grown in grove		10
131	Douglas fir	7	viable; grown in grove		8
132	Douglas fir	12	viable; grown in grove		8
133	Douglas fir	14	viable; grown in grove		10
134	Douglas fir	9	viable; grown in grove		8
135	Douglas fir	10	viable; grown in grove		
136	Douglas fir	21	viable; hedgerow; pruned for high votage wires		14
137	western redcedar	18	viable; hedgerow; pruned for high votage wires		12
138	Douglas fir	39	viable; hedgerow; pruned for high votage wires		25
139	Douglas fir	8	viable; grown in grove		8
140	Pacific yew	9	dead		0
141	western redcedar	12	viable; hedgerow		8
142	Douglas fir	16	viable; hedgerow; pruned for high votage wires		10
143	Douglas fir	9	viable; grown in grove		8
144	Douglas fir	16	viable; hedgerow; pruned for high voltage wires; so-dominant from base		10
145	Douglas fir	11	viable; grown in grove		
146	Douglas fir	10	viable; grown in grove	_	8
147	Douglas fir	9	viable; grown in grove		8
148	Douglas fir	11	viable; grown in grove		8
149	Douglas fir	14	viable: grown in grove		10

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

8" and larger

10/23/2018

150	Douglas fir	15	viable; grown in grove	10
151	Douglas fir	9	viable; grown in grove	8
152	Douglas fir	7	viable; grown in grove	 8
153	Douglas fir	7	viable; grown in grove	8
154	Douglas fir	6	viable; grown in grove	8
155	Douglas fir	9	viable; grown in grove	8
156	Douglas fir	7	viable; grown in grove	8
157	Douglas fir	14	viable; hedgerow; pruned for high voltage wires	10
158	Douglas fir	26	viable; hedgerow; pruned for high voltage wires	16
159	Douglas fir	25	viable; hedgerow; pruned for high voltage wires	16
160	Douglas fir	30	viable; hedgerow; pruned for high voltage wires	20
161	Douglas fir	25	viable; hedgerow; pruned for high voltage wires; co-dominant @ 10 ft.	16
162	Douglas fir	17	viable; hedgerow; pruned for high voltage wires	12
163	Douglas fir	17	viable; grown in grown	12
164	Douglas fir	7	viable; grown in grown	8
165	Douglas fir	23	viable; grown in grown	16
166	Douglas fir	18	viable	12
167	Douglas fir	11	viable	8
168	Douglas fir	9	viable	8
169	Douglas fir	14	viable	10
170	western redcedar	11	viable	8
171	western redcedar	12	viable; co-dominant from base	8
172	western redcedar	17	viable; 3 co-dominant stems from base	12
173	apple	14	orchard pruned @ 18 ft.	8
174	western redcedar	62	viable; co-dominant stems from 45 ft.	25

8" and larger

175	western redcedar	47	viable; co-dominant stems from 3 ft.; multiple tops	25
176	western redcedar	38	viable	25
178	Douglas fir	26	viable; trunk swoop; unstable soil	 16
179	bigleaf maple	16	viable; large surface root	 10
180	Douglas fir	16	viable	 10
181	western redcedar	8	viable	 8
182	western redcedar	11	viable	 8
183	bigleaf maple	14	viable	10
184	Douglas fir	38	viable	 25
185	Douglas fir	11	topped for high-voltage wires	8
186	Douglas fir	30	viable; pruned for high-voltage wires	 20
187	Douglas fir	17	viable	 12
188	Douglas fir	27	viable	 18
189	Douglas fir	21	topped for high-voltage wires	14
190	Douglas fir	21	viable; pruned for high-voltage wires	 14
191	Douglas fir	28	viable; large surface root to northwest	 18
192	Douglas fir	24	viable	16
193	western redcedar	43,23	2 co-dominant stems; large flush cut on trunk @ old inclusion; basal decay; trunk decay	 25
194	western redcedar	64	viable; co-dominant stem broken	25
195	Douglas fir	5	undersize; grows from the base of T194	 8
196	cascara buckthorn	9	viable	8
197	western redcedar	57	viable; hedgerow	 25
198	western redcedar	62	3 co-dominant tops; broken stem; wood borers; loose/shedding bark on root flares	25
199	red alder	15	listed; flagging in top; mechanical damage to trunk	 0
200	Douglas fir	28	viable	18

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

201	western redcedar	15	viable	10
202	Douglas fir	22	viable	14
203	Douglas fir	17	viable; offsite	12
204	cascara buckthorn	0	broken	 o
205	Douglas fir	4	offsite; tagged #205	 8
206	Douglas fir	24	viable; was tagged #205; changed tag to #206	 16
207	Douglas fir	16	viable	 10
208	Douglas fir	23	viable; hedgerow; size estimated	 16
209	Douglas fir	24	viable; hedgerow; size estimated	 16
210	Douglas fir	26	viable; hedgerow; size estimated	 16
211	American elm	18	viable; hedgerow; co-dominant from base	 12
212	American elm	26	viable; hedgerow	 16
213	American elm	6	viable; hedgerow	8
214	American elm	6	viable; hedgerow	 8
215	Douglas fir	10	viable; hedgerow	 8
216	Douglas fir	22	viable; hedgerow	 14
218	sweet cherry	9	viable; hedgerow	
219	Douglas fir	18	viable; hedgerow	 12
220	bigleaf maple	8	viable; hedgerow	8
221	Douglas fir	20	viable; hedgerow	12
222	sweet cherry	8	invasive species	 8
223	Douglas fir	19	viable; hedgerow	 12
224	Douglas fir	19	viable; hedgerow	12
225	western redcedar	26	viable; hedgerow	16
226	bigleaf maple	18	viable; hedgerow; T2002 grows from base	12

10/23/2018

				-
227	bigleaf maple	13	viable; hedgerow	10
228	western redcedar	5	undersize	8
229	western redcedar	10	viable; hedgerow	8
230	western redcedar	6	undersize	8
231	Douglas fir	14	viable; hedgerow	8
232	bigleaf maple	10	viable; hedgerow	8
233	western redcedar	4	undersize	 8
234	Douglas fir	3	undersize	 8
235	Douglas fir	28	viable; hedgerow	18
236	western redcedar	10	viable; hedgerow	8
237	bigleaf maple	3	undersize	 8
238	bigleaf maple	4	undersize	 8
239	western redcedar	3	undersize	8
240	Douglas fir	17	viable; hedgerow	 12
241	bigleaf maple	7	undersize	 8
243	bigleaf maple	5	undersize	 8
244	western redcedar	9	viable; hedgerow	8
245	bigleaf maple	7	undersize	8
246	Douglas fir	33	viable; dead branches	21
247	Douglas fir	18	viable; hedgerow	12
248	red alder	15	dead	 0
249	bigleaf maple	14	viable; co-dominant from base	 8
250	Douglas fir	17	viable; hedgerow	12
251	bigleaf maple	7	undersize	 0
252	bigleaf maple	3	undersize	0

8" and larger

253	bigleaf maple	10	viable; hedgerow	 8
254	western redcedar	9	viable; hedgerow	 8
255	western redcedar	6	undersize	 0
256	Douglas fir	23	viable; hedgerow	 16
257	Douglas fir	24	viable; hedgerow	 16
258	Douglas fir	17	viable; hedgerow	 12
259	western redcedar	6	undersize	 0
261	bigleaf maple	3	undersize	 0
262	bigleaf maple	10	viable; hedgerow	 8
263	Douglas fir	20	viable; hedgerow	14
264	bigleaf maple	10	viable; hedgerow	 8
265	western redcedar	5	undersize	 0
266	Douglas fir	21	viable; hedgerow	 14
267	western redcedar	10	viable; hedgerow	 8
268	Douglas fir	18	viable; hedgerow	 12
269	bigleaf maple	7	undersize	 8
270	western redcedar	6	undersize	
271	Douglas fir	36	viable; dead branches; 75% LCR	 24
272	Douglas fir	41	viable	 25
273	Douglas fir	42	viable; dead branches; 75% LCR	 25
274	western redcedar	23	viable; co-dominant @ 15ft.	 24
275	bigleaf maple	11	viable; hedgerow	8
276	red alder	10	hollow; trunk decay; long cavity	 0
277	European birch	5	undersize	8
278	European birch	11	viable	8

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

279	Douglas fir	28	viable; hedgerow		18
280	Douglas fir	26	viable; hedgerow		16
281	Douglas fir	18	viable; hedgerow		12
282	Douglas fir	20	viable; hedgerow		18
283	Douglas fir	23	viable; hedgerow; small wound at 4 ft.		16
284	Douglas fir	21	viable; hedgerow		13
285	Douglas fir	11	viable; hedgerow; mechanical damage to trunk		8
286	bigleaf maple	10	broken top; tear out on base		88
287	Douglas fir	36	viable; hedgerow; large buttress roots	<u> </u>	_22
288	Douglas fir	33	viable; hedgerow; large buttress roots		20
289	Douglas fir	32	viable; hedgerow		20
290	Douglas fir	30	viable; hedgerow	<u> </u>	20
291	western redcedar	11	viable; hedgerow		8
292	Douglas fir	8	viable		8
293	western hemlock	10	viable		8
294	Douglas fir	11	viable		8
295	Douglas fir	33	viable; large buttress roots		22
296	Douglas fir	22	viable; ivy		14
297	Douglas fir	22	viable; ivy		14
298	Douglas fir	23	viable; ivy		16
299	Douglas fir	25	viable; ivy		16
300	Douglas fir	18	viable		12
301	Douglas fir	16	viable		10
302	western hemlock	13	viable	_	8
303	Douglas fir	8	viable; hedgerow		8

10/23/2018

304	Douglas fir	16	viable	10
305	Douglas fir	12	viable; hedgerow	8
307	bigleaf maple	7	undersize	8
308	Douglas fir	6	undersize	8
309	Douglas fir	18	viable	12
310	Douglas fir	17	viable	12
311	western redcedar	10	viable	 8
312	Douglas fir	16	viable; tagged #312 and #321 in field	10
313	Douglas fir	15	viable	10
314	Douglas fir	19	viable	12
315	Douglas fir	17	viable	12
316	Douglas fir	17	viable	12
317	Douglas fir	17	viable	 12
318	Douglas fir	24	viable	16
319	Douglas fir	17	viable	12
320	Douglas fir	32	co-domionant @ 20 ft.; large inclusion; hazardous	0
322	Douglas fir	16	viable	12
323	Douglas fir	17	viable	12
324	western redcedar	11	viable	 8
325	Douglas fir	17	viable	12
326	gone	0	gone	0
327	western redcedar	7	undersize	 8
328	Douglas fir	17	viable	12
329	Douglas fir	20	viable	14
330	Douglas fir	23	viable	14

8" and larger

331	Douglas fir	16	viable	10
332	western redcedar	11	viable	 8
333	bigleaf maple	6	undersize	8
334	western redcedar	10	viable	8
335	Douglas fir	15	viable	10
336	Douglas fir	19	viable	12
337	Douglas fir	18	viable	12
338	Douglas fir	16	viable	10
339	western redcedar	11	viable	8
340	western redcedar	14	viable	10
341	Douglas fir	10	viable	8
342	Douglas fir	7	undersize; dead	 0
343	sweet cherry	7	undersize; excessive lean	0
344	Douglas fir	9	viable; hedgerow; suppressed	8
345	Douglas fir	16	viable	 10
346	Douglas fir	17	viable	12
347	Douglas fir	7	undersize	8
348	Douglas fir	17	viable	12
349	Douglas fir	15	viable	 10
350	Douglas fir	12	viable	 10
351	Douglas fir	16	viable	10
353	gone	0	gone	 0
353	Douglas fir	17	viable	12
354	Douglas fir	15	co-dominant @ 25ft.; windthrow potential	10
355	bigleaf maple	8	viable	8

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

8" and larger

356	Douglas fir	15	viable	 10
357	western redcedar	30	large co-dominant stems @ 25 ft.	 20
358	Douglas fir	18	viable	12
35 9	apple	8	orchard pruned	 8
360	linden	8	viable	8
361	Douglas fir	19	viable	 12
362	bigleaf maple	17	trunk cavity and decay	 12
363	bigleaf maple	24	viable; 3 co-dominant stems from base	 16
364	apple	10	orchard pruned	 8
364	Douglas fir	16	viable	12
365	bigleaf maple	41	viable	 25
366	western redcedar	10	viable	 8
367	red maple	9	viable	 8
368	Douglas fir	9	viable; suppressed	8
369	bigleaf maple	12	viable	 8
370	western redcedar	18	viable	 12
371	Douglas fir	16	viable	10
372	western redcedar	6	undersize	 8
373	western redcedar	8	viable; hedgerow	8
374	Douglas fir	17	viable	 12
375	Douglas fir	22	viable	14
376	European birch	9	broken top	 8
377	bigleaf maple	11	viable	 8
378	western redcedar	8	viable	 8
379	western redcedar	12	viable	8

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

380	western redcedar	10	viable	8
381	bigleaf maple	34	viable	22
382	Douglas fir	16	viable	10
383	western redcedar	56	viable; co-dominant @ 20 ft.	25
385	Douglas fir	13	viable	8
386	western redcedar	33	trunk decay; basal decay; hollow; broke @ 20 ft.; multiple tops; habitat den in stump	22
387	bigleaf maple	14	viable	8
388	bigleaf maple	22	viable	14
389	western redcedar	9	viable; grows from nurse log	8
390	western redcedar	23	trunk and stem decay; co-dominant	16
391	western redcedar	22	viable; grows from old stump	14
392	western redcedar	8	viable	8
393	Douglas fir	11	viable	 10
394	Douglas fir	10	trunk decay	8
395	Douglas fir	16	viable	10
396	Douglas fir	24	viable; mechanical damage to root flare	16
397	Douglas fir	16	viable	 10
398	Douglas fir	16	viable	 10
399	western redcedar	18	basal decay; trunk decay	12
400	Douglas fir	42	viable	25
401	bigleaf maple	38	mechanical damage to base and buttress roots; mutiple tops @ 40'	25
402	Douglas fir	13	viable	8
403	bigleaf maple	27	large stem broke; trunk decay	18
404	western redcedar	10	viable; grows from nurse log	8
405	bigleaf maple	6	multiple 6-inch stump suckers; basal decay	0

8" and larger

406	western redcedar	29	base is hollow	 0
407	Douglas fir	21	viable	14
408	western hemlock	10	viable; wound on trunk	 8
409	gone	0	gone	 0
410	western redcedar	19	mechanical damage to trunk	12
411	western redcedar	27	mechanical damage to trunk; wood borers	18
412	bigleaf maple	28	viable; co-dominant @ 30'; tear out @ 40'	 18
413	western redcedar	35	terminal decline; severe basal decay	 0
414	western redcedar	9	viable	 8
415	bigleaf maple	21	viable; dead branches	 14
416	gone	0	gone	 0
417	western redcedar	7	undersize; viable	8
418	Douglas fir	13	viable; poor trunk taper	8
419	western redcedar	12	viable	 10
420	western redcedar	9	viable	 8
421	western redcedar	10	viable	 8
422	western redcedar	8	viable	 8
423	western redcedar	9	viable	 8
424	bigleaf maple	10	viable	 8
425	western redcedar	8	viable	 8
426	western redcedar	8	viable	 8
427	western redcedar	8	viable	8
428	Douglas fir	7	undersize; mechanical damage to trunk and stem	8
429	western redcedar	9	viable; co-dodominant from base	8
430	bigleaf maple	38	co-dominant @ 30 ft.; northern stem broken; woodpeckers @ 20 ft.	25

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

10/23/2018

431	red alder	12	viable; 15 degree trunk swoop	10
432	western redcedar	11	viable	8
500	western redcedar	21	dead top; wound on trunk; trunk decay	14
501	western redcedar	21	viable	14
502	Douglas fir	23	viable	16
503	western redcedar	11	viable	8
504	western redcedar	17	viable	12
505	western hemlock	11	trunk decay; long trunk wound	0
506	western redcedar	8	trunk decay; long trunk wound	0
507	western redcedar	10	trunk decay; long trunk wound	0
508	western redcedar	30	viable	20
509	western redcedar	37	viable	25
510	western redcedar	15	viable; suppressed; co-dominant	10
511	bigleaf maple	37	viable; minor wounds on trunk; dead branches	25
512	western redcedar	52	co-dominant from base; basal decay; hollow; hazardous	0
513	western redcedar	14	viable	10
514	western redcedar	19	viable; co-dominant @ 12 ft.	12
515	Douglas fir	34	viable; small wound on trunk; base is swollen	22
516	western redcedar	20	viable	12
517	bigleaf maple	10	viable	8
518	western redcedar	12	viable	8
519	Douglas fir	9	viable	8
520	western hemlock	14	viable; wound on trunk	10
521	western redcedar	18	viable; no tag in field	12
522	western hemlock	11	viable	8

8" and larger

523	western redcedar	18	viable	12
524	western redcedar	42	basal decay; trunk decay for 30 ft.	25
525	western redcedar	15	viable	 10
526	western redcedar	8	listed; hung up in T525	8
527	western redcedar	12	viable	8
528	western redcedar	12	viable	 8
529	western redcedar	9	viable	8
530	western redcedar	18	viable	 12
531	Douglas fir	12	viable; poor trunk taper; windthrow potential	 8
532	Douglas fir	9	viable; poor trunk taper; windthrow potential	8
533	Douglas fir	6	viable; suppressed	 8
534	Douglas fir	15	viable; poor trunk taper	10
535	western hemlock	13	20-foot trunk scar; poor response growth; windthrow potential	8
536	western hemlock	8	35-foot trunk scar; terminal decline	 8
537	western redcedar	17	viable	12
538	western hemlock	10	viable; minor mechanical damage to trunk	8
539	western hemlock	16	viable; grows from nurse stump	 10
540	Douglas fir	8	listed; hung up in T541	 8
541	red alder	10	viable; trunk swoop	8
542	western hemlock	12	viable	8
543	Douglas fir	8	viable	8
544	Douglas fir	9	viable	 8
545	Douglas fir	25	viable	 16
546	Douglas fir	18	viable	 12
547	red alder	17	listed and self-corrected; excessive lean	12

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

8" and larger

548	western redcedar	47	viable; approx. 2 ft. offsite	25
549	red alder	14	viable; on property line	8
550	red alder	17	viable; at creek	 12
551	western red cedar	21	trunk decay	 14
552	western red cedar	22	viable	14
553	western red cedar	25	viable	 16
554	western red cedar	13	viable	 8
555	dead	5	dead	0
556	red alder	6	viable; undersize	 9
557	red alder	15	viable	 10
558	Douglas fir	5	suppressed; undersize	 9
559	red alder	11	viable; at creek	 8
560	western red cedar	33	viable; some trunk decay; at creek	 22
561	western red cedar	50	viable	25
562	red alder	9	excessive lean; trunk bows 60 ft. to south	 9
563	red alder	8	terminal decline	 9
564	bigleaf maple	24	viable	 16
565	red alder	10	viable	9
566	red alder	12	viable	8
567	red alder	12	viable; leans towards site	8
568	bigleaf maple	17	viable	12
569	bigleaf maple	5	undersize; stump sucker	9
570	western hemlock	7	viable; undersize	8
571	western red cedar	27	wound on trunk; trunk decay	18
573	western redcedar	9	viable	8

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

8" and larger

574	western redcedar	9	viable	8
575	bigleaf maple	32	broken stem; trunk decay; re-trenched	22
1003	red alder	22	viable; added by arborist	14
2000	Douglas fir	15	viable; added by arborist; shares stump with T177; unstable soil	10
2001	Douglas fir	12	viable; added by arborist	8
2002	Douglas fir	9	viable; hedgerow; added by arborist	 8
2003	Douglas fir	16	viable; hedgerow; added by arborist	 10
2004	bigleaf maple	10	viable hedgerow; added by arborist	 8
2005	western redcedar	17	viable; hedgerow	 12
2006	western redcedar	10	viable; hedgerow; added by arborist; grows from the base of T263	 8
2007	Douglas fir	16	viable; added by arborist	 10
3000	bigleaf maple	18	viable; no top; naturally re-trenched; added by arborist	 12
3001	red alder	20	off-site; grows over property by 40 ft. listed and self-corrected; added by arborist	 14
3002	red alder	15	viable; trunk swoop; grows from base of T534; added by arborist	 10
3003	red alder	14	4 ft. off-site; leans over site; added by arborist	 8
3004	bigleaf maple	17	2 ft. off-site; added by arborist	 12
3005	bigleaf maple	15	viable; listed and self-corrected; added by arborist	 10
3006	western redcedar	13	viable; on property line; grows on nurse stump; base is 4.5 ft.high; added by arborist	8
3007	western redcedar	8	viable	9
3008	western redcedar	19	viable; 8 ft. off-site	 12
177	Douglas fir	25	viable; shared stump with T2000; unstable soil	16
242A	western redcedar	4	undersize	8
242B	bigleaf maple	6	undersize; dead	0
260A	western redcedar	15	viable; hedgerow	10
260B	bigleaf maple	6	undersize	0

Field work by Ryan Neumann, PN-5539A on 10/12, 10/15, and 10/17/2018

8" and larger

572A	western redcedar	15 wound on base; wood borers	10
572B	western redcedar	43 large trunk swoop from base; broken stem; low vigor	25

10/23/2018

EXHIBIT L

1 40

CURRAN-MCLEOD, INC. CONSULTING ENGINEERS 6655 S.W. HAMPTON STREET, SUITE 210 PORTLAND, OREGON 97223

January 15, 2019

Ms. Emily Meharg City of Sandy 39250 Pioneer Blvd. Sandy, OR 97055

RE: CITY OF SANDY JACOBY HEIGHTSD SUBDIVISION (FILE NO. 18-025 SUB/VAR/FSH/TREE/INT) PRELIMINARY REVIEW

Dear Emily:

We have reviewed the resubmitted preliminary plans and supporting documents for the above noted development and have the following comments:

- 1. We have reviewed the "Geotechnical Report" prepared by Rapid Soil Solutions, LLC, dated February 23, 2018 and recommend that the developer retains appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage and terracing should comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, a final report should be submitted to the City by the Geotechnical Engineer stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the above noted report and the OSSC.
- 2. A demolition permit shall be obtained from the City of Sandy Building Department prior to demolition of the existing structure on-site.
- 3. The preliminary plans don't identify any existing domestic or irrigation wells on-site. Any existing well found on-site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City.
- 4. The preliminary plans don't identify any septic tanks, any on-site sewage disposal system found on-site shall be abandoned in conformance with Clackamas County Water Environment Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City.

PHONE: (503) 684-3478

E-MAIL: cml@cuttan-mcleod.com

FAX: (503) 624-8247

Mr. Emily Meharg January 15, 2019 Page 2 0

- 5. We reviewed the preliminary stormwater calculations were provided with this submittal. The calculations are found to meet the water quality/quantity criteria as stated in the City of Sandy Development Code (SDC) 13.18 Standards and the City of Portland Stormwater Management Manual (SWMM) Standards, that were adopted by reference into the Sandy Development Code. However, a detailed final report stamped by a licensed professional shall be submitted for review with the final construction plans.
- 6. A Wetlands Report outlining the delineated wetlands/ high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). The report shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife.
- FSH Overlay District line at northeast corner of the site should be review by the City of Sandy Planning Department in conformance with Sandy Municipal Code (SDC), section 17.60.30.
- 8. We reviewed the "Traffic Impact Analysis" prepared by Lancaster Engineering dated June 15, 2018 providing that all intersections are projected to continue operating to acceptable levels of service, the sight distance requirements at the Woodstock Street and Newton Street intersections with Jacoby Road are met.

There are no reported crashes at the intersection of Jacoby and Dubarko Road, or at any of the intersections adjacent to the development. However, the crash rate at the intersection of Dubarko Road and Highway 211 exceeds the 90th percentile criteria and meets the warrants for the addition of left turn lanes as identified in the TSP. This intersection should be addressed by ODOT as soon as possible, but not as a component of this proposed development.

- 9. The developer's applicant has submitted a variance related to the 400-foot block length standard on Woodstock Street as required by Sandy Development Code (SDC), section 17.100.120(B). Sheet 8 of the preliminary plans provide a street layout and lots configuration on the adjoining properties, given the wetlands/Tickle Creek is located to the north, a stubbed street is not feasible. Extending the block length for approximately 100 feet appears to be reasonable and is compatible with the location of the existing house on the adjacent property to the east.
- 10. Half Street Improvements should be constructed along the entire site frontage with Jacoby Road to collector street standards to include curbs, sidewalks, street lights, landscaping strip with trees and utilities as appropriate. The improvements shall include an additional 10 feet of right-of-way dedications (60' ROW).

1 350

Mr. Emily Meharg February 20, 2018 Page 3 £ .)

No. 1

 All interior streets (Woodstock Street, Camden Court and Middle Berry Avenue) should be constructed to local street standards (28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot wide sidewalks) in compliance with the City of Sandy Transportation System Plan (TSP), figure 12.

()

- 12. 3/4 Improvements should be required on Newton Street to include 28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot wide sidewalks on the north side of the roadway.
- 13. The developer's engineer should provide a profile design for a minimum of 200 feet for all future street extensions (Newton Street and Woodstock Street) past the project boundary to ensure future grades can be met.
- 14. At the eastern terminus of Newton Street, the paved width needs to be a minimum of 20 feet to allow for two ways traffic. This stub appears to be in excess of 150 feet from the nearest intersection and may require a temporary turnaround or lots being sprinkled as required by Sandy Fire District.
- 15. The plans don't clearly indicate the radius for the proposed cul-de-sac, the minimum paved cul-de-sac radius shall be 48 feet as per the Oregon Fire Code and Metro Code Committee. A review by the Fire Department to confirm compliance will be required.
- 16. All public sanitary sewer, waterline mains to be a minimum of 8-inches in diameter and a minimum of 12-inches in diameter for storm drains and be extended to the plat boundaries where practical to provide future connections to adjoining properties. All utilities are extended to the plat boundary for future connections.
- 17. The access to the detention shall be paved or all-weather surface to a minimum of 12foot in width.
- The detention pond shall meet the requirements of the 2016 City of Portland StormWater Management Manual (SWWM) for landscaping section 2.4.1 and escape route, section 2.30.

We have no concerns about the proceedings with this project subject to the above stated comments.

Mr. Emily Meharg January 15, 2019 Page 4

Very truly yours,

CURRAN-McLEOD, INC.

0

Hassan A. Ibrahim, PE

cc: Mr. Mike Walker, City of Sandy

1- 1- V

 \mathbb{Q}_{i}
EXHIBIT M

REPLINGER & ASSOCIATES LLC TRANSPORTATION ENGINEERING

January 18, 2019

 \square

Ms. Emily Meharg City of Sandy 39250 Pioneer Blvd. Sandy, OR 97055

SUBJECT: REVIEW OF TRANSPORTATION IMPACT ANALYSIS – JACOBY HEIGHTS SUBDIVISION

Dear Emily:

In response to your request, I have reviewed materials submitted in support of the Jacoby Heights Subdivision. The materials consisted of the site plan and the Transportation Impact Analysis (TIA) for the Marshall Ridge Subdivision. The TIA was prepared under the direction of William Farley, PE of Lancaster Engineering. The TIS is dated June 15, 2018.

The TIA describes a proposal to construct a 31-lot subdivision of single-family dwellings. The site is in the southwest part of Sandy, south of Dubarko Road and east of Jacoby Road. The proposed access on Jacoby Road would include an extension of Newton Street and a new access north of Newton Street and south of Cascadia Village Drive.

Overall

I find the TIA addresses the city's requirements and provides an adequate basis to evaluate impacts of the proposed development.

Comments

- 1. Study Area. The study addresses the appropriate intersections. It includes analyses of:
 - OR 211 at Dubarko Road
 - Dubarko Road at SE Jacoby Road
 - SE Jacoby Road at New Site Access
 - SE Jacoby Road at Newton Street
- 2. Traffic Counts. The AM and PM peak hour traffic counts were conducted on May 1 and May 2, 2018 at the study area intersections. The engineer adjusted the May traffic counts to account for seasonal variations according to the procedures defined by the Oregon Department of Transportation (ODOT). The counts appear reasonable.
- 3. Trip Generation. The TIA uses trip generation for single-family houses from the Institute of Transportation Engineers' (ITE) Trip Generation Manual. The site is currently occupied by

Ms. Emily Meharg January 18, 2019 Page 2

one single-family dwelling. The calculations of trip generation were based on 32 additional houses based on a prior site plan. The new site plan indicates one less dwelling. The engineer calculates that the 32 additional houses would produce 23 new AM peak hour trips; 31 PM new peak hour trips; and 292 new daily trips. The calculation of trips generated by the subdivision appears reasonable.

- 4. Trip Distribution. The TIA provided information about trip distribution from the site. The engineer assumed 20 percent of the traffic would travel to and from the north on OR 211; 30 percent would travel to and from the south on OR 211; and 25 percent would travel to and from the west on Dubarko Road west of OR 211; and 25 percent would travel to and from the east on Dubarko Road. Trip distribution patterns were based on existing traffic counts and judgement about likely destinations. All trips to and from the subdivision were assumed to use Jacoby Street to Dubarko Road. The trip distribution seems reasonable.
- 5. Traffic Growth. The TIA uses a 3.2 percent annual increase for OR 211 based on ODOT's Future Volume Tables. For other facilities it uses a 2.0 percent annual growth rated background traffic growth. In addition, the TIA specifically accounts for in-process developments to calculate 2020 background conditions including the nearby Zion Meadows, Marshall Ridge, Mt. View Ridge, and Snowberry Subdivisions. Background volumes were prepared for 2020, the year in which the development is expected to be completed. These assumptions account for future traffic and appear reasonable.
- 6. Analysis. Traffic volumes were calculated for the intersections cited in #1, above. Intersection level-of-service (LOS) and the volume-to-capacity (v/c) ratio were provided. ODOT uses the v/c ratio for its standard of intersection performance. Performance of the intersections was calculated for existing 2018 conditions; 2020 background conditions; and 2020 conditions with the proposed subdivision.

For the intersection of OR 211/Dubarko Road, the engineer indicates that the intersection currently operates at LOS C during both the AM and PM peak hours. By 2020, the intersection's performance will degrade slightly and is calculated to operate at LOS D during both the AM and PM peak hours with or without this development. The poorest performance is predicted during the AM peak hour when the v/c ratio is expected to increase to 0.52. This easily meets ODOT's performance standard.

All three of the other study area intersections is calculated to operate under LOS B or better under all scenarios and will meet city performance standards.

The engineer recommends no mitigation for traffic from this proposal. I concur.

7. Crash Information. The TIA provides information on crashes for the most recent available five-year period (2012 through 2016). For the five-year period, 28 crashes were reported at OR 211/Dubarko Road intersection. The engineer reports that 16 occurred prior to the

Ms. Emily Meharg January 18, 2019 Page 3

intersection realignment and 12 occurred since completion of the project that changed the approach angle of the intersection. The engineer provided a good summary of crashes including crash types and other characteristics. The most prominent type was angle collisions in which eastbound or westbound motorists failed to yield to traffic on OR 211. The crash rate was calculated to be 1.83 crashes per million entering vehicles, a rate that exceeds the 90th percentile rate specified by ODOT in its *Analysis Procedures Manual*. The engineer notes that the high crash rate may indicate the need for improvements and opines that the "installation of the planned turn lanes and traffic signal outlined in the Sandy TSP will likely reduce the crash rate at this intersection."

No crashes were reported at Jacoby Road and Dubarko Road; one was reported at Jacoby Road and Newton Street. Neither location appears to need mitigation for safety issues.

The engineer did not recommend mitigation for any safety issues. I agree that imposing conditions on this subdivision to help address safety issues at OR 211/Dubarko Road is not appropriate.

I recommend continuing to work with ODOT to evaluate and address safety issues at the intersection of OR 211/Dubarko Road and work toward implementation of the intersection improvement project specified in the TSP.

8. Site Plan and Access. The site plan provides for two access points on SE Jacoby Road: an extension of Newton Street, which currently extends west from Jacoby Road; and a new intersection extending into the subdivision from Jacoby Road between Newton Street and Cascadia Village Drive. The site plan also indicates both east-west streets will terminate with stubs for potential connection when land to the east of the proposed subdivision develops.

The new intersection with SE Jacoby Road is proposed to be located approximately 270 feet north of Newton Street and about 380 feet south of Cascadia Village Drive. The intersection location is appropriate.

- 9. Sight Distance. The engineer analyzed sight distance at the new proposed intersection on Jacoby Road and at Newton Street. For the new intersection, the engineer measured sight distance to be 350 feet to the north and 410 feet to the south. For the Newton Street intersection, he measured sight distance to be 545 feet to the north and 595 feet to the south. These values are far in excess of the requirements for a 25-mph speed zone. The engineer concluded sight distance is adequate. I concur.
- 10. Left-Turn Lane and Signal Warrants. The engineer analyzed the subject intersections for left-turn lanes using standard methods based on traffic volumes, travel speeds, and lanes.

Ms. Emily Meharg January 18, 2019 Page 4

For the intersection of OR 211 and Dubarko Road, he concluded left-turn lane warrants for northbound and southbound approaches meet warrants under existing conditions. For the intersections of Dubarko/Jacoby, Jacoby/Newton and Jacoby/New site access, he concluded that left-turn lane would not be met for 2020 conditions with the proposed development.

He also analyzed traffic signal warrants at the OR 211 and Dubarko Road intersection for 2020 conditions. The Sandy TSP specifies a traffic signal and left-turn lanes on OR 211, but under projected 2020 traffic volumes he concluded traffic signal warrants would not be met. I concur.

11. Conclusions and Recommendations. The engineer concludes that traffic operations will be acceptable with the development without mitigation. Though crash data indicates a concern at the intersection of OR 211 and Dubarko Road, he did not recommend mitigation for safety concerns. I concur with these conclusions.

Conclusion and Recommendations

I find the TIA meets City requirements. The TIA demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards. The intersections that will provide access from the subdivision to Jacoby Road are appropriately located and provide adequate sight distance.

The existence of a higher than average crash rate at the intersection of OR 211 and Dubarko Road and traffic volumes that suggest left-turn lanes are warranted on OR 211 at this location are indicators that the city should continue to work with ODOT for implementation of enhancements at this intersection. The analysis indicates that a traffic signal is not yet warranted but left-turn lanes on both the northbound and southbound approaches on OR 211 are warranted. The need for these improvements is justified based on regional growth of which this development is only a minor component.

If you have any questions or need any further information concerning this review, please contact me at <u>replinger-associates@comcast.net</u>.

Sincerely,

John Keplinger

John Replinger, PE Principal

JacobyHeightsSubdTIA011819

1/24/2019

City of Sandy Mail - RE: Jacoby Heights Subdivision (File Mo. 49 025 SUB/VAR/FSH/TREE/INT)



EXHIBIT N

ly Meharg <emeharg@ci.sandy.or.us>

RE: Jacoby Heights Subdivision (File No. 18-025 SUB/VAR/FSH/TREE/INT) 1 message

Kristine Hendrix <Kristine.Hendrix@pgn.com> To: "emeharg@ci.sandy.or.us" <emeharg@ci.sandy.or.us> Wed, Jan 23, 2019 at 2:03 PM

We have reviewed the pre-application description you sent us and we haven't found any conflicts to your project right now.

When the developer is ready to start the project please have them call PGE Service Coordination's at 503-323-6700.



Kristine Hendrix Job Processor

Work Hours 6:30 am to 4:00 pm M - TH & 6:30 am to 10:30 am Fri

Portland General Electric

1705 NE Burnside, Gresham, OR 97030

1 2: (503) 669-5214 | 4: (503) 669-5229 | Kristine.hendrix.@pgn.com

19124 Jacoby Road Sandy OR.pdf 331K

https://mail.google.com/mail/u/0?ik=95a4e5548c&view=pt&search=all&permthid=thread-f%3A1623490304093391505&simpl=msg-f%3A16234903040... 1/1



EXHIBIT O

MEMORANDUM

TO: EMILY MEHARG, ASSOCIATE PLANNER FROM: MIKE WALKER, PUBLIC WORKS DIRECTOR RE: JACOBY HEIGHTS SUBDIVISION DATE: JANUARY 26, 2019

 $\left(\right)$

The following are Public Works' comments on the above-referenced application:

Transportation

The traffic impact analysis and the City Traffic Engineer's review demonstrate that all site access points and analyzed street intersections will meet the City's minimum level of service (LOS) standard at buildout.

The site has approximately 600 LF of frontage on Jacoby Rd., a designated collector. The right-of-way width adjacent to the site is 50 ft. The standard for collector streets is 60 ft. The applicant shall dedicate 10 ft. of right-of-way along the Jacoby Rd. frontage of the site.

Two local streets will intersect Jacoby, Woodstock St. and an extension of Newton St. aligning with the existing street of the same name on the west side of Jacoby. The proposed alignment of Newton St. does not conform with the 50 feet of tangent alignment (measured from the curb line extended) required in Section 17.84.50(H)5b Sandy Municipal Code (SMC). The applicant shall revise the alignment of Newton St. to conform with the standard.

The applicant is proposing a ³⁄₄ street improvement (curb on both sides, 28 ft. wide street with a planter strip and sidewalk on the development site) for Newton St. through the site transitioning to a half-street section at the west line of proposed lot 2. The proposed Newton St. alignment coincides with an existing access easement serving at least two and possibly as many as four existing dwellings to the east. No more than two dwellings may be served by a half-street improvement. As proposed this narrowed section of Newton would serve at least four dwellings and perhaps as many as six. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site.

The applicant shall construct a mid-block ADA ramp on the west side of Jacoby opposite Woodstock St. to provide ADA access to the existing sidewalk on the west side of Jacoby and the neighborhood to the west.

The proposed parking plan shows 21 parking spaces on the south side of Newton St. These spaces should not be counted toward meeting the on-street parking requirement since the south side of the street will eventually develop with driveway and/or streets that will eliminate these spaces or prevent the developer of the site to the south from counting these spaces toward their on-street parking requirement. The applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval.

 \bigcirc

The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of lot 29 to eliminate remnant of Tract A between the access drive and lot 29. In order to maximize street tree planting areas and prevent conflicts with streetlights, water meters and dry utility facilities, shared driveway approaches may be required in cul-de-sac bulbs. Lot 28 and 29 shall share a driveway approach maximum 24 ft. in width. Lot 26 and 27 shall share a driveway approach maximum 24 ft. in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches. The driveway approach for Tract A shall be a maximum of 12 ft. in width.

Utilities

The proposed utility plan shows an 8" sanitary sewer extended south from the existing sewer line in Jacoby Rd. approximately 490 feet to the site. The site will drain to the existing SE sewer pump station. The applicant shall provide a sanitary sewer lateral serving TL 6100 with the sanitary sewer main line extension.

Water service will be provided from a connection to the existing 8" water main at the intersection of Cascadia Village Dr. and Jacoby Rd. There is no other proposed connection to the water distribution system. The applicant shall demonstrate that adequate fire and domestic flow will be available with single point of connection to the water distribution system or connect to the existing 8" water line at the intersection of Newton and Amherst.

Site stormwater will be collected and conveyed to a proposed detention pond located in Tract A. Detained and treated discharge from the detention pond is proposed to be discharged to the adjacent property to the north. The applicant shall demonstrate that pre-development site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way.

The plan depicts a 20 ft. wide common public storm and sanitary easement along the west boundary of lots 9 and 10. The City reserves the right to widen this easement depending on the final design depth of the sanitary and storm sewer lines per section 17.84.90(A)2 SMC.

SANDY OREGON	Ď	EXHIBIT P	/ Meharg <emeharg@ci.sandy.or.us></emeharg@ci.sandy.or.us>
Jacoby Heights	subdivision - trai	nsit comments?	
Andi Howell <ahowell To: Emily Meharg <em< td=""><td>@ci.sandy.or.us> eharg@ci.sandy.or.us></td><td></td><td>Mon, Jan 28, 2019 at 4:25 PM</td></em<></ahowell 	@ci.sandy.or.us> eharg@ci.sandy.or.us>		Mon, Jan 28, 2019 at 4:25 PM
Yes, Emily, the N	W corner as close to	Cascadia Village Dr. as po	ossible. Thanks!
Andi Howell Transit Director			
City of Sandy 16610 Champion Wa Sandy, OR 97055 503-489-0925 Fax: 503-826-0618 ahowell@ci.sandy.or	y us		
Sandy Transit Web a	nd Trip Planner		
Sandy Area Metro			
[Quoted text hidden]			

https://mail.google.com/mail/u/0?ik=95a4e5548c&view=pt&search=all&permmsgid=msg-f%3A1623952247216918648&simpl=msg-f%3A16239522472... 1/1

RECEIVED **EXHIBIT Q** FEB 0 6 2019 No. 18-025 SUB/VAR/FSH/TREE/INT: **CITY OF SANDY** 60)01 Phone Number olorad Address APPLICABLE CRITERIA: Sandy Municipal Code: 17.12 Procedures for Decision Making; 17.14 Request for Interpretation; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 Single Family Residential (SFR); 17.56 Hillside Development; 17.60 Flood and Slope Hazard Overlay District; 17.66 Adjustments and Variances; 17.80 Additional Setbacks on Collector and Arterial Streets; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.86 Parkland and Open Space; 17.92 Landscaping and Screening; 17.98 Parking, Loading, and Access; 17.100 Land Division; 17.102 Urban Forestry; and 15.30 Dark Sky Ordinance.

18-025 SUB VAR FSH TREE INT Jacoby Heights Subdivision Notice

Page 3 of 3

RECEIVED **EXHIBIT R** () FEB 1 1 **2019** No. 18-025 SUB/VAR/FSH/TREE/INT: Valarianse i CITY OF SANDY TO THE COUNTY/CETY PLANNER FOR THE CETT OF SANDY." the city of ore. 100m issues tacing U. A lack tocesight influx of perple moring -10 (mits. <u>(7) The</u> 10 Support said F Interstacture to cease construction of new this point in time Same aught be/iere or adifimal neighborhoods Infrastructure until They have strongh ort the residents who are a ready paying taxes within limits. We need a road (Hard HW4 76) that will more people though Sandly. without a back yo to Kelso Rd. lit need shapping conters this end (my end) of town so that use we don't hale to Sit traffic for 25 minutes just to go Z.S miles to Fred Mey We need Commarcially to build more residential. before we continue without realizing that Sandy continues to operate small town They AS_A a_tulk_ -city: Step pretending and Start working an tinotiming to build me and my children a Chacerbates Acting more porto any the forstation of the residents within the City limits. build have here - I an not against builden houses society it the menstructure is in place to support additional if is clearly not. pople - which Shawn 971.413.4384 Corenze Your Name Phone Number 39343 Stratford Pl. SAUDY OR 97055

Address

APPLICABLE CRITERIA: Sandy Municipal Code: 17.12 Procedures for Decision Making; 17.14 Request for Interpretation; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 Single Family Residential (SFR); 17.56 Hillside Development; 17.60 Flood and Slope Hazard Overlay District; 17.66 Adjustments and Variances; 17.80 Additional Setbacks on Collector and Arterial Streets; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.66 Parkland and Open Space; 17.92 Landscaping and Screening; 17.98 Parking, Loading, and Access; 17.100 Land Division; 17.102 Urban Forestry; and 15.30 Dark Sky Ordinance.

18-025 SUB VAR FSH TREE INT Jacoby Heights Subdivision Notice

Page 3 of 3

EXHIBIT S FEB 1 2 2019 CITY OF SANDY

Dear City Planners,

This is regarding the development of the property 19124 Jacoby Rd file number (18-025 SUB/VAR/FSH/TREE/INT)

As we are the property directly north of the proposed development and having 1.5 acres of wooded property which boarders the Tickle Creek Wetlands, we have the following concerns and would like to know how these will be addressed:

- What type of fence or division will be between our forest property and this new development that will deter trespassing and possible liability to us?
- Where will the sewer come into the neighborhood and will we be required to connect to it at this time?
- Where will the water retention pond be located?
- Does the City spray annually for mosquitos annually around that water retention area?
- We would like to attend this meeting to better understand the requested variances.
- Is there any idea how long we will be subjected to the noise and disturbance of constructing this neighborhood?

Thank you for your consideration of these concerns,

Tim and Cherri Anders 19098 Jacoby Rd Sandy. Oregon 97055 360-903-0644 2/12/2019

SANDY OREGON City of Sandy Mail - Development 18-025 SUB/VAP "SH/TREE/INT

Rebecca Casey <rcasey@ci.sandy.or.us>

Tue, Feb 12, 2019 at 4:35 PM

Development 18-025 SUB/VAR/FSH/TREE/INT

1 message

Tim Anders <tim.anders@gmail.com>

To: planning@ci.sandy.or.us, emeharg@ci.sandy.or.us

This is regarding the development at 19124 Jacoby Rd Sandy OR

Tim Anders

SubdivisionLetter.docx 14K

https://mail.google.com/mail/u/0?ik=3ad3df1da3&view=pt&search=all&permthid=thread-f%3A1625311841495859239&simpl=rmsg-f%3A162531184149... 1/1

EXHIBIT T

CHAPTER 17.102 URBAN FORESTRY

17.102.00	INTENT	1
17.102.10	DEFINITIONS	1
17.102.20	APPLICABILITY	2
17.102.30	PROCEDURES AND APPLICATION REQUIREMENTS	2
17.102.40	PERMIT REVIEW	4
17.102.50	TREE RETENTION AND PROTECTION REQUIREMENTS	4
17.102.60	TREE REPLANTING REQUIREMENTS	5
17.102.70	VARIANCES	5
17.102.80	ENFORCEMENT	5
17.102.90	APPLICABILITY OF THE OREGON FOREST PRACTICES ACT	6
17.102.100	URBAN FORESTRY FUND	6

17.102.00 INTENT

- A. This chapter is intended to conserve and replenish the ecological, aesthetic and economic benefits of urban forests, by regulating tree removal on properties greater than one acre within the Sandy Urban Growth Boundary.
- B. This chapter is intended to facilitate planned urban development as prescribed by the Sandy Comprehensive Plan, through the appropriate location of harvest areas, landing and yarding areas, roads and drainage facilities.
- C. This chapter shall be construed in a manner consistent with Chapter 17.60 Flood and Slope Hazard Overlay District. In cases of conflict, Chapter 17.60 shall prevail.

17.102.10 DEFINITIONS

Technical terms used in this chapter are defined below. See also Chapter 17.10, Definitions.

Urban Forestry Related Definitions:

- **Diameter at Breast Height (DBH)**: The diameter of a tree inclusive of the bark measured 4½ feet above the ground on the uphill side of a tree.
- **Hazard Tree**: A tree located within required setback areas or a tree required to be retained as defined in 17.102.50 that is cracked, split, leaning, or physically damaged to the degree that it is likely to fall and injure persons or property. Hazard trees include diseased trees, meaning those trees with a disease of a nature that, without reasonable treatment or pruning, is likely to spread to adjacent trees and cause such adjacent trees to become diseased or hazard trees.
- **Protected Setback Areas**: Setback areas regulated by the Flood and Slope Hazard Ordinance (FSH), Chapter 17.60 and 70 feet from top of bank of Tickle Creek and 50 feet from top of bank of other perennial streams outside the city limits, within the urban growth boundary.
- Tree: For the purposes of this chapter, tree means any living, standing, woody plant having a trunk 11 inches DBH or greater.
- **Tree Protection Area**: The area reserved around a tree or group of trees in which no grading, access, stockpiling or other construction activity shall occur.

17.102 - 1

• **Tree Removal:** Tree removal means to cut down a tree, 11 inches DBH or greater, or remove 50 percent or more of the crown, trunk, or root system of a tree; or to damage a tree so as to cause the tree to decline and/or die. Tree removal includes topping but does not include normal trimming or pruning of trees.

17.102.20 APPLICABILITY

This chapter applies only to properties within the Sandy Urban Growth Boundary that are greater than one acre including contiguous parcels under the same ownership.

- A. General: No person shall cut, harvest, or remove trees 11 inches DBH or greater without first obtaining a permit and demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance, the applicant shall agree to implement required provisions of this chapter and to allow all inspections to be conducted.
 - 2. Tree removal is subject to the provisions of Chapter 15.44, Erosion Control, Chapter 17.56, Hillside Development, and Chapter 17.60 Flood and Slope Hazard.
- B. Exceptions: The following tree removals are exempt from the requirements of this chapter.
 - 1. Tree removal as required by the city or public utility for the installation or maintenance or repair of roads, utilities, or other structures.
 - 2. Tree removal to prevent an imminent threat to public health or safety, or prevent imminent threat to public or private property, or prevent an imminent threat of serious environmental degradation. In these circumstances, a Type I tree removal permit shall be applied for within seven days following the date of tree removal.

17.102.30 PROCEDURES AND APPLICATION REQUIREMENTS

A person who desires to remove trees shall first apply for and receive one of the following tree cutting permits before tree removal occurs:

A. Type I Permit. The following applications shall be reviewed under a Type I procedure:

- 1. Tree removal on sites within the city limits under contiguous ownership where 50 or fewer trees are requested to be removed.
- 2. Removal of a hazard tree or trees that presents an immediate danger of collapse and represents a clear and present danger to persons or property.
- 3. Removal of up to two trees per year, six inches DBH or greater within the FSH Overlay District as shown on the City Zoning Map and described in Chapter 17.60.
- 4. Tree removal on sites outside the city limits and within the urban growth boundary and outside protected setback areas.
- 5. Removal of up to two trees per year outside the city limits within the UGB and within protected setback areas.
- B. An application for a Type I Tree Removal permit shall be made upon forms prescribed by the City to contain the following information:

17.102 - 2

- 1. Two copies of a scaled site plan to contain the following information:
 - a. Dimensions of the property and parcel boundaries.
 - b. Location and species of trees 11" DBH or greater to be retained.
 - c. Location and type of tree protection measures to be installed.
- 2. A brief narrative describing the project.
- 3. Estimated starting and ending dates.
- 4. A scaled re-planting plan indicating ground cover type, species of trees to be planted, and general location of re-planting.
- 5. An application for removal of a hazard tree within a protected setback area or a tree required to be retained as defined in Chapter 17.102.50 shall also contain a report from a certified arborist or professional forester indicating that the condition or location of the tree presents a hazard or danger to persons or property and that such hazard or danger cannot reasonably be alleviated by treatment or pruning.
- C. Type II Permit. The following applications shall be reviewed under a Type II procedure:
 - 1. Tree removal on sites under contiguous ownership where greater than 50 trees are requested to be removed as further described below:
 - a. Within City Limits: outside of FSH Restricted Development Areas as defined in Chapter 17.60.
- D. An application for a Type II Permit shall contain the same information as required for a Type I permit above in addition to the following:
 - a. A list of property owners on mailing labels within 200 feet of the subject property.
 - b. A written narrative addressing permit review criteria in 17.102.40.
- E. Type III Permit. The following applications shall be reviewed under a Type III procedure:
 - 1. Request for a variance to tree retention requirements as specified in Section 17.102.50 may be permitted subject to the provisions of 17.102.70.
- F. An application for a Type III Permit shall contain the same information as required for a Type I permit in addition to the following:
 - a. A list of property owners on mailing labels within 300 feet of the subject property.
 - b. A written narrative addressing applicable code sections 17.102.50, 17.102.60, and 17.102.70.

17.102.40 PERMIT REVIEW

An application for a Type II or III tree removal permit shall demonstrate that the provisions of Chapter 17.102.50 are satisfied. The Planning Director may require a report from a certified arborist or professional forester to substantiate the criteria for a permit.

- A. The Director shall be responsible for interpreting the provisions of this chapter. The Director may consult with the Oregon Department of Forestry in interpreting applicable provisions of the Oregon Forest Practices Act (OAR Chapter 629). Copies of all forestry operation permit applications will be sent to the Oregon Department of Forestry and Department of Revenue. The City may request comments from the Oregon Department of Forestry, the Oregon Department of Fish & Wildlife or other affected state agencies.
- B. Expiration of Tree Removal Permits. Tree removal permits shall remain valid for a period of one year from the date of issuance or date of final decision by a hearing body, if applicable. A 30-day extension shall be automatically granted by the Planning Director if requested in writing before the expiration of the permit. Permits that have lapsed are void.

17.102.50 TREE RETENTION AND PROTECTION REQUIREMENTS

- A. Tree Retention: The landowner is responsible for retention and protection of trees required to be retained as specified below:
 - 1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous ownership.
 - 2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.
 - 3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.
 - 4. If possible, at least two of the required trees per acre must be of conifer species.
 - 5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements.
- **B.** Tree Protection Area: Except as otherwise determined by the Planning Director, all tree protection measures set forth in this section shall be instituted prior to any development activities and removed only after completion of all construction activity. Tree protection measures are required for land disturbing activities including but not limited to tree removal, clearing, grading, excavation, or demolition work.
 - 1. Trees identified for retention shall be marked with yellow flagging tape and protected by protective barrier fencing placed no less than 10 horizontal feet from the outside edge of the trunk.
 - 2. Required fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade.
 - 3. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles.

17.102 - 4

C. Inspection. The applicant shall not proceed with any tree removal or construction activity, except erosion control measures, until the City has inspected and approved the installation of tree protection measures. Within 15 days of the date of accepting an application for a Type I permit, the city shall complete an onsite inspection of proposed activities and issue or deny the permit. Within 15 days of is suing a Type II or Type III permit, the city shall complete an onsite inspection of proposed activities.

For ongoing forest operations, the permit holder shall notify the city by phone or in writing 24 hours prior to subsequent tree removal. The city may conduct an onsite re-inspection of permit conditions at this time.

17.102.60 TREE REPLANTING REQUIREMENTS

- 1. All areas with exposed soils resulting from tree removal shall be replanted with a ground cover of native species within 30 days of harvest during the active growing season, or by June 1st of the following spring.
- 2. All areas with exposed soils resulting from tree removal occurring between October 1 and March 31 shall also be covered with straw to minimize erosion.
- 3. Removal of hazard trees as defined shall be replanted with two native trees of quality nursery stock for every tree removed.
- 4. Tree Removal allowed within the FSH Overlay District shall be replanted with two native trees of quality nursery stock for every tree removed.
- 5. Tree Removal not associated with a development plan must be replanted following the provisions of OAR Chapter 629, Division 610, Section 020-060

17.102.70 VARIANCES

Under a Type III review process, the Planning Commission may allow newly-planted trees to substitute for retained trees if:

- 1. The substitution is at a ratio of at least two-to-one (i.e., at least two native quality nursery grown trees will be planted for every protected tree that is removed); and
- 2. The substitution more nearly meets the intent of this ordinance due to:
 - a. The location of the existing and proposed new trees, or
 - b. The physical condition of the existing trees or their compatibility with the existing soil and climate conditions; or
 - c. An undue hardship is caused by the requirement for retention of existing trees.
 - d. Tree removal is necessary to protect a scenic view corridor.

17.102.80 ENFORCEMENT

The provisions of Chapter 17.06, Enforcement, shall apply to tree removal that is not in conformance with this chapter. Each unauthorized tree removal shall be considered a separate offense for purposes of assigning penalties under Section 17.06.80. Funds generated as a result of enforcement of this ordinance shall be dedicated to the Urban Forestry Fund established under Section 17.102.100 below.

17.102.90 APPLICABILITY OF THE OREGON FOREST PRACTICES ACT

The following provisions of the Oregon Forest Practices Act (OAR Chapter 629) are adopted by

17.102 - 5

reference for consideration by the City in the review of Forest Operations Plans. Although the Director may seek advice from the Department of Forestry, the Director shall be responsible for interpreting the following provisions.

Division 610 - Reforestation Stocking Standards. Where reforestation is required, the provisions of OAR Chapter 629, Division 610, Section 020-060 shall be considered by the Director, in addition to the requirements of Section 17.102.60.

Division 615 - Treatment of Slash. Slash shall not be placed within the protected setback areas. Otherwise, the Director shall consider the provisions of OAR Chapter 629, Division 615 in determining how to dispose of slash.

Division 620 - Chemical and Other Petroleum Products Rules. The storage, transferring, cleaning of tanks and mixing of chemicals and petroleum products shall occur outside the protected setback areas. Aerial spraying shall not be permitted within the Urban Growth Boundary. Otherwise, the provisions of Chapter 629, Division 620 shall apply.

Division 625 - Road Construction and Maintenance. Forest roads, bridges and culverts shall not be constructed within the protected setback areas, except where permitted within the FSH overlay area as part of an approved urban development. Otherwise, the Director shall consider the provisions of OAR Chapter 629, Division 625 in the review of road, bridge and culvert construction.

Division 630 - Harvesting. Forest harvesting operations, including but not limited to skidding and yarding practices, construction of landings, construction of drainage systems, treatment of waste materials, storage and removal of slash, yarding and stream crossings, shall not be permitted within protected setback areas. Otherwise, the provisions of Chapter 629, Division 630 shall apply.

17.102.100 URBAN FORESTRY FUND CREATED

In order to encourage planting of trees, the City will create a fund or account to be used for tree planting in rights-of-way, city parks, riparian areas, and other public property. The source of funds will be donations, grants, and any other funds the City Council may designate.

Adopted November 18, 2002 Ordinance 2002-10

17.102 - 6



Exhibit U

SUBJECT: File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision

AGENDA DATE: February 25, 2019

DEPARTMENT: Planning Division

Application Submitted: June 25, 2018 Additional Submittal Items: December 5, 2018 Application Complete: December 27, 2018 120-Day Deadline: April 27, 2019

STAFF CONTACT: Emily Meharg, Associate Planner

EXHIBITS:

Applicant's Submittals

- A. Land Use Application Form
- B. Narrative
- C. Project Plan Set
 - Sheet 1: Cover Sheet and Preliminary Plat Map
 - Sheet 2: Existing Conditions and Tree Retention Inventory
 - Sheet 3: Existing Trees with Building Setbacks Map
 - Sheet 4: FSH and Slopes Map
 - Sheet 5: Preliminary Street and Utility Plan
 - Sheet 6: Preliminary Street Tree Plan
 - Sheet 7: Preliminary Parking Plan
 - Sheet 8: Future Street and Pedestrian Trail Plan
 - Sheet 9: Grading and Erosion Control Plan
 - Sheet 10: Cut and Fill Evaluation
- D. Code Interpretation Request
- E. Storm Drainage Design and Calculations for the Jacoby Heights Subdivision
- F. Traffic Impact Analysis
- G. Geotechnical Report
- H. Wetland Determination
- I. Easement between Cory Knight and Ian Bird and Kelly Bird
- J. Letter from Public Works Director Re: Pump Station Capacity
- K. Arborist Report

Agency Comments

- L. City Engineer (January 15, 2019)
- M. Transportation Engineer (January 18, 2019)
- N. PGE (January 23, 2019)
- O. Public Works Director (January 26, 2019)
- P. Transit Director (January 28, 2019)

Public Comments

- Q. Annette Giltner, 35000 SE Colorado Road (February 6, 2019)
- R. Shawn Lorenzen, 39343 Stratford Street (February 11, 2019)
- S. Tim and Cherri Anders, 19098 Jacoby Road (February 12, 2019)

Additional Documents Submitted by Staff

T. Chapter 17.102 of the City of Sandy Development Code

I. BACKGROUND

A. PROCEEDING

Type III Subdivision, Type III Code Interpretation, Type III FSH review, Type III Variance for not orienting towards a transit street, Type III Variance for not providing two tiered lot configuration, Type III Variance to Block Length, and Type II Tree Removal Permit. Note: the proposal includes a newly mapped FSH overlay area. The zone change associated with the new FSH overlay will be processed separately by the City.

B. FACTUAL INFORMATION

- 1. APPLICANT/OWNER: Cory Knight
- 2. PROJECT NAME: Jacoby Heights Subdivision
- 3. LEGAL DESCRIPTION: T2S R4E Section 24A Tax Lot 2300
- 4. SITUS ADDRESS: 19124 Jacoby Road
- 5. PROPERTY LOCATION: East of Jacoby Road, west of Langensand Road, and south of Cascadia Village Drive
- 6. PROPERTY SIZE: 9.68 acres
- 7. PROPOSAL: All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards All Zones, and Chapter 17.102, Urban Forestry.
- 8. COMPREHENSIVE PLAN DESIGNATION: Low Density Residential
- 9. ZONING DISTRICT DESIGNATION: SFR, Single Family Residential
- 10. SERVICE CONSIDERATIONS: The applicant proposes to extend the existing sanitary sewer in Jacoby Road to service the entire subdivision. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. The stormwater for the subdivision would be collected and conveyed to a proposed detention pond located in Tract A at the northwest corner of the site.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

11. RESPONSE FROM GOVERNMENTAL AGENCIES, UTILITY PROVIDERS, AND CITY DEPARTMENTS:

- a. City Manager No comments received
- b. City Engineer– Exhibit L
- c. Transportation Engineer Exhibit M
- d. Public Works Director Exhibit O
- e. Transit Director Exhibit P
- f. PGE Exhibit N
- g. ODOT No comments received
- h. SandyNet Manager No comments received
- i. Police Department No comments received
- j. Fire District No. 72 No comments received
- k. US Postal Service No comments received

C. PUBLIC COMMENTS

Three written comments were received as follows:

- Annette Giltner at 35000 SE Colorado Road (Exhibit Q) does not want to see more development in Sandy.
- Shawn Lorenzen at 39343 Stratford Street (Exhibit R) does not want to see more
 residential units built in Sandy without first strengthening the City's infrastructure to
 support existing residents.
- Tim and Cherri Anders at 19098 Jacoby Road (Exhibit S) expressed concerns about the effect of the proposed subdivision on their property, which is directly adjacent to the north.

D. APPLICABLE CRITERIA:

Sandy Municipal Code: 17.12 Procedures for Decision Making; 17.14 Request for Interpretation; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 Single Family Residential (SFR); 17.56 Hillside Development; 17.60 Flood and Slope Hazard Overlay District; 17.66 Adjustments and Variances; 17.80 Additional Setbacks on Collector and Arterial Streets; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.86 Parkland and Open Space; 17.92 Landscaping and Screening; 17.98 Parking, Loading, and Access; 17.100 Land Division; 17.102 Urban Forestry; 15.30 Dark Sky Ordinance; and 15.44 Erosion Control.

E. BACKGROUND INFORMATION/HISTORY

There is one existing house at 19124 Jacoby Road. The applicant shall obtain a demolition permit from the City of Sandy Building Department prior to demolition of the existing structure(s) on-site. Per the City Engineer (Exhibit L), the preliminary plans don't identify any existing domestic or irrigation wells on site. Any existing domestic or irrigation wells on site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City Planning Division. Per the City Engineer (Exhibit L), the preliminary plans don't identify any septic tanks on site. Any on-site sewage disposal system shall be abandoned in conformance with Clackamas County Water Environmental

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City Planning Division.

F. PROCEDURAL CONSIDERATIONS

This request is being processed as a Type III Subdivision, Type III FSH Review, three Type III Variances, Type III Code Interpretation, and Type II Tree Removal Permit. Notification of the proposal was mailed to property owners within 300 feet of the subject property and to affected agencies on January 29, 2019. A legal notice was published in the Sandy Post on February 6, 2019.

II. ANALYSIS OF CODE COMPLIANCE

CHAPTER 17.14 – REQUEST FOR INTERPRETATION

Request: The applicant is requesting an interpretation regarding whether Chapter 17.92, Landscaping and Screening, applies to subdivisions.

RESPONSE: Staff reviewed the Development Code and relevant ordinances and consulted with the City Attorney and found nothing indicating that Chapter 17.92 does not apply to subdivisions. In fact, the full title of Chapter 17.92 is Landscaping and Screening General Standards – All Zones. Staff interprets this as meaning that Chapter 17.92 applies to all zones, including Single Family Residential. The specific general provision that the applicant is questioning (Section 17.92.10(C)) states that significant trees should be integrated into the design of a development and that trees to be retained shall be protected from damage during construction. Staff finds that a subdivision qualifies as a development and as construction. Furthermore, there is nothing contained in Chapter 17.92 or Chapter 17.102 that restricts using Chapter 17.92 during subdivision review. There is also nothing in the ordinance (Ordinance No. 2002-10) that adopted Chapter 17.102 that speaks on this matter. In addition, there are several Comprehensive Plan Goal 5 policies that support retaining significant trees, most notably, the following:

- Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.
- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.
- Policy 31: The City of Sandy shall undertake a study to designate "significant/heritage" trees or areas of second growth that exist within the City's urban growth boundary. Designated trees shall be protected and preserved where possible.

Recommendation: Staff recommends that the Planning Commission continue to apply Chapter 17.92 to all zones, including during subdivision review.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

CHAPTER 17.30 – ZONING DISTRICT

The Single Family Residential (SFR) zoning on the site allows a minimum of 3 units per acre and a maximum of 5.8 units per acre. The number of dwelling units permitted on a parcel of land is calculated after determining the amount of restricted development area in accordance with Section 17.60, the Flood and Slope Hazard Overlay District. Density transfer is permitted from restricted development areas consistent with the provisions of Section 17.60.90, although no density transfer is proposed.

RESPONSE: The total gross acreage for the entire property is 9.68 acres. The proposal contains 2.08 acres of area dedicated for public right-of-way and 0.47 acres dedicated for public tracts (Tract A). After removal of the right-of-way and public tracts the net site area for the subject property is reduced to 7.13 acres of net site area (NSA). The subject property also contains 1.20 acres of restricted development area; therefore, the unrestricted site area (USA) is 5.93 acres. Based on required density, the SFR land requires a minimum of 18 dwelling units (5.93 USA x 3). The maximum allowed dwelling units is 41 (calculated as the lesser of (a) 7.13 NSA x 5.8 = 41 and (b) 5.93 USA x 5.8 x 1.5 = 52). The proposed 32 dwelling units are within the allowable density range and therefore meet the density requirement.

CHAPTER 17.34 – SINGLE FAMILY RESIDENTIAL (SFR)

The applicant proposes 32 single family detached dwellings in conformance with minimum and maximum density requirements, as detailed above in Chapter 17.30.

17.34.10 PERMITTED USES

RESPONSE: Section 17.34.10 lists single family detached dwellings as a permitted use. The proposed subdivision includes 32 lots for single family detached dwellings. All homes shall provide building design features in compliance with the standards in Section 17.90.150 and Chapter 17.54.

17.34.30 DEVELOPMENT STANDARDS

Туре	Standard
A. Minimum Lot Area - Single detached dwelling	7,500 square ft.
- Other permitted uses	No minimum
A. Minimum Average Lot Width	
- Single detached dwelling	60 ft.
C. Minimum Lot Frontage	20 ft. except as allowed by Section
	17.100.160
D. Minimum Average Lot Depth	No minimum
E. Setbacks (Main Building)	
Front yard	10 ft. minimum
Rear yard	20 ft. minimum
Side yard (interior)	7.5 ft. minimum
Corner Lot	10 ft. minimum on side abutting the street ¹
F. Setbacks (Garage/Carport)	22 ft. minimum for front vehicle access

¹ Must comply with clear vision requirements of Chapter 17.74.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

	15 ft. minimum if entrance is perpendicular to street (subject to Section 17.90.220)5 ft. minimum for alley or rear access
G. Projections into Required Setbacks	See Chapter 17.74
H. Accessory Structures in Required Setbacks	See Chapter 17.74
I. Structure Height	35 ft. maximum
J. Building Site Coverage	No minimum
K. Off-Street Parking	See Chapter 17.98

RESPONSE: The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes shall meet the development standards of Section 17.34.30.

17.34.40 MINIMUM REQUIREMENTS

- A. Must connect to municipal water.
- B. Must connect to municipal sewer if service is currently within 200 hundred feet of the site. Sites more than 200 feet from municipal sewer, may be approved to connect to an alternative disposal system provided all of the following are satisfied:
 - 1. A county septic permit is to be secured and a copy is provided to the city.
 - 2. The property owner executes a waiver of remonstrance to a local improvement district and/or signs a deed restriction agreeing to complete improvements.
 - 3. The minimum size of the property is one acre or is a pre-existing buildable lot, as determined by the city.
 - 4. Site consists of a buildable parcel(s) created through dividing property in the city, which is less than five acres in size.
- C. The location of any real improvements to the property must provide for a future street network to be developed.
- D. Must have frontage or approved access to public streets.

RESPONSE: All lots will be required to connect to City services. The applicant is also required to extend utilities to the furthest extent of the subject property.

CHAPTER 17.56 – HILLSIDE DEVELOPMENT

17.56.10 APPLICABILITY

These regulations shall apply to any parcel with slopes greater than twenty-five percent (25%) as-shown on the Hillside Development Overlay District Map or with slope hazards mapped by the Department of Geology and Mineral Industries (DOGAMI). This chapter shall apply only to activities and uses that require a building, grading, tree removal and/or land use permit.

- A. <u>General</u>. No person shall develop property in areas designated by SDC 17.56.10, without first demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance or land use approval, the applicant shall agree to implement the recommendations of approved studies and to allow all inspections to be conducted.
 - 2. Where a bond, letter of credit or other guarantee is required, the permit shall not be issued until the bond or guarantee has been obtained and approved.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

B. Exemptions:

- 1. An activity or use that avoids slopes of 25% or greater, DOGAMI slope hazard areas, natural drainageways and potentially hazardous analysis areas as defined in Section 17.56.30.A.
- 2. The following activities, regardless of location:
 - a. An excavation that is less than three feet in depth, or which involves less than fifty cubic yards of volume;
 - b. A fill that does not exceed three feet in depth or 50 cubic yards of volume;
 - c. New construction or expansion of a structure resulting in a net increase in ground floor area of less than 1,000 square feet that does not involve grading;
 - d. Emergency actions required to prevent an imminent threat to public health or safety, or prevent imminent danger to public or private property, as determined by the Director; or
 - e. Any land use or activity that does not require a building, grading permit, or land use approval.

RESPONSE: The applicant submitted an FSH and Slopes Map (Exhibit C, Sheet 4) that shows areas of land that exceed 25 percent slope. A majority of the 25 percent or greater slope areas are located in the northeast corner of the property, adjacent to the Tickle Creek basin. The applicant submitted a Geotechnical Report (Exhibit G) by Rapid Soil Solutions LLC that includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations, which was reviewed by the City Engineer. Per the City Engineer (Exhibit L), the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC. The submitted Cut and Fill Evaluation (Exhibit C, Sheet 10) shows proposed cut and fill within the tree protection area south of Trees #419, #523, and #526. The applicant shall update the cut and fill evaluation to include the location of the tree protection fencing as depicted on the existing conditions and tree retention inventory (Exhibit C, Sheet 2). The applicant shall not cut or fill within the proposed tree protection area.

17.56.30 REQUIRED MAP AND STUDIES

- A. Topographic Map Required. To determine the location of potentially hazardous areas, the applicant shall submit a scaled topographic map at two-foot contour intervals for the subject property (site) and for land within 25 feet of the site perimeter. In addition to DOGAMI slide hazard areas and slopes of 25% or greater, potentially hazardous "analysis areas" include land within 25 feet of the top or toe of slopes of 25% or greater and the area 25 feet on either side of drainageways that drain 20 acres or more. This map shall be prepared by a registered engineer or land surveyor and shall show:
 - 1. Slopes of 25-34%;
 - 2. Slopes of 35% and greater;
 - 3. The analysis that is within 25' of slopes that are 25% or greater parallel to and within 25 feet of the top of the 25% slope break;
 - 4. Mapped DOGAMI slide hazard areas;
 - 5. The analysis area within 25 feet of the centerline of drainageways that drain at least 20 acres; and

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- 6. The area (in square feet) for each category listed above for the subject property.
- B. <u>Types of Required Studies</u>. There are three types of geological and engineering studies that may be required by this chapter. See Table 1 under Section 17.56.40, below.
 - <u>Geological Assessments</u> are prepared and stamped by a Certified Engineering Geologist and describe the surface and subsurface conditions of a site, delineate areas of a property that may be subject to specific geologic hazards, and assess the suitability of the site for development. Geotechnical Reports shall be conducted according to the requirements of Appendix A (*Geological Assessments*), shall make recommendations as to whether further studies are required, and may be incorporated into or included as an appendix to the geotechnical report.
 - 2. <u>Engineering Geology Reports</u> are prepared and stamped by a Certified Engineering Geologist and provide detailed descriptions of the geology of the site, professional conclusions and recommendations regarding the effect of geological conditions on the proposed development, and opinions and recommendations covering the adequacy of the site to be developed. Engineering Geology Reports shall be prepared in accordance with the requirements of Appendix B (*Guidelines for Preparing Engineering Geology Reports in Oregon* adopted by the Oregon State Board of Geologist Examiners) and may be incorporated into or included as an appendix to the geotechnical report.
 - 3. <u>Geotechnical Reports</u> are prepared and stamped by a Geotechnical Engineer, evaluate site conditions, and recommend design measures necessary to reduce the development risks and facilitate safe and stable development. Geotechnical Reports shall be conducted according to the requirements of Appendix C (Geotechnical Reports), and may be incorporated into or included as an appendix to the Engineering Geology Report.

RESPONSE: Rapid Soil Solutions LLC completed a Geotechnical Report (Exhibit G) dated February 23, 2018 for the subject site. **The applicant shall follow the recommendations outlined in the Geotechnical Report.**

17.56.40 WHERE STUDIES REQUIRED

Additional geological or engineering studies shall be required, or not required, under the following circumstances:

Situation	Type I Development Applications; Single Family Homes, Duplexes and Accessory Uses	NON-EXEMPT Grading; Type II or III Develop- ment Applications
A. Proposed development avoids slopes of 25% or greater, drainageways, DOGAMI slope hazard areas and all analysis areas:	1. No further requirements	2. No further requirements
B. Development proposed on slopes of 25-35% or analysis areas, but avoids drainageways, DOGAMI hazard areas and slopes of	1. Geological Assessment required; Engineering Geology or Geotechnical Reports may be required*	2. Engineering Geology Report required; Geotechnical Report may be required*

TABLE 1: WHERE STUDIES ARE REQUIRED OR NOT REQUIRED

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

35% or greater:		
C. Development proposed	1. Engineering Geology	2. Engineering Geology
on DOGAMI hazard areas,	Report required;	Report and Geotechnical
slopes of 35% or greater,	Geotechnical Report may be	Report required
or drainageway areas:	required*	

* Whether additional studies are necessary depends on recommendations of base required study.

RESPONSE: The applicant shall submit a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay. This includes Lots 19, 20, 21, 27, 28, and 29. Additional reports may be required depending on the recommendations of the geological assessment. Once the subdivision is platted and building permits are filed for individual home construction the property owner/contractor shall submit Type I Hillside applications with studies as detailed in Table 1 in Section 17.56.40.

17.56.50 COMPLIANCE WITH STUDY CONCLUSIONS AND RECOMMENDATIONS REQUIRED

- A. <u>Professional Standards</u>. The director shall determine whether Geological Assessments, Engineering Geology Reports, or Geotechnical Reports have been prepared in accordance with Section 17.56.30. The director may require additional information or analysis necessary to meet study requirements.
- B. <u>Peer Review</u>. The director may require peer review of any required report, in which case regulated activities and uses shall be reviewed and accepted through the peer review process before any regulated activity will be allowed.
 - 1. A professional or professional firm of the city's choice that meets the qualifications listed in this chapter shall perform the review.
 - 2. The review shall be at the applicant's expense.
 - 3. Review of report submittals shall determine whether required elements are completed, geologic report procedures and assumptions are accepted, and all conclusions and recommendations are supported and reasonable.
- C. <u>Review Criteria</u>. The approval authority shall rely on the conclusions and recommendations of required reports, as modified by peer review, to determine compliance with this chapter.
- D. <u>Conditions of Approval</u>. Conclusions and recommendations stated in approved reports shall be directly incorporated as permit conditions or provide the basis for conditions of approval for the regulated activity or use.
- E. <u>Expiration</u>. Where an approved assessment or report as defined and required by this chapter has been prepared within the last five years for a specific site, and where the proposed land use activity and surrounding site conditions are unchanged, that report may be utilized and a new report is not required. Should environmental conditions associated with the site or surrounding the site change, or if the proposed land use activity or development has materially changed, the applicant shall submit an amendment to the required assessment or report, which may be reviewed and approved through the peer review process.

RESPONSE: The Geotechnical Report includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations. **The applicant shall retain**

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.

CHAPTER 17.60 - FLOOD AND SLOPE HAZARD OVERLAY DISTRICT

17.60.00 INTENT

This chapter is intended to promote the public health, safety and general welfare by minimizing public and private adverse impacts from flooding, erosion, landslides or degradation of water quality consistent with Statewide Planning Goals 6 (Air, Land and Water Resources Quality) and 7 (Areas Subject to Natural Disasters and Hazards) and the Sandy Comprehensive Plan (SCP).

RESPONSE: The applicant is not proposing any development within the Flood and Slope Hazard (FSH) overlay district. However, the proposal includes 6 lots that that are proposed to be platted within the FSH overlay. The applicant shall place the FSH analysis area in a separate tract and shall dedicate the FSH tract to the City. Staff brought the subject subdivision proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board recommends that a tract is created and dedicated to the City and cited three primary reasons that they would like to see the land dedicated: 1) increased protection of the trees located within the FSH buffer area; 2) a larger area through which a connector trail to Tickle Creek Trail could meander; and, 3) to provide a wider riparian buffer that would not only better protect Tickle Creek, but would also enhance the natural beauty along the future extension of the Tickle Creek Trail. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the FSH tract without knowing the existing condition of that area. The applicant shall submit a Phase I environmental assessment and a native vegetation report for the FSH area to be dedicated in the tract. The applicant shall remove all invasive plants and re-plant any areas of exposed soil with native plants. The applicant shall dedicate the FSH tract to the City and shall submit a maintenance bond for two full growing seasons (May 1 - October31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. The proposal also contains two trees that are proposed for removal and appear to be on the FSH boundary line: Trees #510 and #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.

17.60.30 REQUIRED SETBACK AREAS

Setback areas shall be required to protect water quality and maintain slope stability near stream corridors and locally significant wetlands. Setbacks are measured horizontally from, parallel to and upland from the protected feature.

A. <u>Required Setbacks</u>. The required special setback(s) shall be:

- 1. 70 feet from the top of bank of Tickle Creek;
- 2. 50 feet from top of bank along other perennial streams, except for "No Name Creek" east of Towle Drive, as provided in Section 17.60.30.C.2 below.
- 3. 25 feet around the edge of any mapped locally significant wetland; and

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

4. 25 feet from the top of any 25% slope break where the slope break occurs within the FSH overlay district as mapped by the city.

RESPONSE: The applicant submitted a Wetland Determination (Exhibit H) by Environmental Science & Assessment, LLC dated May 30, 2018. The wetland determination concluded there is no evidence of intermittent or perennial flow on the site. Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). The wetland report/delineation shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife. The City Engineer also recommends that the City of Sandy Planning Division review the FSH Overlay District line at the northeast corner of the site.

17.60.50 SPECIAL REPORTS

Where development is proposed on restricted development areas within the FSH overlay district as defined in Section 17.60.20.A, the Director shall require submission of the following special reports. These reports shall be in addition to other information required for specific types of development, and shall be prepared by professionals in their respective fields.

The Director may require one of more of these reports where necessary to address potential adverse impacts from development on buildable land within the FSH overlay district. The Director may exempt Type II permit applications from one or more of these reports where impacts are minimal and the exemption is consistent with the purpose of the FSH overlay zone as stated in Section 17.60.00.

- A. <u>Hydrology and Soils Report</u>. This report shall include information on the hydrological conditions on the site, the effect of hydrologic conditions on the proposed development, the proposed development's impact on surface and groundwater flows to wetlands and streams, and any hydrological or erosion hazards. This report shall also include soils characteristics of the site, their suitability for development, carrying capacity, and erosion or slumping characteristics that might present a hazard to life and property, or adversely affect the use or stability of a public facility or utility. Finally, this report shall include information on the nature, distribution and strength of existing soils; the adequacy of the site for development purposes; and an assessment of grading procedures required to impose the minimum disturbance to the natural state. A licensed professional engineer registered in Oregon shall prepare the hydrology and soils report.
- B. <u>Grading Plan.</u> The grading plan shall be specific to a proposed physical structure or use and shall include information on terrain (two-foot intervals of property), drainage, direction of drainage flow, location of proposed structures and existing structures which may be affected by the proposed grading operations, water quality facilities, finished contours or elevations, including all cut and fill slopes and proposed drainage channels. Project designs including but not limited to locations of surface and subsurface devices, walls, dams, sediment basins, storage reservoirs, and other protective devices shall form part of the submission. The grading plan shall also include: 1) construction phase erosion control plan consistent with the provisions of Chapter 15.44; and 2) schedule of operations. A licensed professional engineer registered in Oregon shall prepare the grading and erosion control plan.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

C. <u>Native Vegetation Report</u>. This report shall consist of a survey of existing vegetative cover, whether it is native or introduced, and how it will be altered by the proposed development. Measures for revegetation with native plant species will be clearly stated, as well as methods for immediate and long-term stabilization of slopes and control of soil erosion. A landscape architect, landscape designer, botanist or arborist with specific knowledge of native plant species, planting and maintenance methods, survival rates, and their ability to control erosion and sedimentation shall prepare the vegetation report. The applicant shall be responsible for replacing any native plant species that do not survive the first two years after planting, and for ensuring the survival of any replacement plants for an additional two years after their replacement.

RESPONSE: Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). The wetland report/delineation shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife.

17.60.80 WATER QUALITY TREATMENT FACILITIES

Tickle Creek, the Sandy River and associated natural drainage ways are vital to Sandy's recreationally based economy and to the quality of life of Sandy residents. Placement of water quality facilities shall be limited as follows:

- A. The water quality facility shall not be constructed in restricted development areas, except where necessary to serve approved development within restricted development areas (e.g., a road) and where no reasonable alternative exists in buildable areas of the site.
- B. Where the approval authority determines that a more efficient and effective regional site exists within the sub-basin, the water quality facility may be constructed off-site.

RESPONSE: The proposed detention pond (Tract A) is located in the northwest corner of the site and is not within the mapped FSH overlay area.

17.60.90 DENSITY TRANSFER PROVISIONS

Residential density transfer may be approved subject to the following:

- A. Required Setback Areas. Density may be transferred from restricted development areas (i.e., steep slopes, protected water features and required setbacks) to buildable portions of the site.
- B. Density Maximum. The maximum gross density for the buildable area of the site shall not exceed 150% of the maximum density allowed by the underlying zoning district for that buildable area.
- C. Housing Types Not Permitted in Underlying Zoning District. Housing types not permitted in the underlying zoning district may only be approved through the PD (planned development) or SAP (specific area plan) process.
- D. Transfer Area. Transfer of density may only occur within the same property and/or to properties contiguous to the primary property. The terms "primary property" identify the legal lot from which density is to be transferred to "secondary property(s)". Further development or land use action on the primary or secondary properties shall be reviewed together in the same application.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

RESPONSE: Since no development is proposed within the FSH area, a density transfer per Section 17.60.90 is allowed for the buildable area of the site. There are 1.20 acres of restricted development area. See Chapter 17.30 for density analysis.

CHAPTER 17.80 - ADDITIONAL SETBACKS ON COLLECTOR AND ARTERIAL STREETS

17.80.10 APPLICABILITY

Collector: Jacoby Road

RESPONSE: Four lots (Lots 9, 10, 30 and 32) are proposed to contain frontage on Jacoby Road, which is classified as a collector street. All structures on lots along Jacoby Road shall be constructed to comply with the standards of Section 17.80.20.

17.80.20 SPECIFIC SETBACKS

Any structure located on streets listed above or identified in the Transportation System Plan as arterials or collectors shall have a minimum setback of 20 feet measured from the property line. This applies to applicable front, rear and side yards.

RESPONSE: All structures shall maintain a minimum 20-foot setback from the Jacoby Road public right-of-way. The Preliminary Plat (Exhibit C, Sheet 1) depicts building envelopes at 20 feet from the Jacoby Road right-of-way.

CHAPTER 17.82 – SPECIAL SETBACKS ON TRANSIT STREETS

17.82.20 BUILDING ORIENTATION

- A. All buildings shall have their primary entrances oriented toward a transit street rather than a parking area, or if not adjacent to a transit street, toward a public right-of-way or private walkway which leads to a transit street.
- B. Buildings shall have a primary entrance connecting directly between the street and building interior. This entrance shall be open to the public during all business hours and shall comply with the accessibility standards of the Uniform Building Code.
- C. In lieu of a building entrance oriented to a transit street, a building's entrance may be enhanced and identified in the following manner:
 - 1. An entrance plaza of at least 150 square feet, at least 100 square feet of which shall be visible from the transit street. The entrance plaza shall be at least 10 feet wide at the narrowest dimension; and
 - 2. A permanent building feature (e.g. a portico, porch or awning) shall be visible from the transit street, signifying an entrance; and
 - 3. Pedestrian-scale lighting shall be required at the entrance; and
 - 4. A clearly marked, convenient, safe and lighted pedestrian route shall be provided to the entrance, from the transit street.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- D. Primary building entrances shall be architecturally emphasized and visible from the street. Building entrances shall incorporate arcades, roofs, porches, alcoves, porticoes, and awnings that protect pedestrians from the rain and sun. Continuous arcades are strongly encouraged.
- E. All building entrances and exits shall be well lit. Lighting shall be a pedestrian scale (3'-12') and the source light shall be shielded to reduce glare.
- F. For commercial buildings with facades over 300 feet in length on a transit street, two or more building entrances on the street must be provided.
- G. If the site has frontage on more than one transit street, the building shall provide one main entrance oriented to a transit street or to a corner where two transit streets intersect.

RESPONSE: Jacoby Road is a designated transit street. Lots 9, 10, 30, and 32 have frontage on Jacoby Road. As part of the Jacoby Heights subdivision proposal, the applicant has requested a variance to the requirements of Section 17.82.20 to allow the front door for the houses on Lots 9, 10, 30, and 32 to face the internal street network instead of Jacoby Road, which is a designated transit street. The variance request is discussed in detail in the analysis of Chapter 17.66 in this staff report.

CHAPTER 17.84 - IMPROVEMENTS REQUIRED WITH DEVELOPMENT

This chapter provides general information regarding improvements required in association with development, and it clarifies the timing, extent, and standards for public and private improvements.

17.84.20 TIMING OF IMPROVEMENTS

- A. All improvements required by the standards in this chapter shall be installed concurrently with development, as follows:
 - 1. Where a land division is proposed, each proposed lot shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to approval of the final plat.
 - 2. Where a land division is not proposed, the site shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to temporary or final occupancy of structures.

RESPONSE: Submission of preliminary street and utility plans during the land use review process is solely for compliance with the data requirements of Section 17.100.60 (D). **Public improvement** plans are subject to a separate review and approval process. Preliminary plat approval does not connote approval of public improvement construction plans.

B. Where specific approval for a phasing plan has been granted for a planned development and/or subdivision, improvements may similarly be phased in accordance with that plan.

RESPONSE: The applicant did not propose a phasing plan with this application.

17.84.30 PEDESTRIAN AND BICYCLIST IMPROVEMENTS

A. Sidewalks shall be required along both sides of all arterial, collector, and local streets, as follows:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- 1. Sidewalks shall be a minimum of 5 ft. wide on local streets. The sidewalks shall be separated from curbs by a tree planting area that provides separation between sidewalk and curb, unless modified in accordance with Subsection 3 below.
- 2. Sidewalks along arterial and collector streets shall be separated from curbs with a planting area, except as necessary to continue an existing curb-tight sidewalk. The planting area shall be landscaped with trees and plant materials approved by the City. The sidewalks shall be a minimum of 6 ft. wide.
- 3. Sidewalk improvements shall be made according to city standards, unless the city determines that the public benefit in the particular case does not warrant imposing a severe adverse impact to a natural or other significant feature such as requiring removal of a mature tree, requiring undue grading, or requiring modification to an existing building. Any exceptions to the standards shall generally be in the following order.
 - a) Narrow landscape strips
 - b) Narrow sidewalk or portion of sidewalk to no less than 4 feet in width
 - c) Eliminate landscape strips
 - d) Narrow on-street improvements by eliminating on-street parking
 - e) Eliminate sidewalks
- 4. The timing of the installation of sidewalks shall be as follows:
 - a) Sidewalks and planted areas along arterial and collector streets shall be installed with street improvements, or with development of the site if street improvements are deferred.
 - b) Sidewalks along local streets shall be installed in conjunction with development of the site, generally with building permits, except as noted in (c) below.
 - c) Where sidewalks on local streets abut common areas, drainageways, or other publicly owned or semi-publicly owned areas, the sidewalks and planted areas shall be installed with street improvements.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) depicts an 8-foot-wide public utility easement along the frontage of all proposed lots to facilitate future installation of utilities. Sidewalks abutting the proposed lots shall be constructed in association with development of the lots. **The applicant shall construct sidewalks along Tract A both on Jacoby Road and Camden Court, prior** to final plat approval. The sidewalks on Camden Court, Woodstock Street, Newton Street, and Middlebury Avenue shall be five feet in width and separated by a planter strip or swale (six feet in width) in areas not transverse by driveways. The sidewalks on Jacoby Road shall be six feet in width and separated by a five foot planter strip.

The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all streets. The section of Newton Street on the east side of the property adjacent to Lots 1 and 2 is proposed to be constructed with half street improvements. This proposed design places the utilities underneath the planter strip; thus, the proposal does not show street trees in front of Lots 1 and 2. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. Street trees shall be installed approximately 30 feet on center in conjunction with issuance of building permits. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted and staked per the City of Sandy standard planting detail; trees shall be tied to the stakes with loosely tied twine. The planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed within one year of installation.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- B. Safe and convenient pedestrian and bicyclist facilities that strive to minimize travel distance to the extent practicable shall be provided in conjunction with new development within and between new subdivisions, planned developments, commercial developments, industrial areas, residential areas, public transit stops, school transit stops, and neighborhood activity centers such as schools and parks, as follows:
 - 1. For the purposes of this section, "safe and convenient" means pedestrian and bicyclist facilities that: are reasonably free from hazards which would interfere with or discourage travel for short trips; provide a direct route of travel between destinations; and meet the travel needs of pedestrians and bicyclists considering destination and length of trip.
 - 2. To meet the intent of "B" above, right-of-ways connecting cul-de-sacs or passing through unusually long or oddly shaped blocks shall be a minimum of 15 ft. wide with 8 feet of pavement.
 - 3. 12 feet wide pathways shall be provided in areas with high bicycle volumes or multiple use by bicyclists, pedestrians, and joggers.
 - 4. Pathways and sidewalks shall be encouraged in new developments by clustering buildings or constructing convenient pedestrian ways. Pedestrian walkways shall be provided in accordance with the following standards:
 - a) The pedestrian circulation system shall be at least five feet in width and shall connect the sidewalk on each abutting street to the main entrance of the primary structure on the site to minimize out of direction pedestrian travel.
 - b) Walkways at least five feet in width shall be provided to connect the pedestrian circulation system with existing or planned pedestrian facilities which abut the site but are not adjacent to the streets abutting the site.
 - c) Walkways shall be as direct as possible and avoid unnecessary meandering.
 - d) Walkway/driveway crossings shall be minimized. Internal parking lot design shall maintain ease of access for pedestrians from abutting streets, pedestrian facilities, and transit stops.
 - e) With the exception of walkway/driveway crossings, walkways shall be separated from vehicle parking or vehicle maneuvering areas by grade, different paving material, painted crosshatching or landscaping. They shall be constructed in accordance with the sidewalk standards adopted by the City. (This provision does not require a separated walkway system to collect drivers and passengers from cars that have parked on site unless an unusual parking lot hazard exists).
 - f) Pedestrian amenities such as covered walk-ways, awnings, visual corridors and benches will be encouraged. For every two benches provided, the minimum parking requirements will be reduced by one, up to a maximum of four benches per site. Benches shall have direct access to the circulation system.

RESPONSE: Five foot wide sidewalks are proposed on local streets and six foot wide sidewalks are proposed along Jacoby Road adjacent to the proposed subdivision. These sidewalks provide safe and convenient pedestrian access within the subdivision and connect the site to neighboring properties. To improve pedestrian connectivity, the applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west.

C. Where a development site is traversed by or adjacent to a future trail linkage identified within the Transportation System Plan, improvement of the trail linkage shall occur concurrent with development. Dedication of the trail to the City shall be provided in accordance with 17.84.80.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

RESPONSE: The City's current TSP maps were created with the former UGB boundaries (pre-June 2017) and did not include the subject property that was brought into the revised UGB boundaries. Therefore, there are no trail linkages identified in the TSP for this property.

E. To ensure improved access between a development site and an existing developed facility such as a commercial center, school, park, or trail system, the Planning Commission or Director may require off-site pedestrian facility improvements concurrent with development.

RESPONSE: Tickle Creek crosses the northeast corner of the subject property and the future extension of the Tickle Creek Trail is planned to extend along the length of Tickle Creek. At the preapplication meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trail on the adjacent property to the east (tax lot 800). Staff brought this proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board also recommends that the connector trail be constructed as part of this development. Staff recommends the Planning Commission require a pedestrian trail connection to the future extension of the Tickle Creek Trail as part of this development. Staff recommends that the trail connection be located in alignment with Middlebury Avenue. Alternatively, staff recommends that Lot 19 be included in the FSH tract and that the trail connection be located on the proposed location of Lot 19. Preserving Lot 19 as a tract would preserve trees, further enhance the Tickle Creek basin, and provide a wide trail connection for switch backs or stairs. If the applicant wants to pursue locating the pedestrian trail connection off-site, the applicant shall provide justification as to why a connection on tax lot 800 makes more sense than a connection developed as part of this proposal. Based on an initial review of topography (using Lidar data), it appears that the applicant's proposed trail connection on tax lot 800 is located on steeper slopes than staff's recommendation so staff is unsure what the justification is for the applicant's proposed trail connection location on tax lot 800. After review of the Lidar data the Parks and Trails Advisory Board was also unsure about the justification for moving the trail connection to tax lot 800.

17.84.40 TRANSIT AND SCHOOL BUS TRANSIT REQUIREMENTS

- A. Development sites located along existing or planned transit routes shall, where appropriate, incorporate bus pull-outs and/or shelters into the site design. These improvements shall be installed in accordance with the guidelines and standards of the transit agency. School bus pull-outs and/or shelters may also be required, where appropriate, as a condition of approval for a residential development of greater than 50 dwelling units where a school bus pick-up point is anticipated to serve a large number of children.
- B. New developments at or near existing or planned transit or school bus transit stops shall design development sites to provide safe, convenient access to the transit system, as follows:
 - 1. Commercial and civic use developments shall provide a prominent entrance oriented towards arterial and collector streets, with front setbacks reduced as much as possible to provide access for pedestrians, bicycles, and transit.
 - 2. All developments shall provide safe, convenient pedestrian walkways between the buildings and the transit stop, in accordance with the provisions of 17.84.30 B.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx
RESPONSE: The Transit Director (Exhibit P) states that the proposed development will require construction of one transit amenity at the northwest corner of the subject property adjacent to Tract A and as close to Cascadia Village Drive as possible. The amenity required is a 7 foot by 9.5 foot concrete pad to accommodate a 5 foot by 7.5 foot bus shelter, equipped with one 6 foot long green metal bench. The applicant shall install one bench (Fairweather model PL-3, powder-coated RAL6028) with a concrete pad to City specifications. The applicant shall contact the Transit Director for specific location, amenity information, and pad engineering specifications.

17.84.50 STREET REQUIREMENTS

- A. Traffic evaluations may be required of all development proposals in accordance with the following:
 - 1. A proposal establishing the scope of the traffic evaluation shall be submitted for review to the City Engineer. The evaluation requirements shall reflect the magnitude of the project in accordance with accepted traffic engineering practices. Large projects should assess all nearby key intersections. Once the scope of the traffic evaluation has been approved, the applicant shall present the results with and an overall site development proposal. If required by the City Engineer, such evaluations shall be signed by a Licensed Professional Civil Engineer or Licensed Professional Traffic Engineer licensed in the State of Oregon.
 - 2. If the traffic evaluation identifies level-of-service conditions less than the minimum standard established in the Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered concurrent with a development proposal.

RESPONSE: The submitted Traffic Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10th Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably per City of Sandy and ODOT standards and are projected to continue to operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development.

The City Engineer (Exhibit L) reviewed the Traffic Impact Analysis and noted the following: "There are no reported crashes at the intersection of Jacoby [Road] and Dubarko Road, or at any of the intersections adjacent to the development. However, the crash rate at the intersection of Dubarko Road and Highway 211 exceeds the 90th percentile criteria and meets the warrants for the addition of left turn lanes as identified in the TSP. This intersection should be addressed by ODOT as soon as possible, but not as a component of this proposed development."

The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and does not recommend any modifications to the TIA as the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards. The City Transportation Engineer states the following: "The intersections that will provide access from the

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

subdivision to Jacoby Road are appropriately located and provide adequate sight distance. The existence of a higher than average crash rate at the intersection of OR 211 and Dubarko Road and traffic volumes that suggest left-turn lanes are warranted on OR 211 at this location are indicators that the city should continue to work with ODOT for implementation of enhancements at this intersection. This analysis indicates that a traffic signal is not yet warranted but left-turn lanes on both the northbound and southbound approaches on OR 211 are warranted. The need for these improvements is justified based on regional growth of which this development is only a minor component."

- B. Location of new arterial streets shall conform to the Transportation System Plan in accordance with the following:
 - 1. Arterial streets should generally be spaced in one-mile intervals.
 - 2. Traffic signals should generally not be spaced closer than 1500 ft. for reasonable traffic progression.

RESPONSE: The proposal does not include any new arterial or collector streets.

- C. Local streets shall be designed to discourage through traffic. NOTE: for the purposes of this section, "through traffic" means the traffic traveling through an area that does not have a local origination or destination. To discourage through traffic and excessive vehicle speeds the following street design characteristics shall be considered, as well as other designs intended to discourage traffic:
 - 1. Straight segments of local streets should be kept to less than a quarter mile in length. As practical, local streets should include traffic calming features, and design features such as curves and "T" intersections while maintaining pedestrian connectivity.
 - 2. Local streets should typically intersect in "T" configurations rather than 4-way intersections to minimize conflicts and discourage through traffic. Adjacent "T" intersections shall maintain a minimum of 150 ft. between the nearest edges of the 2 rights-of-way.
 - 3. Cul-de-sacs should generally not exceed 400 ft. in length nor serve more than 20 dwelling units, except in cases where existing topography, wetlands, or drainage systems or other existing features necessitate a longer cul-de-sac in order to provide adequate access to an area. Cul-de-sacs longer than 400 feet or developments with only one access point may be required to provide an alternative access for emergency vehicle use only, install fire prevention sprinklers, or provide other mitigating measures, determined by the City.

RESPONSE: The proposal includes a cul-de-sac (Camden Court) that does not exceed 400 feet in length. Five lots and Tract A are proposed to gain access from the cul-de-sac.

- D. Development sites shall be provided with access from a public street improved to City standards in accordance with the following:
 - 1. Where a development site abuts an existing public street not improved to City standards, the abutting street shall be improved to City standards along the full frontage of the property concurrent with development.
 - 2. Half-street improvements are considered the minimum required improvement. Three-quarterstreet or full-street improvements shall be required where traffic volumes generated by the development are such that a half-street improvement would cause safety and/or capacity problems. Such a determination shall be made by the City Engineer.
 - 3. To ensure improved access to a development site consistent with policies on orderly urbanization and extension of public facilities the Planning Commission or Director may require off-site

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

improvements concurrent with development. Off-site improvement requirements upon the site developer shall be reasonably related to the anticipated impacts of the development.

- 4. Reimbursement agreements for ³/₄ street improvements (i.e., curb face to curb face) may be requested by the developer per Chapter 12 of the SMC.
- 5. A ¹/₂ street improvement includes curb and pavement 2 feet beyond the center line of the right-ofway. A ³/₄ street improvement includes curbs on both sides of the side and full pavement between curb faces.

RESPONSE: The proposed street network and improvements generally comply with City standards, with the exception of one block variance. There are five streets inside or adjacent to the proposed subdivision requiring the following improvements:

<u>Jacoby Road</u>: Half-street construction of Jacoby Road is required within the subject property boundary to collector street standards, including curb, 6-foot wide concrete sidewalks, street lighting, landscaping strip, street trees, and public utilities. Per the City Engineer (Exhibit L) and Public Works Director (Exhibit O), the improvements on Jacoby Road shall include an additional 10 feet of right-of-way dedication (60' ROW).

<u>Woodstock Street</u>: Woodstock Street requires full-street construction to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities. The applicant has applied for one variance to exceed the maximum block length for the north side of Woodstock Street from Camden Court to the east property boundary. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." **The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. The developer shall improve the street section adjacent to the recommended trail connection tract on Woodstock Street prior to final plat approval.**

<u>Middlebury Avenue</u>: Middlebury Avenue requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

<u>Newton Street</u>: Newton Street is proposed to align with the existing alignment of Newton Street west of Jacoby Road and to then shift to the south such that it aligns with the southern property boundary. The extension of Newton Street requires full-street improvements to local standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, street trees, ADA ramps, and public utilities adjacent to Lot 9, tapering to three-quarter street improvements adjacent to the remaining lots along Newton Street. The applicant is proposing a three-quarter street improvement for Newton Street through the site transitioning to a half-street section at the west line of proposed Lot 2. Per the Public Works Director (Exhibit O), the proposed Newton Street alignment coincides with an existing access easement serving at least two and possibly as many as four existing dwellings to the east. No more than two dwellings may be served by a half-street improvement. As proposed this narrowed section of Newton Street would serve at least four dwellings and perhaps as many as six. **The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site.** Per the City Engineer (Exhibit L), **the three-quarter street improvements on Newton Street shall include a 28-foot wide paved**

surface, curbs on both sides, 5-foot planter strips and 5-foot sidewalks on the north side of the roadway. The preliminary street and utility plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road.

<u>Camden Court</u>: Camden Court requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

- E. As necessary to provide for orderly development of adjacent properties, public streets installed concurrent with development of a site shall be extended through the site to the edge of the adjacent property(ies) in accordance with the following:
 - 1. Temporary dead-ends created by this requirement to extend street improvements to the edge of adjacent properties may be installed without turn-arounds, subject to the approval of the Fire Marshal.
 - 2. In order to assure the eventual continuation or completion of the street, reserve strips may be required.

RESPONSE: The applicant is not proposing any permanent dead-end streets. The applicant proposes that Newton Street, Woodstock Street, and Middlebury Avenue be temporary dead-end streets with construction of this subdivision until such a time as these streets are extended onto the adjoining properties to the east and south. The applicant shall plat a vehicle non-access reserve (VNAR) strip along the south side of the right-of-way of Newton Street and at the east end of Woodstock Street. The dead-ends on Woodstock Street and Newton Street would be approximately 190 feet. The applicant shall work with the Fire Marshall to determine if a temporary turn-around is needed on Woodstock Street and/or Newton Street, or if houses on particular lots need to be sprinkled. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements on Woodstock Street improvements on the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code.

F. Where required by the Planning Commission or Director, public street improvements may be required through a development site to provide for the logical extension of an existing street network or to connect a site with a nearby neighborhood activity center, such as a school or park. Where this creates a land division incidental to the development, a land partition shall be completed concurrent with the development.

RESPONSE: The proposal includes the extension of Newton Street from the Cascadia Village subdivision.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

G. Except for extensions of existing streets, no street names shall be used that will duplicate or be confused with names of existing streets. Street names and numbers shall conform to the established pattern in the surrounding area and be subject to approval of the Director.

RESPONSE: The applicant is proposing the following new street names: Camden Court, Woodstock Street, and Middlebury Avenue. **The City of Sandy reserves the right to name streets.**

- H. Location, grades, alignment, and widths for all public streets shall be considered in relation to existing and planned streets, topographical conditions, public convenience and safety, and proposed land use. Where topographical conditions present special circumstances, exceptions to these standards may be granted by the City Engineer provided the safety and capacity of the street network is not adversely affected. The following standards shall apply:
 - 1. Location of streets in a development shall not preclude development of adjacent properties. Streets shall conform to planned street extensions identified in the Transportation Plan and/or provide for continuation of the existing street network in the surrounding area.
 - 2. Grades shall not exceed 6 percent on arterial streets, 10 percent on collector streets, and 15 percent on local streets.
 - 3. As far as practical, arterial streets and collector streets shall be extended in alignment with existing streets by continuation of the street centerline. When staggered street alignments resulting in "T" intersections are unavoidable, they shall leave a minimum of 150 ft. between the nearest edges of the two rights-of-way.
 - 4. Centerline radii of curves shall not be less than 500 ft. on arterial streets, 300 ft. on collector streets, and 100 ft. on local streets.
 - 5. Streets shall be designed to intersect at angles as near as practicable to right angles and shall comply with the following:
 - a) The intersection of an arterial or collector street with another arterial or collector street shall have a minimum of 100 ft. of straight (tangent) alignment perpendicular to the intersection.
 - b) The intersection of a local street with another street shall have a minimum of 50 ft. of straight (tangent) alignment perpendicular to the intersection.
 - c) Where right angle intersections are not possible, exceptions can be granted by the City Engineer provided that intersections not at right angles have a minimum corner radius of 20 ft. along the right-of-way lines of the acute angle.
 - d) Intersections with arterial streets shall have a minimum curb corner radius of 20 ft. All other intersections shall have a minimum curb corner radius of 10 ft.
 - 6. Right-of-way and improvement widths shall be as specified by the Transportation System Plan. Exceptions to those specifications may be approved by the City Engineer to deal with specific unique physical constraints of the site.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) does not show the correct measurement of 50 feet of tangent alignment at the intersection of Jacoby Road and Newton Street as required by 17.84.50(H.5.b). The applicant measured the 50 feet of tangent alignment from the centerline of Jacoby Road rather than from the curb line extended. **The applicant shall revise the alignment of Newton Street to conform to the 50 feet of tangent alignment standard**.

17.84.60 PUBLIC UTILITY EXTENSIONS

A. All development sites shall be provided with public water, sanitary sewer, and storm drainage.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- B. Where necessary to serve property as specified in "A" above, required public utility installations shall be constructed concurrent with development.
- C. Off-site public utility extensions necessary to fully serve a development site and adjacent properties shall be constructed concurrent with development.
- D. As necessary to provide for orderly development of adjacent properties, public utilities installed concurrent with development of a site shall be extended through the site to the edge of adjacent property(ies).
- E. All public utility installations required with development shall conform to the City's facilities master plans.
- F. Private on-site sanitary sewer and storm drainage facilities may be considered provided all the following conditions exist:
 - 1. Extension of a public facility through the site is not necessary for the future orderly development of adjacent properties;
 - 2. The development site remains in one ownership and land division does not occur (with the exception of land divisions that may occur under the provisions of 17.84.50 F above);
 - 3. The facilities are designed and constructed in accordance with the Uniform Plumbing Code and other applicable codes, and permits and/or authorization to proceed with construction is issued prior to commencement of work.

RESPONSE: The applicant's Preliminary Street and Utility Plan (Exhibit C, Sheet 5) depicts the location and type of proposed public utilities including water, sanitary sewer and stormwater. All public utility installations shall conform to the City's facilities master plans. Staff recommends the applicant revise the utility plan to include broadband fiber locations as detailed by the SandyNet Manager. Per the City Engineer (Exhibit L), all public sanitary sewer and waterline mains shall be a minimum of 8 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and all stormwater drains shall be issued until all public utilities including sanitary sewer are available to serve the development. The applicant shall pay plan review, inspection, and permit fees as determined by the Public Works Director. The utility improvements proposal and requirements for the Jacoby Heights subdivision are further detailed in Sections 17.100.230, 17.100.240, and 17.100.250 below.

17.84.80 FRANCHISE UTILITY INSTALLATIONS

These standards are intended to supplement, not replace or supersede, requirements contained within individual franchise agreements the City has with providers of electrical power, telephone, cable television, and natural gas services (hereinafter referred to as "franchise utilities").

RESPONSE: Private utility services will be submitted for review and approval by service providers and City staff in association with construction plans, and all utility lines will be extended to the perimeter of the site. All franchise utilities shall be installed underground and in conformance with City standards. PGE submitted a comment (Exhibit N) stating they did not find any conflicts related to the project. Per PGE's request, the applicant shall call the PGE Service Coordinators at 503-323-6700 when the developer is ready to start the project.

17.84.90 LAND FOR PUBLIC PURPOSES

- A. Easements for public sanitary sewer, water, storm drain, pedestrian and bicycle facilities shall be provided whenever these facilities are located outside a public right-of-way in accordance with the following:
 - 1. When located between adjacent lots, easements shall be provided on one side of a lot line.
 - 2. The minimum easement width for a single utility is 15 ft. The minimum easement width for two adjacent utilities is 20 ft. The easement width shall be centered on the utility to the greatest extent practicable. Wider easements may be required for unusually deep facilities.
- B. Public utility easements with a minimum width of 5 feet shall be provided adjacent to all street rights-of-way for franchise utility installations.
- C. Where a development site is traversed by a drainageway or water course, a drainage way dedication shall be provided to the City.
- D. Where a development is traversed by, or adjacent to, a future trail linkage identified within the Transportation System Plan, dedications of suitable width to accommodate the trail linkage shall be provided. This width shall be determined by the City Engineer, considering the type of trail facility involved.
- E. Where existing rights-of-way and/or easements within or adjacent to development sites are nonexistent or of insufficient width, dedications may be required. The need for and widths of those dedications shall be determined by the City Engineer.
- F. Where easement or dedications are required in conjunction with land divisions, they shall be recorded on the plat. Where a development does not include a land division, easements and/or dedications shall be recorded on standard document forms provided by the City Engineer.
- G. If the City has an interest in acquiring any portion of a proposed subdivision or planned development site for a public purpose, other than for those purposes listed above, or if the City has been advised of such interest by a school district or other public agency, and there is a reasonable assurance that steps will be taken to acquire the land, the Planning Commission may require those portions of the land be reserved for public acquisition for a period not to exceed 1 year.
- H. Environmental assessments for all lands to be dedicated to the public or City may be required to be provided by the developer. An environmental assessment shall include information necessary for the City to evaluate potential liability for environmental hazards, contamination, or required waste cleanups related to the dedicated land. An environmental assessment shall be completed prior to the acceptance of dedicated lands in accordance with the following:
 - The initial environmental assessment shall detail the history of ownership and general use of the land by past owners. Upon review of the information provided by the grantor, as well as any site investigation by the City, the Director will determine if the risks of potential contamination warrant further investigation. When further site investigation is warranted, a Level I Environmental Assessment shall be provided by the grantor.

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A). Rather than include the FSH overlay area as part of Lots 19, 20, 21, 26, 27, and 28, the applicant shall designate the FSH area as a separate tract on the plat and dedicate the tract to the City. In addition, staff recommends the Planning Commission require a pedestrian trail connection to the future extension of the Tickle Creek Trail as part of this development. Staff recommends that the trail connection be located in alignment with Middlebury Avenue. Alternatively, staff recommends that Lot 19 be included in the FSH tract and that the trail connection be located on the proposed location of Lot 19. The applicant shall include the trail connector tract as part of the FSH tract and shall dedicate the entire tract to the City.

The Plat shall detail the following easements:

- An eight-foot wide public utility easement (PUE) along the frontage of all proposed lots;
- Crossover easements along the common parcel line recorded between pairs of lots that share driveways (Lots 26 and 27, and Lots 28 and 29);
- 20 foot easement along the west boundary of Lots 9 and 10 for public sanitary sewer and public storm line. Per the Public Works Director (Exhibit O), the City reserves the right to widen this easement depending on the final design depth of the sanitary sewer and storm lines per Section 17.84.90(A.2);
- A 15 foot easement between Lot 30 and Tract A for public sanitary sewer;
- A 10 foot private storm drain easement at the rear of Lots 1-18, 22-25, 28, 29, 31, and 32 and a 10 foot private storm drain easement across Lots 19-21 just south of the FSH boundary;
- A vehicle non-access reserve (VNAR) strip in the following locations:
 - Jacoby Road frontage of Lots 9, 10, 30 and 32 and Tract A; and,
 - South side of the right-of-way of Newton Street; and,
 - East end of the right-of-way of Woodstock Street.

17.84.100 MAIL DELIVERY FACILITIES

RESPONSE: The applicant will need to coordinate with the United States Postal Service (USPS) to locate mail facilities and these will be approved by the City and USPS. **Mail delivery facilities shall be** provided by the applicant in conformance with Section 17.84.100 and the standards of the USPS. The applicant shall submit a mail delivery plan, featuring grouped lockable mail facilities, to the City and USPS for review and approval prior to installation of mailboxes.

CHAPTER 17.86 – PARKLAND AND OPEN SPACE

17.86.10 MINIMUM PARKLAND DEDICATION REQUIREMENTS

<u>Parkland Dedication:</u> New residential subdivisions, planned developments, multi-family or manufactured home park developments shall be required to provide parkland to serve existing and future residents of those developments. Multi-family developments which provide some "congregate" services and/or facilities, such as group transportation, dining halls, emergency monitoring systems, etc., but which have individual dwelling units rather than sleeping quarters only, are considered to be multifamily developments for the purpose of parkland dedication. Licensed adult congregate living facilities, nursing homes, and all other similar facilities which provide their clients with individual beds and sleeping quarters, but in which all other care and services are communal and provided by facility employees, are specifically exempt from parkland dedication and system development fee requirements.

- 1. The required parkland shall be dedicated as a condition of approval for the following:
 - a. Tentative plat for a subdivision or partition;

- b. Planned Development conceptual or detailed development plan;
- c. Design review for a multi-family development or manufactured home park; and
- d. Replat or amendment of any site plan for multi-family development or manufactured home park where dedication has not previously been made or where the density of the development involved will be increased.
- 2. Calculation of Required Dedication: The required parkland acreage to be dedicated is based on a calculation of the following formula rounded to the nearest 1/100 (0.00) of an acre:

Required parkland dedication (acres) = (proposed units) x (persons/unit) x 0.0043 (per person park land dedication factor)

a. Population Formula: The following table shall be used to determine the number of persons per unit to be used in calculating required parkland dedication:

Type of Unit	Total Persons Per Unit
Single family residential	3.0
Standard multi-family unit	2.0
Manufactured dwelling park	2.0
Congregate multi-family unit	1.5

Persons per unit, age distribution, and local conditions change with time. The specific formula for the dedication of land will, therefore, be subject to periodic review and amendment.

b. Per Person Parkland Dedication Factor: The total parkland dedication requirement shall be 0.0043 of an acre per person based on the adopted standard of 4.3 acres of land per one thousand of ultimate population per the Parks Master Plan¹. This standard represents the citywide land-topopulation ratio for city parks, and may be adjusted periodically through amendments to the Parks Master Plan.

RESPONSE: The applicant proposes 32 single-family detached dwellings with this subdivision request. Based upon the calculations adopted by the City and specified within Section 17.86.10, the required dedication area is 0.41 acres of public parkland (32 proposed units x 3 persons per unit x .0043=0.4128 rounded to the nearest 1/100 of an acre = 0.41 acres to be dedicated).

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

¹ Parks Master Plan, Implementation Plan section, Pages 4 and 5 indicate a required park acreage total of 64.5 acres. This number, divided by population (2015) of 15,000 equates to 4.3 acres per 1000 population or 0.0043 per person.

17.86.40 CASH IN LIEU OF DEDICATION

At the city's discretion only, the city may accept payment of a fee in lieu of land dedication. The city may require payment in lieu of land when the park land to be dedicated is less than 3 acres. A payment in lieu of land dedication is separate from Park Systems Development Charges, and is not eligible for a credit of Park Systems Development Charges. The amount of the fee in lieu of land dedication (in dollars per acre) shall be set by City Council Resolution, and it shall be based on the typical market value of developed property (finished lots) in Sandy net of related development costs.

- 1. The following factors shall be used in the choice of whether to accept land or cash in lieu:
 - a. The topography, geology, access to, parcel size, and location of land in the development available for dedication;
 - b. Potential adverse/beneficial effects on environmentally sensitive areas;
 - c. Compatibility with the Parks Master Plan, Public Facilities element of the Comprehensive Plan, and the City of Sandy Capital Improvements Program in effect at the time of dedication;
 - d. Availability of previously acquired property; and
 - e. The feasibility of dedication.
- 2. A. Cash in lieu of parkland dedication shall be paid prior to approval of the final plat or as specified below:
 - a. 50 percent of the payment shall be paid prior to final plat approval, and
 - b. The remaining 50 percent of the payment pro-rated equally among the lots, plus an administrative surcharge, shall be paid as specified by City Council Resolution.

RESPONSE: Based on the calculations specified in Section 17.86.10, the applicant is responsible for dedicating 0.41 acres of public parkland based on 32 dwelling units. No parkland is identified on the subject property in the Parks Master Plan. For these reasons, the applicant shall pay a fee in lieu for the required parkland dedication per the adopted Fee Resolution. Per Resolution 2013-14, the required fee in lieu amount is \$241,000 per acre if the entire amount is paid prior to final plat approval. Therefore, based on the current Fee Resolution, the applicant would be required to pay a fee in lieu of dedication for a total of \$98,810 (0.41 acres of land to be dedicated x \$241,000). Alternatively, Ordinance 2013-03 allows the applicant to pay a minimum of 50 percent of the fee to receive final plat approval with the remaining balance to be paid as a proportionate amount with each building permit. If a portion of the fee is deferred, Resolution 2013-14 specifies a per acre fee of \$265,000. Currently, the Fee Resolution requires payment of \$108,650 if a portion of the fee is deferred, a minimum of 50 percent (\$54,325) paid prior to final plat approval and the remaining 50 percent (\$54,325) divided between the 32 lots (\$1,697.66/lot). If the number of lots is modified from 32, the parks fee in lieu will need to be recalculated.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

17.86.50 MINIMUM STANDARDS FOR OPEN SPACE DEDICATION

The applicant through a subdivision or design review process may propose the designation and protection of open space areas as part of that process. This open space will not, however, be counted toward the parkland dedication requirement of Sections 17.86.10 through 17.86.40.

4. Open space areas shall be maintained so that the use and enjoyment thereof is not diminished or destroyed. Open space areas may be owned, preserved, and maintained by any of the following mechanisms or combinations thereof:

a. Dedication to the City of Sandy or an appropriate public agency approved by the City, if there is a public agency willing to accept the dedication. Prior to acceptance of proposed open space, the City may require the developer to submit a Phase I Environmental Site Assessment completed by a qualified professional according to American Society of Testing and Materials (ASTM) standards (ASTM E 1527). The results of this study shall indicate a clean environmental record.

b. Common ownership by a homeowner's association that assumes full responsibility for its maintenance;

c. Dedication of development rights to an appropriate public agency with ownership remaining with the developer or homeowner's association. Maintenance responsibility will remain with the property owner; and/or

d. Deed-restricted private ownership preventing development and/or subsequent subdivision and providing for maintenance responsibilities.

RESPONSE: The applicant is not proposing any dedication of open space. However, the Parks Board recommends that the FSH analysis tract be dedicated to the City. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the tract without knowing the condition that it is in. The applicant shall submit a Phase I environmental assessment and a native vegetation report for the FSH area. The applicant shall remove all invasive plants and re-plant any areas of exposed soil with native plants. The applicant shall dedicate the FSH tract to the City and shall submit a maintenance bond for two full growing seasons (May 1 – October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Although parkland dedication is separate from open space dedication, staff recommends that the Planning Commission grant the applicant shall obtain an appraisal for the FSH analysis area and, with City approval of the appraised value, shall receive a parks fee-in-lieu credit for dedicating the open space FSH analysis area. To be considered for a parks fee in-lieu credit the appraisal will need to account for the limited development potential of the land to be dedicated.

CHAPTER 17.92 – LANDSCAPING AND SCREENING

17.92.10 GENERAL PROVISIONS

C. Significant plant and tree specimens should be preserved to the greatest extent practicable and integrated into the design of a development. Trees of 25-inches or greater circumference measured at

 $W: \label{eq:linear} W: \lab$

a height of 4-1/2 ft. above grade are considered significant. Plants to be saved and methods of protection shall be indicated on the detailed planting plan submitted for approval. Existing trees may be considered preserved if no cutting, filling, or compaction of the soil takes place between the trunk of the tree and the area 5-ft. outside the tree's drip line. Trees to be retained shall be protected from damage during construction by a construction fence located 5 ft. outside the dripline.

L. All landscaping shall be continually maintained, including necessary watering, weeding, pruning and replacing.

RESPONSE: Per Section 17.92.10 (C), trees over 25-inches circumference measured at a height of 4-¹/₂ feet above grade are considered significant and should be preserved to the greatest extent practicable and integrated into the design of a development. A 25-inch circumference tree measured at 4-¹/₂ feet above grade has roughly an eight-inch diameter at breast height (DBH). The applicant shall preserve significant trees (25-inches or greater circumference, or 8-inches or greater DBH) to the greatest extent practicable. The applicant shall install protective fencing located at least 5 feet outside of the dripline around all trees to be retained on the subject property and for trees on adjacent property. Tree retention will be discussed in more detail under Chapter 17.102 in this document. Per Section 17.92.10(L), all landscaping shall be continually maintained, including necessary watering, weeding, pruning, and replacing.

Note: the applicant is requesting a code interpretation to determine if Section 17.92.10(C) applies to subdivisions.

17.92.30 REQUIRED TREE PLANTINGS

Planting of trees is required for all parking lots with 4 or more parking spaces, public street frontages, and along private drives more than 150 feet long. Trees shall be planted outside the street right-of-way except where there is a designated planting strip or City adopted street tree plan.

The City maintains a list of appropriate trees for street tree and parking lot planting situations. Selection of species should be made from the city-approved list. Alternate selections may be approved by the Director following written request. The type of tree used shall determine frequency of trees in planting areas. Trees in parking areas shall be dispersed throughout the lot to provide a canopy for shade and visual relief.

Area/Type of Planting	Canopy	Spacing
Street Tree	Medium	30 ft. on center
Street Tree	Large	50 ft. on center

Trees may not be planted:

- Within 5 ft. of permanent hard surface paving or walkways, unless specific species, special planting techniques and specifications approved by the Director are used.
- Unless approved otherwise by the City Engineer:
 - * Within 10 ft. of fire hydrants and utility poles
 - * Within 20 ft. of street light standards
 - * Within 5 ft. from an existing curb face

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- * Within 10 ft. of a public sanitary sewer, storm drainage or water line
- Where the Director determines the trees may be a hazard to the public interest or general welfare.
- Trees shall be pruned to provide a minimum clearance of 8 ft. above sidewalks and 12 ft. above street and roadway surfaces.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires trees spaced 30 feet on center along street frontages. The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all of the proposed streets. The proposed plan details 84 street trees. Two (2) of the street trees are proposed to be located in driveways in front of Lots 31 and 32; however, there appears to be sufficient space to relocate at least one of the trees outside of the driveway area. The applicant shall update the plan set to remove street trees from the driveways in front of Lots 31 and 32; at least one of the street trees shall be relocated outside of the driveway area. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." With extension of the sidewalk and planter strip to the southern property boundary, the proposal could accommodate a third street tree on the south side of Newton Street. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the sidewalk, planter strip, and one additional street tree. The proposed layout would likely accommodate an additional two (2) street trees on the east end of Newton Street. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. If the plans change in a way that affects the number of street trees (e.g., driveway locations), the applicant shall submit an updated street tree plan for staff review and approval. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted, staked, and the planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed after one growing season (or a maximum of 1 year).

17.92.40 IRRIGATION

Landscaping shall be irrigated, either with a manual or automatic system, to sustain viable plant life.

RESPONSE: Section 17.92.40 requires that all landscaping shall be irrigated, either with a manual or automatic system. As required by Section 17.92.140, the developer and lot owners shall be required to maintain all vegetation planted in the development for two years from the date of completion, and shall replace any dead or dying plants during that period.

17.92.50 TYPES AND SIZES OF PLANT MATERIALS

- A. At least 75% of the required landscaping area shall be planted with a suitable combination of trees, shrubs, or evergreen ground cover except as otherwise authorized by Chapter 17.92.10 F.
- B. Plant Materials. Use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged where possible.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- C. Trees shall be species having an average mature spread of crown greater than 15 feet and having trunks which can be maintained in a clear condition with over 5 feet of clear wood (without branches). Trees having a mature spread of crown less than 15 feet may be substituted by grouping the same so as to create the equivalent of a 15-foot crown spread.
- D. Deciduous trees shall be balled and burlapped, be a minimum of 7 feet in overall height or 1 ¹/₂ inches in caliper measured 6 inches above the ground, immediately after planting. Bare root trees will be acceptable to plant during their dormant season.
- E. Coniferous trees shall be a minimum five feet in height above ground at time of planting.
- F. Shrubs shall be a minimum of 1 gallon in size or 2 feet in height when measured immediately after planting.
- G. Hedges, where required to screen and buffer off-street parking from adjoining properties shall be planted with an evergreen species maintained so as to form a continuous, solid visual screen within 2 years after planting.
- H. Vines for screening purposes shall be a minimum of 1 gallon in size or 30 inches in height immediate after planting and may be used in conjunction with fences, screens, or walls to meet physical barrier requirements as specified.
- I. Groundcovers shall be fully rooted and shall be well branched or leafed. If used in lieu of turf in whole or in part, ground covers shall be planted in such a manner as to provide complete coverage in one year.
- J. Turf areas shall be planted in species normally grown as permanent lawns in western Oregon. Either sod or seed are acceptable. Acceptable varieties include improved perennial ryes and fescues used within the local landscape industry.
- K. Landscaped areas may include architectural features or artificial ground covers such as sculptures, benches, masonry or stone walls, fences, rock groupings, bark dust, decorative hard paving and gravel areas, interspersed with planted areas. The exposed area developed with such features shall not exceed 25% of the required landscaped area. Artificial plants are prohibited in any required landscape area.

RESPONSE: Street trees are typically required to be a minimum caliper of 1.5-inches measured 6 inches from grade. In the past, staff has recommended a larger caliper street tree (2-inches) for subdivision proposals where a substantial number of existing trees were proposed for removal. However, staff has received feedback that fewer species are available at the 2-inch caliper size. To provide species diversity, staff recommends requiring 1.5-inch caliper street trees. All street trees shall be a minimum of 1.5-inches in caliper measured 6 inches above the ground and shall be planted per the City of Sandy standard planting detail. The applicant proposes six distinct street tree species with one tree species per block face. Staff would like to see more diversity in street tree species in general and within each block. The applicant shall update the plan set to detail a minimum of two (2) different tree species per block face for staff review and approval. The proposed street tree list includes paperbark maple and Patmore ash. Due to concerns with Asian Longhorn Beetle and Emerald Ash Borer, staff do not recommend planting new maples or ashes at this time. The applicant shall update the proposed street tree list with at least two additional species from the City of Sandy Street Tree List that aren't maples or ashes. Staff recommends two (or more) species from the following list: cascara (Rhamnus purshiana), kobus magnolia (Magnolia kobus), Skyrocket oak (Quercus robur 'Skyrocket'), Japanese stewartia (Stewartia pseudocamellia), Japanese snowbell (Styrax japonica), or eastern redbud (Cercis canadensis). Providing species diversity adds visual interest and provides additional resistance against disease in conformance with City of Sandy Comprehensive Goal 5 Policy 28. The applicant also proposed Halka ginkgo. Technically, the only ginkgo on the City of Sandy street tree list is Saratoga ginkgo (Ginkgo biloba

'Saratoga'); however, staff recommends that Halka ginkgo be approved as a street tree with this application.

17.92.60 REVEGETATION IN UNLANDSCAPED OR NATURAL AREAS

- A. Areas where natural vegetation has been removed or damaged through grading or construction activity in areas not affected by the landscaping requirements and that are not to be occupied by structures or other improvements shall be replanted.
- B. Plant material shall be watered at intervals sufficient to assure survival and growth.
- C. The use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged to reduce irrigation and maintenance demands.

RESPONSE: The applicant did not submit any plans for re-vegetation of areas damaged through grading/construction, although most of the areas affected by grading will be improved. The applicant shall maintain all unlandscaped and/or revegetated areas for a period of two years following the date of recording of the final plat associated with those improvements.

17.92.130 PERFORMNCE BOND

If weather conditions or other circumstances beyond the control of the developer or owner make completion of the landscaping impossible prior to desired occupancy, an extension of up to 6 months may be applied for by posting "security" equal to 120% of the cost of the landscaping, assuring installation within 6 months. "Security" may consist of a performance bond payable to the city, cash, certified check, time certificates of deposit, assignment of a saving account, letter of credit, or other such assurance of access to funds necessary for completion as shall meet the approval of the City Attorney. Upon acceptance of the security, the developer or owner may be allowed occupancy for a period of up to 180 days. If the installation of the landscaping improvement is not completed within 180 days, the City shall have access to the security to complete the installation and/or revoke occupancy. Upon completion of the installation by the city, any portion of the remaining security shall be assessed against the property and the City shall thereupon have a valid lien against the property, which will come due, and payable.

RESPONSE: The applicant has the option to defer the installation of street trees and/or landscaping for weather-related reasons. Staff recommends the applicant utilize this option rather than install trees and landscaping during the dry summer months. If the applicant chooses to postpone street tree and/or landscaping installation, the applicant shall post a performance bond equal to 120 percent of the cost of the street trees/landscaping, assuring installation within 6 months. The cost of street trees shall be based on the street tree plan and at least \$500 per tree fee, including a three-year maintenance and warranty period. The cost of landscaping shall be based on the average of three estimates from three landscaping contractors; the estimates shall include as separate items all materials, labor, and other costs of the required action, including a two-year maintenance and warranty period.

CHAPTER 17.98 - PARKING, LOADING, AND ACCESS REQUIREMENTS

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

17.98.20 OFF-STREET PARKING REQUIREMENTS

RESPONSE: Section 17.98.20 requires two off-street parking spaces per single family detached dwelling unit. The 32 dwelling units proposed in this subdivision requires 64 off-street parking spaces. Each lot will have a driveway and the ability to construct a double car garage.

17.98.50 SETBACKS

- A. Parking areas, which abut a residential zoning district, shall meet the setback of the most restrictive adjoining residential zoning district.
- B. Required parking shall not be located in a required front or side yard setback area abutting a public street except in industrial districts. For single family and two-family dwellings, required off-street parking may be located in a driveway.
- C. Parking areas shall be setback from a lot line adjoining a street the same distance as the required building setbacks. Regardless of other provisions, a minimum setback of 5 feet shall be provided along the property fronting on a public street. The setback area shall be landscaped as provided in this code.

RESPONSE: Garages are required to be at least 22 feet setback from the front property line to meet setback requirements in the SFR zoning district. The Preliminary Plat (Exhibit C, Sheet 1) details a typical 22 foot garage setback.

17.98.60 DESIGN, SIZE AND ACCESS

All off-street parking facilities, vehicular maneuvering areas, driveways, loading facilities, accessways, and private streets shall conform to the standards set forth in this section.

- A. Parking Lot Design. All areas for required parking and maneuvering of vehicles shall have a durable hard surface such as concrete or asphalt.
- B. Size of Space.
 - 1. A standard parking space shall be 9 feet by 18 feet.
 - 2. A compact parking space shall be 8 feet by 16 feet.
 - 3. Handicapped parking spaces shall be 13 feet by 18 feet. Accessible parking shall be provided for all uses in compliance with the requirements of the State of Oregon (ORS 447.233) and the Americans with Disabilities Act.
 - 4. Parallel parking spaces shall be a length of 22 feet. No more than 35 percent of the parking stalls shall be compact spaces

RESPONSE: The applicant shall comply with the parking standards in Section 17.98.60. The parking areas in front of the proposed garages for all lots need to be at least 10 feet in width by 20 feet in length. Driveways for single family homes are required to be at least 10 feet wide as detailed in Section 17.98.100 below. The garages shall be adequate depth to park a vehicle and the on-street parking spaces shall be at least 22 feet in length.

17.98.80 ACCESS TO ARTERIAL AND COLLECTOR STREETS

- A. Location and design of all accesses to and/or from arterials and collectors (as designated in the Transportation System Plan) are subject to review and approval by the City Engineer. Where practical, access from a lower functional order street may be required. Accesses to arterials or collectors shall be located a minimum of 150 ft. from any other access or street intersection. Exceptions may be granted by the City Engineer. Evaluations of exceptions shall consider posted speed of the street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.
- B. No development site shall be allowed more than one access point to any arterial or collector street (as designated in the Transportation System Plan) except as approved by the City Engineer. Evaluations of exceptions shall consider posted speed of street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.
- C. When developed property is to be expanded or altered in a manner that significantly affects on-site parking or circulation, both existing and proposed accesses shall be reviewed under the standards in A and B above. As a part of an expansion or alteration approval, the City may require relocation and/or reconstruction of existing accesses not meeting those standards.

RESPONSE: Jacoby Heights Subdivision is proposed to create two intersections with Jacoby Road, a designated collector street: Woodstock Street and Newton Street. All lots are required to have driveway access from the internal local street network with a vehicle non-access reserve strip (VNAR) detailed on the face of plat along Jacoby Road.

17.98.100 DRIVEWAYS

- A. A driveway to an off-street parking area shall be improved from the public roadway to the parking area a minimum width of 20 feet for a two-way drive or 12 feet for a one-way drive but in either case not less than the full width of the standard approach for the first 20 feet of the driveway.
- B. A driveway for a single-family dwelling shall have a minimum width of 10 feet.
- C. A driveway for a two-family dwelling shall have a minimum width of 20 feet. A driveway approach must be constructed in accordance with applicable city standards and the entire driveway must be paved with asphalt or concrete.
- D. Driveways, aisles, turnaround areas and ramps shall have a minimum vertical clearance of twelve feet for their entire length and width but such clearance may be reduced in parking structures.
- E. No driveway shall traverse a slope in excess of 15 percent at any point along the driveway length.
- F. The location and design of the driveway shall provide for unobstructed sight per the vision clearance requirements. Requests for exceptions to these requirements will be evaluated by the City Engineer considering the physical limitations of the lot and safety impacts to vehicular, bicycle, and pedestrian traffic.

RESPONSE: The minimum driveway width for a single-family dwelling is 10 feet. The Public Works driveway approach standard detail specifies a maximum of 24 feet wide for a residential driveway

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

approach. The Preliminary Parking Plan (Exhibit C, Sheet 7) details driveway curb cuts for all lots. With the exception of a few lots on the cul-de-sac, all driveways are detailed at approximately 20 feet. The applicant is proposing a shared driveway between Lot 29 and Tract A at approximately 36 feet and a shared driveway between Lot 27 and 28 at approximately 26 feet. In order to maximize street tree planting areas and prevent conflicts with streetlights, water meters and dry utility facilities, shared driveway approaches may be required in cul-de-sac bulbs. Per the Public Works Director (Exhibit O), Lot 28 and 29 shall share a driveway approach maximum 24 feet in width. Lot 26 and 27 shall share a driveway approach maximum 24 feet in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches. The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of Lot 29 to eliminate remnant of Tract A between the access drive and Lot 29. The driveway approach for Tract A shall be a maximum of 12 feet in width. Per the City Engineer (Exhibit L), the access to the detention pond shall be paved or all-weather surface.

17.98.130 PAVING

- A. Parking areas, driveways, aisles and turnarounds shall be paved with concrete, asphalt or comparable surfacing, constructed to city standards for off-street vehicle areas.
- B. Gravel surfacing shall be permitted only for areas designated for non-motorized trailer or equipment storage, propane or electrically powered vehicles, or storage of tracked vehicles.

RESPONSE: As required by Section 17.98.130, all parking, driveway and maneuvering areas shall be constructed of asphalt, concrete, or other approved material.

17.98.140 DRAINAGE

Parking areas, aisles and turnarounds shall have adequate provisions made for the on-site collection of drainage waters to eliminate sheet flow of such waters onto sidewalks, public rights-of-way and abutting private property.

RESPONSE: Section 17.98.140 contains requirements for drainage. Other sections of this order detail the stormwater requirements.

17.98.200 RESIDENTIAL ON-STREET PARKING REQUIREMENTS

A. Residential On-Street Parking Requirements. Residential on-street parking shall conform to the following standards:

- 1. In addition to required off-street parking, all new residential planned developments, subdivisions and partitions shall provide one (1) on-street parking space within 200 feet of each dwelling except as provided in Section 17.98.200(A)(6) below.
- 2. The location of residential on-street parking shall be reviewed for compliance with this section through submittal of a Residential Parking Analysis Plan as required in Section 17.98.10(M).
- 3. Residential on-street parking shall not obstruct required clear vision areas and shall not violate any local or state laws.
- 4. Parallel residential on-street parking spaces shall be 22 feet minimum in length.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- 5. Residential on-street parking shall be measured along the curb from the outside edge of a driveway wing or curb cut. Parking spaces must be set back a minimum of 15 feet from an intersection and may not be located within 10 feet of a fire hydrant.
- 6. Portions of residential on-street parking required by this section may be provided in parking courts that are interspersed throughout a development when the following standards are met:
 - a. No more than eight (8) parking spaces shall be provided in a parking court;
 - b. Parking spaces within a parking court shall be nine (9) feet wide and 18 feet in depth;
 - c. Notwithstanding Section 17.98.70, vehicles parked in a parking court are permitted to back onto the public right-of-way from the parking court;
 - d. A parking court shall be located within 200 feet of the dwellings requiring parking in accordance with the requirements of Section 17.98.10(M);
 - e. No more than two (2) parking courts shall be provided within a block, with only one (1) parking court provided along a block face;
 - f. A parking court shall be paved in compliance with the standards of this chapter and the latest adopted grading and drainage standards;
 - g. If a parking court is adjacent to a public right-of-way, it shall be publicly owned and maintained;
 - h. If a parking court is adjacent to a private drive, it shall be privately owned and maintained. For each parking court there shall be a legal recorded document which includes:
 - i. A legal description of the parking court;
 - ii. Ownership of the parking court;
 - iii. Use rights; and
 - iv. A maintenance agreement and the allocation and/or method of determining liability for maintenance of the parking court;
 - i. A parking court shall be used solely for the parking of operable passenger vehicles.

RESPONSE: The Preliminary Parking Plan (Exhibit C, Sheet 7) identifies a total of 62 on-street parking spaces with at least one (1) on-street parking space within 200 feet of each of the 32 lots. The proposed parking plan shows 21 parking spaces on the south side of Newton Street, including parking spaces for Lots 1-4. These on-street parking spaces should not be counted toward meeting the on-street parking requirement since the south side of the street will eventually develop with driveways and/or streets that will eliminate these spaces or prevent the developer of the site to the south from counting these spaces toward their on-street parking requirement. Per the Public Works Director (Exhibit O), the applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval. No parking courts are proposed. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with Construction Plans. The applicant shall revise the Parking Analysis if required fire hydrants affect on-street parking spaces.

CHAPTER 17.100 – LAND DIVISION

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

Chapter 17.100 presents review procedures, design standards, and improvement requirements for land division. Section 17.100.60 is the subsection relevant to the division of subdivisions.

17.100.60 SUBDIVISIONS

Approval of a subdivision is required for a land division of 4 or more parcels in a calendar year. A twostep procedure is required for subdivision approval: (1) tentative plat review and approval; and (2) final plat review and approval.

RESPONSE: Submittal of preliminary utility plans is solely to satisfy the requirements of Section 17.100.60. Preliminary plat approval does not connote utility or public improvement plan approval which will be reviewed and approved separately upon submittal of public improvement construction plans.

- E. <u>Approval Criteria.</u> The Director or Planning Commission shall review the tentative plat for the subdivision based on the classification procedure (Type II or III) set forth in Section 17.12 and the following approval criteria:
 - The proposed subdivision is consistent with the density, setback and dimensional standards of the base zoning district, unless modified by a Planned Development approval. *RESPONSE:* The applicant requests subdivision approval for a subdivision that is in compliance with most of the applicable development standards. The applicant requests one variance to increase block length beyond the maximum block length standard, one variance to allow houses to not face a transit street, and one variance to allow single-tiered lots. As a result, the application for the subdivision and variances is being processed concurrently through a Type III procedure. The proposal is consistent with density and other dimensional standards of the base zoning district.
 - 2. The proposed subdivision is consistent with the design standards set forth in this chapter. *RESPONSE:* Consistency with design standards in this chapter are discussed under each subsection below. Conditions of approval can be adopted where necessary to bring the proposal into compliance with applicable standards.
 - The proposed street pattern is connected and consistent with the Comprehensive Plan or official street plan for the City of Sandy.
 RESPONSE: The proposed subdivision includes the extension of Newton Street to the east of Jacoby Road and three new proposed local streets, including one cul-de-sac. The subject property was not included in the City's current Transportation System Plan (TSP) because it was outside of the Urban Growth Boundary (UGB) when the TSP was adopted in 2011.
 - 4. Adequate public facilities are available or can be provided to serve the proposed subdivision. *RESPONSE:* All public facilities including water, sanitary sewer and stormwater are available or will be constructed by the applicant to serve the Jacoby Heights Subdivision.
 - 5. All proposed improvements meet City standards. *RESPONSE:* The detailed review of proposed improvements is contained in this report. Staff has identified a few aspects of the proposed subdivision improvements requiring additional information or modification by the applicant, but conditions of approval can be adopted to bring the proposal into compliance with all standards.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

6. The phasing plan, if requested, can be carried out in a manner that meets the objectives of the above criteria and provides necessary public improvements for each phase as it develops. *RESPONSE: The applicant did not submit a phasing plan. This subdivision request is for a 32 lot subdivision in one development phase.*

17.100.70 LAND DIVISION DESIGN STANDARDS

All land divisions shall be in conformance with the requirements of the applicable base zoning district and this chapter, as well as with other applicable provisions of this Code. Modifications to these requirements may be accomplished through a Planned Development. The design standards in this section shall be used in conjunction with street design standards included in the City of Sandy Transportation System Plan and standards and construction specifications for public improvements as set forth in adopted Public Facilities Plans and the Sandy Municipal Code.

RESPONSE: The applicant requested the following three Type III variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff supports the variance to front door orientation, the variance to allow a single-tiered lot configuration along Jacoby Road, and the variance to allow the block length to exceed 400 feet on the north side of Woodstock Street between Camden Court and the east property boundary. The tentative plat shall otherwise be designed to comply with all standards of the City of Sandy Development Code, Transportation System Plan, Facilities Master Plans and Sandy Municipal Code.

17.100.100 STREETS GENERALLY

No subdivision or partition shall be approved unless the development has frontage or approved access to an existing public street. In addition, all streets shall be graded and improved in conformance with the City's construction standards, approved by the City Engineer, in accordance with the construction plans.

A. <u>Street Connectivity Principle.</u> The pattern of streets established through land divisions should be connected to: (a) provide safe and convenient options for cars, bikes and pedestrians; (b) create a logical, recognizable pattern of circulation; and (c) spread traffic over many streets so that key streets (particularly U.S. 26) are not overburdened.

RESPONSE: The proposed development is moderately conducive to walking and biking while accommodating motor vehicles. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north.

B. <u>Transportation Impact Studies.</u> Transportation impact studies may be required by the city engineer to assist the city to evaluate the impact of development proposals, determine reasonable and prudent

transportation facility improvements and justify modifications to the design standards. Such studies will be prepared in accordance with the following:

- 1. A proposal established with the scope of the transportation impact study shall be coordinated with, and agreed to, by the city engineer. The study requirements shall reflect the magnitude of the project in accordance with accepted transportation planning and engineering practices. A professional civil or traffic engineer registered in the State of Oregon shall prepare such studies.
- 2. If the study identifies level-of-service conditions less than the minimum standards established in the Sandy Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered as part of the land use decision for the proposal.

RESPONSE: The submitted Transportation Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10^{ih} Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably per City of Sandy and ODOT standards and are projected to continue to operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development. The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and concurred that the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards.

C. <u>Topography and Arrangement.</u> All streets shall be properly related to special traffic generators such as industries, business districts, schools, and shopping centers and to the pattern of existing and proposed land uses.

RESPONSE: Considering the site's topography, the proposed street layout is acceptable given the topography and residential use of this site, and the topography and use of adjacent properties.

D. <u>Street Spacing</u>. Street layout shall generally use a rectangular grid pattern with modifications as appropriate to adapt to topography or natural conditions.

RESPONSE: The applicant proposes a rectangular pattern of streets with an offset grid. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north. However, future development to the south will be required to align with the proposed intersections in order to maintain a rectangular grid pattern and maximize pedestrian, bicycle, and vehicular connectivity. An offset grid with "T" intersections is not warranted due to the lack of through traffic.

E. <u>Future Street Plan.</u> Future street plans are conceptual plans, street extensions and connections on acreage adjacent to land divisions. They assure access for future development and promote a logical, connected pattern of streets. It is in the interest of the city to promote a logical, connected pattern of streets. All applications for land divisions shall provide a future street plan that shows the pattern of

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

existing and proposed future streets within the boundaries of the proposed land divisions, proposed connections to abutting properties, and extension of streets to adjacent parcels within a 400 foot radius of the study area where development may practically occur.

RESPONSE: The proposed local street plan has been designed to facilitate the traffic needs of this development while ensuring there are no intersection conflicts with future development. Per the City Engineer (Exhibit L), the applicant shall provide a profile design for a minimum of 200 feet for all future street extensions beyond the project boundary to ensure future street grades can be met.

- F. <u>Connections.</u> Except as permitted under Exemptions, all streets, alleys and pedestrian walkways shall connect to other streets within the development and to existing and planned streets outside the development and to undeveloped properties which have no future street plan. Streets shall terminate at other streets or at parks, schools or other public land within a neighborhood.
 - 1. Where practicable, local roads shall align and connect with other roads when crossing collectors and arterials.
 - 2. Proposed streets or street extensions shall be located to provide direct access to existing or planned transit stops, and existing or planned neighborhood activity centers, such as schools, shopping areas and parks.
 - 3. Exemptions.
 - a. A future street plan is not required for partitions of residentially zoned land when none of the parcels may be redivided under existing minimum density standards.
 - b. Standards for street connections do not apply to freeways and other highways with full access control.
 - c. When street connection standards are inconsistent with an adopted street spacing standard for arterials or collectors, a right turn in/right turn out only design including median control may be approved. Where compliance with the standards would result in unacceptable sight distances, an accessway may be approved in place of a street connection.

RESPONSE: The proposed design extends Newton Street east across Jacoby Road into the site. All proposed streets will allow connection with future development to the south and east, with the exception of Camden Court, which is a cul-de-sac. Due to topography and the location of Tickle Creek, it is not practicable to connect Camden Court to the north.

17.100.110 STREET STANDARDS AND CLASSIFICATION

Street standards are illustrated in the figures included at the end of this chapter. Functional definitions of each street type are described in the Transportation System Plan as summarized below.

- A. Major arterials are designed to carry high volumes of through traffic, mixed with some unavoidable local traffic, through or around the city. Major arterials should generally be spaced at 1-mile intervals.
- B. Minor arterials are designed to collect and distribute traffic from major and minor arterials to neighborhood collectors and local streets, or directly to traffic destinations. Minor arterials should generally be spaced at 1-mile intervals.
- C. Residential minor arterials are a hybrid between minor arterial and collector type streets that allow for moderate to high traffic volumes on streets where over 90% of the fronting lots are residential.

- D. Collector streets are designed to collect and distribute traffic from higher type arterial streets to local streets or directly to traffic destinations. Collector streets should generally be spaced at 1/2-mile intervals.
- E. Local streets are designed to provide direct access to abutting property and connect to collector streets. A general spacing of 8-10 local streets per mile is recommended.
- F. Cul-de-sacs and dead end streets are discouraged. If deemed necessary, cul-de-sacs shall be as short as possible and shall not exceed 400 feet in length.
- G. Public access lanes are designed to provide primary access to a limited number of dwellings when the construction of a local street is unnecessary.
- H. Alleys are designed to provide access to multiple dwellings in areas where lot frontages are narrow and driveway spacing requirements cannot be met.

RESPONSE: Jacoby Road is an existing collector street. All proposed streets in the subdivision are local streets, including the extension of the existing Newton Street, which is proposed to extend east of Jacoby Road into the site. Camden Court is a proposed cul-de-sac and does not exceed 400 feet. Per the City Engineer (Exhibit L), the plans don't clearly indicate the radius for the proposed cul-de-sac. The minimum paved cul-de-sac radius shall be 48 feet as per the Oregon Fire Code and Metro Code Committee. A review by the Fire Department to confirm cul-de-sac size compliance shall be required.

17.100.120 BLOCKS AND ACCESSWAYS

- A. <u>Blocks</u>. Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features.
- B. <u>Residential Blocks.</u> Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance.

RESPONSE: The applicant is requesting variances to Sections 17.100.120(A) and (B). The proposal includes one lot (Lot 30) that will be a single-tiered lot and one block (north side of Woodstock Street between Camden Court and the eastern property boundary) that exceeds 400 feet.

D. <u>Pedestrian and Bicycle Access Way Requirements.</u> In any block in a residential or commercial district over 600 feet in length, a pedestrian and bicycle accessway with a minimum improved surface of 10 feet within a 15-foot right-of-way or tract shall be provided through the middle of the block. To enhance public convenience and mobility, such accessways may be required to connect to cul-de-sacs, or between streets and other public or semipublic lands or through greenway systems.

RESPONSE: None of the proposed blocks exceed 600 feet in length. At the pre-application meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trail on the adjacent property to the east (tax lot 800). Staff brought this proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory

Board also recommends that the connector trail be constructed as part of this development. Staff recommends the Planning Commission require a pedestrian trail connection to the future extension of the Tickle Creek Trail as part of this development. Staff recommends that the trail connection be located in alignment with Middlebury Avenue. Alternatively, staff recommends that Lot 19 be included in the FSH tract and that the trail connection be located on the proposed location of Lot 19. If the applicant wants to pursue locating the pedestrian trail connection offsite, the applicant shall provide justification as to why a connection on tax lot 800 makes more sense than a connection developed as part of this proposal.

17.100.130 EASEMENTS

A minimum eight (8) foot public utility easement shall be required along property lines abutting a rightof-way for all lots within a partition or subdivision. Where a partition or subdivision is traversed by a watercourse, drainage way, channel or stream, the land division shall provide a stormwater easement or drainage right-of-way conforming substantially with the lines of such watercourse, and such further width as determined needed for water quality and quantity protection.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) details an 8 foot wide public utility easement along all street frontages. The plat shall detail all proposed easements as detailed in Section 17.84.90 above.

17.100.180 INTERSECTIONS

- A. <u>Intersections.</u> Streets shall be laid out so as to intersect as nearly as possible at right angles. A proposed intersection of two new streets at an angle of less than 75 degrees shall not be acceptable. No more than two streets shall intersect at any one point unless specifically approved by the City Engineer. The city engineer may require left turn lanes, signals, special crosswalks, curb extensions and other intersection design elements justified by a traffic study or necessary to comply with the Development Code.
- B. <u>Curve Radius.</u> All local and neighborhood collector streets shall have a minimum curve radius (at intersections of rights-of-way) of 20 feet, unless otherwise approved by the City Engineer. When a local or neighborhood collector enters on to a collector or arterial street, the curve radius shall be a minimum of 30 feet, unless otherwise approved by the City Engineer.

RESPONSE: The proposed intersections are all right angles and meet the required minimum spacing standard of 150 feet as required in Section 17.84.50(C)(2).

17.100.190 STREET SIGNS

RESPONSE: The applicant will be required to pay for the cost of all street signs and the City will install the street signs.

17.100.210 STREET LIGHTING

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

insufficient. The locations of street light fixtures shall be reviewed in detail with construction plans.

17.100.220 LOT DESIGN

- A. The lot arrangement shall be such that there will be no foreseeable difficulties, for reason of topography or other conditions, in securing building permits to build on all lots in compliance with the Development Code.
- B. The lot dimensions shall comply with the minimum standards of the Development Code. When lots are more than double the minimum lot size required for the zoning district, the subdivider may be required to arrange such lots to allow further subdivision and the opening of future streets to serve such potential lots.
- C. The lot or parcel width at the front building line shall meet the requirements of the Development Code and shall abut a public street other than an alley for a width of at least 20 feet. A street frontage of not less than 15 feet is acceptable in the case of a flag lot division resulting from the division of an unusually deep land parcel which is of a size to warrant division into not more than two parcels.
- D. Double frontage lots shall be avoided except where necessary to provide separation of residential developments from arterial streets or to overcome specific disadvantages of topography or orientation.
- E. Lots shall avoid deriving access from major or minor arterials. When driveway access from major or minor arterials may be necessary for several adjoining lots, the Director or the Planning Commission may require that such lots be served by a common access drive in order to limit possible traffic hazards on such streets. Where possible, driveways should be designed and arranged to avoid requiring vehicles to back into traffic on minor or major arterials.

RESPONSE: The Single Family Residential (SFR) zoning district requires lots at least 7,500 square feet in area. The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes are required to comply with setback standards and maximum building height limitations as required in Chapter 17.34. Lot 30 is a double frontage lots that will take access from Camden Court but also abut Jacoby Road. Lots 19-21 and 27 contain more than double the minimum lot size; however, a large portion of each lot's area is within the FSH and, therefore, cannot be divided into a separate buildable lot. To avoid potential conflict between private property owners and the FSH overlay area in the future, the applicant shall designate the FSH analysis area as a separate tract and dedicate the tract to the City of Sandy. No lots are proposed to be accessed from a major or minor arterial. All lots are required to comply with clear vision requirements at all intersections.

17.100.230 WATER FACILITIES

Water lines and fire hydrants serving the subdivision or partition, and connecting the development to City mains, shall be installed to provide adequate water pressure to serve present and future consumer demand. The materials, sizes, and locations of water mains, valves, service laterals, meter boxes and other required appurtenances shall be in accordance with the standards of the Fire District, the City, and the State.

If the city requires the subdivider to install water lines in excess of eight inches, the city may participate in the oversizing costs. Any oversizing agreements shall be approved by the city manager based upon council policy and dependent on budget constraints. If required water mains will directly serve property

outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement for the proportionate share of the cost.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of water facilities will be reviewed with construction plans. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. There is no other proposed connection to the water distribution system. Per the Public Works Director (Exhibit O), the applicant shall demonstrate that adequate fire and domestic flow will be available with a single point of connection to the water distribution system or connect to the existing 8-inch water line at the intersection of Street and Amherst Street. Per the City Engineer (Exhibit L), all new waterline mains shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The applicant's proposed Utility Plan (Exhibit C, Sheet 5) depicts new hydrants. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with construction plans.

17.100.240 SANITARY SEWERS

Sanitary sewers shall be installed to serve the subdivision and to connect the subdivision to existing mains. Design of sanitary sewers shall take into account the capacity and grade to allow for desirable extension beyond the subdivision.

If required sewer facilities will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement by nonparticipating landowners for the proportionate share of the cost of construction.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of sanitary sewer facilities will be reviewed with construction plans. Per the City Engineer (Exhibit L), all new public sanitary sewer lines shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The proposed utility plan shows an 8-inch sanitary sewer line extended south from the existing sewer line in Jacoby Road approximately 490 feet to the site. The site will drain to the existing SE sewer pump station located on Jacoby Road south of Trillium Avenue. Per the Public Works Director (Exhibit O), the applicant shall provide a sanitary sewer lateral serving Tax Lot 6100 (located directly north of the subject property) with the sanitary sewer main line extension.

17.100.250 SURFACE DRAINAGE AND STORM SEWER SYSTEM

- A. Drainage facilities shall be provided within the subdivision and to connect with off-site drainage ways or storm sewers. Capacity, grade and materials shall be by a design approved by the city engineer. Design of drainage within the subdivision shall take into account the location, capacity and grade necessary to maintain unrestricted flow from areas draining through the subdivision and to allow extension of the system to serve such areas.
- B. In addition to normal drainage design and construction, provisions shall be taken to handle any drainage from preexisting subsurface drain tile. It shall be the design engineer's duty to investigate the location of drain tile and its relation to public improvements and building construction.
- C. The roof and site drainage from each lot shall be discharged to either curb face outlets (if minor quantity), to a public storm drain or to a natural acceptable drainage way if adjacent to the lot.

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A) to be dedicated to the City of Sandy. Detained and treated discharge from the detention pond is proposed to be discharged to the adjacent property to the north. Per the Public Works Director(Exhibit O), the applicant shall demonstrate that the proposed subdivision does not exceed pre-development site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way. Per the City Engineer (Exhibit L), the detention pond shall meet the requirements of the 2016 City of Portland Stormwater Management Manual (SWMM) for landscaping, Section 2.4.1, and escape route, Section 2.30. All new public storm drains shall be a minimum of 12-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The City Engineer (Exhibit L) states the submitted preliminary stormwater calculations meet the water quality and water quantity criteria as stated in the City of Sandy Municipal Code Chapter 13.18 Standards and the City of Portland current Stormwater Management Manual (SWMM) Standards that were adopted by reference into the Sandy Development Code. Per the City Engineer, the applicant shall submit a detailed final stormwater report stamped by a licensed professional to the City for review and approval with the final construction plans.

17.100.260 UNDERGROUND UTILITIES

All subdivisions or major partitions shall be required to install underground utilities (including, but not limited to, electrical and telephone wiring). The utilities shall be installed pursuant to the requirements of the utility company.

RESPONSE: All utilities shall be placed underground.

17.100.270 SIDEWALKS

Sidewalks shall be installed on both sides of a public street and in any special pedestrian way within the subdivision.

RESPONSE: The applicant proposes constructing sidewalks along all public street frontages, with the exception of the proposed three-quarter street section and one-half street section of Newton Street.

17.100.290 STREET TREES

Where planting strips are provided in the public right-of-way, a master street tree plan shall be submitted and approved by the Director. The street tree plan shall provide street trees approximately every 30' on center for all lots.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires installation of trees along all street frontages. Street trees are discussed in Section 17.92.30 of this staff report.

17.100.300 EROSION CONTROL

Grass seed planting shall take place prior to September 30th on all lots upon which a dwelling has not been started but the ground cover has been disturbed. The seeds shall be of an annual rye grass variety and shall be sown at not less than four pounds to each 1000 square feet of land area.

RESPONSE: The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a re-vegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code and as detailed below. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

17.100.310 REQUIRED IMPROVEMENTS

The following improvements shall be installed at no expense to the city, consistent with the design standards of Chapter 17.84, except as otherwise provided in relation to oversizing.

- A. Drainage facilities
- B. Lot, street and perimeter monumentation
- C. Mailbox delivery units
- D. Sanitary sewers
- E. Sidewalks
- F. Street lights
- G. Street name signs
- H. Street trees
- I. Streets
- J. Traffic signs
- K. Underground communication lines, including broadband (fiber), telephone, and cable. Franchise agreements will dictate whether telephone and cable lines are required.
- L. Underground power lines
- M. Water distribution lines and fire hydrants

RESPONSE: The applicant shall be responsible for the installation of all improvements detailed in Section 17.100.310, including fiber facilities. SandyNet requires the developer to work with the City to ensure that broadband infrastructure meets the design standards and adopted procedures as described in Section 17.84.70. Entry monument signs shall be located entirely outside the public rightof-way and clear vision areas as required by Section 17.74.30. If entry signs are desired, staff recommends the applicant submit a detailed plan showing the location of such signage.

CHAPTER 17.102 – URBAN FORESTRY

17.102.20 APPLICABILITY

This chapter applies only to properties within the Sandy Urban Growth Boundary that are greater than one acre including contiguous parcels under the same ownership.

- A. General: No person shall cut, harvest, or remove trees 11 inches DBH or greater without first obtaining a permit and demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance, the applicant shall agree to implement required provisions of this chapter and to allow all inspections to be conducted.
 - 2. Tree removal is subject to the provisions of Chapter 15.44, Erosion Control, Chapter 17.56, Hillside Development, and Chapter 17.60 Flood and Slope Hazard.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

- B. Exceptions: The following tree removals are exempt from the requirements of this chapter.
 - 1. Tree removal as required by the city or public utility for the installation or maintenance or repair of roads, utilities, or other structures.
 - 2. Tree removal to prevent an imminent threat to public health or safety, or prevent imminent threat to public or private property, or prevent an imminent threat of serious environmental degradation. In these circumstances, a Type I tree removal permit shall be applied for within seven days following the date of tree removal.

RESPONSE: Section 17.00.10 states the overall purpose of the Development Code, which is to ensure that development is consistent with the goals and policies of the Sandy Comprehensive Plan. Goal 5 of the Comprehensive Plan addresses natural resources, with a stated intention of establishing policies for conservation of open space and protection of natural and scenic resources in compliance with Statewide Planning Goal 5. The Comprehensive Plan states, "In keeping with the mandate of Statewide Planning Goal 5, Oregon communities are required to provide programs that will 'conserve open space and protect natural and scenic resources.' Additionally, the City of Sandy has the goal of providing parks, open space, and recreational opportunities for its citizens, as well as promoting healthy and visually attractive environments that are in harmony with the natural landscape character of the area." Goal 5 of the Comprehensive Plan includes policies related to natural resources, open space, stream corridor protection, indigenous growth protection, historic and cultural resource protection, and environmental and cultural resource protection. The most relevant policies related to tree retention and mitigation include the following:

- Policy 4: The City promotes development practices which maintain or enhance the environmental quality enjoyed by the residents of the community.
- Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.
- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 18: Where feasible, preserve natural vegetation resource sites through public acquisition, conservation easements or other available methods to permanently limit development.
- Policy 19: Encourage and, if necessary, require reforestation in open space areas to help mitigate the adverse impacts of development.
- Policy 27: The City of Sandy shall promote the planting of trees throughout the community on both public and private properties.
- Policy 28: The City of Sandy shall consider the establishment of a street tree plan that identifies particular trees within different areas of the City but that provide species diversity to add visual interest throughout the community, create individual neighborhood identity, and provide additional resistance through species diversity against disease.
- Policy 29: The City of Sandy shall review existing development standards, including street standards, to assure that the standards support and encourage the maintenance of a sustainable urban forest.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

 Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.

As discussed in further detail below, the 9.68 acre site requires retention of a minimum of 29 trees that are in good condition and are 11-inches DBH or greater. Of the 521 trees inventoried on the site, the applicant is proposing to remove 457. Per Section 17.92.10(C), significant trees (8-inches or greater DBH) should be preserved to the greatest extent practicable. In addition, the City of Sandy Comprehensive Plan encourages development that minimizes disturbance of native tree stands, preserves natural open space, and promotes innovative site and building design to minimize the adverse impacts of development on native tree stands.

17.102.50 TREE RETENTION AND PROTECTION REQUIREMENTS

- A. Tree Retention: The landowner is responsible for retention and protection of trees required to be retained as specified below:
 - 1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous ownership.
 - 2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.
 - 3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.
 - 4. If possible, at least two of the required trees per acre must be of conifer species.
 - 5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements.
- **B.** Tree Protection Area: Except as otherwise determined by the Planning Director, all tree protection measures set forth in this section shall be instituted prior to any development activities and removed only after completion of all construction activity. Tree protection measures are required for land disturbing activities including but not limited to tree removal, clearing, grading, excavation, or demolition work.
 - 1. Trees identified for retention shall be marked with yellow flagging tape and protected by protective barrier fencing placed no less than 10 horizontal feet from the outside edge of the trunk.
 - 2. Required fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade.
 - 3. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles.
- C. **Inspection.** The applicant shall not proceed with any tree removal or construction activity, except erosion control measures, until the City has inspected and approved the installation of tree protection measures. Within 15 days of the date of accepting an application for a Type I permit, the city shall complete an onsite inspection of proposed activities and issue or deny the permit. Within 15 days of issuing a Type II or Type III permit, the city shall complete an onsite inspection of proposed activities.

For ongoing forest operations, the permit holder shall notify the city by phone or in writing 24 hours prior to subsequent tree removal. The city may conduct an onsite re-inspection of permit conditions at this time.

RESPONSE: The subject property is 9.68 acres, which requires a minimum of 29 retention trees that are 11-inches or greater DBH and in good health. Per the submitted Tree Retention Inventory (Exhibit C, Sheet 2), the applicant proposes to retain all trees within the FSH overlay (64 total), including 33 trees that are viable and 11-inches or greater DBH. It appears that there are two trees that are proposed for removal located on the FSH boundary line: Tree #510 and Tree #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay. Of the 33 viable trees identified for retention, six (6) appear to be on or very close to the eastern property boundary (Trees # 527, 528, 530, 549, 550, and 3006), and an additional four (4) include notes such as "poor trunk taper," "windthrow potential," "wound on trunk," or "some trunk decay" (Trees # 520, 531, 534, and 560). Without those nine (9) trees, the proposal would only include 24 retention trees. Therefore, staff recommends that the applicant retain additional trees. In addition, Section 17.92.10(C) of the Development Code states that significant trees (defined as 8inches or greater DBH) should be preserved to the greatest extent practicable and integrated into the design of a development. This is also in line with the Comprehensive Plan, particularly Policies 16 and 17. The applicant shall update the plan set to include a minimum of an additional 10 retention trees 8-inches or greater DBH and in good health. Staff recommends retaining all trees on Lot 19 and including the area as part of the FSH tract dedicated to the City of Sandy. Alternatively, staff recommends the applicant consider directional boring in the utility easements to retain Trees #109, 110, 120, 122, 336, 340, 366, 380, 383 and 392, and/or reducing the square footage of Lots 9 and 10 and creating an approximately 20 foot wide tree preservation buffer tract along Jacoby Road, which shall also be placed in a conservation easement. A tree buffer along Jacoby Road would also support the applicant's request to have the houses on Lots 9 and 10 face the internal street network rather than Jacoby Road.

Section 17.102.50(B.1) requires tree protection fencing be placed no less than 10 horizontal feet from the outside edge of the trunk. Per the Pacific Northwest International Society of Arboriculture (ISA), the ISA defines the critical root zone (CRZ) as "an area equal to a 1-foot radius from the base of the tree's trunk for each 1 inch of the tree's diameter at 4.5 feet above grade (referred to as diameter at breast height)." Often the drip-line is used to estimate a tree's CRZ; however, it should be noted that a tree's roots typically extend well beyond its drip-line. Section 17.92.10(C) requires tree protection fencing located 5 feet beyond the drip-line of a tree, taking into account that there are many roots beyond the drip-line and that trees continue to grow. Tree protection fencing shall be located at least 5 feet beyond the dripline of all retention trees. Tree protection fencing shall be located at least 5 feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. The applicant shall request an inspection of tree protection measures prior to any tree removal, grading, or other construction activity on the site.

OTHER CONSIDERATIONS FOR TREES:

To ensure protection of the required retention trees, the applicant shall record a tree protection covenant specifying protection of these trees and limiting removal without submittal of an Arborist's Report and City approval. This document shall include a sketch identifying the required retention trees. All trees marked for retention shall be retained and protected during construction regardless of desired or proposed building plans; plans for future houses on the

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

proposed lots within the subdivision shall be modified to not encroach on retention trees and associated tree protection fencing.

The properties directly north and east of the subject site (tax lots 6100 and 800) contain a number of existing trees, some of which are located close to the shared property line and have canopies that extend onto the subject property. The applicant shall update the plan set to detail tree protection fencing on the subject property located 5 feet beyond the dripline of all trees located on the properties to the north and east.

CHAPTER 17.66 – ADJUSTMENTS AND VARIANCES

17.66.60 VARIANCES

Variances are a means of requesting a complete waiver or major adjustment to certain development standards. They may be requested for a specific lot or as part of a land division application. The Type II variance process is generally reserved for major adjustments on individual lots, while variances to development standards proposed as part of a land division are processed as a Type III application (requiring a public hearing).

RESPONSE: The applicant proposes the following three Type III variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

17.66.70 TYPE II AND TYPE III VARIANCE CRITERIA

The authority to grant a variance does not include authority to approve a development that is designed, arranged or intended for a use not otherwise approvable in the location. The criteria are as follows:

- A. The circumstances necessitating the variance are not of the applicant's making.
- B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located.
- C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan.
- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land.

F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control.

BLOCK LENGTH VARIANCE

17.100.120 (B) Residential Blocks. "Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance."

Request: The applicant requests approval to exceed the 400 foot block length standard for the north side of Woodstock Street between Camden Court and the eastern property boundary (approximately 426 feet).

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin. The City Engineer reviewed the future street plan (Exhibit C, Sheet 8) and states "given the wetlands/Tickle Creek is [sic] located to the north, a stubbed street is not feasible. Extending the block length for approximately 100 feet appears to be reasonable and is compatible with the location of the existing house on the adjacent property to the east."

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance. Granting a block length variance will better protect the natural resources on the site.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." **RESPONSE:** Approval of the variance will not have an impact on any of the policies or goals of the Comprehensive Plan or any other applicable policies and standards adopted by the City. The applicant's narrative (Exhibit B) states: "Granting this variance will further the City'[s] goal of protecting natural resources and will not adversely affect implementation of the Comprehensive Plan."

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." *RESPONSE:* The proposed development will not be significantly different than if Woodstock Street was limited to 400 feet in length. Inclusion of a connector trail to the future Tickle Creek Trail on the subject property north of Woodstock Street will break up the block length and provide additional pedestrian connectivity.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin.

Recommendation: Staff recommends the Planning Commission approve a variance to allow the north side of Woodstock Street between Camden Court and the eastern property boundary to exceed 400 feet as proposed in the submitted plan set. Staff's recommendation to include the connector trail to the future Tickle Creek trail on the subject property will effectively reduce the block length for pedestrians and will improve pedestrian connectivity in the future.

BUILDING ORIENTATION VARIANCES

The requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located.

Request: The applicant requests a Type III variance to Chapter 17.82.20 to allow the front door of the future homes constructed on Lots 9, 10, 30 and 32 to face the internal local street network instead of Jacoby Road, a designated transit street.

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The Development Code does not allow driveway access to higher classification streets such as Jacoby Road, a collector street. The front door could be oriented to Jacoby Road with a rear or side loaded garage oriented to the internal street network; however, staff recognizes that the front doors on Jacoby Road would essentially be false front doors, which is not the internet of the code.

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." *RESPONSE:* The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity. Orientation of the houses on Lots 9, 10, 30, and 32 to the internal street network will have no negative effect on surrounding properties. Per the applicant's narrative (Exhibit B), "Homes directly across Jacoby Road from the subject property have a similar configuration to what is being proposed."

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

some economic use of the land." **RESPONSE:** Orienting the houses on Lots 9, 10, 30, and 32 to the local street will allow construction of a traditional structure containing a garage and front door on the same street elevation, which will create a consistent look in the neighborhood. However, the requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located. If Planning Commission approves this variance, staff recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30 and 32 per the following:

- Builders of individual lots shall construct a decorative fence on the Jacoby Road facing side of Lots 9, 10, 30, and 32 to enhance the visual appeal of these lots from Jacoby Road. The fence shall include the following design details:
 - Constructed of vertical black metal or faux metal fencing material.
 - No greater than 3-inch gap between vertical pickets.
 - o 4-feet to 6-feet in height.
- The rear elevations of Lot 30 and the side elevation of Lots 9, 10, and 32 shall incorporate all of the following design details where applicable:
 - Decorative gables including two or more of the following:
 - A window with grids.
 - A trimmed vent. The trim must match the trim on the windows and the vent must be at least 4 square feet in area.
 - Cross or diagonal bracing, shingles, trim, corbels, exposed rafter ends, or brackets.
 - Decorative 'belly-band' with an alternative paint color to the siding color, between building floors.
 - *Mixture of siding materials, including shake or horizontal lap siding with an alternative paint color to the primary siding color.*
 - Recessed or covered rear entries (rear elevation only).
 - The covered area must be at least 48 square feet and a minimum of 8 feet wide.
 - The recessed entry must feature vertical support posts.
 - Minimum four-inch wide trim or 12-inch wide shutters around all windows.

Builders shall submit proposed decorative fence and elevation designs for staff review and approval.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." **RESPONSE:** It would create an unconventional design to have the rear of the houses on Lots 30 and 32 facing the proposed Camden Court cul-de-sac. While the front doors of the houses on Lots 9 and 10 could face Jacoby Road, the garages would not be able to gain access from Jacoby Road. A more traditional orientation would be to have both the front door and garage face the local street as requested by the applicant.

Recommendation: Staff recommends the Planning Commission approve a variance to allow the front door of the future homes constructed on Lots 9, 10, 30, and 32 to not face Jacoby Road. Staff further

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx
recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30, and 32.

TWO-TIERED LOT VARIANCE

17.100.120 (A) Blocks. "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Request: The applicant requests approval to provide a one-tier lot configuration for Lot 30 between Jacoby Road and Camden Court, instead of the required two-tier configuration.

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The applicant's narrative (Exhibit B) states that the design of the proposed subdivision, including the single-tiered lot on the cul-de-sac, is due to "topographic and natural resource constraints of the property associated with the location of Tickle Creek and steep slopes." The proposal only includes one single-tiered lot. In addition, the Development Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A), which states: "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." *RESPONSE:* The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." **RESPONSE:** The proposal contains one single-tiered lot (Lot 30) and the proposed development will not be significantly different than if the applicant proposed a two-tiered lot configuration so that Lot 30 was somehow not oriented as a single tiered lot.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The applicant's narrative states "the subject property contains unique circumstances (collector street, steep slopes, and creek) that warrant approval of an alternative design." The Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A) as explained in Criteria A.

Recommendation: Staff recommends the Planning Commission approve the applicant's request to allow one-tiered lot configuration for Lot 30 between Jacoby Road and Camden Court.

CHAPTER 15.30 - DARK SKY ORDINANCE

All exterior lighting is required to conform to the requirements of this chapter.

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined insufficient. The locations of these fixtures shall be reviewed in detail with construction plans. Full cut-off lighting shall be required.

CHAPTER 15.44 - EROSION CONTROL

15.44.20 AREA OF APPLICATION

This chapter applies to all ground disturbing activities within the city limits whether or not a permit is required, unless such activities are otherwise exempted by the Sandy Development Code. All non-permitted ground disturbing activities that are permanent or temporary in nature shall comply with this chapter unless otherwise noted.

RESPONSE: All on-site earthwork activities including any retaining wall construction should follow the requirements of the current edition of the Oregon Structural Specialty Code (OSSC). If the proposal includes a retaining wall, the applicant shall submit additional details on the proposed retaining wall for staff review and approval. All on-site earthwork activities should follow the Geotechnical Report (Exhibit G) prepared by Rapid Soil Solutions LLC, dated February 23, 2018. Any deviations from the initial design parameters shall be reviewed by Rapid Soil Solutions for applicability. The City Engineer (Exhibit L) reviewed the Geotechnical Report and recommends that the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the above noted report and OSSC. Site grading should not in any way impede, impound or inundate the adjoining properties. All the work within the public right-of-way and within the paved area should comply with American Public Works Association (APWA) and City requirements as amended. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a revegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code. The proposed subdivision is greater than one acre which typically requires approval of a DEO 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

15.44.50 APPLICANT'S RESPONSIBILITY

The developer, subsequent developers or property owners shall bear responsibility for the installation, construction, protection of all stockpiles on the site, materials transported from the site, maintenance including re-vegetation of all graded areas, inspection and disposition of erosion control measures. In order to meet the purpose of this chapter, the developer shall:

- C. Apply permanent or temporary soil stabilization to denuded development site areas in conformance with the following schedule:
 - 1. Between October 1 and April 30, all denuded sites shall immediately be provided with either temporary or permanent soil stabilization.
 - 2. Between May 1 and September 30, temporary erosion and sediment control measures to reduce dust and sediment transport shall be applied as soon as practicable, but in no case more than seven days after ground disturbing activity occurs.
 - 3. Ground cover shall be installed on any portion of a site that is denuded for more than six months. Sports fields or playgrounds surrounded by vegetative cover or permanently installed curbing are exempt from this requirement.
 - 4. Temporary measures shall be maintained until permanent measures are established.
 - 5. Plant replacement vegetative cover that does not include plants listed as nuisance or prohibited plants on the City of Portland Plant List (available in the Planning & Development office) or the City's Native Plant List. Agriculture, timber production or residential crop growing activities are exempted from this requirement.
 - 6. Secure or protect soil stockpiles throughout the project with temporary or permanent soil stabilization measures. Depositions of soil may be subject to additional regulations requiring permit, review or erosion and sediment control.
 - 7. Select BMPs from the Erosion Control Manual.

RESPONSE: Section 15.44.50 contains requirements for maintenance of a site including revegetation of all graded areas. The applicant's Erosion Control Plan shall be designed in accordance with the standards of Section 15.44.50.

III. SUMMARY AND CONCLUSION

All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

As detailed in this report, the proposed subdivision generally complies with most Development Code requirements.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

Biggest Issues/Staff Recommendations:

- Add pedestrian connector trail, or provide justification as to why a connection on tax lot 800
 makes more sense than a connection developed as part of this proposal.
- Create separate FSH tract and dedicate it to the City.
- Add additional decorative fence and building features on Jacoby Road for Lots 9, 10, 30, and 32.
- Provide a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay.
- Design Newton Street with a three-quarter street section to the eastern property boundary.
- Identify a minimum of 10 additional retention trees.

IV. RECOMMENDATION

Staff recommends the Planning Commission hold a public hearing to take public testimony regarding the proposal. Staff recommends the Planning Commission **approve** the subdivision application with modifications as recommended in this report. The staff recommendation includes approval of the following three variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff recommends that the Planning Commission interpret the code such that Chapter 17.92, Landscaping and Screening, continue to apply to all zones, including subdivisions.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC report.docx

Exhibit V

Jacoby Heights Subdivision

2/23/2019

Arborist Notes for Jacoby Heights Subdivision

The purpose of these notes is to add details and clarification to the Tree Preservation Plan for Jacoby Heights Subdivision and to address specific trees mentioned in the City of Sandy Staff Report. All trees in the FSH area near Tickle Creek will be preserved regardless of condition. This is a natural area populated by native trees that vary in age, species, and condition. Preserving trees in a stand near the creek is ideal and provides the greatest benefits for tree health, wildlife, and retention of site characteristics. There are three issues mentioned in the City of Sandy Staff Report to be addressed.

- "Section 17.92.10(C) requires tree protection fencing located 5 feet beyond the drip-line of a tree, taking into account that there are many roots beyond the drip-line and that trees continue to grow." (Staff report, pg. 49) The tree protection fence (TPF) will be installed roughly following the south and west limits of the FSH area and terminating at the north and east property lines. (See Figure 1) The TPF will be installed 5 feet beyond the drip-lines of trees being preserved that border the TPF. Trees being preserved within the core of the FSH will have ample protection.
- "The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay." (Staff report, pg. 49) T510 (Tree 510) and T516 will be added to the preservation plan. These two trees are at the edge of the FSH (See Figure 1). A grading cut will be made to the southwest of these two trees. Grading may encroach into the root protection zones (RPZs) of the two trees but will be no closer than 10 ft. from either tree. Grading will slope away from the trees which preserves more of the root crown than a vertical cut. The project arborist will be on-site to supervise grading that occurs within the RPZs of these trees and to prune roots if needed.
- "Of the 33 viable trees identified for retention, six (6) appear to be on or very close to the eastern property boundary (Trees # 527, 528, 530, 549, 550, and 3006), and an additional four (4) include notes such 'poor trunk taper,' 'windthrow potential,' 'wound) on trunk', or 'trunk decay' (Trees # 520, 531, 534, and 560)." (Staff report, pg. 49) Trees listed as "viable" in the "remarks" column of the Tree Table are suitable for preservation and consistent with City of Sandy Code 17.102.50 A 3, which states, "Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest". Defects included in the "remarks" column aid the arborist in determining which trees to preserve in concert with the development plant. For example, a viable tree with "poor trunk taper" or "windthrow potential" can be preserved as part of a stand but may not be suitable for preservation as a single tree after the stand has been removed. Similarly, minor defects noted about "viable" trees that do not necessarily affect survivability or lead to tree mortality are considered if proximity to construction will lead to additional stress that will affect the tree's ability to compartmentalize or outgrow the defect. The mention of these defects does not mean the trees are hazardous or declining.

2/23/2019

As trees being preserved are largely part of, and adjacent to, a mature stand of native trees they have a strong likelihood of survival and reduced risk of windthrow. I would add that because there will be no disturbance or tree removal within the FSH, tree health will be maximized in this area. There are many trees within the FSH which were not listed for preservation because of decline or significant defect but will be retained. (See Table 2.) The retention of these trees is also a great benefit in the preservation of the site's characteristics as these will continue to reproduce, provide habitat, build soil, guard against erosion, and protect developing trees.

17.102.50 TREE RETENTION AND PROTECTION REQUIREMENTS

A. Tree Retention: The landowner is responsible for retention and protection of trees required to be

retained as specified below:

1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous

ownership.

2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.

3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.

4. If possible, at least two of the required trees per acre must be of conifer species.

5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements.

	western			
510	redcedar	15	viable; suppressed; co-dominant	10
	western			
516	redcedar	20	viable	12
	western			
520	hemlock	14	viable; wound on trunk	10
531	Douglas fir	12	viable; poor trunk taper; windthrow potential	8
534	Douglas fir	15	viable; poor trunk taper	10
	western red			
560	cedar	33	viable; some trunk decay; at creek	22

Table 1. (Selections from Arborist's Tree Table.)

2/23/2019



Figure 1. Map of FSH and Tree Preservation Area. (Source: Sheet 2 of Development Plan.)

14

2/23/2019

AG NUMBER	STREET	DAN	CONDITION	ACTION
to no month	CT BCLCS	-	California	
IN1	COLOR FORMER	10	VIARE	CHE-MAN
10.00	LETTERS (PEC CELLAR	- Digitize		C COST INC.
5/8	DOUBLAS FIR		VIABLE	THESERV
-	ARTING HEREITS	14	COLL CORE OF THE	C FRENCE VI
	E BHINE (EU)	10	SHOLE NO TAN INFELD	THERE
	A CONTRACTOR OF A			A RECEIPTION OF
224		42	BASAL DECAY, TRINK	
Dolar.			DECAY FOR 30 PT	T Passan Ve
CONTRACTOR OF	SECOND REPORT		LIDED HING UP IN THET	
0.36	DESIGNN RED CEDAR		525	Process Vo
	CENTRO PER LETAP	-		
\$29	NET TOTAL RED CEDAR		VIABLE	PRESERV
1.442	E IDNING EM	6	-408	PHENENN
	Boldenia and a more than		CARLE COST TOM	
			FORMAL STREET	in the second
837	DOUGLAS PIR		VIABLE POOR TRUK TAPER UNIOTHROU	PRESERVE
			POIDNIAL	
533	DOUBLAS FIR		VACLE, SUPPRESED	- Ballan
1946	Dougleside		TATER	Larginol
dec	NETTERN HETLOCK	10	RESPONSE DROUTH	PRESERVI
			WHETHERE POTENTIAL	
936	SERIER HERLOCK	8	TOP WAL DELAS	FREDERV
60/1		1	SUCCESSION AND	HED
610	NESTERN HEPLOCK	0	HECHANICAL DAPLACE TO	meserv
			TO TRENK	
and the second	PROBAN NEW CO.	-	10.00	PROBNI
540	DOUBLAS FIR		LISTED, HUND SP IN TREEL	PRESERV
841	RED ALDER	10	VIABLE, TRUNK SECOP	- Interiory
740	· · · · · · · · · · · · · · · · · · ·		1	1.28
843	DOUDLAS PHR		VIABLE	THE REAL PROPERTY
544	DOLELAS PH		VIACLE	PAGERY
S-de-	COLLAPE		- HILE	Ingen-
			LISTED, AND SELF-	
SH1	NED ALLEN	0	LEAN	PHEREN
645	RED ALDER	14	VARE OF	THE REAL
	ABD ALL BR		WEAT AT AT A	
Den		14	Here's Permit	Present
16.0	UNITED OF STREET	10.	Viele	1+ BORNES
564	MENTRESCOND CRIDER	i telefond	CONTRACTOR AND INCOME.	FFEREN
905	DEAD	5	CEAD	PRESERV
584	RED ALDER		VIABLE INCIDENTE	TREATIN
- Ma				
COLUMN COLUMN	STELLED FOR	in the second se	And A crep	P PALIE N
Concernance of the second s	Art Nikel Altis Litt Alte		VALLE STRE TO BE	Hard Street
		-	TELAT AT DED	
140			EXCEDONE LEAN TRINK	
963	RED ALDER	,	BOUS 60 PT TO SOUTH	LAICE DAY
56.3	RED ALDER	-	TERMINAL DECLINE	PREDERV
Bad	HED ALCEN	6	VIABLE	(The second
54.6	NTL ALLEN		VALL	A PAL DA
SUT	NEL ALCEN		CONTRACT LEGAL TOWARDS	CHERRY .
	WALKS FALLE	-		1
540	DIGLEAT HAPLE	3		PRESERV
910	LESTIN HETLOCK	1	VIABLE INCERNIZE	-
מיפ	MESTERN RED CEDAR	л	BOUND ON TRUNK, TRUNK	FREDERIV
8724	stamping pro circus	-	WOUND ON BASE, MOOD	and the second second
9 4/A		-	BORERS	- HEUCHY
9700	HEATERN RED CED4R	43	BACE DROKEN STERL LOW	mean
973	REPORTED AND CETAL	1	VIGOR VIGOR	part of the
\$14	CENTERN RED CEDAR	1.	VIABLE	FREDERY
876	DIGLEAR PLAPER	1 10	BRICKEN STEPL TRUNK	page-page
	personal i may bits	-	DECAT, NE-TRENCHED	· · · ·
1000	Lan wran	1. 10	AND APR	1 States
are.	Distant Provide	1	SULF COMMENTED ADDRES	INCOM
	Marka waka	-	In Annual State	
and the second	ETTHE HE LETTER	1	SUPER SHIPS BACK IS 45	Intrasi
	I CONTRACTOR OF THE	1.000	APRIL 10	La Carte
		-		the second s

Table 2. Trees to be preserved. (Source: Sheet 2 of Development Plan.)

2/23/2019

Assumptions and Limiting Conditions

Portland Tree C	Consulting	PO Box 19042	Portland, OR 97280
503.421.3883	info@pdxtre	econsulting.com	CCB 154349

- 1. Client warrants any legal description provided to the Consultant is correct and titles and ownerships to property are good and marketable. Consultant shall not be responsible for incorrect information provided by Client.
- 2. Consultant can neither guarantee nor be responsible for the accuracy of information provided by others.
- 3. The Consultant shall not be required to give testimony or attend court or hearings unless subsequent contractual arrangements are made, including additional fees.
- 4. The report and any values expressed therein represent the opinion of the Consultant, and the Consultant's fee is in no way contingent upon the reporting of a specified value, a stipulated result, the occurrence of a subsequent event, nor upon any finding to be reported.
- 5. Sketches, drawings and photographs in the report are intended as visual aids and may not be to scale. The reproduction of information generated by others will be for coordination and ease of reference. Inclusion of such information does not warrant the sufficiency or accuracy of the information by the Consultant.
- 6. Unless expressed otherwise, information in the report covers only items that were examined and reflects the condition at the time of inspection. The inspection is limited to visual examination of accessible items without laboratory analysis, dissection, excavation, probing, or coring, unless otherwise stated.
- 7. There is no warranty or guarantee, expressed or implied, that problems or deficiencies of the plants or property in question may not arise in the future.
- 8. The report is the completed work product. Any additional work, including production of a site plan, addenda and revisions, construction of tree protection measures, tree work, or inspection of tree protection measures, for example, must be contracted separately.
- 9. Loss or alteration of any part of the report invalidates the entire report.

Ryan Neumann

ISA Certified Arborist PN-5539A

TRAQ Qualified

Exhibit W

Date: February 25, 2019
To: Sandy Planning Commission
From: Tracy Brown, Tracy Brown Planning Consultants, LLC Ray Moore, All County Surveyors & Planners, Inc

RE: Review of Staff Recommended Conditions - File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision

<u>Introduction</u>: This document reviews each of the recommended conditions contained in the staff report for this project. The conditions we will discuss are shown in **bold** type and conditions we agree with are in shaded text. Our presentation groups these conditions into the following categories:

- FSH Overlay and Trail Conditions 4, 5, 19
- Trees Conditions 1, 2, 14, 15, 16, 17, 18
- Planning Conditions Conditions 3, 6, 12, 13, 25, 26, 27, 32, 35, 78
- Engineering Conditions Conditions 28, 30, 31, 34, 38, 40, (48, 52, 59) regarding utilities) same as 45, 49, 67, (65, 72, 73) similar regarding erosion control, (66, 68, 69, 71) similar regarding geotechnical reporting, 79

<u>Exhibits</u>

- 1. NWI/LWI
- 2. TSP Pedestrian System Plan
- 3. Tree Retention Exhibit A (trees 549,550, 3006)
- 4. Tree Retention Exhibit B (trees 527,528, 529, 530)

FSH/Trees/Landscaping

1. Staff recommends that the Planning Commission continue to apply Chapter 17.92 to all zones, including during subdivision review.

Response: The applicant requests this condition be removed. A formal request for interpretation was submitted with this application because we believe staff is incorrectly applying the requirements of Chapter 17.92, specifically Section 17.92.10(C) to this project. Chapter 17.14 contains standards for submitting requests for interpretations and identifies both minor (single property) and major (multiple property or zone) interpretations. Our submittal for this request is contained in Exhibit D beginning on page 132 of the online application materials. Since the Commission's decision on the interpretation affects several of the staff recommended conditions below, we request the Commission make a decision on this request prior to reviewing the rest of the application.

2. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.

Response: The project Arborist has provided additional information to address this condition that shows the applicant intends to retain these two trees as requested with a recommended modified root protection zone. This information addresses this condition.

3. The wetland report/delineation shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife.

Page 1 of 14

Response: It appears this condition from the City Engineer (Exhibit L) recommended the applicant submit a Wetlands Report outlining the delineated wetlands/high water level based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). It is unclear if the City Engineer reviewed the wetlands report submitted as part of the application (Exhibit H). This report indicates there are no wetlands on the property. In addition, a review of both the NWI and LWI do not show any wetlands on the property. Since there are no wetlands on the property, we ask that this condition be removed.

- 4. The applicant shall designate the FSH area as a separate tract on the plat and dedicate the tract to the City. In addition, staff recommends the Planning Commission require a pedestrian trail connection to the future extension of the Tickle Creek Trail as part of this development. Staff recommends that the trail connection be located in alignment with Middlebury Avenue. Alternatively, staff recommends that Lot 19 be included in the FSH tract and that the trail connection be located on the proposed location of Lot 19. The applicant shall include the trail connector tract as part of the FSH tract and shall dedicate the entire tract to the City. *Response:* This condition recommends the applicant dedicate all of the FSH Overlay on the property to the City and provide a trail connection to a future Tickle Creek Trail. There are several problems with this recommendation:
 - First, this requirement represents a taking of the applicant's property. As shown on submitted plans no development is proposed within the FSH Overlay and Section 17.60.20(C) does not prevent lots from being platted in the FSH Overlay only that lots cannot be platted entirely in the FSH Overlay. In addition, the Development Code does not contain language requiring this property to be dedicated to the City. In order to ensure this area receives adequate protection, a conservation easement can be platted over the FSH Overlay portion of these lots.
 - Another problem with this recommendation is that Lots 19, 20, and 21 will not comply with the 7,500 square foot minimum lot size standard if the entire FSH Overlay is put in a separate tract.
 - There is no legal justification for requiring the applicant to dedicate property or build a trail connecting to the Tickle Creek Trail. Section 17.84.90 (D) contains the following language: "Where a development is traversed by, or adjacent to, a future trail linkage identified within the Transportation System Plan, dedications of suitable width to accommodate the trail linkage shall be provided."

The property is neither traversed by or adjacent to an extension of the Tickle Creek Trail. A review of adopted Planning documents including the Parks Master Plan and the Transportation System Plan do not show a trail along Tickle Creek on the subject property. The nearest segment of the trail labeled as P22 in the TSP ends at Jacoby Road about 600 feet north of the site. (See Exhibit)

• This condition also recommends the Planning Commission consider taking Lot 19 for a trail connection. The value of Lot 19 is approximately \$130,000 and it is unclear if staff is also recommending the applicant be compensated for this property.

Page 2 of 14

With these factors in mind, the applicant requests this condition be removed.

5. The applicant shall submit a Phase I environmental assessment and a native vegetation report for the FSH area. The applicant shall remove all invasive plants and re-plant any areas of exposed soil with native plants. The applicant shall dedicate the FSH tract to the City and shall submit a maintenance bond for two full growing seasons (May 1 – October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Although parkland dedication is separate from open space dedication, staff recommends that the Planning Commission grant the applicant a parks fee-in-lieu credit for dedication of the open space FSH analysis area. The applicant shall obtain an appraisal for the FSH analysis area and, with City approval of the appraised value, shall receive a parks fee-in-lieu credit for dedicating the open space FSH analysis area.

Response: Condition 5 is dependent on the approval of Condition 4. This condition recommends in addition to the applicant dedicating property to the City that he also be required to complete a list of items before the City will accept the dedication.

- Phase I environmental assessment. (\$\$)
- Native vegetation report (\$\$)
- Remove invasive plants and plant native plants (\$\$)
- Maintain the property and provide a maintenance bond for two years (\$\$)
- Property Appraisal (\$\$)

If everything checks out on the property after completion of these items, staff recommends the Planning Commission grant a parks fee-in-lieu credit for this area based on an appraised value. There are several problems with this condition:

- We estimate the cost of providing the list of items staff is recommending to be about \$30,000. Because the property has limited development potential being in the FSH Overlay, the cost of providing these items is likely to exceed the parks fee-in-lieu credit.
- It is our understanding the reason for some of these items particularly the removal of invasive plants and planting natives and maintaining the area is because the Public Works Director has expressed concerns regarding the City maintaining this area.
- Also, it is unclear if the Planning Commission has the legal authority to approve fee reductions as recommended in this condition.

With these factors in mind, the applicant requests this Condition be removed.

6. All landscaping shall be continually maintained, including necessary watering, weeding, pruning, and replacing.

Response: This condition is vague and should either be clarified or removed. What landscaping is this referring to? It appears this condition is referring to Section 17.92.10(L) and should be removed. It is also very similar to Condition 8 below.

Page 3 of 14

- 7. The applicant shall update the plan set to remove street trees from the driveways in front of Lots 31 and 32; at least one of the street trees shall be relocated outside of the driveway area.
- 8. As required by Section 17.92.140, the developer and lot owners shall be required to maintain all vegetation planted in the development for two years from the date of completion, and shall replace any dead or dying plants during that period.

Response: This Condition depends on the Planning Commission's decision on the Code clarification of Section 17.92. If it is decides 17.92 applies, then this condition needs to be clarified as to what "vegetation" needs to be maintained. If it is found that Section 17.92 does not apply to subdivisions, this condition needs to be removed.

- 9. All street trees shall be a minimum of 1.5-inches in caliper measured 6 inches above the ground and shall be planted per the City of Sandy standard planting detail.
- 10. The applicant shall update the plan set to detail a minimum of two (2) different tree species per block face for staff review and approval.
- 11. The applicant shall update the proposed street tree list with at least two additional species from the City of Sandy Street Tree List that aren't maples or ashes.
- 12. The applicant shall maintain all unlandscaped and/or revegetated areas for a period of two years following the date of recording of the final plat associated with those improvements. **Response:** This condition comes from Chapter 17.92 and is basically the same as condition 8.
- 13. If the applicant chooses to postpone street tree and/or landscaping installation, the applicant shall post a performance bond equal to 120 percent of the cost of the street trees/landscaping, assuring installation within 6 months. The cost of street trees shall be based on the street tree plan and at least \$500 per tree fee, including a three-year maintenance and warranty period. The cost of landscaping shall be based on the average of three estimates from three landscaping contractors; the estimates shall include as separate items all materials, labor, and other costs of the required action, including a two-year maintenance and warranty period. **Response**: The underlined text in this condition is inconsistent.
- 14. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay. **Response:** This condition is the same as condition #2 above.

15. The applicant shall update the plan set to include a minimum of an additional 10 retention trees 8-inches or greater DBH and in good health. Staff recommends retaining all trees on Lot 19 and including the area as part of the FSH tract dedicated to the City of Sandy. Alternatively, staff recommends the applicant consider directional boring in the utility easements to retain Trees #109, 110, 120, 122, 336, 340, 366, 380, 383 and 392, and/or reducing the square footage of Lots 9 and 10 and creating an approximately 20 foot wide tree preservation buffer tract along Jacoby Road, which shall also be placed in a conservation easement.

Response: It appears from reading the staff report that the basis for this condition is that staff doesn't believe the application complies with tree retention standards. The subject property is 9.68 acres requiring a minimum of 29 trees, 11-inches or greater DBH to be retained per Chapter 17.102. This chapter allows retained trees to be located

Page 4 of 14

anywhere on the site at the landowner's discretion and trees proposed for retention shall be healthy and likely to grow to maturity. All of the trees proposed for retention are within the FSH, are located on the applicant's property, and have assessed by our Arborist to be in "viable" condition.

Pages 48 and 49 of the staff report reviews the tree plan by stating, "Of the 33 viable trees identified for retention, six (6) appear to be on or very close to the eastern property boundary (Trees # 527, 528, 530, 549, 550, and 3006), and an additional four (4) include notes such as "poor trunk taper," "windthrow potential," "wound on trunk," or "some trunk decay" (Trees # 520, 531, 534, and 560)." Staff concludes this discussion by saying that 10 more trees eight-inches or greater need to be retained.

Staff is correct that the above list of trees are close to the property line. Tree 3006 is on the property line, the remaining trees are entirely on the subject property. See the attached tree survey exhibit completed by the surveyor.

In addition, the project Arborist has provided additional notes and testimony that the trees staff is concerned about are "viable" should be counted towards tree retention. In addition, as included on the submitted tree inventory, 11 viable trees 8-inches to 10-inches in DBH are also within the tree protection area. With the addition of trees #510 and #516, the applicant will be protecting 35 "viable" trees 11-inches or greater DBH. One tree is on the property line, so to be conservative the applicant has 34 "viable" trees meeting Section 17-102.50 Tree Retention Requirements. As explained above the site must retain at least 29 trees. The proposed 34 trees exceed this requirement. In addition to the 34 trees there are 11 more 8-inch to 10-inch trees that will be protected. Given this information the proposal complies with the code as submitted and this condition is unnecessary and should be removed.

The condition suggests three options for retaining the 10 additional trees.

- Dedicate Lot 19 The approximate value of Lot 19 is \$130,000. Losing this value could jeopardize the project's viability, unless staff is also recommending the applicant be compensated for this. The applicant rejects this option without compensation.
- Retain trees on the back of several lots As shown on submitted plans grading does not make it feasible to protect the listed trees located along the back of several lots.
- Create a conservation easement on Lots 9 and 10 This option is also problematic in that as shown on the submitted Utility Plan this is the location the primary sanitary sewer and storm drainage lines will be placed. It is not possible to construct these utilities and save trees in this area.
- 16. The applicant shall preserve significant trees (25-inches or greater circumference, or 8-inches or greater DBH) to the greatest extent practicable.

Response: As noted in the applicant's interpretation request regarding Chapter 17.92, the wording in this condition has no legal basis. Chapter 17.92 uses the word "should" and not the word "shall" and these words carry different meanings. It appears staff is trying to change the code to protect more trees. The applicant requests this condition

Page 5 of 14

be removed because as discussed in condition 15 the proposal complies with tree retention standards and this condition will only lead to confusion in the future.

17. The applicant shall install protective fencing located at least 5 feet outside of the dripline around all trees to be retained on the subject property and for trees located on adjacent properties to the north and east. Tree protection fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. The applicant shall request an inspection of tree protection measures prior to any tree removal, grading, or other construction activity on the site.

Response: The requirements of this condition conflict with the tree protection requirements in Chapter 17.102 and should be removed. Since the Code is in conflict, and somewhat generic, we will protect all saved trees as required by a certified arborist. This condition could be rewritten as such.

18. To ensure protection of the required retention trees, the applicant shall record a tree protection covenant specifying protection of these trees and limiting removal without submittal of an Arborist's Report and City approval. This document shall include a sketch identifying the required retention trees. All trees marked for retention shall be retained and protected during construction regardless of desired or proposed building plans; plans for future houses on the proposed lots within the subdivision shall be modified to not encroach on retention trees and associated tree protection fencing.

Response: The applicant does not have a problem with this condition as long as protected trees area limited to those within the FSH Overlay. As noted by the Arborist, the applicant proposes restricting all access to the FSH Overlay/Tree Protection area during development.

19. Staff recommends the Planning Commission require a pedestrian trail connection to the future extension of the Tickle Creek Trail as part of this development. Staff recommends that the trail connection be located in alignment with Middlebury Avenue. Alternatively, staff recommends that Lot 19 be included in the FSH tract and that the trail connection be located on the proposed location of Lot 19. If the applicant wants to pursue locating the pedestrian trail connection off-site, the applicant shall provide justification as to why a connection on tax lot 800 makes more sense than a connection developed as part of this proposal. **Response**: As discussed in condition 4 above, City adopted planning documents do not identify an extension of the Tickle Creek Trail across the subject property and this condition is unwarranted. Our plans show the location of a possible trail connection if it was reasonable to require such a facility in the future. The reason the future trail is shown off-site was because the extension of Woodstock Street will be adjacent to the FSH area. This location will allow more room for a future trail to traverse the hillside with switchbacks. There is about 50-feet of elevation difference between the future street and the Creek. The proposed trail length shown will provide for a trail at about 6% grade. If you were to construct a trail in a linear path, in line with Middlebury Avenue, the "path" would be a continuous set of stairs. We feel a more natural trail may be desired as you enter the FSH area. The applicant's Engineer can provide

additional information regarding the proposed trail location as needed. This condition should be removed.

Individual House Conditions

- 20. All structures on lots along Jacoby Road shall be constructed to comply with the standards of Section 17.80.20.
- 21. All homes shall meet the development standards of Section 17.34.30.
- 22. Lot 28 and 29 shall share a driveway approach maximum 24 feet in width. Lot 26 and 27 shall share a driveway approach maximum 24 feet in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches. The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of Lot 29 to eliminate remnant of Tract A between the access drive and Lot 29. The driveway approach for Tract A shall be a maximum of 12 feet in width.
- 23. Staff recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30 and 32 per the following:
 - Builders of individual lots shall construct a decorative fence on the Jacoby Road facing side of Lots 9, 10, 30, and 32 to enhance the visual appeal of these lots from Jacoby Road. The fence shall include the following design details:
 - o Constructed of vertical black metal or faux metal fencing material.
 - o No greater than 3-inch gap between vertical pickets.
 - o 4-feet to 6-feet in height.
 - The rear elevations of Lot 30 and the side elevation of Lots 9, 10, and 32 shall incorporate all of the following design details where applicable:
 - o Decorative gables including two or more of the following:
 - A window with grids.
 - A trimmed vent. The trim must match the trim on the windows and the vent must be at least 4 square feet in area.
 - Cross or diagonal bracing, shingles, trim, corbels, exposed rafter ends, or brackets.
 - Decorative 'belly-band' with an alternative paint color to the siding color, between building floors.
 - o Mixture of siding materials, including shake or horizontal lap siding with an alternative paint color to the primary siding color.
 - o Recessed or covered rear entries (rear elevation only).
 - The covered area must be at least 48 square feet and a minimum of 8 feet wide.
 - The recessed entry must feature vertical support posts.
 - o Minimum four-inch wide trim or 12-inch wide shutters around all windows.
- 24. Builders shall submit proposed decorative fence and elevation designs for staff review and approval.
- 25. Staff recommends the Planning Commission approve a variance to allow the front door of the future homes constructed on Lots 9, 10, 30, and 32 to not face Jacoby Road. Staff further recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30, and 32.

Response: This is an approval recommendation and is also a repeat of Condition 23.

Page 7 of 14

26. Staff recommends the Planning Commission approve the applicant's request to allow onetiered lot configuration for Lot 30 between Jacoby Road and Camden Court. *Response:* This is not a condition but an approval recommendation.

Transportation/Streets

- 27. The applicant shall construct sidewalks along Tract A both on Jacoby Road and Camden Court, prior to final plat approval. The sidewalks on Camden Court, Woodstock Street, Newton Street, and Middlebury Avenue shall be five feet in width and separated by a planter strip or swale (<u>six feet in width</u>) in areas not transverse by driveways. The sidewalks on Jacoby Road shall be six feet in width and separated by a five foot planter strip. *Response: The local street standard requires a five foot planter not six feet as stated.*
- 28. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The three-quarter street improvements on Newton Street shall include a 28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot sidewalks on the north side of the roadway. *Response:* The applicant's proposal to narrow this street was intended to allow the home on the adjacent property to the East to be saved. We request the following statement be added to the end of the condition to allow some flexibility at the time of final design. "...The City Engineer may accept other reasonable design options."
- 29. The applicant shall revise the alignment of Newton Street to conform to the 50 feet of tangent alignment standard.
- 30. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road.

Response: The applicant does not have an easement from the neighbor. The applicant proposes to end the curbs and pavement back from the property line a sufficient distance to provide a 2:1 slope from the new improvements to the property line. Building retaining walls at the ends of public streets can be problematic and expensive for the next developer to remove when the street is extended. The applicant would rather pay a fee-in-leu for the improvements. This fee can be 125% of the actual cost to allow for future increase in material and construction costs. The applicant requests the following language be added to the end of this condition. "...As an option the developer may pay the City a fee-in-leu of 125% of the cost to extend the street to the property line."

31. The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. The developer shall improve the street section adjacent to the recommended trail connection tract on Woodstock Street prior to final plat approval.

Page 8 of 14

Response: This is the same as item #30. The applicant does not have an easement from the neighbor and to require this could be burdensome. The applicant proposes to end the curbs and pavement back from the property line a sufficient distance to provide a 2:1 slope from the new improvements to the property line. Building retaining walls at the ends of public streets can be problematic and expensive for the next developer to remove when the street is extended. The applicant would rather pay a fee-in-leu for the improvements. The fee can be 125% of the actual cost to allow for future increase in material and construction costs. The applicant requests the following be added to the end of this condition. "...As an option the developer may pay the City a fee-in-leu of 125% of the cost to extend the street to the property line."

32. The improvements on Jacoby Road shall include an additional 10 feet of right-of-way dedication (60' ROW).

Response: This condition could be read to require dedication of an "additional 10 feet" in addition to the 10 foot dedication proposed. I don't think that was the intent. Revise to eliminate "an additional".

- 33. The applicant shall provide a profile design for a minimum of 200 feet for all future street extensions beyond the project boundary to ensure future street grades can be met.
- 34. The minimum paved cul-de-sac radius shall be 48 feet as per the Oregon Fire Code and Metro Code Committee. A review by the Fire Department to confirm cul-de-sac size compliance shall be required.

Response: This Condition conflicts with the City of Sandy code requirements. "Section 17.96.60 (4) Dead End Streets. Cul-de-sacs over 400 ft. in length shall have a standard cul-de-sac bulb with a 38-ft. curbside radius." The Cul-de-sac shown on the applicant's plans meets this requirement. This condition should be removed to avoid confusion.

- 35. Staff recommends the Planning Commission approve a variance to allow the north side of Woodstock Street between Camden Court and the eastern property boundary to exceed 400 feet as proposed in the submitted plan set. Staff's recommendation to include the connector trail to the future Tickle Creek trail on the subject property will effectively reduce the block length for pedestrians and will improve pedestrian connectivity in the future. *Response: This sounds like a Finding not a condition. Please remove this condition for clarity.*
- 36. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. If the plans change in a way that affects the number of street trees (e.g., driveway locations), the applicant shall submit an updated street tree plan for staff review and approval.
- 37. Street trees shall be installed approximately 30 feet on center in conjunction with issuance of building permits. Street trees are required to be a minimum caliper of 1.5- inches measured 6 inches from grade. Trees shall be planted and staked per the City of Sandy standard planting detail; trees shall be tied to the stakes with loosely tied twine. The planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed after one growing season (or a maximum of 1 year).

Page 9 of 14

38. The applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west.

Response: This condition cannot be met. The existing street grade and sidewalk on Jacoby Road at this location is approximately 7.5%. This exceeds the 2% landing that is required to meet ADA. Please remove this condition.

39. The applicant shall install one bench (Fairweather model PL-3, powder-coated RAL6028) with a concrete pad to City specifications. The applicant shall contact the Transit Director for specific location, amenity information, and pad engineering specifications

40. The applicant shall plat a vehicle non-access reserve (VNAR) strip along the south side of the right-of-way of Newton Street and at the east end of Woodstock Street. Response: We recommend that the City allow access for the neighbors to enter the new paved street as needed. The condition could be re-written to allow some flexibility at the time of final design. "...the VNAR can be adjusted as needed to allow access to the new road with City Engineer approval."

- 41. The applicant shall work with the Fire Marshall to determine if a temporary turn-around is needed on Woodstock Street and/or Newton Street, or if houses on particular lots need to be sprinkled.
- 42. The City of Sandy reserves the right to name streets.
- 43. The access to the detention pond shall be paved or all-weather surface.

Utilities

- 44. All public utility installations shall conform to the City's facilities master plans. Staff recommends the applicant revise the utility plan to include broadband fiber locations as detailed by the SandyNet Manager.
- **45.** All public sanitary sewer and waterline mains shall be a minimum of 8 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. No building permits will be issued until all public utilities including sanitary sewer are available to serve the development. The applicant shall pay plan review, inspection, and permit fees as determined by the Public Works Director.
- 46. Any existing domestic or irrigation wells on site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City Planning Division.
- 47. Any on-site sewage disposal system shall be abandoned in conformance with Clackamas County Water Environmental Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City Planning Division.
- 48. All new public sanitary sewer lines shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. *Response:* This is the same as condition 45. Please remove this condition for clarity.

Page 10 of 14

- 49. The applicant shall provide a sanitary sewer lateral serving Tax Lot 6100 (located directly north of the subject property) with the sanitary sewer main line extension. *Response:* We ask that the City reimburse the developer for the cost of this lateral. Please include this to the end of the condition "...City will reimburse the developer for the cost of the new sewer lateral."
- 50. The applicant shall demonstrate that the proposed subdivision does not exceed pre-development site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way.
- 51. The detention pond shall meet the requirements of the 2016 City of Portland Stormwater Management Manual (SWMM) for landscaping, Section 2.4.1, and escape route, Section 2.30.

52. All new public storm drains shall be a minimum of 12-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. *Response:* This is the same as condition 45. Please remove this condition for clarity.

- 53. The applicant shall submit a detailed final stormwater report stamped by a licensed professional to the City for review and approval with the final construction plans.
- 54. All utilities shall be placed underground.
- 55. All franchise utilities shall be installed underground and in conformance with City standards.
- 56. The applicant shall call the PGE Service Coordinators at 503-323-6700 when the developer is ready to start the project.
- 57. The applicant shall demonstrate that adequate fire and domestic flow will be available with a single point of connection to the water distribution system or connect to the existing 8-inch water line at the intersection of Newton Street and Amherst Street.
- 58. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with construction plans.
- 59. All new waterline mains shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. *Response:* This is the same as condition 45. Please remove this condition for clarity.
- 60. The applicant shall be responsible for the installation of all improvements detailed in Section 17.100.310, including fiber facilities. SandyNet requires the developer to work with the City to ensure that broadband infrastructure meets the design standards and adopted procedures as described in Section 17.84.70.
- 61. The locations of street light fixtures shall be reviewed in detail with construction plans.

Parking

- 62. The applicant shall comply with the parking standards in Section 17.98.60.
- 63. All parking, driveway and maneuvering areas shall be constructed of asphalt, concrete, or other approved material.
- 64. The applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval. The applicant shall revise the Parking Analysis if required fire hydrants affect on-street parking spaces.

Geotech/Grading/Erosion

Page 11 of 14

Response: A number of the conditions in this section are redundant. For example, conditions 65, 72, and 73 are similar regarding erosion control and conditions 66, 68, 69, 71 are similar regarding geotechnical reporting. We request these conditions be combined for clarity.

- 65. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a re-vegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code and as detailed below. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.
- 66. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.
- 67. The applicant shall update the cut and fill evaluation to include the location of the tree protection fencing as depicted on the existing conditions and tree retention inventory (Exhibit C, Sheet 2). The applicant shall not cut or fill within the proposed tree protection area. Response: For flexibility at time of construction, please include this phrase to the end of the condition. "...without approval of a certified arborist."
- 68. The applicant shall follow the recommendations outlined in the Geotechnical Report.
- 69. The applicant shall submit a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay. This includes Lots 19, 20, 21, 27, 28, and 29. Additional reports may be required depending on the recommendations of the geological assessment. Once the subdivision is platted and building permits are filed for individual home construction the property owner/contractor shall submit Type I Hillside applications with studies as detailed in Table 1 in Section 17.56.40.
- 70. All on-site earthwork activities including any retaining wall construction should follow the requirements of the current edition of the Oregon Structural Specialty Code (OSSC). If the proposal includes a retaining wall, the applicant shall submit additional details on the proposed retaining wall for staff review and approval.
- 71. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the Page 12 of 14

above noted report and OSSC. Site grading should not in any way impede, impound or inundate the adjoining properties. All the work within the public right-of-way and within the paved area should comply with American Public Works Association (APWA) and City requirements as amended. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite.

- 72. All erosion control and grading shall comply with Section 15.44 of the Municipal Code. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.
- 73. The applicant's Erosion Control Plan shall be designed in accordance with the standards of Section 15.44.50.

Miscellaneous

- 74. If entry signs are desired, staff recommends the applicant submit a detailed plan showing the location of such signage.
- 75. Public improvement plans are subject to a separate review and approval process. Preliminary plat approval does not connote utility or public improvement plan approval which will be reviewed and approved separately upon submittal of public improvement construction plans.
- 76. The applicant shall obtain a demolition permit from the City of Sandy Building Department prior to demolition of the existing structure(s) on-site.
- 77. Mail delivery facilities shall be provided by the applicant in conformance with Section 17.84.100 and the standards of the USPS. The applicant shall submit a mail delivery plan, featuring grouped lockable mail facilities, to the City and USPS for review and approval prior to installation of mailboxes.
- 78. The applicant shall pay a fee in lieu for the required parkland dedication per the adopted Fee Resolution. Based on the current Fee Resolution, the applicant would be required to pay a fee in lieu of dedication for a total of \$98,810 (0.41 acres of land to be dedicated x \$241,000). Currently, the Fee Resolution requires payment of \$108,650 if a portion of the fee is deferred, a minimum of 50 percent (\$54,325) paid prior to final plat approval and the remaining 50 percent (\$54,325) divided between the 32 lots (\$1,697.66/lot). Response: This condition is dependent on the decision of other conditions above.
- 79. The Plat shall detail the following easements:
 - An eight-foot wide public utility easement (PUE) along the frontage of all proposed lots;
 - Crossover easements along the common parcel line recorded between pairs of lots that share driveways (Lots 26 and 27, and Lots 28 and 29);
 - 20 foot easement along the west boundary of Lots 9 and 10 for public sanitary sewer and public storm line. Per the Public Works Director (Exhibit O), the City reserves the right to widen this easement depending on the final design depth of the sanitary sewer and storm lines per Section 17.84.90(A.2);
 - A 15 foot easement between Lot 30 and Tract A for public sanitary sewer;
 - A 10 foot private storm drain easement at the rear of Lots 1-18, 22-25, 28, 29, 31, and 32 and a 10 foot private storm drain easement across Lots 19-21 just south of the FSH boundary;

Page 13 of 14

- A vehicle non-access reserve (VNAR) strip in the following locations:
- Jacoby Road frontage of Lots 9, 10, 30 and 32 and Tract A; and,
- South side of the right-of-way of Newton Street; and,
- **Response:** This condition is the same as #40. We recommend that the City allow access for the neighbors to enter the new paved street as needed. The condition could be re-written to allow some flexibility at time of final design. "...the VNAR can be adjusted as needed to allow access to the new road with City Engineer approval."
- East end of the right-of-way of Woodstock Street.

Page 14 of 14











Exhibit X

28. The applicant shall eliminate the proposed curve at the east end of Newton and continue the threequarter street improvement to the east boundary of the site. The three-quarter street improvements on Newton Street shall include a 28-foot wide paved surface, curbs on both sides, 5foot planter strips and 5-foot sidewalks on the north side of the roadway.

Response: The applicant's proposal to narrow this street was intended to allow the home on the adjacent property to the East to be saved. We request the following statement be added to the end of the condition to allow some flexibility at the time of final design. "...**The City Engineer may accept other reasonable design options**."

All the City has is the application before it. The City shouldn't speculate on what may or may not happen on the adjacent property. The adjacent property owner has been noticed and if they have a better idea of what the future holds for their property or the existing dwelling than they should submit testimony or comments reflecting this. A 40 ft. right-of-way dedication and a ³/₄-street improvement should be required to serve more than two single-family dwellings.

If the applicant wants to accommodate future development of the parcel to the east then the applicant needs to figure out how to do this and still provide the minimum ³/₄-street section on the development site or acquire land to accommodate it on adjacent properties.

30. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road.

Response: The applicant does not have an easement from the neighbor. You should not be writing conditions contingent on neighbor participation for development which does not occur on their parcel. The applicant proposes to end the curbs and pavement back from the property line a sufficient distance to provide a 2:1 slope from the new improvements to the property line. Building retaining walls at the ends of public streets can be problematic and expensive for the next developer to remove when the street is extended. The applicant would rather pay a fee-in-leu for the improvements. This fee can be 125% of the actual cost to allow for future increase in material and construction costs. The applicant requests the following language be added to the end of this condition. "...As an option the developer may pay the City a fee-in-leu of 125% of the cost to extend the street to the property line."

There is no need for an easement, wall, etc. and the curbs and improvements do not end "back from the property line" The applicant's submittal shows the Jacoby frontage improvements ending at the south line of the site without walls, easements, etc. I recommend that this condition should be left as is.

31. The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. The developer shall improve the street section adjacent to the recommended trail connection tract on Woodstock Street prior to final plat approval.

Response: This is the same as item #30. The applicant does not have an easement from the neighbor and to require this could be burdensome. You should not be writing conditions contingent on neighbor participation for development which does not occur on their parcel. The applicant proposes to end the curbs and pavement back from the property line a

sufficient distance to provide a 2:1 slope from the new improvements to the property line. Building retaining walls at the ends of public streets can be problematic and expensive for the next developer to remove when the street is extended. The applicant would rather pay a fee-in-leu for the improvements. The fee can be 125% of the actual cost to allow for future increase in material and construction costs. The applicant requests the following be added to the end of this condition. "...As an option the developer may pay the City a fee-in-leu of 125% of the cost to extend the street to the property line."

The condition gives the applicant *the option* of a retaining wall or an easement to allow extension of the street. The condition doesn't *require* neighbor participation nor is it contingent on same. Based on the grading plan the wall would be a fill wall, less about 2 feet in height. This is neither problematic or expensive. The bottom of the wall would be above any required utilities (which are shown extending all the way to the property line. The typical method when extending a street with a fill wall is to remove the top course or two and leave the rest of the wall in place and bury it when the street is extended, no removal required. Even if removal would be required it would be no more expensive than removing the soil (whether fill or cut) at the end of the street when extended. I recommend that the condition be left as is.

The minimum paved cul-de-sac radius shall be 48 feet as per the Oregon Fire Code and Metro Code Committee. A review by the Fire Department to confirm cul-de-sac size compliance shall be required. *Response:* This Condition conflicts with the City of Sandy code requirements. "Section 17.96.60 (4) Dead End Streets. Cul-de-sacs over 400 ft. in length shall have a standard cul-de-sac bulb with a 38-ft. curbside radius..." The Cul-de-sac shown on the applicant's plans meets this requirement. This condition should be removed to avoid confusion. Our proposed cul-de-sac has a 39-ft curbside radius, which is not only greater than that required by the city code, but is also identical to the curb radius of the recently approved Marshal Ridge Subdivision.

Section 17.96 relates to Manufactured Home Parks in general and section 17.96.60(4) relates to culde-sacs in private drives within manufactured home parks in particular. The subject site is zoned SFR and thus the standard in 17.96.60(4) neither conflicts with or applies to the condition. The fact that a cul-de-sac of this size was approved on another development does not allow staff to ignore the Fire Code standard on this site.

The cul-de-sac bulb dimensions are not solely for emergency vehicle access. The larger radius allows enough room for on-street parking, garbage pickup, street sweeping and snow plowing. Even if lots 26-30 were constructed with residential fire protection systems I would not recommend reducing the cul-de-sac bulb radius. I recommend that the condition be left as is.

38. The applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west.

Response: This condition cannot be met. The existing street grade and sidewalk on Jacoby Road at this location is approximately 7.5%. This exceeds the 2% landing that is required to meet ADA. Please remove this condition.

This location isn't at an intersection – the condition is requiring a mid-block ramp on the existing street opposite the proposed intersection. Regardless, this standard should be easily achievable if the street grade is 7.5%. The ODOT standard detail (attached) allows a 15' long transition between the landing and the point where the existing street grade is matched. I recommend that the condition be left as is.

49. The applicant shall provide a sanitary sewer lateral serving Tax Lot 6100 (located directly north of the subject property) with the sanitary sewer main line extension.

Response: We ask that the City reimburse the developer for the cost of this lateral. Please include this to the end of the condition "...City will reimburse the developer for the cost of the new sewer lateral."

Section 17.84.60(C) states: "Off-site public facility extensions necessary to fully serve a development site and adjacent properties shall be constructed concurrent with development." The request to install a short-side lateral for a single dwelling is not onerous. I believe the adjacent property owner included this request in a public comment.

The City will not reimburse the developer for the cost of the lateral. The applicant may apply for an Advanced Financed Public Improvement reimbursement from the benefitting property owner(s) per section 12.14 of the Municipal Code.



North Pacific Design, Inc. Landscape Architecture

917 Oak Street, Suite 300 Portland, Oregon 97205 Tele: 503-294-9905

Exhibit Y

City of Sandy

Department of Community Development 38348 Pioneer Boulevard Sandy, Oregon

Parks Master Plan Park Facility Plan

Sheet	Scale: _1" = 800'
Date: May 15, 1997	North

Exhibit Z

To: City of Sandy City Council, Planning Commission and Planning Staff.

From: Brad Picking

I have reviewed the Planning Commission Agenda and Staff report for the proposed Jacoby Heights Subdivision. I do not have any concerns with the proposed subdivision and I am in favor of good land use planning and projects that can provide additional quality housing to the City of Sandy, such as this project.

My concern is with the Staff's interpretation of Chapter 17.92. The purpose of this letter is to convey my knowledge regarding the proposed code interpretation. As you may or may not know, I was part of the original committee that helped develop code section 17.102 Urban Forestry. Part of the reason 17.102 was created, was to provide some tree removal restrictions that would be applied to subdivision development. Prior to that time there was none. It was understood at that time, the Code 17.92 was not applicable to residential subdivisions. 17.92 was intended to provide guidance to landscaping, buffering and screening. This section does apply to all zones as indicated, but not subdivision developments.

Here is Staff's recommendation: "Staff recommends that the Planning Commission interpret the code such that Chapter 17.92, Landscaping and Screening, continue to apply to all zones, including subdivisions."

Code Section 17.92 clearly states that this section applies to all Zones. "Subdivisions" are not zones, they are a type of development. Staff's recommendation is worded in such a way to make you recommend that it continue to apply to all zones and subdivisions. The problem with their recommendation is they are adding words to the code that are not there, "Subdivisions". You cannot add words to codes without proper procedures.

I understand that there is a new committee being formed to start the process of re-wording Code Sections 17.92 and 17.102. I believe this would be the proper and legal process of changing the words to the Code.



SUBJECT: File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision

AGENDA DATE: April 8, 2019

DEPARTMENT: Planning Division

Application Submitted: June 25, 2018 Additional Submittal Items: December 5, 2018 Application Complete: December 27, 2018 Planning Commission Hearing: February 25, 2019 PC Continued Hearing: April 8, 2019 120-Day Deadline: June 7, 2019

STAFF CONTACT: Emily Meharg, Associate Planner

EXHIBITS:

- Applicant's Submittals
- A. Land Use Application Form
- B. Narrative
- C. Project Plan Set
 - Sheet 1: Cover Sheet and Preliminary Plat Map
 - Sheet 2: Existing Conditions and Tree Retention Inventory
 - Sheet 3: Existing Trees with Building Setbacks Map
 - Sheet 4: FSH and Slopes Map
 - Sheet 5: Preliminary Street and Utility Plan
 - Sheet 6: Preliminary Street Tree Plan
 - Sheet 7: Preliminary Parking Plan
 - Sheet 8: Future Street and Pedestrian Trail Plan
 - Sheet 9: Grading and Erosion Control Plan
 - Sheet 10: Cut and Fill Evaluation
- D. Code Interpretation Request
- E. Storm Drainage Design and Calculations for the Jacoby Heights Subdivision
- F. Traffic Impact Analysis
- G. Geotechnical Report
- H. Wetland Determination
- I. Easement between Cory Knight and Ian Bird and Kelly Bird
- J. Letter from Public Works Director Re: Pump Station Capacity
- K. Arborist Report

Agency Comments

- L. City Engineer (January 15, 2019)
- M. Transportation Engineer (January 18, 2019)
- N. PGE (January 23, 2019)
- O. Public Works Director (January 26, 2019)
- P. Transit Director (January 28, 2019)

Public Comments

- Q. Annette Giltner, 35000 SE Colorado Road (February 6, 2019)
- R. Shawn Lorenzen, 39343 Stratford Street (February 11, 2019)
- S. Tim and Cherri Anders, 19098 Jacoby Road (February 12, 2019)

Additional Documents Submitted by Staff

- T. Chapter 17.102 of the City of Sandy Development Code
- U. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision Staff Report from February 25, 2019 Planning Commission meeting

Documents Submitted at the February 25, 2019 Planning Commission Hearing

- V. Arborist Notes for Jacoby Heights Subdivision
- W. Review of Staff Recommended Conditions File No. 18-025 SUB/VAR/FSH/TREE/INT Jacoby Heights Subdivision memo by Tracy Brown

Additional Agency Comments after the February 25, 2019 PC Hearing

X. Public Works Director and City Engineer

Additional Documents Submitted by Staff after the February 25, 2019 PC Hearing

Y. Park Facilities Plan

Additional Comments Submitted the day of the February 25, 2019 PC Hearing Z. Brad Picking

I. BACKGROUND

A. PROCEEDING

Type III Subdivision, Type III Code Interpretation, Type III FSH review, Type III Variance for not orienting towards a transit street, Type III Variance for not providing two tiered lot configuration, Type III Variance to Block Length, and Type II Tree Removal Permit. Note: the proposal includes a newly mapped FSH overlay area. The zone change associated with the new FSH overlay will be processed separately by the City.

B. FACTUAL INFORMATION

- 1. APPLICANT/OWNER: Cory Knight
- 2. PROJECT NAME: Jacoby Heights Subdivision
- LEGAL DESCRIPTION: T2S R4E Section 24A Tax Lot 2300
- 4. SITUS ADDRESS: 19124 Jacoby Road
- 5. PROPERTY LOCATION: East of Jacoby Road, west of Langensand Road, and south of Cascadia Village Drive
- 6. PROPERTY SIZE: 9.68 acres
- 7. PROPOSAL: All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

- 8. COMPREHENSIVE PLAN DESIGNATION: Low Density Residential
- 9. ZONING DISTRICT DESIGNATION: SFR, Single Family Residential
- 10. SERVICE CONSIDERATIONS: The applicant proposes to extend the existing sanitary sewer in Jacoby Road to service the entire subdivision. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. The stormwater for the subdivision would be collected and conveyed to a proposed detention pond located in Tract A at the northwest corner of the site.

11. RESPONSE FROM GOVERNMENTAL AGENCIES, UTILITY PROVIDERS, AND CITY DEPARTMENTS:

- a. City Manager No comments received
- b. City Engineer- Exhibit L
- c. Transportation Engineer Exhibit M
- d. Public Works Director Exhibit O
- e. Transit Director Exhibit P
- f. PGE Exhibit N
- g. ODOT No comments received
- h. SandyNet Manager No comments received
- i. Police Department No comments received
- j. Fire District No. 72 No comments received
- k. US Postal Service No comments received

C. PUBLIC COMMENTS

Three written comments were received as follows:

- Annette Giltner at 35000 SE Colorado Road (Exhibit Q) does not want to see more development in Sandy.
- Shawn Lorenzen at 39343 Stratford Street (Exhibit R) does not want to see more
 residential units built in Sandy without first strengthening the City's infrastructure to
 support existing residents.
- Tim and Cherri Anders at 19098 Jacoby Road (Exhibit S) expressed concerns about the effect of the proposed subdivision on their property, which is directly adjacent to the north.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

D. APPLICABLE CRITERIA:

Sandy Municipal Code: 17.12 Procedures for Decision Making; 17.14 Request for Interpretation; 17.18 Processing Applications; 17.22 Notices; 17.30 Zoning Districts; 17.34 Single Family Residential (SFR); 17.56 Hillside Development; 17.60 Flood and Slope Hazard Overlay District; 17.66 Adjustments and Variances; 17.80 Additional Setbacks on Collector and Arterial Streets; 17.82 Special Setbacks on Transit Streets; 17.84 Improvements Required with Development; 17.86 Parkland and Open Space; 17.92 Landscaping and Screening; 17.98 Parking, Loading, and Access; 17.100 Land Division; 17.102 Urban Forestry; 15.30 Dark Sky Ordinance; and 15.44 Erosion Control.

E. BACKGROUND INFORMATION/HISTORY

There is one existing house at 19124 Jacoby Road. The applicant shall obtain a demolition permit from the City of Sandy Building Department prior to demolition of the existing structure(s) on-site. Per the City Engineer (Exhibit L), the preliminary plans don't identify any existing domestic or irrigation wells on site. Any existing domestic or irrigation wells on site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City Planning Division. Per the City Engineer (Exhibit L), the preliminary plans don't identify any septic tanks on site. Any on-site sewage disposal system shall be abandoned in conformance with Clackamas County Water Environmental Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City Planning Division.

F. PROCEDURAL CONSIDERATIONS

This request is being processed as a Type III Subdivision, Type III FSH Review, three Type III Variances, Type III Code Interpretation, and Type II Tree Removal Permit. Notification of the proposal was mailed to property owners within 300 feet of the subject property and to affected agencies on January 29, 2019. A legal notice was published in the Sandy Post on February 6, 2019.

The Planning Commission reviewed the subject subdivision request at a public hearing on February 25, 2019. The meeting focused primarily on the applicant's request for a code interpretation. Due to inclement weather, staff suggested that the meeting be continued. The continued meeting date was set for April 8, 2019. This staff report is an update to the staff report that was presented at the February 25, 2019 Planning Commission meeting.

At the February 25, 2019 Planning Commission hearing, the applicant contested some of staff's proposed conditions. Due to inclement weather and the general consensus to continue the hearing, staff did not have an opportunity to respond to various issues raised by the applicant. Below is a partial list of some of the objections brought up by the applicant's representatives, followed by staff's response.

- Applicant's Representative: Subdivisions don't include landscaping. Staff's Response: Many subdivisions include parking tracts, stormwater detention areas, and pedestrian tracts that do have landscaping requirements. All subdivisions are required to install street trees.
- Applicant's Representative: The pre-application notes were not comprehensive. Staff's Response: Pre-application notes are never comprehensive. Every proposal is unique and

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx
often issues are only discovered after a thorough review of the proposal. All pre-app notes include the following language: "*This analysis includes a review of those code sections that may conflict with the proposed design as submitted. This review is not intended to be a comprehensive analysis of all applicable code sections.*"

- Applicant's Arborist: Excavating within critical root zones could lead to tree failure. Staff's Response: We completely agree.
- Applicant's Arborist: "I agree that the 10 feet from the trunk is not [couldn't hear] enough tree protection for a lot of trees. It needs to be at least the drippage [sic]. The 17.92 says the drippage [sic] plus 5 feet, which I think would be ample." Locating tree protective fencing 5 feet beyond the dripline is good. Staff's Response: We completely agree.
- Applicant's Arborist: Recommends using a 6 foot metal fence. Staff's Response: We completely agree and that is what Section 17.102.50(B.2) of our code says.
- Applicant's Representative: Lot 19 is worth \$130,000. Staff's Response: A majority of Lot 19 contains slopes in excess of 25 percent. Staff is curious who appraised this lot to be \$130,000 as staff does not have evidence of that appraisal amount.

Prior to submitting construction plans or building permits, including grading and erosion control permits, the applicant shall update the plan set and associated documents based on the conditions of approval determined by the Planning Commission and shall submit a full set of the updated plans to Planning Division staff for review and approval.

II. ANALYSIS OF CODE COMPLIANCE

CHAPTER 17.14 – REQUEST FOR INTERPRETATION

Request: The applicant is requesting an interpretation regarding whether Chapter 17.92, Landscaping and Screening, applies to subdivisions.

RESPONSE: Staff reviewed the Development Code and relevant ordinances and consulted with the City Attorney and found nothing indicating that Chapter 17.92 does not apply to subdivisions. In fact, the full title of Chapter 17.92 is Landscaping and Screening General Standards – All Zones. Staff interprets this as meaning that Chapter 17.92 applies to all zones, including Single Family Residential. Chapter 17.92 includes standards related to required tree plantings (including street trees and parking court trees), minimum sizes for trees and plants, and revegetation standards, all of which are highly applicable to subdivision development. The specific general provision that the applicant is questioning (Section 17.92.10(C)) states that significant trees should be integrated into the design of a development and that trees to be retained shall be protected from damage during construction. Staff finds that a subdivision qualifies as a development and as construction. Furthermore, there is nothing contained in Chapter 17.92 or Chapter 17.102 that restricts using Chapter 17.92 during subdivision review. There is also nothing in the ordinance (Ordinance No. 2002-10) that adopted Chapter 17.102 that speaks on this matter. In addition, there are several Comprehensive Plan Goal 5 policies that support retaining significant trees, most notably, the following:

 Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.
- Policy 31: The City of Sandy shall undertake a study to designate "significant/heritage" trees or areas of second growth that exist within the City's urban growth boundary. Designated trees shall be protected and preserved where possible.

At the February 25, 2019 Planning Commission hearing, the City Attorney stated: "There's really nothing in 17.92 or 17.102 that indicates 17.92 would not be applicable during subdivision review." The City Attorney further stated: "When you have [a tension between two code sections] normally what you would do, unless there's an interpretation to the contrary, would be to try to harmonize both 17.102 and 17.92 when it comes to tree preservation." Members of the Planning Commission reiterated that the code clearly states that Chapter 17.92 applies to all zones. Commissioner Carlton got clarification from the City Attorney that all zones means all zones, including SFR, R-1, R-2, etc. Commissioner Crosby stated: "It would be hard to interpret something against what it says in the title 'All Zones." After further discussion, Commissioner Carlton made a motion to interpret Chapter 12.92 (sic) to apply to all zones including subdivisions, but that Section 17.92.10(C) not apply to residential subdivisions. After consultation with the City Attorney on the wording of the motion, Commissioner Carlton restated the motion as "the tree retention standards in 17.102 are the standards to apply to subdivisions versus the tree retention standards in 17.92.10(C)" and Commissioner Logan seconded the motion. The Planning Commissioners voted 4-0 to carry the motion. Staff was confused by the motion and unsure what the basis was for rendering the interpretation that 17.92.10(C) does not apply to residential subdivisions, but the remainder of Chapter 17.92 does apply to subdivisions since the motion contradicted the City Attorney's statement and did not seem to include an analysis or reason for the code interpretation. Staff reached out to the Planning Commission chair for clarification and was provided the following rationale:

"Here's what I recall was our thinking. There is a conflict between the tree retention standards in 17.92.10 (C) and 17.102 in regards to their applicability to subdivisions, and it has produced confusion for developers. (Although 17.92 clearly states in its title, "All Zones," oddly enough the chart in it does not list all zones. That caused both John and I to cock our heads.) To resolve the conflict and avoid confusion (to a point...there will always be confusion!), we simply pointed to 102 and said, 'This is the code to be applied to subdivisions, in place of 17.92.10 (C), in regards to tree retention.""

Staff is still confused as the tree retention standards in 17.92.10(C) and 17.102 don't conflict. Furthermore, the landscaping chart in Section 17.92.20 that the Planning Commission referenced simply lists zones that have minimum landscaping standards; the SFR, R-1, and R-2 zones do not have minimum standards so do not appear in the chart. Staff is still not clear as to what section of the code the Planning Commission is citing as the basis for their interpretation.

Recommendation: Since the Planning Commission decides to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

decision.

CHAPTER 17.30 - ZONING DISTRICT

The Single Family Residential (SFR) zoning on the site allows a minimum of 3 units per acre and a maximum of 5.8 units per acre. The number of dwelling units permitted on a parcel of land is calculated after determining the amount of restricted development area in accordance with Section 17.60, the Flood and Slope Hazard Overlay District. Density transfer is permitted from restricted development areas consistent with the provisions of Section 17.60.90, although no density transfer is proposed.

RESPONSE: The total gross acreage for the entire property is 9.68 acres. The proposal contains 2.08 acres of area dedicated for public right-of-way and 0.47 acres dedicated for public tracts (Tract A). After removal of the right-of-way and public tracts the net site area for the subject property is reduced to 7.13 acres of net site area (NSA). The subject property also contains 1.20 acres of restricted development area; therefore, the unrestricted site area (USA) is 5.93 acres. Based on required density, the SFR land requires a minimum of 18 dwelling units (5.93 USA x 3). The maximum allowed dwelling units is 41 (calculated as the lesser of (a) 7.13 NSA x 5.8 = 41 and (b) 5.93 USA x 5.8 x 1.5 = 52). The proposed 32 dwelling units are within the allowable density range and therefore meet the density requirement.

CHAPTER 17.34 - SINGLE FAMILY RESIDENTIAL (SFR)

The applicant proposes 32 single family detached dwellings in conformance with minimum and maximum density requirements, as detailed above in Chapter 17.30.

17.34.10 PERMITTED USES

RESPONSE: Section 17.34.10 lists single family detached dwellings as a permitted use. The proposed subdivision includes 32 lots for single family detached dwellings. All homes shall provide building design features in compliance with the standards in Section 17.90.150 and Chapter 17.54.

17.34.30 DEVELOPMENT STANDARDS

Туре	Standard
A. Minimum Lot Area - Single detached dwelling	7,500 square ft.
- Other permitted uses	No minimum
A. Minimum Average Lot Width	
- Single detached dwelling	60 ft.
C. Minimum Lot Frontage	20 ft. except as allowed by Section
	17.100.160
D. Minimum Average Lot Depth	No minimum
E. Setbacks (Main Building)	
Front yard	10 ft. minimum
Rear yard	20 ft. minimum
Side yard (interior)	7.5 ft. minimum
Corner Lot	

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

	10 ft. minimum on side abutting the street ¹
F. Setbacks (Garage/Carport)	22 ft. minimum for front vehicle access
	15 ft. minimum if entrance is perpendicular
	to street (subject to Section 17.90.220)
	5 ft. minimum for alley or rear access
G. Projections into Required Setbacks	See Chapter 17.74
H. Accessory Structures in Required Setbacks	See Chapter 17.74
I. Structure Height	35 ft. maximum
J. Building Site Coverage	No minimum
K. Off-Street Parking	See Chapter 17.98

RESPONSE: The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes shall meet the development standards of Section 17.34.30.

17.34.40 MINIMUM REQUIREMENTS

- A. Must connect to municipal water.
- B. Must connect to municipal sewer if service is currently within 200 hundred feet of the site. Sites more than 200 feet from municipal sewer, may be approved to connect to an alternative disposal system provided all of the following are satisfied:
 - 1. A county septic permit is to be secured and a copy is provided to the city.
 - 2. The property owner executes a waiver of remonstrance to a local improvement district and/or signs a deed restriction agreeing to complete improvements.
 - 3. The minimum size of the property is one acre or is a pre-existing buildable lot, as determined by the city.
 - 4. Site consists of a buildable parcel(s) created through dividing property in the city, which is less than five acres in size.
- C. The location of any real improvements to the property must provide for a future street network to be developed.
- D. Must have frontage or approved access to public streets.

RESPONSE: All lots will be required to connect to City services. The applicant is also required to extend utilities to the furthest extent of the subject property.

CHAPTER 17.56 – HILLSIDE DEVELOPMENT

17.56.10 APPLICABILITY

These regulations shall apply to any parcel with slopes greater than twenty-five percent (25%) as-shown on the Hillside Development Overlay District Map or with slope hazards mapped by the Department of Geology and Mineral Industries (DOGAMI). This chapter shall apply only to activities and uses that require a building, grading, tree removal and/or land use permit.

A. <u>General</u>. No person shall develop property in areas designated by SDC 17.56.10, without first demonstrating compliance with this chapter.

¹ Must comply with clear vision requirements of Chapter 17.74.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- 1. As a condition of permit issuance or land use approval, the applicant shall agree to implement the recommendations of approved studies and to allow all inspections to be conducted.
- 2. Where a bond, letter of credit or other guarantee is required, the permit shall not be issued until the bond or guarantee has been obtained and approved.

B. Exemptions:

- 1. An activity or use that avoids slopes of 25% or greater, DOGAMI slope hazard areas, natural drainageways and potentially hazardous analysis areas as defined in Section 17.56.30.A.
- 2. The following activities, regardless of location:
 - a. An excavation that is less than three feet in depth, or which involves less than fifty cubic yards of volume;
 - b. A fill that does not exceed three feet in depth or 50 cubic yards of volume;
 - c. New construction or expansion of a structure resulting in a net increase in ground floor area of less than 1,000 square feet that does not involve grading;
 - d. Emergency actions required to prevent an imminent threat to public health or safety, or prevent imminent danger to public or private property, as determined by the Director; or
 - e. Any land use or activity that does not require a building, grading permit, or land use approval.

RESPONSE: The applicant submitted an FSH and Slopes Map (Exhibit C, Sheet 4) that shows areas of land that exceed 25 percent slope. A majority of the 25 percent or greater slope areas are located in the northeast corner of the property, adjacent to the Tickle Creek basin. The applicant submitted a Geotechnical Report (Exhibit G) by Rapid Soil Solutions LLC that includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations, which was reviewed by the City Engineer. Per the City Engineer (Exhibit L), the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC. The submitted Cut and Fill Evaluation (Exhibit C, Sheet 10) shows proposed cut and fill within the tree protection area south of Trees #419, #523, and #526. The applicant shall update the cut and fill evaluation to include the location of the tree protection fencing as depicted on the existing conditions and tree retention inventory (Exhibit C, Sheet 2). The applicant shall not cut or fill within the proposed tree protection area.

17.56.30 REQUIRED MAP AND STUDIES

- A. Topographic Map Required. To determine the location of potentially hazardous areas, the applicant shall submit a scaled topographic map at two-foot contour intervals for the subject property (site) and for land within 25 feet of the site perimeter. In addition to DOGAMI slide hazard areas and slopes of 25% or greater, potentially hazardous "analysis areas" include land within 25 feet of the top or toe of slopes of 25% or greater and the area 25 feet on either side of drainageways that drain 20 acres or more. This map shall be prepared by a registered engineer or land surveyor and shall show:
 - 1. Slopes of 25-34%;
 - 2. Slopes of 35% and greater;

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- 3. The analysis that is within 25' of slopes that are 25% or greater parallel to and within 25 feet of the top of the 25% slope break;
- 4. Mapped DOGAMI slide hazard areas;
- 5. The analysis area within 25 feet of the centerline of drainageways that drain at least 20 acres; and
- 6. The area (in square feet) for each category listed above for the subject property.
- B. <u>Types of Required Studies</u>. There are three types of geological and engineering studies that may be required by this chapter. See Table 1 under Section 17.56.40, below.
 - <u>Geological Assessments</u> are prepared and stamped by a Certified Engineering Geologist and describe the surface and subsurface conditions of a site, delineate areas of a property that may be subject to specific geologic hazards, and assess the suitability of the site for development. Geotechnical Reports shall be conducted according to the requirements of Appendix A (*Geological Assessments*), shall make recommendations as to whether further studies are required, and may be incorporated into or included as an appendix to the geotechnical report.
 - 2. Engineering Geology Reports are prepared and stamped by a Certified Engineering Geologist and provide detailed descriptions of the geology of the site, professional conclusions and recommendations regarding the effect of geological conditions on the proposed development, and opinions and recommendations covering the adequacy of the site to be developed. Engineering Geology Reports shall be prepared in accordance with the requirements of Appendix B (*Guidelines for Preparing Engineering Geology Reports in Oregon* adopted by the Oregon State Board of Geologist Examiners) and may be incorporated into or included as an appendix to the geotechnical report.
 - 3. <u>Geotechnical Reports</u> are prepared and stamped by a Geotechnical Engineer, evaluate site conditions, and recommend design measures necessary to reduce the development risks and facilitate safe and stable development. Geotechnical Reports shall be conducted according to the requirements of Appendix C (Geotechnical Reports), and may be incorporated into or included as an appendix to the Engineering Geology Report.

RESPONSE: Rapid Soil Solutions LLC completed a Geotechnical Report (Exhibit G) dated February 23, 2018 for the subject site. **The applicant shall follow the recommendations outlined in the Geotechnical Report.**

17.56.40 WHERE STUDIES REQUIRED

Additional geological or engineering studies shall be required, or not required, under the following circumstances:

Situation	Type I Development Applications; Single Family Homes, Duplexes and Accessory Uses	NON-EXEMPT Grading; Type II or III Develop- ment Applications
A. Proposed development avoids slopes of 25% or greater, drainageways, DOGAMI slope hazard areas and all analysis areas:	1. No further requirements	2. No further requirements

TABLE 1: WHERE STUDIES ARE REQUIRED OR NOT REQUIRED

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

B. Development proposed	1. Geological Assessment	2. Engineering Geology
on slopes of 25-35% or	required; Engineering	Report required;
analysis areas, but avoids	Geology or Geotechnical	Geotechnical Report may
drainageways, DOGAMI	Reports may be required*	be required*
hazard areas and slopes of		_
35% or greater:		
C. Development proposed	1. Engineering Geology	2. Engineering Geology
on DOGAMI hazard areas,	Report required;	Report and Geotechnical
slopes of 35% or greater,	Geotechnical Report may be	Report required
or drainageway areas:	required*	

* Whether additional studies are necessary depends on recommendations of base required study.

RESPONSE: The applicant shall submit a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay. This includes Lots 19, 20, 21, 27, 28, and 29. Additional reports may be required depending on the recommendations of the geological assessment. Once the subdivision is platted and building permits are filed for individual home construction the property owner/contractor shall submit Type I Hillside applications with studies as detailed in Table 1 in Section 17.56.40.

17.56.50 COMPLIANCE WITH STUDY CONCLUSIONS AND RECOMMENDATIONS REQUIRED

- A. <u>Professional Standards</u>. The director shall determine whether Geological Assessments, Engineering Geology Reports, or Geotechnical Reports have been prepared in accordance with Section 17.56.30. The director may require additional information or analysis necessary to meet study requirements.
- B. <u>Peer Review</u>. The director may require peer review of any required report, in which case regulated activities and uses shall be reviewed and accepted through the peer review process before any regulated activity will be allowed.
 - 1. A professional or professional firm of the city's choice that meets the qualifications listed in this chapter shall perform the review.
 - 2. The review shall be at the applicant's expense.
 - 3. Review of report submittals shall determine whether required elements are completed, geologic report procedures and assumptions are accepted, and all conclusions and recommendations are supported and reasonable.
- C. <u>Review Criteria</u>. The approval authority shall rely on the conclusions and recommendations of required reports, as modified by peer review, to determine compliance with this chapter.
- D. <u>Conditions of Approval</u>. Conclusions and recommendations stated in approved reports shall be directly incorporated as permit conditions or provide the basis for conditions of approval for the regulated activity or use.
- E. <u>Expiration</u>. Where an approved assessment or report as defined and required by this chapter has been prepared within the last five years for a specific site, and where the proposed land use activity and surrounding site conditions are unchanged, that report may be utilized and a new report is not required. Should environmental conditions associated with the site or surrounding the site change, or if the proposed land use activity or development has materially changed, the applicant shall submit

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

an amendment to the required assessment or report, which may be reviewed and approved through the peer review process.

RESPONSE: The Geotechnical Report includes construction recommendations related to site preparation, excavation, structural fills, and drainage considerations. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.

CHAPTER 17.60 - FLOOD AND SLOPE HAZARD OVERLAY DISTRICT

17.60.00 INTENT

This chapter is intended to promote the public health, safety and general welfare by minimizing public and private adverse impacts from flooding, erosion, landslides or degradation of water quality consistent with Statewide Planning Goals 6 (Air, Land and Water Resources Quality) and 7 (Areas Subject to Natural Disasters and Hazards) and the Sandy Comprehensive Plan (SCP).

RESPONSE: The applicant is not proposing any development within the Flood and Slope Hazard (FSH) overlay district. However, the proposal includes 6 lots that that are proposed to be platted within the FSH overlay. Staff brought the subject subdivision proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board recommends that a tract is created and dedicated to the City and cited three primary reasons that they would like to see the land dedicated: 1) increased protection of the trees located within the FSH buffer area; 2) a larger area through which a connector trail to Tickle Creek Trail could meander; and, 3) to provide a wider riparian buffer that would not only better protect Tickle Creek, but would also enhance the natural beauty along the future extension of the Tickle Creek Trail. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the FSH tract without knowing the existing condition of that area and recommended that, prior to dedicating the FSH tract to the City, the applicant would need to: 1) submit a Phase I environmental assessment and a native vegetation report for the FSH area to be dedicated in the tract, 2) remove all invasive plants and re-plant any areas of exposed soil with native plants, and 3) submit a maintenance bond for two full growing seasons (May 1) - October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Staff supported these recommendations in the February 25, 2019 staff report; however, the applicant's representative indicated that the applicant did not want to incur the expense of having to restore the FSH tract and conduct a Phase I environmental assessment. If the FSH tract is not dedicated to the City, it will not be necessary to perform a Phase I environmental assessment. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants. Due to the site layout, Lots 19-21 will end up less than 7,500 square feet once the FSH area is put into a separate tract. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

area and allowing the rear setback to be reduced to 15 feet. This will help avoid confusion over property ownership, fence lines, and unnecessary or illegal removal of riparian vegetation or vegetation on steep slopes.

The proposal also contains two trees that are proposed for removal and appear to be on the FSH boundary line: Trees #510 and #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.

17.60.30 REQUIRED SETBACK AREAS

Setback areas shall be required to protect water quality and maintain slope stability near stream corridors and locally significant wetlands. Setbacks are measured horizontally from, parallel to and upland from the protected feature.

- A. <u>Required Setbacks</u>. The required special setback(s) shall be:
 - 1. 70 feet from the top of bank of Tickle Creek;
 - 2. 50 feet from top of bank along other perennial streams, except for "No Name Creek" east of Towle Drive, as provided in Section 17.60.30.C.2 below.
 - 3. 25 feet around the edge of any mapped locally significant wetland; and
 - 4. 25 feet from the top of any 25% slope break where the slope break occurs within the FSH overlay district as mapped by the city.

RESPONSE: The applicant submitted a Wetland Determination (Exhibit H) by Environmental Science & Assessment, LLC dated May 30, 2018. The wetland determination concluded there is no evidence of intermittent or perennial flow on the site. Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). Initially, the City Engineer required that the wetland report/delineation shall be concurred with by the State of Oregon Division of State Lands (ODSL) and the US Army Corps of Engineers (COE) and the Oregon Department of Fish and Wildlife. In a subsequent email to the applicant's representative, the City Engineer decided his original comment had been met and there was no need to include concurrence as a condition of approval. The City Engineer recommends that the City of Sandy Planning Division review the FSH Overlay District line at the northeast corner of the site.

17.60.50 SPECIAL REPORTS

Where development is proposed on restricted development areas within the FSH overlay district as defined in Section 17.60.20.A, the Director shall require submission of the following special reports. These reports shall be in addition to other information required for specific types of development, and shall be prepared by professionals in their respective fields.

The Director may require one of more of these reports where necessary to address potential adverse impacts from development on buildable land within the FSH overlay district. The Director may exempt Type II permit applications from one or more of these reports where impacts are minimal and the exemption is consistent with the purpose of the FSH overlay zone as stated in Section 17.60.00.

A. <u>Hydrology and Soils Report</u>. This report shall include information on the hydrological conditions on the site, the effect of hydrologic conditions on the proposed development, the proposed

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

development's impact on surface and groundwater flows to wetlands and streams, and any hydrological or erosion hazards. This report shall also include soils characteristics of the site, their suitability for development, carrying capacity, and erosion or slumping characteristics that might present a hazard to life and property, or adversely affect the use or stability of a public facility or utility. Finally, this report shall include information on the nature, distribution and strength of existing soils; the adequacy of the site for development purposes; and an assessment of grading procedures required to impose the minimum disturbance to the natural state. A licensed professional engineer registered in Oregon shall prepare the hydrology and soils report.

- B. <u>Grading Plan.</u> The grading plan shall be specific to a proposed physical structure or use and shall include information on terrain (two-foot intervals of property), drainage, direction of drainage flow, location of proposed structures and existing structures which may be affected by the proposed grading operations, water quality facilities, finished contours or elevations, including all cut and fill slopes and proposed drainage channels. Project designs including but not limited to locations of surface and subsurface devices, walls, dams, sediment basins, storage reservoirs, and other protective devices shall form part of the submission. The grading plan shall also include: 1) construction phase erosion control plan consistent with the provisions of Chapter 15.44; and 2) schedule of operations. A licensed professional engineer registered in Oregon shall prepare the grading and erosion control plan.
- C. <u>Native Vegetation Report</u>. This report shall consist of a survey of existing vegetative cover, whether it is native or introduced, and how it will be altered by the proposed development. Measures for revegetation with native plant species will be clearly stated, as well as methods for immediate and long-term stabilization of slopes and control of soil erosion. A landscape architect, landscape designer, botanist or arborist with specific knowledge of native plant species, planting and maintenance methods, survival rates, and their ability to control erosion and sedimentation shall prepare the vegetation report. The applicant shall be responsible for replacing any native plant species that do not survive the first two years after planting, and for ensuring the survival of any replacement plants for an additional two years after their replacement.

RESPONSE: Per the City Engineer (Exhibit L), a Wetlands Report outlining the delineated wetlands/high water level appears to be necessary based on the National Wetland Inventory (NWI) or the Local Wetland Inventory (LWI). The applicant submitted a wetland determination (Exhibit H) that the City Engineer determined satisfies the requirement.

17.60.80 WATER QUALITY TREATMENT FACILITIES

Tickle Creek, the Sandy River and associated natural drainage ways are vital to Sandy's recreationally based economy and to the quality of life of Sandy residents. Placement of water quality facilities shall be limited as follows:

- A. The water quality facility shall not be constructed in restricted development areas, except where necessary to serve approved development within restricted development areas (e.g., a road) and where no reasonable alternative exists in buildable areas of the site.
- B. Where the approval authority determines that a more efficient and effective regional site exists within the sub-basin, the water quality facility may be constructed off-site.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: The proposed detention pond (Tract A) is located in the northwest corner of the site and is not within the mapped FSH overlay area.

17.60.90 DENSITY TRANSFER PROVISIONS

Residential density transfer may be approved subject to the following:

- A. Required Setback Areas. Density may be transferred from restricted development areas (i.e., steep slopes, protected water features and required setbacks) to buildable portions of the site.
- B. Density Maximum. The maximum gross density for the buildable area of the site shall not exceed 150% of the maximum density allowed by the underlying zoning district for that buildable area.
- C. Housing Types Not Permitted in Underlying Zoning District. Housing types not permitted in the underlying zoning district may only be approved through the PD (planned development) or SAP (specific area plan) process.
- D. Transfer Area. Transfer of density may only occur within the same property and/or to properties contiguous to the primary property. The terms "primary property" identify the legal lot from which density is to be transferred to "secondary property(s)". Further development or land use action on the primary or secondary properties shall be reviewed together in the same application.

RESPONSE: Since no development is proposed within the FSH area, a density transfer per Section 17.60.90 is allowed for the buildable area of the site. There are 1.20 acres of restricted development area. See Chapter 17.30 for density analysis.

CHAPTER 17.80 - ADDITIONAL SETBACKS ON COLLECTOR AND ARTERIAL STREETS

17.80.10 APPLICABILITY

Collector: Jacoby Road

RESPONSE: Four lots (Lots 9, 10, 30 and 32) are proposed to contain frontage on Jacoby Road, which is classified as a collector street. All structures on lots along Jacoby Road shall be constructed to comply with the standards of Section 17.80.20.

17.80.20 SPECIFIC SETBACKS

Any structure located on streets listed above or identified in the Transportation System Plan as arterials or collectors shall have a minimum setback of 20 feet measured from the property line. This applies to applicable front, rear and side yards.

RESPONSE: All structures shall maintain a minimum 20-foot setback from the Jacoby Road public right-of-way. The Preliminary Plat (Exhibit C, Sheet 1) depicts building envelopes at 20 feet from the Jacoby Road right-of-way.

CHAPTER 17.82 – SPECIAL SETBACKS ON TRANSIT STREETS

17.82.20 BUILDING ORIENTATION

A. All buildings shall have their primary entrances oriented toward a transit street rather than a parking area, or if not adjacent to a transit street, toward a public right-of-way or private walkway which leads to a transit street.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- B. Buildings shall have a primary entrance connecting directly between the street and building interior. This entrance shall be open to the public during all business hours and shall comply with the accessibility standards of the Uniform Building Code.
- C. In lieu of a building entrance oriented to a transit street, a building's entrance may be enhanced and identified in the following manner:
 - 1. An entrance plaza of at least 150 square feet, at least 100 square feet of which shall be visible from the transit street. The entrance plaza shall be at least 10 feet wide at the narrowest dimension; and
 - 2. A permanent building feature (e.g. a portico, porch or awning) shall be visible from the transit street, signifying an entrance; and
 - 3. Pedestrian-scale lighting shall be required at the entrance; and
 - 4. A clearly marked, convenient, safe and lighted pedestrian route shall be provided to the entrance, from the transit street.
- D. Primary building entrances shall be architecturally emphasized and visible from the street. Building entrances shall incorporate arcades, roofs, porches, alcoves, porticoes, and awnings that protect pedestrians from the rain and sun. Continuous arcades are strongly encouraged.
- E. All building entrances and exits shall be well lit. Lighting shall be a pedestrian scale (3'-12') and the source light shall be shielded to reduce glare.
- F. For commercial buildings with facades over 300 feet in length on a transit street, two or more building entrances on the street must be provided.
- G. If the site has frontage on more than one transit street, the building shall provide one main entrance oriented to a transit street or to a corner where two transit streets intersect.

RESPONSE: Jacoby Road is a designated transit street. Lots 9, 10, 30, and 32 have frontage on Jacoby Road. As part of the Jacoby Heights subdivision proposal, the applicant has requested a variance to the requirements of Section 17.82.20 to allow the front door for the houses on Lots 9, 10, 30, and 32 to face the internal street network instead of Jacoby Road, which is a designated transit street. The variance request is discussed in detail in the analysis of Chapter 17.66 in this staff report.

CHAPTER 17.84 - IMPROVEMENTS REQUIRED WITH DEVELOPMENT

This chapter provides general information regarding improvements required in association with development, and it clarifies the timing, extent, and standards for public and private improvements.

17.84.20 TIMING OF IMPROVEMENTS

- A. All improvements required by the standards in this chapter shall be installed concurrently with development, as follows:
 - 1. Where a land division is proposed, each proposed lot shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to approval of the final plat.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

2. Where a land division is not proposed, the site shall have required public and franchise utility improvements installed or financially guaranteed in accordance with the provisions of Chapter 17 prior to temporary or final occupancy of structures.

RESPONSE: Submission of preliminary street and utility plans during the land use review process is solely for compliance with the data requirements of Section 17.100.60 (D). **Public improvement** plans are subject to a separate review and approval process. Preliminary plat approval does not connote approval of public improvement construction plans.

B. Where specific approval for a phasing plan has been granted for a planned development and/or subdivision, improvements may similarly be phased in accordance with that plan.

RESPONSE: The applicant did not propose a phasing plan with this application.

17.84.30 PEDESTRIAN AND BICYCLIST IMPROVEMENTS

- A. Sidewalks shall be required along both sides of all arterial, collector, and local streets, as follows:
 - 1. Sidewalks shall be a minimum of 5 ft. wide on local streets. The sidewalks shall be separated from curbs by a tree planting area that provides separation between sidewalk and curb, unless modified in accordance with Subsection 3 below.
 - 2. Sidewalks along arterial and collector streets shall be separated from curbs with a planting area, except as necessary to continue an existing curb-tight sidewalk. The planting area shall be landscaped with trees and plant materials approved by the City. The sidewalks shall be a minimum of 6 ft. wide.
 - 3. Sidewalk improvements shall be made according to city standards, unless the city determines that the public benefit in the particular case does not warrant imposing a severe adverse impact to a natural or other significant feature such as requiring removal of a mature tree, requiring undue grading, or requiring modification to an existing building. Any exceptions to the standards shall generally be in the following order.
 - a) Narrow landscape strips
 - b) Narrow sidewalk or portion of sidewalk to no less than 4 feet in width
 - c) Eliminate landscape strips
 - d) Narrow on-street improvements by eliminating on-street parking
 - e) Eliminate sidewalks
 - 4. The timing of the installation of sidewalks shall be as follows:
 - a) Sidewalks and planted areas along arterial and collector streets shall be installed with street improvements, or with development of the site if street improvements are deferred.
 - b) Sidewalks along local streets shall be installed in conjunction with development of the site, generally with building permits, except as noted in (c) below.
 - c) Where sidewalks on local streets abut common areas, drainageways, or other publicly owned or semi-publicly owned areas, the sidewalks and planted areas shall be installed with street improvements.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) depicts an 8-foot-wide public utility easement along the frontage of all proposed lots to facilitate future installation of utilities. Sidewalks abutting the proposed lots shall be constructed in association with development of the lots. The applicant shall construct sidewalks along Tract A both on Jacoby Road and Camden Court, prior to final plat approval. The sidewalks on Camden Court, Woodstock Street, Newton Street, and

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Middlebury Avenue shall be five feet in width and separated by a five foot planter strip (or 6 foot swale) in areas not transverse by driveways. The sidewalks on Jacoby Road shall be six feet in width and separated by a five foot planter strip.

The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all streets. The section of Newton Street on the east side of the property adjacent to Lots 1 and 2 is proposed to be constructed with half street improvements. This proposed design places the utilities underneath the planter strip; thus, the proposal does not show street trees in front of Lots 1 and 2. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a three-quarter street. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. Street trees shall be installed approximately 30 feet on center in conjunction with issuance of building permits. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted and staked per the City of Sandy standard planting detail; trees shall be tied to the stakes with loosely tied twine. The planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed within one year of installation.

- B. Safe and convenient pedestrian and bicyclist facilities that strive to minimize travel distance to the extent practicable shall be provided in conjunction with new development within and between new subdivisions, planned developments, commercial developments, industrial areas, residential areas, public transit stops, school transit stops, and neighborhood activity centers such as schools and parks, as follows:
 - 1. For the purposes of this section, "safe and convenient" means pedestrian and bicyclist facilities that: are reasonably free from hazards which would interfere with or discourage travel for short trips; provide a direct route of travel between destinations; and meet the travel needs of pedestrians and bicyclists considering destination and length of trip.
 - 2. To meet the intent of "B" above, right-of-ways connecting cul-de-sacs or passing through unusually long or oddly shaped blocks shall be a minimum of 15 ft. wide with 8 feet of pavement.
 - 3. 12 feet wide pathways shall be provided in areas with high bicycle volumes or multiple use by bicyclists, pedestrians, and joggers.
 - 4. Pathways and sidewalks shall be encouraged in new developments by clustering buildings or constructing convenient pedestrian ways. Pedestrian walkways shall be provided in accordance with the following standards:
 - a) The pedestrian circulation system shall be at least five feet in width and shall connect the sidewalk on each abutting street to the main entrance of the primary structure on the site to minimize out of direction pedestrian travel.
 - b) Walkways at least five feet in width shall be provided to connect the pedestrian circulation system with existing or planned pedestrian facilities which abut the site but are not adjacent to the streets abutting the site.
 - c) Walkways shall be as direct as possible and avoid unnecessary meandering.
 - d) Walkway/driveway crossings shall be minimized. Internal parking lot design shall maintain ease of access for pedestrians from abutting streets, pedestrian facilities, and transit stops.
 - e) With the exception of walkway/driveway crossings, walkways shall be separated from vehicle parking or vehicle maneuvering areas by grade, different paving material, painted

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

crosshatching or landscaping. They shall be constructed in accordance with the sidewalk standards adopted by the City. (This provision does not require a separated walkway system to collect drivers and passengers from cars that have parked on site unless an unusual parking lot hazard exists).

f) Pedestrian amenities such as covered walk-ways, awnings, visual corridors and benches will be encouraged. For every two benches provided, the minimum parking requirements will be reduced by one, up to a maximum of four benches per site. Benches shall have direct access to the circulation system.

RESPONSE: Five foot wide sidewalks are proposed on local streets and six foot wide sidewalks are proposed along Jacoby Road adjacent to the proposed subdivision. These sidewalks provide safe and convenient pedestrian access within the subdivision and connect the site to neighboring properties. To improve pedestrian connectivity, **the applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west. At the February 25, 2019 Planning Commission hearing, the applicant's engineer stated it would be impossible to design an ADA ramp at that location on Jacoby Road due to the slope. The City Engineer and Public Works Director stated that the mid-block ramp should be easily achievable if the street grade is 7.5 percent and recommend the condition remain as written.**

C. Where a development site is traversed by or adjacent to a future trail linkage identified within the Transportation System Plan, improvement of the trail linkage shall occur concurrent with development. Dedication of the trail to the City shall be provided in accordance with 17.84.80.

RESPONSE: The City's current TSP maps were created with the former UGB boundaries (pre-June 2017) and did not include the subject property that was brought into the revised UGB boundaries. Therefore, there are no trail linkages identified in the TSP for this property.

E. To ensure improved access between a development site and an existing developed facility such as a commercial center, school, park, or trail system, the Planning Commission or Director may require off-site pedestrian facility improvements concurrent with development.

RESPONSE: Tickle Creek crosses the northeast corner of the subject property and the future extension of the Tickle Creek Trail is planned to extend along the length of Tickle Creek. At the preapplication meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trail on the adjacent property to the east (tax lot 800). Staff brought this proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board also recommends that the connector trail be constructed as part of this development. While the Park Facilities Plan (Exhibit Y) depicts the future Tickle Creek Trail located along Tickle Creek in the northeast corner of the subject property, the Park Facilities Plan doesn't go as far as identifying specific trail connections to the proposed future Tickle Creek Trail alignment. Without specific trail connectors identified explicitly on the plan, staff cannot compel the applicant to include a trail connecting to the future Tickle Creek Trail. However, staff and the Parks & Trails Advisory Board recommend the applicant include a connector trail on the subject property that provides access to the future Tickle Creek Trail.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.84.40 TRANSIT AND SCHOOL BUS TRANSIT REQUIREMENTS

- A. Development sites located along existing or planned transit routes shall, where appropriate, incorporate bus pull-outs and/or shelters into the site design. These improvements shall be installed in accordance with the guidelines and standards of the transit agency. School bus pull-outs and/or shelters may also be required, where appropriate, as a condition of approval for a residential development of greater than 50 dwelling units where a school bus pick-up point is anticipated to serve a large number of children.
- B. New developments at or near existing or planned transit or school bus transit stops shall design development sites to provide safe, convenient access to the transit system, as follows:
 - 1. Commercial and civic use developments shall provide a prominent entrance oriented towards arterial and collector streets, with front setbacks reduced as much as possible to provide access for pedestrians, bicycles, and transit.
 - 2. All developments shall provide safe, convenient pedestrian walkways between the buildings and the transit stop, in accordance with the provisions of 17.84.30 B.

RESPONSE: The Transit Director (Exhibit P) states that the proposed development will require construction of one transit amenity at the northwest corner of the subject property adjacent to Tract A and as close to Cascadia Village Drive as possible. The amenity required is a 7 foot by 9.5 foot concrete pad to accommodate a 5 foot by 7.5 foot bus shelter, equipped with one 6 foot long green metal bench. The applicant shall install one bench (Fairweather model PL-3, powder-coated RAL6028) with a concrete pad to City specifications. The applicant shall contact the Transit Director for specific location, amenity information, and pad engineering specifications.

17.84.50 STREET REQUIREMENTS

- A. Traffic evaluations may be required of all development proposals in accordance with the following:
 - A proposal establishing the scope of the traffic evaluation shall be submitted for review to the City Engineer. The evaluation requirements shall reflect the magnitude of the project in accordance with accepted traffic engineering practices. Large projects should assess all nearby key intersections. Once the scope of the traffic evaluation has been approved, the applicant shall present the results with and an overall site development proposal. If required by the City Engineer, such evaluations shall be signed by a Licensed Professional Civil Engineer or Licensed Professional Traffic Engineer licensed in the State of Oregon.
 - 2. If the traffic evaluation identifies level-of-service conditions less than the minimum standard established in the Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered concurrent with a development proposal.

RESPONSE: The submitted Traffic Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10th Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

per City of Sandy and ODOT standards and are projected to continue to operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development.

The City Engineer (Exhibit L) reviewed the Traffic Impact Analysis and noted the following: "There are no reported crashes at the intersection of Jacoby [Road] and Dubarko Road, or at any of the intersections adjacent to the development. However, the crash rate at the intersection of Dubarko Road and Highway 211 exceeds the 90th percentile criteria and meets the warrants for the addition of left turn lanes as identified in the TSP. This intersection should be addressed by ODOT as soon as possible, but not as a component of this proposed development."

The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and does not recommend any modifications to the TIA as the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards. The City Transportation Engineer states the following: "The intersections that will provide access from the subdivision to Jacoby Road are appropriately located and provide adequate sight distance. The existence of a higher than average crash rate at the intersection of OR 211 and Dubarko Road and traffic volumes that suggest left-turn lanes are warranted on OR 211 at this location are indicators that the city should continue to work with ODOT for implementation of enhancements at this intersection. This analysis indicates that a traffic signal is not yet warranted but left-turn lanes on both the northbound and southbound approaches on OR 211 are warranted. The need for these improvements is justified based on regional growth of which this development is only a minor component."

- B. Location of new arterial streets shall conform to the Transportation System Plan in accordance with the following:
 - 1. Arterial streets should generally be spaced in one-mile intervals.
 - 2. Traffic signals should generally not be spaced closer than 1500 ft. for reasonable traffic progression.

RESPONSE: The proposal does not include any new arterial or collector streets.

- C. Local streets shall be designed to discourage through traffic. NOTE: for the purposes of this section, "through traffic" means the traffic traveling through an area that does not have a local origination or destination. To discourage through traffic and excessive vehicle speeds the following street design characteristics shall be considered, as well as other designs intended to discourage traffic:
 - 1. Straight segments of local streets should be kept to less than a quarter mile in length. As practical, local streets should include traffic calming features, and design features such as curves and "T" intersections while maintaining pedestrian connectivity.
 - 2. Local streets should typically intersect in "T" configurations rather than 4-way intersections to minimize conflicts and discourage through traffic. Adjacent "T" intersections shall maintain a minimum of 150 ft. between the nearest edges of the 2 rights-of-way.
 - 3. Cul-de-sacs should generally not exceed 400 ft. in length nor serve more than 20 dwelling units, except in cases where existing topography, wetlands, or drainage systems or other existing features necessitate a longer cul-de-sac in order to provide adequate access to an area. Cul-de-sacs longer than 400 feet or developments with only one access point may be required to provide

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

an alternative access for emergency vehicle use only, install fire prevention sprinklers, or provide other mitigating measures, determined by the City.

RESPONSE: The proposal includes a cul-de-sac (Camden Court) that does not exceed 400 feet in length. Five lots and Tract A are proposed to gain access from the cul-de-sac.

- D. Development sites shall be provided with access from a public street improved to City standards in accordance with the following:
 - 1. Where a development site abuts an existing public street not improved to City standards, the abutting street shall be improved to City standards along the full frontage of the property concurrent with development.
 - 2. Half-street improvements are considered the minimum required improvement. Three-quarterstreet or full-street improvements shall be required where traffic volumes generated by the development are such that a half-street improvement would cause safety and/or capacity problems. Such a determination shall be made by the City Engineer.
 - 3. To ensure improved access to a development site consistent with policies on orderly urbanization and extension of public facilities the Planning Commission or Director may require off-site improvements concurrent with development. Off-site improvement requirements upon the site developer shall be reasonably related to the anticipated impacts of the development.
 - 4. Reimbursement agreements for ³/₄ street improvements (i.e., curb face to curb face) may be requested by the developer per Chapter 12 of the SMC.
 - 5. A ¹/₂ street improvement includes curb and pavement 2 feet beyond the center line of the right-ofway. A ³/₄ street improvement includes curbs on both sides of the side and full pavement between curb faces.

RESPONSE: The proposed street network and improvements generally comply with City standards, with the exception of one block variance. There are five streets inside or adjacent to the proposed subdivision requiring the following improvements:

<u>Jacoby Road</u>: Half-street construction of Jacoby Road is required within the subject property boundary to collector street standards, including curb, 6-foot wide concrete sidewalks, street lighting, landscaping strip, street trees, and public utilities. The site has approximately 600 linear feet of frontage on Jacoby Road, a designated collector. The right-of-way width adjacent to the site is 50 feet; the standard for collector streets is 60 feet. Per the City Engineer (Exhibit L) and Public Works Director (Exhibit O), the applicant shall dedicate 10 feet of right-of-way along the Jacoby Road frontage of the site.

<u>Woodstock Street</u>: Woodstock Street requires full-street construction to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities. The applicant has applied for one variance to exceed the maximum block length for the north side of Woodstock Street from Camden Court to the east property boundary. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." **The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. At the February 25, 2019 Planning Commission hearing, the applicant's representative indicated the applicant doesn't have an easement from the neighbor**

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

and obtaining one could be burdensome. In response, the Public Works Director and City Engineer note: "The condition gives the applicant the option of a retaining wall or an easement to allow extension of the street. The condition doesn't require neighbor participation nor is it contingent on same. Based on the grading plan the wall would be a fill wall, less about 2 feet in height. This is neither problematic or [sic] expensive. The bottom of the wall would be above any required utilities (which are shown extending all the way to the property line. The typical method when extending a street with a fill wall is to remove the top course or two and leave the rest of the wall in place and bury it when the street is extended, no removal required. Even if removal would be required it would be no more expensive than removing the soil (whether fill or cut) at the end of the street when extended." The Public Works Director and City Engineer recommend the condition remain as written.

<u>Middlebury Avenue</u>: Middlebury Avenue requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

<u>Newton Street</u>: Newton Street is proposed to align with the existing alignment of Newton Street west of Jacoby Road and to then shift to the south such that it aligns with the southern property boundary. The extension of Newton Street requires full-street improvements to local standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, street trees, ADA ramps, and public utilities adjacent to Lot 9, tapering to three-quarter street improvements adjacent to the remaining lots along Newton Street. The applicant is proposing a three-quarter street improvement for Newton Street through the site transitioning to a half-street section at the west line of proposed Lot 2. Per the Public Works Director (Exhibit O), the proposed Newton Street alignment coincides with an existing access easement serving at least two and possibly as many as four existing dwellings to the east. No more than two dwellings may be served by a half-street improvement. As proposed this narrowed section of Newton Street would serve at least four dwellings and perhaps as many as six. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. Per the City Engineer (Exhibit L), the three-quarter street improvements on Newton Street shall include a 28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot sidewalks on the north side of the roadway. At the February 25, 2019 Planning Commission hearing, the applicant's representative's memo (Exhibit W) indicated the applicant's "proposal to narrow this street was intended to allow the home on the adjacent property to the East to the saved." The Public Works Director and City Engineer (Exhibit X) provided the following response: "All the City has is the application before it. The City shouldn't speculate on what may or may not happen on the adjacent property. The adjacent property owner has been noticed and if they have a better idea of what the future holds for their property or the existing dwelling than [sic] they should submit testimony or comments reflecting this. A 40 ft. right-of-way dedication and a ³/₄-street improvement should be required to serve more than two single-family dwellings. If the applicant wants to accommodate future development of the parcel to the east then the applicant needs to figure out how to do this and still provide the minimum ³/₄-street section on the development site or acquire land to accommodate it on adjacent properties. The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a three-quarter street.

The preliminary street and utility plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW dedication for staff review and approval.

<u>Camden Court</u>: Camden Court requires full-street improvements to local street standards including concrete curbs, 5-foot wide concrete sidewalks, street lighting, 5-foot wide planter strip, street trees, ADA ramps, and public utilities.

- E. As necessary to provide for orderly development of adjacent properties, public streets installed concurrent with development of a site shall be extended through the site to the edge of the adjacent property(ies) in accordance with the following:
 - 1. Temporary dead-ends created by this requirement to extend street improvements to the edge of adjacent properties may be installed without turn-arounds, subject to the approval of the Fire Marshal.
 - 2. In order to assure the eventual continuation or completion of the street, reserve strips may be required.

RESPONSE: The applicant is not proposing any permanent dead-end streets. The applicant proposes that Newton Street, Woodstock Street, and Middlebury Avenue be temporary dead-end streets with construction of this subdivision until such a time as these streets are extended onto the adjoining properties to the east and south. The applicant shall plat a vehicle non-access reserve (VNAR) strip at the east end of Woodstock Street and along the south side of the right-of-way of Newton Street, with the exception of the easternmost end of the south side of Newton Street to allow access to the newly constructed street from the existing gravel drive; the length of the VNAR for the south side of Newton Street shall be determined during Construction Plan review. The dead-ends on Woodstock Street and Newton Street would be approximately 190 feet. The applicant shall work with the Fire Marshall to determine if a temporary turn-around is needed on Woodstock Street and/or Newton Street, or if houses on particular lots need to be sprinkled. The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the street improvements on Woodstock Street ending approximately 7 feet from the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code.

F. Where required by the Planning Commission or Director, public street improvements may be required through a development site to provide for the logical extension of an existing street network or to connect a site with a nearby neighborhood activity center, such as a school or park. Where this creates a land division incidental to the development, a land partition shall be completed concurrent with the development.

RESPONSE: The proposal includes the extension of Newton Street from the Cascadia Village subdivision.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

G. Except for extensions of existing streets, no street names shall be used that will duplicate or be confused with names of existing streets. Street names and numbers shall conform to the established pattern in the surrounding area and be subject to approval of the Director.

RESPONSE: The applicant is proposing the following new street names: Camden Court, Woodstock Street, and Middlebury Avenue. **The City of Sandy reserves the right to name streets.**

- H. Location, grades, alignment, and widths for all public streets shall be considered in relation to existing and planned streets, topographical conditions, public convenience and safety, and proposed land use. Where topographical conditions present special circumstances, exceptions to these standards may be granted by the City Engineer provided the safety and capacity of the street network is not adversely affected. The following standards shall apply:
 - 1. Location of streets in a development shall not preclude development of adjacent properties. Streets shall conform to planned street extensions identified in the Transportation Plan and/or provide for continuation of the existing street network in the surrounding area.
 - 2. Grades shall not exceed 6 percent on arterial streets, 10 percent on collector streets, and 15 percent on local streets.
 - 3. As far as practical, arterial streets and collector streets shall be extended in alignment with existing streets by continuation of the street centerline. When staggered street alignments resulting in "T" intersections are unavoidable, they shall leave a minimum of 150 ft. between the nearest edges of the two rights-of-way.
 - 4. Centerline radii of curves shall not be less than 500 ft. on arterial streets, 300 ft. on collector streets, and 100 ft. on local streets.
 - 5. Streets shall be designed to intersect at angles as near as practicable to right angles and shall comply with the following:
 - a) The intersection of an arterial or collector street with another arterial or collector street shall have a minimum of 100 ft. of straight (tangent) alignment perpendicular to the intersection.
 - b) The intersection of a local street with another street shall have a minimum of 50 ft. of straight (tangent) alignment perpendicular to the intersection.
 - c) Where right angle intersections are not possible, exceptions can be granted by the City Engineer provided that intersections not at right angles have a minimum corner radius of 20 ft. along the right-of-way lines of the acute angle.
 - d) Intersections with arterial streets shall have a minimum curb corner radius of 20 ft. All other intersections shall have a minimum curb corner radius of 10 ft.
 - 6. Right-of-way and improvement widths shall be as specified by the Transportation System Plan. Exceptions to those specifications may be approved by the City Engineer to deal with specific unique physical constraints of the site.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) does not show the correct measurement of 50 feet of tangent alignment at the intersection of Jacoby Road and Newton Street as required by 17.84.50(H.5.b). The applicant measured the 50 feet of tangent alignment from the centerline of Jacoby Road rather than from the curb line extended. **The applicant shall revise the alignment of Newton Street to conform to the 50 feet of tangent alignment standard.**

17.84.60 PUBLIC UTILITY EXTENSIONS

A. All development sites shall be provided with public water, sanitary sewer, and storm drainage.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- B. Where necessary to serve property as specified in "A" above, required public utility installations shall be constructed concurrent with development.
- C. Off-site public utility extensions necessary to fully serve a development site and adjacent properties shall be constructed concurrent with development.
- D. As necessary to provide for orderly development of adjacent properties, public utilities installed concurrent with development of a site shall be extended through the site to the edge of adjacent property(ies).
- E. All public utility installations required with development shall conform to the City's facilities master plans.
- F. Private on-site sanitary sewer and storm drainage facilities may be considered provided all the following conditions exist:
 - 1. Extension of a public facility through the site is not necessary for the future orderly development of adjacent properties;
 - 2. The development site remains in one ownership and land division does not occur (with the exception of land divisions that may occur under the provisions of 17.84.50 F above);
 - 3. The facilities are designed and constructed in accordance with the Uniform Plumbing Code and other applicable codes, and permits and/or authorization to proceed with construction is issued prior to commencement of work.

RESPONSE: The applicant's Preliminary Street and Utility Plan (Exhibit C, Sheet 5) depicts the location and type of proposed public utilities including water, sanitary sewer and stormwater. All public utility installations shall conform to the City's facilities master plans. Staff recommends the applicant revise the utility plan to include broadband fiber locations as detailed by the SandyNet Manager. Per the City Engineer (Exhibit L), all public sanitary sewer and waterline mains shall be a minimum of 8 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. No building permits will be issued until all public utilities including sanitary sewer are available to serve the development. The applicant shall pay plan review, inspection, and permit fees as determined by the Public Works Director. The utility improvements proposal and requirements for the Jacoby Heights subdivision are further detailed in Sections 17.100.230, 17.100.240, and 17.100.250 below.

17.84.80 FRANCHISE UTILITY INSTALLATIONS

These standards are intended to supplement, not replace or supersede, requirements contained within individual franchise agreements the City has with providers of electrical power, telephone, cable television, and natural gas services (hereinafter referred to as "franchise utilities").

RESPONSE: Private utility services will be submitted for review and approval by service providers and City staff in association with construction plans, and all utility lines will be extended to the perimeter of the site. All franchise utilities shall be installed underground and in conformance with City standards. PGE submitted a comment (Exhibit N) stating they did not find any conflicts related to the project. Per PGE's request, the applicant shall call the PGE Service Coordinators at 503-323-6700 when the developer is ready to start the project.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.84.90 LAND FOR PUBLIC PURPOSES

- A. Easements for public sanitary sewer, water, storm drain, pedestrian and bicycle facilities shall be provided whenever these facilities are located outside a public right-of-way in accordance with the following:
 - 1. When located between adjacent lots, easements shall be provided on one side of a lot line.
 - 2. The minimum easement width for a single utility is 15 ft. The minimum easement width for two adjacent utilities is 20 ft. The easement width shall be centered on the utility to the greatest extent practicable. Wider easements may be required for unusually deep facilities.
- B. Public utility easements with a minimum width of 5 feet shall be provided adjacent to all street rights-of-way for franchise utility installations.
- C. Where a development site is traversed by a drainageway or water course, a drainage way dedication shall be provided to the City.
- D. Where a development is traversed by, or adjacent to, a future trail linkage identified within the Transportation System Plan, dedications of suitable width to accommodate the trail linkage shall be provided. This width shall be determined by the City Engineer, considering the type of trail facility involved.
- E. Where existing rights-of-way and/or easements within or adjacent to development sites are nonexistent or of insufficient width, dedications may be required. The need for and widths of those dedications shall be determined by the City Engineer.
- F. Where easement or dedications are required in conjunction with land divisions, they shall be recorded on the plat. Where a development does not include a land division, easements and/or dedications shall be recorded on standard document forms provided by the City Engineer.
- G. If the City has an interest in acquiring any portion of a proposed subdivision or planned development site for a public purpose, other than for those purposes listed above, or if the City has been advised of such interest by a school district or other public agency, and there is a reasonable assurance that steps will be taken to acquire the land, the Planning Commission may require those portions of the land be reserved for public acquisition for a period not to exceed 1 year.
- H. Environmental assessments for all lands to be dedicated to the public or City may be required to be provided by the developer. An environmental assessment shall include information necessary for the City to evaluate potential liability for environmental hazards, contamination, or required waste cleanups related to the dedicated land. An environmental assessment shall be completed prior to the acceptance of dedicated lands in accordance with the following:
 - The initial environmental assessment shall detail the history of ownership and general use of the land by past owners. Upon review of the information provided by the grantor, as well as any site investigation by the City, the Director will determine if the risks of potential contamination warrant further investigation. When further site investigation is warranted, a Level I Environmental Assessment shall be provided by the grantor.

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A). Rather than include the FSH overlay area as part of Lots 19, 20, 21, 26, 27, and 28, the applicant shall designate the FSH area as a separate tract on the plat, which shall be owned and maintained by an HOA.

The Plat shall detail the following easements:

• An eight-foot wide public utility easement (PUE) along the frontage of all proposed lots;

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Crossover easements along the common parcel line recorded between pairs of lots that share driveways (Lots 26 and 27, and Lots 28 and 29);
- 20 foot easement along the west boundary of Lots 9 and 10 for public sanitary sewer and public storm line. Per the Public Works Director (Exhibit O), the City reserves the right to widen this easement depending on the final design depth of the sanitary sewer and storm lines per Section 17.84.90(A.2);
- A 15 foot easement between Lot 30 and Tract A for public sanitary sewer;
- A 10 foot private storm drain easement at the rear of Lots 1-18, 22-25, 28, 29, 31, and 32 and a 10 foot private storm drain easement across Lots 19-21 just south of the FSH boundary;
- A blanket conservation easement and pedestrian easement over the entirety of the FSH tract;
- An 8 foot wide pedestrian maintenance access easement to the FSH tract (could be located on the east edge of Lot 19 or elsewhere); and
- A vehicle non-access reserve (VNAR) strip in the following locations:
 - Jacoby Road frontage of Lots 9, 10, 30 and 32 and Tract A; and,
 - South side of the right-of-way of Newton Street, with the exception of the easternmost section as determined necessary to allow access to the newly constructed Newton Street from the existing gravel driveway during Construction Plan review; and,
 - East end of the right-of-way of Woodstock Street.

17.84.100 MAIL DELIVERY FACILITIES

RESPONSE: The applicant will need to coordinate with the United States Postal Service (USPS) to locate mail facilities and these will be approved by the City and USPS. **Mail delivery facilities shall be** provided by the applicant in conformance with Section 17.84.100 and the standards of the USPS. The applicant shall submit a mail delivery plan, featuring grouped lockable mail facilities, to the City and USPS for review and approval prior to installation of mailboxes.

CHAPTER 17.86 – PARKLAND AND OPEN SPACE

17.86.10 MINIMUM PARKLAND DEDICATION REQUIREMENTS

<u>Parkland Dedication:</u> New residential subdivisions, planned developments, multi-family or manufactured home park developments shall be required to provide parkland to serve existing and future residents of those developments. Multi-family developments which provide some "congregate" services and/or facilities, such as group transportation, dining halls, emergency monitoring systems, etc., but which have individual dwelling units rather than sleeping quarters only, are considered to be multifamily developments for the purpose of parkland dedication. Licensed adult congregate living facilities, nursing homes, and all other similar facilities which provide their clients with individual beds and sleeping quarters, but in which all other care and services are communal and provided by facility employees, are specifically exempt from parkland dedication and system development fee requirements.

1. The required parkland shall be dedicated as a condition of approval for the following:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- a. Tentative plat for a subdivision or partition;
- b. Planned Development conceptual or detailed development plan;
- c. Design review for a multi-family development or manufactured home park; and
- d. Replat or amendment of any site plan for multi-family development or manufactured home park where dedication has not previously been made or where the density of the development involved will be increased.
- 2. Calculation of Required Dedication: The required parkland acreage to be dedicated is based on a calculation of the following formula rounded to the nearest 1/100 (0.00) of an acre:

Required parkland dedication (acres) = (proposed units) x (persons/unit) x 0.0043 (per person park land dedication factor)

a. Population Formula: The following table shall be used to determine the number of persons per unit to be used in calculating required parkland dedication:

Type of Unit	Total Persons Per Unit
Single family residential	3.0
Standard multi-family unit	2.0
Manufactured dwelling park	2.0
Congregate multi-family unit	1.5

Persons per unit, age distribution, and local conditions change with time. The specific formula for the dedication of land will, therefore, be subject to periodic review and amendment.

b. Per Person Parkland Dedication Factor: The total parkland dedication requirement shall be 0.0043 of an acre per person based on the adopted standard of 4.3 acres of land per one thousand of ultimate population per the Parks Master Plan¹. This standard represents the citywide land-topopulation ratio for city parks, and may be adjusted periodically through amendments to the Parks Master Plan.

RESPONSE: The applicant proposes 32 single-family detached dwellings with this subdivision request. Based upon the calculations adopted by the City and specified within Section 17.86.10, the required dedication area is 0.41 acres of public parkland (32 proposed units x 3 persons per unit x .0043=0.4128 rounded to the nearest 1/100 of an acre = 0.41 acres to be dedicated).

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

¹ Parks Master Plan, Implementation Plan section, Pages 4 and 5 indicate a required park acreage total of 64.5 acres. This number, divided by population (2015) of 15,000 equates to 4.3 acres per 1000 population or 0.0043 per person.

17.86.40 CASH IN LIEU OF DEDICATION

At the city's discretion only, the city may accept payment of a fee in lieu of land dedication. The city may require payment in lieu of land when the park land to be dedicated is less than 3 acres. A payment in lieu of land dedication is separate from Park Systems Development Charges, and is not eligible for a credit of Park Systems Development Charges. The amount of the fee in lieu of land dedication (in dollars per acre) shall be set by City Council Resolution, and it shall be based on the typical market value of developed property (finished lots) in Sandy net of related development costs.

- 1. The following factors shall be used in the choice of whether to accept land or cash in lieu:
 - a. The topography, geology, access to, parcel size, and location of land in the development available for dedication;
 - b. Potential adverse/beneficial effects on environmentally sensitive areas;
 - c. Compatibility with the Parks Master Plan, Public Facilities element of the Comprehensive Plan, and the City of Sandy Capital Improvements Program in effect at the time of dedication;
 - d. Availability of previously acquired property; and
 - e. The feasibility of dedication.
- 2. A. Cash in lieu of parkland dedication shall be paid prior to approval of the final plat or as specified below:
 - a. 50 percent of the payment shall be paid prior to final plat approval, and
 - b. The remaining 50 percent of the payment pro-rated equally among the lots, plus an administrative surcharge, shall be paid as specified by City Council Resolution.

RESPONSE: Based on the calculations specified in Section 17.86.10, the applicant is responsible for dedicating 0.41 acres of public parkland based on 32 dwelling units. No parkland is identified on the subject property in the Parks Master Plan. For these reasons, the applicant shall pay a fee in lieu for the required parkland dedication per the adopted Fee Resolution. Per Resolution 2013-14, the required fee in lieu amount is \$241,000 per acre if the entire amount is paid prior to final plat approval. Therefore, based on the current Fee Resolution, the applicant would be required to pay a fee in lieu of dedication for a total of \$98,810 (0.41 acres of land to be dedicated x \$241,000). Alternatively, Ordinance 2013-03 allows the applicant to pay a minimum of 50 percent of the fee to receive final plat approval with the remaining balance to be paid as a proportionate amount with each building permit. If a portion of the fee is deferred, Resolution 2013-14 specifies a per acre fee of \$265,000. Currently, the Fee Resolution requires payment of \$108,650 if a portion of the fee is deferred, a minimum of 50 percent (\$54,325) paid prior to final plat approval and the remaining 50 percent (\$54,325) divided between the 32 lots (\$1,697.66/lot). If the number of lots is modified from 32, the parks fee in lieu will need to be recalculated.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.86.50 MINIMUM STANDARDS FOR OPEN SPACE DEDICATION

The applicant through a subdivision or design review process may propose the designation and protection of open space areas as part of that process. This open space will not, however, be counted toward the parkland dedication requirement of Sections 17.86.10 through 17.86.40.

4. Open space areas shall be maintained so that the use and enjoyment thereof is not diminished or destroyed. Open space areas may be owned, preserved, and maintained by any of the following mechanisms or combinations thereof:

a. Dedication to the City of Sandy or an appropriate public agency approved by the City, if there is a public agency willing to accept the dedication. Prior to acceptance of proposed open space, the City may require the developer to submit a Phase I Environmental Site Assessment completed by a qualified professional according to American Society of Testing and Materials (ASTM) standards (ASTM E 1527). The results of this study shall indicate a clean environmental record.

b. Common ownership by a homeowner's association that assumes full responsibility for its maintenance;

c. Dedication of development rights to an appropriate public agency with ownership remaining with the developer or homeowner's association. Maintenance responsibility will remain with the property owner; and/or

d. Deed-restricted private ownership preventing development and/or subsequent subdivision and providing for maintenance responsibilities.

RESPONSE: The applicant is not proposing any dedication of open space. However, the Parks Board recommends that the FSH analysis tract be dedicated to the City. The Public Works Director expressed some maintenance concerns about the City taking on ownership of the tract without knowing the condition that it is in and recommended that, prior to dedicating the FSH tract to the City, the applicant would need to: 1) submit a Phase I environmental assessment and a native vegetation report for the FSH area to be dedicated in the tract, 2) remove all invasive plants and re-plant any areas of exposed soil with native plants, and 3) submit a maintenance bond for two full growing seasons (May 1 -October 31) to ensure 80 percent plant survival or replacement and that the tract remains in good condition. Staff supported these recommendations in the February 25, 2019 staff report; however, the applicant's representative indicated that the applicant did not want to incur the expense of having to restore the FSH tract and conduct a Phase I environmental assessment. If the FSH tract is not dedicated to the City, it will not be necessary to perform a Phase I environmental assessment. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. Due to the site layout, Lots 19-21 will end up less than 7,500 square feet once the FSH area is put into a separate tract. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot area and allowing the rear setback to be reduced to 15 feet. This will help avoid confusion over property ownership, fence lines, and unnecessary or illegal removal of riparian vegetation and vegetation on steep slopes.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

5. In the event that any private owner of open space fails to maintain it according to the standards of this Code, the City of Sandy, following reasonable notice, may demand that the deficiency of maintenance be corrected, and may enter the open space for maintenance purposes. All costs thereby incurred by the City shall be charged to those persons having the primary responsibility for maintenance of the open space.

RESPONSE: The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants. In the event that the HOA fails to maintain the FSH tract according to the standards of the Sandy Municipal Code, the City of Sandy, following reasonable notice, may demand that the deficiency of maintenance be corrected, and may enter the FSH tract area for maintenance purposes. All costs thereby incurred by the City shall be charged to the HOA.

CHAPTER 17.92 – LANDSCAPING AND SCREENING

17.92.10 GENERAL PROVISIONS

- C. Significant plant and tree specimens should be preserved to the greatest extent practicable and integrated into the design of a development. Trees of 25-inches or greater circumference measured at a height of 4-1/2 ft. above grade are considered significant. Plants to be saved and methods of protection shall be indicated on the detailed planting plan submitted for approval. Existing trees may be considered preserved if no cutting, filling, or compaction of the soil takes place between the trunk of the tree and the area 5-ft. outside the tree's drip line. Trees to be retained shall be protected from damage during construction by a construction fence located 5 ft. outside the dripline.
- L. All landscaping shall be continually maintained, including necessary watering, weeding, pruning and replacing.

RESPONSE: Per Section 17.92.10 (C), trees over 25-inches circumference measured at a height of 4-¹/₂ feet above grade are considered significant and should be preserved to the greatest extent practicable and integrated into the design of a development. A 25-inch circumference tree measured at 4-¹/₂ feet above grade has roughly an eight-inch diameter at breast height (DBH). Based on the Planning Commission interpretation from February 25, 2019, Subsection 17.92.10(C) does not apply to subdivisions. Tree protection fencing and tree retention will be discussed in more detail under Chapter 17.102 in this document. Per Section 17.92.10(L), all landscaping shall be continually maintained, including necessary watering, weeding, pruning, and replacing.

Note: the applicant is requesting a code interpretation to determine if Section 17.92.10(C) applies to subdivisions.

17.92.30 REQUIRED TREE PLANTINGS

Planting of trees is required for all parking lots with 4 or more parking spaces, public street frontages, and along private drives more than 150 feet long. Trees shall be planted outside the street right-of-way except where there is a designated planting strip or City adopted street tree plan.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

The City maintains a list of appropriate trees for street tree and parking lot planting situations. Selection of species should be made from the city-approved list. Alternate selections may be approved by the Director following written request. The type of tree used shall determine frequency of trees in planting areas. Trees in parking areas shall be dispersed throughout the lot to provide a canopy for shade and visual relief.

Area/Type of Planting	Canopy	Spacing
Street Tree	Medium	30 ft. on center
Street Tree	Large	50 ft. on center

Trees may not be planted:

- Within 5 ft. of permanent hard surface paving or walkways, unless specific species, special planting techniques and specifications approved by the Director are used.
- Unless approved otherwise by the City Engineer:
 - * Within 10 ft. of fire hydrants and utility poles
 - * Within 20 ft. of street light standards
 - * Within 5 ft. from an existing curb face
 - * Within 10 ft. of a public sanitary sewer, storm drainage or water line
- Where the Director determines the trees may be a hazard to the public interest or general welfare.
- Trees shall be pruned to provide a minimum clearance of 8 ft. above sidewalks and 12 ft. above street and roadway surfaces.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires trees spaced 30 feet on center along street frontages. The submitted Street Tree Plan (Exhibit C, Sheet 6) identifies street trees along all of the proposed streets. The proposed plan details 84 street trees. Two (2) of the street trees are proposed to be located in driveways in front of Lots 31 and 32; however, there appears to be sufficient space to relocate at least one of the trees outside of the driveway area. The applicant shall update the plan set to remove street trees from the driveways in front of Lots 31 and 32; at least one of the street trees shall be relocated outside of the driveway area.

The applicant is proposing to mass grade the buildable portion of the site. This will remove top soil and heavily compact the soil. In order to maximize the success of the required street trees, the applicant shall aerate the planter strips to a depth of 3 feet prior to planting street trees. The applicant shall either aerate the planter strip soil at the subdivision stage and install fencing around the planter strips to protect the soil from compaction or shall aerate the soil at the individual home construction phase. The applicant shall call for an inspection with the City after aerating the soil and before planting the street trees.

The Preliminary Street and Utility Plan (Exhibit C, Sheet 5) shows the sidewalk and planter strip improvements on the south side of Newton Street across from Lot 9 ending before the development site boundary. Section 17.84.50(E) requires extension of street improvements "to the edge of adjacent properties." With extension of the sidewalk and planter strip to the southern property boundary, the proposal could accommodate a third street tree on the south side of Newton Street. **The applicant shall extend the street improvements on Newton Street across from Lot 9 to the**

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW *dedication for staff review and approval.* The proposed layout would likely accommodate an additional two (2) street trees on the east end of Newton Street. Per the Public Works Director (Exhibit O), the applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. If the plans change in a way that affects the number of street trees (e.g., driveway locations), the applicant shall submit an updated street tree plan for staff review and approval. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted, staked, and the planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed after one growing season (or a maximum of 1 year).

17.92.40 IRRIGATION

Landscaping shall be irrigated, either with a manual or automatic system, to sustain viable plant life.

RESPONSE: Section 17.92.40 requires that all landscaping shall be irrigated, either with a manual or automatic system. As required by Section 17.92.140, the developer and lot owners shall be required to maintain all vegetation planted in the development for two years from the date of completion, and shall replace any dead or dying plants during that period.

17.92.50 TYPES AND SIZES OF PLANT MATERIALS

- A. At least 75% of the required landscaping area shall be planted with a suitable combination of trees, shrubs, or evergreen ground cover except as otherwise authorized by Chapter 17.92.10 F.
- B. Plant Materials. Use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged where possible.
- C. Trees shall be species having an average mature spread of crown greater than 15 feet and having trunks which can be maintained in a clear condition with over 5 feet of clear wood (without branches). Trees having a mature spread of crown less than 15 feet may be substituted by grouping the same so as to create the equivalent of a 15-foot crown spread.
- D. Deciduous trees shall be balled and burlapped, be a minimum of 7 feet in overall height or 1 ¹/₂ inches in caliper measured 6 inches above the ground, immediately after planting. Bare root trees will be acceptable to plant during their dormant season.
- E. Coniferous trees shall be a minimum five feet in height above ground at time of planting.
- F. Shrubs shall be a minimum of 1 gallon in size or 2 feet in height when measured immediately after planting.
- G. Hedges, where required to screen and buffer off-street parking from adjoining properties shall be planted with an evergreen species maintained so as to form a continuous, solid visual screen within 2 years after planting.
- H. Vines for screening purposes shall be a minimum of 1 gallon in size or 30 inches in height immediate after planting and may be used in conjunction with fences, screens, or walls to meet physical barrier requirements as specified.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- I. Groundcovers shall be fully rooted and shall be well branched or leafed. If used in lieu of turf in whole or in part, ground covers shall be planted in such a manner as to provide complete coverage in one year.
- J. Turf areas shall be planted in species normally grown as permanent lawns in western Oregon. Either sod or seed are acceptable. Acceptable varieties include improved perennial ryes and fescues used within the local landscape industry.
- K. Landscaped areas may include architectural features or artificial ground covers such as sculptures, benches, masonry or stone walls, fences, rock groupings, bark dust, decorative hard paving and gravel areas, interspersed with planted areas. The exposed area developed with such features shall not exceed 25% of the required landscaped area. Artificial plants are prohibited in any required landscape area.

RESPONSE: Street trees are typically required to be a minimum caliper of 1.5-inches measured 6 inches from grade. In the past, staff has recommended a larger caliper street tree (2-inches) for subdivision proposals where a substantial number of existing trees were proposed for removal. However, staff has received feedback that fewer species are available at the 2-inch caliper size. To provide species diversity, staff recommends requiring 1.5-inch caliper street trees. All street trees shall be a minimum of 1.5-inches in caliper measured 6 inches above the ground and shall be planted per the City of Sandy standard planting detail. The applicant proposes six distinct street tree species with one tree species per block face. Staff would like to see more diversity in street tree species in general and within each block. The applicant shall update the plan set to detail a minimum of two (2) different tree species per block face for staff review and approval. The proposed street tree list includes paperbark maple and Patmore ash. Due to concerns with Asian Longhorn Beetle and Emerald Ash Borer, staff do not recommend planting new maples or ashes at this time. The applicant shall update the proposed street tree list with at least two additional species from the City of Sandy Street Tree List that aren't maples or ashes. Staff recommends two (or more) species from the following list: cascara (Rhamnus purshiana), kobus magnolia (Magnolia kobus), Skyrocket oak (Ouercus robur 'Skyrocket'), Japanese stewartia (Stewartia pseudocamellia), Japanese snowbell (Styrax japonica), or eastern redbud (Cercis canadensis). Providing species diversity adds visual interest and provides additional resistance against disease in conformance with City of Sandy Comprehensive Goal 5 Policy 28. The applicant also proposed Halka ginkgo. Technically, the only ginkgo on the City of Sandy street tree list is Saratoga ginkgo (Ginkgo biloba 'Saratoga'); however, staff recommends that Halka ginkgo be approved as a street tree with this application.

17.92.60 REVEGETATION IN UNLANDSCAPED OR NATURAL AREAS

- A. Areas where natural vegetation has been removed or damaged through grading or construction activity in areas not affected by the landscaping requirements and that are not to be occupied by structures or other improvements shall be replanted.
- B. Plant material shall be watered at intervals sufficient to assure survival and growth.
- C. The use of native plant materials or plants acclimatized to the Pacific Northwest is encouraged to reduce irrigation and maintenance demands.

RESPONSE: The applicant did not submit any plans for re-vegetation of areas damaged through grading/construction, although most of the areas affected by grading will be improved. The applicant shall maintain all unlandscaped and/or revegetated areas for a period of two years following the date of recording of the final plat associated with those improvements.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.92.130 PERFORMNCE BOND

If weather conditions or other circumstances beyond the control of the developer or owner make completion of the landscaping impossible prior to desired occupancy, an extension of up to 6 months may be applied for by posting "security" equal to 120% of the cost of the landscaping, assuring installation within 6 months. "Security" may consist of a performance bond payable to the city, cash, certified check, time certificates of deposit, assignment of a saving account, letter of credit, or other such assurance of access to funds necessary for completion as shall meet the approval of the City Attorney. Upon acceptance of the security, the developer or owner may be allowed occupancy for a period of up to 180 days. If the installation of the landscaping improvement is not completed within 180 days, the City shall have access to the security to complete the installation and/or revoke occupancy. Upon completion of the installation by the city, any portion of the remaining security shall be assessed against the property and the City shall thereupon have a valid lien against the property, which will come due, and payable.

RESPONSE: The applicant has the option to defer the installation of street trees and/or landscaping for weather-related reasons. Staff recommends the applicant utilize this option rather than install trees and landscaping during the dry summer months. If the applicant chooses to postpone street tree and/or landscaping installation, the applicant shall post a performance bond equal to 120 percent of the cost of the street trees/landscaping, assuring installation within 6 months. The cost of street trees shall be based on the street tree plan and at least \$500 per tree fee, including a three-year maintenance and warranty period. The cost of landscaping shall be based on the average of three estimates from three landscaping contractors; the estimates shall include as separate items all materials, labor, and other costs of the required action, including a two-year maintenance and warranty period.

CHAPTER 17.98 - PARKING, LOADING, AND ACCESS REQUIREMENTS

17.98.20 OFF-STREET PARKING REQUIREMENTS

RESPONSE: Section 17.98.20 requires two off-street parking spaces per single family detached dwelling unit. The 32 dwelling units proposed in this subdivision requires 64 off-street parking spaces. Each lot will have a driveway and the ability to construct a double car garage.

17.98.50 SETBACKS

- A. Parking areas, which abut a residential zoning district, shall meet the setback of the most restrictive adjoining residential zoning district.
- B. Required parking shall not be located in a required front or side yard setback area abutting a public street except in industrial districts. For single family and two-family dwellings, required off-street parking may be located in a driveway.
- C. Parking areas shall be setback from a lot line adjoining a street the same distance as the required building setbacks. Regardless of other provisions, a minimum setback of 5 feet shall be provided along the property fronting on a public street. The setback area shall be landscaped as provided in this code.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: Garages are required to be at least 22 feet setback from the front property line to meet setback requirements in the SFR zoning district. The Preliminary Plat (Exhibit C, Sheet 1) details a typical 22 foot garage setback.

17.98.60 DESIGN, SIZE AND ACCESS

All off-street parking facilities, vehicular maneuvering areas, driveways, loading facilities, accessways, and private streets shall conform to the standards set forth in this section.

- A. Parking Lot Design. All areas for required parking and maneuvering of vehicles shall have a durable hard surface such as concrete or asphalt.
- B. Size of Space.
 - 1. A standard parking space shall be 9 feet by 18 feet.
 - 2. A compact parking space shall be 8 feet by 16 feet.
 - 3. Handicapped parking spaces shall be 13 feet by 18 feet. Accessible parking shall be provided for all uses in compliance with the requirements of the State of Oregon (ORS 447.233) and the Americans with Disabilities Act.
 - 4. Parallel parking spaces shall be a length of 22 feet. No more than 35 percent of the parking stalls shall be compact spaces

RESPONSE: The applicant shall comply with the parking standards in Section 17.98.60. The parking areas in front of the proposed garages for all lots need to be at least 10 feet in width by 20 feet in length. Driveways for single family homes are required to be at least 10 feet wide as detailed in Section 17.98.100 below. The garages shall be adequate depth to park a vehicle and the on-street parking spaces shall be at least 22 feet in length.

17.98.80 ACCESS TO ARTERIAL AND COLLECTOR STREETS

- A. Location and design of all accesses to and/or from arterials and collectors (as designated in the Transportation System Plan) are subject to review and approval by the City Engineer. Where practical, access from a lower functional order street may be required. Accesses to arterials or collectors shall be located a minimum of 150 ft. from any other access or street intersection. Exceptions may be granted by the City Engineer. Evaluations of exceptions shall consider posted speed of the street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.
- B. No development site shall be allowed more than one access point to any arterial or collector street (as designated in the Transportation System Plan) except as approved by the City Engineer. Evaluations of exceptions shall consider posted speed of street on which access is proposed, constraints due to lot patterns, and effects on safety and capacity of the adjacent public street, bicycle and pedestrian facilities.
- C. When developed property is to be expanded or altered in a manner that significantly affects on-site parking or circulation, both existing and proposed accesses shall be reviewed under the standards in A and B above. As a part of an expansion or alteration approval, the City may require relocation and/or reconstruction of existing accesses not meeting those standards.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

RESPONSE: Jacoby Heights Subdivision is proposed to create two intersections with Jacoby Road, a designated collector street: Woodstock Street and Newton Street. All lots are required to have driveway access from the internal local street network with a vehicle non-access reserve strip (VNAR) detailed on the face of plat along Jacoby Road.

17.98.100 DRIVEWAYS

- A. A driveway to an off-street parking area shall be improved from the public roadway to the parking area a minimum width of 20 feet for a two-way drive or 12 feet for a one-way drive but in either case not less than the full width of the standard approach for the first 20 feet of the driveway.
- B. A driveway for a single-family dwelling shall have a minimum width of 10 feet.
- C. A driveway for a two-family dwelling shall have a minimum width of 20 feet. A driveway approach must be constructed in accordance with applicable city standards and the entire driveway must be paved with asphalt or concrete.
- D. Driveways, aisles, turnaround areas and ramps shall have a minimum vertical clearance of twelve feet for their entire length and width but such clearance may be reduced in parking structures.
- E. No driveway shall traverse a slope in excess of 15 percent at any point along the driveway length.
- F. The location and design of the driveway shall provide for unobstructed sight per the vision clearance requirements. Requests for exceptions to these requirements will be evaluated by the City Engineer considering the physical limitations of the lot and safety impacts to vehicular, bicycle, and pedestrian traffic.

RESPONSE: The minimum driveway width for a single-family dwelling is 10 feet. The Public Works driveway approach standard detail specifies a maximum of 24 feet wide for a residential driveway approach. The Preliminary Parking Plan (Exhibit C, Sheet 7) details driveway curb cuts for all lots. With the exception of a few lots on the cul-de-sac, all driveways are detailed at approximately 20 feet. The applicant is proposing a shared driveway between Lot 29 and Tract A at approximately 36 feet and a shared driveway between Lot 27 and 28 at approximately 26 feet. In order to maximize street tree planting areas and prevent conflicts with streetlights, water meters and dry utility facilities, shared driveway approaches may be required in cul-de-sac bulbs. Per the Public Works Director (Exhibit O), Lot 28 and 29 shall share a driveway approach maximum 24 feet in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches. The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of Lot 29 to eliminate remnant of Tract A between the access drive and Lot 29. The driveway approach for Tract A shall be a maximum of 12 feet in width. Per the City Engineer (Exhibit L), the access to the detention pond shall be paved or all-weather surface.

17.98.130 PAVING

A. Parking areas, driveways, aisles and turnarounds shall be paved with concrete, asphalt or comparable surfacing, constructed to city standards for off-street vehicle areas.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

B. Gravel surfacing shall be permitted only for areas designated for non-motorized trailer or equipment storage, propane or electrically powered vehicles, or storage of tracked vehicles.

RESPONSE: As required by Section 17.98.130, all parking, driveway and maneuvering areas shall be constructed of asphalt, concrete, or other approved material.

17.98.140 DRAINAGE

Parking areas, aisles and turnarounds shall have adequate provisions made for the on-site collection of drainage waters to eliminate sheet flow of such waters onto sidewalks, public rights-of-way and abutting private property.

RESPONSE: Section 17.98.140 contains requirements for drainage. Other sections of this order detail the stormwater requirements.

17.98.200 RESIDENTIAL ON-STREET PARKING REQUIREMENTS

A. Residential On-Street Parking Requirements. Residential on-street parking shall conform to the following standards:

- 1. In addition to required off-street parking, all new residential planned developments, subdivisions and partitions shall provide one (1) on-street parking space within 200 feet of each dwelling except as provided in Section 17.98.200(A)(6) below.
- 2. The location of residential on-street parking shall be reviewed for compliance with this section through submittal of a Residential Parking Analysis Plan as required in Section 17.98.10(M).
- 3. Residential on-street parking shall not obstruct required clear vision areas and shall not violate any local or state laws.
- 4. Parallel residential on-street parking spaces shall be 22 feet minimum in length.
- 5. Residential on-street parking shall be measured along the curb from the outside edge of a driveway wing or curb cut. Parking spaces must be set back a minimum of 15 feet from an intersection and may not be located within 10 feet of a fire hydrant.
- 6. Portions of residential on-street parking required by this section may be provided in parking courts that are interspersed throughout a development when the following standards are met:
 - a. No more than eight (8) parking spaces shall be provided in a parking court;
 - b. Parking spaces within a parking court shall be nine (9) feet wide and 18 feet in depth;
 - c. Notwithstanding Section 17.98.70, vehicles parked in a parking court are permitted to back onto the public right-of-way from the parking court;
 - d. A parking court shall be located within 200 feet of the dwellings requiring parking in accordance with the requirements of Section 17.98.10(M);
 - e. No more than two (2) parking courts shall be provided within a block, with only one (1) parking court provided along a block face;
 - f. A parking court shall be paved in compliance with the standards of this chapter and the latest adopted grading and drainage standards;

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- g. If a parking court is adjacent to a public right-of-way, it shall be publicly owned and maintained;
- h. If a parking court is adjacent to a private drive, it shall be privately owned and maintained. For each parking court there shall be a legal recorded document which includes:
 - i. A legal description of the parking court;
 - ii. Ownership of the parking court;
 - iii. Use rights; and
 - iv. A maintenance agreement and the allocation and/or method of determining liability for maintenance of the parking court;
- i. A parking court shall be used solely for the parking of operable passenger vehicles.

RESPONSE: The Preliminary Parking Plan (Exhibit C, Sheet 7) identifies a total of 62 on-street parking spaces with at least one (1) on-street parking space within 200 feet of each of the 32 lots. The proposed parking plan shows 21 parking spaces on the south side of Newton Street, including parking spaces for Lots 1-4. These on-street parking spaces should not be counted toward meeting the on-street parking requirement since the south side of the street will eventually develop with driveways and/or streets that will eliminate these spaces or prevent the developer of the site to the south from counting these spaces toward their on-street parking requirement. Per the Public Works Director (Exhibit O), the applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval. No parking courts are proposed. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with Construction Plans. The applicant shall revise the Parking Analysis if required fire hydrants affect on-street parking spaces.

CHAPTER 17.100 - LAND DIVISION

Chapter 17.100 presents review procedures, design standards, and improvement requirements for land division. Section 17.100.60 is the subsection relevant to the division of subdivisions.

17.100.60 SUBDIVISIONS

Approval of a subdivision is required for a land division of 4 or more parcels in a calendar year. A twostep procedure is required for subdivision approval: (1) tentative plat review and approval; and (2) final plat review and approval.

RESPONSE: Submittal of preliminary utility plans is solely to satisfy the requirements of Section 17.100.60. Preliminary plat approval does not connote utility or public improvement plan approval which will be reviewed and approved separately upon submittal of public improvement construction plans.

- E. <u>Approval Criteria</u>. The Director or Planning Commission shall review the tentative plat for the subdivision based on the classification procedure (Type II or III) set forth in Section 17.12 and the following approval criteria:
 - 1. The proposed subdivision is consistent with the density, setback and dimensional standards of the base zoning district, unless modified by a Planned Development approval.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx
RESPONSE: The applicant requests subdivision approval for a subdivision that is in compliance with most of the applicable development standards. The applicant requests one variance to increase block length beyond the maximum block length standard, one variance to allow houses to not face a transit street, and one variance to allow single-tiered lots. As a result, the application for the subdivision and variances is being processed concurrently through a Type III procedure. The proposal is consistent with density and other dimensional standards of the base zoning district.

- 2. The proposed subdivision is consistent with the design standards set forth in this chapter. *RESPONSE:* Consistency with design standards in this chapter are discussed under each subsection below. Conditions of approval can be adopted where necessary to bring the proposal into compliance with applicable standards.
- The proposed street pattern is connected and consistent with the Comprehensive Plan or official street plan for the City of Sandy.
 RESPONSE: The proposed subdivision includes the extension of Newton Street to the east of Jacoby Road and three new proposed local streets, including one cul-de-sac. The subject property was not included in the City's current Transportation System Plan (TSP) because it was
- 4. Adequate public facilities are available or can be provided to serve the proposed subdivision. *RESPONSE:* All public facilities including water, sanitary sewer and stormwater are available or will be constructed by the applicant to serve the Jacoby Heights Subdivision.

outside of the Urban Growth Boundary (UGB) when the TSP was adopted in 2011.

- 5. All proposed improvements meet City standards. *RESPONSE:* The detailed review of proposed improvements is contained in this report. Staff has identified a few aspects of the proposed subdivision improvements requiring additional information or modification by the applicant, but conditions of approval can be adopted to bring the proposal into compliance with all standards.
- 6. The phasing plan, if requested, can be carried out in a manner that meets the objectives of the above criteria and provides necessary public improvements for each phase as it develops. *RESPONSE: The applicant did not submit a phasing plan. This subdivision request is for a 32 lot subdivision in one development phase.*

17.100.70 LAND DIVISION DESIGN STANDARDS

All land divisions shall be in conformance with the requirements of the applicable base zoning district and this chapter, as well as with other applicable provisions of this Code. Modifications to these requirements may be accomplished through a Planned Development. The design standards in this section shall be used in conjunction with street design standards included in the City of Sandy Transportation System Plan and standards and construction specifications for public improvements as set forth in adopted Public Facilities Plans and the Sandy Municipal Code.

RESPONSE: The applicant requested the following three Type III variances:

• Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff supports the variance to front door orientation, the variance to allow a single-tiered lot configuration along Jacoby Road, and the variance to allow the block length to exceed 400 feet on the north side of Woodstock Street between Camden Court and the east property boundary. The tentative plat shall otherwise be designed to comply with all standards of the City of Sandy Development Code, Transportation System Plan, Facilities Master Plans and Sandy Municipal Code.

17.100.100 STREETS GENERALLY

No subdivision or partition shall be approved unless the development has frontage or approved access to an existing public street. In addition, all streets shall be graded and improved in conformance with the City's construction standards, approved by the City Engineer, in accordance with the construction plans.

A. <u>Street Connectivity Principle.</u> The pattern of streets established through land divisions should be connected to: (a) provide safe and convenient options for cars, bikes and pedestrians; (b) create a logical, recognizable pattern of circulation; and (c) spread traffic over many streets so that key streets (particularly U.S. 26) are not overburdened.

RESPONSE: The proposed development is moderately conducive to walking and biking while accommodating motor vehicles. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north.

- B. <u>Transportation Impact Studies.</u> Transportation impact studies may be required by the city engineer to assist the city to evaluate the impact of development proposals, determine reasonable and prudent transportation facility improvements and justify modifications to the design standards. Such studies will be prepared in accordance with the following:
 - 1. A proposal established with the scope of the transportation impact study shall be coordinated with, and agreed to, by the city engineer. The study requirements shall reflect the magnitude of the project in accordance with accepted transportation planning and engineering practices. A professional civil or traffic engineer registered in the State of Oregon shall prepare such studies.
 - 2. If the study identifies level-of-service conditions less than the minimum standards established in the Sandy Transportation System Plan, improvements and funding strategies mitigating the problem shall be considered as part of the land use decision for the proposal.

RESPONSE: The submitted Transportation Impact Analysis (Exhibit F) was completed by Lancaster Engineering and is dated June 15, 2018. The traffic assumptions are based on the 10th Edition Trip Generation handbook. The analysis is based on the construction of 32 single-family homes and the removal of one existing house, for a net increase of 31 units. The trip rates indicate that upon full occupancy the subdivision will generate about 23 trips during the morning peak hour and 31 trips during the evening peak hour, with a weekday total of 292 trips. The study found that adequate sight distances can be made available on Jacoby Road at the site access intersections of both Woodstock Street and Newton Street. The study also found that all study intersections are operating acceptably per City of Sandy and ODOT standards and are projected to continue to

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

operate acceptably upon build-out of the proposed development through year 2020. The Traffic Impact analysis concludes that no operational mitigation is necessary or recommended, no safety mitigation is recommended, no sight distance mitigations are necessary or recommended, and no turn lane or signal mitigations are recommended as part of this development. The City Transportation Engineer (Exhibit M) reviewed the Traffic Impact Analysis and concurred that the analysis demonstrates that the development can be accommodated without requiring mitigation to meet adopted operational standards.

C. <u>Topography and Arrangement.</u> All streets shall be properly related to special traffic generators such as industries, business districts, schools, and shopping centers and to the pattern of existing and proposed land uses.

RESPONSE: Considering the site's topography, the proposed street layout is acceptable given the topography and residential use of this site, and the topography and use of adjacent properties.

D. <u>Street Spacing</u>. Street layout shall generally use a rectangular grid pattern with modifications as appropriate to adapt to topography or natural conditions.

RESPONSE: The applicant proposes a rectangular pattern of streets with an offset grid. Due to topography and the location of Tickle Creek, it does not make sense to extend the north-south streets (Camden Court and Middlebury Avenue) to the north. However, future development to the south will be required to align with the proposed intersections in order to maintain a rectangular grid pattern and maximize pedestrian, bicycle, and vehicular connectivity. An offset grid with "T" intersections is not warranted due to the lack of through traffic.

E. <u>Future Street Plan.</u> Future street plans are conceptual plans, street extensions and connections on acreage adjacent to land divisions. They assure access for future development and promote a logical, connected pattern of streets. It is in the interest of the city to promote a logical, connected pattern of streets. All applications for land divisions shall provide a future street plan that shows the pattern of existing and proposed future streets within the boundaries of the proposed land divisions, proposed connections to abutting properties, and extension of streets to adjacent parcels within a 400 foot radius of the study area where development may practically occur.

RESPONSE: The proposed local street plan has been designed to facilitate the traffic needs of this development while ensuring there are no intersection conflicts with future development. Per the City Engineer (Exhibit L), the applicant shall provide a profile design for a minimum of 200 feet for all future street extensions beyond the project boundary to ensure future street grades can be met.

- F. <u>Connections.</u> Except as permitted under Exemptions, all streets, alleys and pedestrian walkways shall connect to other streets within the development and to existing and planned streets outside the development and to undeveloped properties which have no future street plan. Streets shall terminate at other streets or at parks, schools or other public land within a neighborhood.
 - 1. Where practicable, local roads shall align and connect with other roads when crossing collectors and arterials.
 - 2. Proposed streets or street extensions shall be located to provide direct access to existing or planned transit stops, and existing or planned neighborhood activity centers, such as schools, shopping areas and parks.
 - 3. Exemptions.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- a. A future street plan is not required for partitions of residentially zoned land when none of the parcels may be redivided under existing minimum density standards.
- b. Standards for street connections do not apply to freeways and other highways with full access control.
- c. When street connection standards are inconsistent with an adopted street spacing standard for arterials or collectors, a right turn in/right turn out only design including median control may be approved. Where compliance with the standards would result in unacceptable sight distances, an accessway may be approved in place of a street connection.

RESPONSE: The proposed design extends Newton Street east across Jacoby Road into the site. All proposed streets will allow connection with future development to the south and east, with the exception of Camden Court, which is a cul-de-sac. Due to topography and the location of Tickle Creek, it is not practicable to connect Camden Court to the north.

17.100.110 STREET STANDARDS AND CLASSIFICATION

Street standards are illustrated in the figures included at the end of this chapter. Functional definitions of each street type are described in the Transportation System Plan as summarized below.

- A. Major arterials are designed to carry high volumes of through traffic, mixed with some unavoidable local traffic, through or around the city. Major arterials should generally be spaced at 1-mile intervals.
- B. Minor arterials are designed to collect and distribute traffic from major and minor arterials to neighborhood collectors and local streets, or directly to traffic destinations. Minor arterials should generally be spaced at 1-mile intervals.
- C. Residential minor arterials are a hybrid between minor arterial and collector type streets that allow for moderate to high traffic volumes on streets where over 90% of the fronting lots are residential.
- D. Collector streets are designed to collect and distribute traffic from higher type arterial streets to local streets or directly to traffic destinations. Collector streets should generally be spaced at 1/2-mile intervals.
- E. Local streets are designed to provide direct access to abutting property and connect to collector streets. A general spacing of 8-10 local streets per mile is recommended.
- F. Cul-de-sacs and dead end streets are discouraged. If deemed necessary, cul-de-sacs shall be as short as possible and shall not exceed 400 feet in length.
- G. Public access lanes are designed to provide primary access to a limited number of dwellings when the construction of a local street is unnecessary.
- H. Alleys are designed to provide access to multiple dwellings in areas where lot frontages are narrow and driveway spacing requirements cannot be met.

RESPONSE: Jacoby Road is an existing collector street. All proposed streets in the subdivision are local streets, including the extension of the existing Newton Street, which is proposed to extend east of Jacoby Road into the site. Camden Court is a proposed cul-de-sac and does not exceed 400 feet. Per the City Engineer (Exhibit L), the plans don't clearly indicate the radius for the proposed cul-de-sac. In the February 25, 2019 staff report, the City Engineer and Public Works Director recommended that the minimum paved cul-de-sac radius be 48 feet as per the Oregon Fire Code and Metro Code Committee. The applicant's representative (Exhibit W) indicated the applicant wants to provide a smaller cul-de-sac and cited the cul-de-sac requirements for manufactured home parks as a basis for allowing a smaller cul-de-sac radius. They also citied other cul-de-sacs in Sandy that

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

don't meet the minimum radius requirement. The City Engineer and Public Works Director (Exhibit X) stated: "The fact that a cul-de-sac of this size was approved on another development does not allow staff to ignore the Fire Code standard on this site. The cul-de-sac bulb dimensions are not solely for emergency vehicle access. The larger radius allows enough room for on-street parking, garbage pickup, street sweeping and snow plowing. Even if lots 26-30 were constructed with residential fire protection systems I would not recommend reducing the cul-de-sac bulb radius." After further discussion, the Public Works Director determined that a minimum 42 foot paved cul-de-sac radius would suffice if the Fire Department is willing to sign off on a 42 foot radius cul-de-sac. The minimum paved cul-de-sac radius shall be 42 feet; a review by the Fire Department to confirm cul-de-sac size compliance shall be required prior to approval of the construction plans.

17.100.120 BLOCKS AND ACCESSWAYS

- A. <u>Blocks.</u> Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features.
- B. <u>Residential Blocks</u>. Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance.

RESPONSE: The applicant is requesting variances to Sections 17.100.120(A) and (B). The proposal includes one lot (Lot 30) that will be a single-tiered lot and one block (north side of Woodstock Street between Camden Court and the eastern property boundary) that exceeds 400 feet.

D. <u>Pedestrian and Bicycle Access Way Requirements.</u> In any block in a residential or commercial district over 600 feet in length, a pedestrian and bicycle accessway with a minimum improved surface of 10 feet within a 15-foot right-of-way or tract shall be provided through the middle of the block. To enhance public convenience and mobility, such accessways may be required to connect to cul-de-sacs, or between streets and other public or semipublic lands or through greenway systems.

RESPONSE: None of the proposed blocks exceed 600 feet in length. At the pre-application meeting, staff informed the applicant that staff would like to see a connection from the proposed subdivision to the future Tickle Creek Trail. Staff recommended the trail connection extend north from the intersection of Middlebury Avenue and Woodstock Street. The submitted Future Street and Pedestrian Trail Plan (Exhibit C, Sheet 8) details the future pedestrian trail connection to the future Tickle Creek Trail on the adjacent property to the east (tax lot 800). Staff brought this proposal to the January 30, 2019 Parks and Trails Advisory Board meeting. The Parks and Trails Advisory Board also recommends that the connector trail be constructed as part of this development. While the Park Facilities Plan (Exhibit Y) depicts the future Tickle Creek Trail located along Tickle Creek in the northeast corner of the subject property, the Park Facilities Plan doesn't go as far as identifying specific trail connections to the proposed future Tickle Creek Trail alignment. Without specific trail connectors identified explicitly on the plan, staff cannot compel the applicant to include a trail connecting to the future Tickle Creek Trail. However, staff and the Parks & Trails Advisory Board recommend the applicant include a connector trail on the subject property that provides access to the future Tickle Creek Trail.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.100.130 EASEMENTS

A minimum eight (8) foot public utility easement shall be required along property lines abutting a rightof-way for all lots within a partition or subdivision. Where a partition or subdivision is traversed by a watercourse, drainage way, channel or stream, the land division shall provide a stormwater easement or drainage right-of-way conforming substantially with the lines of such watercourse, and such further width as determined needed for water quality and quantity protection.

RESPONSE: The Preliminary Plat (Exhibit C, Sheet 1) details an 8 foot wide public utility easement along all street frontages. The plat shall detail all proposed easements as detailed in Section 17.84.90 above.

17.100.180 INTERSECTIONS

- A. <u>Intersections.</u> Streets shall be laid out so as to intersect as nearly as possible at right angles. A proposed intersection of two new streets at an angle of less than 75 degrees shall not be acceptable. No more than two streets shall intersect at any one point unless specifically approved by the City Engineer. The city engineer may require left turn lanes, signals, special crosswalks, curb extensions and other intersection design elements justified by a traffic study or necessary to comply with the Development Code.
- B. <u>Curve Radius.</u> All local and neighborhood collector streets shall have a minimum curve radius (at intersections of rights-of-way) of 20 feet, unless otherwise approved by the City Engineer. When a local or neighborhood collector enters on to a collector or arterial street, the curve radius shall be a minimum of 30 feet, unless otherwise approved by the City Engineer.

RESPONSE: The proposed intersections are all right angles and meet the required minimum spacing standard of 150 feet as required in Section 17.84.50(C)(2).

17.100.190 STREET SIGNS

RESPONSE: The applicant will be required to pay for the cost of all street signs and the City will install the street signs.

17.100.210 STREET LIGHTING

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined insufficient. The locations of street light fixtures shall be reviewed in detail with construction plans.

17.100.220 LOT DESIGN

- A. The lot arrangement shall be such that there will be no foreseeable difficulties, for reason of topography or other conditions, in securing building permits to build on all lots in compliance with the Development Code.
- B. The lot dimensions shall comply with the minimum standards of the Development Code. When lots are more than double the minimum lot size required for the zoning district, the subdivider may be

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

required to arrange such lots to allow further subdivision and the opening of future streets to serve such potential lots.

- C. The lot or parcel width at the front building line shall meet the requirements of the Development Code and shall abut a public street other than an alley for a width of at least 20 feet. A street frontage of not less than 15 feet is acceptable in the case of a flag lot division resulting from the division of an unusually deep land parcel which is of a size to warrant division into not more than two parcels.
- D. Double frontage lots shall be avoided except where necessary to provide separation of residential developments from arterial streets or to overcome specific disadvantages of topography or orientation.
- E. Lots shall avoid deriving access from major or minor arterials. When driveway access from major or minor arterials may be necessary for several adjoining lots, the Director or the Planning Commission may require that such lots be served by a common access drive in order to limit possible traffic hazards on such streets. Where possible, driveways should be designed and arranged to avoid requiring vehicles to back into traffic on minor or major arterials.

RESPONSE: The Single Family Residential (SFR) zoning district requires lots at least 7,500 square feet in area. The proposed lots range in size from 7,500 square feet to 21,183 square feet. All homes are required to comply with setback standards and maximum building height limitations as required in Chapter 17.34. Lot 30 is a double frontage lots that will take access from Camden Court but also abut Jacoby Road. Lots 19-21 and 27 contain more than double the minimum lot size; however, a large portion of each lot's area is within the FSH and, therefore, cannot be divided into a separate buildable lot. To avoid potential conflict between private property owners and the FSH overlay area in the future, the applicant shall designate the FSH analysis area as a separate tract, which shall be owned and maintained by an HOA. No lots are proposed to be accessed from a major or minor arterial. All lots are required to comply with clear vision requirements at all intersections.

17.100.230 WATER FACILITIES

Water lines and fire hydrants serving the subdivision or partition, and connecting the development to City mains, shall be installed to provide adequate water pressure to serve present and future consumer demand. The materials, sizes, and locations of water mains, valves, service laterals, meter boxes and other required appurtenances shall be in accordance with the standards of the Fire District, the City, and the State.

If the city requires the subdivider to install water lines in excess of eight inches, the city may participate in the oversizing costs. Any oversizing agreements shall be approved by the city manager based upon council policy and dependent on budget constraints. If required water mains will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement for the proportionate share of the cost.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of water facilities will be reviewed with construction plans. Water service will be provided from a connection to the existing 8-inch water main at the intersection of Cascadia Village Drive and Jacoby Road. There is no other proposed connection to the water distribution system. Per the Public Works Director (Exhibit O), the applicant shall demonstrate that adequate fire and domestic flow will be available with a single point of connection to the water distribution system or connect to the existing 8-inch water line at the intersection of Newton Street and Amherst Street. Per the City Engineer (Exhibit L), all new waterline mains shall be a minimum of 8-inches in diameter and shall be

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

extended to the plat boundaries where practical to provide future connections to adjoining properties. The applicant's proposed Utility Plan (Exhibit C, Sheet 5) depicts new hydrants. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with construction plans.

17.100.240 SANITARY SEWERS

Sanitary sewers shall be installed to serve the subdivision and to connect the subdivision to existing mains. Design of sanitary sewers shall take into account the capacity and grade to allow for desirable extension beyond the subdivision.

If required sewer facilities will directly serve property outside the subdivision, the city may enter into an agreement with the subdivider setting forth methods for reimbursement by nonparticipating landowners for the proportionate share of the cost of construction.

RESPONSE: City facilities have adequate capacity to serve the proposed development. The specific details of sanitary sewer facilities will be reviewed with construction plans. Per the City Engineer (Exhibit L), all new public sanitary sewer lines shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The proposed utility plan shows an 8-inch sanitary sewer line extended south from the existing sewer line in Jacoby Road approximately 490 feet to the site. The site will drain to the existing SE sewer pump station located on Jacoby Road south of Trillium Avenue. Per the Public Works Director (Exhibit O), the applicant shall provide a sanitary sewer main line extension. At the February 25, 2019 Planning Commission hearing, the applicant's representative (Exhibit W) requested that the City reimburse the developer for the cost of the lateral. The Public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The public Works Director and City Engineer (Exhibit X) stated "The City will not reimburse the developer for the cost of the lateral. The applicant may apply for an Advanced Financed Public Improvement reimbursement from the benefitting property owner(s) per section 12.14 of the Municipal Code."

17.100.250 SURFACE DRAINAGE AND STORM SEWER SYSTEM

- A. Drainage facilities shall be provided within the subdivision and to connect with off-site drainage ways or storm sewers. Capacity, grade and materials shall be by a design approved by the city engineer. Design of drainage within the subdivision shall take into account the location, capacity and grade necessary to maintain unrestricted flow from areas draining through the subdivision and to allow extension of the system to serve such areas.
- B. In addition to normal drainage design and construction, provisions shall be taken to handle any drainage from preexisting subsurface drain tile. It shall be the design engineer's duty to investigate the location of drain tile and its relation to public improvements and building construction.
- C. The roof and site drainage from each lot shall be discharged to either curb face outlets (if minor quantity), to a public storm drain or to a natural acceptable drainage way if adjacent to the lot.

RESPONSE: The applicant proposes a 20,283 square foot public stormwater detention pond (Tract A) to be dedicated to the City of Sandy. Detained and treated discharge from the detention pond is proposed to be discharged to the adjacent property to the north. Per the Public Works Director(Exhibit O), the applicant shall demonstrate that the proposed subdivision does not exceed pre-development site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way. Per the City Engineer (Exhibit L), the detention pond shall meet the

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

requirements of the 2016 City of Portland Stormwater Management Manual (SWMM) for landscaping, Section 2.4.1, and escape route, Section 2.30. All new public storm drains shall be a minimum of 12-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. The City Engineer (Exhibit L) states the submitted preliminary stormwater calculations meet the water quality and water quantity criteria as stated in the City of Sandy Municipal Code Chapter 13.18 Standards and the City of Portland current Stormwater Management Manual (SWMM) Standards that were adopted by reference into the Sandy Development Code. Per the City Engineer, the applicant shall submit a detailed final stormwater report stamped by a licensed professional to the City for review and approval with the final construction plans.

17.100.260 UNDERGROUND UTILITIES

All subdivisions or major partitions shall be required to install underground utilities (including, but not limited to, electrical and telephone wiring). The utilities shall be installed pursuant to the requirements of the utility company.

RESPONSE: All utilities shall be placed underground.

17.100.270 SIDEWALKS

Sidewalks shall be installed on both sides of a public street and in any special pedestrian way within the subdivision.

RESPONSE: The applicant proposes constructing sidewalks along all public street frontages, with the exception of the proposed three-quarter street section and one-half street section of Newton Street.

17.100.290 STREET TREES

Where planting strips are provided in the public right-of-way, a master street tree plan shall be submitted and approved by the Director. The street tree plan shall provide street trees approximately every 30' on center for all lots.

RESPONSE: As required by Section 17.92.30, the development of the subdivision requires installation of trees along all street frontages. Street trees are discussed in Section 17.92.30 of this staff report.

17.100.300 EROSION CONTROL

Grass seed planting shall take place prior to September 30th on all lots upon which a dwelling has not been started but the ground cover has been disturbed. The seeds shall be of an annual rye grass variety and shall be sown at not less than four pounds to each 1000 square feet of land area.

RESPONSE: The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a re-vegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code and as detailed below. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

17.100.310 REQUIRED IMPROVEMENTS

The following improvements shall be installed at no expense to the city, consistent with the design standards of Chapter 17.84, except as otherwise provided in relation to oversizing.

- A. Drainage facilities
- B. Lot, street and perimeter monumentation
- C. Mailbox delivery units
- D. Sanitary sewers
- E. Sidewalks
- F. Street lights
- G. Street name signs
- H. Street trees
- I. Streets
- J. Traffic signs
- K. Underground communication lines, including broadband (fiber), telephone, and cable. Franchise agreements will dictate whether telephone and cable lines are required.
- L. Underground power lines
- M. Water distribution lines and fire hydrants

RESPONSE: The applicant shall be responsible for the installation of all improvements detailed in Section 17.100.310, including fiber facilities. SandyNet requires the developer to work with the City to ensure that broadband infrastructure meets the design standards and adopted procedures as described in Section 17.84.70. Entry monument signs shall be located entirely outside the public rightof-way and clear vision areas as required by Section 17.74.30. If entry signs are desired, staff recommends the applicant submit a detailed plan showing the location of such signage.

CHAPTER 17.102 – URBAN FORESTRY

17.102.20 APPLICABILITY

This chapter applies only to properties within the Sandy Urban Growth Boundary that are greater than one acre including contiguous parcels under the same ownership.

- A. General: No person shall cut, harvest, or remove trees 11 inches DBH or greater without first obtaining a permit and demonstrating compliance with this chapter.
 - 1. As a condition of permit issuance, the applicant shall agree to implement required provisions of this chapter and to allow all inspections to be conducted.
 - 2. Tree removal is subject to the provisions of Chapter 15.44, Erosion Control, Chapter 17.56, Hillside Development, and Chapter 17.60 Flood and Slope Hazard.
- B. Exceptions: The following tree removals are exempt from the requirements of this chapter.
 - 1. Tree removal as required by the city or public utility for the installation or maintenance or repair of roads, utilities, or other structures.
 - 2. Tree removal to prevent an imminent threat to public health or safety, or prevent imminent threat to public or private property, or prevent an imminent threat of serious environmental

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

degradation. In these circumstances, a Type I tree removal permit shall be applied for within seven days following the date of tree removal.

RESPONSE: Section 17.00.10 states the overall purpose of the Development Code, which is to ensure that development is consistent with the goals and policies of the Sandy Comprehensive Plan. Goal 5 of the Comprehensive Plan addresses natural resources, with a stated intention of establishing policies for conservation of open space and protection of natural and scenic resources in compliance with Statewide Planning Goal 5. The Comprehensive Plan states, "In keeping with the mandate of Statewide Planning Goal 5, Oregon communities are required to provide programs that will 'conserve open space and protect natural and scenic resources.' Additionally, the City of Sandy has the goal of providing parks, open space, and recreational opportunities for its citizens, as well as promoting healthy and visually attractive environments that are in harmony with the natural landscape character of the area." Goal 5 of the Comprehensive Plan includes policies related to natural resources, open space, stream corridor protection, indigenous growth protection, historic and cultural resource protection, and environmental and cultural resource protection. The most relevant policies related to tree retention and mitigation include the following:

- Policy 4: The City promotes development practices which maintain or enhance the environmental quality enjoyed by the residents of the community.
- Policy 7: Employ development regulations to preserve and protect open space and environmentally sensitive lands, integrate the natural environment of Sandy into project designs, minimize the creation of impervious surface, and discourage the unnecessary clearing of trees and other natural vegetation.
- Policy 16: Require development and construction projects to minimize disturbance of significant indigenous growth and to enhance the cohesive quality of tree stands through sensitive site design and construction methods.
- Policy 17: Promote innovative site and building designs which reduce the adverse impacts of development on native tree stands and other areas of significant vegetation.
- Policy 18: Where feasible, preserve natural vegetation resource sites through public acquisition, conservation easements or other available methods to permanently limit development.
- Policy 19: Encourage and, if necessary, require reforestation in open space areas to help mitigate the adverse impacts of development.
- Policy 27: The City of Sandy shall promote the planting of trees throughout the community on both public and private properties.
- Policy 28: The City of Sandy shall consider the establishment of a street tree plan that identifies particular trees within different areas of the City but that provide species diversity to add visual interest throughout the community, create individual neighborhood identity, and provide additional resistance through species diversity against disease.
- Policy 29: The City of Sandy shall review existing development standards, including street standards, to assure that the standards support and encourage the maintenance of a sustainable urban forest.
- Policy 30: The City of Sandy shall consider incentives to encourage the preservation of significant trees.

As discussed in further detail below, the 9.68 acre site requires retention of a minimum of 29 trees that are in good condition and are 11-inches DBH or greater. Of the 521 trees inventoried on the site, the applicant is proposing to remove 457. Per Section 17.92.10(C), significant trees (8-inches or greater DBH) should be preserved to the greatest extent practicable. In addition, the City of Sandy Comprehensive Plan encourages development that minimizes disturbance of native tree

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

stands, preserves natural open space, and promotes innovative site and building design to minimize the adverse impacts of development on native tree stands.

17.102.50 TREE RETENTION AND PROTECTION REQUIREMENTS

- A. Tree Retention: The landowner is responsible for retention and protection of trees required to be retained as specified below:
 - 1. At least three trees 11 inches DBH or greater are to be retained for every one-acre of contiguous ownership.
 - 2. Retained trees can be located anywhere on the site at the landowner's discretion before the harvest begins. Clusters of trees are encouraged.
 - 3. Trees proposed for retention shall be healthy and likely to grow to maturity, and be located to minimize the potential for blow-down following the harvest.
 - 4. If possible, at least two of the required trees per acre must be of conifer species.
 - 5. Trees within the required protected setback areas may be counted towards the tree retention standard if they meet these requirements.
- **B.** Tree Protection Area: Except as otherwise determined by the Planning Director, all tree protection measures set forth in this section shall be instituted prior to any development activities and removed only after completion of all construction activity. Tree protection measures are required for land disturbing activities including but not limited to tree removal, clearing, grading, excavation, or demolition work.
 - 1. Trees identified for retention shall be marked with yellow flagging tape and protected by protective barrier fencing placed no less than 10 horizontal feet from the outside edge of the trunk.
 - 2. Required fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade.
 - 3. No construction activity shall occur within the tree protection zone, including, but not limited to dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles.
- C. **Inspection.** The applicant shall not proceed with any tree removal or construction activity, except erosion control measures, until the City has inspected and approved the installation of tree protection measures. Within 15 days of the date of accepting an application for a Type I permit, the city shall complete an onsite inspection of proposed activities and issue or deny the permit. Within 15 days of issuing a Type II or Type III permit, the city shall complete an onsite inspection of proposed activities.

For ongoing forest operations, the permit holder shall notify the city by phone or in writing 24 hours prior to subsequent tree removal. The city may conduct an onsite re-inspection of permit conditions at this time.

RESPONSE: The subject property is 9.68 acres, which requires a minimum of 29 retention trees that are 11-inches or greater DBH and in good health. Per the submitted Tree Retention Inventory (Exhibit C, Sheet 2), the applicant proposes to retain all trees within the FSH overlay (64 total), including 33 trees that are viable and 11-inches or greater DBH. Of the 33 viable trees identified for retention, six (6) appear to be on or very close to the eastern property boundary (Trees # 527, 528, 530, 549, 550, and 3006). The applicant's representative submitted a memo (Exhibit W) at the

 $W: \verb|City Hall|Planning|REPORTS|2018| 18-025 \ SUB_VAR_FSH_TREE_INT \ Jacoby \ Heights \ Subdivision \ PC \ continuance \ report. docx \ Subdivision \ PC \ continuance \ report. \ Subdivision \ report. \ Subdivision \ PC \ continuance \ report. \ Subdivision \ PC \ continuance \ report. \ Subdivision \ report. \ Subdivision \ Subdivision \ report. \ Subdivision \ Subdivision \ report. \ Subdivision \ report. \ Subdivision \ report. \ Subdivision \ report. \ Subdivision \ Subdivision \ report. \$

February 25, 2019 Planning Commission hearing that confirmed Tree #3006 is on the property line and Trees # 527, 528, 530, 549, and 550 are all within 5 feet of the property line. However, the roots and canopies of these trees clearly cross the property boundary; thus, staff is concerned that these trees could be impacted by actions occurring on the neighboring property to the east, which is not in City limits. The applicant shall either identify an additional six (6) retention trees on the subject property, or shall obtain a recorded tree conservation easement on the neighboring property to the east for Trees # 527, 528, 530, 549, 550, and 3006. The tree conservation easement shall include the area under the dripline of the tree plus an additional 5 feet beyond the dripline. If the applicant chooses to retain additional trees, staff recommends retaining all trees on Lot 19 and including the area as part of the FSH tract to be owned and maintained by the HOA. Alternatively, staff recommends the applicant consider directional boring in the utility easements to retain Trees #109, 110, 120, 122, 336, 340, 366, 380, 383 and 392, and/or reducing the square footage of Lots 9 and 10 and creating an approximately 20 foot wide tree preservation buffer tract along Jacoby Road, which would also be placed in a conservation easement. A tree buffer along Jacoby Road would also support the applicant's request to have the houses on Lots 9 and 10 face the internal street network rather than Jacoby Road.

Of the 33 viable trees, an additional four (4) include notes such as "poor trunk taper," "windthrow potential," "wound on trunk," or "some trunk decay" (Trees # 520, 531, 534, and 560). At the February 25, 2019 Planning Commission hearing, the applicant's arborist submitted Arborist Notes for Jacoby Heights subdivision (Exhibit V). The notes state that "minor defects noted about 'viable' trees that do not necessarily affect survivability or lead to tree mortality are considered if proximity to construction will lead to additional stress that will affect the tree's ability to compartmentalize or outgrow the defect. The mention of these defects does not mean the trees are hazardous or declining. As trees being preserved are largely part of, and adjacent to, a mature stand of native trees they have a strong likelihood of survival and reduced risk of windthrow." Staff recommends obtaining a second opinion on the health of these four (4) trees and their likelihood of growing to maturity. **The applicant shall obtain a third-party arborist opinion to evaluate Trees # 520, 531, 534, and 560, and shall pay the third-party arborist fee.**

It appears that there are two trees that are proposed for removal located on the FSH boundary line: Tree #510 and Tree #516. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay. Both Tree #510 and Tree #516 are viable per the arborist report.

Section 17.102.50(B.1) requires tree protection fencing be placed no less than 10 horizontal feet from the outside edge of the trunk. Per the Pacific Northwest International Society of Arboriculture (ISA), the ISA defines the critical root zone (CRZ) as "an area equal to a 1-foot radius from the base of the tree's trunk for each 1 inch of the tree's diameter at 4.5 feet above grade (referred to as diameter at breast height)." Often the drip-line is used to estimate a tree's CRZ; however, it should be noted that a tree's roots typically extend well beyond its drip-line. In addition, trees continue to grow, and roots continue to extend. Thus, a proactive approach to tree protection would take into consideration the fact that the tree and its root zone will continue to grow. The arborist notes for Jacoby Heights Subdivision submitted by the applicant's arborist at the February 25, 2019 Planning Commission hearing (Exhibit V) state: "The TPF [tree protection fence] will be installed 5 feet beyond the drip-lines of trees being preserved that border the TPF. Trees being preserved within the core of the FSH will have ample protection." With regards to Trees # 510 and 516, the arborist's notes state: "T510 (Tree 510) and T516 (Tree 516) will be added to the preservation plan. These two

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

trees are at the edge of the FSH. A grading cut will be made to the southwest of these two trees. Grading may encroach into the root protection zones (RPZs) of the two trees but will be no closer than 10 ft. from either tree. Grading will slope away from the trees which preserves more of the root crown than a vertical cut. The project arborist will be onsite to supervise grading that occurs within the RPZs of these trees and to prune roots if needed." As stated by the applicant's arborist, tree protection fencing shall be located at least 5 feet beyond the dripline of all retention trees, with the exception of Trees #510 and #516. For Trees #510 and #516, tree protection fencing shall be placed as close as possible to 5 feet beyond the dripline but may encroach in the critical root zone under supervision of the arborist, who shall be on site to supervise grading that occurs within the critical root zones of Trees # 510 and #516. In no case shall grading or other construction activity occur within 10 feet from the trunk of Tree #510 or Tree #516. Tree protection fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade. Staff recommends 6 foot tall no-jump horse fencing. Staff also recommends the applicant affix a laminated sign (minimum 8.5 inches by 11 inches) to the tree protection fencing indicating that the area behind the fence is a tree retention area and that the fence shall not be removed or relocated. No construction activity shall occur within the tree protection zone, including, but not limited to, dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. The applicant shall request an inspection of tree protection measures prior to any tree removal, grading, or other construction activity on the site.

OTHER CONSIDERATIONS FOR TREES:

To ensure protection of the required retention trees, the applicant shall record a tree protection covenant specifying protection of these trees and limiting removal without submittal of an Arborist's Report and City approval. This document shall include a sketch identifying the required retention trees. All trees marked for retention shall be retained and protected during construction regardless of desired or proposed building plans; plans for future houses on the proposed lots within the subdivision shall be modified to not encroach on retention trees and associated tree protection fencing.

The properties directly north and east of the subject site (tax lots 6100 and 800) contain a number of existing trees, some of which are located close to the shared property line and have canopies that extend onto the subject property. The applicant shall update the plan set to detail tree protection fencing on the subject property located 5 feet beyond the dripline of all trees located on the properties to the north and east.

CHAPTER 17.66 – ADJUSTMENTS AND VARIANCES

17.66.60 VARIANCES

Variances are a means of requesting a complete waiver or major adjustment to certain development standards. They may be requested for a specific lot or as part of a land division application. The Type II variance process is generally reserved for major adjustments on individual lots, while variances to development standards proposed as part of a land division are processed as a Type III application (requiring a public hearing).

RESPONSE: The applicant proposes the following three Type III variances:

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

17.66.70 TYPE II AND TYPE III VARIANCE CRITERIA

The authority to grant a variance does not include authority to approve a development that is designed, arranged or intended for a use not otherwise approvable in the location. The criteria are as follows:

- A. The circumstances necessitating the variance are not of the applicant's making.
- B. The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located.
- C. Granting of the variance will not adversely affect implementation of the Comprehensive Plan.
- D. The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity.
- E. The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land.
- F. Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control.

BLOCK LENGTH VARIANCE

17.100.120 (B) Residential Blocks. "Blocks fronting local streets shall not exceed 400 feet in length, unless topographic, natural resource, or other similar physical conditions justify longer blocks. Blocks may exceed 400 feet if approved as part of a Planned Development, Specific Area Plan, adjustment or variance."

Request: The applicant requests approval to exceed the 400 foot block length standard for the north side of Woodstock Street between Camden Court and the eastern property boundary (approximately 426 feet).

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin. The City Engineer reviewed the future street plan (Exhibit C, Sheet 8) and states "given the wetlands/Tickle Creek is [sic] located to the north, a stubbed street is not feasible. Extending the block length for

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

approximately 100 feet appears to be reasonable and is compatible with the location of the existing house on the adjacent property to the east."

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance. Granting a block length variance will better protect the natural resources on the site.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." **RESPONSE:** Approval of the variance will not have an impact on any of the policies or goals of the Comprehensive Plan or any other applicable policies and standards adopted by the City. The applicant's narrative (Exhibit B) states: "Granting this variance will further the City'[s] goal of protecting natural resources and will not adversely affect implementation of the Comprehensive Plan."

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." *RESPONSE:* The proposed development will not be significantly different than if Woodstock Street was limited to 400 feet in length. Inclusion of a connector trail to the future Tickle Creek Trail on the subject property north of Woodstock Street will break up the block length and provide additional pedestrian connectivity.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The topography of the site is such that there cannot be a north-south street north of Woodstock Street without encroaching in the FSH. By not creating an intersection along the north side of Woodstock Street, the design better protects the FSH area and Tickle Creek basin.

Recommendation: Staff recommends the Planning Commission approve a variance to allow the north side of Woodstock Street between Camden Court and the eastern property boundary to exceed 400 feet as proposed in the submitted plan set.

BUILDING ORIENTATION VARIANCES

The requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located.

Request: The applicant requests a Type III variance to Chapter 17.82.20 to allow the front door of the future homes constructed on Lots 9, 10, 30 and 32 to face the internal local street network instead of Jacoby Road, a designated transit street.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The Development Code does not allow driveway access to higher classification streets such as Jacoby Road, a collector street. The front door could be oriented to Jacoby Road with a rear or side loaded garage oriented to the internal street network; however, staff recognizes that the front doors on Jacoby Road would essentially be false front doors, which is not the intern of the code.

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." *RESPONSE:* The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity. Orientation of the houses on Lots 9, 10, 30, and 32 to the internal street network will have no negative effect on surrounding properties. Per the applicant's narrative (Exhibit B), "Homes directly across Jacoby Road from the subject property have a similar configuration to what is being proposed."

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." *RESPONSE:* Orienting the houses on Lots 9, 10, 30, and 32 to the local street will allow construction of a traditional structure containing a garage and front door on the same street elevation, which will create a consistent look in the neighborhood. However, the requirement of building entrances oriented to transit streets, such as Jacoby Road, is to provide a pleasant and enjoyable pedestrian experience by connecting activities within a structure to the adjacent sidewalk where transit amenities are located. If Planning Commission approves this variance, staff recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30 and 32 per the following:

- Builders of individual lots shall construct a decorative fence on the Jacoby Road facing side of Lots 9, 10, 30, and 32 to enhance the visual appeal of these lots from Jacoby Road. The fence shall include the following design details:
 - *Constructed of vertical black metal or faux metal fencing material.*
 - No greater than 3-inch gap between vertical pickets.
 - o 4-feet to 6-feet in height.
- The rear elevations of Lot 30 and the side elevation of Lots 9, 10, and 32 shall incorporate all of the following design details where applicable:
 - Decorative gables including two or more of the following:
 - A window with grids.
 - A trimmed vent. The trim must match the trim on the windows and the vent must be at least 4 square feet in area.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

- Cross or diagonal bracing, shingles, trim, corbels, exposed rafter ends, or brackets.
- Decorative 'belly-band' with an alternative paint color to the siding color, between building floors.
- *Mixture of siding materials, including shake or horizontal lap siding with an alternative paint color to the primary siding color.*
- Recessed or covered rear entries (rear elevation only).
 - The covered area must be at least 48 square feet and a minimum of 8 feet wide.
 - The recessed entry must feature vertical support posts.
- o Minimum four-inch wide trim or 12-inch wide shutters around all windows.

Builders shall submit proposed decorative fence and elevation designs for staff review and approval.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* It would create an unconventional design to have the rear of the houses on Lots 30 and 32 facing the proposed Camden Court cul-de-sac. While the front doors of the houses on Lots 9 and 10 could face Jacoby Road, the garages would not be able to gain access from Jacoby Road. A more traditional orientation would be to have both the front door and garage face the local street as requested by the applicant.

Recommendation: Staff recommends the Planning Commission approve a variance to allow the front door of the future homes constructed on Lots 9, 10, 30, and 32 to not face Jacoby Road. Staff further recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30, and 32.

TWO-TIERED LOT VARIANCE

17.100.120 (A) Blocks. "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

Request: The applicant requests approval to provide a one-tier lot configuration for Lot 30 between Jacoby Road and Camden Court, instead of the required two-tier configuration.

Criteria A. of Section 17.66.70 states "The circumstances necessitating the variance are not of the applicant's making." **RESPONSE:** The applicant's narrative (Exhibit B) states that the design of the proposed subdivision, including the single-tiered lot on the cul-de-sac, is due to "topographic and natural resource constraints of the property associated with the location of Tickle Creek and steep slopes." The proposal only includes one single-tiered lot. In addition, the Development Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A), which states: "Blocks shall have sufficient width to provide for two tiers of lots at appropriate depths. However, exceptions to the block width shall be allowed for blocks that are adjacent to arterial streets or natural features."

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

Criteria B. of Section 17.66.70 states "The hardship does not arise from a violation of this Code, and approval will not allow otherwise prohibited uses in the district in which the property is located." **RESPONSE:** The applicant has not violated the Development Code and the uses allowed on the lots will be the same with or without approval of this variance.

Criteria C. of Section 17.66.70 states "Granting of the variance will not adversely affect implementation of the Comprehensive Plan." *RESPONSE:* The variance will not have an impact on any of the policies or goals of the Comprehensive Plan.

Criteria D. of Section 17.66.70 states "The variance authorized will not be materially detrimental to the public welfare or materially injurious to other property in the vicinity." **RESPONSE:** Approval of the variance will not be materially detrimental or injurious to other property owners in the vicinity.

Criteria E. of Section 17.66.70 states "The development will be the same as development permitted under this code and City standards to the greatest extent that is reasonably possible while permitting some economic use of the land." **RESPONSE:** The proposal contains one single-tiered lot (Lot 30) and the proposed development will not be significantly different than if the applicant proposed a two-tiered lot configuration so that Lot 30 was somehow not oriented as a single tiered lot.

Criteria F. of Section 17.66.70 states "Special circumstances or conditions apply to the property which do not apply generally to other properties in the same zone or vicinity, and result from lot size or shape (legally existing prior to the effective date of this Code), topography, or other circumstances over which the applicant has no control." *RESPONSE:* The applicant's narrative states "the subject property contains unique circumstances (collector street, steep slopes, and creek) that warrant approval of an alternative design." The Code allows an exception for blocks that are adjacent to arterial streets or natural features. Though Jacoby Road is a collector street, staff believes approving the applicant's request to allow one-tiered lot configuration would be in line with the intent of Section 17.100.120(A) as explained in Criteria A.

Recommendation: Staff recommends the Planning Commission approve the applicant's request to allow one-tiered lot configuration for Lot 30 between Jacoby Road and Camden Court.

CHAPTER 15.30 - DARK SKY ORDINANCE

All exterior lighting is required to conform to the requirements of this chapter.

RESPONSE: Chapter 15.30 contains the City of Sandy's Dark Sky Ordinance. The applicant will need to install street lights along all street frontages wherever street lighting is determined insufficient. The locations of these fixtures shall be reviewed in detail with construction plans. Full cut-off lighting shall be required.

CHAPTER 15.44 - EROSION CONTROL

15.44.20 AREA OF APPLICATION

This chapter applies to all ground disturbing activities within the city limits whether or not a permit is

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

required, unless such activities are otherwise exempted by the Sandy Development Code. All nonpermitted ground disturbing activities that are permanent or temporary in nature shall comply with this chapter unless otherwise noted.

RESPONSE: All on-site earthwork activities including any retaining wall construction should follow the requirements of the current edition of the Oregon Structural Specialty Code (OSSC). If the proposal includes a retaining wall, the applicant shall submit additional details on the proposed retaining wall for staff review and approval. All on-site earthwork activities should follow the Geotechnical Report (Exhibit G) prepared by Rapid Soil Solutions LLC, dated February 23, 2018. Any deviations from the initial design parameters shall be reviewed by Rapid Soil Solutions for applicability. The City Engineer (Exhibit L) reviewed the Geotechnical Report and recommends that the applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the above noted report and OSSC. Site grading should not in any way impede, impound or inundate the adjoining properties. All the work within the public right-of-way and within the paved area should comply with American Public Works Association (APWA) and City requirements as amended. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a revegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.

15.44.50 APPLICANT'S RESPONSIBILITY

The developer, subsequent developers or property owners shall bear responsibility for the installation, construction, protection of all stockpiles on the site, materials transported from the site, maintenance including re-vegetation of all graded areas, inspection and disposition of erosion control measures. In order to meet the purpose of this chapter, the developer shall:

- C. Apply permanent or temporary soil stabilization to denuded development site areas in conformance with the following schedule:
 - 1. Between October 1 and April 30, all denuded sites shall immediately be provided with either temporary or permanent soil stabilization.
 - 2. Between May 1 and September 30, temporary erosion and sediment control measures to reduce dust and sediment transport shall be applied as soon as practicable, but in no case more than seven days after ground disturbing activity occurs.
 - 3. Ground cover shall be installed on any portion of a site that is denuded for more than six months. Sports fields or playgrounds surrounded by vegetative cover or permanently installed curbing are exempt from this requirement.
 - 4. Temporary measures shall be maintained until permanent measures are established.
 - 5. Plant replacement vegetative cover that does not include plants listed as nuisance or prohibited plants on the City of Portland Plant List (available in the Planning & Development

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

office) or the City's Native Plant List. Agriculture, timber production or residential crop growing activities are exempted from this requirement.

- 6. Secure or protect soil stockpiles throughout the project with temporary or permanent soil stabilization measures. Depositions of soil may be subject to additional regulations requiring permit, review or erosion and sediment control.
- 7. Select BMPs from the *Erosion Control Manual*.

RESPONSE: Section 15.44.50 contains requirements for maintenance of a site including revegetation of all graded areas. The applicant's Erosion Control Plan shall be designed in accordance with the standards of Section 15.44.50.

III. SUMMARY AND CONCLUSION

All County Surveyors & Planners submitted an application on behalf of Cory Knight to subdivide 9.68 acres at 19124 Jacoby Road into a 32 lot subdivision. The 32 lots range in size from 7,500 square feet to 21,183 square feet. The subdivision proposal includes the extension of Newton Street into the site and three proposed local streets: Camden Court, Woodstock Street, and Middlebury Avenue. The applicant is requesting three (3) variances: a variance to Section 17.82.20 to allow future homes to not orient to Jacoby Road, a variance to Section 17.100.120(A) to allow a one-tiered lot configuration between Jacoby Road and Camden Court, and a variance to Section 17.100.120(B) to exceed a 400 foot block length on the north side of Woodstock Street between Camden Court and the east property line. In addition, the applicant is requesting a code interpretation related to the tree protection standards of Chapter 17.92, Landscaping and Screening General Standards – All Zones, and Chapter 17.102, Urban Forestry.

As detailed in this report, the proposed subdivision generally complies with most Development Code requirements.

Biggest Issues/Staff Recommendations:

- Create a separate FSH tract, which will be owned and maintained by an HOA. Approve variance to allow Lots 19-21 to be less than 7,500 square feet and allow rear setback to be reduced to 15 feet.
- Add additional decorative fence and building features on Jacoby Road for Lots 9, 10, 30, and 32.
- Provide a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay.
- Design Newton Street with a three-quarter street section to the eastern property boundary. Provide the minimum three-quarter street improvements on Newton Street on the subject property or acquire land on adjacent properties to accommodate a three-quarter street.
- Identify an additional six (6) retention trees on the subject property, or obtain a recorded tree conservation easement on the neighboring property to the east for Trees # 527, 528, 530, 549, 550, and 3006. The tree conservation easement shall include the area under the dripline of the tree plus an additional 5 feet beyond the dripline.
- Obtain a second opinion from the third-party arborist to evaluate Trees # 520, 531, 534, and 560, and pay the third-party arborist fee.

IV. RECOMMENDATION

Staff recommends the Planning Commission hold a public hearing to take public testimony regarding the proposal. Staff recommends the Planning Commission **approve** the subdivision application with modifications as recommended in this report. The staff recommendation includes approval of the following three variances:

- Variance to Section 17.82.20 to have the front door for the houses along Jacoby Road face the interior local street network instead of Jacoby Road, which is designated as a transit street;
- Variance to Section 17.100.120(A) to provide a one-tiered lot configuration rather than a twotiered lot configuration along Jacoby Road;
- Variance to Section 17.100.120(B) to increase the maximum block length beyond 400 feet for the north side of the proposed Woodstock Street between Camden Court and the east property boundary.

Staff is also recommending that the FSH analysis area be a separate tract owned and maintained by an HOA. This would result in Lots 19-21 being less than the minimum 7,500 square feet; thus, staff is recommending approval of a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum and allowing the rear setback to be reduced to 15 feet.

Since the Planning Commission decided to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation decision.

W:\City Hall\Planning\REPORTS\2018\18-025 SUB_VAR_FSH_TREE_INT Jacoby Heights Subdivision PC continuance report.docx

18-025 SUB/VAR/FSH/TREE Jacoby Heights Subdivision

Recommendations and Conditions

Note: This is a list of all recommendations and conditions in the staff report in the order in which they appear. This list contains duplicates.

- 1. The applicant shall obtain a demolition permit from the City of Sandy Building Department prior to demolition of the existing structure(s) on-site.
- Any existing domestic or irrigation wells on site shall be located, identified, capped, disconnected or abandoned in conformance with OAR 690-220-0030. A copy of the Oregon Water Resources Department (OWRD) abandonment certificate shall be submitted to the City Planning Division.
- 3. Any on-site sewage disposal system shall be abandoned in conformance with Clackamas County Water Environmental Services (WES) regulations. A copy of the septic tank removal certificate shall be submitted to the City Planning Division.
- 4. Prior to submitting construction plans or building permits, including grading and erosion control permits, the applicant shall update the plan set and associated documents based on the conditions of approval determined by the Planning Commission and shall submit a full set of the updated plans to Planning Division staff for review and approval.
- Since the Planning Commission decides to interpret the code such that one specific subsection of Chapter 17.92 does not apply to residential subdivisions, staff requests that the Planning Commission provide a revised motion with a more clear basis for the code interpretation decision.
- 6. All homes shall provide building design features in compliance with the standards in Section 17.90.150 and Chapter 17.54.
- 7. All homes shall meet the development standards of Section 17.34.30.
- 8. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.
- The applicant shall update the cut and fill evaluation to include the location of the tree protection fencing as depicted on the existing conditions and tree retention inventory (Exhibit C, Sheet 2). The applicant shall not cut or fill within the proposed tree protection area.
- 10. The applicant shall follow the recommendations outlined in the Geotechnical Report.
- 11. The applicant shall submit a geological assessment specific to each lot proposed to be developed with a single family house on 25 percent or greater slopes that fall outside the mapped FSH overlay. This includes Lots 19, 20, 21, 27, 28, and 29. Additional reports may be

required depending on the recommendations of the geological assessment. Once the subdivision is platted and building permits are filed for individual home construction the property owner/contractor shall submit Type I Hillside applications with studies as detailed in Table 1 in Section 17.56.40.

- 12. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on all lots (Lots 1-32) and all of the work is in compliance with the above noted report and OSSC.
- 13. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants.
- 14. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot area and allowing the rear setback to be reduced to 15 feet.
- 15. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.
- All structures on lots along Jacoby Road shall be constructed to comply with the standards of Section 17.80.20.
- 17. Public improvement plans are subject to a separate review and approval process. Preliminary plat approval does not connote approval of public improvement construction plans.
- 18. The applicant shall construct sidewalks along Tract A both on Jacoby Road and Camden Court, prior to final plat approval. The sidewalks on Camden Court, Woodstock Street, Newton Street, and Middlebury Avenue shall be five feet in width and separated by a five foot planter strip (or 6 foot swale) in areas not transverse by driveways. The sidewalks on Jacoby Road shall be six feet in width and separated by a five foot planter strip.
- 19. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a three-quarter street.
- 20. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. Street trees shall be installed approximately 30 feet on center in conjunction with issuance of building permits. Street trees are required to be a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted and staked per the City of Sandy standard planting detail; trees shall be tied to the stakes with loosely tied twine. The planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed within one year of installation.

- 21. The applicant shall construct a mid-block ADA ramp on the west side of Jacoby Road opposite Woodstock Street to provide ADA access to the existing sidewalk on the west side of Jacoby Road and the neighborhood to the west.
- 22. The applicant shall install one bench (Fairweather model PL-3, powder-coated RAL6028) with a concrete pad to City specifications. The applicant shall contact the Transit Director for specific location, amenity information, and pad engineering specifications.
- 23. The applicant shall dedicate 10 feet of right-of-way along the Jacoby Road frontage of the site.
- 24. The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code.
- 25. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. Per the City Engineer (Exhibit L), the three-quarter street improvements on Newton Street shall include a 28-foot wide paved surface, curbs on both sides, 5-foot planter strips and 5-foot sidewalks on the north side of the roadway.
- 26. The applicant shall provide the minimum three-quarter street improvements on Newton Street on the subject property or shall acquire land on adjacent properties to accommodate a threequarter street.
- 27. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW dedication for staff review and approval.
- 28. The applicant shall plat a vehicle non-access reserve (VNAR) strip at the east end of Woodstock Street and along the south side of the right-of-way of Newton Street, with the exception of the easternmost end of the south side of Newton Street to allow access to the newly constructed street from the existing gravel drive; the length of the VNAR for the south side of Newton Street shall be determined during Construction Plan review.
- 29. The applicant shall work with the Fire Marshall to determine if a temporary turn-around is needed on Woodstock Street and/or Newton Street, or if houses on particular lots need to be sprinkled.
- 30. The applicant shall extend the street improvements on Woodstock Street to the east line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code.
- 31. The City of Sandy reserves the right to name streets.
- 32. The applicant shall revise the alignment of Newton Street to conform to the 50 feet of tangent alignment standard.
- 33. All public utility installations shall conform to the City's facilities master plans. Staff recommends the applicant revise the utility plan to include broadband fiber locations as detailed by the SandyNet Manager. Per the City Engineer (Exhibit L), all public sanitary sewer and waterline

mains shall be a minimum of 8 inches in diameter and all stormwater drains shall be a minimum of 12 inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties. No building permits will be issued until all public utilities including sanitary sewer are available to serve the development. The applicant shall pay plan review, inspection, and permit fees as determined by the Public Works Director.

- 34. All franchise utilities shall be installed underground and in conformance with City standards.
- 35. The applicant shall call the PGE Service Coordinators at 503-323-6700 when the developer is ready to start the project.
- 36. The applicant shall designate the FSH area as a separate tract on the plat, which shall be owned and maintained by an HOA.
- 37. The Plat shall detail the following easements:
 - An eight-foot wide public utility easement (PUE) along the frontage of all proposed lots;
 - Crossover easements along the common parcel line recorded between pairs of lots that share driveways (Lots 26 and 27, and Lots 28 and 29);
 - 20 foot easement along the west boundary of Lots 9 and 10 for public sanitary sewer and public storm line. Per the Public Works Director (Exhibit O), the City reserves the right to widen this easement depending on the final design depth of the sanitary sewer and storm lines per Section 17.84.90(A.2);
 - A 15 foot easement between Lot 30 and Tract A for public sanitary sewer;
 - A 10 foot private storm drain easement at the rear of Lots 1-18, 22-25, 28, 29, 31, and 32 and a 10 foot private storm drain easement across Lots 19-21 just south of the FSH boundary;
 - A blanket conservation easement and pedestrian easement over the entirety of the FSH tract;
 - An 8 foot wide pedestrian maintenance access easement to the FSH tract (could be located on the east edge of Lot 19 or elsewhere); and
 - A vehicle non-access reserve (VNAR) strip in the following locations:
 - Jacoby Road frontage of Lots 9, 10, 30 and 32 and Tract A; and,
 - South side of the right-of-way of Newton Street, with the exception of the easternmost section as determined necessary to allow access to the newly constructed Newton Street from the existing gravel driveway during Construction Plan review; and,
 - East end of the right-of-way of Woodstock Street.
- 38. Mail delivery facilities shall be provided by the applicant in conformance with Section 17.84.100 and the standards of the USPS. The applicant shall submit a mail delivery plan, featuring grouped lockable mail facilities, to the City and USPS for review and approval prior to installation of mailboxes.

- 39. For these reasons, the applicant shall pay a fee in lieu for the required parkland dedication per the adopted Fee Resolution.
- 40. Based on the current Fee Resolution, the applicant would be required to pay a fee in lieu of dedication for a total of \$98,810 (0.41 acres of land to be dedicated x \$241,000).
- 41. Currently, the Fee Resolution requires payment of \$108,650 if a portion of the fee is deferred, a minimum of 50 percent (\$54,325) paid prior to final plat approval and the remaining 50 percent (\$54,325) divided between the 32 lots (\$1,697.66/lot).
- 42. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract.
- 43. Staff recommends supporting a variance to allow Lots 19-21 to be less than the 7,500 square foot minimum required lot area and allowing the rear setback to be reduced to 15 feet.
- 44. The applicant shall place the FSH analysis area in a separate tract, which shall be owned and maintained by a Homeowner's Association (HOA). The applicant shall place a blanket conservation easement and pedestrian easement over the entirety of the FSH tract. The HOA shall be responsible for maintaining the FSH tract in an ecologically functional state, including but not limited to removal of all invasive plants and replanting any areas of exposed soil with native plants. In the event that the HOA fails to maintain the FSH tract according to the standards of the Sandy Municipal Code, the City of Sandy, following reasonable notice, may demand that the deficiency of maintenance be corrected, and may enter the FSH tract area for maintenance purposes. All costs thereby incurred by the City shall be charged to the HOA.
- 45. All landscaping shall be continually maintained, including necessary watering, weeding, pruning, and replacing.
- 46. The applicant shall update the plan set to remove street trees from the driveways in front of Lots 31 and 32; at least one of the street trees shall be relocated outside of the driveway area.
- 47. The applicant shall aerate the planter strips to a depth of 3 feet prior to planting street trees. The applicant shall either aerate the planter strip soil at the subdivision stage and install fencing around the planter strips to protect the soil from compaction or shall aerate the soil at the individual home construction phase. The applicant shall call for an inspection with the City after aerating the soil and before planting the street trees.
- 48. The applicant shall extend the street improvements on Newton Street across from Lot 9 to the south line of the development site and obtain slope easements or construct retaining walls as necessary to comply with this section of the development code. Improvements shall include extension of the Newton Street sidewalk and planter strip, one additional street tree, and landscaping the additional 17.70 feet of right-of-way dedication at the southeast corner of Newton Street and Jacoby Road. The applicant shall submit a landscaping plan for the 17.70 feet of ROW dedication for staff review and approval.
- 49. The applicant shall eliminate the proposed curve at the east end of Newton and continue the three-quarter street improvement to the east boundary of the site. The applicant shall update the plan set to show street trees in the planter strip adjacent to Lots 1 and 2. If the plans change in a way that affects the number of street trees (e.g., driveway locations), the applicant shall submit an updated street tree plan for staff review and approval. Street trees are required to be

a minimum caliper of 1.5-inches measured 6 inches from grade. Trees shall be planted, staked, and the planter strip shall be graded and backfilled as necessary, and bark mulch, vegetation, or other approved material installed prior to occupancy. Tree ties shall be removed after one growing season (or a maximum of 1 year).

- 50. As required by Section 17.92.140, the developer and lot owners shall be required to maintain all vegetation planted in the development for two years from the date of completion, and shall replace any dead or dying plants during that period.
- 51. All street trees shall be a minimum of 1.5-inches in caliper measured 6 inches above the ground and shall be planted per the City of Sandy standard planting detail.
- 52. The applicant shall update the plan set to detail a minimum of two (2) different tree species per block face for staff review and approval.
- 53. The applicant shall update the proposed street tree list with at least two additional species from the City of Sandy Street Tree List that aren't maples or ashes.
- 54. The applicant shall maintain all unlandscaped and/or revegetated areas for a period of two years following the date of recording of the final plat associated with those improvements.
- 55. If the applicant chooses to postpone street tree and/or landscaping installation, the applicant shall post a performance bond equal to 120 percent of the cost of the street trees/landscaping, assuring installation within 6 months. The cost of street trees shall be based on the street tree plan and at least \$500 per tree fee, including a three-year maintenance and warranty period. The cost of landscaping shall be based on the average of three estimates from three landscaping contractors; the estimates shall include as separate items all materials, labor, and other costs of the required action, including a two-year maintenance and warranty period.
- 56. The applicant shall comply with the parking standards in Section 17.98.60.
- 57. Lot 28 and 29 shall share a driveway approach maximum 24 feet in width. Lot 26 and 27 shall share a driveway approach maximum 24 feet in width. The applicant shall grant crossover easements along the common parcel line to lots with shared driveway approaches.
- 58. The applicant shall realign the access drive proposed to serve the stormwater facility in Tract A or revise the west line of Lot 29 to eliminate remnant of Tract A between the access drive and Lot 29.
- 59. The driveway approach for Tract A shall be a maximum of 12 feet in width. Per the City Engineer (Exhibit L), the access to the detention pond shall be paved or all-weather surface.
- 60. All parking, driveway and maneuvering areas shall be constructed of asphalt, concrete, or other approved material.
- 61. The applicant shall revise and resubmit the on-street parking plan for staff review and approval prior to final plat approval.
- 62. The applicant shall revise the Parking Analysis if required fire hydrants affect on-street parking spaces.
- 63. Preliminary plat approval does not connote utility or public improvement plan approval which will be reviewed and approved separately upon submittal of public improvement construction plans.
- 64. The applicant shall provide a profile design for a minimum of 200 feet for all future street extensions beyond the project boundary to ensure future street grades can be met.

- 65. The minimum paved cul-de-sac radius shall be 42 feet; a review by the Fire Department to confirm cul-de-sac size compliance shall be required prior to approval of the construction plans.
- 66. The locations of street light fixtures shall be reviewed in detail with construction plans.
- 67. The applicant shall designate the FSH analysis area as a separate tract, which shall be owned and maintained by an HOA.
- 68. The applicant shall demonstrate that adequate fire and domestic flow will be available with a single point of connection to the water distribution system or connect to the existing 8-inch water line at the intersection of Newton Street and Amherst Street. Per the City Engineer (Exhibit L), all new waterline mains shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties.
- 69. The location of fire hydrants will be reviewed by the Sandy Fire Department in more detail with construction plans.
- 70. All new public sanitary sewer lines shall be a minimum of 8-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties.
- 71. The applicant shall provide a sanitary sewer lateral serving Tax Lot 6100 (located directly north of the subject property) with the sanitary sewer main line extension.
- 72. The applicant shall demonstrate that the proposed subdivision does not exceed predevelopment site runoff discharges to this same point and provide information on the dimensions and slope of the existing drainage way. Per the City Engineer (Exhibit L), the detention pond shall meet the requirements of the 2016 City of Portland Stormwater Management Manual (SWMM) for landscaping, Section 2.4.1, and escape route, Section 2.30.
- 73. All new public storm drains shall be a minimum of 12-inches in diameter and shall be extended to the plat boundaries where practical to provide future connections to adjoining properties.
- 74. The applicant shall submit a detailed final stormwater report stamped by a licensed professional to the City for review and approval with the final construction plans.
- 75. All utilities shall be placed underground.
- 76. The applicant shall submit a grading and erosion control permit and request an inspection of installed devices prior to any additional grading onsite. The grading and erosion control plan shall include a re-vegetation plan for all areas disturbed during construction of the subdivision. All erosion control and grading shall comply with Section 15.44 of the Municipal Code and as detailed below. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.
- 77. The applicant shall be responsible for the installation of all improvements detailed in Section 17.100.310, including fiber facilities. SandyNet requires the developer to work with the City to ensure that broadband infrastructure meets the design standards and adopted procedures as described in Section 17.84.70.
- 78. If entry signs are desired, staff recommends the applicant submit a detailed plan showing the location of such signage.

- 79. The applicant shall either identify an additional six (6) retention trees on the subject property, or shall obtain a recorded tree conservation easement on the neighboring property to the east for Trees # 527, 528, 530, 549, 550, and 3006. The tree conservation easement shall include the area under the dripline of the tree plus an additional 5 feet beyond the dripline.
- The applicant shall obtain a third-party arborist opinion to evaluate Trees # 520, 531, 534, and 560, and shall pay the third-party arborist fee.
- 81. The applicant shall update the plan set to retain Trees #510 and #516 to avoid tree removal in the FSH overlay.
- 82. Tree protection fencing shall be located at least 5 feet beyond the dripline of all retention trees, with the exception of Trees #510 and #516. For Trees #510 and #516, tree protection fencing shall be placed as close as possible to 5 feet beyond the dripline but may encroach in the critical root zone under supervision of the arborist, who shall be on site to supervise grading that occurs within the critical root zones of Trees # 510 and #516. In no case shall grading or other construction activity occur within 10 feet from the trunk of Tree #510 or Tree #516. Tree protection fencing shall be a minimum of six feet tall supported with metal posts placed no farther than ten feet apart installed flush with the initial undisturbed grade. Staff recommends 6 foot tall no-jump horse fencing. Staff also recommends the applicant affix a laminated sign (minimum 8.5 inches by 11 inches) to the tree protection zone, including, but not limited to, dumping or storage of materials such as building supplies, soil, waste items, equipment, or parked vehicles. The applicant shall request an inspection of tree protection measures prior to any tree removal, grading, or other construction activity on the site.
- 83. To ensure protection of the required retention trees, the applicant shall record a tree protection covenant specifying protection of these trees and limiting removal without submittal of an Arborist's Report and City approval. This document shall include a sketch identifying the required retention trees. All trees marked for retention shall be retained and protected during construction regardless of desired or proposed building plans; plans for future houses on the proposed lots within the subdivision shall be modified to not encroach on retention trees and associated tree protection fencing.
- 84. The applicant shall update the plan set to detail tree protection fencing on the subject property located 5 feet beyond the dripline of all trees located on the properties to the north and east.
- 85. Staff recommends the Planning Commission approve a variance to allow the north side of Woodstock Street between Camden Court and the eastern property boundary to exceed 400 feet as proposed in the submitted plan set.
- 86. Staff recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30 and 32 per the following:
 - Builders of individual lots shall construct a decorative fence on the Jacoby Road facing side of Lots 9, 10, 30, and 32 to enhance the visual appeal of these lots from Jacoby Road. The fence shall include the following design details:
 - o Constructed of vertical black metal or faux metal fencing material.
 - \circ $\,$ No greater than 3-inch gap between vertical pickets.

- 4-feet to 6-feet in height.
- The rear elevations of Lot 30 and the side elevation of Lots 9, 10, and 32 shall incorporate all of the following design details where applicable:
 - Decorative gables including two or more of the following:
 - A window with grids.
 - A trimmed vent. The trim must match the trim on the windows and the vent must be at least 4 square feet in area.
 - Cross or diagonal bracing, shingles, trim, corbels, exposed rafter ends, or brackets.
 - Decorative 'belly-band' with an alternative paint color to the siding color, between building floors.
 - Mixture of siding materials, including shake or horizontal lap siding with an alternative paint color to the primary siding color.
 - o Recessed or covered rear entries (rear elevation only).
 - The covered area must be at least 48 square feet and a minimum of 8 feet wide.
 - The recessed entry must feature vertical support posts.
 - Minimum four-inch wide trim or 12-inch wide shutters around all windows.
- 87. Builders shall submit proposed decorative fence and elevation designs for staff review and approval.
- 88. Staff recommends the Planning Commission approve a variance to allow the front door of the future homes constructed on Lots 9, 10, 30, and 32 to not face Jacoby Road. Staff further recommends the applicant add additional design elements and decorative fencing along the Jacoby Road facing sides of Lots 9, 10, 30, and 32.
- 89. Staff recommends the Planning Commission approve the applicant's request to allow one-tiered lot configuration for Lot 30 between Jacoby Road and Camden Court.
- 90. The locations of these fixtures shall be reviewed in detail with construction plans. Full cut-off lighting shall be required.
- 91. All on-site earthwork activities including any retaining wall construction should follow the requirements of the current edition of the Oregon Structural Specialty Code (OSSC). If the proposal includes a retaining wall, the applicant shall submit additional details on the proposed retaining wall for staff review and approval.
- 92. The applicant shall retain appropriate professional geotechnical services for observation of construction of earthwork and grading activities. The grading setbacks, drainage, and terracing shall comply with the Oregon Structural Specialty Code (OSSC) requirements and the geotechnical report recommendations and conclusions as indicated in the report. When the grading is completed, the applicant shall submit a final report by the Geotechnical Engineer to the City stating that adequate inspections and testing have been performed on the lots and all of the work is in compliance with the above noted report and OSSC. Site grading should not in any way impede, impound or inundate the adjoining properties. All the work within the public right-of-way and within the paved area should comply with American Public Works Association (APWA) and City requirements as amended. The applicant shall submit a grading and erosion

control permit and request an inspection of installed devices prior to any additional grading onsite.

- 93. All erosion control and grading shall comply with Section 15.44 of the Municipal Code. The proposed subdivision is greater than one acre which typically requires approval of a DEQ 1200-C Permit. The applicant shall submit confirmation from DEQ if a 1200-C Permit will not be required.
- 94. The applicant's Erosion Control Plan shall be designed in accordance with the standards of Section 15.44.50.